

Not Nothing

Nonemptiness in Team Semantics





INSTITUTE FOR LOGIC, LANGUAGE AND COMPUTATION



Aleksi Anttila

Not Nothing

Nonemptiness in Team Semantics

Aleksi Anttila

Not Nothing

Nonemptiness in Team Semantics

ILLC Dissertation Series DS-2025-04



INSTITUTE FOR LOGIC, LANGUAGE AND COMPUTATION

For further information about ILLC-publications, please contact

Institute for Logic, Language and Computation Universiteit van Amsterdam Science Park 107 1098 XG Amsterdam phone: +31-20-525 6051 e-mail: illc@uva.nl homepage: http://www.illc.uva.nl/

The research for this doctoral thesis was supported in part by the Academy of Finland (grant 336283) and by the European Research Council (ERC) under the European Union's Horizon 2020 research and innovation programme (Grant Agreement No. 101020762).

Copyright © 2025 by Aleksi Anttila

Cover and bookmark photos by Aleksi Anttila. Back cover illustration by Fernanda Sousa-Duarte. Design by Aleksi Anttila and Fernanda Sousa-Duarte. Printed and bound by Ipskamp Printing.

ISBN: 978-94-6473-737-0

Not Nothing Nonemptiness in Team Semantics

ACADEMISCH PROEFSCHRIFT

ter verkrijging van de graad van doctor aan de Universiteit van Amsterdam op gezag van de Rector Magnificus prof. dr. ir. P.P.C.C. Verbeek ten overstaan van een door het College voor Promoties ingestelde commissie, in het openbaar te verdedigen in de Agnietenkapel op dinsdag 1 april 2025, te 16.00 uur

> door Aleksi Ilari Anttila geboren te Helsinki

Promotiecommissie

Promotores:

Copromotores:

Overige leden:

dr. F. Yang prof. dr. R. Iemhoff dr. A. Baltag prof. dr. J. Kontinen dr. N. Bezhanishvili dr. I.A. Ciardelli prof. dr. F. Roelofsen dr. A. Özgün

dr. M.D. Aloni

prof. dr. ing. R.A.M. van Rooij

Universiteit van Amsterdam Universiteit van Amsterdam

Universiteit Utrecht

Universiteit Utrecht Universiteit van Amsterdam University of Helsinki Universiteit van Amsterdam Universiteit van Amsterdam Universiteit van Amsterdam

Faculteit der Natuurwetenschappen, Wiskunde en Informatica

Lumille ja Naolle

Contents

| Ac | Acknowledgments xi | | | | |
|----|--------------------|---|----|--|--|
| Cł | napter | r Sources and Author Contributions | XV | | |
| 1 | Ove | rview | 1 | | |
| 2 | Stat | e-based Modal Logics for Free Choice | 11 | | |
| | 2.1 | Introduction | 11 | | |
| | 2.2 | Bilateral State-Based Modal Logic | 15 | | |
| | 2.3 | Expressive Power and Normal Forms | 20 | | |
| | | 2.3.1 Bisimulation | 21 | | |
| | | 2.3.2 Expressive Completeness | 24 | | |
| | 2.4 | Axiomatizations | 31 | | |
| | | 2.4.1 $BSML^{\vee}$ | 31 | | |
| | | 2.4.2 $BSML^{\circ}$ | 45 | | |
| | | 2.4.3 BSML | 53 | | |
| | 2.5 | Conclusion | 60 | | |
| 3 | Con | vex Propositional and Modal Team Logics | 63 | | |
| | 3.1 | Introduction | 63 | | |
| | 3.2 | Convex Properties | 67 | | |
| | | 3.2.1 Propositional Properties | 67 | | |
| | | 3.2.2 Modal Properties | 77 | | |
| | 3.3 | Convex and Union-closed Properties | 79 | | |
| | | 3.3.1 Propositional Properties | 79 | | |
| | | 3.3.2 Modal Properties | 82 | | |
| | 3.4 | Uniform Definability and Uniform Extensions | 83 | | |
| | 3.5 | Conclusion | 88 | | |

| 4 | Further Remarks on the Dual Negation in Team Logics | | | | |
|---|---|---|----|--|--|
| | 4.1 | Introduction | 9 | | |
| | | 4.1.1 The Dual Negation and Burgess' Theorem | 9 | | |
| | | 4.1.2 Burgess Theorems for Modal and Propositional Team Logics . | 9 | | |
| | | 4.1.3 Remarks on Burgess' Theorem | 9 | | |
| | 4.2 | Burgess Theorems for BSML and $BSML^{\vee}$ | 9 | | |
| | | 4.2.1 Preliminaries | 9 | | |
| | | 4.2.2 Bicompleteness and Incompatibility | 10 | | |
| | | 4.2.3 The Theorems | 10 | | |
| | 4.3 | Burgess Theorems for Propositional Team Logics | 11 | | |
| | | 4.3.1 Bicompleteness Without Failure of Replacement | 11 | | |
| | | 4.3.2 Hawke and Steinert-Threlkeld's Logic | 11 | | |
| | | 4.3.3 The Propositional Fragments of BSML and $BSML^{\vee}$ | 11 | | |
| | | 4.3.4 Propositional Dependence Logic | 12 | | |
| | 4.4 | Interpretations of the Incompatibility Notions | 12 | | |
| | | 4.4.1 Information States; Factual Information; Persistency | 12 | | |
| | | 4.4.2 Expressivist Epistemic Information; Factual Contradictions vs. | | | |
| | | Epistemic Contradictions | 13 | | |
| | | 4.4.3 Pragmatic Information via the Neglect-zero Tendency; Factual | | | |
| | | Contradictions vs. Pragmatic Contradictions | 13 | | |
| | | 4.4.4 Contradictions with Inquisitive Content | 13 | | |
| | 4.5 | Concluding Remarks | 13 | | |
| 5 | Axio | omatizing Modal Inclusion Logic and its Variants | 14 | | |
| | 5.1 | Introduction | 14 | | |
| | 5.2 | Preliminaries | 14 | | |
| | 5.3 | Expressive Completeness and Normal Forms | 15 | | |
| | 5.4 | Axiomatizations | 15 | | |
| | | 5.4.1 $ML(\subseteq)$ | 15 | | |
| | | 5.4.2 $ML(\nabla)$ and $ML(\nabla)$ | 17 | | |
| | 5.5 | Concluding Remarks and Directions for Further Research | 17 | | |
| | 5.6 | Appendix: Translation of $ML(\subseteq)$ into First-order Inclusion Logic | 17 | | |
| 6 | A D | A Deep-Inference Sequent Calculus for a Propositional Team Logic 1 | | | |
| | 6.1 | Introduction | 18 | | |
| | 6.2 | Preliminaries | 18 | | |
| | | 6.2.1 Syntax, Semantics, and Closure Properties | 18 | | |
| | | 6.2.2 Resolutions and Normal Form | 18 | | |
| | 6.3 | The System GT | 19 | | |
| | 6.4 | Partial Resolutions and Cutfree Completeness | 19 | | |
| | 6.5 | Weakening, Inversion, Contraction | 19 | | |
| | 6.6 | Countermodel Construction | 20 | | |
| | 6.7 | Cut Elimination and Derivation Normal Form | 20 | | |

| 6.8 6.9 | Variant System with Independent Contexts | 214 215 |
|------------|--|------------|
| Bibliog | raphy | 219 |
| Samenv | atting | 233 |
| Abstrac | t | 235 |

Acknowledgments

First, a few words about how this thesis came into being: although the thesis was finished at the ILLC, I began work on it at the Department of Mathematics and Statistics, University of Helsinki, under the supervision of Fan Yang. Following Fan's departure to Utrecht University, I first followed her on an extended research visit to Utrecht and eventually transferred to the ILLC at the end of 2023, with Robert van Rooij, Maria Aloni, and Fan serving as (co)promotors. For a period of a few months between my Utrecht research visit and my transfer to Amsterdam, I was supervised by Jouko Väänänen and funded by his ERC grant.

I am grateful, most of all, to the *de facto* supervisors of this thesis, Fan Yang and Maria Aloni, for all their support, guidance, and advice. The cover photo and illustration are a tribute to Fan and Maria—they depict verifiers, falsifiers, and zero-models (as in Maria's [6]), with bamboo steamers playing the part of teams, as per Fan's Chinese translation of 'team' as '屉' (one possible translation of 'steamer' being '蒸屉').

Fan always goes above and beyond for her students and collaborators, and I have been one of the foremost beneficiaries of her consideration and efforts during these past four years. On top of her actual supervision work, Fan spent a great deal of time arranging my transfer to the Netherlands, knowing that it would be best for us to continue working together in person. I am grateful to her for our many stimulating discussions, for teaching me how to write proofs (at least somewhat better than before), for painstakingly checking my work on many occasions and gently correcting my mistakes, for pushing me to present my work whenever I can, and for showing me—in the form of her own papers, presentations, lectures, and supervision—what to aspire to as an academic. I also want to thank Fan for being such a kind and encouraging person.

Maria's creative and insightful ideas serve as the background and motivation for much of this thesis. I am grateful to have played a part in her project. I want to thank Maria for her great enthusiasm, for her detailed and nuanced feedback on my work, for helping keep me alert to philosophical and linguistic concerns, and for always hearing me out and working through my (usually very disorganized) ideas with me. I am also grateful for her wise and kind mentorship throughout these years. I am very grateful to Jouko Väänänen for his always insightful comments, for suggesting many new interesting ideas and research projects to me, and for his assistance with the funding for this thesis and with my transfer to Amsterdam. I also want to thank Jouko for his considerate guidance and advice.

I want to thank Robert van Rooij for acting as my promotor and for assisting with my transfer to Amsterdam, as well as for his helpful comments on the thesis and corrections on the samenvatting.

I am very grateful to my coauthors: Fan, Maria, Matilda Häggblom, Rosalie Iemhoff, and Søren Knudstorp. Thank you for all your great ideas. Our joint work sessions were the most enjoyable part of the preparation of this thesis; I hope to continue working with all of you in the future!

I also want to thank everyone else with whom I discussed my work in detail, and whose suggestions helped shape my ideas and improve the thesis: Nick Bezhanishvili, Ivano Ciardelli, Marco Degano, Josef Doyle, Pietro Galliani, Marianna Girlando, Lorenz Hornung, Luca Incurvati, Tomasz Klochowicz, Juha Kontinen, Elio La Rosa, Marius Tritschler, Jouko Väänänen, Fei Xue, and Woxuan Zhou. I will note here in particular that I began work on the material in Chapter 4 because Juha asked me whether his and Jouko's results in [94] hold for BSML.

I want to thank the committee for taking the time to read and evaluate this thesis: Alexandru Baltag, Nick Bezhanishvili, Ivano Ciardelli, Rosalie Iemhoff, Juha Kontinen, Aybüke Özgün, and Floris Roelofsen.

I am very grateful to Dick de Jongh for his patient and thoughtful guidance during my master's, for teaching me intuitionistic logic, and for suggesting I apply for Fan's PhD position.

Thank you to the Helsinki logic and mathematics community (plus friends and affiliates), in particular to Jouko, Juha, and Juliette Kennedy for your all you taught me, for your mentorship and guidance, as well as for assisting me and welcoming me so warmly. Thank you also to everyone else who taught me, worked with me, and helped make the community what it is: Fan, Matilda, Fausto Barbero, Teemu Hankala, Miika Hannula, Lauri Hella, Minna Hirvonen, Åsa Hirvonen, Tapani Hyttinen, Jonne Iso-Tuisku, Ulla Karhumäki, Siiri Kivimäki, Mika Koskenoja, Kerkko Luosto, Miguel Moreno, Joni Puljujärvi, Davide Quadrellaro, Otto Rajala, Jenna Räsänen, Gabriel Sandu, Tapio Saarinen, Max Sandström, Boban Velickovic, Andrés Villaveces, Jonni Virtema, Vadim Weinstein, Philip Welch, and Ur Ya'ar.

I want to thank everyone at Utrecht University for providing such a warm, welcoming, and stimulating work environment throughout my research visit and beyond. Thank you in particular to Rosalie, Jan Broersen, Colin Caret, Daniel Cohnitz, Raheleh Jalali, Robin Martinot, Nima Motamed, and Albert Visser. I am very grateful to everyone who assisted in my transfer to the Netherlands, including Fan, Maria, Jouko, Juha, Robert, Rosalie, Tuğba Altin, Ewout Arends, Jan Broersen, Peter van Ormondt, and Alexandra Zieglerová.

Thank you to the faculty, staff, and students at the ILLC for making it the rich, exciting, and friendly research institute it is. Thank you to our Nøthing is Logical-group and collaborators for all the interesting research and events—Maria, Marco, Tomasz, Søren, Sonia Ramotowska, Woxuan, Nick, Reinhard Muskens, Peter, Lorenzo Pinton, Floris, Giorgio Sbardolini, Haoyu Wang, Jialiang Yan, and Fan. To my office mate NihiLists and friends Marco, Søren, and Tomasz: I am sure our paths will cross again. Thank you to Floris as well as Thom van Gessel for introducing me to team semantics. Alexandru, thank you for all the interesting courses you taught as well as for assisting me with my applications. Thank you to everyone else I learned from, met at events, and had interesting discussions with: Pieter Adriaans, Rodrigo Almeida, Andrés Arteaga, Bahram Assadian, Orestis Dimou Belegratis, Johan van Benthem, Benno van den Berg, Elsbeth Brouwer, Balder ten Cate, Paul Dekker, Milica Denić, Josef Doyle, Émile Enguehard, Iris van der Giessen, Marianna Girlando, Angelica Hill, Lorenz Hornung, Luca Incurvati, Franziska Jahnke, Victor Joss, Johannes Kloibhofer, Benedikt Löwe, Franci Mangraviti, Robin Martinot, Joel Maxson, Dean McHugh, Takanobu Nakamura, Thijs Ossenkoppele, Aybüke Özgün, Simone Picenni, Miriam Rey, Pablo Rivas-Robledo, Vita Saitta, Thomas Schindler, Julian Schlöder, Sonja Smets, Martin Stokhof, Blaž Istenič Urh, Yde Venema, Haitian Wang, Minghong Xu, Fei Xue, and Morgane Ziegler. See you in the pit, Pablo.

Thank you to everyone I had the opportunity to teach alongside and learn from: Elsbeth, Mika, Fan, Marco, Maria, Juha, and Franziska, as well as to my students. Thank you in particular to Matilda and Lorenz for letting me help supervise you.

Thank you to the logic communities at Tsinghua University and Peking University for your hospitality during our long visit, and for all the interesting discussions we had. Thank you in particular to Jialiang, Fenrong Liu, Yanjing Wang, Haoyu Wang, and Yunsong Wang.

Thank you, Marco, for all the help you provided with producing the final version of the thesis. Fernanda, thank you for your wonderful illustration and for helping me design the cover. Thank you to Robin and Robert for their corrections to the AI-translated samenvatting, and to Rosalie for her help arranging this (any remaining mistakes are due to my failing to incorporate this detailed feedback in a satisfactory way.)

I remain extremely grateful to Ole Hjortland and Toby Meadows, who first taught me logic in St Andrews many years ago.

Finally, thank you to my friends and family for all their support.

Aleksi Anttila Amsterdam, February 2025

Chapter Sources and Author Contributions

This thesis is based on the following papers/manuscripts. In all cases, the authors are listed alphabetically and authorship is shared equally. All authors for a given project contributed equally to all aspects of the project unless specified otherwise. Chapters 2 and 5, which are based on published/accepted papers, reproduce those papers with some very minor additions and modifications. The other chapters, which are based on preprints/unpublished manuscripts, may differ substantially from any possible future publications based on their content.

Chapter 2 is based on:

Maria Aloni, Aleksi Anttila, and Fan Yang. "State-Based Modal Logics for Free Choice". In: *Notre Dame J. Formal Logic* (2024), pp. 1–47. URL: https://doi.org/10.1215/00294527-2024-0027

This paper, in turn, is partly based on Anttila's M.Sc. thesis [12], jointly supervised by Aloni and Yang. (It should be emphasized here that the paper contains a significant number of results which are not in the M.Sc. thesis, including the specific axiomatization which was the main goal of this project, and the material was completely rewritten.) Most of the writing is due to Anttila. This work has been presented (by Anttila) at Logic Colloquium, 2021; Workshop on Logics of Dependence and Independence, ESSLLI, 2021; and Scandinavian Logic Symposium, 2022.

Chapter 3 is based on:

Aleksi Anttila and Søren Brinck Knudstorp. *Convex Propositional and Modal Team Logics*. Manuscript. 2024

Knudstorp first showed that the logics $PL_{\vee}(NE)$ and BSML are convex and came up with the expressive completeness proofs for these union-closed convex logics included in Section 3.3; Anttila then developed the convex logics in Section 3.2 and showed

their expressive completeness by building on the ideas in Knudstorp's proof. The authors jointly developed and polished this material, and added the material on uniform definability and uniform extensions in Section 3.4.

Chapter 4 is based on:

Aleksi Anttila. Further remarks on the dual negation in team logics. 2024. arXiv: 2410.07067 [math.LO]. URL: https://arxiv.org/abs/2410.07067

This work has been presented at Dutch Logic PhD Day, 2023; and the 4th Tsinghua Interdisciplinary Workshop on Logic, Language, and Meaning, 2024

Chapter 5 is based on:

Aleksi Anttila, Matilda Häggblom, and Fan Yang. "Axiomatizing modal inclusion logic and its variants". In: *Arch. Math. Logic* (2024). Forthcoming. arXiv: 2312.02285 [math.LO]. URL: https://arxiv.org/abs/2312.02285

This paper, in turn, is partly based on Häggblom's M.Sc. thesis [63], jointly supervised by Yang and Anttila. (It should be emphasized here that although the paper is based on Häggblom's M.Sc. thesis, the authors contributed equally to this project. As with Chapter 2, the paper contains a significant number of results which are not in the M.Sc. thesis, and the material was completely rewritten.) This work has been presented (by Häggblom) at Logic Colloquium, 2023.

Chapter 6 is based on:

Aleksi Anttila, Rosalie Iemhoff, and Fan Yang. Deep-inference Sequent Calculi for Propositional Team Logics. Manuscript. 2024

The main idea for the system and most of the writing is due to Anttila. This work has been presented (by Anttila) at PhDs in Logic, 2023.

The samenvatting was translated from the abstract using ChatGPT, with corrections by Robin Martinot and Robert van Rooij.

Chapter 1

Overview

This dissertation collects together five papers which focus primarily on the expressive power, axiomatizations, and proof theory of propositional and modal logics with *team semantics*.

In team semantics, formulas are interpreted with respect to sets of evaluation points called *teams* rather than single evaluation points as in standard Tarskian semantics. The basic idea can be adapted to multiple settings: teams can be for instance, sets of first-order assignments, sets of propositional valuations, or sets of possible worlds. Team semantics was introduced by Hodges [78, 79] to provide a compositional semantics for Hintikka and Sandu's *independence-friendly logic* (IF) [77, 75]. It was later further developed by Väänänen in his work on *dependence logic* (D) [119]; this gave rise to a research programme focused in particular on so-called *logics of dependency*—logics with team semantics such as dependence logic, *inclusion logic* [51, 54], and *independence logic* [59] furnished with *atoms of dependency* expressing that certain relationships hold between the values of the elements in a team. Independently, Ciardelli, Groenendijk, and Roelofsen (building in particular on ideas in dynamic semantics [68, 61, 117, 123]) developed *inquisitive logic* [60, 41, 38, 35] which also essentially employs team semantics (see [133, 34]).

Teams can have different interpretations—they can represent, for instance, databases, data sets, and information states; accordingly, team semantics and team-based logics have found application in many different fields. The atoms of dependency mentioned above, for instance, correspond to dependency notions studied in relational database theory, and axiomatizations and implication problems involving these notions can be studied in the setting of team logics [65, 66]. Other areas these logics have been applied to include formal semantics and philosophical logic [38, 33, 67, 101, 7]), quantum information theory [83, 4, 1], and social choice theory [102].

One of the primary motivations for the development and study of logics with team semantics has been the capacity of teams to represent, in a perspicuous and simple manner, phenomena that would be difficult or impossible to model (at least in an equally satisfactory way) in a setting based on standard singleton evaluation points. By way of example, let us consider *propositional dependence logic* [139]. This logic extends classical propositional logic with *dependence atoms* =(p,q) expressing that the truth value of q is *functionally dependent* on that of p. Consider the following table, which represents a propositional team $t = \{v_1, v_2, v_3\}$ —a set of valuations v_1, v_2, v_3 over a finite set $\{p,q,r\}$ of propositional variables:

| | p | q | r |
|-----------------------|---|---|---|
| v_1 | 1 | 0 | 1 |
| v_2 | 1 | 0 | 0 |
| <i>v</i> ₃ | 0 | 1 | 0 |

The atom =(p,q) is true in this team because the value of q is functionally dependent on that of p in that there is a function f from the set $t[p] = \{v(p) | v \in t\}$ of truth values of p in the team to the set of truth values of q such that f(v(p)) = v(q) for all $v \in t$: for any valuation v with v(p) = 1, we have v(q) = 0, and for any valuation v with v(p) = 0, we have v(q) = 1. On the other hand, the atom =(p,r) is not true, because if v(p) = 1, v(r) might be 1 or 0.

A dependence atom, then, expresses directly that such a functional dependence fact holds at a (team-semantic) evaluation point, and this is only meaningful due to the fact that the evaluations points are sets rather than singletons: a single valuation assigns a single value to each propositional variable, so functional dependence can never fail at a single valuation.

In other words, the semantics of the dependence atom leverages the nature of the evaluation points as sets to directly express a phenomenon—functional dependence—that can be naturally represented using sets. Many other team-semantic atoms and connectives function in a similar way: their truth conditions at an evaluation point (a team) make use not only of the values assigned to variables at that point, but also the second dimension provided by the nature of the evaluation point as a set—a multitude—and the structure of the set itself (its cardinality, etc.), as well as the interactions between these dimensions (as we saw in the dependence atom example: =(p,q) is true in a team just in case any pair of elements in the team which agree on the truth value of p also agree on the truth value of q).

Let us consider another example. *Propositional inquisitive logic* [41, 38, 35] employs the *global* or *inquisitive disjunction* \forall (see also [2]) to formalize questions: $p \forall \neg p$, for instance, represents the question as to whether or not p is the case. This disjunction has the standard disjunction semantics, defined with respect to teams:

$$t \models \phi \lor \psi \iff t \models \phi \text{ or } t \models \psi.$$

Valuations represent possible states of affairs or possible worlds in the familiar way, whereas teams—sets of valuations—represent *information states*: if, for instance, it is raining according to all worlds in a team/information state, then according to the information embodied in this state, it is raining. A propositional variable *p* is satisfied

by a team just in case it is satisfied by all valuations in the team; its satisfaction by a team represents the fact that the sentence corresponding to p is established in the corresponding information state as described above. (A negated variable $\neg p$, similarly, is satisfied by a team just in case it is satisfied by all valuations in the team in the usual way.) On the other hand, if a question such as $p \vee \neg p$ is satisfied by a team, this represents the fact that the question can be truthfully resolved given the information in the state—the information in the state implies that one of the possible answers to the question is true. The formula $p \vee \neg p$ is satisfied by a team just in case p is satisfied by the team, or $\neg p$ is—the question as to whether p can be truthfully resolved in an information state just in case either the information state establishes that p, or it establishes that not p. Consider the following figure, where the nodes are valuations (labelled according to what they make true/false: $v_{p\overline{q}}(p) = 1$ and $v_{p\overline{q}}(q) = 0$, etc.), and the encircled areas are teams:



Here the team $\{v_{pq}, v_{p\overline{q}}\}$ satisfies p because all valuations in the team evaluate p as true. The corresponding information state therefore establishes that p, and hence the question $p \vee \neg p$ can be resolved in the affirmative in this state. The team $\{v_{pq}, v_{\overline{p}q}\}$ does not satisfy p (because it contains the valuation $v_{\overline{p}q}$, with $v_{\overline{p}q}(p) = 0$), and it does not satisfy $\neg p$ (because it contains the valuation v_{pq} , with $v_{pq}(\neg p) = 0$), and hence there is not enough information in the corresponding state to resolve this question.

The information state-interpretation leverages the nature of the evaluation points as sets in that sets can incorporate multiple possibilities, thus representing uncertainty or lack of information. A single possible world assigns a single definite truth value to each propositional variable, which represents every fact being settled in such a world: for each propositional variable, either v(p) = 1, or v(p) = 0 (whence $v(\neg p) = 1$). As we saw above, a team need not satisfy one of p and $\neg p$ because it can contain possibilities instantiating both options—this represents uncertainty concerning p.

The interpretation of $\forall \forall$ builds on the information-state interpretation and exploits the sethood of teams in a further way: notice that given the standard Tarskian semantics, there is only one way for a formula α to be satisfied in that there is a single maximal information state $\{v | v \models \alpha\}$ that establishes α . All further refinements of this state—all subsets of the set—also establish α , and hence whatever information they add is superfluous to the truth value of α . For a question such as $p \forall \neg p$, on the other hand, there is no such single maximal information state—a team satisfies $p \forall \neg p$ just in case it is either a subset of $\{v | v \models p\}$, or a subset of $\{v | v \models \neg p\}$. The two ways in which the formula can be satisfied represent the two distinct ways the question can be resolved—the two possible answers to the question.

Now, one basic structural feature of sets is emptiness/nonemptiness: a team, unlike a valuation, can be empty. Four of the five papers in this dissertation focus mainly on logics leveraging this particular feature of teams in some way, so it functions as a loose guiding theme for this dissertation and overview. The two most prominent types of team-based logic introduced above which inspired much of the subsequent work in team semantics—dependence logic and inquisitive logic—do not exploit this feature in any significant way (in a sense we explain directly below). There are, however, a large number of logics that do make significant use of emptiness/nonemptiness, and they come with their own conceptual and technical motivations for embracing the use of this feature.

Before discussing logics which do make use of emptiness/nonemptiness, let us explain the sense in which dependence logic and inquisitive logic do not make use of this particular structural feature of sets. This is reflected in the *team-semantic closure properties* satisfied by the formulas of these logics (or, simply, by these logics). The closure properties describe how formulas of team-based logics interact with the structural properties of teams, and they play an important conceptual and technical role in the study of these logics—as we discuss presently, many of the results in this dissertation concern expressive completeness results with respect to classes of team properties satisfying certain closure properties. The closure properties of dependence logic and inquisitive logic relevant to our present purposes are the *empty team property*—each formula of either of these logics is true in the empty team—and the *downward closure property*: if a formula of one of these logics is true in a team t, it is also true in all subteams of t.

The empty team property directly reflects the fact that the emptiness/nonemptiness of the evaluation team is not exploited in the semantics of these logics in a way that would allow formulas to express that the evaluation team is nonempty. More simply, nonemptiness cannot be expressed in the languages of these logics because all their formulas are true in the empty team. Due to this property, the empty team, from the point of view of these logics, is simply a state of absurdity in which even contradictions are true.¹ One of the reasons these logics abide by the empty team property has to do with the conceptual difficulties associated with apprehending the empty team as anything but an absurdity state that makes everything true. Consider, for instance, the following passage from Rönnholm [109, p. 192]:

By interpreting teams as epistemic sets of uncertainty, the empty team does not make much sense. This is because (at least) the "real" state of affairs should always be included in the set of all possibilities. Moreover,

¹It is interesting to note that \perp , the syntactic representation of contradiction, while true in no evaluation point whatsoever in classical logics with standard Tarskian semantics, has the following truth conditions in these logics, and indeed in most team-based logics: \perp is true in *t* just in case *t* = \emptyset .

by interpreting teams as databases, the empty team would mean *not having a database at all*...

While the empty team property can be seen as reflecting the fact that the emptiness/nonemptiness of the evaluation team itself is not exploited in the semantics of these logics, downward closure, on the other hand, reflects the fact that the truth conditions of these logics do not include conditions to the effect that in order for a formula to be true in the evaluation team, the team must contain elements (or non-empty subsets) satisfying certain properties. This type of condition can also be seen as a condition involving emptiness/nonemptiness, albeit in a less direct way.

Let us consider an example. Many of the papers in this dissertation are concerned with operators which can be used to represent *epistemic modals* such as the 'might' in 'It might be raining'. One such operator is the operator ∇ (introduced in [73], and studied in Chapter 5), which has the following truth conditions:

$$t \models \nabla \phi \iff [t = \emptyset \text{ or } \exists s \subseteq t : s \neq \emptyset \text{ and } s \models \phi].$$

That is, $\nabla \varphi$ is true in a team just in case the team is empty, or it contains a nonempty subteam that satisfies φ . How does this represent an epistemic modal? Recalling the information state-interpretation of teams, the idea is that if, for instance, there are some worlds in an information state according to which it is raining—or, equivalently, there is a nonempty substate of the state consisting of such worlds—then, for all that one knows given the information embodied in the state, it might be raining.² It is easy to see that while $\nabla \varphi$ always has the empty team property, we have, for instance, that ∇p is not downward closed (where, as above, p is true in a team t just in case it is true in all $v \in t$). Therefore, operators such as ∇ are not expressible in dependence logic or in inquisitive logic, which are downward closed.

In many frameworks (see, e.g., [35, 123, 25, 67]), downward closure is called *persistency*. This has to do with the fact that, on the information state-interpretation, downward closure corresponds to the following property of the information or content communicated by formulas: if one is in an information state that establishes some specific piece of information, then the process of becoming more informed by ruling out more possibilities (i.e., moving to a substate of the initial information state) does not cause one to lose the piece of information that had already been established. The content communicated by formulas which are not downward closed is not persistent in this way—we discuss this briefly in Chapter 4.

We now move on to logics which do make use (in the sense characterized above) of the emptiness/nonemptiness feature of teams. As expected, these logics break the empty team property or downward closure, or both. We mention one important family of logics from the literature before discussing the logics we focus on in this dissertation.

²It seems that the first account of epistemic modals along these lines was in Veltman's dynamic framework in [123]. See also [117, 129, 25, 73, 107, 140, 67]. We consider multiple different 'might'-operators in Chapters 3, 4, and 5.

Consider the Boolean or contradictory negation ~, with the semantics

$$t \vDash \sim \varphi \iff t \not\vDash \varphi.$$

That is, the Boolean negation of φ is true in a team just in case φ is not true—this negation is the classical negation with respect to teams. Clearly, for instance, ~ *p* neither has the empty team property nor is downward closed. The Boolean negation was essentially considered already by Hintikka, who developed an extension of independence-friendly logic (IF) with this negation in order to compensate for some of the apparent shortcomings of the *dual negation* of IF [75] (discussed in Chapter 4). Väänänen, similarly, introduced *team logic*³, an extension of first-order dependence logic (D) with ~. This connective is much more powerful than the others we consider here (clearly it can capture much more than only emptiness/nonemptiness phenomena); for instance, while D is expressively equivalent to existential second-order logic [91]. For more on logics featuring this negation, see [96, 97].

Two of the logics of dependency mentioned above, inclusion logic and independence logic, have the empty team property but are not downward closed. Their semantics include conditions of the kind described above that mandate the existence of elements satisfying certain properties in teams. For instance, in Chapter 5 we study *extended modal inclusion logic* (introduced in [73]). This logic extends classical modal logic with *inclusion atoms* $\alpha \subseteq \beta$ (where α and β are classical formulas, interpretable on single worlds), where

$$t \vDash \alpha \subseteq \beta \iff [\forall w \in t : \exists v \in t : w \vDash \alpha \iff v \vDash \beta].$$

That is, $\alpha \subseteq \beta$ is true in a team *t* just in case for all worlds *w* in *t*, there exists a world *v* in *t* that gives the same truth value to β as *w* does to α . These atoms import the notion of *inclusion dependencies* from database theory into team semantics (just as dependence atoms may be seen as importing the notion of functional dependencies). While the contribution of nonemptiness in the semantics of these atoms is less direct, we will see in Chapter 5 that this logic is equivalent in expressive power to an extension of classical modal logic with the operator ∇ discussed above (indeed, observe that for classical α , $\nabla \alpha$ is definable using these atoms: $\nabla \alpha \equiv \top \subseteq \alpha$.)

Finally, and most crucially for our purposes, we have the *nonemptiness atom* NE (introduced in [121, 140])—which this dissertation is named after—with the semantics:

$$t \models \text{NE} \iff t \neq \emptyset.$$

That is, NE is true in a team just in case the team is nonempty. This atom, then, isolates the emptiness/nonemptiness feature of teams, and expresses nothing but nonemptiness

³The term 'team logic' is ambiguous. In this dissertation, we generally use it to refer to any logic primarily intended to be interpreted using team semantics.

facts. Given that, as discussed above, it is not immediately clear from the interpretations of team semantics what it would mean for the empty team to fail to make something true, not much use has been made in the literature of this atom.

Recently, however, Aloni [6] introduced *bilateral state-based modal logic*, an extension of classical modal logic (interpreted on teams) with NE. Aloni employs BSML to account for *free choice* inferences (see, e.g., [128, 87]) and related linguistic phenomena. In these inferences, conjunctive meanings are unexpectedly derived from disjunctive sentences; for instance:

> You may go to the beach or to the cinema. → You may go to the beach and you may go to the cinema.

That is, upon hearing the first of these sentences, people are inclined to infer the second. On Aloni's account, speakers, when interpreting language, create mental structures that represent reality; and inferences such as the above are due to a tendency in human cognition to disregard empty structures in this process of interpretation (the *neglect-zero tendency*). This tendency is modelled using a *pragmatic enrichment function* which, given a formula α of classical modal logic, recursively appends ' \wedge NE' to each subformula of α . We discuss this account in more detail in Chapter 2.

The bulk of this dissertation is dedicated to the study of the logic and expressive import of nonemptiness, carried out via an investigation of the mathematical properties of BSML and related logics. Most of the results are technical⁴; they concern, as mentioned above, the expressive power, axiomatizations, and proof theory of these logics.

An important role is played by *expressive completeness theorems* with respect to the team-semantic closure properties (such as downward closure). To give an example of such a theorem, propositional dependence logic is expressively complete with respect to the class of all downward-closed *team properties* with the empty team property [139] (where a team property is a class of teams). This means that the class of properties definable by formulas of propositional dependence logic (where φ defines the property consisting of all teams in which φ is true) is precisely the class of all downward-closed properties with the empty team properties can be extended to team properties in a natural and obvious way). These theorems allow for concise and tractable characterizations and classification of these logics, and provide an effective tool, as we will see, for proving definability results, completeness of axiomatizations (see, e.g., [139, 140]), and other properties such as uniform interpolation [43].

We conclude by giving a brief summary of the contents of each paper.

In the first paper, *State-based Modal Logics for Free Choice* [7] (Chapter 2), we focus on Aloni's BSML. We introduce two extensions of BSML: BSML^{\circ}, or BSML with

⁴Chapter 4 also includes some conceptual discussion; note in particular that we touch there on the failure of persistency (Sections 4.4.2 and 4.4.3), and, somewhat indirectly, on the interpretation of NE and hence on what it might mean to simply express that a team is nonempty (Section 4.4.3).

the *emptiness operator* \oslash , an operator which can be used to cancel the effects of NE, and BSML^V, or BSML with the global disjunction V. We prove expressive complete ness theorems for these extensions: we show that BSML^V is expressively complete for the class of all modal team properties invariant under bounded team bisimulation. We also show that BSML^{\oslash} is expressively complete for the class of all *union-closed* modal properties invariant under bounded team bisimulation, where a formula is union closed if its truth in a nonempty collection of teams implies its truth in the union of the collection (and similarly for team properties).

Building on the propositional axiomatizations in [140], and following a strategy commonly used for axiomatizing propositional and modal team-based logics (see, e.g., [139, 140]), we make use of normal forms obtained from the expressive completeness theorems to provide natural deduction axiomatizations for the extensions as well as for BSML.

While we obtain expressive completeness theorems for extensions of BSML in the first paper, the expressive power of BSML itself is left as an open problem in that paper. This is solved in the second paper, *Convex Propositional and Modal Team Logics* [16] (Chapter 3). In this paper, we focus more generally on expressive completeness results involving the closure property of *convexity*—a formula is convex just in case if it is true in two teams, then it is also true in all the teams between these two teams with respect to set inclusion. While convexity has been extensively studied in formal semantics (and other fields) [19, 23, 55], it has thus far not received much attention in the team semantics literature (with the exception of [72, 36]). Convexity is a natural extension of downward closure to a setting in which the empty team property fails in that a formula is downward closed if and only if it is convex and additionally has the property that if it is true in any team whatsoever, it is also true in the empty team, and hence has the empty team property.

We show that BSML is expressively complete for the class of all convex and unionclosed modal properties (invariant under bounded bisimulation), and we show an analogous result for the propositional fragment of BSML as well as for another modal extension of this fragment.

We also introduce propositional and modal logics which we prove to be expressively complete with respect to the class of *all* convex propositional/modal properties. The connective involved in expressing emptiness/nonemptiness facts in these convex logics is a variant \bullet of the might-operator ∇ discussed above.

There is an interesting sense, related to the failure of *closure under uniform substitution* in team-based logics, in which one of our novel convex propositional logics extends propositional dependence logic, and in which another of them extends propositional inquisitive logic; we generalize the notion of *uniform definability* [32, 52, 133, 135, 37, 72] from the team semantics literature to make this notion of extension precise.

In the third paper, *Further Remarks on the Dual Negation in Team Logics* [13] (Chapter 4), we take a closer look at the *bilateral negation* of BSML. This is essentially

the same notion as the *dual* or *game-theoretical negation* of independence-friendly logic and early formulations of dependence logic. In IF and D, the dual negation exhibits an extreme degree of semantic indeterminacy in that for any pair of sentences φ and ψ of IF/D, if φ and ψ are *incompatible* in the sense that they share no models, there is a sentence θ of IF/D such that $\varphi \equiv \theta$ and $\psi \equiv \neg \theta$ (as shown originally by Burgess [29] in the context of the prenex fragment of *Henkin quantifier logic* [74], which is equivalent in expressive power to IF/D [48, 124, 75, 119]; and later generalized to arbitrary formulas by Kontinen and Väänänen [94]).

The most natural analogue of Burgess' notion of incompatibility in the setting of modal team semantics of BSML is what we call \perp -*incompatibility*: two formulas φ and ψ are \perp -incompatible just in case if they are both jointly true in a team, the team must be empty, i.e., $\varphi, \psi \models \bot$. While the negation \neg of BSML does conform to this notion of incompatibility in the sense that φ and $\neg \varphi$ are always \perp -incompatible, it can be shown that a BSML-formula and its negation also conform to another, stronger notion of incompatibility: φ and ψ are ground-incompatible if the truth of φ in t and of ψ in s implies that the intersection of t and s is empty. Ground-incompatibility is a notion that is sensitive to the emptiness/nonemptiness information in formulas which are not downward closed: we show that while ground-incompatibility is equivalent to 1-incompatibility in a downward-closed setting, it is strictly stronger when downward closure fails (that is, a ground-incompatible pair is always ⊥-incompatible, but the converse implication fails in a non-downward-closed setting). We proceed to show that an analogue of Burgess' theorem holds for BSML if we employ ground-incompatibility as our notion of incompatibility, while no such analogue holds if \perp -incompatibility is used.

We then go on to prove analogues of Burgess' theorem for multiple different propositional and modal team-based logics, defining new notions of incompatibility as required. These notions are, more properly, *pair properties*: classes of ordered pairs of team properties. Together with its converse, a Burgess-theorem analogue can be seen as an expressive completeness theorem with respect to the relevant pair property; we make this precise by defining the notion of *bicompleteness*.

We also consider some intuitive interpretations of the incompatibility notions we define, connecting some of them with *epistemic contradictions* such as 'It is raining but it might not be raining', and some with what we call *pragmatic contradictions*— contradictions obtained using Aloni's pragmatic enrichment function.

In the fourth article, Axiomatizing Modal Inclusion Logic and its Variants [14] (Chapter 5), we consider extended modal inclusion logic, already introduced above the extension of classical modal logic with inclusion atoms $\alpha \subseteq \beta$ (a logic which has the empty team property, but breaks downward closure). We recall the expressive completeness theorem for this logic (it is complete for the class of all union-closed modal properties with the empty team property which are invariant under bounded bisimulation) given in [73], and refine this proof slightly using formulas defined in [89]. We then leverage expressive completeness to provide a natural deduction axiomatization for this logic, making use of essentially the same strategy as that employed in Chapter 2. We similarly recall (likewise from [73]) the expressive completeness theorem for the extension of classical modal logic with the ∇ -operator, which is expressively equivalent to extended modal inclusion logic, and axiomatize this extension as well. Finally, we introduce yet another variant ∇ of ∇ , and show the expressive completeness of and axiomatize the extension of classical modal logic with ∇ (this extension is also expressively equivalent to modal inclusion logic).

In the fifth article, A Deep-Inference Sequent Calculus for a Propositional Team Logic [15] (Chapter 6), we provide a sequent calculus for PL(W) (see, e.g., [112, 139, 33, 138]), the team-based propositional logic with both the split disjunction \vee and the global disjunction W—a logic similar to both propositional dependence logic and propositional inquisitive logic. This logic has the empty team property and is downward closed.

While there are many natural deduction- and Hilbert-style axiomatizations of propositional team logics in the literature (see, e.g., [41, 112, 107, 139, 33, 140, 96, 39, 137]), the development of sequent calculus systems and of proof theory in general for these logics has been slower. One of the main reasons for this is that typically these logics are not closed under uniform substitution. Many proof-theoretic techniques depend on the universal applicability of rules, but given the failure of closure under uniform substitution in team logics, the axiomatizations of these logics typically feature rules with syntactic restrictions which are not sound for all formulas.

Our solution to this issue is to employ a standard classical G3-style sequent calculus for the fragment of PL(W) corresponding to classical propositional logic (the W-free fragment), with syntactic restrictions limiting some of these rules to classical formulas. This classical subsystem is supplemented with *deep-inference* (see, e.g., [62, 27, 28]) rules for the global disjunction W—that is, rules which allow one to introduce the global disjunction (almost) anywhere within a formula, rather than only as its main connective. The deep-inference rules allow us to work around the difficulties engendered by the syntactically restricted rules. The resulting system is a simple minimal extension of the classical G3-style subsystem, and we show that it satisfies various desirable properties: it admits height-preserving weakening, contraction and inversion; it supports a procedure for constructing cutfree proofs and countermodels similar to that for G3cp; and cut elimination holds as a corollary of cut elimination for the G3-style subsystem together with a normal form theorem for cutfree derivations in the system.

Chapter 2 State-based Modal Logics for Free Choice

This chapter is based on:

Maria Aloni, Aleksi Anttila, and Fan Yang. "State-Based Modal Logics for Free Choice". In: *Notre Dame J. Formal Logic* (2024), pp. 1–47. URL: https://doi.org/10.1215/00294527-2024-0027

Abstract We study the mathematical properties of bilateral state-based modal logic (BSML), a modal logic employing state-based semantics (also known as team semantics), which has been used to account for free choice inferences and related linguistic phenomena. This logic extends classical modal logic with a nonemptiness atom which is true in a state if and only if the state is nonempty. We introduce two extensions of BSML and show that the extensions are expressively complete, and develop natural deduction axiomatizations for the three logics.

2.1 Introduction

In this article, we study *bilateral state-based modal logic* (BSML), a modal logic employing *team semantics*. BSML was introduced by Aloni in [6] to account for *free choice* (FC) inferences. In FC-inferences, conjunctive meanings are unexpectedly derived from disjunctive sentences:

- (1) Free choice (see Kamp [87], von Wright [128])
 - (a) You may go to the beach or to the cinema.

→ You may go to the beach and you may go to the cinema.

(b) $\diamondsuit(b \lor c) \rightsquigarrow \diamondsuit b \land \diamondsuit c$

The novel hypothesis at the core of the account in [6] is that FC and other related inferences are a consequence of a tendency in human cognition to disregard structures

| [■,■,■] | [■,□,■] | $[]; [\Delta, \Delta, \Delta]; [\diamondsuit, \blacktriangle, \blacklozenge]$ |
|--------------|---------------|---|
| (a) Verifier | (b) Falsifier | (c) Zero-models |

Figure 2.1: Models for the sentence *Every square is black*.

that verify sentences by virtue of some empty configuration (*neglect-zero tendency*); see Bott, Schlotterbeck, and Klein [26]. Models that verify sentences by virtue of an empty witness-set are called *zero-models* (see Figure 2.1 for an illustration).

BSML is introduced as a framework allowing for the formalization of the neglectzero tendency and the rigorous study of its impact on linguistic interpretation. It extends classical modal logic (ML) with a special nonemptiness atom (NE) requiring the adoption of team semantics. In team semantics (introduced originally by Hodges [78, 79] and developed further in the literature on *dependence logic* by Väänänen [119, 120]), formulas are interpreted with respect to sets of evaluation points called *teams*, rather than single points. In the type of modal team semantics employed by BSML, teams are sets of possible worlds. In [6], teams are also called states since they represent speakers' information states; we will use these terms interchangeably. The atom NE (introduced by Yang and Väänänen in [121] and [140]) is true (or sup*ported*) in a state if and only if the state is not empty: $s \models \text{NE}$ iff $s \neq \emptyset$. This atom is used, in [6], to define a *pragmatic enrichment function* []⁺ whose core effect is to disallow zero-models; it is then shown that pragmatic enrichment yields nontrivial effects including the prediction (given certain preconditions) of both narrow-scope $(\diamondsuit(b \lor c) \rightsquigarrow \diamondsuit b \land \diamondsuit c)$ and wide-scope $(\diamondsuit b \lor \diamondsuit c \rightsquigarrow \diamondsuit b \land \diamondsuit c)$ FC-inferences, as well as their cancellation under negation (see (2) below). See Figure 2.2 for illustrations of zero-models and the narrow-scope FC-prediction in BSML.¹ See [6] for more discussion and for comparisons with other accounts of FC.

- (2) Cancellation of FC under negation (see Alonso-Ovalle [10])
 - (a) You are not allowed to eat the cake or the ice cream.
 - → You are not allowed to eat either one.
 - (b) $\neg \diamondsuit(\alpha \lor \beta) \rightsquigarrow \neg \diamondsuit \alpha \land \neg \diamondsuit \beta$

BSML is closely related to modal logics in the lineage of dependence logic (see [119]), such as modal dependence (see Hella et al. [71], Väänänen [120], Yang [134]),

¹The state $\{w_q\}$ in Figure 2.2(a) supports $p \lor q$. This is because we can find substates of $\{w_q\}$ supporting each disjunct: $\{w_q\}$ itself supports q, and the empty state vacuously supports p. But $\{w_q\}$ is a zero-model for $p \lor q$ because no *nonempty* substate supports the first disjunct. Pragmatic enrichment has the effect of ruling out this zero-model— $\{w_q\}$ does not support $[p \lor q]^+$. The state $\{w_p, w_q\}$ in Figure 2.2(b) is a nonzero verifier for $p \lor q$, so it supports $[p \lor q]^+$. The state $\{w_{\infty}\}$ in Figure 2.2(c) is a zero-model for $\diamondsuit(p \lor q)$, and it does not support $\diamondsuit p \land \diamondsuit q$. The state $\{w_{pq}\}$, on the other hand, is a nonzero verifier for $\diamondsuit(p \lor q)$. We will see that such a verifier must also verify $\diamondsuit p \land \diamondsuit q$ —in other words, narrow-scope FC is predicted by the fact that $[\diamondsuit(p \lor q)]^+ \vDash \diamondsuit p \land \diamondsuit q$. (See Section 2.2 for the definition of the pragmatic enrichment function and more details.)



Figure 2.2: Zero-models and FC in BSML. Worlds are labelled according to what is true in them: w_p stands for a world where only p is true, w_{pq} for one in which only p and q are, etc. The accessibility relation is depicted using arrows.

inclusion (see Anttila, Häggblom, and Yang [14], Hella and Stumpf [73], Kontinen et al. [89]), independence (see Kontinen et al. [90]), and team logic (see [89], Lück [97], Müller [99]). It is also similar to modal inquisitive logic (see Ciardelli [34, 35]). However, the specific combination of connectives (NE, \lor , \diamondsuit , \neg) on which the account of FC in [6] depends is unique to BSML. Let us briefly discuss each of these in turn.

The most crucial component in the account of FC is the nonemptiness atom NE. This atom also sets BSML apart from dependence and inquisitive logic in that formulas containing NE might not be *downward closed* and might not have the *empty state* (or empty team) property. Logics which violate these properties are commonly known as *team logics* (see, e.g., [119, 140, 97]). Most team logics incorporate the *Boolean negation* ~ ($s \models \sim \varphi$ iff $s \neq \varphi$). This negation is very strong in terms of expressive power: the extension of classical modal logic ML with ~ (known as *modal team logic* (MTL)) can express all modally definable state properties (see [89]). BSML, being essentially ML together with NE, is a more modest extension. The propositional fragment of BSML (i.e., classical propositional logic extended with NE) was studied already in [140].

The disjunction \lor in BSML is the standard disjunction in the dependence logic lineage, commonly referred to as the *tensor disjunction* or *local disjunction*. Inquisitive logic makes use of a different disjunction, the *global disjunction* \lor (also known as the *inquisitive disjunction*):

$$s \vDash \varphi \lor \psi \quad \text{iff} \quad \text{there are } t, u \subseteq s \text{ s.t. } s = t \cup u \text{ and } t \vDash \varphi \text{ and } u \vDash \psi;$$
$$s \vDash \varphi \lor \psi \quad \text{iff} \quad s \vDash \varphi \text{ or } s \vDash \psi.$$

The predictions in [6] rely on the use of \lor , but it is interesting to note that alternative accounts of FC have been given in inquisitive semantics, making use of \lor instead of \lor (see, e.g., Aloni [5], Aloni and Ciardelli [8], Nygren [101]; see also [6] for a comparison).

The modalities \diamond and \Box of BSML (essential for the FC-predictions) are equivalent to ones considered in an early version of modal inquisitive logic (see [34]), and are

closely related to the modalities in possibility semantics (see Humberstone [81]). They are distinct from the standard modalities of modal team semantics—we refer the reader to Anttila [12] for a comparison.

For its predictions involving negation (such as (2)), BSML relies on a *bilateral* semantics: in addition to support/assertion conditions, each formula is also provided with *anti-support*/rejection conditions. The semantics of the *bilateral negation* \neg are then defined using the anti-support conditions.²

In this article, we study the logical properties of BSML, as well as two new extensions of BSML: BSML with the global disjunction \vee (BSML^{\heartsuit}), and BSML with the novel *emptiness operator* \oslash (BSML^{\oslash}). The emptiness operator is a natural counterpart to NE which can be used to cancel out its effects: a state *s* supports $\oslash \varphi$ if and only if *s* supports φ or *s* is empty ($s \models \oslash \varphi$ iff $s \models \varphi$ or $s = \varnothing$), meaning that, for instance, $\oslash NE$ is always supported. While these extensions may prove to have interesting applications of their own (we return to this in the conclusion), our introduction of them is motivated primarily by technical considerations: as we show in the first part of the article, each extension is expressively complete for a natural class of state properties. Moreover, in the second part of the article we axiomatize each of the three logics, and the expressive completeness of the extensions plays a crucial role in our proof of the completeness of these axiomatizations.

In the first part, then, we focus on expressive power. We characterize the expressiveness of our logics in terms of the notion of *state (or team) bisimulation* as introduced in the literature on modal team semantics (see [71, 89]; cf. *inquisitive bisimulation* in Ciardelli and Otto [40]). We show that BSML^V is expressively complete for the class of all state properties invariant under bounded bisimulation (meaning that BSML^V and MTL are equivalent in expressive power; see [89]), and that BSML[©] is expressively complete for the class of all *union-closed* state properties invariant under bounded bisimulation. These results build on similar results in, for instance, [71], [73], [89], and Sevenster [114]. BSML, as we will demonstrate, is union closed but cannot express all union-closed properties.

In the second part, we develop natural deduction axiomatizations for each of the three logics. These systems build on systems presented in [140] for logics which are essentially the propositional fragments of BSML and $BSML^{\vee}$. For other similar natural deduction axiomatizations for modal state-based logics, see, for instance, [14] and [134].

²Bilateralism is typically associated with inferentialism in the proof-theoretic tradition (see, e.g., Price [104], Smiley [115], Rumfitt [110], Restall [108], Wansing [125], Incurvati and Schlöder [86], and Wansing and Ayhan [126]; see also Schroeder-Heister [113] for an overview). Other systems employing bilateral semantics include first-degree entailment logic (FDE) (see Anderson and Belnap [11], Dunn [46], and Belnap [21, 22]) and truthmaker semantics (see van Fraassen [122], and Fine [49]). Recently, Aher [3] and Willer [127] presented bilateral accounts of FC in an inquisitive and a dynamic setting, respectively. The bilateral negation in BSML is essentially the same notion as the *dual* or *game-theoretic negation* of *independence-friendly logic* (see Hintikka and Sandu [77], and Hintikka [75]) and some formulations of dependence logic (see [119, 120]).

The article is structured as follows. In Section 2.2, we define the syntax and semantics of BSML and the extensions, and discuss some basic properties of these logics. In Section 2.3, we show the expressive power results described above. We also derive some simple consequences of these results such as the finite model property (for each of the logics) and uniform interpolation (for BSML^{\vee} and BSML^{\oslash}). In Section 2.4, we present our natural deduction axiomatizations, and in Section 2.5 we conclude by noting some open problems. Our work is partly based on Anttila's M.Sc. thesis [12] (supervised by Aloni and Yang), which contains preliminary versions of some of the results.

2.2 Bilateral State-Based Modal Logic

In this section, we define the syntax and semantics of BSML and its extensions, discuss the basic properties of these logics, and recall standard notions and results from team semantics.

2.2.1. DEFINITION (Syntax). Fix a (countably infinite) set Prop of propositional variables. The set of formulas of *bilateral state-based modal logic* BSML is generated by

$$\varphi ::= p \mid \bot \mid \neg \varphi \mid (\varphi \land \varphi) \mid (\varphi \lor \varphi) \mid \diamondsuit \varphi \mid \mathsf{NE},$$

where $p \in \text{Prop.}$ Classical modal logic ML is the NE-free fragment of BSML.

BSML^{\forall} is BSML extended with the *global disjunction* \forall : the set of formulas of BSML^{\forall} is generated by the definition for BSML augmented with the case $\varphi \lor \varphi$.

BSML^{\circ} is BSML extended with the *emptiness operator* \diamond : the set of formulas of BSML^{\circ} is generated by the definition for BSML augmented with the case $\diamond \phi$.

Throughout the article, we use the first Greek letters $\alpha, \beta, \gamma, ...$ to stand for formulas of ML (also called *classical formulas*). We write $P(\varphi)$ for the set of propositional variables in φ and $\varphi(p_1, ..., p_n)$ if $P(\varphi) \subseteq \{p_1, ..., p_n\}$. We write $\varphi(\psi/p)$ for the result of replacing all occurrences of p in φ by ψ .

2.2.2. DEFINITION. A (*Kripke*) model (over $X \subseteq Prop$) is a triple M = (W, R, V), where

- W is a nonempty set, whose elements are called (*possible*) worlds;
- $R \subseteq W \times W$ is a binary relation, called the *accessibility relation*;
- $-V: X \rightarrow \wp(W)$ is a function, called the *valuation*.

We call a subset $s \subseteq W$ of W a *state* on M.

For any world *w* in *M*, define, as usual, $R[w] := \{v \in W \mid wRv\}$. Similarly, for any state *s* on *M*, define $R[s] := \bigcup_{w \in S} R[w]$.

In the standard world-based Kripke semantics for modal logic, formulas are evaluated with respect to worlds: one writes $M, w \models \varphi$ if φ is true in the world w in the model M. In our state-based semantics, formulas are instead evaluated with respect to states. We will also make use of two fundamental semantic notions—*support* and *antisupport*—rather than one. As noted in Section 2.1, states in BSML represent speakers' information states. Support of a formula by (or in) a state represents that what the formula expresses is assertible given the information in the state; anti-support, similarly, represents rejectability in a state.

2.2.3. DEFINITION (Semantics). For a model M = (W, R, V) over X, a state *s* on *M*, and formula φ with $P(\varphi) \subseteq X$, the notions of φ being *supported/anti-supported* by *s* in *M*, written $M, s \models \varphi/M, s \dashv \varphi$ (or simply $s \models \varphi/s \dashv \varphi$), are defined recursively as follows:

| $M, s \vDash p$ | $: \iff$ | for all $w \in s : w \in V(p)$; |
|---|------------|--|
| $M, s \dashv p$ | : ⇔ | for all $w \in s : w \notin V(p)$; |
| $M, s \vDash \bot$ $M, s \dashv \bot$ | : ⇔ | $s = \emptyset;$ always the case; |
| $M, s \vDash \text{NE}$ $M, s \dashv \text{NE}$ | : ⇔ : ⇔ | $s \neq \emptyset;$ $s = \emptyset;$ |
| $M, s \vDash \neg \varphi$ $M, s \rightrightarrows \neg \varphi$ | :⇔ :⇔ | $egin{aligned} M, s &= oldsymbol{arphi}; \ M, s &\models oldsymbol{arphi}; \end{aligned}$ |
| $M, s \vDash \varphi \land \psi$ $M, s \dashv \varphi \land \psi$ | : ⇔ : ⇔ | $M, s \vDash \varphi$ and $M, s \vDash \psi$; there exist t, u s.t. $s = t \cup u$ and $M, t \dashv \varphi$ and $M, u \dashv \psi$; |
| $M, s \vDash \varphi \lor \psi$ $M, s \dashv \varphi \lor \psi$ | : ⇔ : ⇔ | there exist <i>t</i> , <i>u</i> s.t. $s = t \cup u$ and $M, t \models \varphi$ and $M, u \models \psi$; $M, s \dashv \varphi$ and $M, s \dashv \psi$; |
| $M, s \vDash \varphi \lor \psi$ $M, s \dashv \varphi \lor \psi$ | : ⇔ : ⇔ | $M, s \vDash \varphi \text{ or } M, s \vDash \psi;$ $M, s \dashv \varphi \text{ and } M, s \dashv \psi;$ |
| $M, s \vDash \diamondsuit \varphi$ $M, s \dashv \diamondsuit \varphi$ | :⇔ :⇔ | for all $w \in s$ there exists $t \subseteq R[w]$ s.t. $t \neq \emptyset$ and $M, t \models \varphi$; for all $w \in s : M, R[w] = \varphi$; |
| $M, s \vDash \oslash \varphi$ $M, s \dashv \oslash \varphi$ | : ⇔ : ⇔ | $M, s \vDash \varphi \text{ or } s = \emptyset;$ $M, s \dashv \varphi.$ |
For a set Φ of formulas, we write $M, s \models \Phi$ if $M, s \models \varphi$ for all $\varphi \in \Phi$. We say that Φ *entails* ψ , written $\Phi \models \psi$, if for all models M and states s on M: $M, s \models \Phi$ implies $M, s \models \psi$. We also write simply $\varphi_1, \ldots, \varphi_n \models \psi$ for $\{\varphi_1, \ldots, \varphi_n\} \models \psi$, and $\Phi, \varphi \models \psi$ for $\Phi \cup \{\varphi\} \models \psi$. We write $\models \varphi$ for $\emptyset \models \varphi$ and in this case say that φ is *valid*. If both $\varphi \models \psi$ and $\psi \models \varphi$, then we write $\varphi \equiv \psi$ and say that φ and ψ are *equivalent*. If both $\varphi \equiv \psi$ and $\neg \varphi \equiv \neg \psi$, then φ and ψ are said to be *strongly equivalent*, written $\varphi \equiv \psi$.

The box modality \Box is defined as the dual of the diamond: $\Box \varphi := \neg \Diamond \neg \varphi$; the resulting support/anti-support clauses are:

$$\begin{array}{ll} M, s \vDash \Box \varphi & \iff & \text{for all } w \in s : M, R[w] \vDash \varphi; \\ M, s \dashv \Box \varphi & \iff & \text{for all } w \in s \text{ there exists } t \subseteq R[w] \text{ s.t. } t \neq \emptyset \text{ and } M, t \dashv \varphi. \end{array}$$

We refer to the atom \perp as the *weak contradiction*. The original syntax for BSML in [6] does not include \perp , but rather defines it as $\perp := p \land \neg p$ for some fixed p. Including \perp in our syntax allows us to simplify parts of our exposition. We also define the *strong contradiction* $\parallel := \perp \land NE$, and the *strong tautology* $\pi := \neg \bot$. The weak contradiction is supported only in the empty state, the strong contradiction in no state, and the strong tautology in all states. The atom NE (supported in all nonempty states) can also be viewed as the *weak tautology* \top ; accordingly, we let $\top := NE$. Note that we have the following equivalences:

$$\neg \bot \equiv \Pi, \qquad \neg \Pi \equiv \bot, \qquad \neg \amalg \equiv \Pi, \qquad \neg \top \equiv \bot.$$

We use these contradictions and tautologies to interpret the empty disjunctions and conjunction:

$$\bigvee \varnothing := \bot, \qquad \qquad \bigvee \varnothing := \amalg, \qquad \qquad \bigwedge \varnothing := \pi.$$

We now give some examples to illustrate the semantics—consider Figure 2.3. A (local) disjunction $\varphi \lor \psi$ is supported in s if s can be split into two states t and u such that $t \models \varphi$ and $u \models \psi$. Note that one or both of these substates might be empty; for instance, in Figure 2.3(a) we have $\{w_q\} \models p \lor q$. One can force these substates to be nonempty using NE: we have $\{w_q\} \neq (p \land NE) \lor (q \land NE)$, whereas $\{w_p, w_q\} \models (p \land NE)$ NE) \lor ($q \land$ NE). A global disjunction, on the other hand, is supported if either disjunct is supported: $\{w_q\} \models (p \land NE) \lor (q \land NE)$, whereas $\{w_p, w_q\} \not\models (p \land NE) \lor (q \land NE)$. The emptiness operator \oslash is essentially a restricted variant of \lor in that $\oslash \varphi \equiv \varphi \lor \bot$: $\{w_q\} \models$ $(p \land NE) \lor (q \land NE)$ and $\{w_p, w_q\} \models (p \land NE) \lor (q \land NE)$. A diamond formula $\Diamond \varphi$ is supported in s if for each world w in s there is a nonempty substate t of the state of all worlds R[w] accessible from w that supports φ ; for example, in Figure 2.3(b) we have $s_b \models \Diamond q$ and $s_b \not\models \Diamond p$. A box formula $\Box \varphi$ is supported in s if for each $w \in s$, R[w] as a whole supports φ : $s_b \not\models \Box q$ and $s_b \models \Box p \lor \Box q$. It is easy to verify that the conjunction and (local) disjunction distribute over the global disjunction: $\varphi \land (\psi \lor \chi) \equiv$ $(\varphi \land \psi) \lor (\varphi \land \chi)$ and $\varphi \lor (\psi \lor \chi) \equiv (\varphi \lor \psi) \lor (\varphi \lor \chi)$; whereas the modalities do not distribute over \lor , as, for example, $s_b \models \Box(p \lor q)$ but $s_b \not\models \Box p \lor \Box q$. We have instead $\Diamond(\varphi \lor \psi) \equiv \Diamond \varphi \lor \Diamond \psi \text{ and } \Box(\varphi \lor \psi) \equiv \Box \varphi \lor \Box \psi.$



Figure 2.3: Examples of the semantics

Let us also revisit the pragmatic enrichment function $[]^+: ML \rightarrow BSML$ described in Section 2.1, which can now be formally and recursively defined as:

$$\begin{array}{ll} [p]^+ & := & p \wedge \operatorname{NE}; \\ [\circ \alpha]^+ & := & \circ ([\alpha]^+) \wedge \operatorname{NE} & \text{for } \circ \in \{\neg, \diamondsuit, \square\}; \\ [\alpha \bigtriangleup \beta]^+ & := & ([\alpha]^+ \bigtriangleup [\beta]^+) \wedge \operatorname{NE} & \text{for } \bigtriangleup \in \{\land, \lor\}. \end{array}$$

It is easy to see that []⁺ disallows zero-models as described in Section 2.1. Consider, for instance, $[p \lor q]^+ = ((p \land NE) \lor (q \land NE)) \land NE \equiv (p \land NE) \lor (q \land NE)$. The state $\{w_q\}$ is a zero-model for $p \lor q$ since $\{w_q\} \models p \lor q$ but $\{w_q\} \nvDash (p \land NE) \lor (q \land NE)$, whereas $\{w_p, w_q\}$ is a nonzero verifier since $\{w_p, w_q\} \models (p \land NE) \lor (q \land NE)$. For a formula $\diamond (\alpha \lor \beta)$ as in the antecedent of a narrow-scope FC-inference, we have $[\diamond (\alpha \lor \beta)]^+ \equiv$ $\diamond ((\alpha \land NE) \lor (\beta \land NE)) \land NE$. It is then also easy to see that $[\diamond (\alpha \lor \beta)]^+ \models \diamond \alpha \land \diamond \beta$ a nonzero verifier for $\diamond (\alpha \lor \beta)$ must verify $\diamond \alpha \land \diamond \beta$. In Figure 2.3, s_b is a zero-model for $\diamond (p \lor q)$ and s_c is a nonzero verifier: we have $s_b \models p \lor q, s_b \nvDash \diamond ((p \land NE) \lor (q \land NE))$, and $s_b \nvDash \diamond p \land \diamond q$; whereas $s_c \models \diamond ((p \land NE) \lor (q \land NE))$ and $s_c \models \diamond p \land \diamond q$.

Similarly, to many other logics employing team semantics (see, e.g., Ciardelli and Roelofsen [41], Yang and Väänänen [140]), BSML and its extensions do not admit uniform substitution: $\varphi(p) \models \psi(p)$ need not imply $\varphi(\chi/p) \models \psi(\chi/p)$. For instance, we have $\models p \lor \neg p$ and $p \lor p \models p$, whereas $\notin \text{NE} \lor \neg \text{NE}$ and $(p \lor \neg p) \lor (p \lor \neg p) \notin p \lor \neg p$.

Closely related is the failure of preservation of equivalence under replacement: $\varphi \equiv \psi$ need not imply $\theta(\varphi/p) \equiv \theta(\psi/p)$. For instance, we have $\neg(\varphi \lor \psi) \equiv \neg(\varphi \lor \psi)$ but $\neg \neg(\varphi \lor \psi) \equiv \varphi \lor \psi \equiv \varphi \lor \psi \equiv \neg \neg(\varphi \lor \psi)$, and $\bot \equiv \neg \operatorname{NE}$ but $\neg \bot \equiv \Pi \equiv \operatorname{NE} \equiv \neg \neg \operatorname{NE}$. This failure is occasioned by the bilateral negation: it is easy to show that if *p* is not within the scope of \neg , then it does follow that $\theta(\varphi/p) \equiv \theta(\psi/p)$. Strong equivalence, on the other hand, *is* preserved under replacement, as we prove in the following proposition (cf. a similar replacement theorem in [119, Lemma 3.25]). We write $\varphi[\chi]$ (and sometimes simply $[\chi]$ or χ) to refer to a specific occurrence of the subformula χ in φ , and $\varphi[\psi/\chi]$ for the result of replacing this occurrence of χ in φ with ψ . This notation is also extended to sequences of occurrences $\varphi[\chi_1, \dots, \chi_n]$ and sequences or replacements $\varphi[\psi_1/\chi_1, \dots, \psi_n/\chi_n]$ in the obvious way. **2.2.4.** PROPOSITION (Replacement with respect to strong equivalence). For any formulas θ , φ , and ψ , if $\varphi \models \psi$, then $\theta[\varphi/p] \models \theta[\psi/p]$.

Proof:

This is by induction on θ . We only give a detailed proof for a few interesting cases.

Case 1: $\theta = \neg \eta$. By the induction hypothesis, $\eta[\varphi/p] \equiv \eta[\psi/p]$, so $\neg \eta[\varphi/p] \equiv \neg \eta[\psi/p]$ and $\neg \neg \eta[\varphi/p] \equiv \eta[\varphi/p] \equiv \eta[\psi/p] \equiv \neg \neg \eta[\psi/p]$. Therefore, $\neg \eta[\varphi/p] \equiv \neg \eta[\psi/p]$.

Case 2: $\theta = \Diamond \eta$. By the induction hypothesis, we have $\eta[\varphi/p] \equiv \eta[\psi/p]$. To show $\Diamond \eta[\varphi/p] \models \Diamond \eta[\psi/p]$, assume $s \models \Diamond \eta[\varphi/p]$. Then for all $w \in s$ there is a nonempty $t \subseteq R[w]$ such that $t \models \eta[\varphi/p]$. But then also for each such t we have $t \models \eta[\psi/p]$, so $s \models \Diamond \eta[\psi/p]$. The proof of $\Diamond \eta[\psi/p] \models \Diamond \eta[\varphi/p]$ is analogous. To show $\neg \Diamond \eta[\varphi/p] \models \neg \langle \eta[\psi/p]$, assume $s \models \neg \Diamond \eta[\varphi/p]$. Then $R[w] \models \neg \eta[\varphi/p]$ for all $w \in s$. But then also $R[w] \models \neg \eta[\psi/p]$ for all $w \in s$, so $s \models \neg \Diamond \eta[\psi/p]$. The proof of $\neg \Diamond \eta[\psi/p] \models \neg \eta[\psi/p]$ for all $w \in s$, so $s \models \neg \Diamond \eta[\psi/p]$.

Case 3: $\theta = \oslash \eta$. By the induction hypothesis, we have $\eta[\varphi/p] \equiv \eta[\psi/p]$. We then have $\oslash \eta[\varphi/p] \equiv \eta[\varphi/p] \lor \bot \equiv \eta[\psi/p] \lor \bot \equiv \oslash \eta[\psi/p]$ and $\neg \oslash \eta[\varphi/p] \equiv \neg \eta[\varphi/p] \equiv \neg \eta[\psi/p]$. \Box

An easy induction shows that a formula φ and its negation $\neg \varphi$ are *incompatible* in the sense that if $M, s \vDash \varphi$ and $M, t \vDash \neg \varphi$, then $s \cap t = \emptyset$ (for a proof, see [12, Proposition 3.3.9]). It also appears (see Anttila [13], or Theorem 4.2.19 in Chapter 4) that one can prove the converse of (a reformulation of) this fact: if $M, s \vDash \varphi$ and $M, t \vDash \psi$ implies $s \cap t = \emptyset$, then there is a formula θ such that $\varphi \equiv \theta$ and $\psi \equiv \neg \theta$ (cf. the similar results for Henkin sentences and their contraries in Burgess [29] and for dependence logic with the dual negation in Kontinen and Väänänen [94]).

As usual, each formula can be transformed into an equivalent formula in *negation normal form*; that is, a formula in each occurrence of negation directly precedes a propositional variable p or the atom NE, or is part of the defined connective \Box .

2.2.5. FACT (Negation normal form). For any formula φ , there exists a formula φ' in negation normal form such that $\varphi \equiv \varphi'$. Moreover, if $\varphi \in BSML$, then $\varphi \equiv \varphi'$.

Proof:

Follows easily from the following (strong) equivalences:

$$\neg \neg \varphi \boxminus \varphi; \qquad \neg (\varphi \land \psi) \boxminus \neg \varphi \lor \neg \psi; \qquad \neg (\varphi \lor \psi) \equiv \neg \varphi \land \neg \psi; \\ \neg \Diamond \varphi \boxminus \Box \neg \varphi; \qquad \neg (\varphi \lor \psi); \boxminus \neg \varphi \land \neg \psi; \qquad \neg \oslash \varphi \equiv \neg \varphi. \qquad \Box$$

Throughout the article, we make extensive use of the following properties.

2.2.6. DEFINITION. We say that a formula φ

- is *downward closed*, provided that $[M, s \models \varphi \text{ and } t \subseteq s] \implies M, t \models \varphi$;
- is union closed, provided that $[M, s \models \varphi \text{ for all } s \in S \neq \emptyset] \implies M, \bigcup S \models \varphi;$
- has the *empty state property*, provided that $M, \emptyset \vDash \varphi$ for all M;
- is *flat*, provided that *M*,*s* ⊨ ϕ \iff *M*, {*w*} ⊨ ϕ for all *w* ∈ *s*.

It is easy to check that a formula is flat if and only if it is downward closed, union closed, and has the empty state property. A simple induction shows the following.

2.2.7. FACT. If a formula φ does not contain NE, then it is downward closed and has the empty state property. If φ does not contain \mathbb{V} , then it is union closed.

In particular, formulas of BSML and $BSML^{\circ}$ are union closed; formulas of ML are flat (as they are downward and union closed, and have the empty state property).

It is also easy to verify that $\Diamond \varphi$ and $\Box \varphi$ are always flat. It follows from the flatness of ML-formulas that the state-based semantics for ML correspond with the standard (single-world) Kripke semantics in the sense of the following fact.

2.2.8. FACT. For any formula $\alpha \in ML$,

 $M, s \models \alpha \quad \iff \quad M, \{w\} \models \alpha \text{ for all } w \in s \quad \iff \quad M, w \models \alpha \text{ for all } w \in s.$

Proof:

The first equivalence follows from flatness, and the second from $\{w\} \models \alpha$ iff $w \models \alpha$ (proved by an easy induction).

As an immediate consequence of Fact 2.2.8, we have that for any set $\Delta \cup \{\alpha\}$ of ML-formulas, $\Delta \vDash \alpha$ in the state-semantics sense if and only if $\Delta \vDash \alpha$ in the usual single-world-semantics sense. For this reason, when discussing ML-formulas, we simply use the notation \vDash to refer to both state-based and world-based entailment (similarly for \equiv).

2.3 Expressive Power and Normal Forms

In this section, we study the expressive power of BSML and its extensions. We measure the expressive power of the logics in terms of the *properties*—sets of pointed models—they can express. It is well known that the expressive power of classical modal logic ML is characterized by *bisimulation-invariance*: the properties expressible in ML are invariant (closed) under bisimulation. This also holds in the state-based modal setting: our logics are invariant under *state bisimulation*. In Section 2.3.1, we introduce state bisimulation and prove bisimulation-invariance; in Section 2.3.2, we apply this notion to show expressive completeness results for our logics.

2.3.1 Bisimulation

We briefly recall some standard results concerning bisimulation for classical modal logic and standard Kripke semantics. (For more comprehensive discussion, we refer the reader to, for instance, Blackburn, de Rijke, and Venema [24], Goranko and Otto [56]).

Throughout this section, we fix a finite set $X \subseteq Prop$ of propositional variables. We omit mention of X whenever doing so does not result in confusion in order to keep our notation light. A *pointed model* (over X) is a pair (M, w), where M is a model over X and $w \in W$.

2.3.1. DEFINITION (Bisimilarity). For any $k \in \mathbb{N}$ and any (M, w) (over $Y \supseteq X$) and (M', w') (over $Y' \supseteq X$), we write $M, w \rightleftharpoons_k^X M', w'$ (or simply $w \rightleftharpoons_k w'$) and say that (M, w) and (M', w') are X, *k*-bisimilar if the following recursively defined relation holds:

$$-M, w \rightleftharpoons_0^X M, w : \iff$$
 for all $p \in X$ we have $M, w \models p$ iff $M', w' \models p$.

$$-M, w \rightleftharpoons_{k+1}^{\mathsf{X}} M', w' : \longleftrightarrow M, w \rightleftharpoons_0 M', w' \text{ and }$$

- (a) [forth] for all $v \in R[w]$ there is a $v' \in R'[w']$ such that $M, v \rightleftharpoons_k M', v'$;
- (b) [back] for all $v' \in R'[w']$ there is a $v \in R[w]$ such that $M, v \rightleftharpoons_k M', v'$.

The modal depth $md(\varphi)$ of a formula φ is defined recursively as:

$$- md(p) = md(NE) = 0;$$

$$- md(\neg \phi) = md(\phi \phi) = md(\phi);$$

$$- md(\phi \land \psi) = md(\phi \lor \psi) = md(\phi \lor \psi) = \max\{md(\phi), md(\psi)\};$$

$$- md(\diamondsuit \phi) = md(\phi) + 1.$$

We say that (M, w) and (M, w') are X, *k*-equivalent $(k \in \mathbb{N})$, written $M, w \equiv_k^X M', w'$ (or simply $w \equiv_k w'$), if for all $\alpha(X) \in ML$ with $md(\alpha) \le k$: $M, w \models \alpha \iff M', w' \models \alpha$.

2.3.2. DEFINITION (Hintikka formulas). Let $k \in \mathbb{N}$, and let (M, w) be a pointed model over $Y \supseteq X$. We define the *kth Hintikka formula* $\chi_{M,w}^{X,k}$ (or simply χ_w^k) of (M, w) recursively as:

$$\chi_{M,w}^{\mathsf{X},0} \qquad := \qquad \bigwedge \{p \mid p \in \mathsf{X}, w \in V(p)\} \land \bigwedge \{\neg p \mid p \in \mathsf{X}, w \notin V(p)\}, \\ \chi_{M,w}^{\mathsf{X},k+1} \qquad := \qquad \chi_{w}^{k} \land \bigwedge_{v \in R[w]} \diamondsuit \chi_{v}^{k} \land \Box \bigvee_{v \in R[w]} \chi_{v}^{k}.$$

It is easy to see that there are only finitely many nonequivalent *k*th Hintikka formulas for a given finite X—this is why we may assume that the conjunction and the disjunction in $\chi_{M,w}^{X,k+1}$ are finite and hence that $\chi_{M,w}^{X,k+1}$ is well defined.

2.3.3. THEOREM (see [24, 56]). We have

 $w \equiv_k w' \quad \Longleftrightarrow \quad w \rightleftharpoons_k w' \quad \Longleftrightarrow \quad w' \vDash \chi_w^k \quad \Longleftrightarrow \quad \chi_w^k \equiv \chi_{w'}^k.$

In the context of standard Kripke semantics, we measure the expressive power of a logic in terms of the classes of pointed models expressible in the logic.

2.3.4. DEFINITION. A *property* (over X) is a class of pointed models over X. Each formula α *expresses* a property (over X $\supseteq P(\phi)$)

$$\llbracket \alpha \rrbracket_{\mathsf{X}} \coloneqq \{ (M, w) \text{ over } \mathsf{X} \mid M, w \vDash \alpha \}.$$

We say that a logic *L* is *expressively complete* for a class of properties \mathbb{P} , written $\llbracket L \rrbracket = \mathbb{P}$, if for each finite X, the class \mathbb{P}_X of properties over X in \mathbb{P} is precisely the class of properties over X expressible by formulas of *L* with, that is, if

$$\mathbb{P}_{\mathsf{X}} = \llbracket L \rrbracket_{\mathsf{X}} := \{ \llbracket \alpha \rrbracket_{\mathsf{X}} \mid \alpha \in L \}.$$

A property \mathcal{P} is *invariant under* X, *k*-bisimulation if $(M, w) \in \mathcal{P}$ and $M, w \rightleftharpoons_k^X M', w'$ imply $(M', w') \in \mathcal{P}$. A property \mathcal{P} over X is *invariant under bounded bisimulation* if \mathcal{P} is invariant under X, *k*-bisimulation for some $k \in \mathbb{N}$. Theorem 2.3.3 allows us to prove a world-based expressive completeness theorem for ML: ML can express all worldbased properties that are invariant under bounded bisimulation.

2.3.5. THEOREM (World-based expressive completeness of ML).

 $\llbracket ML \rrbracket = \{ \mathcal{P} \mid \mathcal{P} \text{ is invariant under bounded bisimulation} \}.$

Proof:

For the inclusion \subseteq , for any $\alpha \in ML$, we have that $[\alpha]$ is invariant under *k*-bisimulation for $k = md(\alpha)$ by Theorem 2.3.3. For the converse inclusion \supseteq , for any \mathcal{P} invariant under *k*-bisimulation, it follows easily from Theorem 2.3.3 that

$$M', w' \vDash \bigvee_{(M,w)\in\mathcal{P}} \chi_w^k \iff (M', w') \in \mathcal{P} \quad (\text{i.e.}, \mathcal{P} = \llbracket \bigvee_{(M,w)\in\mathcal{P}} \chi_w^k \rrbracket \in \llbracket \mathrm{ML} \rrbracket).$$

Note that since there are only finitely many nonequivalent Hintikka formulas χ_w^k for a given finite X, we may assume that the disjunction in the above formula $\bigvee_{(M,w)\in\mathcal{P}} \chi_w^k$ is finite and hence that the formula is well defined.

The formula $\bigvee_{(M,w)\in\mathcal{P}} \chi_w^k$ in the above proof can be viewed as a characteristic formula for \mathcal{P} . The proof also yields a disjunctive normal form for formulas of ML: given that $\llbracket \alpha \rrbracket$ is invariant under *k*-bisimulation for $k = md(\alpha)$, $\llbracket \alpha \rrbracket = \llbracket \bigvee_{(M,w)\in\llbracket \alpha \rrbracket} \chi_w^{md(\alpha)} \rrbracket$, that is, $\alpha \equiv \bigvee_{(M,w)\in\llbracket \alpha \rrbracket} \chi_w^{md(\alpha)}$.

We now introduce state-based analogues of the preceding notions and results. A *pointed (state) model* (over X) is a pair (M,s), where M is a model over X and s is a state on M. For a given $k \in \mathbb{N}$, (M,s) and (M,s') are X,*k-equivalent* (in a logic L), written $M, s \equiv_{k}^{X} M', s'$, if for all $\varphi(X) \in L$ with $md(\varphi) \leq k : M, s \models \varphi \iff M', s' \models \varphi$.

State-based bisimulation (first introduced in [71, 89]) is a natural generalization of the world-based notion: two pointed state models are bisimilar if for each world in one there is a bisimilar world in the other, and vice versa.

2.3.6. DEFINITION (State bisimilarity). For any $k \in \mathbb{N}$, (M, s) (over $Y \supseteq X$) and (M', s') (over $Y' \supseteq X$) are X, *k*-bisimilar, written $M, s \rightleftharpoons_k^X M', s'$ (or simply $s \rightleftharpoons_k s'$), if

- [forth] for each $w \in s$ there is some $w' \in s'$ such that $M, w \rightleftharpoons_k M', w'$;
- [back] for each $w' \in s'$ there is some $w \in s$ such that $M, w \rightleftharpoons_k M', w'$.

State bisimulation is clearly a conservative extension of world-based bisimulation in that $\{w\} \rightleftharpoons_k \{w'\}$ iff $w \rightleftharpoons_k w'$. We thus use the same symbol \rightleftharpoons_k for both world-based and state based bisimulation, and we sometimes write simply $w \rightleftharpoons_k s'$ for $\{w\} \rightleftharpoons_k s'$. It is easy to see that for any $k \in \mathbb{N}$, if $s \rightleftharpoons_k s'$, then $s \rightleftharpoons_n s'$ for all n < k (for $w \rightleftharpoons_k w'$ similarly implies $w \rightleftharpoons_n w'$ for all n < k).

We list below some useful properties of state bisimulation (see [71] for a proof).

2.3.7. LEMMA. If $s \rightleftharpoons_{k+1} s'$, then:

- (*i*) $R[s] \rightleftharpoons_k R'[s'];$
- (ii) for all $t, u \subseteq s$ such that $s = t \cup u$ there are $t', u' \subseteq s'$ such that $s' = t' \cup u', t \rightleftharpoons_{k+1} t',$ and $u \rightleftharpoons_{k+1} u'$.

As expected, state k-bisimilarity implies k-equivalence in each of the three logics.

2.3.8. THEOREM (Bisimulation invariance). If $s \rightleftharpoons_k s'$, then $s \equiv_k s'$.

Proof:

For any φ , we show by induction on φ that if $s \rightleftharpoons_k s'$ for $k = md(\varphi)$, then $s \vDash \varphi$ iff $s' \vDash \varphi$. By Fact 2.2.5, we may assume that φ is in negation normal form. Most cases can be found in [71]; note that the \lor -case follows by Lemma 2.3.7(ii). The case $\varphi = \oslash \psi$ follows by $\oslash \psi \equiv \psi \lor \neg NE$ and the other cases. We now show the cases for the modalities. We only give the detailed proof for one direction of each implication.

Case 1: $\varphi = \Diamond \psi$. Suppose that $s \rightleftharpoons_k s'$ for $k = md(\Diamond \psi) = md(\psi) + 1$, and $s \models \Diamond \psi$. Let $w' \in s'$. By $s \rightleftharpoons_k s'$, there is some $w \in s$ such that $w \rightleftharpoons_k w'$. Since $s \models \Diamond \psi$, there is a nonempty $t \subseteq R[w]$ such that $t \models \psi$. By $w \rightleftharpoons_k w'$, we obtain $R[w] \rightleftharpoons_{k-1} R'[w']$ by Lemma 2.3.7(i). Then by Lemma 2.3.7(ii), there is some $t' \subseteq R'[w']$ such that $t \rightleftharpoons_{k-1} t'$. By the induction hypothesis, $t' \models \psi$. Since $t \neq \emptyset$ and $t \rightleftharpoons_{k-1} t'$, we also have $t' \neq \emptyset$. Hence, $s' \models \Diamond \psi$. *Case 2*: $\varphi = \Box \psi$. Suppose that $s \rightleftharpoons_k s'$ for $k = md(\Box \psi) = md(\psi) + 1$, and $s \models \Box \psi$. Let $w' \in s'$. By $s \rightleftharpoons_k s'$, there is a $w \in s$ such that $w \rightleftharpoons_k w'$. Since $s \models \Box \psi$, we have $R[w] \models \psi$. By $w \rightleftharpoons_k w'$, we obtain $R[w] \rightleftharpoons_{k-1} R'[w']$ by Lemma 2.3.7(i). Then by the induction hypothesis, $R'[w'] \models \psi$, so $s' \models \Box \psi$.

2.3.2 Expressive Completeness

We move on to the main results concerning the expressive power of our logics. In the classical world-based setting we measured the expressive power of a logic in terms of the classes of pointed models expressible in the logic; we now use classes of pointed state models.

2.3.9. DEFINITION. A *state property* (over X) is a class of pointed state models over X. Each formula φ *expresses* a property (over $X \supseteq P(\varphi)$)

$$\|\varphi\|_{\mathsf{X}} \coloneqq \{(M,s) \text{ over } \mathsf{X} \,|\, M, s \vDash \varphi\}.$$

We say that a logic *L* is *expressively complete* for a class of properties \mathbb{P} , written $||L|| = \mathbb{P}$, if for each finite X, the class \mathbb{P}_X of properties over X in \mathbb{P} is precisely the class of properties over X expressible by formulas of *L*, that is, if

$$\mathbb{P}_{\mathsf{X}} = \|L\|_{\mathsf{X}} \coloneqq \{\|\varphi\|_{\mathsf{X}} \mid \varphi \in L\}.$$

We say that a state property \mathcal{P} is *invariant under* X,*k-bisimulation* ($k \in \mathbb{N}$) if

$$[M, s \rightleftharpoons_k^{\mathsf{X}} M', s'] \Longrightarrow (M', s') \in \mathcal{P},$$

and that \mathcal{P} over X is *invariant under bounded bisimulation* if \mathcal{P} is invariant under X, *k*-bisimulation for some $k \in \mathbb{N}$. Similarly, \mathcal{P} is *union closed* if

$$[(M,s) \in \mathcal{P} \text{ for all } s \in S \neq \emptyset] \Longrightarrow (M, \bigcup S) \in \mathcal{P},$$

and \mathcal{P} is *flat* if

$$(M,s) \in \mathcal{P} \iff (M,\{w\}) \in \mathcal{P}$$
 for all $w \in s$.

Clearly, \mathcal{P} is flat if and only if \mathcal{P} is union closed and *downward closed* (i.e., $(M, s) \in \mathcal{P}$ and $t \subseteq s$ imply $(M, t) \in \mathcal{P}$), and has the *empty state property* (i.e., $(M, \emptyset) \in \mathcal{P}$ for all M).

We aim to show three results. First, $BSML^{\vee}$ is expressively complete for

 $\mathbb{B} \coloneqq \{\mathcal{P} \mid \mathcal{P} \text{ is invariant under bounded bisimulation}\}.$

Second, $BSML^{\circ}$ is complete for

 $\mathbb{U} := \{\mathcal{P} \mid \mathcal{P} \text{ is union closed and invariant under bounded bisimulation}\}.$

Third, although the properties expressed by formulas of BSML are union closed and invariant under bounded bisimulation, BSML is not complete for \mathbb{U} . Using the notions we have introduced, it is also easy to show that ML is complete for

 $\mathbb{F} := \{ \mathcal{P} \mid \mathcal{P} \text{ is flat and invariant under bounded bisimulation} \}.$

We include below a brief proof of this folklore result as seeing it alongside the other expressive completeness theorems is instructive. Writing $||L_1|| \subset ||L_2||$ if $||L_1||_X \subseteq ||L_2||_X$ for all finite X and $||L_2||_Y \notin ||L_1||_Y$ for some finite Y, we therefore have

$$\|\mathbf{ML}\| \subset \|\mathbf{BSML}\| \subset \|\mathbf{BSML}^{\oslash}\| \subset \|\mathbf{BSML}^{\bigtriangledown}\|$$

We begin by using Hintikka formulas $\chi_w^k \in ML$ for pointed models to construct Hintikka formulas $\chi_s^k \in ML$ and $\theta_s^k \in BSML$ for pointed state models.

2.3.10. DEFINITION (Hintikka formulas for states). For any pointed state model (M, s) over $Y \supseteq X$ and any $k \in \mathbb{N}$, the *kth Hintikka formula* $\chi_{M,s}^{X,k}$ (or simply χ_s^k) and the *kth strong Hintikka formula* $\theta_{M,s}^{X,k}$ (or simply θ_s^k) of (M, s) are defined as

$$\chi_{M,s}^{\mathbf{X},k} \coloneqq \bigvee_{w \in s} \chi_{w}^{k},$$
$$\theta_{M,s}^{\mathbf{X},k} \coloneqq \bigvee_{w \in s} (\chi_{w}^{k} \wedge \mathrm{NE}).$$

Recall that we stipulate that $\bigvee \emptyset = \bot$. Since there are only finitely many nonequivalent χ_w^k for a given finite X, we may assume that the disjunctions in θ_s^k and χ_w^k are finite, and hence that the formulas are well defined. These formulas function as characteristic formulas for states in the sense of the following proposition.

2.3.11. PROPOSITION.

(*i*)
$$M', s' \models \chi_s^k \iff$$
 there is a state $t \subseteq s$ such that $M, t \rightleftharpoons_k M', s'$
(*ii*) $M', s' \models \theta_s^k \iff M, s \rightleftharpoons_k M', s'.$

Proof:

We prove the two items simultaneously. If $s = \emptyset$, then $\theta_{\emptyset}^k = \chi_{\emptyset}^k = \bot$, and $s' \models \bot \iff s' = \emptyset \iff s \rightleftharpoons_k s' \iff \exists t \subseteq s : t \rightleftharpoons_k s'$. Now assume that $s \neq \emptyset$.

 \Leftarrow : We first prove (i). Suppose that $t \rightleftharpoons_k s'$ for some $t \subseteq s$. Then for any $w' \in s'$, there exists $w \in t \subseteq s$ such that $w \rightleftharpoons_k w'$, which, by Theorem 2.3.3, implies that $w' \models \chi_w^k$ and so $w' \models \chi_s^k$. Since the classical formula χ_s^k is flat (see Fact 2.2.7), we conclude that $s' \models \chi_s^k$.

For (ii), suppose $s \rightleftharpoons_k s'$. First observe that the back condition for $s \rightleftharpoons_k s'$ implies that there is some $t \subseteq s$ such that $t \rightleftharpoons_k s'$. Thus, by (i), we have $s' \models \chi_s^k$, which means that for every $w \in s$, there exists $s'_w \subseteq s'$ such that $s' = \bigcup_{w \in s} s'_w$ and $s'_w \models \chi_w^k$. Meanwhile,

| Hintikka formula | Logic | Characterization effect | Defined in |
|---|--|---|-------------------|
| χ^k_w | ML | $w' \vDash \chi_w^k \iff w \rightleftharpoons_k w'$ | Definition 2.3.2 |
| $\chi_s^k = \bigvee_{w \in s} \chi_w^k$ | ML | $s' \vDash \chi_s^k \iff t \rightleftharpoons_k s'$ | Definition 2.3.10 |
| | | for some $t \subseteq s$ | |
| $\theta_s^k = \bigvee_{w \in s} (\chi_w^k \wedge \text{NE})$ | BSML | $s' \vDash \theta_s^k \iff s \rightleftharpoons_k s'$ | Definition 2.3.10 |
| | | | |
| Normal form | Logic | Characterization effect | Defined in |
| $v_{\mathcal{P}}^k = \bigvee_{(M,s)\in\mathcal{P}} \chi_s^k$ | ML | $M, s \models v_{\mathcal{P}}^k \iff$ | Definition 2.3.12 |
| | | $(M,s)\in\mathcal{P}\ (\mathcal{P}\in\mathbb{F})$ | |
| $\xi_{\mathcal{P}}^{k} = \bigvee_{(M,s)\in\mathcal{P}} \theta_{s}^{k}$ | $BSML^{\!$ | $M, s \models \xi_{\mathcal{P}}^k \iff$ | Definition 2.3.14 |
| | | $(M,s) \in \mathcal{P} \ (\mathcal{P} \in \mathbb{B})$ | |
| $\zeta_{\mathcal{P}}^{k} = \bigvee_{(M,s)\in\mathcal{P}} \oslash \boldsymbol{\theta}_{s}^{k}$ | $BSML^{\oslash}$ | $M, s \vDash \zeta_{\mathcal{P}}^k \iff$ | Definition 2.3.16 |
| | | $(M,s) \in \mathcal{P} \ (\mathcal{P} \in \mathbb{U}^{\varnothing} \subseteq \mathbb{U})$ | |
| NE $\wedge \zeta_{\mathcal{P}}^k$ | $BSML^{\oslash}$ | $M, s \vDash \operatorname{NE} \land \zeta_{\mathcal{P}}^k \iff$ | Definition 2.3.16 |
| , | | $(M,s) \in \mathcal{P} \ (\dot{\mathcal{P}} \in \mathbb{U}^{NE} \subseteq \mathbb{U})$ | |

Table 2.2: Characteristic formulas

by the forth condition for $s \rightleftharpoons_k s'$, we know that for every $w \in s$, there exists $w' \in s'$ such that $w \rightleftharpoons_k w'$, which implies that $w' \models \chi_w^k$ by Theorem 2.3.3. Thus, we may assume without loss of generality that each $s'_w \neq \emptyset$ (for we can simply include the world w' in s'_w), giving that $s'_w \models \chi_w^k \land NE$. Hence, $s' \models \theta_s^k$.

 \implies : For (i), suppose $s' \models \chi_s^k$. Then, for each $w \in s$, there exists a subset $s'_w \subseteq s'$ such that $s' = \bigcup_{w \in s} s'_w$ and $s'_w \models \chi_w^k$. By the empty state property of classical formulas and Theorem 2.3.3, we have that either $s'_w = \emptyset$ or $w \rightleftharpoons_k s'_w$. Let $t = \{w \in s \mid s'_w \neq \emptyset\} \subseteq s$. Clearly, the forth condition for $t \rightleftharpoons_k s'$ is satisfied. To verify the back condition, observe that for any $v' \in s' = \bigcup_{w \in s} s'_w$, there exists $w \in s$ such that $v' \in s'_w \neq \emptyset$. Thus, $w \in t$ and $w \rightleftharpoons_k s'_w$, whereby $w \rightleftharpoons_k v'$.

For (ii), suppose $s' \models \theta_s^k$. Then, for every $w \in s$, there exists $s'_w \subseteq s'$ such that $s' = \bigcup_{w \in s} s'_w$ and $s'_w \models \chi^k_w \land NE$. We then have $s'_w \neq \emptyset$ and $w \rightleftharpoons_k s'$ (by Theorem 2.3.3 again). Analogous to the above proof for (i), we define $t = \{w \in s \mid s'_w \neq \emptyset\} \subseteq s$ and obtain $t \rightleftharpoons_k s'$ by the same argument. But since now each $s'_w \neq \emptyset$, we have actually t = s and thus $s \rightleftharpoons_k s'$.

The formulas χ_s^k are discussed also in [71], and the formulas χ_s^k and θ_s^k are modal versions of the propositional characteristic formulas defined by the third author and Väänänen in [139] and [140]. Table 2.2 presents a summary of the different characteristic formulas we consider.

Table 2.2 also illustrates the main results of the remaining part of this section: we will use Hintikka formulas χ_s^k and θ_s^k for states to construct characteristic formulas for different types of state properties, and the formula for a given type of property will, in turn, be used to show that the logic it belongs to is expressively complete with respect

to that type of property. Each of the characteristic formulas for a type of property also exemplifies the disjunctive normal form for the logic in question. Similar disjunctive normal forms are studied in the modal context in, for example, [71], [73], and [89]; and in the propositional context in, for example, [41], [139], and [140].

We now give the proofs of our expressive completeness results in order. We start with the folklore result that ML is complete for \mathbb{F} . Hereafter, we often abbreviate a pointed state model (M,s) in a property \mathcal{P} simply as s.

2.3.12. DEFINITION (Characteristic formulas for properties in \mathbb{F}). For any $\mathcal{P} \in \mathbb{F}_X$ and $k \in \mathbb{N}$, the *kth characteristic formula* $v_{\mathcal{P}}^{X,k} \in ML$ of \mathcal{P} is defined as

$$\boldsymbol{v}_{\mathcal{P}}^{\mathsf{X},k} \coloneqq \bigvee_{s \in \mathcal{P}} \boldsymbol{\chi}_{s}^{k}.$$

As before, the formula is clearly well defined given a specific finite X.

2.3.13. PROPOSITION (ML is expressively complete for \mathbb{F}).

 $\|\mathbf{ML}\| = \mathbb{F} := \{\mathcal{P} \mid \mathcal{P} \text{ is flat and invariant under bounded bisimulation}\}.$

Proof:

The inclusion \subseteq follows from Theorem 2.3.8 and Fact 2.2.7. For the converse inclusion \supseteq , we show that for any $\mathcal{P} \in \mathbb{F}$ invariant under *k*-bisimulation,

$$M', s' \vDash \bigvee_{(M,s)\in\mathcal{P}} \chi_s^k \iff (M', s') \in \mathcal{P} \quad (\text{i.e.}, \mathcal{P} = \| v_{\mathcal{P}}^k \| \in \| \mathrm{ML} \|).$$

 \Leftarrow : Suppose $s' \in \mathcal{P}$. Since $s' \rightleftharpoons_k s'$, we have $s' \vDash \chi_{s'}^k$ by Proposition 2.3.11(i), which implies that $s' \vDash \bigvee_{s \in \mathcal{P}} \chi_s^k$ by the empty state property of χ_s^k .

 $\implies: \text{Suppose } s' \models \bigvee_{s \in \mathcal{P}} \chi_s^k. \text{ Then for each } s \in \mathcal{P}, \text{ there exists } t'_s \subseteq s' \text{ such that } s' = \bigcup_{s \in \mathcal{P}} t'_s \text{ and } t'_s \models \chi_s^k. \text{ The latter means, by Proposition 2.3.11(i), that there exists } t \subseteq s \text{ such that } t'_s \rightleftharpoons_k t. \text{ Since } \mathbb{F} \text{ is downward closed and invariant under } k-\text{bisimulation, we have } t'_s \in \mathcal{P} \text{ for each } s \in \mathcal{P}. \text{ Finally, since } \mathcal{P} \text{ is union closed, and also using the fact that since } \mathcal{P} \text{ has the empty state property, it is not the empty property, we conclude that } s' = \bigcup_{s \in \mathcal{P}} t'_s \in \mathcal{P}. \square$

The theorem also yields an alternative normal form for formulas in ML: for an arbitrary ML-formula α , since $\|\alpha\|$ is invariant under *k*-bisimulation for $k = md(\alpha)$, we have $\|\alpha\| = \|v_{\|\alpha\|}^{md(\alpha)}\|$, that is, $\alpha \equiv \bigvee_{s \in \|\alpha\|} \chi_s^{md(\alpha)}$.

Next, we show the first of our main results: $BSML^{\vee}$ is complete for \mathbb{B} , and is thus the expressively strongest logic. The proof is similar to the one given for ML and \mathbb{F} , but the addition of NE and \vee allows us to fully characterize states and to capture all properties invariant under bounded bisimulation.

2.3.14. DEFINITION (Characteristic formulas for properties in \mathbb{B}). For any $\mathcal{P} \in \mathbb{B}_X$ and $k \in \mathbb{N}$, the *kth characteristic formula* $\xi_{\mathcal{P}}^{X,k} \in \text{BSML}^{\mathbb{V}}$ of \mathcal{P} is defined as

$$\boldsymbol{\xi}_{\mathcal{P}}^{\mathcal{X},k} \coloneqq \bigvee_{s \in \mathcal{P}} \boldsymbol{\theta}_{s}^{k}.$$

Recall that we stipulate that $W \varnothing = \bot$.

2.3.15. THEOREM (BSML^{\vee} is expressively complete for \mathbb{B}).

 $\|BSML^{\vee}\| = \mathbb{B} = \{\mathcal{P} \mid \mathcal{P} \text{ is invariant under bounded bisimulation}\}.$

Proof:

The inclusion \subseteq follows from Theorem 2.3.8. For the nontrivial inclusion \supseteq , we show that for any $\mathcal{P} \in \mathbb{B}$ invariant under *k*-bisimulation,

$$M', s' \models \bigvee_{(M,s)\in\mathcal{P}} \theta_s^k \iff (M',s')\in\mathcal{P} \quad (\text{i.e.}, \mathcal{P} = \|\xi_{\mathcal{P}}^k\| \in \|\text{BSML}^{\vee}\|).$$

If $\mathcal{P} = \emptyset$, then $\xi_{\mathcal{P}}^k = \mathbb{I}$, and thus $\mathcal{P} = \|\mathbb{I}\| = \|\xi_{\mathcal{P}}^k\|$. Now, suppose $\mathcal{P} \neq \emptyset$. If $s' \models \bigcup_{s \in \mathcal{P}} \theta_s^k$, then $s' \models \theta_s^k$ for some $s \in \mathcal{P}$. By Proposition 2.3.11(ii), we have $s \rightleftharpoons_k s'$, which implies that $s' \in \mathcal{P}$ by invariance under *k*-bisimulation. Conversely, if $s' \in \mathcal{P}$, since $s' \rightleftharpoons_k s'$, we have $s' \models \theta_{s'}^k$ by Proposition 2.3.11(ii) and so $s' \models \bigcup_{s \in \mathcal{P}} \theta_s^k$. \Box

As above, this also provides us with a disjunctive normal form: for each BSML^V - formula φ : $\|\varphi\| = \|\xi_{\|\varphi\|}^{md(\varphi)}\|$, that is, $\varphi \equiv \bigvee_{s \in \|\varphi\|} \theta_s^{md(\varphi)}$.

The expressive completeness of $BSML^{\circ}$ for \mathbb{U} is proved in a similar way, but there is a small twist in this case. First define

 $\mathbb{U}^{\varnothing} := \{ \mathcal{P} \in \mathbb{U} \mid \mathcal{P} \text{ has the empty state property} \},\$ $\mathbb{U}^{\text{NE}} := \{ \mathcal{P} \in \mathbb{U} \mid \mathcal{P} \text{ does not have the empty state property} \}$

so that $\mathbb{U} = \mathbb{U}^{\varnothing} \cup \mathbb{U}^{\text{NE}}$. Note that if $\mathcal{P} \in \mathbb{U}^{\text{NE}}$ so that for some *M* we have $(M, \emptyset) \notin \mathcal{P}$, then in fact by invariance under *k*-bisimulation $(M', \emptyset) \notin \mathcal{P}$ for all *M'*.

2.3.16. DEFINITION (Characteristic formulas for properties in \mathbb{U}). For any $\mathcal{P} \in \mathbb{U}_X$ and $k \in \mathbb{N}$, let

$$\zeta_{\mathcal{P}}^{\mathsf{X},k} \coloneqq \bigvee_{s \in \mathcal{P}} \oslash \boldsymbol{\theta}_{s}^{k}.$$

If $\mathcal{P} \in \mathbb{U}^{\varnothing}$, then the *characteristic formula* of \mathcal{P} is $\zeta_{\mathcal{P}}^k$. If $\mathcal{P} \in \mathbb{U}^{NE}$, then the *characteristic formula* of \mathcal{P} is NE $\land \zeta_{\mathcal{P}}^k$.

2.3.17. THEOREM (BSML^{\circ} is expressively complete for U).

 $\|BSML^{\otimes}\| = \mathbb{U}$ $= \{\mathcal{P} \mid \mathcal{P} \text{ is union closed and invariant under bounded bisimulation} \}.$

Proof:

The inclusion \subseteq follows from Fact 2.2.7 and Theorem 2.3.8. For the nontrivial inclusion \supseteq , we first show that for any $\mathcal{P} \in \mathbb{U}^{\emptyset}$ invariant under *k*-bisimulation,

$$M', s' \models \bigvee_{(M,s)\in\mathcal{P}} \oslash \theta_s^k \iff (M', s') \in \mathcal{P} \quad (\text{i.e.}, \mathcal{P} = \|\zeta_{\mathcal{P}}^k\| \in \|\text{BSML}^{\oslash}\|).$$

 $\stackrel{\longleftarrow}{\longleftarrow} : \text{Suppose that } s' \in \mathcal{P} \in \mathbb{U}^{\varnothing}. \text{ Since } s' \rightleftharpoons_k s', \text{ by Proposition 2.3.11(ii), we have} \\ s' \vDash \theta_{s'}^k, \text{ whereby } s' \vDash \oslash \theta_{s'}^k. \text{ Hence, } s' \vDash \bigvee_{s \in \mathcal{P}} \oslash \theta_s^k \text{ by the empty state property of } \oslash \theta_s^k. \\ \stackrel{\longleftarrow}{\Longrightarrow} : \text{ Suppose that } s' \vDash \bigvee_{s \in \mathcal{P}} \oslash \theta_s^k. \text{ Then for each } s \in \mathcal{P}, \text{ there exists } t'_s \subseteq s' \text{ such} \end{cases}$

 \implies : Suppose that $s' \models \bigvee_{s \in \mathcal{P}} \otimes \theta_s^k$. Then for each $s \in \mathcal{P}$, there exists $t'_s \subseteq s'$ such that $s' = \bigcup_{s \in \mathcal{P}} t'_s$ and $t'_s \models \otimes \theta_s^k$. The latter implies, by Proposition 2.3.11(ii), that $t'_s = \emptyset$ or $t'_s \rightleftharpoons_k s$. Since $\mathcal{P} \in \mathbb{U}^{\emptyset}$ has the empty state property, and since it is invariant under *k*-bisimulation, we have $t'_s \in \mathcal{P}$ for each $s \in \mathcal{P}$. Finally, since \mathcal{P} is union closed, and also using the fact that since \mathcal{P} has the empty state property, it is not the empty property, we conclude that $s' = \bigcup_{s \in \mathcal{P}} t'_s \in \mathcal{P}$.

Next, we show that for any $\mathcal{P} \in \mathbb{U}^{NE}$ invariant under *k*-bisimulation,

$$M', s' \vDash \operatorname{NE} \land \bigvee_{(M,s)\in\mathcal{P}} \oslash \theta_s^k \iff (M',s') \in \mathcal{P}$$

(i.e., $\mathcal{P} = \|\operatorname{NE} \land \zeta_{\mathcal{P}}^k\| \in \|\operatorname{BSML}^{\oslash}\|).$

If $\mathcal{P} = \emptyset$, then $\zeta_{\mathcal{P}}^k = \bot$, and thus $\mathcal{P} = ||\bot|| = ||NE \land \bot|| = ||NE \land \zeta_{\mathcal{P}}^k||$. Now, suppose $\mathcal{P} \neq \emptyset$.

 \implies : Suppose $s' \in \mathcal{P} \in \mathbb{U}^{NE}$. By the same argument as in the \mathbb{U}^{\varnothing} -case, we obtain $s' \models \bigvee_{s \in \mathcal{P}} \oslash \Theta_s^k$. Moreover, since $\mathcal{P} \in \mathbb{U}^{NE}$, we have $s' \neq \varnothing$ so $s' \models NE$.

 $\iff: \text{Suppose } s' \models \text{NE} \land \bigvee_{s \in P} \oslash \theta_s^k. \text{ As in the } \mathbb{U}^{\varnothing}\text{-case, for each } s \in \mathcal{P}\text{, there exists}$ $t'_s \subseteq s' \text{ such that } s' = \bigcup_{s \in \mathcal{P}} t'_s \text{ and } t'_s = \emptyset \text{ or } t'_s \rightleftharpoons_k s. \text{ Let } \mathcal{Q} \coloneqq \{s \in \mathcal{P} \mid t'_s \rightleftharpoons_k s\} \text{ so that } s' = \bigcup_{s \in \mathcal{Q}} t'_s. \text{ Since } \mathcal{P}\text{ is invariant under } k\text{-bisimulation, we have } t'_s \in \mathcal{P}\text{ for each } s \in \mathcal{Q}\text{,}$ and since $s' \models \text{NE}$, we have $s' \neq \emptyset$ so that $\mathcal{Q} \neq \emptyset$. Finally, since \mathcal{P} is union closed and $\mathcal{Q} \neq \emptyset$, we conclude $s' = \bigcup_{s \in \mathcal{Q}} t'_s \in \mathcal{P}$.

Again the proof yields a disjunctive normal form: for any BSML^{\oslash}-formula φ , either $\varphi \equiv \zeta_{\|\varphi\|}^{md(\varphi)}$ or $\varphi \equiv \text{NE} \land \zeta_{\|\varphi\|}^{md(\varphi)}$.

Finally, we remark that while BSML is union closed and invariant under bisimulation, it is not complete for \mathbb{U} . To prove this, we show first that for BSML-formulas, the empty state property implies the downward closure property. Note that while converse might also appear to hold, it does not—for instance, the formula μ is downward closed but does not have the empty state property.

2.3.18. LEMMA. For any $\varphi \in BSML$, if φ has the empty state property, then φ is downward closed.

Proof:

This is by induction on φ (assumed to be in negation normal form). We only show the nontrivial cases.

Let $\varphi = \psi \land \chi$. If $M, \emptyset \models \psi \land \chi$, then $M, \emptyset \models \psi$ and $M, \emptyset \models \chi$. Therefore, by the induction hypothesis, ψ and χ are downward closed. So if $s \models \psi \land \chi$ and $t \subseteq s$, then $t \models \psi$ and $t \models \chi$, and therefore $t \models \psi \land \chi$.

Let $\varphi = \psi \lor \chi$. If $M, \emptyset \models \psi \lor \chi$, then clearly $M, \emptyset \models \psi$ and $M, \emptyset \models \chi$. So by the induction hypothesis, ψ and χ are downward closed. If $s \models \psi \lor \chi$, and $t \subseteq s$, then there are s_1, s_2 such that $s = s_1 \cup s_2, s_1 \models \psi$ and $s_2 \models \chi$. By downward closure, we have $t \cap s_1 \models \psi$ and $s \cap t_2 \models \chi$ and therefore $t = (t \cap s_1) \cup (t \cap s_2) \models \psi \lor \chi$.

2.3.19. FACT (BSML is not expressively complete for U).

 $\|BSML\| \subset \mathbb{U}$ = { $\mathcal{P} \mid \mathcal{P}$ is invariant under bounded bisimulation and union closed}.

Proof:

The inclusion \subseteq follows from Fact 2.2.7 and Theorem 2.3.8. For inequality, let

$$\mathcal{P} \coloneqq \|(p \land \mathrm{NE}) \lor (\neg p \land \mathrm{NE})\| \cup \|\bot\|.$$

Clearly, $\mathcal{P} \in \mathbb{U}$. Assume for contradiction that $\|\varphi\| = \mathcal{P}$ for some φ in BSML. Then φ has the empty state property so it is downward closed by Lemma 2.3.18. Let $(M, s) \in \|(p \land \text{NE}) \lor (\neg p \land \text{NE})\| \subseteq \mathcal{P}$. Then there are t, u such that $s = t \cup u, t \models p \land \text{NE}$, and $u \models \neg p \land \text{NE}$. By $\|\varphi\| = \mathcal{P}$, we have $s \models \varphi$. By downward closure, $t \models \varphi$, so $(M, t) \in \mathcal{P}$. But $t \nvDash (p \land \text{NE}) \lor (\neg p \land \text{NE})$ and $t \nvDash \bot$; a contradiction.

The property $\mathcal{P} \in \mathbb{U}$ in the above proof is clearly expressible in the expressively complete logic BSML^{\circ} as $\mathcal{P} = \| \oslash ((p \land NE) \lor (\neg p \land NE)) \|$.

We conclude this section by noting some consequences of the expressive power results. First, it is shown by D'Agostino in [43] that any state-based modal logic with the locality property and which is also *forgetting* enjoys *uniform interpolation*. Each of our logics has the locality property, and it follows from our expressive completeness results that BSML^{\vee} and BSML^{\otimes} are forgetting. Therefore BSML^{\vee} and BSML^{\otimes} enjoy uniform interpolation. (See [43] for details; for the locality requirement, see Yang [137].)

Second, the finite model property for our logics follows from that for ML, by a simple argument that makes use of disjoint unions of models. The *disjoint union* $\biguplus_{i \in I} M_i$ of the models $\{M_i \mid i \in I \neq \emptyset\}$ is defined as usual. Recall in particular that the domain of the disjoint union is defined as $\bigcup_{i \in I} (W_i \times \{i\})$. We also extend the notion to state properties in a natural way: the *disjoint union* of a nonempty property $\mathcal{P} = \{(M_i, s_i) \mid i \in I\}$ is $\biguplus \mathcal{P} := (\biguplus_{i \in I} M_i, \biguplus_{i \in I} s_i)$, where $\biguplus_{i \in I} s_i := \bigcup_{i \in I} (s_i \times \{i\})$. To simplify notation, we define the disjoint union \oiint of the empty property to be (M, \emptyset) for some fixed M. We refer to an element (w, i) in the disjoint union as simply w, and similarly for states. The following is standard (see, e.g., [24, 56]).

2.4. Axiomatizations

2.3.20. PROPOSITION. For all $i \in I, w \in W_i, k \in \mathbb{N}, X \subseteq \text{Prop} : M_i, w \rightleftharpoons_k^X \bigcup_{i \in I} M_i, w$. Consequently, for all $s \subseteq W_i : M_i, s \rightleftharpoons_k \bigcup_{i \in I} M_i, s$.

2.3.21. PROPOSITION (Finite model property). *If* $\neq \varphi$, *then there is some finite model M and state s such that* $M, s \neq \varphi$.

Proof:

Let (M, s) be such that $M, s \neq \varphi$, and let $X = P(\varphi)$ and $k = md(\varphi)$. If $s = \emptyset$, then we clearly have $M', s \neq \varphi$ for all finite M'. Now assume that $s \neq \emptyset$. Since there are only finitely many X,*k*-bisimilarity types (or Hintikka formulas), we can pick a finite $t \subseteq s$ such that $t \rightleftharpoons_k^X s$. Clearly for any $w \in t$ we have $\chi_w^k \not\equiv \bot$; therefore, by the finite model property for ML, there is some finite (M_w, v_w) such that $M_w, v_w \models \chi_w^k$. Then by Theorem 2.3.3, $M_w, v_w \rightleftharpoons_k M, w$. By Proposition 2.3.20, $\biguplus_{w \in t} M_w, v_w \rightleftharpoons_k M_w, v_w$ for all $w \in t$; then also $\biguplus_{w \in t} M_w, v_w \rightleftharpoons_k M, w$ for all $w \in t$, so $\biguplus_{w \in t} M_w, \{v_w \mid w \in t\} \rightleftharpoons_k M, t$, whence $\biguplus_{w \in t} M_w, \{v_w \mid w \in t\} \neq \varphi$ by Theorem 2.3.8, where $\biguplus_{w \in t} M_w$ is clearly finite. \Box

Together with the completeness results to be proved in Section 2.4, this implies that our logics BSML, $BSML^{\emptyset}$, and $BSML^{\mathbb{W}}$ are decidable.

2.4 Axiomatizations

In this section, we introduce sound and complete natural deduction systems for BSML and its extensions. The strategy we employ in proving completeness makes essential use of the expressive power results in Section 2.3: for the expressively complete logics $BSML^{\vee}$ and $BSML^{\otimes}$, we show that each formula is provably equivalent to some formula in disjunctive normal form—completeness then reduces to showing it for formulas in normal form. As for BSML, for which no similar normal form is available, our strategy involves simulating the $BSML^{\vee}$ -normal form by using sets of BSML-formulas. For this reason we first axiomatize the extensions.

2.4.1 BSML^V

We first focus on the strongest logic $BSML^{\vee}$. The propositional fragment of $BSML^{\vee}$ corresponds to the logic PT^+ studied in [140] (with the caveat that the negation in PT^+ is not the bilateral negation). Our system for $BSML^{\vee}$ is essentially an extension (and simplification) of the propositional system for PT^+ presented in [140], with additional rules for the bilateral negation and the modalities.

Before we present the system, let us issue a word of caution. Recall from Section 2.2 that BSML and its extensions are not closed under uniform substitution. Due to this, our systems will *not* admit the usual *uniform substitution* rule. Note in particular that in the presentation of our rules, the metavariables α and β range, as before, exclusively over formulas of ML.

Most rules in the system for $BSML^{\vee}$ are also rules in the systems for BSML and $BSML^{\circ}$. For ease of reference, we first present these shared rules; all rules involving \vee are grouped together at the end.

2.4.1. DEFINITION (Natural deduction system for BSML^{\forall}). The following rules comprise a natural deduction system for BSML^{\forall}.

(a) Rules for \wedge :

| D_1 D_2 | D | D | |
|--|--|--|--|
| $\frac{\varphi \psi}{\varphi \land \psi} \land I$ | $\frac{\varphi \wedge \psi}{\varphi} \wedge E$ | $\frac{-\varphi \wedge \psi}{\psi} \wedge E$ | |

(b) Rules for \neg :



(*) The undischarged assumptions in D do not contain NE.

(c) Rules for \lor :

$$\frac{D}{\varphi \lor \psi} \bigvee |(*) \qquad \frac{D}{\varphi \lor \varphi} \lor W \qquad \frac{D}{\psi \lor \varphi} \operatorname{Com} \lor \qquad \frac{D}{(\varphi \lor \chi)} \operatorname{Assv}$$

$$\frac{[\varphi] \quad [\psi]}{\varphi \lor \psi} \qquad \frac{[\varphi]}{\chi} \qquad \frac{[\psi]}{\chi} \qquad \frac{D}{\varphi \lor \psi} \xrightarrow{D} \operatorname{Assv}$$

$$\frac{[\varphi] \quad [\psi]}{\varphi \lor \psi} \qquad \frac{[\varphi]}{\chi} \qquad \frac{[\psi]}{\varphi \lor \psi} \qquad \frac{[\psi]}{\chi} \qquad \frac{\nabla (\psi \lor \chi)}{\chi} \operatorname{Assv}$$

$$\frac{[\varphi] \quad [\psi]}{\varphi \lor \psi} \qquad \frac{[\psi]}{\chi} \qquad \frac{\nabla (\psi \lor \chi)}{\chi} \lor \operatorname{Mon}(\dagger)$$

$$(*) \psi \text{ does not contain NE.}$$

(†) The undischarged assumptions in D_1, D_2 do not contain NE.

- (‡) χ does not contain \vee .
- (d) Rules for \perp and NE:

$$\frac{D}{\frac{\bot \lor \varphi}{\varphi}} \bot \mathsf{E} \qquad \qquad \frac{D}{\frac{\amalg \lor \varphi}{\psi}} \, \bot \mathsf{Ctr}$$

(e) Basic modal rules:



(f) Rules governing the interaction of the modalities and connectives:



(g) Propositional rules involving \vee (BSML^{\vee}-specific):

$$\frac{D}{\varphi \lor \psi} \lor I \qquad \frac{D}{\varphi \lor \psi} \lor I \qquad \frac{D}{\varphi \lor \psi} \bigvee I \qquad \frac{D}{\varphi \lor \psi} \frac{D}{\chi} \frac$$

(h) Modal rules for \lor (BSML^{\lor}-specific):

We write $\Phi \vdash_{\text{BSML}^{\vee}} \psi$ (or simply $\Phi \vdash \psi$) if ψ is derivable from formulas in Φ in the system. We also write $\Phi, \varphi_1, \dots, \varphi_n \vdash \psi$ for $\Phi \cup \{\varphi_1, \dots, \varphi_n\} \vdash \psi$. We say that φ and ψ are *provably equivalent*, written $\varphi \dashv \vdash \psi$, if $\varphi \vdash \psi$ and $\psi \vdash \varphi$.

The conjunction \land and global disjunction \lor have the standard introduction and elimination rules. The rules involving \lor must be constrained due to the failure of downward/union closure occasioned by the presence of NE and \lor . For instance, $p \neq p \lor NE$,

so unconstrained \lor -introduction is not sound. To ensure soundness, the introduced disjunct must have the empty state property, and so in \lor l we require that the introduced disjunct does not contain NE. Alternatively, the introduced disjunct may simply be the premise again, as in \lor -weakening \lor W. \lor -elimination is similarly constrained to ensure that the subderivations D_1 and D_2 do not depend on formulas which are not downward closed, and to ensure that the consequent χ of the rule is union closed. The commutativity and associativity of \lor , and the distributivity of \lor over \lor need to be included as rules; for the derivable algebraic properties of the connectives, see Proposition 2.4.6 below. The rules for \neg include both constrained standard rules as well as rules corresponding to the equivalences noted in Section 2.2. The propositional rules involving \bot and NE— \bot E, \bot Ctr (\bot -contraction), NEl and \neg NEE—are self-explanatory. Note that with \bot we have *ex falso* with respect to classical formulas, and with \bot , we have it with respect to all formulas.

2.4.2. LEMMA. $\bot \vdash \varphi$.

Proof:

 $\bot \vdash \bot \lor \lor by \lor I$, and then $\bot \lor \lor \vdash \varphi$ by $\bot Ctr$.

The basic modal rules (e) are standard. While the modalities do not distribute over \forall , a weaker distributivity in which \forall is switched with \lor —the conversion rules Conv $\Diamond \forall \lor$ and Conv $\Box \forall \lor$ —does hold. The \diamond -separation rule \diamond Sep corresponds to FC-entailment for pragmatically enriched formulas as described in Sections 2.1 and 2.2. The \Box -instantiation rule \Box Inst characterizes the fact that $\Box \varphi$ implies $\Diamond \varphi$ in case accessible worlds exist. The two join rules \diamond Join and $\Box \diamond$ Join allow one to graft together witnessing accessible states.

2.4.3. THEOREM (Soundness of BSML^{\forall}). If $\Phi \vdash \psi$, then $\Phi \models \psi$.

Proof:

This is by induction on the length k of possible derivations of $\Phi \vdash \psi$. We only include the cases for the novel rules involving modalities, as well as for $\lor E$; most of the other cases can be found in [140]. The base case (k = 1) is simple (see [140]). For the inductive case, assume the result holds for all derivations of length at most k. We consider the different possibilities for what the final rule in the derivation $\Phi \vdash \psi$ can be.

VE: Assume that we have derivations of length at most k witnessing $\Phi \vdash \varphi \lor \psi$; $\Phi_1, \varphi \vdash \chi$; and $\Phi_2, \psi \vdash \chi$; that for all $\eta \in \Phi_1 \cup \Phi_2$, η does not contain NE, and that χ does not contain \lor . By the induction hypothesis, $\Phi \models \varphi \lor \psi$; $\Phi_1, \varphi \models \chi$; and $\Phi_2, \psi \models \chi$. We show that $\Phi, \Phi_1, \Phi_2 \models \chi$. Assume that $s \models \Phi \cup \Phi_1 \cup \Phi_2$. Then $s \models \varphi \lor \psi$, so there are some t_1, t_2 such that $s = t_1 \cup t_2$; $t_1 \models \varphi$; and $t_2 \models \psi$. Since no $\eta \in \Phi_1 \cup \Phi_2$ contains NE, each such η is downward closed. So $t_1 \models \eta_1$ for all $\eta_1 \in \Phi_1$, and $t_2 \models \eta_2$ for all $\eta_2 \in \Phi_2$. Thus $t_1 \models \chi$ and $t_2 \models \chi$. Since χ does not contain \lor , it is union closed; therefore, $s \models \chi$.

2.4. Axiomatizations

Conv $\Diamond \forall \forall \forall$: It suffices to show that $\Diamond (\varphi \lor \psi) \equiv \Diamond \varphi \lor \Diamond \psi$; \exists is easy. For \vDash , let $s \vDash \Diamond (\varphi \lor \psi)$. If $s = \emptyset$, then clearly $s \vDash \Diamond \varphi \lor \Diamond \psi$. Otherwise if $s \neq \emptyset$, let $w \in s$. Then there is a nonempty $t \subseteq R[w]$ such that $t \vDash \varphi \lor \psi$, so that $t \vDash \varphi$ or $t \vDash \psi$. Letting

$$s_1 := \{ w \in s \mid \exists t \subseteq R[w] : t \neq \emptyset \text{ and } t \models \varphi \}$$
 and
$$s_2 := \{ w \in s \mid \exists t \subseteq R[w] : t \neq \emptyset \text{ and } t \models \psi \},$$

we have $s = s_1 \cup s_2$. Clearly $s_1 \models \Diamond \varphi$ and $s_2 \models \Diamond \psi$, so $s \models \Diamond \varphi \lor \Diamond \psi$.

Conv \square \lor \lor : This is analogous to Conv \diamondsuit \lor \lor .

 \diamond Sep: It suffices to show that $\diamond(\varphi \lor (\psi \land NE)) \vDash \diamond \psi$; let $s \vDash \diamond(\varphi \lor (\psi \land NE))$. If $s = \emptyset$, then clearly $s \vDash \diamond \psi$. If $s \neq \emptyset$, let $w \in s$. Then there is a nonempty $t \subseteq R[w]$ such that $t \vDash \varphi \lor (\psi \land NE)$. Therefore, there are t_1, t_2 such that $t = t_1 \cup t_2$; $t_1 \vDash \varphi$; and $t_2 \vDash \psi \land NE$. Then $t_2 \neq \emptyset$ and $t_2 \vDash \psi$, and clearly $t_2 \subseteq R[w]$. Thus $s \vDash \diamond \psi$.

◊ Join: It suffices to show that $\Diamond \varphi, \Diamond \psi \models \Diamond (\varphi \lor \psi)$; let $s \models \Diamond \varphi$ and $s \models \Diamond \psi$. If $s = \emptyset$, then clearly $s \models \Diamond (\varphi \lor \psi)$. Otherwise if $s \neq \emptyset$, let $w \in s$. Since $s \models \Diamond \varphi$ and $s \models \Diamond \psi$, there are nonempty $t_1, t_2 \subseteq R[w]$ such that $t_1 \models \varphi$ and $t_2 \models \psi$. Therefore, $t_1 \cup t_2 \models \varphi \lor \psi$. Clearly $t_1 \cup t_2$ is nonempty and $t_1 \cup t_2 \subseteq R[w]$, so $s \models \Diamond (\varphi \lor \psi)$.

□ Inst: It suffices to show that $\Box(\phi \land NE) \models \Diamond \phi$; let $s \models \Box(\phi \land NE)$. If $s = \emptyset$, clearly $s \models \Diamond \phi$. If $s \neq \emptyset$, let $w \in s$. Then $R[w] \models \phi \land NE$, so $R[w] \neq \emptyset$, and therefore $s \models \Diamond \phi$.

 $\Box \diamond \text{ Join: It suffices to show } \Box \phi, \diamond \psi \vDash \Box(\phi \lor \psi); \text{ let } s \vDash \Box \phi \text{ and } s \vDash \diamond \psi. \text{ If } s = \emptyset, \text{ then clearly } s \vDash \Box(\phi \lor \psi). \text{ If } s \neq \emptyset, \text{ let } w \in s. \text{ Since } s \vDash \Box \phi, \text{ we have } R[w] \vDash \phi, \text{ and since } s \vDash \diamond \psi, \text{ there is some nonempty } t \subseteq R[w] \text{ such that } t \vDash \psi. \text{ Since } R[w] = R[w] \cup t, \text{ we have } R[w] \vDash \phi \lor \psi, \text{ and so } s \vDash \Box(\phi \lor \psi). \Box$

As expected, a limited replacement lemma holds for our system (cf. Proposition 2.2.4, the semantic replacement lemma with respect to strong equivalence).

2.4.4. LEMMA (Replacement). Suppose that θ contains a specific occurrence $\theta[p]$ of p which is not in the scope of \neg (unless the \neg forms part of $\Box = \neg \Diamond \neg$). Then $\varphi \vdash \psi$ implies $\theta[\varphi/p] \vdash \theta[\psi/p]$. In particular, if $\varphi \dashv \vdash \psi$, then $\theta[\varphi/p] \dashv \vdash \theta[\psi/p]$.

Proof:

This is by a routine induction on θ , where the modality cases are proved by applying \Diamond Mon and \Box Mon. \Box

We showed in Fact 2.2.5 that every formula is equivalent semantically to one in negation normal form. This fact also holds in the proof system.

2.4.5. PROPOSITION (Negation normal form). Every formula φ is provably equivalent to a formula in negation normal form.

Proof:

The result follows by repeated applications of $\neg \neg \varphi \dashv \vdash \varphi$ (by $\neg \neg E$); $\neg(\varphi \land \psi) \dashv \vdash \neg \varphi \lor \neg \psi$, $\neg(\varphi \lor \psi) \dashv \vdash \neg \varphi \land \neg \psi$ and $\neg(\varphi \lor \psi) \dashv \vdash \neg \varphi \land \neg \psi$ (by the DM-rules); and

 $\neg \diamondsuit \varphi \dashv \vdash \Box \neg \varphi$ (by Inter $\diamondsuit \Box$).

Standard commutativity, associativity and distributivity laws for \land and \lor can be derived in the usual way in our system. Laws involving \lor hold with some restrictions.

2.4.6. PROPOSITION.

- (i) $\varphi \land (\psi \lor \chi) \vdash (\varphi \land \psi) \lor (\varphi \land \chi)$ if φ does not contain NE, and $(\varphi \land \psi) \lor (\varphi \land \chi) \vdash \varphi \land (\psi \lor \chi)$ if φ does not contain \lor ;
- (*ii*) $\varphi \lor (\psi \land \chi) \vdash (\varphi \lor \psi) \land (\varphi \lor \chi);$
- (*iii*) $\varphi \lor (\psi \lor \chi) \vdash (\varphi \lor \psi) \lor (\varphi \lor \chi);$
- $(iv) \ \varphi \lor (\psi \lor \chi) \dashv \vdash (\varphi \lor \psi) \lor (\varphi \lor \chi).$

Proof:

This is easy.

We often apply the above three basic lemmas (Lemmas 2.4.4–2.4.6) without explicit reference to them. Note that all results in this section which do not involve $\forall \forall$ (such as (i) and (ii) of Proposition 2.4.6) also hold for BSML^{\oslash} and BSML. Some such results are given BSML^{\forall}-specific derivations in this section. The BSML^{\oslash}- and BSML-derivations can be found in the sequel.

We now move on to prove the completeness theorem for the system. Our strategy consists in showing that each formula is provably equivalent to one in disjunctive normal form—as stated in the following lemma—and then making use of the semantic and proof-theoretic properties of formulas in this form.

2.4.7. LEMMA. For each $\varphi \in \text{BSML}^{\vee}$, each $k \ge md(\varphi)$, and each finite $X \supseteq P(\varphi)$, there is some property \mathcal{P} over X such that

$$\varphi \dashv \vdash \bigvee_{s \in \mathcal{P}} \theta_s^{\mathsf{X},k}.$$

The proof of the above lemma is involved. We withhold this proof for now, and first present the main completeness argument.

As a first step, let us note that our system is a conservative extension of the smallest normal logic \mathbf{K} —it is easy to see by inspecting our rules, that, for instance, the axioms and rules of the Hilbert-style system for \mathbf{K} are derivable (see also [134] for a detailed proof of the analogous fact for a similar system). Therefore, since \mathbf{K} is complete with respect to the class of all Kripke models, the same holds for the classical fragment of our system.

2.4.8. PROPOSITION (Classical completeness). For any $\Delta \cup \{\alpha\} \subseteq ML : \Delta \models \alpha$ if and only if $\Delta \vdash \alpha$.

2.4. Axiomatizations

As an easy corollary, the *k*th Hintikka formulas of *k*-bisimilar pointed models are provably equivalent.

2.4.9. LEMMA.

- (i) If $w \rightleftharpoons_k w'$, then $\chi_w^k \dashv \vdash \chi_w^k$.
- (*ii*) If $s \rightleftharpoons_k s'$, then $\chi_s^k \dashv \chi_{s'}^k$.

Proof:

For (i), we have $\chi_w^k \equiv \chi_{w'}^k$ by Theorem 2.3.3. Thus $\chi_w^k \dashv \chi_{w'}^k$ follows from Proposition 2.4.8. Item (ii) follows from (i) and Proposition 2.4.8.

Note that, strictly speaking, Hintikka formulas are only guaranteed to be defined for finite pointed models; for an infinite pointed model, we in effect choose some finite *k*-bisimilar pointed model and treat the *k*th Hintikka formula of this finite model as that of the infinite model. The lemma above ensures that our choice of finite representative does not matter proof-theoretically and hence that our use of these representatives in this section is admissible. It follows from results we show that similar provable equivalence results hold for all the characteristic formulas we make use of, and hence that in all cases the use of these representatives is admissible. We now show that the strong Hintikka formulas of two bisimilar pointed models are likewise provably equivalent.

2.4.10. LEMMA. If $s \rightleftharpoons_k s'$, then $\theta_s^k \dashv \theta_{s'}^k$.

Proof:

Suppose that $s \rightleftharpoons_k s'$. The two directions are symmetric; we only give the detailed proof for $\theta_s^k \vdash \theta_{s'}^k$, that is, $\bigvee_{w \in s} (\chi_w^k \land NE) \vdash \bigvee_{w' \in s'} (\chi_{w'}^k \land NE)$. If $s = \emptyset$, then clearly $s' = \emptyset$, so that $\theta_s^k = \theta_{s'}^k = \bot$ and $\theta_s^k \vdash \theta_{s'}^k$. Otherwise if $s \neq \emptyset$, then for any $w \in s$, there exists by assumption a (nonempty) substate $s'_w \subseteq s'$ such that $w \rightleftharpoons_k w'$ for all $w' \in s'_w$. We may assume that s'_w is the maximal such substate. By Lemma 2.4.9, we have that for any $w' \in s'_w$, $\chi_w^k \vdash \chi_{w'}^k$ implying $\chi_w^k \land NE \vdash \chi_{w'}^k \land NE$. Now, by repeatedly applying the rule $\lor W$, we obtain $\chi_{w'}^k \land NE \vdash \bigvee_{u \in s'_w} (\chi_{w'}^k \land NE)$. Since w' is also k-bisimilar to every element $u \in s'_w$, we have by Lemma 2.4.9 again that $\chi_{w'}^k \land NE \vdash \chi_u^k \land NE$. Thus, by $\lor Mon$, we obtain $\bigvee_{u \in s'_w} (\chi_{w'}^k \land NE) \vdash \bigvee_{u \in s'_w} (\chi_u^k \land NE)$. Putting all these together, we derive $\chi_w^k \land NE \vdash \bigvee_{u \in s'_w} (\chi_u^k \land NE)$; that is, $\chi_w^k \land NE \vdash \theta_{s'_w}$. Hence, by repeatedly applying $\lor Mon$ and $\lor E$, we obtain $\bigvee_{w \in s} (\chi_w^k \land NE) \vdash \bigvee_{w \in s} \theta_{s'_w} \vdash$

Hence, by repeatedly applying \lor Mon and \lor E, we obtain $\bigvee_{w \in s} (\chi_w^k \land NE) \vdash \bigvee_{w \in s} \theta_{s'_w} \vdash \theta_{\bigcup_{w \in s} s'_w}$. Observe that since $s \rightleftharpoons_k s'$, we have $s' = \bigcup_{w \in s} s'_w$, whereby $\theta_{\bigcup_{w \in s} s'_w} = \theta_{s'}$. Finally, we conclude that $\theta_s \vdash \theta_{s'}$.

On the other hand, if two states are not *k*-bisimilar, their strong Hintikka formulas are contradictory.

2.4.11. LEMMA. If $s \neq_k s'$, then $\theta_s^k, \theta_{s'}^k \vdash \bot$.

To prove this fact, we need two lemmas. The first is an analogous result for Hintikka formulas for worlds.

2.4.12. LEMMA. If $w \neq_k w'$ for all $w' \in s'$, then $\bigvee_{w' \in s'} \chi_{w'}^k \vdash \neg \chi_w^k$.

Proof:

For any $w' \in s'$, we have $w' \not\models \chi_w^k$ by Theorem 2.3.3 and therefore $w' \vdash \neg \chi_w^k$ by Proposition 2.4.8. So $\bigvee_{w' \in s'} \chi_{w'}^k \vdash \neg \chi_w^k$ by $\lor \mathsf{E}$.

The second lemma states a basic fact concerning the strong contradiction.

2.4.13. LEMMA. $(\alpha \land \text{NE}) \lor \varphi, \neg \alpha \vdash \bot$.

Proof:

We have that

$$((\alpha \land NE) \lor \varphi) \land \neg \alpha \qquad \vdash \qquad ((\alpha \land NE) \land \neg \alpha) \lor (\varphi \land \neg \alpha) \qquad (Prop. 2.4.6)$$
$$\vdash \qquad (\bot \land NE) \lor (\varphi \land \neg \alpha) \qquad (\neg E)$$
$$\vdash \qquad \square \qquad (\bot Ctr)$$

Proof of Lemma 2.4.11:

If $s \neq_k s'$, then either there is a $w \in s$ such that $w \neq_k w'$ for all $w' \in s'$, or there is a $w' \in s'$ such that $w \neq_k w'$ for all $w \in s$. We may without loss of generality assume the former. Then

$$\theta_{s}^{k} \wedge \theta_{s'}^{k} = \theta_{s}^{k} \wedge \bigvee_{w' \in s'} (\chi_{w'}^{k} \wedge \operatorname{NE})$$

$$\vdash \theta_{s}^{k} \wedge \bigvee_{w' \in s'} \chi_{w'}^{k}$$

$$\vdash ((\chi_{w}^{k} \wedge \operatorname{NE}) \vee \theta_{s \setminus \{w\}}^{k}) \wedge \neg \chi_{w}^{k} \quad (\text{Lemma 2.4.12})$$

$$\vdash \blacksquare \quad (\text{Lemma 2.4.13})$$

We show also that strong Hintikka formulas $\theta_s^{X,k}$ are monotone with respect to the parameters k and X in the sense of the following lemma.

2.4.14. LEMMA.

- (*i*) If $n \leq k$, then $\theta_{s}^{k} \vdash \theta_{s}^{n}$.
- (*ii*) If $Y \subseteq X$, then $\theta_s^{X,k} \vdash \theta_s^{Y,k}$.

Proof:

It follows from Theorem 2.3.3 that $\chi_w^k \models \chi_w^n$ and $\chi_w^{X,k} \models \chi_w^{Y,k}$ for any $w \in s$. Therefore, $\chi_w^k \models \chi_w^n$ and $\chi_w^{X,k} \models \chi_w^{Y,k}$ by Proposition 2.4.8. Then also $\chi_w^k \land \text{NE} \models \chi_w^n \land \text{NE}$ and $\chi_w^{X,k} \land \text{NE} \models \chi_w^{Y,k} \land \text{NE}$, and the results therefore follow by repeated applications of $\lor \text{Mon.}$

Our final lemma leading to the completeness proof concerns some standard properties of the disjunction elimination rule.

2.4.15. LEMMA.

- (*i*) If Φ , $\bigvee_{i \in J} \varphi_i \not\vdash \psi$, then $\Phi, \varphi_i \not\vdash \psi$ for some $j \in J$.
- (*ii*) If $\{\bigvee_{j\in J_i} \varphi_j \mid i \in I\} \neq \psi$, then $\{\varphi_i \mid i \in I\} \neq \psi$ for some collection $\{\varphi_i \mid i \in I\}$ such that for each $i \in I$, $\varphi_i \in \{\varphi_j \mid j \in J_i\}$.

Proof:

See, for example, Ciardelli, Iemhoff, and Yang [39]. Item (i) is proved easily by applying $\forall E$. For (ii), choose an enumeration of *I*; the result then follows by an inductive proof making use of (i).

We are now ready to prove completeness. Our proof is similar to the one given in [140] for the completeness of the system PT^+ , though note that the proof in [140] is for weak completeness; here we directly prove strong completeness.

2.4.16. THEOREM (Completeness of BSML^{\forall}). If $\Phi \vDash \psi$, then $\Phi \vdash \psi$.

Proof:

Assume that $\Phi \nvDash \psi$. We show that $\Phi \nvDash \psi$. Put $\Phi = \{\varphi_i \mid i \in I\}$. For each $i \in I$, let $k_i := \max\{md(\varphi_i), md(\psi)\}$, and $X_i := \mathsf{P}(\varphi_i) \cup \mathsf{P}(\psi)$. By Lemma 2.4.7,

$$\varphi_i \dashv \vdash \bigvee_{s \in \mathcal{P}_i} \theta_s^{X_i, k_i} \tag{2.1}$$

for some properties \mathcal{P}_i over X_i . Then also $\{ \bigvee_{s \in \mathcal{P}_i} \theta_s^{X_i, k_i} | i \in I \} \notin \psi$. By Lemma 2.4.15, there is a collection of formulas $\Phi' = \{ \theta_{s_i}^{X_i, k_i} | s_i \in \mathcal{P}_i \text{ and } i \in I \}$ such that $\Phi' \notin \psi$.

Observe that $\Phi' \not\models \bot$, since otherwise clearly for some $\theta_{s_i}^{X_i,k_i}, \theta_{s_j}^{X_j,k_j} \in \Phi'$ we have $s_i \not\models_m^{\mathsf{M}} s_j$ for $m = \min\{k_i, k_j\}$ and $\mathsf{M} = \mathsf{X}_i \cap \mathsf{X}_j$, so that by Lemma 2.4.11, $\theta_{s_i}^{\mathsf{M},m}, \theta_{s_j}^{\mathsf{M},m} \vdash \bot$. Then by Lemma 2.4.14, $\theta_{s_i}^{X_i,k_i}, \theta_{s_j}^{X_j,k_j} \vdash \bot$, so that $\Phi' \vdash \bot$, whence $\Phi' \vdash \psi$ by Lemma 2.4.2; a contradiction.

Thus, we can let *t* be such that $t \models \Phi'$. By (2.1) and soundness, we have $\Phi' \models \varphi_i$ for each $i \in I$. Therefore, $t \models \Phi$. To show that $\Phi \not\models \psi$, it suffices to show that $t \not\models \psi$. Assume otherwise. Take an $i \in I$. By Lemma 2.4.7, we have

$$\boldsymbol{\psi} \dashv \vdash \bigvee_{r \in \mathcal{Q}} \boldsymbol{\theta}_r^{\mathsf{X}_i, k_i} \tag{2.2}$$

for some property Q over X_i . Since $t \models \psi$, by (2.2) and soundness, we have $t \models \theta_r^{X_i,k_i}$ for some $r \in Q$. Meanwhile, $t \models \Phi'$ implies $t \models \theta_{s_i}^{X_i,k_i}$. Thus, by Proposition 2.3.11(ii), we have $r \rightleftharpoons_{k_i}^{X_i} t \rightleftharpoons_{k_i}^{X_i} s_i$, whence $\theta_{s_i}^{X_i,k_i} \vdash \theta_r^{X_i,k_i}$ by Lemma 2.4.10. Then by $\forall I$ and (2.2), $\theta_{s_i}^{X_i,k_i} \vdash \bigvee_{r \in Q} \theta_r^{X_i,k_i} \vdash \psi$. But then $\Phi' \vdash \psi$, which is a contradiction. \Box

We derive the compactness of $BSML^{\vee}$ as well as that of the weaker logics BSML $BSML^{\oslash}$ as a corollary of strong completeness. The remainder of this section is devoted to the proof of Lemma 2.4.7 (provable equivalence of the normal form). We start with two technical lemmas concerning the behavior of NE in disjunctions.

2.4.17. LEMMA. $\varphi \lor (\psi \land \text{NE}) \dashv \vdash (\varphi \lor (\psi \land \text{NE})) \land \text{NE}.$

Proof:

 \dashv follows by \land E. We give a BSML^V-specific derivation for \vdash :

| | $\varphi \lor (\psi \land \operatorname{NE})$ | |
|---|--|---------------|
| ⊢ | $(\varphi \lor (\psi \land \operatorname{NE})) \land (\bot \lor \operatorname{NE})$ | (NEl) |
| ⊢ | $((\varphi \lor (\psi \land \operatorname{NE})) \land \bot) \lor ((\varphi \lor (\psi \land \operatorname{NE})) \land \operatorname{NE})$ | |
| ⊢ | $((\varphi \land \bot) \lor ((\psi \land \operatorname{NE}) \land \bot)) \lor ((\varphi \lor (\psi \land \operatorname{NE})) \land \operatorname{NE})$ | (Prop. 2.4.6) |
| ⊢ | $((\varphi \land \bot) \lor (\psi \land \amalg)) \lor ((\varphi \lor (\psi \land \operatorname{NE})) \land \operatorname{NE})$ | |
| ⊢ | $((\varphi \lor (\psi \land \operatorname{NE})) \land \operatorname{NE}) \lor ((\varphi \lor (\psi \land \operatorname{NE})) \land \operatorname{NE})$ | (⊥LCtr) |
| ⊢ | $(\varphi \lor (\psi \land \operatorname{NE})) \land \operatorname{NE}$ | |

2.4.18. LEMMA. $\varphi \lor \psi \vdash \varphi \lor \psi \lor ((\varphi \land \text{NE}) \lor ((\psi \land \text{NE}))).$

Proof:

We have that

2.4. Axiomatizations

The following lemma further characterizes the interactions between NE and the disjunctions. From item (iv) below, it follows that formulas which are in ML-normal form $\bigvee_{(M,w)\in \llbracket \alpha \rrbracket} \chi_w^k$ can be converted into BSML^V-normal form $\bigvee_{\mathcal{P}\subseteq \llbracket \alpha \rrbracket} \bigvee_{w\in \mathcal{P}} (\chi_w^k \land NE)$.

2.4.19. LEMMA.

(*i*) NE, $\bigvee_{i \in I} \varphi_i \vdash \bigvee_{\varnothing \neq J \subseteq I} \bigvee_{j \in J} (\varphi_j \land NE).$ (*ii*) NE $\land \bigvee_{i \in I} \alpha_i \dashv \vdash \bigvee_{\varnothing \neq J \subseteq I} \bigvee_{j \in J} (\alpha_j \land NE).$ (*iii*) $\bigvee_{i \in I} \varphi_i \vdash \bigvee_{J \subseteq I} \bigvee_{j \in J} (\varphi_j \land NE).$ (*iv*) $\bigvee_{i \in I} \alpha_i \dashv \vdash \bigvee_{J \subseteq I} \bigvee_{j \in J} (\alpha_j \land NE).$

(i) This is by induction on k = |I|. If k = 0, then $\bigvee \emptyset = \bot$, and by Lemma 2.4.2 we have $NE \land \bot \vdash \psi$ for any ψ .

If k = 1, then $\bigvee \{\varphi\} = \varphi$, and φ , NE $\vdash \varphi \land$ NE by \land I; and $\bigvee \{\bigvee \{\varphi \land NE\}\} = \varphi \land NE$. If k = 2, then by Lemma 2.4.18 we have $\varphi \lor \psi \vdash \varphi \lor \psi \lor ((\varphi \land NE) \lor (\psi \land NE))$, so that clearly NE, $\varphi \lor \psi \vdash (\varphi \land NE) \lor (\psi \land NE) \lor ((\varphi \land NE) \lor (\psi \land NE))$.

 $\frac{1}{2} = \frac{1}{2} + \frac{1}$

For the case |I| = k + 1, by the induction hypothesis,

$$\operatorname{NE}, \bigvee_{i \in (I \setminus \{x\})} \varphi_i \vdash \bigvee_{\varnothing \neq J \subseteq (I \setminus \{x\})} \bigvee_{j \in J} (\varphi_j \wedge \operatorname{NE}), \qquad (2.3)$$

where $x \in I$. Then we have that

$$NE, \bigvee_{i \in I} \varphi_{i}$$

$$\vdash NE \land (\varphi_{x} \lor \bigvee_{i \in (I \setminus \{x\})} \varphi_{i})$$

$$\vdash (\varphi_{x} \land NE) \lor (NE \land \bigvee_{i \in (I \setminus \{x\})} \varphi_{i}) \lor ((\varphi_{x} \land NE) \lor (NE \land \bigvee_{i \in (I \setminus \{x\})} \varphi_{i})) \quad (Case \ k = 2)$$

$$\vdash (\varphi_{x} \land NE) \lor (\bigvee_{\emptyset \neq J \subseteq (I \setminus \{x\})} \bigvee_{j \in J} (\varphi_{j} \land NE)) \lor ((\varphi_{x} \land NE) \lor (\bigvee_{\emptyset \neq J \subseteq (I \setminus \{x\})} \bigvee_{j \in J} (\varphi_{j} \land NE))$$

$$\vdash (\varphi_{x} \land NE) \lor (\bigvee_{\emptyset \neq J \subseteq (I \setminus \{x\})} \bigvee_{j \in J} (\varphi_{j} \land NE)) \lor (\bigvee_{\emptyset \neq J \subseteq (I \setminus \{x\})} ((\varphi_{x} \land NE) \lor \bigvee_{j \in J} (\varphi_{j} \land NE)))$$

$$\vdash (\bigvee_{\emptyset \neq J \subseteq I} \bigvee_{j \in J} (\varphi_{j} \land NE))$$

(ii) \vdash by (i). \dashv : For any nonempty $J \subseteq I$, we have that $\bigvee_{j \in J} (\alpha_j \land NE) \vdash NE$ by Lemma 2.4.17. We also have that for any $j \in J$, $\alpha_j \land NE \vdash \bigvee_{i \in I} \alpha_i$ by $\land E$ and $\lor I$. Therefore, $\bigvee_{j \in J} (\alpha_j \land NE) \vdash \bigvee_{i \in I} \alpha_i$ by $\lor E$. And so $\bigvee_{\emptyset \neq J \subseteq I} \bigvee_{j \in J} (\alpha_j \land NE) \vdash NE \land \bigvee_{i \in I} \alpha_i$ by $\lor E$.

(iii) We have that

$$\bigvee_{i \in I} \varphi_i \qquad \vdash \qquad (\bot \lor \operatorname{NE}) \land \bigvee_{i \in I} \varphi_i \qquad (\operatorname{NE})$$

$$\vdash \qquad (\bot \land \bigvee_{i \in I} \varphi_i) \lor (\operatorname{NE} \land \bigvee_{i \in I} \varphi_i)$$

$$\vdash \qquad \bot \lor \bigvee_{\varnothing \neq J \subseteq I} \bigvee_{j \in J} (\varphi_j \land \operatorname{NE}) \qquad ((i))$$

$$\vdash \qquad \qquad \bigvee_{J \subseteq I} \bigvee_{j \in J} (\varphi_j \land \operatorname{NE}) \qquad (\lor \varnothing = \bot)$$

 $(iv) \vdash by (iii)$. \dashv : We have that

$$\bigvee_{J \subseteq I} \bigvee_{j \in J} (\alpha_j \wedge \mathrm{NE}) \qquad \vdash \qquad \bot \lor \bigvee_{\varnothing \neq J \subseteq I} \bigvee_{j \in J} (\alpha_j \wedge \mathrm{NE}) \qquad (\lor \varnothing = \bot)$$

$$\perp \forall \left(\mathsf{NE} \land \bigvee_{i \in I} \alpha_i \right) \tag{(ii)}$$

$$\bigvee_{i \in I} \alpha_i \qquad (\forall \mathsf{E}, \mathsf{Prop.}\ 2.4.8)$$

It now follows that each classical formula is equivalent to one in $BSML^{\vee}$ -normal form, which, in our inductive proof for Lemma 2.4.10, takes care of the case in which the formula is classical.

 \vdash

⊢

2.4.20. LEMMA. For each $\alpha \in ML$, each $k \ge md(\alpha)$, and each finite $X \supseteq P(\alpha)$, there is some property \mathcal{P} over X such that

$$\alpha \dashv \vdash \bigvee_{s \in \mathcal{P}} \theta_s^{\mathsf{X},\mathsf{k}}.$$

Proof:

 $\llbracket \alpha \rrbracket$ (over X) is invariant under *k*-bisimulation so by the proof of Theorem 2.3.5, we have $\alpha \equiv \bigvee_{w \in \llbracket \alpha \rrbracket} \chi_w^k$. Then by Proposition 2.4.8, $\alpha \dashv \vdash \bigvee_{w \in \llbracket \alpha \rrbracket} \chi_w^k$, and therefore by Lemma 2.4.19(iv):

$$\alpha \to \bigcup_{w \in \llbracket \alpha \rrbracket} \chi_w^k \to \bigcup_{\mathcal{P} \subseteq \llbracket \alpha \rrbracket w \in \mathcal{P}} (\chi_w^k \wedge \operatorname{NE}) = \bigvee_{s \in \wp(\llbracket \alpha \rrbracket)} \bigvee_{w \in s} \theta_s^k,$$

where $\wp(\llbracket \alpha \rrbracket)$ is the state property $\{ \bigcup \{ (M, \{w\}) \mid (M, w) \in \mathcal{P} \} \mid \mathcal{P} \subseteq \llbracket \alpha \rrbracket \}$.

Now, for the modality cases in our inductive proof for Lemma 2.4.10, we need to show normal form provable equivalence for formulas of the forms $\Diamond \xi_{\mathcal{P}}^k$ and $\Box \xi_{\mathcal{P}}^k$.

These formulas are flat, so by Proposition 2.3.13 we have $\Diamond \xi_{\mathcal{P}}^k \equiv \alpha$ and $\Box \xi_{\mathcal{P}}^k \equiv \beta$ for some $\alpha, \beta \in ML$. Given Lemma 2.4.20, it is thus sufficient to show that $\Diamond \xi_{\mathcal{P}}^k \dashv \alpha$ and $\Box \xi_{\mathcal{P}}^k \dashv \beta$; that is, the modality cases can be reduced to the case for classical formulas. As will become clear later, this further reduces to showing that formulas of the form $\Diamond \theta_s^k$ or $\Box \theta_s^k$ are provably equivalent to classical formulas. We now work towards proving this result. First, two technical lemmas are in order.

2.4.21. LEMMA. $\Diamond \varphi \vdash \Diamond (\varphi \land \text{NE}).$

Proof:

We give a $BSML^{\vee}$ -specific derivation:

$$\begin{array}{cccc} \diamond \varphi & \vdash & \diamond (\varphi \land (\bot \lor NE)) & (NEl) \\ \vdash & \diamond ((\varphi \land \bot) \lor (\varphi \land NE)) \\ \vdash & \diamond (\bot \lor (\varphi \land NE)) \\ \vdash & \diamond \bot \lor \diamond (\varphi \land NE) & (\diamond \lor \lor \mathsf{Conv}) \\ \vdash & \bot \lor \diamond (\varphi \land NE) & (\mathsf{Prop. 2.4.8, \lor Mon}) \\ \vdash & \diamond (\varphi \land NE) & (\bot E) \end{array}$$

2.4.22. LEMMA. For all φ which do not contain \lor and for all ψ :

$$(i) \, \diamond((\varphi \wedge \operatorname{NE}) \lor \psi) \dashv \vdash \diamond(\varphi \lor \psi) \land \diamond \varphi;$$

$$(ii) \ \Box((\phi \land \operatorname{NE}) \lor \psi) \dashv \vdash \Box(\phi \lor \psi) \land \diamondsuit \phi.$$

Proof:

(i) \vdash : By \diamondsuit Sep. \dashv : We have that

$$\begin{array}{cccc} \diamond(\varphi \lor \psi) \land \diamond \varphi & \vdash & \diamond(\varphi \lor \psi) \land \diamond(\varphi \land \text{NE}) & (\text{Lemma 2.4.21}) \\ \vdash & \diamond(\varphi \lor (\varphi \land \text{NE}) \lor \psi) & (\diamond \text{Join}) \\ \vdash & \diamond(((\varphi \lor (\varphi \land \text{NE})) \land \text{NE}) \lor \psi) & (\text{Lemma 2.4.17, } \lor \text{Mon}) \\ \vdash & \diamond((\varphi \land \text{NE}) \lor \psi) & (\lor \text{E}) \end{array}$$

(ii) \vdash : We have that

$$\Box((\varphi \land NE) \lor \psi) \vdash \Box(\varphi \lor \psi) \land \Box((\varphi \land NE) \lor \psi)$$

$$\vdash \Box(\varphi \lor \psi) \land \Box(((\varphi \land NE) \lor \psi) \land NE) \quad (Lemma 2.4.17)$$

$$\vdash \Box(\varphi \lor \psi) \land \diamondsuit((\varphi \land NE) \lor \psi) \quad (\Box Inst)$$

$$\vdash \Box(\varphi \lor \psi) \land \diamondsuit \varphi \qquad (\diamondsuit Sep)$$

$$\Box(\varphi \lor \psi) \land \Diamond \varphi \qquad (\Diamond \operatorname{Sep})$$

 \dashv : Similar to \dashv of (i), using $\Box \diamondsuit$ Join instead of \diamondsuit Join.

We now show that formulas of the form $\Diamond \theta_s^k$ or $\Box \theta_s^k$ are provably equivalent to classical formulas.

2.4.23. LEMMA.

$$(i) \ \Diamond \ \theta_s^k \dashv \vdash \Diamond \ \chi_s^k \land \bigwedge_{w \in s} \Diamond \ \chi_w^k.$$
$$(ii) \ \Box \ \theta_s^k \dashv \vdash \Box \ \chi_s^k \land \bigwedge_{w \in s} \Diamond \ \chi_w^k.$$

Proof:

(i) If $s = \emptyset$, then $\Diamond \theta_s^k = \Diamond \bot = \Diamond \chi_s^k \dashv \vdash \Diamond \chi_s^k \land \Pi = \Diamond \chi_s^k \land \bigwedge_{w \in s} \Diamond \chi_w^k$. Otherwise if $s \neq \emptyset$, let $\theta_s^k = (\chi_{w_1}^k \land \text{NE}) \lor \cdots \lor (\chi_{w_n}^k \land \text{NE})$. Then by Lemma 2.4.22(i),

$$\diamond \theta_{s}^{k} = \qquad \diamond \bigvee_{w \in s} (\chi_{w}^{k} \wedge \operatorname{NE})$$

$$\dashv \vdash \qquad \diamond (\bigvee_{i=1}^{n-1} (\chi_{w_{i}}^{k} \wedge \operatorname{NE}) \vee \chi_{w_{n}}^{k}) \wedge \diamond \chi_{w_{n}}^{k}$$

$$\dashv \vdash \qquad \diamond (\bigvee_{i=1}^{n-2} (\chi_{w_{i}}^{k} \wedge \operatorname{NE}) \vee \chi_{w_{n-1}}^{k} \vee \chi_{w_{n}}^{k}) \wedge \diamond \chi_{w_{n-1}}^{k} \wedge \diamond \chi_{w_{n}}^{k}$$

$$\vdots$$

$$\dashv \vdash \qquad \diamond \chi_{s}^{k} \wedge \bigwedge_{w \in s} \diamond \chi_{w}^{k}$$

(ii) This is similar to (i), using Lemma 2.4.22(ii) instead of Lemma 2.4.22(i).

We are finally ready to prove Lemma 2.4.7, the normal form lemma. **Proof of Lemma 2.4.7:**

This is by induction on the complexity of φ . By Proposition 2.4.5, we may assume that φ is in negation normal form. Let $k \ge md(\varphi)$.

If $\varphi = p$ or $\varphi = \neg p$, we apply Lemma 2.4.20. If $\varphi = \neg \text{NE}$, we have $\neg \text{NE} \dashv \vdash \bot$ by NEE, and the result then follows by Lemma 2.4.20. If $\varphi = \text{NE}$, we have that

$$NE \qquad \dashv \vdash \qquad NE \land (p \lor \neg p) \qquad (Prop. 2.4.8)$$

$$\dashv \vdash \qquad NE \land \bigvee_{s \in \mathcal{P}} \theta_s^k \qquad (Lemma 2.4.20)$$

$$\dashv \vdash \qquad NE \land (\bot \lor \bigvee_{s \neq \emptyset \in \mathcal{P}} \theta_s^k)$$

$$\dashv \vdash \qquad (NE \land \bot) \lor \bigvee_{s \neq \emptyset \in \mathcal{P}} \theta_s^k \qquad (Prop. 2.4.6, Lemma 2.4.17)$$

$$\dashv \vdash \qquad \bigvee_{s \in \mathcal{P}} \theta_s^k \qquad (Lemma 2.4.2)$$

If $\varphi = \psi \land \chi$, then $k \ge md(\varphi) = \max\{md(\psi), md(\chi)\}$, so by the induction hypothesis there are \mathcal{P} and \mathcal{Q} such that $\psi \dashv \vdash \bigvee_{s \in \mathcal{P}} \theta_s^k$ and $\chi \dashv \vdash \bigvee_{t \in \mathcal{Q}} \theta_t^k$. Let $\mathcal{R} := \{r \mid \exists s \in \mathcal{P}, \exists t \in \mathcal{Q} : s \rightleftharpoons_k r \rightleftharpoons_k t\}$. We have that

$$\boldsymbol{\psi} \wedge \boldsymbol{\chi} \quad \dashv \vdash \quad \bigvee_{s \in \mathcal{P}} \boldsymbol{\theta}_s^k \wedge \bigvee_{t \in \mathcal{Q}} \boldsymbol{\theta}_t^k$$

44

2.4. Axiomatizations

If $\varphi = \psi \lor \chi$, then $k \ge md(\varphi) = \max\{md(\psi), md(\chi)\}$, and by the induction hypothesis we have properties \mathcal{P}, \mathcal{Q} as in the conjunction case. Then

 $s \in \mathcal{R}$

$$\begin{split} \psi \lor \chi & \dashv \vdash & \bigvee_{s \in \mathcal{P}} \theta_s^k \lor \bigvee_{t \in \mathcal{Q}} \theta_t^k \\ \dashv \vdash & \bigvee_{s \in \mathcal{P}} \bigvee_{t \in \mathcal{Q}} (\theta_s^k \lor \theta_t^k) & (\text{Prop. 2.4.6}) \\ \dashv \vdash & \bigvee_{s \in \mathcal{P}} \bigvee_{t \in \mathcal{Q}} \theta_{s \uplus t}^k \end{split}$$

If $\varphi = \psi \lor \chi$, then the result follows immediately by the induction hypothesis.

If $\varphi = \Diamond \psi$, then $k - 1 \ge md(\varphi) - 1 = md(\psi)$, so by the induction hypothesis there is a property \mathcal{P} such that $\psi \dashv \vdash \bigvee_{s \in \mathcal{P}} \theta_s^{k-1}$. Thus,

$$\diamondsuit \psi \qquad \dashv \vdash \qquad \diamondsuit \bigotimes_{s \in \mathcal{P}} \theta_s^{k-1}$$

$$\dashv \vdash \qquad \bigvee_{s \in \mathcal{P}} \diamondsuit \theta_s^{k-1} \qquad (\operatorname{Conv} \diamondsuit \lor \lor)$$

$$\dashv \vdash \qquad \bigvee_{s \in \mathcal{P}} (\diamondsuit \chi_s^{k-1} \land \bigwedge_{w \in s} \diamondsuit \chi_w^{k-1}) \qquad (\operatorname{Lemma 2.4.23(i)})$$

This formula is classical and of modal depth at most k, so we are done by Lemma 2.4.20.

The case $\varphi = \Box \psi$ is similar to the case for $\Diamond \psi$, except we use Conv $\Box \lor \lor$ instead of Conv $\Diamond \lor \lor$ and Lemma 2.4.23(ii) instead of Lemma 2.4.23(i).

2.4.2 BSML[⊘]

For the BSML^{\oslash}-system, we remove the rules concerning \lor from the BSML^{\heartsuit}-system and add rules for the emptiness operator \oslash . Recall that \oslash corresponds essentially to

one specific type of global disjunction, as $\oslash \psi \equiv \psi \boxtimes \bot$. The introduction rules for \oslash capture (the right-to-left entailment in) this equivalence by imitating instances of \lor -introduction which yield $\psi \vdash \psi \boxtimes \bot$ and $\bot \vdash \psi \boxtimes \bot$. The elimination rules for \oslash capture something stronger than (the left-to-right entailment in) $\oslash \psi \equiv \psi \boxtimes \bot$: we also encode the fact that \lor , \land , and \oslash distribute over \circledast —and that \neg and \diamondsuit do not—directly into these rules. To that end, we call a formula occurrence $[\psi] \boxtimes$ -distributive in χ if $[\psi]$ is not within the scope of any \neg or \diamondsuit in χ (where recall that \Box is an abbreviation of $\neg \diamondsuit \neg$). For example, [p] is not \circledast -distributive in $\Box(p \lor q) = \neg \diamondsuit \neg (p \lor q)$ but it *is* \circledast -distributive in the subformula $p \lor q$. Given the pertinent equivalence and distributivity facts, we have that if $[\oslash \psi]$ is \lor -distributive in φ , then $\varphi \equiv \varphi[\psi \lor \bot / \oslash \psi] = \varphi[\psi / \oslash \psi] \lor \varphi[\bot / \oslash \psi]$ (and similarly for \Box). The elimination rules for \oslash capture (the left-to-right entailments in) these equivalences by imitating the \circledast -rules as applied to formulas of the form $\varphi[\psi / \oslash \psi] \lor \varphi[\bot / \oslash \psi]$ and $\diamondsuit / \Box(\varphi[\psi / \oslash \psi] \lor \varphi[\bot / \oslash \psi]$).

2.4.24. DEFINITION (Natural deduction system for BSML^{\oslash}). The natural deduction system for BSML^{\oslash} includes all rules not involving \lor from the system for BSML^{\lor} (boxes (a)–(f)) and the following rules for \oslash :



The introduction rules \oslash NEl and \oslash I imitate the BSML^{\lor} -rules NEl and \lor I, respectively. The elimination rule \oslash E imitates the rule \lor E; and \diamondsuit \oslash E and $\Box \oslash$ E imitate \diamondsuit \lor \lor Conv and $\Box \lor \lor$ Conv, respectively. The anti-support clause for $\oslash \varphi$ is characterized by the elimination rule $\neg \oslash$ E. The soundness of these new rules follows from the equivalences noted above.

2.4.25. THEOREM (Soundness of BSML^{\otimes}). *If* $\Phi \vdash \psi$, *then* $\Phi \models \psi$.

To provide a simple demonstration of the new rules, we derive the useful fact that \oslash cancels out the effects of appending \land NE to classical formulas.

2.4. Axiomatizations

2.4.26. LEMMA. $\alpha \rightarrow = \emptyset(\alpha \land \text{NE}).$

Proof:

 $\neg: \bot \vdash \alpha \text{ by } \neg \text{I and } \neg \text{E, and } \alpha \land \text{NE} \vdash \alpha \text{ so } \oslash(\alpha \land \text{NE}) \vdash \alpha \text{ by } \oslash \text{E.} \vdash: \alpha \vdash \alpha \land \oslash \text{NE} \text{ by } \oslash \text{NE}.$ $\bigcirc \text{NEI. We have } \alpha \land \oslash \text{NE}[\text{NE} / \oslash \text{NE}] = \alpha \land \text{NE, and } \alpha \land \text{NE} \vdash \oslash(\alpha \land \text{NE}) \text{ by } \oslash \text{I. On the other hand, } \alpha \land \oslash \text{NE}[\bot / \oslash \text{NE}] = \alpha \land \bot, \text{ and } \alpha \land \bot \vdash \bot \vdash \oslash(\alpha \land \text{NE}) \text{ by } \oslash \text{I. Therefore, } \oslash \text{NE} \land \alpha \vdash \oslash(\alpha \land \text{NE}) \text{ by } \oslash \text{E.}$

Our strategy for proving completeness is similar to that used in Section 2.4.1. We again require a crucial normal form provable equivalence result. This is stated in the next lemma; the proof is once more withheld until the end of the section.

2.4.27. LEMMA. For each $\varphi \in BSML^{\otimes}$ and each $k \ge md(\varphi)$, there is some property \mathcal{P} such that

$$\boldsymbol{\varphi} \dashv \vdash \bigvee_{s \in \mathcal{P}} \oslash \boldsymbol{\theta}_s^k \quad or \quad \boldsymbol{\varphi} \dashv \vdash \operatorname{NE} \land \bigvee_{s \in \mathcal{P}} \oslash \boldsymbol{\theta}_s^k.$$

A number of lemmas which were proved specifically for BSML^{\vee} in Section 2.4.1 also hold for BSML^{\circ}, and their BSML^{\circ}-proofs are similar. In particular, Lemma 2.4.4 (replacement) follows by a similar inductive argument, where the new case $\theta = \odot \eta$ is proved by applying $\oslash E$ and $\oslash I$. Proposition 2.4.5 (negation normal form) is proved by a similar argument, using the additional equivalence $\neg \oslash \varphi \dashv \vdash \neg \varphi$ (by $\neg \oslash E$).

Now, observe that a formula in BSML^{\circ}-normal form can be converted into an equivalent formula in the normal form of the expressively stronger logic BSML^{\vee}, as stated in the following lemma (we omit the easy proof).

2.4.28. LEMMA.
$$\bigvee_{s \in \mathcal{P}} \oslash \theta_s^k \equiv \bigvee_{\mathcal{Q} \subseteq \mathcal{P}} \theta_{t \neq \mathcal{Q}}^k.$$

While the global disjunction \mathbb{V} is not in the language of BSML^{\oslash}, we show in the next lemma that the formulas $\bigvee_{s \in \mathcal{P}} \otimes \theta_s^k$ and $\bigvee_{\mathcal{Q} \subseteq \mathcal{P}} \theta_{t \not Q}^k$ are also proof-theoretically interchangeable in the sense that if rules for both \otimes and \mathbb{V} were available, then the two formulas would derive the same consequences and be derivable from the same premises. Item (i) below corresponds to the entailment $\bigvee_{\mathcal{Q} \subseteq \mathcal{P}} \theta_{t \not Q}^k \models \bigvee_{s \in \mathcal{P}} \otimes \theta_s^k$, and item (ii) simulates the other direction $\bigvee_{s \in \mathcal{P}} \otimes \theta_s^k \models \bigvee_{\mathcal{Q} \subseteq \mathcal{P}} \theta_{t \not Q}^k$.

2.4.29. LEMMA.

- (i) For any $\mathcal{Q} \subseteq \mathcal{P}$, we have $\theta_{t}^k \mathcal{Q} \vdash \bigvee_{s \in \mathcal{P}} \otimes \theta_s^k$.
- (*ii*) Suppose that $\bigvee_{s \in \mathcal{P}} \otimes \theta_s^k$ is \forall -distributive in φ . If $\Phi, \varphi[\theta_{t \in \mathcal{Q}}^k / \bigvee_{s \in \mathcal{P}} \otimes \theta_s^k] \vdash \psi$ for all $\mathcal{Q} \subseteq \mathcal{P}$, then $\Phi, \varphi \vdash \psi$.

Proof:

(i) Clearly, $\theta_{t \mid Q}^{k} \to \bigvee_{s \in Q} \theta_{s}^{k}$. Next, we derive $\bigvee_{s \in Q} \theta_{s}^{k} \mapsto \bigvee_{s \in Q} \otimes \theta_{s}^{k}$ by $\otimes I$ and \vee Mon. Since $\bigvee_{s \in Q} \otimes \theta_{s}^{k} \mapsto \bigvee_{s \in Q} \otimes \theta_{s}^{k} \vee \bigvee_{t \in \mathcal{P} \setminus Q} \bot$ (by $\vee I$), we obtain $\bigvee_{s \in Q} \otimes \theta_{s}^{k} \mapsto \bigvee_{s \in Q} \otimes \theta_{s}^{k} \vee \bigvee_{t \in \mathcal{P} \setminus Q} \boxtimes \theta_{t}^{k}$ by $\otimes I$ —that is, $\bigvee_{s \in Q} \otimes \theta_{s}^{k} \mapsto \bigvee_{s \in \mathcal{P}} \otimes \theta_{s}^{k}$. Putting all these together, we conclude that $\theta_{t \mid Q}^{k} \mapsto \bigvee_{s \in \mathcal{P}} \otimes \theta_{s}^{k}$.

(ii) Let $\mathcal{P} = \{s_1, \ldots, s_n\}$, and thus $\zeta_{\mathcal{P}}^k = \bigvee_{s \in \mathcal{P}} \oslash \theta_s^k = \oslash \theta_{s_1}^k \lor \cdots \lor \oslash \theta_{s_n}^k$. Observe that for every $\mathcal{Q} \subseteq \mathcal{P}$, we have $\theta_{(\ddagger)\mathcal{Q}}^k \dashv \vdash \bigvee_{s_i \in \mathcal{Q}} \theta_{s_i}^k \dashv \vdash \bigvee_{i=1}^n \tau_i$, where $\tau_i = \theta_{s_i}^k$ if $s_i \in \mathcal{Q}$, and $\tau_i = \bot$ if $s_i \notin \mathcal{Q}$, by $\lor I$ and $\bot E$. Now, by assumption, we have $\Phi, \varphi[\theta_{(\ddagger)\mathcal{Q}}^k / \zeta_{\mathcal{P}}^k] \vdash \psi$ and $\Phi, \varphi[\theta_{(\ddagger\{s_n\})}^k / \zeta_{\mathcal{P}}^k] \vdash \psi$. Since $\theta_{(\ddagger)\mathcal{Q}}^k \dashv \vdash \bigvee_{i=1}^n \bot$ and $\theta_{(\ddagger\{s_n\})}^k \dashv \vdash (\bigvee_{i=1}^{n-1} \bot) \lor \theta_{s_n}^k$, we obtain

$$\Phi, \varphi[(\bigvee_{i=1}^{n-2} \bot) \lor \bot \lor \bot/\zeta_{\mathcal{P}}^{k}] \vdash \psi \quad \text{and} \quad \Phi, \varphi[(\bigvee_{i=1}^{n-2} \bot) \lor \bot \lor \theta_{s_{n}}^{k}/\zeta_{\mathcal{P}}^{k}] \vdash \psi$$

Then by $\oslash E$, we obtain

$$\Phi, \varphi[(\bigvee_{i=1}^{n-2} \bot) \lor \bot \lor \oslash \theta_{s_n}^k / \zeta_{\mathcal{P}}^k] \vdash \Psi.$$
(2.4)

Next, by assumption again, Φ , $\varphi[\theta_{\forall \{s_{n-1}\}}^k / \zeta_{\mathcal{P}}^k] \vdash \psi$ and Φ , $\varphi[\theta_{\forall \{s_{n-1}, s_n\}}^k / \zeta_{\mathcal{P}}^k] \vdash \psi$, which imply

$$\Phi, \varphi[(\bigvee_{i=1}^{n-2} \bot) \lor \theta_{s_{n-1}}^k \lor \bot/\zeta_{\mathcal{P}}^k] \vdash \psi \quad \text{and} \quad \Phi, \varphi[(\bigvee_{i=1}^{n-2} \bot) \lor \theta_{s_{n-1}}^k \lor \theta_{s_n}^k/\zeta_{\mathcal{P}}^k] \vdash \psi.$$

Again, by $\oslash E$, we obtain

$$\Phi, \varphi[(\bigvee_{i=1}^{n-2} \bot) \lor \theta_{s_{n-1}}^k \lor \oslash \theta_{s_n}^k / \zeta_{\mathcal{P}}^k] \vdash \psi.$$
(2.5)

Applying $\oslash \mathsf{E}$ to (2.4) and (2.5), we obtain $\Phi, \varphi[(\bigvee_{i=1}^{n-2} \bot) \lor \oslash \theta_{s_{n-1}}^k \lor \oslash \theta_{s_n}^k / \zeta_{\mathcal{P}}^k] \vdash \psi$. Repeating this argument, we eventually get $\Phi, \varphi \vdash \psi$, as required. \Box

Given this proof-theoretic correspondence between the normal forms, we can show completeness for BSML^{\circ} using an argument that is similar to that used for BSML^{\vee}. This time we first establish weak completeness, and then derive the strong completeness theorem as a corollary of weak completeness and compactness. (The strategy we employed for the direct proof of the strong completeness of BSML^{\vee} does not work for BSML^{\circ} due to the structure of the BSML^{\circ}-normal form.)

2.4.30. THEOREM (Weak completeness of BSML^{\circ}). If $\varphi \models \psi$, then $\varphi \vdash \psi$.

Proof:

Let $k = \max\{md(\varphi), md(\psi)\}$. By Lemma 2.4.27,

$$\varphi \dashv \vdash \eta_{\varphi} \land \bigvee_{s \in \mathcal{P}} \oslash \theta_{s}^{k}$$
 and $\psi \dashv \vdash \eta_{\psi} \land \bigvee_{t \in \mathcal{Q}} \oslash \theta_{t}^{k}$, (2.6)

where each of η_{φ} and η_{ψ} is either NE or (for notational convenience) π . Since $\varphi \models \psi$, $\eta_{\varphi} = \pi$ implies $\eta_{\psi} = \pi$. Each of the remaining possibilities—that is, (a) $\eta_{\varphi} = \eta_{\psi} = \pi$; (b) $\eta_{\varphi} = \eta_{\psi} = \text{NE}$; (c) $\eta_{\varphi} = \text{NE}$ and $\eta_{\psi} = \pi$ —implies by (2.6) and soundness,

$$\bigvee_{s\in\mathcal{P}}\otimes\boldsymbol{\theta}_s^k\vDash\bigvee_{t\in\mathcal{Q}}\otimes\boldsymbol{\theta}_t^k$$

so that by Lemma 2.4.28,

$$\bigvee_{\mathcal{P}'\subseteq\mathcal{P}} \boldsymbol{\theta}_{\boldsymbol{t}'\mathcal{P}'}^{k} \vDash \bigvee_{\mathcal{Q}'\subseteq\mathcal{Q}} \boldsymbol{\theta}_{\boldsymbol{t}'\mathcal{Q}'}^{k}.$$
(2.7)

Let $\mathcal{P}' \subseteq \mathcal{P}$, and $u \models \theta_{\forall \mathcal{P}'}^k$. By (2.7), we have $u \models \bigvee_{\mathcal{Q}' \subseteq \mathcal{Q}} \theta_{\forall \mathcal{Q}'}^k$. Thus, $u \models \theta_{\forall \mathcal{Q}'}^k$ for some $\mathcal{Q}' \subseteq \mathcal{Q}$. By Proposition 2.3.11(ii), $\forall \mathcal{P}' \rightleftharpoons_k u \rightleftharpoons_k \forall \mathcal{Q}'$. Then by Lemma 2.4.10, $\theta_{\forall \mathcal{P}'}^k \vdash \theta_{\forall \mathcal{Q}'}^k$. By Lemma 2.4.29(i), $\theta_{\forall \mathcal{Q}'}^k \vdash \bigvee_{t \in \mathcal{Q}} \otimes \theta_t^k$, so also $\theta_{\forall \mathcal{P}'}^k \vdash \bigvee_{t \in \mathcal{Q}} \otimes \theta_t^k$. Hence, by Lemma 2.4.29(ii), $\bigvee_{s \in \mathcal{P}} \otimes \theta_s^k \vdash \bigvee_{t \in \mathcal{Q}} \otimes \theta_t^k$, and therefore $\varphi \vdash \psi$.

2.4.31. COROLLARY (Strong completeness of BSML^{\circ}). *If* $\Phi \vDash \psi$, *then* $\Phi \vdash \psi$.

Proof:

This is by Theorem 2.4.30 and compactness.

In the remainder of this section we give the inductive proof of Lemma 2.4.27 (normal form provable equivalence). As in Section 2.4.1, we first show normal form provable equivalence for classical formulas.

2.4.32. LEMMA. For each $\alpha \in ML$ and each $k \ge md(\alpha)$, there is some property \mathcal{P} such that

$$\alpha \dashv \vdash \bigvee_{s \in \mathcal{P}} \oslash \theta_s^k.$$

Proof:

Since $[\![\alpha]\!]$ is invariant under *k*-bisimulation, by the proof of Theorem 2.3.5, we have $\alpha \equiv \bigvee_{w \in [\![\alpha]\!]} \chi_w^k$. Then by Proposition 2.4.8, $\alpha \dashv \vdash \bigvee_{w \in [\![\alpha]\!]} \chi_w^k$, and by Lemma 2.4.26 and $\lor Mon$,

$$\bigvee_{w \in \llbracket \alpha \rrbracket} \chi_w^k \to \bigvee_{w \in \llbracket \alpha \rrbracket} \oslash (\chi_w^k \land \operatorname{NE}) = \bigvee_{\{w\} \in \wp(\llbracket \alpha \rrbracket)} \oslash \theta_{\{w\}}^k.$$

Again following the argument in Section 2.4.1, recall that in our inductive proof of Lemma 2.4.27, another important step consists in showing that formulas of the form $\Diamond \theta_s^k$ or $\Box \theta_s^k$ are provably equivalent to classical formulas. In Section 2.4.1, this was proved in Lemma 2.4.23, which made use of Lemma 2.4.22. The two lemmas leading to Lemma 2.4.22 (namely, Lemmas 2.4.17 and 2.4.21) were given, in Section 2.4.1,

only BSML^{\vee}-specific derivations. We now show that these two lemmas can also be derived in the system for BSML^{\circ}; we may then also make use of Lemmas 2.4.22 and 2.4.23 in BSML^{\circ}.

Proof of Lemma 2.4.17 (in BSML[©]):

Proof of Lemma 2.4.21 (in $BSML^{\circ}$):

We derive $\Diamond \varphi \vdash \Diamond (\varphi \land NE)$ as follows:

| $\diamond \phi$ | F | $\Diamond(\phi \land \oslash \operatorname{NE})$ | (⊘nel) |
|-----------------|---|---|------------------------|
| | ⊢ | $\Diamond(\varphi \land \text{NE}) \lor \Diamond(\varphi \land \bot)$ | $(\Diamond \oslash E)$ |
| | F | $\Diamond(\phi \land \text{NE}) \lor \bot$ | (Prop. 2.4.8) |
| | ⊢ | $\Diamond(\phi \wedge \text{NE})$ | (⊥E) |
| | | | |

With all the pieces at hand, we are now ready to give the full proof of Lemma 2.4.27. **Proof of Lemma 2.4.27:**

This is by induction on the complexity of φ (assumed without loss of generality to be in negation normal form). Let $k \ge md(\varphi)$.

If $\varphi = p$ or $\varphi = \neg p$ or $\varphi = \neg NE$, then the result follows by Lemma 2.4.32 and $\neg NEE$. If $\varphi = NE$, then we have $NE \dashv \vdash NE \land (p \lor \neg p)$ by Proposition 2.4.8, and $NE \land (p \lor \neg p) \dashv \vdash NE \land \bigvee_{s \in \mathcal{P}} \oslash \theta_s^k$ by Lemma 2.4.32.

If $\varphi = \psi \land \chi$, then $k \ge md(\varphi) = \max\{md(\psi), md(\chi)\}$, so by the induction hypothesis there are properties \mathcal{P}, \mathcal{Q} such that

$$\Psi \dashv \vdash \eta_{\Psi} \land \bigvee_{s \in \mathcal{P}} \oslash \theta_{s}^{k} \quad \text{and} \quad \chi \dashv \vdash \eta_{\chi} \land \bigvee_{t \in \mathcal{Q}} \oslash \theta_{t}^{k}, \quad (2.8)$$

where each of η_{ψ} and η_{χ} is either π or NE. Let

$$\mathcal{R} \coloneqq \{r \mid \biguplus \mathcal{P}' \rightleftharpoons_k r \rightleftharpoons_k \biguplus \mathcal{Q}' \text{ for some } \mathcal{P}' \subseteq \mathcal{P} \text{ and some } \mathcal{Q}' \subseteq \mathcal{Q}\}$$

We will show that

$$\bigvee_{s\in\mathcal{P}} \oslash \theta_s^k \wedge \bigvee_{t\in\mathcal{Q}} \oslash \theta_t^k \dashv \vdash \bigvee_{r\in\mathcal{R}} \oslash \theta_r^k,$$
(2.9)

from which the result will follow, since if $\eta_{\psi} = \eta_{\chi} = \pi$, then $\varphi \land \psi \dashv \vdash \bigvee_{r \in \mathcal{R}} \oslash \theta_r^k$; and if $\eta_{\psi} = \text{NE}$ or $\eta_{\chi} = \text{NE}$, then $\varphi \land \psi \dashv \vdash \text{NE} \land \bigvee_{s \in \mathcal{P}} \oslash \theta_s^k \land \bigvee_{t \in \mathcal{Q}} \oslash \theta_t^k \dashv \vdash \text{NE} \land \bigvee_{r \in \mathcal{R}} \oslash \theta_r^k$.

Now, for the direction \vdash of (2.9), by Lemma 2.4.29(ii), it suffices to show that $\theta_{\exists \mathcal{P}'}^k, \theta_{\exists \mathcal{Q}'}^k \vdash \bigvee_{s \in \mathcal{R}} \otimes \theta_s^k$ for all $\mathcal{P}' \subseteq \mathcal{P}$ and $\mathcal{Q}' \subseteq \mathcal{Q}$. If $\exists \mathcal{P}' \neq_k \exists \mathcal{Q}'$, then $\theta_{\exists \mathcal{P}'}^k, \theta_{\exists \mathcal{Q}'}^k \vdash \mathcal{Q}'$

 $\blacksquare \vdash \bigvee_{r \in \mathcal{R}} \oslash \theta_r^k \text{ by Lemmas 2.4.11 and 2.4.2. If } \forall \mathcal{P}' \rightleftharpoons_k \forall \mathcal{Q}' \rightleftharpoons_k u \in \mathcal{R}, \text{ then } \theta_{\iota}^k_{!} \mathcal{P}' \dashv \theta_u^k$ by Lemma 2.4.10. Next, we derive by applying $\oslash I$ and $\lor I$ that $\theta_u^k \vdash \oslash \theta_u^k \lor \downarrow \vdash \oslash \theta_u^k \lor$ $\bigvee_{u\neq r\in\mathcal{R}} \oslash \theta_r^k \vdash \bigvee_{r\in\mathcal{R}} \oslash \theta_r^k.$

For the converse direction \dashv of (2.9), we only give the detailed proof for $\bigvee_{r \in \mathcal{R}} \oslash \theta_r^k \vdash$ $\bigvee_{s\in\mathcal{P}}\otimes\theta_s^k$. Given Lemma 2.4.29(ii), it suffices to show that $\theta_{l+\mathcal{R}'}^k \vdash \bigvee_{s\in\mathcal{P}}\otimes\theta_s^k$ for all $\mathcal{R}' \subseteq \mathcal{R}$. Observe that $\mathcal{R}' = \{r \mid r \rightleftharpoons_k \ \forall \mathcal{P}' \text{ and } \mathcal{P}' \in \mathcal{X}\}$ for some $\mathcal{X} \subseteq \wp(\mathcal{P})$ and $\exists \mathcal{R}' \rightleftharpoons_k \exists \bigcup \mathcal{X}$. Thus, $\theta_{\exists \mathcal{R}'}^k \dashv \theta_{\exists \bigcup \mathcal{X}}^k$ by Lemma 2.4.10. Since $\bigcup \mathcal{X} \subseteq \mathcal{P}$, we have by Lemma 2.4.29(i) that $\theta_{l+j}^k \vdash \bigvee_{s \in \mathcal{P}} \otimes \theta_s^k$.

If $\varphi = \psi \lor \chi$, then $k \ge md(\varphi) = \max\{md(\psi), md(\chi)\}$, so by the induction hypothesis we have (2.8). If η_{ψ} = NE and η_{χ} = π , let

$$\mathcal{R} \coloneqq \{\biguplus \mathcal{P}' \uplus \biguplus \mathcal{Q}' \mid \mathcal{P}' \subseteq \mathcal{P} \text{ s.t. } \biguplus \mathcal{P}' \neq_k (M, \emptyset), \text{ and } \mathcal{Q}' \subseteq \mathcal{Q}\},\$$

where *M* is some arbitrary model. We show that $\psi \lor \chi \dashv \vdash \mathbb{NE} \land \bigvee_{r \in \mathcal{R}} \oslash \theta_r^k$.

For the direction \vdash , given Lemma 2.4.29(ii), it suffices to show that for all $\mathcal{P}' \subseteq \mathcal{P}$ and $\mathcal{Q}' \subseteq \mathcal{Q}$, $(\operatorname{NE} \land \theta_{!+\mathcal{P}'}^k) \lor \theta_{!+\mathcal{Q}'}^k \vdash \operatorname{NE} \land \bigvee_{r \in \mathcal{R}} \oslash \theta_r^k$. If $\biguplus \mathcal{P}' \rightleftharpoons_k (M, \varnothing)$, then $\theta_{!+\mathcal{P}'}^k \dashv \vdash \bot$ and we have $(NE \land \theta_{\forall \mathcal{P}'}^k) \lor \theta_{\forall \mathcal{Q}'}^k \vdash \bot \lor \theta_{\forall \mathcal{Q}'}^k \vdash NE \land \bigvee_{r \in \mathcal{R}} \oslash \theta_r^k$ by $\bot Ctr.$ Otherwise $\forall \mathcal{P}' \neq_k \emptyset$, and thus $\forall \mathcal{P}' \forall \forall \mathcal{Q}' \in \mathcal{R}$. We have

$$(\mathrm{NE} \land \boldsymbol{\theta}_{\mathrm{t}}^{k} \mathcal{P}') \lor \boldsymbol{\theta}_{\mathrm{t}}^{k} \mathcal{Q}' \vdash \boldsymbol{\theta}_{\mathrm{t}}^{k} \mathcal{P}' \lor \boldsymbol{\theta}_{\mathrm{t}}^{k} \mathcal{Q}' \dashv \vdash \boldsymbol{\theta}_{\mathrm{t}}^{k} \mathcal{P}' \uplus \boldsymbol{t} \mathcal{Q}' \vdash \bigvee_{r \in \mathcal{R}} \oslash \boldsymbol{\theta}_{r}^{k},$$

by \oslash I and \lor I, and $(NE \land \theta^k_{\flat \mathcal{P}'}) \lor \theta^k_{\flat \mathcal{Q}'} \vdash NE$ by Lemma 2.4.17. For the direction \dashv , given Lemma 2.4.29(ii), it suffices to show that for all $\mathcal{R}' \subseteq \mathcal{R}$, $\mathrm{NE} \land \theta_{t \neq \mathcal{R}'}^k \vdash (\mathrm{NE} \land \bigvee_{s \in \mathcal{P}} \oslash \theta_s^k) \lor \bigvee_{t \in \mathcal{Q}} \oslash \theta_t^k. \text{ If } \mathcal{R}' = \emptyset, \text{ then } \mathrm{NE} \land \theta_{t \neq \mathcal{R}'}^k = \mathrm{NE} \land \bot = \bot, \text{ and } \mathbb{R}' \models \mathbb{R}'$ the implication follows by Lemma 2.4.2. Otherwise

$$\mathcal{R}' = \{ \biguplus \mathcal{P}' \uplus \biguplus \mathcal{Q}' \mid \mathcal{P}' \in \mathcal{X} \text{ and } \mathcal{Q}' \in \mathcal{Y} \}$$

for some nonempty $\mathcal{X} \subseteq \wp(\mathcal{P})$ and nonempty $\mathcal{Y} \subseteq \wp(\mathcal{Q})$ with $\forall \mathcal{P}' \neq_k (M, \emptyset)$ for all $\mathcal{P}' \in \mathcal{X}$. Since $\forall \mathcal{R}' \rightleftharpoons_k \forall \bigcup \mathcal{X} \forall \forall \bigcup \mathcal{Y}$, by Lemma 2.4.10 we have

$$\theta^k_{\uplus \mathcal{R}'} \dashv \vdash \theta^k_{\uplus \sqcup \mathcal{X} \uplus \uplus \sqcup \mathcal{Y}} \dashv \vdash \theta^k_{\uplus \sqcup \mathcal{X}} \lor \theta^k_{\uplus \sqcup \mathcal{Y}}$$

Observe that $\bigcup \mathcal{X} \neq_k (M, \emptyset)$, which implies $\theta_{\bigcup \mathcal{X}}^k \neq \bot$, so that $\theta_{\bigcup \mathcal{X}}^k \vdash \text{NE} \land \theta_{\bigcup \mathcal{X}}^k$ by Lemma 2.4.17. Thus, $\theta_{\bigcup \mathcal{X}}^k \lor \theta_{\bigcup \mathcal{Y}}^k \vdash (\text{NE} \land \theta_{\bigcup \mathcal{X}}^k) \lor \theta_{\bigcup \mathcal{Y}}^k$. Finally, we derive by Lemma 2.4.29(i) that $(NE \land \theta^k_{\forall \cup \mathcal{X}}) \lor \theta^k_{\forall \cup \cup \mathcal{Y}} \vdash (NE \land \bigvee_{s \in \mathcal{P}} \oslash \theta^k_s) \lor \bigvee_{t \in \mathcal{Q}} \oslash \theta^k_t.$ The other cases are similar; for example, if $\eta_{\psi} = \eta_{\chi} = NE$, one uses

$$\mathcal{R} := \{ \biguplus \mathcal{P}' \uplus \biguplus \mathcal{Q}' \mid \mathcal{P}' \subseteq \mathcal{P} \text{ and } \mathcal{Q}' \subseteq \mathcal{Q} \text{ s.t. } \biguplus \mathcal{P}' \neq_k \emptyset \text{ and } \biguplus \mathcal{Q}' \neq_k \emptyset \},\$$

and if $\eta_{\psi} = \eta_{\chi} = \pi$, we clearly have $\varphi \lor \psi \dashv \vdash \bigvee_{u \in \mathcal{P} \cup \mathcal{Q}} \oslash \theta_u^k$.

If $\varphi = \oslash \psi$, then $k \ge md(\varphi) = md(\psi)$, so by the induction hypothesis we have the equivalence for ψ as in (2.8). We show that $\oslash(\eta_{\psi} \land \bigvee_{s \in \mathcal{P}} \oslash \theta_s^k) \dashv \bigvee_{s \in \mathcal{P}} \oslash \theta_s^k$. For the direction \vdash , we have $\eta_{\psi} \land \bigvee_{s \in \mathcal{P}} \oslash \theta_s^k \vdash \bigvee_{s \in \mathcal{P}} \oslash \theta_s^k$ and $\bot \vdash \bigvee_{s \in \mathcal{P}} \bot \vdash \bigvee_{s \in \mathcal{P}} \oslash \theta_s^k$ by \oslash I. Thus we conclude that $\oslash(\eta_{\psi} \land \bigvee_{s \in \mathcal{P}} \oslash \theta_s^k) \vdash \bigvee_{s \in \mathcal{P}} \oslash \theta_s^k$ by \oslash E. For the converse direction \dashv , if $\eta_{\psi} = \pi$, we have $\bigvee_{s \in \mathcal{P}} \oslash \theta_s^k \dashv \vdash \pi \land \bigvee_{s \in \mathcal{P}} \oslash \theta_s^k \vdash \oslash(\pi \land \bigvee_{s \in \mathcal{P}} \oslash \theta_s^k)$ by \oslash I. If $\eta_{\psi} =$ NE, we have $\bigvee_{s \in \mathcal{P}} \oslash \theta_s^k \vdash \bigvee_{s \in \mathcal{P}} \oslash \theta_s^k \land \oslash$ NE by \oslash NEI. Then since $\bigvee_{s \in \mathcal{P}} \oslash \theta_s^k \land$ NE \vdash $\oslash(\bigvee_{s \in \mathcal{P}} \oslash \theta_s^k \land NE)$ by \oslash I and $\bigvee_{s \in \mathcal{P}} \oslash \theta_s^k \land \bot \vdash \bot \vdash \oslash(\bigvee_{s \in \mathcal{P}} \oslash \theta_s^k \land NE)$ by \oslash I, we have $\bigvee_{s \in \mathcal{P}} \oslash \theta_s^k \land \oslash$ NE $\vdash \oslash(\bigvee_{s \in \mathcal{P}} \oslash \theta_s^k \land NE)$ by \oslash E.

If $\varphi = \Diamond \psi$, then $k - 1 \ge md(\varphi) - 1 = md(\psi)$, so by the induction hypothesis there is a property \mathcal{P} such that $\psi \dashv \vdash \eta_{\psi} \land \bigvee_{s \in \mathcal{P}} \oslash \theta_s^{k-1}$, where $\eta_{\psi} = \pi$ or $\eta_{\psi} = \text{NE}$. Let n = k - 1. By Lemma 2.4.32, it suffices to show that ψ is equivalent to some classical formula of modal depth at most k = n + 1. Now, if $\eta_{\psi} = \text{NE}$, we have $\Diamond \psi \dashv \vdash \Diamond (\eta_{\psi} \land \bigvee_{s \in \mathcal{P}} \oslash \theta_s^n) \dashv \vdash \Diamond \bigvee_{s \in \mathcal{P}} \oslash \theta_s^n$ by Lemma 2.4.21; the same equivalence holds also when $\eta_{\psi} = \pi$. It is then sufficient to show that

$$\Diamond \bigvee_{s \in \mathcal{P}} \oslash \theta_s^n \dashv \vdash \bigvee_{\mathcal{P}' \subseteq \mathcal{P}} \Diamond \theta_{\mathfrak{t}}^n \mathcal{P}', \qquad (2.10)$$

for then we have $\bigvee_{\mathcal{P}'\subseteq\mathcal{P}} \diamondsuit \theta_{\forall\mathcal{P}'}^n \dashv \vdash \bigvee_{\mathcal{P}'\subseteq\mathcal{P}} (\diamondsuit \chi_{\forall\mathcal{P}'}^n \land \bigwedge_{w\in \forall\mathcal{P}'} \diamondsuit \chi_w^n)$ by Lemma 2.4.23(i). To prove the direction \dashv of (2.10), for any $\mathcal{P}'\subseteq\mathcal{P}$ we have $\theta_{\forall\mathcal{P}'}^n \vdash \bigvee_{s\in\mathcal{P}} \oslash \theta_s^n$ by

To prove the direction \dashv of (2.10), for any $\mathcal{P}' \subseteq \mathcal{P}$ we have $\theta_{\boxtimes \mathcal{P}'}^n \vdash \bigvee_{s \in \mathcal{P}} \otimes \theta_s^n$ by Lemma 2.4.29(i), so $\Diamond \theta_{\boxtimes \mathcal{P}'}^n \vdash \Diamond \bigvee_{s \in \mathcal{P}} \otimes \theta_s^n$ by \Diamond Mon. The result then follows by $\lor E$. For the converse direction \vdash , let $\mathcal{P} = \{s_1, \dots, s_m\}$. Repeatedly applying $\Diamond \oslash E$ gives

$$\bigotimes_{i=1}^{m} \oslash \theta_{s_{i}}^{n} \vdash \diamondsuit(\theta_{s_{1}}^{n} \lor \oslash \theta_{s_{2}}^{n} \lor \cdots \lor \oslash \theta_{s_{m}}^{n}) \lor \diamondsuit(\bot \lor \oslash \theta_{s_{2}}^{n} \lor \cdots \lor \oslash \theta_{s_{m}}^{n})$$

$$\vdash \diamondsuit(\theta_{s_{1}}^{n} \lor \theta_{s_{2}}^{n} \lor \oslash \theta_{s_{3}}^{n} \lor \cdots \lor \oslash \theta_{s_{m}}^{n}) \lor \diamondsuit(\theta_{s_{1}}^{n} \lor \bot \lor \oslash \theta_{s_{3}}^{n} \lor \cdots \lor \oslash \theta_{s_{m}}^{n})$$

$$\lor \diamondsuit(\bot \lor \theta_{s_{2}}^{n} \lor \oslash \theta_{s_{3}}^{n} \lor \cdots \lor \oslash \theta_{s_{m}}^{n}) \lor \diamondsuit(\bot \lor \bot \lor \oslash \theta_{s_{3}}^{n} \lor \cdots \lor \oslash \theta_{s_{m}}^{n})$$

$$\vdots$$

$$\vdash \bigvee_{\tau_{1} \in \{s_{1}, \varnothing\}} \cdots \bigvee_{\tau_{m} \in \{s_{m}, \varnothing\}} \diamondsuit(\theta_{\tau_{1}}^{n} \lor \cdots \lor \theta_{\tau_{m}}^{n}).$$

$$(\text{recall } \theta_{\varnothing}^{n} = \bot)$$

For each sequence τ_1, \ldots, τ_m as above, putting $\mathcal{P}' = \{\tau_i \mid \tau_i = s_i, 1 \le i \le m\}$, it is easy to see that $\Diamond(\theta_{\tau_1}^n \lor \cdots \lor \theta_{\tau_m}^n) \dashv \vdash \Diamond \bigvee_{\tau_i \in \mathcal{P}'} \theta_{\tau_i}^n \dashv \vdash \Diamond \theta_{i+|\mathcal{P}'}^n$. Thus,

$$\bigvee_{\tau_1\in\{s_1,\varnothing\}}\cdots\bigvee_{\tau_m\in\{s_m,\varnothing\}}\diamondsuit(\theta_{\tau_1}^n\vee\cdots\vee\theta_{\tau_m}^n)\vdash\bigvee_{\mathcal{P}'\subseteq\mathcal{P}}\diamondsuit\theta_{t}^n_{\mathcal{P}'}.$$

If $\varphi = \Box \psi$, then $n = k - 1 \ge md(\varphi) - 1 = md(\psi)$, so by the induction hypothesis, we have the same normal form equivalence for ψ as in the diamond case. Again, by Lemma 2.4.32 it suffices to show that $\Box \psi$ is provably equivalent to a classical formula of modal depth at most k. For the case $\eta_{\psi} = \pi$, we have $\Box \psi \dashv \Box \bigvee_{s \in \mathcal{P}} \oslash \theta_s^n$. One can show that $\Box \bigvee_{s \in \mathcal{P}} \oslash \theta_s^n$ is provably equivalent to a classical formula of modal depth at most k with a proof analogous to that given for $\diamondsuit \bigvee_{s \in \mathcal{P}} \oslash \theta_s^n$ in the case for $\varphi = \diamondsuit \psi$ (using $\Box \text{Mon}$, $\Box \oslash \text{E}$, and Lemma 2.4.23(ii) instead of $\diamondsuit \text{Mon}$, $\diamondsuit \oslash \text{E}$, and Lemma 2.4.23(i), respectively). For the case $\eta_{\psi} = \text{NE}$, we have $\Box \psi \dashv \Box (\text{NE} \land$ $\bigvee_{s \in \mathcal{P}} \oslash \theta_s^n) \dashv \Box \bigvee_{s \in \mathcal{P}} \oslash \theta_s^n \land \bigvee_{s \in \mathcal{P}} \oslash \theta_s^n$ by Lemma 2.4.22(ii). We have already shown that each conjunct in this formula is provably equivalent to a classical formula.
2.4.3 BSML

In this final section, we axiomatize our base logic BSML. As with $BSML^{\vee}$ and PT^+ , the propositional fragment of BSML corresponds to the logic CPL^+ studied and axiomatized in [140] (with the caveat that CPL^+ does not feature a bilateral negation). To construct the system for BSML, we remove all rules involving \vee from the system for BSML^{\vee}, and add rules which simulate the removed rules.

2.4.33. DEFINITION (Natural deduction system for BSML). The natural deduction system for BSML includes all rules not involving \forall from the system for BSML^{\forall} (boxes (a)–(f)) and the following rules simulating the behavior of \forall :



Recall that $[\psi]$ is w-distributive in χ if $[\psi]$ is not within the scope of any \neg or \diamondsuit in χ . So given $\psi \equiv \psi \land (NE \otimes \bot)$, and given that \land and \lor distribute over \otimes , if $[\psi]$ is \otimes -distributive in φ , then $\varphi \equiv \varphi[\psi \land (NE \otimes \bot)/\psi] \equiv \varphi[\psi \land NE/\psi] \lor \varphi[\psi \land \bot/\psi]$. The $\bot NE$ -translation rule $\bot NE$ Trs captures this equivalence by simulating \otimes -elimination as applied to $\varphi[\psi \land NE/\psi] \lor \varphi[\psi \land \bot/\psi]$. The rule $\diamondsuit \bot NE$ Trs captures the equivalence $\diamondsuit \varphi \equiv \diamondsuit(\varphi[\psi \land NE/\psi] \lor \varphi[\psi \land \bot/\psi]) \equiv \diamondsuit \varphi[\psi \land NE/\psi] \lor \diamondsuit \varphi[\psi \land \bot/\psi]$ by simulating $\diamondsuit \lor Conv$; similarly for $\Box \bot NE$ Trs and $\Box \lor \lor Conv$. The soundness of the new rules follows from these equivalences, and so we have the following result.

2.4.34. THEOREM (Soundness of BSML). *If* $\Phi \vdash \psi$, *then* $\Phi \models \psi$.

Observe that the proofs given in Section 2.4.1 for Lemma 2.4.4 (replacement) and Proposition 2.4.5 (negation normal form) also suffice to establish these results for the BSML-system. In what follows we can thus always assume that formulas are in negation normal form.

We now turn to the proof of the completeness theorem for the system. From our expressivity analysis in Section 2.3, we know that every formula φ in BSML (being

also a formula in the stronger logic BSML^V) is equivalent to a formula in BSML^Vnormal form: $\varphi \equiv \bigvee_{s \in \|\varphi\|} \theta_s^k$, where $k = md(\varphi)$. While the global disjunction in the normal form is not in the language of BSML, we will still be able to make use of this disjunctive normal form in our completeness proof, as we were also able to do in our proof for BSML^O. Adapting a similar strategy employed in [140] for CPL⁺, we associate with each formula φ a set of BSML-formulas φ^f of certain syntactic form, called *realizations* of φ . The realizations of φ correspond to the disjuncts θ_s^k in the normal form for φ —we will show that $\varphi \equiv \bigvee_{s \in \|\varphi\|} \theta_s^k \equiv \bigvee \{\varphi^f \mid \varphi^f \text{ is a realization for} \varphi\}$ (cf. Lemma 2.4.28)—and φ can be simulated in the completeness proof using its realizations: we also show that φ and $\bigvee \{\varphi^f \mid \varphi^f \text{ is a realization for } \varphi\}$ are prooftheoretically interchangeable (cf. Lemma 2.4.29).

Realizations for φ are defined by replacing \vee -distributive occurrences $[\eta]$ in φ by disjuncts θ_s^k in their respective normal forms $\bigvee_{\|\eta\|} \theta_s^k$ (where $k \ge md(\varphi)$). The equivalence between φ and $\bigvee_{\{\varphi^f \mid \varphi^f \text{ is a realization for } \varphi\}}$ then follows by the \vee distributivity of these occurrences. In order to secure the proof-theoretic interchangeability result, we require each η to be replaced to be either classical or NE—we first essentially show that the required results hold for formulas that can be constructed using only such occurrences (that is, they hold for formulas of the form $\eta_1 \circ_1 \eta_2 \circ_2 \cdots \circ_{n-1} \eta_n$ where each $\eta_i \in ML$ or $\eta_i = NE$, and each \circ_i is either \wedge or \vee), and later show that every formula is provably equivalent to a formula of this form.

Formally, we define realizations relative to \forall -distributive sequences, and we show that our results hold with respect to such sequences. A \forall -distributive sequence of φ is a sequence $a = \langle \eta_1, \dots, \eta_n \rangle$ of distinct \forall -distributive occurrences of formulas η_i in φ such that each $\eta_i \in ML$ or $\eta_i = NE$ for all $1 \le i \le n$.

Let $a = \langle \eta_1, ..., \eta_n \rangle$ be a \forall -distributive sequence of φ , and let $X \supseteq P(\varphi)$ be some finite set of propositional variables. A *realizing function over* a *and* X is a function $f : \{1, ..., n\} \rightarrow ||\pi||_X$ such that $f_i \models \eta_i$ for all $1 \le i \le n$, where $||\pi||_X = \{(M, s) \mid M, s \models \pi\}$ is the class of all pointed state models over X. For any $k \ge md(\varphi)$, the (k, f)-realization of φ , written $\varphi^{k, f}$ (or simply φ^f), is the BSML-formula

$$\varphi[\theta_{f_1}^k/\eta_1,\ldots,\theta_{f_n}^k/\eta_n],$$

where recall that each $\theta_{f_i}^k$ is the strong Hintikka formula for the state f_i . We write $\mathcal{F}^{X,a}$ (or simply \mathcal{F}^a) for the set of all realizing functions over a and X.

Most results in this section hold for all finite $X \supseteq P(\varphi)$ and all $k \ge md(\varphi)$ where φ is clear from the context; we usually omit mention of these conditions. As promised, each formula is equivalent to the global disjunction of all its realizations.

2.4.35. LEMMA. Let a be a \lor -distributive sequence of φ . Then $\varphi \equiv \bigvee_{f \in \mathcal{F}^a} \varphi^f$.

Proof:

By the proof of Theorem 2.3.15, $\eta_i \equiv \bigvee_{s \in \|\eta_i\|} \theta_s^k$ for each *i*. The result then follows since a is \forall -distributive, and \land and \lor distribute over \forall .

2.4. Axiomatizations

As explained above, we will also show that in the system for BSML, every formula is proof-theoretically interchangeable with the global disjunction of its realizations. This is expressed formally in the following lemma.

2.4.36. LEMMA. Let a be a \vee -distributive sequence of φ . Then

- (*i*) $\varphi^f \vdash \varphi$ for all $f \in \mathcal{F}^a$, and
- (*ii*) *if* $\Phi, \varphi^f \vdash \psi$ *for all* $f \in \mathcal{F}^a$ *, then* $\Phi, \varphi \vdash \psi$ *.*

The proof of the above requires several technical lemmas. For item (i), we need to show that Lemma 2.4.17 is also derivable in the system for BSML.

Proof of Lemma 2.4.17 (in BSML):

We prove $\varphi \lor (\psi \land \text{NE}) \dashv \vdash (\varphi \lor (\psi \land \text{NE})) \land \text{NE}$. Let $\chi := \varphi \lor (\psi \land \text{NE})$. Then $\chi[\chi \land \bot/\chi] = (\varphi \lor (\psi \land \text{NE})) \land \bot \vdash (\varphi \land \bot) \lor (\psi \land \text{NE} \land \bot)$ by Proposition 2.4.6, and $(\varphi \land \bot) \lor (\psi \land \text{NE} \land \bot) \vdash (\varphi \land \bot) \lor \bot \vdash (\varphi \lor (\psi \land \text{NE})) \land \text{NE}$ by $\bot \text{Ctr.}$ And $\chi[\chi \land \text{NE}/\chi] = (\varphi \lor (\psi \land \text{NE})) \land \text{NE}$ here $\varphi \lor (\psi \land \text{NE}) \vdash (\varphi \lor (\psi \land \text{NE})) \land \text{NE}$ by $\bot \text{NE}$ for $\psi \lor (\psi \land \text{NE}) \land \text{NE}$.

Proof of Lemma 2.4.36(i):

By Lemma 2.4.4, it suffices to show that $\theta_{f_i}^k \vdash \eta_i$ for all $1 \le i \le n$. If $\eta_i = \text{NE}$, then $f_i \ne \emptyset$ so $\theta_{f_i}^k$ is not \bot . We have $\theta_{f_i}^k = \bigvee_{w \in f_i} (\chi_w^k \land \text{NE}) \vdash \text{NE}$ by Lemma 2.4.17. If $\eta_i = \alpha \in \text{ML}$, we clearly have that $\theta_{f_i}^k \vdash \chi_{f_i}^k$. Since $f_i \models \alpha$, we have $f_i \subseteq [\alpha]$ by flatness, which implies $\chi_{f_i}^k \models \chi_{[\alpha]}^k \equiv \alpha$. Therefore, $\chi_{f_i}^k \vdash \alpha$ by Proposition 2.4.8.

To prove Lemma 2.4.36(ii), we first show that for a \vee -distributive sequence $a = \langle \eta \rangle$ of length 1, the entailment $\varphi[\eta] \models \varphi[\pi \land \eta/\eta] \models \varphi[\bigvee_{s \in \|\pi\|_{X}} \theta_{s}^{X,k} \land \eta/\eta]$ can be simulated in our system in the sense of the following lemma.

2.4.37. LEMMA. Let $[\eta]$ be \forall -distributive in φ . Let $k \in \mathbb{N}$, and let $X \subseteq$ Prop be finite. If $\Phi, \varphi[\theta_s^{X,k} \land \eta/\eta] \vdash \psi$ for all $s \in ||\pi||_X$, then $\Phi, \varphi \vdash \psi$.

Proof:

Consider $\chi_{\llbracket\Pi\rrbracket}^k := \bigvee_{w \in \llbracket\Pi\rrbracket_X} \chi_w^k$, where $\llbracket\Pi\rrbracket_X = \{w \mid w \models \Pi\}$. Since there are only finitely many nonequivalent *k*th Hintikka formulas χ_w^k over X, we may assume without loss of generality that $\chi_{\llbracket\Pi\rrbracket}^k = \chi_{w_1}^k \lor \cdots \lor \chi_{w_n}^k$, for some worlds w_1, \ldots, w_n from some models M_1, \ldots, M_n . Clearly, $\chi_{\llbracket\Pi\rrbracket}^k \equiv \Pi$, which implies $\vdash \chi_{\llbracket\Pi\rrbracket}^k$ by Proposition 2.4.8. Thus, we have $\varphi \vdash \varphi[\chi_{\llbracket\Pi\rrbracket}^k \land \eta/\eta]$ by Lemma 2.4.4.

Now, for any $\tau_1, \ldots, \tau_n \in \{NE, \bot\}$, consider the state $s = \{w_i \mid \tau_i = NE\}$. By applying $\bot E$ and $\lor I$, we derive $\bigvee_{i=1}^n (\chi_{w_i}^k \land \tau_i) \dashv \vdash \bigvee_{w_i \in s} (\chi_{w_i}^k \land NE) = \theta_s^k$. Thus, we have $\varphi[\bigvee_{i=1}^n (\chi_{w_i}^k \land \tau_i) \land \eta/\eta] \dashv \vdash \varphi[\theta_s^k \land \eta/\eta]$. Therefore, by assumption we have

$$\Phi, \varphi[((\chi_{w_1}^k \wedge \bot) \vee \cdots \vee (\chi_{w_n}^k \wedge \bot)) \wedge \eta/\eta] \vdash \psi \quad \text{and} \quad$$

$$\Phi, \varphi[\left((\chi_{w_1}^k \wedge \bot) \vee \cdots \vee (\chi_{w_n}^k \wedge \operatorname{NE})\right) \wedge \eta/\eta] \vdash \psi,$$

from which $\Phi, \varphi[((\chi_{w_1}^k \wedge \bot) \vee \cdots \vee (\chi_{w_{n-1}}^k \wedge \bot) \vee \chi_{w_n}^k) \wedge \eta/\eta] \vdash \psi$ follows by \bot NETrs. Repeating this argument, we finally get $\Phi, \varphi[\chi_{[[T]]}^k \wedge \eta/\eta] \vdash \psi$, whence $\Phi, \varphi \vdash \psi$. \Box

Next, in (i) of the following lemma ((ii) is used later) we show that for a \forall -distributive sequence $a = \langle \eta_1, \dots, \eta_n \rangle$ of arbitrary length, we can simulate the entailment

$$\varphi[\bigvee_{s\in \|\pi\|_{\mathcal{X}}} \theta_s^{\mathcal{X},k} \wedge \eta_1/\eta_1, \ldots, \bigvee_{s\in \|\pi\|_{\mathcal{X}}} \theta_s^{\mathcal{X},k} \wedge \eta_n/\eta_n] \vDash \bigvee_{f\in \mathcal{F}^{\mathsf{X},\mathsf{a}}} \varphi^{k,f}.$$

2.4.38. LEMMA. Let a be a \lor -distributive sequence of φ . Then:

(i) for any $s_1, \ldots, s_n \in ||\pi||_X$, there is some $f \in \mathcal{F}^a$ such that

$$\varphi[\theta_{s_1}^{\mathsf{X},k} \wedge \eta_1/\eta_1,\ldots,\theta_{s_n}^{\mathsf{X},k} \wedge \eta_n/\eta_n] \vdash \varphi^{k,f};$$

(*ii*) for any $f \in \mathcal{F}^a$, there are some $s_1, \ldots, s_n \in ||\pi||_X$ such that

$$\varphi[\theta_{s_1}^{\mathsf{X},k} \wedge \eta_1/\eta_1,\ldots,\theta_{s_n}^{\mathsf{X},k} \wedge \eta_n/\eta_n] \vdash \varphi^{k,f}.$$

Proof:

(i) We first give the proof for the case when $a = \langle \eta_1 \rangle$ is a sequence of length 1. Let $(M_1, s_1) \in ||\pi||_X$. Case 1: $\eta_1 = \text{NE}$. If $s_1 \neq \emptyset$, let $f_1 := (M_1, s_1)$. Since $s_1 \models \text{NE}$, we have $f \in \mathcal{F}^a$. We derive $\varphi[\theta_{s_1}^k \wedge \text{NE}/\text{NE}] \vdash \varphi[\theta_{s_1}^k/\text{NE}] = \varphi^f$ by Lemma 2.4.4. If $s_1 = \emptyset$, pick any (N, t) such that $t \neq \emptyset$ and let $f_1 := (N, t)$. Since $t \models \text{NE}$, we have $f \in \mathcal{F}^a$. Now, we derive $\theta_{s_1}^k \wedge \text{NE} = \bot \wedge \text{NE} = \bot \vdash \theta_t^k$ by Lemma 2.4.2. Thus, $\varphi[\theta_{s_1}^k \wedge \text{NE}/\text{NE}] \vdash \varphi[\theta_t^k/\text{NE}] = \varphi^f$ follows from Lemma 2.4.4.

Case 2: $\eta_1 = \alpha \in ML$. Let $r = \{w \in s_1 \mid w \models \alpha\}$. Define $f_1 := (M_1, r)$, which yields a realizing function over $\langle \eta_1 \rangle$ (as $r \models \alpha$). We show that $\varphi[\theta_{s_1}^k \land \alpha/\alpha] \vdash \varphi^f$, that is, $\varphi[\theta_{s_1}^k \land \alpha/\alpha] \vdash \varphi[\theta_r^k/\alpha]$. By Lemma 2.4.4, it suffices to show that $\theta_{s_1}^k \land \alpha \vdash \theta_r^k$. Now, by Proposition 2.4.6, we derive

$$\theta_{s_1}^k \wedge \alpha \vdash (\theta_r^k \vee \theta_{s_1 \setminus r}^k) \wedge \alpha \vdash \theta_r^k \vee (\theta_{s_1 \setminus r}^k \wedge \alpha) \vdash \theta_r^k \vee (\chi_{s_1 \setminus r}^k \wedge \alpha).$$

Clearly, $\chi_{s_1\setminus r}^k \vDash \neg \alpha$ and thus $\chi_{s_1\setminus r}^k \vdash \neg \alpha$ by Proposition 2.4.8. Thus, we further derive $\theta_r^k \lor (\chi_{s_1\setminus r}^k \land \alpha) \vdash \theta_r^k \lor (\neg \alpha \land \alpha) \vdash \theta_r^k \lor \bot \vdash \theta_r^k$ by $\bot \vDash$. We thus conclude that $\theta_{s_1}^k \land \alpha \vdash \theta_r^k$.

Now let $\mathbf{a} = \langle \eta_1, \dots, \eta_n \rangle$ with n > 1. Let $\varphi' := \varphi[\theta_{s_2}^k \wedge \eta_2/\eta_2, \dots, \theta_{s_n}^k \wedge \eta_n/\eta_n]$ and $\mathbf{a}' := \langle \eta_1 \rangle$. Applying what we just proved for φ' and \mathbf{a}' we have $\varphi'[\theta_{s_1}^k \wedge \eta_1/\eta_1] \vdash \varphi'^g$ for some $g \in \mathcal{F}^{\mathbf{a}'}$. Repeating this argument *n* times, we can find some $f \in \mathcal{F}^{\mathbf{a}}$ such that $\varphi[\theta_{s_1}^k \wedge \eta_1/\eta_1, \dots, \theta_{s_n}^k \wedge \eta_n/\eta_n] \vdash \varphi^f$.

 $\varphi[\theta_{s_1}^k \land \eta_1/\eta_1, \dots, \theta_{s_n}^k \land \eta_n/\eta_n] \vdash \varphi^f.$ (ii) Define $(M_i, s_i) \coloneqq f_i$. Since $\theta_{s_i}^k \land \eta_i \vdash \theta_{s_i}^k$ for each *i*, we derive by Lemma 2.4.4 that $\varphi[\theta_{s_1}^k \land \eta_1/\eta_1, \dots, \theta_{s_n}^k \land \eta_n/\eta_n] \vdash \varphi[\theta_{s_1}^k/\eta_1, \dots, \theta_{s_n}^k/\eta_n] = \varphi^f.$

2.4. Axiomatizations

Putting together Lemmas 2.4.37 and 2.4.38(i), we can simulate $\varphi \models \bigvee_{f \in \mathcal{F}^a} \varphi^f$; that is, we can prove Lemma 2.4.36(ii).

Proof of Lemma 2.4.36(ii):

Assume that $\Phi, \varphi^f \vdash \psi$ for all $f \in \mathcal{F}^a$. By multiple applications of Lemma 2.4.37, it suffices to show that $\Phi, \varphi[\theta_{s_1}^k \land \eta_1/\eta_1, \dots, \theta_{s_n}^k \land \eta_n/\eta_n] \vdash \psi$ for all $s_1, \dots, s_n \in ||\pi||_X$. This follows by Lemma 2.4.38(i) and our assumption.

We have now settled that relative to w-distributive sequences, each formula is prooftheoretically interchangeable with the global disjunction of all its realizations. For a formula φ that can be constructed solely using the occurrences in such a sequence a, the realizations over a correspond with (and are, in fact, provably equivalent to) the disjuncts θ_s^k in the normal form $\bigvee_{s \in \|\varphi\|} \theta_s^k$ for φ . We must now show that every formula is provably equivalent to some formula of this form. Formally, we say that a w-distributive sequence $a = \langle \eta_1, \dots, \eta_n \rangle$ of a formula is in addition a w-distributive partition of φ if φ is of the form $[\eta_1] \circ_1 [\eta_2] \circ_2 \cdots \circ_{n-1} [\eta_n]$, where each \circ_i is either \wedge or \vee . We show the following.

2.4.39. LEMMA. For all φ there is a φ' such that φ' has a partition and $\varphi \dashv \varphi'$.

Before we prove this lemma, let us confirm that realizations over \forall -distributive partitions are indeed provably equivalent to strong Hintikka formulas θ_s^k . Hereafter, to simplify notation, we write $\theta_0^k := \bot$ and view **0** as a null state such that $\mathbf{0} \models \bot$.

2.4.40. LEMMA. Let a be a \lor -distributive partition of φ . Then for all $f \in \mathcal{F}^{X,a}$, we have that $\varphi^{k,f} \rightarrow \vdash \theta_s^{X,k}$ for some $s \in ||\pi||_X$ or $s = \mathbf{0}$.

Proof:

It suffices to show that for all states *s* and *t* (including the null state), $\theta_s^k \wedge \theta_t^k \dashv \mapsto \theta_u^k$ and $\theta_s^k \vee \theta_t^k \dashv \mapsto \theta_v^k$ for some (null or nonnull) states *u* and *v*. If one of *s* and *t* is **0**, then $\theta_s^k \wedge \theta_t^k \dashv \mapsto \theta_0^k$ and $\theta_s^k \vee \theta_t^k \dashv \mapsto \theta_0^k$ by Lemma 2.4.2 and \bot Ctr. If neither state is **0**, then $\theta_s^k \vee \theta_t^k \dashv \mapsto \theta_{s \lor t}^k$. For $\theta_s^k \wedge \theta_t^k$, if $s \neq_k t$, then $\theta_s^k \wedge \theta_t^k \dashv \mapsto \theta_0^k$ by Lemmas 2.4.11 and 2.4.2; if $s \Rightarrow_k t$, then $\theta_s^k \dashv \mapsto \theta_t^k$ by Lemma 2.4.10, so clearly $\theta_s^k \wedge \theta_t^k \dashv \mapsto \theta_s^k$.

We now turn to Lemma 2.4.39. This result is, as it turns out, an easy corollary of the fact that all formulas of the form $\Diamond \varphi$ or $\Box \varphi$ are provably equivalent to classical formulas. To establish this fact, we first recall that in Lemma 2.4.23 in Section 2.4.1, we proved that formulas of the form $\Diamond \theta_s^k$ or $\Box \theta_s^k$ are provably equivalent to classical formulas. Lemma 2.4.23 depends on Lemma 2.4.22, which in turn depends on Lemmas 2.4.17 and 2.4.21, which were given BSML^V-specific derivations in Section 2.4.1. We have already shown Lemma 2.4.17 for BSML, and directly below we show Lemma 2.4.21; we may then also make use of Lemmas 2.4.22 and 2.4.23 in BSML.

Proof of Lemma 2.4.21 (in BSML):

By $\Diamond \perp \text{NETrs}, \Diamond \varphi \vdash \Diamond (\varphi \land \text{NE}) \lor \Diamond (\varphi \land \bot)$. We have $\Diamond (\varphi \land \text{NE}) \lor \Diamond (\varphi \land \bot) \vdash \Diamond (\varphi \land \bot)$

NE) $\lor \diamond \bot \vdash \diamond (\varphi \land \text{NE}) \lor \bot$ by Proposition 2.4.8, and finally $\diamond (\varphi \land \text{NE}) \lor \bot \vdash \diamond (\varphi \land \text{NE})$ by $\bot E$.

We also need the following modal analogue of Lemma 2.4.37.

2.4.41. LEMMA. Let $[\eta]$ be \forall -distributive in φ . For any $k \in \mathbb{N}$ and any finite $X \subseteq \text{Prop}$:

$$\Diamond \varphi \vdash \bigvee_{s \in \|\pi\|_{\mathsf{X}}} \Diamond \varphi [\theta_s^{\mathsf{X},k} \land \eta/\eta] \quad and \quad \Box \varphi \vdash \bigvee_{s \in \|\pi\|_{\mathsf{X}}} \Box \varphi [\theta_s^{\mathsf{X},k} \land \eta/\eta]$$

Proof:

We prove the result for $\Diamond \varphi$. Consider $\chi_{\llbracket \pi \rrbracket}^k = \chi_{w_1}^k \lor \cdots \lor \chi_{w_n}^k$, where w_1, \ldots, w_n are some worlds from some models M_1, \ldots, M_n . Since $\vdash \chi_{\llbracket \pi \rrbracket}^k$, we derive $\Diamond \varphi \vdash \Diamond \varphi[\chi_{\llbracket \pi \rrbracket}^k \land \eta/\eta]$ by Lemma 2.4.4. Put $\varphi' \coloneqq \varphi[\chi_{\llbracket \pi \rrbracket}^k \land \eta/\eta]$. By applying $\Diamond \perp \text{NETrs}$ repeatedly, we derive

$$\begin{split} & \diamond \varphi' \vdash & \diamond \varphi'[\chi_{w_{1}}^{k} \wedge \operatorname{NE}/\chi_{w_{1}}^{k}] \lor \diamond \varphi'[\chi_{w_{1}}^{k} \wedge \bot/\chi_{w_{1}}^{k}] \\ & \vdash & \diamond \varphi'[\chi_{w_{1}}^{k} \wedge \operatorname{NE}/\chi_{w_{1}}^{k}, \chi_{w_{2}}^{k} \wedge \operatorname{NE}/\chi_{w_{2}}^{k}] \lor \diamond \varphi'[\chi_{w_{1}}^{k} \wedge \operatorname{NE}/\chi_{w_{1}}^{k}, \chi_{w_{2}}^{k} \wedge \bot/\chi_{w_{2}}^{k}] \lor \\ & \diamond \varphi'[\chi_{w_{1}}^{k} \wedge \bot/\chi_{w_{1}}^{k}, \chi_{w_{2}}^{k} \wedge \operatorname{NE}/\chi_{w_{2}}^{k}] \lor \diamond \varphi'[\chi_{w_{1}}^{k} \wedge \bot/\chi_{w_{1}}^{k}, \chi_{w_{2}}^{k} \wedge \bot/\chi_{w_{2}}^{k}] \\ & \cdots \\ & \vdash & \bigvee_{\tau_{1}, \dots, \tau_{n} \in \{\operatorname{NE}, \bot\}} \diamond \varphi'[\chi_{w_{1}}^{k} \wedge \tau_{1}/\chi_{w_{1}}^{k}, \dots, \chi_{w_{n}}^{k} \wedge \tau_{n}/\chi_{w_{n}}^{k}] \\ & \vdash & \bigvee_{\tau_{1}, \dots, \tau_{n} \in \{\operatorname{NE}, \bot\}} \diamond \varphi[\bigvee_{i=1}^{n} (\chi_{w_{i}}^{k} \wedge \tau_{i}) \wedge \eta/\eta] \\ & \vdash & \bigvee_{\tau_{1}, \dots, \tau_{n} \in \{\operatorname{NE}, \bot\}} \diamond \varphi[\bigvee_{i=1}^{n} (\chi_{w_{i}}^{k} \wedge \tau_{i}) \wedge \eta/\eta] \end{split}$$

Now, consider each disjunct of the above formula, with some arbitrary fixed $\tau_1, \ldots, \tau_n \in \{NE, \bot\}$. Let $s = \{w_i \mid \tau_i = NE\} \subseteq \{w_1, \ldots, w_n\}$. By applying $\bot E$ and $\lor I$, we derive $\bigvee_{i=1}^n (\chi_{w_i}^k \wedge \tau_i) \dashv \bigcup_{w_i \in S} (\chi_{w_i}^k \wedge NE) = \theta_s^k$, whence by Lemma 2.4.4, $\diamondsuit \varphi[\bigvee_{i=1}^n (\chi_{w_i}^k \wedge \tau_i) \land \eta/\eta] \vdash \diamondsuit \varphi[\theta_s^k \land \eta/\eta]$. We then clearly also have $\bigvee_{\tau_1, \ldots, \tau_n \in \{NE, \bot\}} \diamondsuit \varphi[\bigvee_{i=1}^n (\chi_{w_i}^k \land \tau_i) \land \eta/\eta] \vdash \bigcup_{s \subseteq \{w_1, \ldots, w_n\}} \diamondsuit \varphi[\theta_s^k \land \eta/\eta]$. Observe that $\|\Pi\|_X$ is (modulo *k*-bisimulation) $\{(M_1 \uplus \cdots \uplus M_n, s) \mid s \subseteq \{w_1, \ldots, w_n\}\}$. Thus, we have $\bigvee_{s \subseteq \{w_1, \ldots, w_n\}} \diamondsuit \varphi[\theta_s^k \land \eta/\eta] \vdash \bigvee_{s \in \|\Pi\|_X} \diamondsuit \varphi[\theta_s^k \land \eta/\eta]$. Putting all these together, we obtain $\diamondsuit \varphi \vdash \bigvee_{s \in \|\Pi\|_X} \diamondsuit \varphi[\theta_s^k \land \eta/\eta]$.

The $\Box \varphi$ -result is proved analogously, using $\Box \perp \text{NETrs}$ in place of $\Diamond \perp \text{NETrs}$. \Box

We can now show that formulas of the form $\Diamond \varphi$ or $\Box \varphi$ are provably equivalent to classical formulas, and then derive the partition lemma as a corollary.

2.4.42. LEMMA. $\Diamond \varphi$ and $\Box \varphi$ are provably equivalent to classical formulas.

Proof:

We prove the two results simultaneously but only give the details for $\Diamond \varphi$; the details for $\Box \varphi$ are similar. We first prove that the result holds in case φ has a partition a. Let $k \ge md(\varphi)$, and let $X \supseteq P(\varphi)$ be finite. We first show that

$$\Diamond \varphi \dashv \vdash \bigvee_{f \in \mathcal{F}^{\mathsf{X},\mathsf{a}}} \Diamond \varphi^{k,f}$$

For the direction \neg , for a given $f \in \mathcal{F}^a$, we have $\varphi^f \vdash \varphi$ by Lemma 2.4.36(i), which implies $\Diamond \varphi^f \vdash \Diamond \varphi$; therefore, $\bigvee_{f \in \mathcal{F}^a} \Diamond \varphi^f \vdash \Diamond \varphi$ follows by $\lor \mathsf{E}$. For the converse direction \vdash , by repeated applications of Lemma 2.4.41, we have

$$\Diamond \varphi \vdash \bigvee_{s_1 \in \|\pi\|_{\mathsf{X}}} \cdots \bigvee_{s_n \in \|\pi\|_{\mathsf{X}}} \Diamond \varphi [\theta_{s_1}^k \land \eta_1/\eta_1, \dots, \theta_{s_n}^k \land \eta_n/\eta_n]$$

By Lemma 2.4.38(i) and (ii) together with \lor Mon and \diamondsuit Mon, we derive

$$\bigvee_{s_1\in \|\pi\|_{\mathsf{X}}} \cdots \bigvee_{s_n\in \|\pi\|_{\mathsf{X}}} \diamondsuit \varphi[\theta_{s_1}^k \land \eta_1/\eta_1, \ldots, \theta_{s_n}^k \land \eta_n/\eta_n] \vdash \bigvee_{f\in \mathcal{F}^{\mathsf{a}}} \diamondsuit \varphi^f.$$

It now suffices to show that $\bigvee_{f \in \mathcal{F}^a} \Diamond \varphi^f$ is provably equivalent to a classical formula. We show this by showing that each disjunct $\Diamond \varphi^f$ is so. By Lemma 2.4.40, we have $\varphi^f \to \Theta_s^k$ for some state *s*. Thus, also $\Diamond \varphi^f \to \Theta_s^k$. If $s = \mathbf{0}$, then $\Diamond \Theta_s^k = \Diamond \bot \to (\Diamond \bot \to \Box)$ by Proposition 2.4.8 and Lemma 2.4.21. If $s \neq \mathbf{0}$, then by Lemma 2.4.23(i), we know that $\Diamond \Theta_s^k$ is provably equivalent to a classical formula.

We now show the general case by induction on the modal depth of φ . Note that by ¬NEE, we may assume without loss of generality that $\varphi = \varphi(\perp/\neg NE)$. If $md(\varphi) = 0$, then there is clearly a partition of φ , so the result follows from what we just proved. If $md(\varphi) = n + 1$, then for all subformulas of φ of form $\Diamond \psi$ or $\Box \psi$, we have $md(\psi) \leq n$, and thus by the induction hypothesis $\Diamond \psi \dashv \vdash \alpha_{\Diamond \psi}$ and $\Box \psi \dashv \vdash \alpha_{\Box \psi}$ for some $\alpha_{\Diamond \psi}, \alpha_{\Box \psi} \in ML$. By Lemma 2.4.4, $\varphi \dashv \vdash \varphi' := \varphi[\alpha_{\bigcirc 1}\psi_1/\bigcirc_1\psi_1, \dots, \alpha_{\bigcirc n}\psi_n/\bigcirc_n\psi_n]$, where $[\bigcirc_1\psi_1], \dots, [\bigcirc_n\psi_n]$ are all the subformula occurrences of the form $\Diamond \psi$ or $\Box \psi$ in φ such that no $[\bigcirc_i\psi_i]$ is a suboccurrence of any other occurrence of the form $\Diamond \psi$ or $\Box \psi$ or $\Box \psi$ in φ . The formula φ' clearly has a partition, so $\Diamond \varphi' \dashv \vdash \alpha \in ML$; then also $\Diamond \varphi \dashv \vdash \alpha$.

Proof of Lemma 2.4.39:

Apply Lemma 2.4.42, \neg NEE, and Lemma 2.4.4 to replace all subformula occurrences of the form $\Diamond \psi$, $\Box \psi$, or \neg NE in φ with classical formulas.

We are now ready to prove the completeness theorem for the BSML-system.

2.4.43. THEOREM (Completeness of BSML). If $\Phi \vDash \psi$, then $\Phi \vdash \psi$.

Proof:

Assume that $\Phi \not\vdash \psi$. We show that $\Phi \not\models \psi$. Let $\Phi = \{\varphi_i \mid i \in I\}$. By Lemma 2.4.39,

we may assume that each φ_i has a partition a_i and ψ has a partition a. For each i, let $k_i := \max\{md(\varphi_i), md(\psi)\}$ and $X_i := \mathsf{P}(\varphi_i) \cup \mathsf{P}(\psi)$. Now, for an arbitrary $i \in I$, since $\Phi \setminus \{\varphi_i\}, \varphi_i \not\vdash \psi$, by Lemma 2.4.36(ii), we know there is a realizing function $f_i \in \mathcal{F}^{X_i, a_i}$ such that $\Phi \setminus \{\varphi_i\}, \varphi_i^f \not\vdash \psi$. Continuing to argue in a similar way, one can find for each $i \in I$ a realizing function $f_i \in \mathcal{F}^{X_i, a_i}$ such that $\{\varphi_i^{k_i, f_i} \mid f_i \in \mathcal{F}^{X_i, a_i}, i \in I\} \not\vdash \psi$. By Lemma 2.4.40, each $\varphi_i^{k_i, f_i}$ is provably equivalent to $\theta_{s_i}^{X_i, k_i}$ for some s_i . Thus, $\Phi' = \{\theta_{s_i}^{X_i, k_i} \mid i \in I\} \not\vdash \psi$ as well.

Observe that $\Phi' \not\models \bot$, since otherwise either $s_i = \mathbf{0}$ for some $i \in I$, or $s_i \neq_m^M s_j$ for some $i, j \in I$ and for $m = \min\{k_i, k_j\}$ and $M = X_i \cap X_j$. In the former case, $\theta_{s_i}^{X_i, k_i} = \bot$ and thus $\Phi' \vdash \psi$; a contradiction. In the latter case, $\theta_{s_i}^{M,m}, \theta_{s_j}^{M,m} \vdash \bot$ by Lemma 2.4.11, so that by Lemma 2.4.14, $\theta_{s_i}^{X_i, k_i}, \theta_{s_i}^{X_j, k_j} \vdash \bot$. Thus, $\Phi' \vdash \psi$; a contradiction again.

Now, let *t* be such that $t \models \Phi'$. For each $i \in I$, we have $t \models \theta_{s_i}^{k_i} \dashv \varphi_i^{f_i}$, and thus $t \models \varphi_i^{f_i}$ by soundness. By Lemma 2.4.35, we further conclude that $t \models \varphi_i$, whereby $t \models \Phi$. To show that $\Phi \not\models \psi$, it then suffices to show that $t \not\models \psi$. Assume otherwise. Take $i \in I$. By Lemma 2.4.35, $t \models \psi^{k_i,g}$ for some $g \in \mathcal{F}^{X_i,a}$. In view of Lemma 2.4.40, we have $t \models \theta_r^{X_i,k_i}$ for some (nonnull) state *r* with $\psi^{k_i,g} \dashv \theta_r^{X_i,k_i}$. On the other hand, since $t \models \Phi'$, we have $t \models \theta_{s_i}^{X_i,k_i}$. Then by Proposition 2.3.11(ii), we have $r \rightleftharpoons_{k_i}^{X_i} t \rightleftharpoons_{k_i}^{X_i} s_i$. Therefore, by Lemma 2.4.10, $\theta_{s_i}^{X_i,k_i} \vdash \theta_r^{X_i,k_i} \vdash \psi^{k_i,g}$. As $\psi^{k_i,g} \vdash \psi$ (by Lemma 2.4.36(i)), we are forced to conclude that $\Phi' \vdash \psi$, which is a contradiction.

2.5 Conclusion

In this article, we presented natural deduction axiomatizations for BSML and the two extensions we introduced, $BSML^{\heartsuit}$ (BSML with the global/inquisitive disjunction \heartsuit) and $BSML^{\oslash}$ (BSML with the novel emptiness operator \oslash). We also proved the expressive completeness of the two extensions: $BSML^{\heartsuit}$ is expressively complete for the class of all state properties invariant under bounded bisimulation; $BSML^{\oslash}$ for the class of all union-closed state properties invariant under bounded bisimulation. We conclude by noting an additional preliminary result known to us, and listing possible directions for further investigation.

We saw that BSML is union closed but not expressively complete for union-closed properties. It appears, however, that one can find a different natural class of state properties for which BSML *is* complete: according to a very recent unpublished result (see Knudstorp [88], and Anttila and Knudstorp [16]; or Section 3.3.2 in Chapter 3), BSML is complete for the class of all properties \mathcal{P} such that \mathcal{P} is invariant under bounded bisimulation, union-closed, and also *convex*, where \mathcal{P} is convex if $[(M,s) \in \mathcal{P}$ and $(M,t) \in \mathcal{P}$ and $t \subseteq u \subseteq s$ implies $(M,u) \in \mathcal{P}$.

The semantics of the quantifiers in first-order dependence logic are analogous to the semantics of the modalities used in modal dependence logic, which are distinct from

the BSML-modalities \diamondsuit and \Box . Defining first-order quantifiers whose semantics are analogous to those of \diamondsuit and \Box may yield first-order state-based logics with interesting properties.

In inquisitive semantics, the global disjunction \mathbb{V} is used to model the meanings of questions. Similarly, in the context of the extension BSML^V, this disjunction might allow one to model questions and their interaction with free choice phenomena (3), and to arrive at a full account of examples of overt free choice cancellation (4), which were left open in [6].

- 3. Can I have coffee or tea? $\Diamond (c \lor t) \lor \neg \Diamond (c \lor t)$
 - (a) Yes \rightsquigarrow you can have coffee and you can have tea.
 - (b) No \rightsquigarrow you cannot have either.
- 4. You may have coffee or tea, I don't know which.

Finally, given that the emptiness operator \oslash is a natural counterpart to NE and that it can be used to cancel out the effects of pragmatic enrichment in the sense that $\oslash(\alpha \land NE) \equiv \alpha$, investigating the applications of this operator in formal semantics may yield some interesting results.

Chapter 3 Convex Propositional and Modal Team Logics

This chapter is based on:

Aleksi Anttila and Søren Brinck Knudstorp. Convex Propositional and Modal Team Logics. Manuscript. 2024

Abstract We prove expressive completeness results for convex propositional and modal team logics, where a logic is convex if, for each of its formulas, if the formula is true in two teams, then it is also true in all the teams between these two teams with respect to set inclusion. We introduce multiple propositional/modal logics which are expressively complete for the class of all convex propositional/modal team properties. We also answer an open question concerning the expressive power of classical propositional logic with team semantics extended with the nonemptiness atom NE-we show that this logic expressively complete for the class all of convex and union-closed propositional team properties. A modal analogue of this result additionally yields an expressive completeness theorem for Aloni's Bilateral State-based Modal Logic. There is a specific sense in which one of the novel propositional convex logics extends propositional dependence logic, and in which another of them extends propositional inquisitive logic. This sense is related to the notion of uniform definability studied in the team semantics literature. We introduce a generalization of uniform definability and define distinct notions of extension making use of this generalization in order to clarify and make precise the notion of extension pertaining to the convex logics, as well as to succintly express a number of facts we discuss relating to definability of connectives.

3.1 Introduction

In *team semantics*—originally introduced by Hodges [78, 79] to provide a compositional semantics for Hintikka and Sandu's *independence-friendly logic* [77, 75] and

later refined by Väänänen in his work on *dependence logic* [119]; also independently developed as a semantics for *inquisitive logic* chiefly by Ciardelli, Groenendijk, and Roelofsen [41, 38, 35]—formulas are interpreted with respect to sets of evaluation points called *teams*, as opposed to single evaluation points as in standard Tarskian semantics. In propositional team semantics, teams are sets of propositional valuations; in modal team semantics [120], teams are sets of possible worlds; etc. We refer to logics which are primarily intended to be interpreted using team semantics as *team logics*.

Team-semantic *closure properties* such as *downward closure* (a formula φ is downward closed just in case its truth in a team implies truth in all subteams— $[t \models \phi]$ and $s \subseteq t$] \implies $s \models \varphi$) and *union closure* (φ is union closed iff given a nonempty collections of teams T, if $t \models \varphi$ for all $t \in T$, then $\bigcup T \models \varphi$) play an important role in the study of team logics. They allow for concise and tractable characterizations and classification of these logics, and provide an effective tool for proving definability results, completeness of axiomatizations (see, e.g., [139, 140]), and other properties such as uniform interpolation [43]. In addition to their useful formal properties, the closure properties are also conceptually suggestive-for instance, on the common interpretation of team logics in which teams represent *information states*, a formula φ being downward closed can be thought of as representing the fact that the kind of information or content expressed by φ is *persistent* in the sense that if φ is established in an information state t ($t \models \phi$), then moving from t to a more informed state s by ruling out some possibilities $(s \subseteq t)$ does not invalidate the information that φ ($s \models \varphi$) (see, e.g., [123, 38, 129, 25, 6]). (Compare the sentence 'It is raining' and the sentence 'It might be raining' with the epistemic modality 'might'. The former is persistent: if I know that it is raining-if my information state truthfully establishes that it is raining—then there is no further information that could invalidate 'It is raining'. 'It might be raining', on the other hand, is not persistent: if, for all that I know, it might be raining, and I learn that it is, in fact, not raining, the information or content expressed by 'It might be raining' is invalidated by the further information.)

In this paper, we focus on the *convexity* closure property: a formula φ is convex if its truth in teams *s* and *t* implies its truth in all teams *u* with $s \subseteq u \subseteq t$ (see Figure 3.1). Intuitively, φ is convex if there are no "gaps" in the (team-based) property $\|\varphi\| =$ $\{t \mid t \models \varphi\}$ expressed by φ —the meaning of φ is continuous (convexity is sometimes referred to as *continuity*, as in [23]). Convexity can also be seen as a natural minimal generalization of downward closure: clearly, if a formula is downward closed, it is also convex. Shifting from a downward-closed setting to a convex one can potentially allow for the expression of many additional interesting and natural properties; for instance, the epistemic modality-example from above, while not expressible by any downwardclosed formula, can, as we will see, be formalized using a simple convex formula.

The notion of convexity defined above is a particularization to the setting of propositional team semantics of a more general notion of gaplessness in meanings. There have been many proposals to the effect that such an absence of gaps is a common or even essential feature of the meanings of simple lexicalized expressions—that convexity constitutes a linguistic or cognitive universal of some kind. Some prominent early



Figure 3.1: Examples of convex/non-convex subsets of a lattice. Note that in a powerset lattice, upward-directedness corresponds to union closure.

examples of such claims include Barwise and Cooper's monotonicity constraint for the meanings of simple noun phrases [19]; van Benthem's expectation that "reasonable" quantifiers exhibit convexity [23]; and Gärdenfors' proposal that concepts conform to convexity [55].

In spite of its naturalness and prominence in the literature, convexity has not received much attention in the study of team logics (with the exception of the recent [72, 36]). In this paper, we aim to further the understanding of convexity in team semantics by proving *expressive completeness theorems* for propositional and modal team logics with respect to convex classes of properties.

In the first part of the paper, we introduce propositional and modal logics which we prove to be expressively complete with respect to the class of all convex propositional/modal properties—that is, we show that all formulas of these logics are convex, and that all convex properties can be expressed by formulas of these logics.

Most team logics are conservative extensions of well-known logics with standard single-evaluation-point Tarskian semantics, in the sense that these logics include a fragment such that for each formula α of the fragment,

 $t \models \alpha \iff v \models \alpha$ for all $v \in t$;

that is, α is true (according to the team semantics of the logic in question) in a team t just in case it is true in all elements of the team (according to the standard semantics of the fragment); and the set of connectives of the fragment is *functionally complete* for the logic with the standard semantics. We are concerned in particular with propositional logics which extend classical propositional logic and modal logics which extend the smallest normal modal logic **K**—we call each of these fragments the *classical basis* of any team logic that extends it.

The standard classical basis in the team logic literature in the lineage of Väänänen's dependence logic includes the *split disjunction* \lor (also known as the *tensor disjunction*

and the *local disjunction*). We show that in a setting in which all convex properties are definable, \lor does not preserve convexity—that is, one can find convex formulas φ and ψ such that $\varphi \lor \psi$ is not convex. To obtain logics which are complete for all convex properties, we must therefore break with this lineage and either modify the split disjunction or opt for a different classical basis. The first of our propositional convex logics, *convex propositional dependence logic* $PL_{\forall}(=(\cdot), \bullet)$, extends one of the most prominent propositional team logics, *propositional dependence logic* $PL_{\lor}(=(\cdot))$ (expressively complete for all downward-closed properties with the *empty team property*) [139] with an operator \bullet corresponding to the epistemic 'might' discussed above, and replaces the \lor of $PL_{\lor}(=(\cdot))$ with a variant \lor (also employed in Hodges' original formulation of team semantics [78]) designed to force downward closure and hence convexity.

Propositional inquisitive logic $PL_{\rightarrow}(\mathbb{W})$ is another of the most of prominent propositional team logics. Like propositional dependence logic, $PL_{\rightarrow}(\mathbb{W})$ is also expressively complete for the class of all downward-closed propositional properties with the empty team property, making it another candidate for a team logic with a natural convex extension. The classical basis of $PL_{\rightarrow}(\mathbb{W})$ (unlike that of $PL_{\vee}(=(\cdot))$) does preserve convexity, and we show that $PL_{\rightarrow}(\bullet)$, the extension of this classical basis with the \bullet operator, is expressively complete for all convex propositional properties. $PL_{\rightarrow}(\mathbb{W})$ extends its classical basis with the *inquisitive disjunction* \mathbb{W} (also known as the global disjunction) which, like the split disjunction, does not preserve convexity in a convex setting; we also introduce a convex variant of $PL_{\rightarrow}(\mathbb{W})$, convex inquisitive logic $PL_{\rightarrow}(\mathbb{W}, \bullet)$, which incorporates a variant \mathbb{W} of \mathbb{W} , which, similarly to \mathbb{V} , forces convexity by forcing downward closure; and show that this variant is, like $PL_{\mathbb{W}}(=(\cdot), \bullet)$ and $PL_{\rightarrow}(\bullet)$, complete for the class of all convex properties.

We then move to the setting of modal team semantics; as with \lor and \lor , we show that the standard diamond modality \Leftrightarrow (the *global diamond*) employed in the modal team logics literature in the dependence logic lineage [120, 73, 90] fails to preserve convexity in a convex setting. To obtain modal team logics complete for the class of all (bisimulation-invariant) convex modal properties, we instead extend our convex propositional logics with the *flat modalities* \diamondsuit and \Box employed in some formulations of modal inquisitive logic [34] and, more recently, in Aloni's Bilateral State-based Modal Logic (BSML) [6].

In the second part of the paper, we focus on a subclass of convex properties: convex union-closed properties. Union-closed logics such as *inclusion logic* [51, 137] are another prominent family of team logics. We show that in the more restricted union-closed setting, the split disjunction \lor *does* preserve convexity, so we may make use of the standard dependence logic classical basis in formulating a logic expressively complete for this class of properties. Indeed, we answer a problem that was left open in [140] by showing that PL_V(NE), the extension of this basis with the *nonemptiness atom* NE—true in a team just in case the team is nonempty—is complete for all convex and union-closed propositional properties. A modal analogue of this yields an expressive completeness result for BSML; this answers another open problem [7]. As with \lor , the

global diamond \otimes preserves convexity in a union-closed setting, and we show that the extension of PL_V(NE) with the \otimes and global box \square is also complete for all convex and union-closed modal properties.

The facts we show concerning \lor , \lor , and \diamondsuit are related to the *failure of uniform substitution* exhibited by many team logics. The notion of *uniform definability* [32, 52, 133, 135, 37, 72] arises from this failure, and our facts concerning \lor , \lor , and \diamondsuit also imply facts about the uniform definability of these connectives in the logics we consider. In the final part of this paper, we define a generalization of uniform definability and use this generalization to formulate multiple senses in which one team logic may be said to extend another team logic. These notions of extension, together with the facts concerning \lor and \lor , then allow us to articulate more precisely the sense in which our convex logics extend the downward-closed logics $PL_{\lor}(=(\cdot))$ and $PL_{\rightarrow}(\lor)$.

The paper is structured as follows. In Section 3.2, we work in the convex setting. We introduce the propositional logics $PL_{\forall}(=(\cdot), \bullet)$, $PL_{\rightarrow}(\forall, \bullet)$, and $PL_{\rightarrow}(\bullet)$, and show that each of them is expressively complete with respect to the class of all convex propositional properties. We then show that modal extensions of these logics are expressively complete with respect to the class of all convex modal properties invariant under bounded bisimulation. We further show that the disjunctions \lor and \Downarrow and the modality \otimes can break convexity in a convex setting. In Section 3.3, we move on to the convex and union-closed setting. We show that the logic $PL_{\lor}(NE)$ is expressively complete with respect to the class of all convex and union-closed propositional properties, and that two distinct modal extensions of $PL_{\lor}(NE)$ are expressively complete with respect to the class of all convex and union-closed propositional properties, and that two distinct modal extensions of $PL_{\lor}(NE)$ are expressively complete with respect to the class of all convex and union-closed modal properties invariant under bounded bisimulation. In Section 3.4, we define a generalization of uniform definability and use this notion to distinguish multiple senses in which one team logic can extend another. In Section 3.5, we conclude by listing some open problems.

3.2 Convex Properties

In Section 3.2.1, we introduce three logics which we show to be expressively complete with respect to the class of all convex properties. In Section 3.2.2, we introduce modal extensions of these logics and show modal analogues of the expressive completeness result.

3.2.1 Propositional Properties

We define the syntax and semantics of the different classical bases we consider, of our three convex logics, and of propositional dependence logic and propositional inquisitive logic; recall basic facts about propositional team semantics and team-semantic closure properties; show that the tensor disjunction \lor and the global disjunction \lor fail to preserve convexity in a convex setting; and show our expressive completeness result for PL_{\lor}(NE).

| Downward-closed logic | Classical basis | Convex variant/logic | Classical basis |
|--------------------------------|--------------------|---|-----------------|
| $PL_{\vee}(=(\cdot))$ | PL_{\vee} | $PL_{v}(=(\cdot), \bullet)$ | PL_{v} |
| $PL_{\rightarrow}(\mathbb{W})$ | PL_{\rightarrow} | $PL_{\rightarrow}(\mathbb{W}, \bullet)$ | PL→ |
| | | $PL_{\rightarrow}(\blacklozenge)$ | PL→ |

Table 3.1: The logics and their classical bases

Preliminaries

Propositional dependence logic $PL_{\vee}(=(\cdot))$ is an extension of classical propositional logic with dependence atoms $=(p_1, \ldots, p_n, p)$; our convex variant $PL_{\vee}(=(\cdot), \bullet)$ extends classical propositional logic with both dependence atoms and the *epistemic might*operator \bullet . The fragment PL_{\vee} of $PL_{\vee}(=(\cdot))$ corresponding to classical propositional logic (the *classical basis* of $PL_{\vee}(=(\cdot))$) is different from that of $PL_{\vee}(=(\cdot), \bullet)$ (PL_{\vee}), with the former featuring the *split disjunction* \vee , and the latter a variant \vee of \vee . *Propositional inquisitive logic* $PL_{\rightarrow}(\vee)$ extends its classical basis PL_{\rightarrow} with the *global* or *inquisitive disjunction* \vee . Our convex variant $PL_{\rightarrow}(\vee, \bullet)$ extends this basis with the operator \bullet and a variant \vee of \vee ; and our third convex logic $PL_{\rightarrow}(\bullet)$, with only the operator \bullet . See Table 3.1.

Fix a countably infinite set of proposition letters P.

3.2.1. DEFINITION (Syntax). The formulas of *classical propositional logic* (*with* \lor /*with* \lor /*with* \rightarrow) PL_{\lor}/PL_{\lor}/PL_{\Rightarrow} are given by the BNF-grammars

$$\boldsymbol{\alpha} ::= p \left| \perp \right| \neg \boldsymbol{\alpha} \left| \boldsymbol{\alpha} \land \boldsymbol{\alpha} \right| \boldsymbol{\alpha} \lor \boldsymbol{\alpha}; \tag{PL}_{\vee}$$

$$\alpha ::= p \mid \bot \mid \neg \alpha \mid \alpha \land \alpha \mid \alpha \lor \alpha; \tag{PL}_{\forall}$$

$$\alpha ::= p \mid \bot \mid \alpha \land \alpha \mid \alpha \to \alpha; \tag{PL}_{\to})$$

where $p \in P$. The formulas of propositional dependence logic $PL_{\vee}(=(\cdot))/convex$ propositional dependence logic $PL_{\vee}(=(\cdot), \bullet)/propositional$ inquisitive logic $PL_{\rightarrow}(\vee)/convex$ propositional inquisitive logic $PL_{\rightarrow}(\vee, \bullet)/PL_{\rightarrow}(\bullet)$ are given by the BNF-grammars

$$\varphi ::= p \mid \bot \mid \neg \alpha \mid \varphi \land \varphi \mid \varphi \lor \varphi \mid = (p_1, \dots, p_n, p); \qquad (PL_{\lor}(=(\cdot)))$$

$$\varphi ::= p \mid \bot \mid \neg \beta \mid \varphi \land \varphi \mid \varphi \lor \varphi \mid = (p_1, \dots, p_n, p) \mid \blacklozenge \varphi; \qquad (PL_{\forall}(=(\cdot), \bigstar))$$

$$\varphi ::= p \left| \bot \right| \varphi \land \varphi \left| \varphi \to \varphi \right| \varphi \lor \varphi; \tag{PL}_{\rightarrow}(\forall))$$

$$\varphi ::= p \mid \bot \mid \varphi \land \varphi \mid \varphi \to \varphi \mid \varphi \lor \varphi \mid \bullet \varphi; \qquad (PL_{\rightarrow}(\lor, \bullet))$$

$$\varphi ::= p \mid \bot \mid \varphi \land \varphi \mid \varphi \to \varphi \mid \bullet \varphi; \tag{PL}_{\to}(\bullet))$$

where $p, p_1, \ldots, p_n \in \mathsf{P}$, $\alpha \in \mathsf{PL}_{\vee}$, and $\beta \in \mathsf{PL}_{\vee}$.

We use the first Greek letters α and β to range exclusively over classical formulas (formulas of $PL_{\vee}/PL_{\vee}/PL_{\vee}$). We write $P(\varphi)$ for the set of proposition letters appearing in φ , and $\varphi(X)$ if $P(\varphi) \subseteq X \subseteq P$. We write $\varphi(\psi_1/p_1, \dots, \psi_n/p_n)$ for the result of replacing all occurrences of p_i in φ by ψ_i , for $1 \le i \le n$.

3.2. Convex Properties

A (propositional) *team t with domain* $X \subseteq P$ (or a *team over* X) is a set of valuations with domain X: $t \subseteq 2^X$.

3.2.2. DEFINITION (Semantics). Given a team *t* over X, the *truth* of a formula $\varphi(X)$ in *t* (written $t \models \varphi$) is defined by the following recursive clauses:

| : ⇔ | $v(p) = 1$ for all $v \in t$. |
|----------|---|
| $: \iff$ | $t = \emptyset$. |
| $: \iff$ | $\{v\} \not\models \alpha \text{ for all } v \in t.$ |
| $: \iff$ | $t \vDash \varphi \ and \ t \vDash \Psi.$ |
| $: \iff$ | <i>there exist</i> s, u <i>such that</i> $s \models \varphi$, $u \models \psi$, <i>and</i> $t = s \cup u$. |
| $: \iff$ | <i>there exist</i> s, u <i>such that</i> $s \models \varphi$, $u \models \psi$, <i>and</i> $t \subseteq s \cup u$. |
| $: \iff$ | for all $s \subseteq t$: if $s \models \varphi$ then $s \models \psi$. |
| $: \iff$ | $\forall v, w \in t : [\forall 1 \le i \le n : v(p_i) = w(p_i)] \Longrightarrow v(p) = w(p).$ |
| $: \iff$ | <i>there exists</i> $s \subseteq t$ <i>such that</i> $s \neq \emptyset$ <i>and</i> $s \models \varphi$. |
| : ⇔ | $t \vDash \varphi \text{ or } t \vDash \psi.$ |
| $: \iff$ | there exists $s \supseteq t$ such that $s \vDash \varphi$ or $s \vDash \psi$. |
| | |

We say that a set of formulas Γ *entails* φ , written $\Gamma \vDash \varphi$, if for all teams *t*, if $t \vDash \gamma$ for all $\gamma \in \Gamma$, then $t \vDash \varphi$. We write simply $\varphi_1, \ldots, \varphi_n \vDash \varphi$ for $\{\varphi_1, \ldots, \varphi_n\} \vDash \varphi$ and $\vDash \varphi$ for $\emptyset \vDash \varphi$, where \emptyset is the empty set of formulas. If both $\varphi \vDash \psi$ and $\psi \vDash \varphi$, we say that φ and ψ are *equivalent*, and write $\varphi \equiv \psi$.

We also define (in PL_→ and its extensions) $\neg \varphi := \varphi \rightarrow \bot$; (in all logics) $\top := \neg \bot$; (in logics with \blacklozenge) $\bot := \blacklozenge \bot$; (in the classical bases) $\alpha \lor \beta := \neg (\neg \alpha \land \neg \beta)$. It can be verified that the truth conditions of the same symbol are always the same, regardless of the logic. Note that in PL_→ and its extensions, $t \models \neg \varphi \iff \forall s \subseteq t : [s \models \varphi \text{ implies } s = \varnothing]$. Note also that $t \models \top$ is always the case and $t \models \bot$ is never the case. We stipulate that $\forall \varphi := \bot, \forall \varphi := \bot, \land \varphi := \bot, \lor \varphi := \bot$.

There are many sets of connectives which are *functionally complete* for classical propositional logic. One can define team-based versions of these sets of connectives to obtain different versions of team-based classical propositional logic such as the classical bases above. Whereas, due to the functional completeness of the different alternatives, the choice of connectives often makes no substantial difference when one is studying the properties of classical propositional logic on its own, this choice may become more significant when the logic is extended with nonclassical connectives. This is the case in our setting: the choice of classical basis of a team logic alters its expressive power, as we see below.

Propositional dependence logic $PL_{\vee}(=(\cdot))$ uses the classical basis PL_{\vee} with the split disjunction \vee , where $\varphi \vee \psi$ is true in a team *t* just in case the team can be split into subteams *s* and *u* such that φ is true in *s* and ψ is true in *u*. PL_{\vee} is the standard classical propositional basis in the dependence logic literature (being used, for instance,

in propositional dependence logic [139] and propositional inclusion logic [137]), and \lor is the canonical 'classical' disjunction in this literature (in the sense of Fact 3.2.5, below). Note that one of the two subteams can be empty; for instance, we have that $\{v_{p\overline{q}}\} \models p \lor \neg p$ (where $v_{p\overline{q}}(p) = 1$ and $v_{p\overline{q}}(q) = 0$) because $\{v_{p\overline{q}}\} = \{v_{p\overline{q}}\} \cup \emptyset$ where $\{v_{p\overline{q}}\} \models p$ and $\emptyset \models \neg p$. PL $_{\lor}(=(\cdot))$ extends PL $_{\lor}$ with dependence atoms $=(p_1, \ldots, p_n, p)$. The intuitive meaning of $=(p_1, \ldots, p_n, p)$ is that the truth values of p_1, \ldots, p_n jointly determine the truth value of p. A unary dependence atom =(p) is called a *constancy atom*—=(p) is true in a team just in case the truth value of p is constant across the team ($t \models =(p)$ iff [$\forall v \in t : v(p) = 1$ or $\forall v \in t : v(p) = 0$]).

The classical basis PL_{\forall} replaces \lor in PL_{\lor} with the variant \lor (used in Hodges' original formulation of team semantics [78]), where $\varphi \lor \psi$ is true in a team *t* just in case *t* is a subteam of the union of some *s* which makes φ and some *u* which makes ψ true. This variant always preserves convexity, whereas \lor does not, as we see below. Our convex variant of dependence logic $PL_{\lor}(=(\cdot), \bullet)$ extends PL_{\lor} with the epistemic might-operator \bullet , where $\bullet \varphi$ is true in *t* just in case *t* contains a nonempty subteam in which φ is true. Its name is due to the fact that similar operators have been used to model the meanings of epistemic modalities such as the 'might' in 'it might be raining' (see, e.g., [123, 129, 25]). The idea is that a team represents an information state, and if an information state contains some possible states of affairs in which it is raining, and hence, for all that one knows given the information in the state, it might be raining.

PL→, featuring the *intuitionistic implication*—where $t \models \varphi \rightarrow \psi$ just in case whenever φ is true in a subteam of t, so is ψ — is the classical basis for propositional inquisitive logic PL→(\forall). PL→(\forall) extends PL→ with the *inquisitive* or *global* disjunction \forall which has the classical disjunction satisfaction clause (with respect to teams), but which, as we will see, behaves nonclassically. The inquisitive disjunction is used to model the meanings of question in inquisitive logic/semantics; for instance, $p \lor q$ represents the question 'p or q?'—the question is true (or *supported*) in a team just in case one of its answers is true. PL→(\forall , \bullet) replaces \forall in PL→(\forall) with \forall , which, similarly to \forall , guarantees the preservation of convexity, whereas \forall does not. PL→(\bullet), the \forall free fragment of PL→(\forall , \bullet), is also sufficiently strong to capture all convex properties (we introduce the stronger logic PL→(\forall , \bullet) because there is an interesting sense in which PL→(\forall , \bullet) extends PL→(\forall), whereas it is an open question whether PL→(\bullet) also extends PL→(\forall) in this way—see Section 3.4).

3.2.3. DEFINITION (Closure properties). We say that

| φ is downward closed | iff | $[t \vDash \varphi \text{ and } s \subseteq t] \Rightarrow s \vDash \varphi;$ |
|--|-----|--|
| $\pmb{\varphi}$ is upward closed | iff | $[t \vDash \varphi \text{ and } s \supseteq t] \Rightarrow s \vDash \varphi;$ |
| φ is <i>convex</i> | iff | $[t \vDash \varphi, s \vDash \varphi, \text{ and } s \subseteq u \subseteq t] \Rightarrow u \vDash \varphi;$ |
| ϕ is union closed | iff | $[t \vDash \varphi \text{ for all } t \in T \neq \emptyset] \Rightarrow \bigcup T \vDash \varphi;$ |
| φ has the <i>empty team property</i> | iff | $\varnothing \vDash \varphi;$ |
| φ is flat | iff | $t \vDash \varphi \Leftrightarrow [\{v\} \vDash \varphi \text{ for all } v \in t].$ |
| | | |

It is straightforward to verify that:

3.2.4. FACT.

- (i) φ is flat iff φ is downward and union closed, and φ has the empty team property.
- (ii) φ is downward closed iff φ is convex and [if $\varphi \neq \bot$, then $\emptyset \models \varphi$].
- (iii) $\varphi(X)$ is upward closed iff φ is convex and $[if \varphi \neq \bot]$, then $2^X \models \varphi]$.

Classical formulas (formulas of $PL_{\vee}/PL_{\vee}/PL_{\rightarrow}$) are flat, and we also have:

3.2.5. FACT. For classical α :

$$t \models \alpha \iff \{v\} \models \alpha \text{ for all } v \in t \iff v \models \alpha \text{ for all } v \in t.$$

Here \models on the right is the usual single-valuation truth relation for classical propositional logic. Therefore, extensions of our classical bases are also conservative extensions of classical logic—we have that for any set $\Gamma \cup \{\alpha\}$ of classical formulas,

$$\Gamma \vDash \alpha \iff \Gamma \vDash_c \alpha$$
,

where \vDash_c is the entailment relation for single-valuation semantics. Given these facts, we use the notations $\{v\} \vDash \alpha$ and $v \vDash \alpha$ interchangeably whenever α is classical, and similarly for $\Gamma \vDash \alpha$ and $\Gamma \vDash_c \alpha$.

Formulas with \bullet may clearly violate downward closure and the empty team property. Formulas of each of our nonclassical logics need not be union closed. Consider, for instance, $\bullet p \to q$. We have that $\{v_{pq}\} \models \bullet p \to q$ and $\{v_{\overline{pq}}\} \models \bullet p \to q$, but $\{v_{pq}, v_{\overline{pq}}\} \not\models \bullet p \to q$. Similarly, for $\varphi :==(q, p,)/\varphi := p \lor \neg p/\varphi := p \lor \neg p$ we have $\{v_{pq}\} \models \varphi$ and $\{v_{\overline{pq}}\} \models \varphi$, but $\{v_{pq}, v_{\overline{pq}}\} \not\models \varphi$. However, we do have:

3.2.6. PROPOSITION. Formulas of $PL_{\vee}(=(\cdot))$ and $PL_{\rightarrow}(\vee)$ are downward closed (and hence also convex). Formulas of $PL_{\vee}(=(\cdot), \blacklozenge)$, $PL_{\rightarrow}(\vee, \blacklozenge)$, and $PL_{\rightarrow}(\blacklozenge)$ are convex.

Proof:

By induction on the structure of formulas φ . Most cases are straightforward—note in particular that = (p_1, \ldots, p_n, p) , $\varphi \rightarrow \psi$, $\varphi \lor \psi$, and $\varphi \lor \psi$ are always downward closed, and hence convex, and that $\diamond \varphi$ is always upward closed, and hence convex.

As mentioned above, whereas the variant \vee always preserves convexity, the split disjunction \vee need not do so, and in fact, as implied by the fact below, no logic expressively complete for the class of all convex properties can incorporate \vee —this is why we swap \vee for \vee in our convex variant of dependence logic $PL_{\vee}(=(\cdot), \bullet)$. The situation with \mathbb{V} and \mathbb{W} is analogous.

3.2.7. FACT. There are convex φ, ψ such that $\varphi \lor \psi$ is not convex. Similarly, there are convex φ, ψ such that $\varphi \lor \psi$ is not convex.

Proof:

For the first part, let $\varphi := (((p \land NE) \lor (\neg p \land NE)) \rightarrow \bot) \land \blacklozenge r$. Clearly the first conjunct is downward closed (and hence convex) and the second is upward closed (and hence convex); therefore, since conjunction preserves convexity, φ is convex. Now observe that $\{v_{\overline{p}r}\} \models \varphi \lor \varphi; \{v_{\overline{p}r}, v_{p\overline{r}}\} \nvDash \varphi \lor \varphi$; and $\{v_{\overline{p}r}, v_{p\overline{r}}, v_{pr}\} \models \varphi \lor \varphi$, where $\{v_{\overline{p}r}\} \subseteq \{v_{\overline{p}r}, v_{p\overline{r}}\} \subseteq \{v_{\overline{p}r}, v_{p\overline{r}}\} \subseteq \{v_{\overline{p}r}, v_{p\overline{r}}\}$.

For the second part, note that p and $\bullet q$ are convex. We have $\{v_{p\overline{q}}\} \vDash p \lor \bullet q$; $\{v_{p\overline{q}}, v_{\overline{pq}}\} \nvDash p \lor \bullet q$; and $\{v_{p\overline{q}}, v_{\overline{pq}}, v_{pq}\} \vDash p \lor \bullet q$, and also that $\{v_{p\overline{q}}\} \subseteq \{v_{p\overline{q}}, v_{\overline{pq}}\} \subseteq \{v_{p\overline{q}}, v_{\overline{pq}}\}$.

The above is related to the fact that, as with many team-based logics, the nonclassical logics we consider are not *closed under uniform substitution*: $\varphi \models \psi$ need not imply $\varphi(\chi/p) \models \psi(\chi/p)$. For instance, we have $p \lor p \models p$, but $(p \lor \neg p) \lor (p \lor \neg p) \not\models (p \lor \neg p) (consider the team <math>\{v_p, v_{\overline{p}}\}$, and $p \land (q \lor r) \models (p \land q) \lor (p \land r)$, but $\bullet \top \land (q \lor r) \not\models (\bullet \top \land q) \lor (\bullet \top \land r)$ (consider the team $\{v_{q\overline{r}}\}$). We return to Fact 3.2.7 and its connection with closure under uniform substitution in Section 3.4.

Expressive Completeness

We measure the expressive power of the logics in terms of the properties—classes of teams—expressible in them.

3.2.8. DEFINITION (Properties and Expressive Completeness). A (*propositional team*) property over X is a class of (propositional) teams over X. For each formula $\varphi(X)$, we denote by $\|\varphi\|_X$ (or simply $\|\varphi\|$) the property over X expressed by φ :

$$\|\boldsymbol{\varphi}\|_{\mathsf{X}} \coloneqq \{t \in 2^{\mathsf{X}} \mid t \models \boldsymbol{\varphi}\}.$$

Given a class of properties \mathbb{P} and $X \subseteq P$, we let

 $\mathbb{P}_{\mathsf{X}} \coloneqq \{\mathcal{P} \text{ is a property over } \mathsf{X} \mid \mathcal{P} \in \mathbb{P}\}.$

We say that a logic L is *expressively complete* for a class of properties \mathbb{P} , written $\|L\| = \mathbb{P}$, if for each finite $X \subseteq P$,

$$\|L\|_{\mathsf{X}} \coloneqq \{\|\varphi\|_{\mathsf{X}} \mid \varphi \text{ is a formula of } L\} = \mathbb{P}_{\mathsf{X}}.$$

That is, L is expressively complete for \mathbb{P} if (\subseteq) each property $\|\varphi\|$ definable by a formula φ of L is in \mathbb{P} , and (\supseteq) each property in \mathbb{P} over a finite X is definable by a formula of L. We also write $\|L\| \subseteq \mathbb{P}$ to mean that for each finite $X \subseteq P$, $\|L\|_X \subseteq \mathbb{P}_X$, etc.

The definition of closure properties is extended to team properties in the obvious way. For instance, a property \mathcal{P} is downward closed if $[t \in \mathcal{P} \text{ and } s \subseteq t]$ implies $s \in \mathcal{P}$. Let $\mathbb{C}/\mathbb{C}\mathbb{U}/\mathbb{D}\mathbb{E}/\mathbb{U}/\mathbb{F}$ be the class of all convex/convex and union-closed/downward-closed and empty-team-property/upward-closed/flat properties, respectively.

72

3.2. Convex Properties

We show that each of $PL_{\forall}(=(\cdot), \bullet)$, $PL_{\rightarrow}(\forall, \bullet)$, and $PL_{\rightarrow}(\bullet)$ is complete for \mathbb{C} , i.e., $\|PL_{\forall}(=(\cdot), \bullet)\| = \|PL_{\rightarrow}(\forall, \bullet)\| = \|PL_{\rightarrow}(\bullet)\| = \mathbb{C}$. Note that have already shown, in Proposition 3.2.6, that $\|PL_{\forall}(=(\cdot), \bullet)\| \subseteq \mathbb{C}$, $\|PL_{\rightarrow}(\forall, \bullet)\| \subseteq \mathbb{C}$, and $\|PL_{\rightarrow}(\bullet)\| \subseteq \mathbb{C}$.

To show the other direction, we construct, for each property \mathcal{P} in \mathbb{C}_X , a formula in $\|PL_{\forall}(=(\cdot), \bullet)\|_X$ that expresses that \mathcal{P} : a characteristic formula for \mathcal{P} —and similarly for the other logics. We begin by recalling characteristic formulas for valuations and teams $\chi_v^X, \chi_t^X \in PL_{\vee}/PL_{\forall}/PL_{\rightarrow}$ from the literature (see, e.g., [41, 71, 139, 140]). Fix a finite $X \subseteq P$. For a valuation *v*, let

$$\chi_{v}^{\mathsf{X}} \coloneqq \bigwedge \{p \mid p \in \mathsf{X}, v(p) = 1\} \land \bigwedge \{\neg p \mid p \in \mathsf{X}, v(p) = 0\}$$

It is then easy to see that:

$$w \vDash \chi_v^{\mathsf{X}} \iff w \upharpoonright \mathsf{X} = v \upharpoonright \mathsf{X},$$

and if $w, v \in 2^{\times}$, then $w \models \chi_v^{\times} \iff w = v$. We usually write simply χ_v . For a team *t*, we let:

$$\chi_t^{\mathsf{X}} := \bigvee_{v \in t} \chi_v^{\mathsf{X}}.$$

Then:

 $s \models \chi_t^X \iff s \upharpoonright X \subseteq t \upharpoonright X,$ (where for a team $u, u \upharpoonright X := \{v \upharpoonright X \mid v \in u\}$)

and if $t, s \subseteq 2^X$, then $s \models \chi_t^X \iff s \subseteq t$. Again, we usually write simply χ_t . Note that since for a given finite X, there are only finitely many χ_v^X , we may assume the disjunction in χ_t^X to be finite and therefore for the formula to be well-defined.¹ Observe also that we have used the defined disjunction $\forall = \neg \land \neg$ (available in each of our logics) in the definition of χ_t ; it is, however, easy to check that $\forall_{v \in t} \chi_v \equiv \bigvee_{v \in t} \chi_v \equiv \forall_{v \in t} \chi_v$.

It is instructive to present our construction of the characteristic formulas for properties in a schematic manner. Note first that the empty property $\mathcal{P} = \emptyset$ is convex, and that it is expressible in each of logics using the formula(s) \bot . As for nonempty properties, we construct, in each of our logics, for such property \mathcal{P} , a formula $\chi_{\mathcal{P}}^{\mathbb{D}}$ such that

$$t \vDash \chi_{\mathcal{P}}^{\mathbb{D}} \iff \exists s \in \mathcal{P} : t \subseteq s, \tag{(*)}$$

and a formula $\chi_{\mathcal{P}}^{\mathbb{U}}$ such that

$$t \vDash \chi_{\mathcal{P}}^{\mathbb{U}} \iff \exists s \in \mathcal{P} : t \supseteq s.$$

The formulas $\chi_{\mathcal{P}}^{\mathbb{D}}$ are characteristic formulas for nonempty downward-closed properties: observe that for nonempty downward-closed \mathcal{P} , $\|\chi_{\mathcal{P}}^{\mathbb{D}}\| = \mathcal{P}$; similarly the formulas $\chi_{\mathcal{P}}^{\mathbb{U}}$ are characteristic formulas for nonempty upward-closed properties. Using these formulas, we construct characteristic formulas for nonempty convex properties as follows:

¹More precisely, what we do is choose, for each infinite *t*, some finite *s* such that $\{\chi_{v}^{X} | v \in t\} = \{\chi_{v}^{X} | v \in s\}$ to act as the representative of *t*, and define $\chi_{t}^{X} := \chi_{s}^{X}$. Similar remarks apply to the characteristic formulas for properties defined below—observe that this is what allows us to treat all properties over a finite X as if they were finite.

3.2.9. LEMMA. Let $X \subseteq P$ be finite, and for each $\mathcal{P} \neq \emptyset$ over X, let $\chi_{\mathcal{P}}^{X,\mathbb{D}}$ and $\chi_{\mathcal{P}}^{X,\mathbb{U}}$ be such that for any $t \subseteq 2^X$, $t \models \chi_{\mathcal{P}}^{X,\mathbb{D}} \iff \exists s \in \mathcal{P} : t \subseteq s$ and $t \models \chi_{\mathcal{P}}^{X,\mathbb{U}} \iff \exists s \in \mathcal{P} : t \supseteq s$. Then for any convex $\mathcal{P} \neq \emptyset$ over X, $\|\chi_{\mathcal{P}}^{X,\mathbb{D}} \wedge \chi_{\mathcal{P}}^{X,\mathbb{U}}\|_{X} = \mathcal{P}$.

Proof:

 $\supseteq: \text{ For any } t \in \mathcal{P}, t \subseteq t \subseteq t, \text{ whence } t \models \chi_{\mathcal{P}}^{\mathbb{D}} \land \chi_{\mathcal{P}}^{\mathbb{U}}. \\ \subseteq: \text{ If } t \models \chi_{\mathcal{P}}^{\mathbb{D}} \land \chi_{\mathcal{P}}^{\mathbb{U}}, \text{ then for some } u, s \in \mathcal{P}, s \subseteq t \subseteq u, \text{ whence } t \in \mathcal{P} \text{ by convexity.} \qquad \Box$

We first construct the formulas $\chi_{\mathcal{P}}^{\mathbb{U}}$. These can be constructed in the same manner in each of our logics:

3.2.10. LEMMA. Let $X \subseteq P$ be finite. For $\mathcal{P} = \{t_1, \ldots, t_n\} \neq \emptyset$ over X, let

$$\chi_{\mathcal{P}}^{\mathsf{X},\mathbb{U}} \coloneqq \bigwedge_{\nu_1 \in t_1,\ldots,\nu_n \in t_n} \bullet (\chi_{\nu_1}^{\mathsf{X}} \lor \ldots \lor \chi_{\nu_n}^{\mathsf{X}}).$$

Then for any $t \subseteq 2^{\mathsf{X}}, t \models \chi_{\mathcal{P}}^{\mathsf{X},\mathbb{U}} \iff \exists t_i \in \mathcal{P} : t \supseteq t_i.$

Proof:

 $\iff: \text{Let } t_i \in \mathcal{P} \text{ be such that } t \supseteq t_i. \text{ For each } v_i \in t_i, \text{ we have that } v_i \models \chi_{v_i}, \text{ so that also} \\ v_i \models \chi_{v_1} \lor \ldots \lor \chi_{v_n} \text{ for any } v_1 \in t_1 \ldots v_{i-1} \in t_{i-1}, v_{i+1} \in t_{i+1}, \ldots v_n \in t_n. \text{ Therefore, since } v_i \in \\ t_i \subseteq t, t \models \bullet (\chi_{v_1} \lor \ldots \lor \chi_{v_n}). \text{ Repeating, this argument, we have } t \models \bigwedge_{v_1 \in t_1, \ldots, v_n \in t_n} \bullet (\chi_{v_1} \lor \ldots \lor \chi_{v_n}).$

 $\implies: \text{Let } t \models \bigwedge_{v_1 \in t_1, \dots, v_n \in t_n} \bullet (\chi_{v_1} \lor \dots \lor \chi_{v_n}). \text{ If } \mathcal{P} \text{ has the empty team property, we have } \emptyset \subseteq t \text{ where } \emptyset \in \mathcal{P}; \text{ we may therefore assume } \mathcal{P} \text{ does not have the empty team property, whence } t_i \neq \emptyset \text{ for all } 1 \leq i \leq n, \text{ whence } \bigwedge_{v_1 \in t_1, \dots, v_n \in t_n} \bullet (\chi_{v_1} \lor \dots \lor \chi_{v_n}) \text{ is not simply } \top (\text{recall that } \bigwedge \emptyset = \top). \text{ Assume for contradiction that } t_i \notin t \text{ for all } t_i \in \mathcal{P}. \text{ Then for each } t_i \in \mathcal{P} \text{ there is some } w_i \in t_i \text{ such that } w_i \notin t. \text{ By } t \models \bigwedge_{v_1 \in t_1, \dots, v_n \in t_n} \bullet (\chi_{v_1} \lor \dots \lor \chi_{v_n}), \text{ there is, for each } v_1 \in t_1, \dots, v_n \in t_n, \text{ a nonempty } t_{v_1, \dots, v_n} \subseteq t \text{ such that } t_{v_1, \dots, v_n} \models \chi_{v_1} \lor \dots \lor \chi_{v_n}. \text{ Therefore, in particular, there is a nonempty } t_{w_1, \dots, w_n} \subseteq t \text{ such that } t_{w_1, \dots, w_n} \models \chi_{w_1} \lor \dots \lor \chi_{w_n}. \text{ Then } t_{w_1, \dots, w_n} \subseteq \bigcup_{1 \leq i \leq n} \{w_i\} \text{ and } t_{w_1, \dots, w_n} \neq \emptyset. \text{ Given } t_{w_1, \dots, w_n} \subseteq t \text{ we have } t \cap \bigcup_{1 \leq i \leq n} \{w_i\} \neq \emptyset, \text{ contradicting the fact that } w_i \notin t \text{ for all } t_i \in \mathcal{P}. \square$

We now turn to the formulas $\chi_{\mathcal{P}}^{\mathbb{D}}$. We will construct these in a distinct manner in each of the three logics $PL_{\forall}(=(\cdot), \bullet)$, $PL_{\rightarrow}(\forall, \bullet)$ and $PL_{\rightarrow}(\bullet)$. We begin constructing these formulas by recalling the following expressive completeness results for propositional dependence logic and propositional inquisitive logic:

3.2.11. THEOREM ([41, 139]). Each of $PL_{\vee}(=(\cdot))$ and $PL_{\rightarrow}(\vee)$ is expressively complete for the class of all downward-closed properties with the empty team property:

$$\|\mathsf{PL}_{\vee}(=(\cdot))\| = \|\mathsf{PL}_{\rightarrow}(\vee)\| = \mathbb{DE}.$$

3.2. Convex Properties

We make use of the characteristic formulas used to prove the above and the similarity between these logics and our convex logics to construct the formulas $\chi_{\mathcal{P}}^{\mathbb{D}}$. Note that although the formulas $\chi_{\mathcal{P}}^{\mathbb{D}}$ are characteristic formulas for nonempty downward-closed properties (equivalently, for properties in \mathbb{DE}), it need not be the case that any given characteristic formula $\chi_{\mathcal{P}}$ for a property in \mathbb{DE} satisfies property (*). However, it turns out that by manipulating the formulas of $PL_{\vee}(=(\cdot))/PL_{\rightarrow}(\vee)$ used to prove Theorem 3.2.11, we are in each case able find similar formulas $\chi_{\mathcal{P}}^{\mathbb{D}}$ with the desired property (*) in our convex logics.

Let us first consider the characteristic formulas in $PL_{\rightarrow}(\mathbb{W})$, which we will use to construct $\chi_{\mathcal{P}}^{\mathbb{D}}$ in $PL_{\rightarrow}(\mathbb{W}, \bullet)$ and $PL_{\rightarrow}(\bullet)$. These formulas are of the form $\bigvee_{s \in \mathcal{P}} \chi_s^X$. It is easy to see that these formulas already satisfy property (*); it therefore suffices to show that we can find equivalent formulas in $PL_{\rightarrow}(\mathbb{W}, \bullet)$ and $PL_{\rightarrow}(\bullet)$. In the former case this is trivial given that for any downward-closed (and hence for any classical) φ and ψ , we have $\varphi \otimes \psi \equiv \varphi \otimes \psi$.

3.2.12. LEMMA. Let $X \subseteq P$ be finite. For $\mathcal{P} \neq \emptyset$ over X, let

$$\chi_{\mathsf{P}}^{\mathcal{X},\mathbb{D}} \coloneqq \bigvee_{s \in \mathcal{P}} \chi_s^{\mathsf{X}}.$$

Then for any $t \subseteq 2^{X}$, $t \models \chi_{\mathcal{P}}^{X,\mathbb{D}} \iff \exists s \in \mathcal{P} : t \subseteq s$.

Proof:

Immediate from the definitions and the fact that $s \vDash \chi_s$.

As for $PL_{\rightarrow}(\bullet)$, we show below that there is a formula equivalent to $\bigvee_{s \in \mathcal{P}} \chi_s$ in $PL_{\rightarrow}(\bullet)$ by showing that the global disjunction of any collection of flat formulas is definable in $PL_{\rightarrow}(\bullet)$. Note that since $PL_{\rightarrow}(\forall, \bullet)$ is a syntactic extension of $PL_{\rightarrow}(\bullet)$, the below also yields a formula in $PL_{\rightarrow}(\forall, \bullet)$ which is equivalent to $\bigvee_{s \in \mathcal{P}} \chi_s$, whence the disjunction \forall and the formula $\bigvee_{s \in \mathcal{P}} \chi_s$ defined above are not required to prove the expressive completeness of $PL_{\rightarrow}(\forall, \bullet)$. We discuss our reasons for defining this extension with a disjunction which does not in this case yield an increase in expressive power in Section 3.4.

3.2.13. PROPOSITION. For any nonempty collection $\{\varphi_i\}_{i \in I}$ of flat formulas,

$$\bigwedge_{i\in I} \left(\left(\bigwedge_{j\in I\setminus\{i\}} \bullet \neg \varphi_j \right) \to \varphi_i \right) \equiv \bigvee_{i\in I} \varphi_i \equiv \bigvee_{i\in I} \varphi_i.$$

Proof:

We show the first equivalence.

 $\models: \text{Let } t \models \bigwedge_{i \in I} ((\bigwedge_{j \in I \setminus \{i\}} \bullet \neg \varphi_j) \to \varphi_i) \text{ and assume for contradiction that } t \not\models \varphi_i \text{ for each } i \in I. \text{ By flatness, for each } i \in I \text{ there is some } v_i \in t \text{ such that } \{v_i\} \not\models \varphi_i \text{ whence also } \{v_i\} \models \neg \varphi_i. \text{ Then for each } i \in I, t \models \bullet \neg \varphi_i. \text{ By } t \models (\bigwedge_{j \in I \setminus \{i\}} \bullet \neg \varphi_i) \to \varphi_i, \text{ we have } t \models \varphi_i \text{ for all } i \in I, a \text{ contradiction. So for some } i \in I, t \models \varphi_i, \text{ whence } t \models \bigvee_{i \in I} \varphi_i.$

 $\exists: \text{Let } t \models \bigvee_{i \in I} \varphi_i. \text{ We can then fix some } i \in I \text{ such that } t \models \varphi_i. \text{ Let } s ⊆ t \text{ be such that } s \models \land_{j \in I \setminus \{i\}} \bullet \neg \varphi_j. \text{ By downward closure, also } s \models \varphi_i, \text{ so } t \models (\land_{j \in I \setminus \{i\}} \bullet \neg \varphi_j) \rightarrow \varphi_i. \text{ Now let } k \in I \text{ be such that } k \neq i. \text{ Let } s ⊆ t, \text{ and assume for contradiction that } s \models \land_{j \in I \setminus \{k\}} \bullet \neg \varphi_j. \text{ Then } s \models \bullet \neg \varphi_i, \text{ so there is some } u ⊆ s \text{ such that } u \neq \emptyset \text{ and } u \models \neg \varphi_i. \text{ But we have } u ⊆ s ⊆ t, \text{ whence by downward closure, } u \models \varphi_i. \text{ Then by } u \models \neg \varphi_i \text{ we have } u = \emptyset, \text{ a contradiction. Therefore, vacuously } t \models (\land_{j \in I \setminus \{k\}} \bullet \neg \varphi_j) \rightarrow \varphi_k.$

It remains only to show that we can define formulas $\chi_{\mathcal{P}}^{\mathbb{D}}$ with property (*) in PL_v(= (·), •). We make use of the characteristic formulas of PL_v(=(·)) used to prove Theorem 3.2.11. These are defined by first letting, for each finite $X \subseteq P$, $\gamma_0^X := \bot$, $\gamma_1^X := \bigwedge_{p \in X} =(p)$, and for $n \ge 2$, $\gamma_n^X := \bigvee_n \gamma_1$. Then it is easy to see that for $t \subseteq 2^X$, we have $t \models \gamma_n^X \iff |t| \le n$, where |t| is the size of t. One then defines, for each nonempty $t \subseteq 2^X$, $\xi_t^X := \gamma_{|t|-1}^X \lor \chi_{|T|_X \lor t}^X$. It can then be shown that for $t \subseteq 2^X$, $t \models \xi_s^X \iff s \notin t$. Finally, the characteristic formula for a $\mathcal{P} \in \mathbb{D}\mathbb{E}$ over X is given by $\bigwedge_{s \in ||T||_X \lor \mathcal{P}} \xi_s^X$. Now, it can be verified that if \mathcal{P} is not downward closed, its characteristic formula as defined here need not have property (*). Given the properties of the formulas ξ_s , we do, however, have:

3.2.14. LEMMA ([139]). Let $X \subseteq P$ be finite. For \mathcal{P} over X such that $\emptyset \notin \mathcal{P}$, and $t \subseteq 2^X$:

$$t \vDash \bigwedge_{s \in \mathcal{P}} \xi_s^{\mathsf{X}} \iff s \notin t \text{ for all } s \in \mathcal{P}.$$

We use this to construct our formulas $\chi_{\mathcal{P}}^{\mathbb{D}}$ with property (*):

3.2.15. LEMMA. Let $X \subseteq P$ be finite. For $\mathcal{P} \neq \emptyset$ over X, let

$$\chi_{\mathcal{P}}^{\mathsf{X},\mathbb{D}} \coloneqq \bigwedge_{u \in \mathcal{Q}} \xi_u^{\prime\mathsf{X}}$$

where $\xi_u^{\prime X} \in PL_{\forall}(=(\cdot), \bullet)$ is defined by replacing each \lor in ξ_u^X with \forall , and $Q := \{u \subseteq 2^X \mid u \notin s \text{ for all } s \in \mathcal{P}\}$. Then for any $t \subseteq 2^X$, $t \models \chi_{\mathcal{P}}^{X,\mathbb{D}} \iff \exists s \in \mathcal{P} : t \subseteq s$.

Proof:

Note that for all downward-closed formulas φ and ψ , $\varphi \lor \psi \equiv \varphi \lor \psi$, and that, since $\mathcal{P} \neq \emptyset$, we have $\emptyset \notin \mathcal{Q}$. Therefore, by Lemma 3.2.14, we have that $t \models \chi_{\mathcal{P}}^{\mathbb{D}} \iff u \notin t$ for all $u \in \mathcal{Q}$. We show that $u \notin t$ for all $u \in \mathcal{Q} \iff \exists s \in \mathcal{P} : t \subseteq s$.

 \implies : Assume for contradiction that there is no $s \in \mathcal{P}$ such that $t \subseteq s$. Then $t \in \mathcal{Q}$, contradicting $u \notin t$ for all $u \in \mathcal{Q}$.

 $\iff: \text{Let } s \in \mathcal{P} \text{ be such that } t \subseteq s. \text{ Assume for contradiction that there is some } u \in \mathcal{Q}$ such that $u \subseteq t$. Then $u \subseteq t \subseteq s$, contradicting the definition of \mathcal{Q} .

And we are done:

3.2.16. THEOREM. Each of $PL_{\forall}(=(\cdot), \blacklozenge)$, $PL_{\rightarrow}(\forall, \blacklozenge)$, and $PL_{\rightarrow}(\blacklozenge)$ is expressively complete for the class of all convex properties:

$$\|\mathsf{PL}_{\forall}(=(\cdot), \bullet)\| = \|\mathsf{PL}_{\rightarrow}(\forall, \bullet)\| = \|\mathsf{PL}_{\rightarrow}(\bullet)\| = \mathbb{C}.$$

Proof:

The direction $||L|| \subseteq \mathbb{C}$ is by Proposition 3.2.6 for each relevant L. For the direction $\mathbb{C} \subseteq ||L||$, let $\mathcal{P} \in \mathbb{C}_X$. If $\mathcal{P} = \emptyset$, then $\mathcal{P} = ||\mu||_X \in ||L||_X$. If $\mathcal{P} \neq \emptyset$, the result follows by Lemmas/Propositions 3.2.9, 3.2.10, 3.2.12, 3.2.13, and 3.2.15.

3.2.2 Modal Properties

We define two distinct modal extensions of each of our classical bases. In each case, one extension uses what we will call the *flat modalities* \diamond and \Box , also used in early versions of modal inquisitive logic [34] and in Aloni's Bilateral State-based Modal Logic [6]; the other the *global modalities* \Leftrightarrow and \Box , used in modal dependence logic and other modal logics of dependence [120, 73, 90]. We call each of these new classical bases ML_{*i*,*j*}, where $i \in \{\lor, \lor, \rightarrow\}$ and $j \in \{\diamondsuit, \diamondsuit\}$, *classical modal logic*; they are defined in the obvious way. The modal extensions ML_{$\forall,j}(=(\cdot), \bullet$), ML_{\rightarrow,j}(\forall, \bullet) and ML_{\rightarrow,j}(\bullet), where $j \in \{\diamondsuit, \diamondsuit\}$ are likewise defined in the obvious way, except we define dependence atoms in ML_{$\forall,\diamondsuit}(=(\cdot), \bullet$) as follows:</sub></sub>

$$=(\alpha_1,\ldots,\alpha_n,\alpha),$$

where $\alpha, \alpha_1, ..., \alpha_n, \alpha \in ML_{\forall, \Diamond}$ (and similarly for $ML_{\forall, \Diamond}(=(\cdot), \bullet)$). That is, we now allow all classical formulas to appear in dependence atoms. (This leads to an increase in expressive power; for instance, it can be shown that the analogue of Theorem 3.2.11 holds for $ML_{\forall, \Diamond}(=(\cdot))$ with these extended dependence atoms, but not for the variant which only allows propositional variables in dependence atoms [47, 71].)

In this section, we show modal analogues of our expressive completeness theorems for the extensions of our convex logics with the flat modalities, and we show that no such analogues can be obtained for the extensions with the global modalities.

Modal team logics are interpreted on teams in standard Kripke models.

3.2.17. DEFINITION. A (*Kripke*) model (over $X \subseteq P$) is a triple M = (W, R, V), where

- W is a nonempty set, whose elements are called (*possible*) worlds;
- $R \subseteq W \times W$ is a binary relation, called the *accessibility relation*;
- $-V: X \rightarrow \wp(W)$ is a function, called the *valuation*.

We call a subset $t \subseteq W$ of W a *team* on M.

For any world *w* in *M*, define, as usual, $R[w] := \{v \in W \mid wRv\}$. Similarly, for any team *t* on *M*, define $R[t] := \bigcup_{w \in t} R[w]$ and $R^{-1}[t] := \{v \in W \mid \exists w \in t : vRw\}$. We write *tRs* and say that *s* is a *successor team* of *t* if $s \subseteq R[t]$ and $t \subseteq R^{-1}[s]$.

The modal semantics for most connectives are the obvious analogues of their propositional semantics (or instance, a team t on M makes p true—written $M, t \models p$ —just in case $t \subseteq V(p)$). We only explicitly give the semantics for the dependence atoms and the modalities.

| $M,t \models = (\alpha_1,\ldots,\alpha_n,\alpha)$ | $: \iff$ | $\forall v, w \in t : [\forall 1 \le i \le n : \{v\} \vDash \alpha_i \iff \{w\} \vDash \alpha_i] \Longrightarrow$ |
|---|----------|---|
| | | $[\{v\} \vDash \alpha \iff \{w\} \vDash \alpha].$ |
| $M,t \vDash \diamondsuit \varphi$ | $: \iff$ | $\forall w \in t : \exists s \subseteq R[w] : s \neq \emptyset \text{ and } M, s \models \varphi.$ |
| $M,t \vDash \Box \varphi$ | $: \iff$ | $\forall w \in t : M, R[s] \vDash \varphi.$ |
| $M,t \vDash \otimes \varphi$ | $: \iff$ | $\exists s \subseteq W : tRs \text{ and } M, s \vDash \varphi.$ |
| $M,t \vDash \Box \phi$ | : ⇔ | $M, R[t] \vDash \varphi.$ |

The semantics of the local modalities \diamond and \Box are defined by stating that a condition that applies to worlds must hold for all worlds in a team; this clearly makes all formulas $\diamond \varphi$ and $\Box \varphi$ flat. The global modalities \Leftrightarrow and \Box , on the other hand, make use of conditions which apply globally to teams.

We define the modal analogues of the closure properties (Definition 3.2.3) in the obvious way. It is then easy to see that the modal analogues of Facts 3.2.4 and Fact 3.2.5 (for each of our new classical bases) hold. Given the flatness of $\Diamond \varphi$ and $\Box \varphi$, an easy extension of Proposition 3.2.6 yields:

3.2.18. PROPOSITION. Formulas of $ML_{\forall,\Diamond}(=(\cdot), \bullet)$, $ML_{\rightarrow,\Diamond}(\forall, \bullet)$ and $ML_{\rightarrow,\Diamond}(\bullet)$ are convex.

The extensions with the global modalities are, however, not convex, as the fact below shows. The fact further shows that, as with \lor and \lor , the global diamond \otimes does not preserve convexity in a convex setting and so no logic with \otimes can be complete for the class of all convex modal properties.

3.2.19. FACT. There are (i) $\psi \in ML_{\rightarrow,\otimes}(\blacklozenge)$ and (ii) $\chi \in ML_{\forall,\otimes}(=(\cdot), \blacklozenge)$ that are not convex. There is (iii) a convex φ such that $\otimes \varphi$ is not convex.

Proof:

Consider the formula $\varphi := ((\bullet p \land \bullet \neg p) \to \bot) \land \bullet r$ and observe that $\varphi \equiv =(p) \land \bullet r$. It is easy to see that φ is convex; to show (i–iii) it therefore suffices to show that $\otimes \varphi$ is not convex. Consider the following model M = (W, R, V) (with *R* represented using arrows):



We have $M, \{w_{\overline{p}r}\} \models \varphi$ and $\{w_{\overline{p}r}\}R\{w_{\overline{p}r}\}$, whence $M, \{w_{\overline{p}r}\} \models \otimes \varphi$. We also have $M, \{w_{pr}, w_{p\overline{r}}\} \models \varphi$ and $\{w_{pr}, w_{p\overline{r}}, w_{\overline{p}r}\}R\{w_{pr}, w_{p\overline{r}}\}$, whence $M, \{w_{pr}, w_{p\overline{r}}, w_{\overline{p}r}\} \models \otimes \varphi$. But we have $M, \{w_{p\overline{r}}, w_{\overline{p}r}\} \not\models \otimes \varphi$. For the only $t \subseteq W$ such that $\{w_{p\overline{r}}, w_{\overline{p}r}\}Rt$ are

 $\{w_{p\bar{r}}, w_{\bar{p}r}\}\$ and $\{w_{p\bar{r}}\}\$, and for neither of these do we have $t \models \varphi$.

We move on to the modal analogues of Theorem 3.2.16 for the logics with the flat modalities $ML_{\forall,\diamondsuit}(=(\cdot), \bullet), ML_{\rightarrow,\diamondsuit}(\forall, \bullet)$ and $ML_{\rightarrow,\diamondsuit}(\bullet)$. We omit the details: given Proposition 3.2.18, the proofs of these results are almost completely analogous to that of Theorem 3.2.16 (with one departure, which we comment on below). One can define natural analogues of team properties, expressive completeness, and the formulas χ^X_{ν} and χ^X_t in the modal setting (see [71, 89] for details). One can also define a notion of bisimulation appropriate for modal team semantics (see [71, 89]), and it is easy to show that each of our logics is invariant under this notion of bisimulation (cf. the bisimulation invariance results for similar logics in [73, 7, 14]). The one departure which must be made from the strategy followed in Section 3.2.15. These require a more complicated definition; see [71] for details. The modal analogue of proof of Theorem 3.2.16 yields:

3.2.20. THEOREM. Each of $ML_{\forall,\diamondsuit}(=(\cdot), \blacklozenge)$, $ML_{\rightarrow,\diamondsuit}(\forall, \blacklozenge)$, and $ML_{\rightarrow,\diamondsuit}(\blacklozenge)$ is expressively complete for the class of all convex modal properties invariant under bounded bisimulation.

3.3 Convex and Union-closed Properties

In Section 3.3.1, we show that the logic $PL_{\vee}(NE)$ is expressively complete with respect to the class of all convex and union-closed propositional properties. In Section 3.3.2, we show modal analogues of this result for two distinct modal extensions of $PL_{\vee}(NE)$.

3.3.1 Propositional Properties

 $PL_{\vee}(NE)$ is an extension of the classical basis PL_{\vee} with the *nonemptiness atom* NE, true in a team just in case the team is nonempty:

$$t \vDash \text{NE} \iff t \neq \emptyset.$$

We define \top , $\alpha \lor \beta$, $\forall \emptyset$ and $\land \emptyset$ as before, and now define $\bot := \bot \land NE$ and $\bullet \varphi := (\varphi \land NE) \lor \top$. It is easy to see that the truth conditions of the same symbol are still always the same, regardless of the logic.

 $PL_{\vee}(NE)$ clearly violates downward closure and the empty team property, but we do have convexity and union closure—interestingly, in contrast with Fact 3.2.7, the split disjunction \vee does preserve convexity in a convex and union-closed setting.

3.3.1. PROPOSITION. Formulas of $PL_{\vee}(NE)$ are convex and union closed.

Proof:

By induction on the structure of formulas φ . Most cases are straightforward—note in particular that NE is upward closed and therefore also convex.

We only explicitly show that $\varphi \lor \psi$ is convex whenever φ and ψ are convex and union closed. Let $s \models \varphi \lor \psi$, $t \models \varphi \lor \psi$, and $s \subseteq u \subseteq t$. Then $s = s_{\varphi} \cup s_{\psi}$ and $t = t_{\varphi} \cup t_{\psi}$ where $t_{\varphi} \models \varphi$, etc. Let $u_{\varphi} \coloneqq (s_{\varphi} \cup t_{\varphi}) \cap u$ and $u_{\psi} \coloneqq (s_{\psi} \cup t_{\psi}) \cap u$. We have $s_{\varphi} \cup t_{\varphi} \models \varphi$ and $s_{\psi} \cup t_{\psi} \models \psi$ by union closure, whence $u_{\varphi} \models \varphi$ by convexity since $s_{\varphi} \subseteq u_{\varphi} \subseteq s_{\varphi} \cup t_{\varphi}$. Similarly, $u_{\psi} \models \psi$. Clearly $u = u_{\varphi} \cup u_{\psi}$, whence $u \models \varphi \lor \psi$.

The above shows $\|PL_{\vee}(NE)\| \subseteq \mathbb{CU}$. We now further show—solving a problem that was left open in [140]—that $\|PL_{\vee}(NE)\| = \mathbb{CU}$ —that is, that $PL_{\vee}(NE)$ is expressively complete for \mathbb{CU} .

We prove this in two distinct ways—it is instructive to see both of these proofs as they break down the characteristics of union-closed convex properties in different ways and feature distinct (if similar) characteristic formulas. For the first proof we use the formula $\parallel = \perp \land NE$ for the empty property, and construct, for each nonempty property \mathcal{P} , a formula $\chi_{\mathcal{P}}^{\mathbb{U}}$ such that

$$t \vDash \chi_{\mathcal{P}}^{\mathbb{U}} \iff \exists s \in \mathcal{P} : t \supseteq s,$$

and a formula $\chi_{\mathcal{P}}^{\mathbb{F}}$ such that

$$t \vDash \chi_{\mathcal{P}}^{\mathbb{F}} \iff \exists s \in \mathcal{P} : t \subseteq \bigcup \mathcal{P}$$

As before, the formulas $\chi^{\mathbb{D}}_{\mathcal{P}}$ are characteristic formulas for nonempty downward-closed properties; the formulas $\chi_{\mathcal{P}}^{\mathbb{F}}$ are characteristic formulas for flat properties: for flat \mathcal{P} , $\|\chi_{\mathcal{D}}^{\mathbb{F}}\| = \mathcal{P}$. Using these formulas, we construct characteristic formulas for nonempty union-closed convex properties as follows:

3.3.2. LEMMA. Let $X \subseteq P$ be finite, and for each $\mathcal{P} \neq \emptyset$ over X, let $\chi_{\mathcal{P}}^{X,\mathbb{F}}$ and $\chi_{\mathcal{P}}^{X,\mathbb{U}}$ be such that for any $t \subseteq 2^X$, $t \models \chi_{\mathcal{P}}^{X,\mathbb{F}} \iff t \subseteq \bigcup \mathcal{P}$ and $t \models \chi_{\mathcal{P}}^{X,\mathbb{U}} \iff \exists s \in \mathcal{P} : t \supseteq s$. Then for any union-closed convex $\mathcal{P} \neq \emptyset$ over X, $\|\chi_{\mathcal{P}}^{X,\mathbb{U}} \wedge \chi_{\mathcal{P}}^{X,\mathbb{U}}\|_{X} = \mathcal{P}$.

Proof:

 $\begin{array}{l} \supseteq: \text{ For any } t \in \mathcal{P}, t \subseteq t \subseteq \bigcup \mathcal{P}, \text{ whence } t \models \chi_{\mathcal{P}}^{\mathbb{F}} \land \chi_{\mathcal{P}}^{\mathbb{U}}. \\ \subseteq: \text{ If } t \models \chi_{\mathcal{P}}^{\mathbb{F}} \land \chi_{\mathcal{P}}^{\mathbb{U}}, \text{ then for some } s \in \mathcal{P}, s \subseteq t \subseteq \bigcup \mathcal{P}. \text{ We have } \bigcup \mathcal{P} \in \mathcal{P} \text{ by union closure} \end{array}$ and the fact that $\mathcal{P} \neq \emptyset$, whence $t \in \mathcal{P}$ by convexity.

Clearly given $\bullet \varphi \equiv (\varphi \land \text{NE}) \lor \top$, we can construct the formulas $\chi_{\mathcal{P}}^{\mathbb{U}}$ analogously to how we did in Lemma 3.2.10. As for the formulas $\chi_{\mathcal{P}}^{\mathbb{F}}$, we use the following (which can used to prove that PL_{\vee} is expressively complete for the class of all flat properties [140]):

3.3. Convex and Union-closed Properties

3.3.3. LEMMA. Let $X \subseteq P$ be finite. For \mathcal{P} over X, let

$$\chi_{\mathcal{P}}^{\mathsf{X},\mathbb{F}} \coloneqq \bigvee_{s\in\mathcal{P}} \chi_s^{\mathsf{X}}.$$

Then for any $t \subseteq 2^{X}$, $t \models \chi_{\mathcal{P}}^{X,\mathbb{F}} \iff t \subseteq \bigcup \mathcal{P}$.

Proof:

Letting $\mathcal{P} = \{s_1, \ldots, s_n\}$, we have that

$$t \models \bigvee_{s \in \mathcal{P}} \chi_s \iff [\exists t_1, \dots, t_n : t = \bigcup_{i=1}^n t_i \text{ and } t_i \models \chi_{s_i} \text{ for each } 1 \le i \le n]$$
$$\iff [\exists t_1, \dots, t_n : t = \bigcup_{i=1}^n t_i \text{ and } t_i \subseteq s_i \text{ for each } 1 \le i \le n] \iff t \subseteq \bigcup \mathcal{P}. \quad \Box$$

We have finished proving the required lemmas for the first proof. In the second proof, we make a case distinction as to whether or not \mathcal{P} has the empty team property.

3.3.4. LEMMA. For each finite X and each $\mathcal{P} = \{t_1, \ldots, t_n\} \in \mathbb{CU}_X$:

- (a) If \mathcal{P} is the empty property, $\mathcal{P} = \| \mathbb{I} \|_{X}$.
- (b) If \mathcal{P} has the empty team property, $\mathcal{P} = \| \bigvee_{s \in \mathcal{P}} \chi_s \|_{X}$.
- (c) If $\mathcal{P} \neq \emptyset$ and \mathcal{P} does not have the empty team property,

$$\mathcal{P} = \left\| \bigvee_{v_1 \in t_1, \dots, v_n \in t_n} ((\chi_{v_1} \vee \dots \vee \chi_{v_n}) \wedge \mathrm{NE}) \right\|_{\mathsf{X}}$$

Proof:

Item (a) is obvious. For item (b), note that by Fact 3.2.4, \mathcal{P} is downward closed and hence also flat. By Lemma 3.3.3, $t \models \bigvee_{s \in \mathcal{P}} \chi$ iff $t \subseteq \bigcup \mathcal{P}$, and, if \mathcal{P} is flat, clearly $t \subseteq \bigcup \mathcal{P}$ iff $t \in \mathcal{P}$.

We now show item (c). For the direction \subseteq , let $t_i \in \mathcal{P}$. For each $v_i \in t_i$, we have that $\{v_i\} \models \chi_{v_i} \land \text{NE}$, so that also (using the empty team property of classical formulas), $\{v_i\} \models (\chi_{v_1} \lor \ldots \lor \chi_{v_n}) \land \text{NE}$ for any $v_1 \in t_1 \ldots v_{i-1} \in t_{i-1}, v_{i+1} \in t_{i+1}, \ldots v_n \in t_n$. Therefore,

$$\{v_i\} \vDash \bigvee_{v_1 \in t_1} \dots \bigvee_{v_{i-1} \in t_{i-1}} \bigvee_{v_{i+1} \in t_{i+1}} \dots \bigvee_{v_n \in t_n} ((\chi_{v_1} \lor \dots \lor \chi_{v_n}) \land NE),$$

whence $t_i = \bigcup_{v_i \in t_i} \{v_i\} \models \bigvee_{v_1 \in t_1, \dots, v_n \in t_n} ((\chi_{v_1} \lor \dots \lor \chi_{v_n}) \land NE).$

For the direction \supseteq , let $s \models \bigvee_{v_1 \in t_1, \dots, v_n \in t_n} ((\chi_{v_1} \lor \dots \lor \chi_{v_n}) \land \text{NE})$. By the fact that \mathcal{P} does not have the empty team property, no $t_i \in \mathcal{P}$ is the empty team. We then have that $s = \bigcup_{v_1 \in t_1, \dots, v_n \in t_n} t_{v_1, \dots, v_n}$ where $t_{v_1, \dots, v_n} \models (\chi_{v_1} \lor \dots \lor \chi_{v_n}) \land \text{NE}$. (Note that if \mathcal{P} did have the empty team property, we would have $\bigvee_{v_1 \in t_1, \dots, v_n \in t_n} ((\chi_{v_1} \lor \dots \lor \chi_{v_n}) \land \text{NE}) \equiv \bigvee \emptyset \equiv \bot$.)

We show that $s \subseteq \bigcup \mathcal{P}$ and that $t_i \subseteq s$ for some $t_i \in \mathcal{P}$. Since $\bigcup \mathcal{P} \in \mathcal{P}$ by the union closure of \mathcal{P} (as well as the fact that $\mathcal{P} \neq \emptyset$), we will then have $t_i \in \mathcal{P}$ by the convexity of \mathcal{P} .

 $s \subseteq \bigcup \mathcal{P}$: We have that $s = \bigcup_{v_1 \in t_1, \dots, v_n \in t_n} t_{v_1, \dots, v_n}$, and for each t_{v_1, \dots, v_n} , by $t_{v_1, \dots, v_n} \models (\chi_{v_1} \lor \dots \lor \chi_{v_n}) \land NE$ we have that $t_{v_1, \dots, v_n} \subseteq \bigcup_{1 \le i \le n} \{v_i\} \subseteq \bigcup \mathcal{P}$.

 $t_i \subseteq s$ for some $t_i \in \mathcal{P}$: Assume for contradiction that $t_i \notin s$ for all $t_i \in \mathcal{P}$. Then for each $t_i \in \mathcal{P}$ there is some $w_i \in t_i$ such that $w_i \notin s$. We have that $t_{w_1,...,w_n} \models (\chi_{w_1} \lor ... \lor \chi_{w_n}) \land \text{NE}$ so $t_{w_1,...,w_n} \subseteq \bigcup_{1 \le i \le n} \{w_i\}$ and $t_{w_1,...,w_n} \neq \emptyset$. We also have $t_{w_1,...,w_n} \subseteq s$, so $s \cap \bigcup_{1 \le i \le n} \{w_i\} \neq \emptyset$, contradicting the fact that $w_i \notin s$ for all $t_i \in \mathcal{P}$.

Putting together Proposition 3.3.1, the Lemmas 3.3.2, 3.2.10, and 3.3.3 (for the first proof), and Lemma 3.3.4 (for the second proof), we have shown:

3.3.5. THEOREM. $PL_{\vee}(NE)$ is expressively complete for convex union-closed properties:

$$\|\mathbf{PL}_{\vee}(\mathbf{NE})\| = \mathbb{CU}.$$

To conclude this section, let us comment on the relationship between the connectives we have studied which break downward closure—NE and \diamond . As we have observed, in a setting with NE and \lor (and \top and \land), $\diamond \varphi$ is definable as ($\varphi \land NE$) $\lor \top$. On the other hand, NE is definable using \diamond (and \top) as $\diamond \top$. It clearly follows that the extension of PL_v with \diamond is also expressively complete for \mathbb{CU} . However, \diamond is stronger than NE in that swapping out \diamond with NE in any of our convex logics yields a logic that, while convex, is no longer complete for \mathbb{C} —an easy induction shows that these logics are *downward closed modulo the empty team*: for each formula φ of one of the logics, if $t \models \varphi$, $s \subseteq t$, and $s \neq \emptyset$, then $s \models \varphi$. Clearly there are convex properties which are not downward closed modulo the empty team (e.g., $\| \diamond p \land \diamond \neg p \| \in \mathbb{C}$).

3.3.2 Modal Properties

We define the syntax and the semantics of the modal extensions $ML_{\vee,\diamondsuit}(NE)$ and $ML_{\vee,\diamondsuit}(NE)$ in the obvious way. In contrast with the extensions featuring the global modalities defined in Section 3.2.2, the extension $ML_{\vee,\diamondsuit}(NE)$ is convex— the situation with \diamondsuit is analogous to that with \vee in that whereas \diamondsuit does not preserve convexity in a convex setting, it does preserve convexity in a convex and union-closed setting:

3.3.6. PROPOSITION. Formulas of $ML_{\vee,\Diamond}(NE)$ and $ML_{\vee,\Diamond}(NE)$ are convex and union closed.

Proof:

By induction on the structure of formulas φ . Most cases are straightforward—note in particular that $\Diamond \varphi$ and $\Box \varphi$ are flat and therefore convex and union closed. We only show explicitly that (i) $\Diamond \varphi$ is union closed provided that φ is union closed; and (ii) $\Diamond \varphi$ is convex provided that φ is convex and union closed.

For (i), let $T \neq \emptyset$ be such that for all $t \in T$, $M, t \models \otimes \varphi$. Then for each $t \in T$, there is an $s_t \subseteq W$ such that tRs_t and $M, s_t \models \varphi$. By union closure, $M, \bigcup_{t \in T} s_t \models \varphi$. By $s_t \subseteq R[t]$ for all $t \in T$, it follows that $\bigcup_{t \in T} s_t \subseteq \bigcup_{t \in T} R[t] = R[\bigcup T]$, and by $t \subseteq R^{-1}[s_t]$ for all $t \in T$, it follows that $\bigcup T = \bigcup_{t \in T} t \subseteq \bigcup_{t \in T} R^{-1}[s_t] = R^{-1}[\bigcup_{t \in T} s_t]$; therefore $\bigcup TR \bigcup_{t \in T} s_t$ whence $M, \bigcup T \models \otimes \varphi$.

For (ii), let $M, t \models \varphi$ and $M, s \models \varphi$ and $s \subseteq u \subseteq t$. Then there are $t', s' \subseteq W$ such that $tRt', sR's, M, t' \models \varphi$, and $M, s' \models \varphi$. By union closure, $M, t' \cup s' \models \varphi$. Define $u' := \{w \in t' \cup s' \mid \exists v \in u : vRw\}$. We will show (a) uRu' and (b) $s' \subseteq u'$. Then by (b) we will have $s' \subseteq u' \subseteq t' \cup s'$ whence by convexity, $M, u' \models \varphi$; so that by (a) we will have $M, u \models \otimes \varphi$.

For (a), clearly $u' \subseteq R[u]$. To show $u \subseteq R^{-1}[u']$, let $v \in u$. Then $v \in t \subseteq R^{-1}[t']$, so there is a $w \in t'$ such that vRw. But then $w \in u'$, whence $v \in R^{-1}[u']$. Therefore $u \subseteq R^{-1}[u']$, and so uRu'.

For (b), let $w \in s'$. Since $s' \subseteq R[s]$, there is a $v \in s$ such that vRw. Then $v \in u$, so $w \in u'$.

We can further show the modal analogue of Theorem 3.3.5 for each of these extensions. As in Section 3.2.2, we omit the details; the proofs are completely analogous to the propositional proof.

3.3.7. THEOREM. Each of $ML_{\vee,\Diamond}(NE)$ and $ML_{\vee,\Diamond}(NE)$ is expressively complete for the class of all convex union-closed modal properties invariant under bounded bisimulation.

Aloni's Bilateral State-based Modal (BSML) is essentially $ML_{\vee,\Diamond}(NE)$ extended with a *bilateral negation* which does not affect the expressive power of the logic (see [7, 13]). Therefore, the above also establishes that BSML is expressively complete for the class of all convex union-closed modal properties invariant under bounded bisimulation; this solves a problem that was left open in [7].

3.4 Uniform Definability and Uniform Extensions

In this section, we generalize the notion of uniform definability [32, 52, 133, 135, 37, 72] from the team semantics literature, use this generalization to articulate multiple senses of what it means for one team logic to extend another, and apply these definitions to clarify the relationships between the logics we have studied as well as the downward-closed logics $PL_{\vee}(=(\cdot))$ and $PL_{\rightarrow}(\vee)$.

Recall that each of $PL_{\vee}(=(\cdot))$ and $PL_{\rightarrow}(\vee)$ is expressively complete for the class of all downward-closed properties with the empty team property, whence they are expressively equivalent in the sense that whatever property is expressible in one is also expressible in the other: $\|PL_{\vee}(=(\cdot))\| = \|PL_{\rightarrow}(\vee)\| = \mathbb{D}\mathbb{E}$. Given that the split disjunction \vee preserves downward closure and the empty team property (in that for any φ and ψ with these properties, $\varphi \lor \psi$ also has these properties), this also has the consequence not only that for any $\varphi, \psi \in PL_{\vee}(=(\cdot))$ we have $\|\varphi \lor \psi\| \in \|PL_{\rightarrow}(\vee)\|$, but also that for any $\varphi, \psi \in PL_{\rightarrow}(\mathbb{W})$ we have $\|\varphi \lor \psi\| \in \|PL_{\rightarrow}(\mathbb{W})\|$ —and more generally, each property expressed by a split disjunction of downward-closed formulas is expressible in $PL_{\rightarrow}(\mathbb{W})$. However, even though, for each particular split disjunction $\varphi \lor \psi$ of $PL_{\rightarrow}(\mathbb{W})$ -formulas, there is some $PL_{\rightarrow}(\mathbb{W})$ -formula $\theta_{\varphi \lor \psi}$ that is equivalent to $\varphi \lor \psi$, it can be shown [37] that \lor is not *uniformly definable* in $PL_{\rightarrow}(\mathbb{W})$ in that there is no *context* $\theta_{\lor}[\cdot_1, \cdot_2]$ of $PL_{\rightarrow}(\mathbb{W})$ (where a context is a formula $\theta_{\lor}(\cdot_1, \cdot_2, p_1, \dots, p_n)$ with designated atoms \cdot_i) such that for any $\varphi, \psi \in PL_{\rightarrow}(\mathbb{W})$, $\varphi \lor \psi \equiv \theta_{\lor}[\varphi, \psi]$ (where $\theta_{\lor}[\varphi, \psi] \coloneqq \theta_{\lor}(\varphi/\cdot_1, \psi/\cdot_2)$). Similarly, it has been shown [135] that neither \mathbb{W} nor \rightarrow is uniformly definable in $PL_{\lor}(=(\cdot))$ (while, again, each $\|\varphi \lor \psi\|$ and each $\|\varphi \rightarrow \psi\|$ is expressible in $PL_{\lor}(=(\cdot))$).

This disconnect between expressibility and uniform definability is only possible due to the failure of closure of uniform substitution in these logics. Clearly, if each connective of one propositional team logic L₁ is uniformly definable in the other L₂ and vice versa, each property expressible in one is also expressible in the other: $\|L_1\| = \|L_2\|$. And assuming closure under uniform substitution, $\|L_1\| = \|L_2\|$ implies the uniform definability of each connective of one logic in the other. For if L₁ can express, say, $\|p \circ q\|$ where \circ is a binary connective of L₂, then there is² a context $\theta_{\circ}[\cdot_1, \cdot_2]$ of L₁ such that $\theta_{\circ}[p,q] \equiv p \circ q$. Then for any formulas φ, ψ of L₁, by closure under uniform substitution, also $\theta_{\circ}[\varphi, \psi] \equiv \varphi \circ \psi$.

Now, to connect this discussion with our concerns, let us first recall Fact 3.2.7, and restate and reprove this fact in a more abstract manner. Given properties \mathcal{P} and \mathcal{Q} , we write $\mathcal{P} \lor \mathcal{Q} := \{t \cup s \mid t \in \mathcal{P} \text{ and } s \in \mathcal{Q}\}$ and $\mathcal{P} \lor \mathcal{Q} := \mathcal{P} \cup \mathcal{Q}$; and given classes of properties \mathbb{P} and \mathbb{Q} , we write $\mathbb{P} \lor \mathbb{P} \subseteq \mathbb{Q}$ if $\mathcal{P}, \mathcal{Q} \in \mathbb{P}_X$ implies $\mathcal{P} \lor \mathcal{Q} \in \mathbb{Q}_X$, and similarly for $\mathbb{P} \lor \mathbb{P} \subseteq \mathbb{Q}$.

3.4.1. FACT. For any logic L, (i) if $\mathbb{C} \vee \mathbb{C} \subseteq ||L||$, then $||L|| \notin \mathbb{C}$, and (ii) if $\mathbb{C} \vee \mathbb{C} \subseteq ||L||$, then $||L|| \notin \mathbb{C}$.

Proof:

(i) Let $\mathcal{P} := \{\{v_1\}, \{v_2, v_3\}\}$ and $\mathcal{Q} := \{\{v_1\}\}$ (where the v_i are valuations). Then $\mathcal{P}, \mathcal{Q} \in \mathbb{C}$, but $\mathcal{P} \lor \mathcal{Q} = \{\{v_1\}, \{v_1, v_2, v_3\}\} \notin \mathbb{C}$.

(ii) Let $\mathcal{P} := \{\{v_1, v_2, v_3\}\}$ and $\mathcal{Q} := \{\{v_1\}\}$. Then $\mathcal{P}, \mathcal{Q} \in \mathbb{C}$, but we have $\mathcal{P} \lor \mathcal{Q} = \{\{v_1\}, \{v_1, v_2, v_3\}\} \notin \mathbb{C}$.

That is, if each split disjunction of convex formulas/properties is expressible in L, then L is not convex (and similarly for \forall). Given this fact and Theorem 3.2.16, there are, for instance, formulas φ, ψ of $PL_{\forall}(=(\cdot), \bullet)$ such that $\|\varphi \lor \psi\| \notin \|PL_{\forall}(=(\cdot), \bullet)\|$ —it is not the case that each split disjunction of $PL_{\forall}(=(\cdot), \bullet)$ -formulas is expressible in $PL_{\forall}(=(\cdot), \bullet)$; *a fortiori*, the split disjunction is not uniformly definable in $PL_{\forall}(=(\cdot), \bullet)$.

However, it is not difficult to see, given our results, that $\mathbb{D} \lor \mathbb{D} \subseteq ||PL_{\lor}(=(\cdot), \bullet)||$. Furthermore, given that for downward-closed φ and ψ , $\varphi \lor \psi \equiv \varphi \lor \psi$, the split disjunction

²We are assuming here that L₁ and L₂ have the *locality property*: for any formula φ of one of these logics, if $t \upharpoonright P(\varphi) = s \upharpoonright P(\varphi)$, then $t \vDash \varphi \iff s \vDash \varphi$. Some team logics, interestingly, lack this property—see, e.g., [51].

of any two downward-closed formulas *is* definable by the context $\theta_{\vee}[\cdot_1, \cdot_2] := \cdot_1 \vee \cdot_2$ in $PL_{\vee}(=(\cdot), \bullet)$. Therefore, in particular, the split disjunction restricted to $||PL_{\vee}(=(\cdot))|| = \mathbb{D}\mathbb{E}$ is definable in this sense in $PL_{\vee}(=(\cdot), \bullet)$, and in fact each connective of $PL_{\vee}(=(\cdot))$ is definable in $PL_{\vee}(=(\cdot), \bullet)$ when restricted to $||PL_{\vee}(=(\cdot))||$. And so, while $PL_{\vee}(=(\cdot), \bullet)$ is not a syntactic extension of $PL_{\vee}(=(\cdot))$, and while it is not an extension in the familiar sense in which each connective of $PL_{\vee}(=(\cdot))$ is definable in $PL_{\vee}(=(\cdot), \bullet)^3$ (since \vee is not uniformly definable in $PL_{\vee}(=(\cdot), \bullet)$), it is not only an expressive extension of $PL_{\vee}(=(\cdot))$ (in that $||PL_{\vee}(=(\cdot))|| \subseteq ||PL_{\vee}(=(\cdot), \bullet)||$), but also an extension in the deeper sense that every function/context definable in $PL_{\vee}(=(\cdot))$ (whose domain is a subset of some Cartesian power of $||PL_{\vee}(=(\cdot))||$) is definable in $PL_{\vee}(=(\cdot), \bullet)$. As defined in the literature, the concept of uniform definability in a logic L applies only to definability as restricted to ||L||. In order to make our observations precise, we must therefore generalize this notion; this is the aim of the rest of this section.

An *n*-ary context $\theta[\cdot_1, \ldots, \cdot_n]$ for a logic L is a formula θ of L with distinguished atoms \cdot_1, \ldots, \cdot_n (it may also contain other atoms such as propositional variables). Given formulas $\varphi_1, \ldots, \varphi_n$, we write $\theta[\varphi_1, \ldots, \varphi_n]$ for the formula $\theta(\varphi_1/\cdot_1, \ldots, \varphi_n/\cdot_n)$. Each *n*-ary logical connective \circ clearly defines a context $\circ[\cdot_1, \ldots, \cdot_n] := \circ(\cdot_1, \ldots, \cdot_n)$. We use the symbol for a logical connective \circ to refer interchangeably both to the connective and to the context it defines.

We restrict our attention to logics whose contexts (and connectives) are *compositional* in that given classes of propositional properties $\mathbb{P}_1, \ldots, \mathbb{P}_n, \mathbb{P}_{n+1}$ and a finite $X \subseteq P$, each such *n*-ary context θ defines a function $\theta_X^{\mathbb{P}_1, \ldots, \mathbb{P}_n, \mathbb{P}_{n+1}} : \mathbb{P}_{1X} \times \ldots \times \mathbb{P}_{nX} \to \mathbb{P}_{n+1X}$ (or simply θ) given by

$$\theta_{\mathsf{X}}^{\mathbb{P}_1,\ldots,\mathbb{P}_n,\mathbb{P}_{n+1}}(\|\varphi_1\|_{\mathsf{X}},\ldots,\|\varphi_n\|_{\mathsf{X}})=\|\varphi_{n+1}\|_{\mathsf{X}}$$

for any $\varphi_1, \ldots, \varphi_n, \varphi_{n+1}$ with $\mathsf{P}(\varphi_i) \subseteq \mathsf{X}$ and $\|\varphi_i\| \subseteq \mathbb{P}_i$ for $1 \le i \le n+1$. ⁴ We also use a logical connective \circ to refer to each such function it defines. We abbreviate $\theta_{\mathsf{X}}^{\mathbb{P}, \ldots, \mathbb{P}, \mathbb{P}}$ as $\theta_{\mathsf{X}}^{\mathbb{P}}$ (or simply $\theta^{\mathbb{P}}$).

We now recall again the notion of uniform definability from the literature.

3.4.2. DEFINITION (Uniform definability). An *n*-ary logical connective \circ is *uniformly definable* in a logic L if there exists an *n*-ary context θ_{\circ} of L such that for all $\varphi_1, \ldots, \varphi_n \in L$, $\circ(\varphi_1, \ldots, \varphi_n) \equiv \theta_{\circ}[\varphi_1, \ldots, \varphi_n]$.

Equivalently, using the functions above, \circ is uniformly definable in L if there is a context θ_{\circ} in L such that for all finite $X \subseteq P$, $\circ_{X}^{\|L\|} = \theta_{\circ X}^{\|L\|}$ (which we may write simply as $\circ^{\|L\|} = \theta_{\circ}^{\|L\|}$).

Generalizing this notion as described above, we have:

³It is in this sense that, for instance, we would say that the variant of classical propositional logic featuring the connectives \neg and \land extends the fragment of classical logic featuring only the connective \lor .

⁴We assume here that if φ is a formula of two distinct logics, then the property $\|\varphi\|_X$ defined by φ according to the semantics of one of these logics is identical to the property as defined according to the semantics of the other logic.

3.4.3. DEFINITION (Generalized uniform definability). An *n*-ary logical connective (or context) \circ is *uniformly definable with respect to* $\mathbb{P}_1, \ldots, \mathbb{P}_n, \mathbb{P}_{n+1}$ in a logic L if there exists an *n*-ary context θ_\circ for L such that for all finite $X \subseteq P$, $\circ_X^{\mathbb{P}_1, \ldots, \mathbb{P}_n, \mathbb{P}_{n+1}} = \theta_{\circ X}^{\mathbb{P}_1, \ldots, \mathbb{P}_n, \mathbb{P}_{n+1}}$. We say that \circ is *uniformly definable with respect to* \mathbb{P} in L if it is uniformly definable with respect to \mathbb{P} in L if it is uniformly definable with respect to \mathbb{P} .

We say that \circ is *locally uniformly definable* in L if it is uniformly definable in L with respect to $\|L\|$, and that it is *globally uniformly definable* in L if it is uniformly definable in L with respect to A, where A is the class of all properties.

Note that uniform definability as in Definition 3.4.2 is the same as local uniform definability.

We are now ready to define the notion of extension described above—*inner local uniform extension*. We will also define/recall other interesting notions of extension to contrast with this notion. The notions are listed in order of strength—we have:

syntactic extension \implies global uniform extension \implies outer local uniform extension \implies inner local uniform extension \implies expressive extension

Let L_1 and L_2 be team logics.

Syntactic extension: We say that L_1 is a *syntactic extension* of L_2 if the set of logical connectives of L_2 is a subset of that of L_1 .

Example: $PL_{\forall}(=(\cdot), \bullet)$ is a syntactic extension of PL_{\forall} , but not of $PL_{\forall}(=(\cdot))$.

Global uniform extension: L_1 is a *global uniform extension* of L_2 if each logical connective of L_2 is globally uniformly definable in L_1 .

Example: Since NE = $\bullet \top$, PL \rightarrow (\bullet) is clearly a global uniform extension (but not a syntactic extension) of the extension PL \rightarrow (NE) of PL \rightarrow with NE.

Outer local uniform extension: L_1 is an *outer local uniform extension* of L_2 if each logical connective of L_2 is locally uniformly definable in L_1 .

Example: Consider the extension $PL_{\forall}(=(\cdot), \bot)$ of PL_{\forall} with dependence atoms and \bot , where \bot is now a primitive with its usual semantics. It is not difficult to check (given Theorem 3.2.11) that $\|PL_{\forall}(=(\cdot), \bot)\| = \mathbb{D}$, whence $\mathbb{D} = \|PL_{\forall}(=(\cdot), \bot)\| = \|PL_{\lor}(=(\cdot))\| = \mathbb{D}\mathbb{E}$. As we have observed before, $\lor^{\mathbb{D}} = \lor^{\mathbb{D}}$, whence \lor is locally uniformly definable in $PL_{\forall}(=(\cdot), \bot)$. Clearly, therefore, $PL_{\forall}(=(\cdot), \bot)$ is an outer local uniform extension of $PL_{\lor}(=(\cdot), \bot)$. Note, however, that given Fact 3.4.1, there can be no context θ_{\lor} of $PL_{\forall}(=(\cdot), \bot)$ such that $\theta_{\lor}^{\mathbb{C}} = \lor^{\mathbb{C}}$; therefore there can also be no such context θ_{\lor} such that $\theta_{\lor}^{\mathbb{A}} = \lor^{\mathbb{A}}$. And so $PL_{\forall}(=(\cdot), \bot)$ is not a global uniform extension of $PL_{\lor}(=(\cdot))$.

Inner local uniform extension: L_1 is an *inner local uniform extension* of L_2 if each logical connective \circ of L_2 is uniformly definable in L_1 with respect to $||L_2||$.

3.4. Uniform Definability and Uniform Extensions

Examples: Given that $\vee^{\mathbb{D}\mathbb{E}} = \vee^{\mathbb{D}\mathbb{E}}$ and $\vee^{\mathbb{D}\mathbb{E}} = \vee^{\mathbb{D}\mathbb{E}}$, we have that \vee is uniformly definable in $PL_{\vee}(=(\cdot), \bullet)$ with respect to $\mathbb{D}\mathbb{E} = \|PL_{\vee}(=(\cdot))\|$, and that \vee is uniformly definable in $PL_{\rightarrow}(\forall, \bullet)$ with respect to $\mathbb{D}\mathbb{E} = \|PL_{\rightarrow}(\forall)\|$. It is then easy to see that $PL_{\vee}(=(\cdot), \bullet)$ is an inner local uniform extension of $PL_{\vee}(=(\cdot))$, and similarly with $PL_{\rightarrow}(\forall, \bullet)$ and $PL_{\rightarrow}(\forall)$. Given Fact 3.4.1, however, $PL_{\vee}(=(\cdot), \bullet)$ is not an outer local uniform extension of $PL_{\vee}(=(\cdot))$, and sufficient extension of $PL_{\rightarrow}(\forall)$.

Expressive extension: L_1 is an *expressive extension* of L_2 if $||L_2|| \subseteq ||L_1||$.

Examples: $PL_{\vee}(=(\cdot))$ is an expressive extension of $PL_{\rightarrow}(\vee)$ (and vice versa), but not of $PL_{\vee}(=(\cdot), \bullet)$. We mentioned above that \vee and \rightarrow are not (locally) uniformly definable in $PL_{\vee}(=(\cdot))$. They are therefore not uniformly definable in $PL_{\vee}(=(\cdot))$ with respect to $\|PL_{\rightarrow}(\vee)\| = \|PL_{\vee}(=(\cdot))\|$; therefore, $PL_{\vee}(=(\cdot))$ is not an inner local uniform extension of $PL_{\rightarrow}(\vee)$.

We noted above that in a setting in which closure under uniform substitution holds, expressive equivalence and uniform definability of connectives collapse into one another. Now observe further that, essentially by the same argument, the notions of expressive extension, inner/outer local uniform extension and global uniform extension are all equivalent in such a setting.

To repeat what was already mentioned above, if L_1 is an inner local uniform extension of L_2 , then each function definable only by the means available in L_2 (in that such a function is definable by a context of L_2 and its domain is a subset of some Cartesian power of $||L_2||$) is also definable by a context of L_1 . It is for this reason that we call $PL_{\forall}(=(\cdot), \bullet)$ a convex variant of $PL_{\forall}(=(\cdot))$, and $PL_{\rightarrow}(\forall, \bullet)$ a convex variant of $PL_{\rightarrow}(\forall)$; it is not clear to us whether $PL_{\rightarrow}(\bullet)$ is also an inner local uniform extension of $PL_{\rightarrow}(\forall)$ (see Open problem 3 below), and we therefore refrain from calling it a convex variant of $PL_{\rightarrow}(\forall)$.

Let us conclude by noting one result and three open problems concerning uniform definability and inner local uniform extensionhood in our logics.

As mentioned above, it is shown in [135] that \rightarrow is not uniformly definable in PL_v(= (·)). Essentially the same proof shows that \rightarrow is also not uniformly definable in PL_v(= (·), •), whence PL_v(=(·), •) is not an inner local uniform extension of PL_{\rightarrow}(\forall , •) or of PL_{\rightarrow}(\diamond).

Open problem(s) 1: Is \vee uniformly definable in $PL_{\rightarrow}(\bullet)$ or in $PL_{\rightarrow}(\forall, \bullet)$? (Is $PL_{\vee}(=(\cdot), \bullet)$ an inner local uniform extension of $PL_{\rightarrow}(\bullet)/PL_{\rightarrow}(\forall, \bullet)$?) It might be possible to adapt the proof from [37] showing that \vee is not uniformly definable in $PL_{\rightarrow}(\forall)$ to show that this is not the case.

(Note that dependence atoms are uniformly definable in both $PL_{\rightarrow}(\mathbb{W}, \bullet)$ and in $PL_{\rightarrow}(\bullet)$. For $PL_{\rightarrow}(\mathbb{W}, \bullet)$, we have (following the definition of dependence atoms in $PL_{\rightarrow}(\mathbb{W})$, which follows essentially from [2]):

$$=(p_1,\ldots,p_n,p)\equiv((p_1 \boxtimes \neg p_1) \land \ldots (p_n \boxtimes \neg p_n)) \to (p \boxtimes \neg p).$$

For $PL_{\rightarrow}(\bullet)$, we use the above in conjunction with Proposition 3.2.13.)

Open problem(s) 3: Is \forall uniformly definable with respect to $\mathbb{D}\mathbb{E}$ in $PL_{\rightarrow}(\bullet)$? (Is $PL_{\rightarrow}(\bullet)$ an inner local uniform extension of $PL_{\rightarrow}(\forall)$?) It seems likely that if one solves this, one also solves whether \forall is uniformly definable in $PL_{\rightarrow}(\bullet)$. (Is $PL_{\rightarrow}(\bullet)$ an inner local uniform extension of $PL_{\rightarrow}(\forall, \bullet)$?) Observe that in Proposition 3.2.13, we only showed that \forall and \forall are uniformly definable in $PL_{\rightarrow}(\bullet)$ with respect to \mathbb{F} .

3.5 Conclusion

In this paper, we introduced the logics $PL_{\forall}(=(\cdot), \bullet)$, $PL_{\rightarrow}(\forall, \bullet)$ and $PL_{\rightarrow}(\bullet)$, which we showed to be expressively complete for the class of all convex propositional properties. We also introduced modal extensions of these logics and showed modal analogues of the expressive completeness theorems for the extensions of the convex logics with the flat modalities \diamondsuit and \square . We then examined some union-closed convex logics from the literature: we showed that the propositional logic $PL_{V}(NE)$ is expressively complete for the class of all union-closed convex propositional properties, and we showed modal analogues of this result for the modal extensions $ML_{\vee,\Diamond}(NE)$, $ML_{\vee,\Diamond}(NE)$ and BSML of $PL_{V}(NE)$. We also showed that while the split disjunction \vee and the global diamond \otimes preserve convexity in a convex and union-closed setting, they do not preserve convexity in general. Similarly, while the global disjunction W preserves downward closure (and hence convexity) in a downward-closed setting, it does not preserve convexity in general. Finally, we generalized the notion of uniform definability from the literature in order to articulate the sense in which our convex variant $PL_{\forall}(=(\cdot), \bullet)$ of dependence logic $PL_{\vee}(=(\cdot))$ our convex variant $PL_{\rightarrow}(\forall, \bullet)$ of inquisitive logic extend their downward-closed counterparts. The convex variants are inner local uniform extensions of the downward-closed counterparts: each connective of the counterpart is uniformly definable in the convex variant with respect to class of properties expressible in the counterpart.

We note two more open problems to add to the list in Section 3.4:

Open problem(s) 4: Axiomatizations of the logics studied. There is a commonlyused strategy for finding axiomatizations of propositional and modal team logics (see, for instance, [41, 139, 140, 134, 137]) that makes heavy use of the characteristic formulas for properties provided by expressive completeness results such as those established in this paper. Given our expressive completeness results, we expect it to be possible to use this strategy to construct axiomatizations of the logics we have studied. It would be particularly interesting to see axiomatizations of the convex propositional logics, to see how these axiomatizations differ from those of the downward-closed logics $PL_{\rightarrow}(\mathbb{W})$ and $PL_{\vee}(=(\cdot))$. (Note that $PL_{\vee}(NE)$ has already been axiomatized in [140], and $ML_{\odot,\vee}(NE)$ and BSML in [7]. An extension of $ML_{\odot,\vee}(NE)$ has been
3.5. Conclusion

axiomatized in [12].)

Open problem(s) 5: First-order dependence logic (D) coincides in expressive power with the downward-monotone (or downward-closed) fragment of existential second-order logic (ESO) in the sense that D and this fragment define the same class of team properties [93]. Similarly, we have, for instance, that first-order *independence logic* coincides with full ESO [51]; D with the *Boolean negation* coincides with full second-order logic (SO) [91], as does first-order independence logic with \rightarrow [132, 133]; D with \rightarrow coincides with the downward-closed fragment of SO [132, 133]; and there are other team logics coinciding in this way with the union-closed fragment of ESO [80], as well as with both full first-order logic and its downward-closed fragment [95]. Is it possible to construct a convex variant of first-order dependence logic (or indeed any natural team logic, perhaps dissimilar to dependence logic) which coincides in this sense with the convex fragment of ESO? Similarly, can we find a natural team logic equivalent to the convex union-closed fragment of ESO? How about the convex (and convex and union-closed) fragments of first-order and second-order logic?

Chapter 4

Further Remarks on the Dual Negation in Team Logics

This chapter is based on:

Aleksi Anttila. Further remarks on the dual negation in team logics. 2024. arXiv: 2410.07067 [math.LO]. URL: https://arxiv.org/abs/2410.07067

Abstract The dual or game-theoretical negation \neg of independence-friendly logic (IF) and dependence logic (D) exhibits an extreme degree of semantic indeterminacy in that for any pair of sentences φ and ψ of IF/D, if φ and ψ are incompatible in the sense that they share no models, there is a sentence θ of IF/D such that $\varphi \equiv \theta$ and $\psi \equiv \neg \theta$ (as shown originally by Burgess in the equivalent context of the prenex fragment of Henkin quantifier logic). We show that by adjusting the notion of incompatibility employed, analogues of this result can be established for a number of modal and propositional team logics, including Aloni's bilateral statebased modal logic, Hawke and Steinert-Threlkeld's semantic expressivist logic for epistemic modals, as well as propositional dependence logic with the dual negation. Together with its converse, a result of this type can be seen as an expressive completeness theorem with respect to the relevant incompatibility notion; we formulate a notion of expressive completeness for pairs of properties to make this precise.

4.1 Introduction

4.1.1 The Dual Negation and Burgess' Theorem

Henkin quantifier logic (H) extends classical first-order logic (FO) with *Henkin quantifiers* [74] (also known as *branching* or *partially ordered quantifiers*) such as the following:

$$\begin{pmatrix} \forall x & \exists y \\ \forall u & \exists v \end{pmatrix} \varphi(x, y, u, v).$$
(4.1)

The intuitive meaning of (4.1) is that the value of *y* depends only on that of *x*, and the value of *v* only on that of *u*—(4.1) is equivalent to the existential second-order sentence $\exists f \exists g \forall x \forall u \varphi(x, f(x), u, g(u))$ (cf. the FO-sentence $\forall x \exists y \forall u \exists v \varphi(x, y, u, v)$). Hintikka and Sandu's *independence-friendly logic* (IF) [77, 75] similarly extends FO with *slashed quantifiers* $(\exists y/\forall x)$, with the intended meaning of $(\exists y/\forall x)$ being that the value of *y* must be chosen independently of the value of *x* (the IF version of (4.1) is $\forall x \exists y \forall u (\exists v/\forall x) \varphi(x, y, u, v)$). Finally, *dependence logic* (D) [119], Väänänen's refinement of IF, replaces the slashed quantifiers with *dependence atoms*: the atom = (x_1, \ldots, x_n, y) asserts that the value of *y* is functionally determined by those of x_1, \ldots, x_n (the dependence logic version of (4.1) is $\forall x \exists y \forall u \exists v (\varphi(x, y, u, v) \land = (u, v))$)).

The prenex fragment of H (denoted H_p; the formulas of H_p are of the prenex form $Q\varphi$, where the prefix Q is a set of quantifiers with a partial order and the matrix φ is a quantifier-free FO-formula) is expressively equivalent to existential second-order logic Σ_1^1 [124, 48], and the same holds for the full logics IF [75] and D [119]. These logics are therefore not closed under classical negation ~ (where ~ φ is true just in case φ is false).¹ The original semantics for IF were game-theoretical, with the meaning of $(\exists y/\forall x)$ captured with a game rule to the effect that the player choosing the value of y must do so without knowing the value of x—the game is one of *imperfect information* (whereas in the game for FO the players have perfect information). This naturally led to the adoption (in both IF and D) of what is known as the *dual* or *game-theoretical* negation \neg . The rule associated with this negation in the semantic game is simply the standard negation rule (also used in the game-theoretical semantics for FO) whereby the players switch their verifier/falsifier roles², and indeed, IF and D are conservative extensions of FO, with the dual negation extending the classical negation of FO.

In the context of the full logics, however, the dual negation exhibits non-classical behavior. Most germane to our purposes is the fact that *preservation of equivalence under replacement* fails in negated contexts—for instance, $\varphi \equiv \psi$ need not imply $\neg \varphi \equiv \neg \psi$. (A concrete example: in dependence logic we have $\neg = (x, y) \equiv \bot$, but $\neg \neg = (x, y) \equiv = (x, y) \not\equiv \bot \equiv \neg \neg \bot$.) Put another way, each sentence φ defines a class of models $\|\varphi\| = \{\mathcal{M} \mid \mathcal{M} \models \varphi\}$ which we may think of as the meaning of φ , and whereas negating a sentence in FO corresponds to the semantic operation of complementation on these classes of models, the dual negation fails to correspond to any operation on such

¹The fact that Σ_1^1 is not closed under ~ follows, for instance, from the compactness of Σ_1^1 together with the fact that one can define a Σ_1^1 -sentence which is true in a model just in case the model is infinite.

²The truth conditions of the dual negation can equivalently be obtained by first defining the conditions for negated classical atoms by translating the standard Tarskian truth conditions for such negated atoms to the semantic framework one is working with (e.g. game-theoretic-semantics or team semantics), and then defining those for complex formulas by stipulating that the following dual (or almost dual) equivalences are to hold: $\varphi \equiv \neg \neg \varphi$; $\neg(\varphi \land \psi) \equiv \neg \varphi \lor \neg \psi$; $\neg(\varphi \lor \psi) \equiv \neg \varphi \land \neg \psi$; $\neg \forall x \varphi \equiv \exists x \neg \varphi$; $\neg \exists x \varphi \equiv \forall x \neg \varphi$; $\neg(\exists y/\forall x) \varphi \equiv \forall x \neg \varphi$ (in IF); and $\neg =(x_1, \dots, x_n, y) \equiv \bot$ (in D).

4.1. Introduction



Figure 4.1: Failure of determination: in FO, $\|\neg \varphi\|$ is the complement of $\|\varphi\|$. In H_p, given only $\|\varphi\|$, $\|\neg \varphi\|$ can be any class disjoint with $\|\varphi\|$ that is expressible in H_p.

classes: the class of models $\|\varphi\|$ of φ does not determine the class of models $\|\neg\varphi\|$ of its dual negation. This does not mean that $\|\neg\varphi\|$ is fully unconstrained: one can show that φ and $\neg\varphi$ must be incompatible in that they share no models ($\|\varphi\| \cap \|\neg\varphi\| = \emptyset$), with the effect that despite some standard inferential principles involving negation (such as negation introduction $\Gamma, \varphi \models \bot \implies \Gamma \models \neg \varphi$) being invalidated due to the lack of determination, others (such as *ex falso* $\varphi, \neg \varphi \models \psi$) remain valid.

Burgess [29], however, showed that this failure of determination is extreme in the sense that incompatibility is the *only* constraint that $\|\varphi\|$ places on $\|\neg\varphi\|$. That is, he showed (in the context of H_p^3) that for any sentences φ, ψ , if φ and ψ share no models $(\|\varphi\| \cap \|\psi\| = \emptyset)$, then there is a sentence θ such that $\varphi \equiv \theta$ and $\psi \equiv \neg\theta (\|\varphi\| = \|\theta\|$ and $\|\psi\| = \|\neg\theta\|$). So knowing only $\|\varphi\|$ (without knowing the sentence φ) tells us nothing whatsoever about $\|\neg\varphi\|$, save for the fact that the two classes are disjoint and the fact that $\|\neg\varphi\|$ is expressible in H_p : for a given $\|\varphi\|$, take any class \mathcal{X} expressible in H_p and disjoint with $\|\varphi\|$. By Burgess' theorem, there is then a θ with $\|\varphi\| = \|\theta\|$ and $\mathcal{X} = \|\neg\theta\|$, so for all we know, $\|\neg\varphi\|$ might be \mathcal{X} . See Figure 4.1. Dechesne [44] later reformulated the result for IF. Kontinen and Väänänen [94] (working in D and IF) generalized the theorem to arbitrary formulas (that may contain free variables). Mann (in [98]) proved an analogue of the result for sentences for the *perfect-recall* fragment of IF.⁴

4.1.2 Burgess Theorems for Modal and Propositional Team Logics

In this paper, we prove analogues of Burgess' theorem for a number of modal and propositional *team logics*.

Team logics are logics such as dependence logic which are primarily intended to

³Burgess proved his result for sentence of H_p and their *contraries*, where the contrary of $Q\varphi$ is obtained by swapping all quantifiers in Q with their duals (\forall with \exists and vice versa), and negating the FO matrix φ . This is equivalent to the dual negation in that given a sentence φ of H_p and an equivalent sentence ψ of D or IF, the contrary of φ and the dual negation of ψ are also equivalent.

⁴See also [20], which corrects for the failure of determination in that it introduces an extension of IF featuring a negation with a natural game-theoretic interpretation which does not exhibit this failure.

be interpreted using *team semantics*. In team semantics (introduced by Hodges [78, 79] to provide a compositional semantics for IF, and independently developed as a semantics for *inquisitive logic* chiefly by Ciardelli, Groenendijk, and Roelofsen [41, 38, 35]), formulas are evaluated with respect to *sets* of evaluation points called *teams* rather than single points, as in standard Tarskian semantics (so, for instance, in first-order team semantics, teams are sets of variable assignments; in modal team semantics [120], teams are sets of possible worlds, etc.).⁵

Contemporary team logics in the lineage of dependence logic typically do not incorporate the dual negation. However, interest in this negation has recently been reinvigorated due to team logics in the philosophical logic and formal semantics literature which employ *bilateral* notions of negation similar to the dual negation. We are concerned with two such logics in particular: Aloni's *bilateral state-based modal logic* (BSML) [6], and Hawke and Steinert-Threlkeld's semantic expressivist logic for epistemic modals [67]. These are both propositional modal team logics developed to account for linguistic phenomena such as *free choice inferences* and *epistemic contradictions*. The semantics of each of these logics can be formulated bilaterally—that is, using both a positive primitive semantic relation \models (interpreted as assertibility), as well as a negative primitive relation \exists (interpreted as rejectability). The bilateral semantics is used, in each of these logics, to define a *bilateral negation* \neg : for a team *s*, $s \models \neg \varphi$ iff $s \exists \varphi$, and $s \exists \neg \varphi$ iff $s \models \varphi$. The resulting bilateral negations are very similar to the dual negation in that they exhibit failure of determination, and, as we will show, an analogue of Burgess' theorem holds for each of them.

In the first part of this paper, we show the analogue for BSML (we additionally show one for BSML^V, the extension of BSML with the *global* or *inquisitive disjunction* \forall). This demonstrates, as noted above, that the negation in BSML can be seen as a type of dual negation, and establishes that what we will call Burgess theorems can also be obtained in the propositional modal setting. BSML and BSML^V differ substantially from D and IF not only in being propositional and modal rather than first order, but also in that formulas of D and IF are *downward closed*—a formula is downward closed if its truth in a team *t* implies truth in all subteams $s \subseteq t$ of *t*—whereas those of BSML and BSML^V need not be. This gives rise to an interesting technical detail: we identify

⁵The term 'team logic' is from Väänänen [119], who originally used it to refer to the extension D with the *Boolean* or *contradictory negation* ~ (where ~ φ is true in a team iff φ is false), and it is now frequently used to refer to any logic employing team semantics that incorporates ~. The term is used both in this sense and in the more general sense (whereby 'team logics' are logics intended to be interpreted using team semantics) in the literature.

We noted earlier that D and IF are not closed under ~, and referred to ~ as the 'classical negation'. This name is appropriate when discussing the relationship between D, IF, and Σ_1^1 because on the level on which D and IF correspond to Σ_1^1 (essentially the level of teams), ~ corresponds to the standard, classical negation of second-order logic. In the context of team semantics, the name 'classical negation' would potentially be confusing because there is another level (essentially the level of assignments or worlds—the elements of teams) on which ~ behaves nonclassically (on this level it is the dual negation rather than ~ which corresponds, in the FO-fragments of D and IF, to the classical negation of FO). See Fact 4.2.4 and Section 4.3.1.

4.1. Introduction

two distinct natural incompatibility notions which are equivalent to one another (and to the notion employed by Burgess) in a downward-closed setting, but which come apart when downward closure fails (cf. [97, Section 6.4]). We show that the weaker of these notions (\perp -*incompatibility*) is not sufficiently strong to yield a Burgess theorem for BSML or BSML^V, but that the stronger notion (*ground-incompatibility*) does suffice.

Burgess observed that his theorem can be seen as a result on the expressive power of H_p :

The Enderton-Walkoe theorem says that for any PC [pseudo-elementary (i.e., Σ_1^1 -definable) class of models], call it *K*, there is a Henkin sentence θ such that $K = ||\theta||$. The corollary just proved allows this theorem to be strengthened to say that for any two disjoint PCs, call them K_0 and K_1 , there is a Henkin sentence θ such that $K_0 = ||\theta||$ and $K_1 = ||\neg \theta||$. [29, Notation amended.]

Dechesne [44] similarly formulates the theorem as an expressivity result for pairs. We develop this idea further by combining it with our distinct notions of incompatibility, which are essentially properties of pairs (of classes of pointed models, in the modal setting): a Burgess theorem, together with its converse, can be seen as establishing that the logic in question is expressively complete for the class of pairs satisfying the relevant pair property/incompatibility notion.⁶ For instance, the results described above imply that whereas D is expressively complete both for relevant pairs that are \pm -incompatible as well as those that ground-incompatible, BSML is only complete for ground-incompatible relevant pairs. We formulate a notion of expressive completeness for pairs (*bicompleteness*) to succinctly describe these results.

Whereas the logics in the first part of the paper employ modal team semantics team semantics on Kripke models—in the second part of the paper, we consider logics with dual-like negations interpreted using propositional team semantics. The simpler propositional setting allows us to readily prove multiple Burgess/bicompleteness theorems for multiple notions of incompatibility⁷, giving us a better handle on the notion of bicompleteness as well as on the diverse behavior this type of negation can be induced to exhibit. We show Burgess theorems for Hawke and Steinert-Threlkeld's logic HS [67] (which we treat here as a propositional logic despite its featuring a modality since it is interpreted on propositional teams); for the propositional fragments of BSML, and BSML^V (both introduced in [140]); and for *propositional dependence logic* [139] with the dual negation, the propositional version of D. These theorems require the introduction of yet more pair properties/incompatibility notions. While our proofs for the modal logics (as well as that for the propositional fragment of BSML^V) are analogous to Burgess'⁸ (these results are essentially corollaries of interpolation for

⁶I am grateful to Søren Brinck Knudstorp for suggesting this idea to me.

⁷We will speak interchangeably of incompatibility notions and pair properties, but it should be noted that many of the pair properties we identify do not correspond to any concept of 'incompatibility' as ordinarily conceived.

⁸This is why we choose to first work in the more complicated modal setting.

classical modal/propositional logic; Burgess', similarly, is a corollary of interpolation for FO), we employ different methods to prove the other propositional theorems: the HS-result follows easily from the basic properties of the negation in HS, while the results for the propositional fragment of BSML and propositional dependence logic are shown to follow from the expressive completeness theorems for these logics. We also demonstrate, in the second part, what happens when the notion of bicompleteness is applied to propositional logics which do not exhibit failure of determination (for instance, classical propositional logic interpreted on teams), and provide an example of maximal failure of determination by introducing a logic which is bicomplete for all propositional pair properties.

While we distinguish the different pair properties/incompatibility notions for technical reasons, many of the notions we consider are also conceptually suggestive. In the third part of the paper, we sketch some possible intuitive interpretations of some of the notions.

We prove the BSML- and $BSML^{\vee}$ -theorems in Section 4.2 and the propositional theorems in Section 4.3. We consider possible interpretations of the incompatibility notions in Section 4.4. We conclude, in Section 4.5, with some discussion of the results. In the remainder of this introduction, I comment briefly on how Burgess' theorem has been interpreted.

4.1.3 Remarks on Burgess' Theorem

Burgess himself intended his theorem to serve, in part, as a point against IF and Hintikka's philosophical ambitions. He writes:

In recent years Hintikka and co-workers have revived a variant version of the logic of Henkin sentences under the label "independence-friendly" logic, have restated many theorems about existential second-order sentences for this "new" logic, and have made very large claims about the philosophical importance of the theorems thus restated. In discussion, pro and con, of such philosophical claims it has not been sufficiently emphasized that contrariety [dual negation], the only kind of "negation" available, fails to correspond to any operation on classes of models. For this reason it seemed worthwhile to set down, in the form of the corollary above, a clear statement of just how total the failure is. [29]

Accordingly, the result is occasionally cited as attesting that the behaviour of the negation is anomalous or problematic (as in [82]). A common gloss has it that Burgess establishes or shows that the dual negation is "not a semantic operation".

I do not intend to assess the merits of Burgess' argument, or to argue for or against the dual negation here. What I do hope to do is to point out why the common characterization of the remark is potentially misleading in multiple ways.⁹

⁹It is also worth pointing out, in connection with the passage from Burgess concerning Hintikka

4.1. Introduction

My first qualm with the gloss is that it is not Burgess' theorem that establishes the "non-semantic" nature of the dual negation. The failure of the negation to correspond to any operation on classes of models is an easily observable, simple fact (recall our dependence logic example: $\neg = (x, y) \equiv \bot$, but $\neg \neg = (x, y) \not\equiv \neg \neg \bot$). The theorem is a further fact, characterized by Burgess as concerning the degree of this failure. Indeed, observe that in the passage above, the theorem functions as a kind of reductio of the failure itself: the (implied) problem with the negation is its failure to correspond to any operation on classes of models. The theorem merely serves to demonstrate the extreme level of this failure, and this extremeness is, in turn, implied to constitute part of what Burgess appears to view as the absurd consequences of adopting an operator that fails to so correspond. The failure itself is independent of the theorem. The results in this paper help to reinforce this point: we prove variants of Burgess' theorem with respect to multiple distinct pair properties/notions of incompatibility, which we may think of as constituting a way of measuring the degree of failure of correspondence in that a Burgess theorem with respect to a weaker notion of incompatibility corresponds to a higher degree of failure (see Section 4.5)—we have, then, failure of determination in multiple different logics, regardless of whether Burgess' original theorem holds for that logic (and also regardless of whether some variant of the theorem can be established), with the Burgess theorems providing insight into the degree and nature of the failure in each case.

Second, as essentially already pointed out by Hodges [78], and later by many others [119, 44, 94, 58] (and as is already implicit in our discussion of bicompleteness), there is a sense in which the dual negation is a semantic operation, which, while perhaps trivial, should not be ignored. Indeed, Hodges' [78] goal in developing team semantics was to demonstrate that logics of imperfect information such as IF could be given a semantics in which all connectives are semantic operations in the sense that replacing a subformula occurrence with one with the same meaning does not change the meaning of a formula. Instead of taking the meaning of φ to be $\|\varphi\|$, let it be represented by the pair $(\|\varphi\|, \|\neg \varphi\|)$. Then the meaning of $\neg \varphi$ is determined by that of φ since to obtain the former from the latter, one need only flip the elements of the pair: $(\|\neg \varphi\|, \|\varphi\|)$.

One could of course perform a similar trick with any other operator or connective no matter its semantics, but there are potentially good reasons for insisting that, applied to the negation, this move is not entirely *ad hoc*. For instance, on the view known both

and IF, that Hintikka did also consider an extension of IF with the Boolean/contradictory negation ~ (*extended independence-friendly logic*), and that he ultimately viewed each negation as indispensable [75, 76]. Consider, for instance, the following passage [75, p. 154]:

^{...}*in any sufficiently rich language, there will be two different notions of negation present.* Or if you prefer a different formulation, our ordinary concept of negation is intrinsically ambiguous. The reason is that one of the central things we certainly want to express in our language is the contradictory negation. But ... a contradictory negation is not selfsufficient. In order to have actual rules for dealing with negation, one must also have the dual negation present, however implicitly.

as *rejectivism* and as *bilateralism* (see, for instance, [104, 105, 115, 110]), the speech act of rejection is not reducible to that of assertion—these notions should be treated as being on par, with neither being conceptually reducible to the other. If one endorses this view, it is natural to conceive of the pair $(\|\varphi\|^+, \|\varphi\|^-)$ as the full meaning of φ , where $\|\varphi\|^+$ denotes the models in which in φ is assertible, and $\|\varphi\|^-$ those in which it is rejectable. And if, as in BSML, one additionally associates rejectability with a negation operator (cf. varieties of rejectivism with a specialized rejection operator such as [115, 110]), one may equate $(\|\varphi\|^+, \|\varphi\|^-)$ with $(\|\varphi\|, \|\neg\varphi\|)$.¹⁰ Whether it is possible to formulate a tenable form of rejectivism featuring the dual negation in this way is beyond the scope of this paper; at any rate, given the clear similarity between Hodges' notion of meaning and that of the rejectivists, it seemed worthwhile to make this connection explicit (rejectivism has not thus far, to my knowledge, been discussed in the literature on the dual negation). It should also be noted that, as alluded to in endnote 8, bilateral notions of meaning of this kind have been proposed for reasons other than rejectivism.

4.2 Burgess Theorems for BSML and BSML^{\vee}

In this section, we list the required preliminaries concerning the logics BSML and $BSML^{\vee}$ and team semantics (Section 4.2.1); reformulate Burgess' theorem as an expressive completeness theorem for pairs satisfying a specific notion of incompatibility (Section 4.2.2); distinguish between multiple notions of incompatibility and examine how they are related (Section 4.2.2); and show Burgess theorems for BSML and $BSML^{\vee}$ using some of these incompatibility notions (Section 4.2.3).

4.2.1 Preliminaries

We first define the syntax and semantics of BSML and BSML^{\vee} (from [6, 7]); then recall the definitions of standard team-semantic closure properties and list some results relating these properties to our logics (from [7, 16]); and, finally, recall basic notions and results from modal logic (see, e.g., [24, 56]), together with team-based analogues (from [71, 7]).

¹⁰The bilateral semantics of BSML and Hawke and Steinert-Threlkeld's logic are not motivated by rejectivism *per se*; the primary intended function of bilateralism in these logics is, rather, to allow these logics to correctly capture and predict linguistic data. Aloni [6] makes a point in defence of failure of replacement by drawing on these empirical considerations:

As we will see, however, this non-classical behavior is precisely what we need to explain the effects of pragmatic enrichment in negative contexts. [...] when replacement under \neg holds, the rejection conditions are derivable from the support conditions, and so without failure of replacement bilateralism would not give different predictions from unilateral systems and so it would be empirically unjustified.

4.2. Burgess Theorems for BSML and $BSML^{\vee}$

BSML is an extension of classical modal logic with the *nonemptiness atom* NE (introduced in [121, 140]), which is true in a team just in case the team is nonempty. BSML^{\vee} is BSML extended with the *global* or *inquisitive disjunction* \vee , used in inquisitive logic [41, 38, 35] (see Sections 4.3.1 and 4.4.4) to model the meanings of questions.

4.2.1. DEFINITION (Syntax of ML, BSML, and BSML^{\vee}). Fix a (countably infinite) set Prop of propositional variables. The set of formulas of *bilateral state-based modal logic* BSML is generated by:

$$\varphi ::= p \mid \perp \mid \neg \varphi \mid (\varphi \land \varphi) \mid (\varphi \lor \varphi) \mid \diamondsuit \varphi \mid \text{NE}$$

where $p \in \text{Prop. } Classical modal logic ML is the NE-free fragment of BSML. BSML^V is the extension of BSML with the binary connective <math>\mathbb{V}$.

We use the first Greek letters α, β to refer exclusively to formulas of ML (also called *classical formulas*). We write P(φ) for the set of propositional variables in φ .

A Kripke model M = (W, R, V) over $X \subseteq$ Prop is defined as usual, where in particular, $V : X \rightarrow \wp(W)$. We call a subset $s \subseteq W$ of W a (modal) *team* on M. For any world win M, define, as usual, $R[w] := \{v \in W \mid wRv\}$. Similarly, for any team s on M, define $R[s] := \bigcup_{w \in S} R[w]$.

4.2.2. DEFINITION (Semantics of ML, BSML and BSML^{\forall}). For a model M = (W, R, V) over X, a team *s* on *M*, and formula φ with $P(\varphi) \subseteq X$, the notions of φ being *supported by/anti-supported by s* in *M*, written $M, s \models \varphi/M, s \dashv \varphi$ (or simply $s \models \varphi/s \dashv \varphi$), are defined recursively as follows:

| $M, s \models p$ $M, s \dashv p$ | $: \iff$ | for all $w \in s : w \in V(p)$; for all $w \in s : w \notin V(p)$; |
|-------------------------------------|----------|---|
| $M, s \exists p$ | . ⇔ | $\text{for all } w \in S \cdot w \notin V(p),$ |
| $M, s \vDash \bot$ | $: \iff$ | $s = \emptyset;$ |
| $M, S \dashv \bot$ | | always the case; |
| $M, s \vDash \text{NE}$ | : ⇔ | $s \neq \emptyset;$ |
| $M, s \dashv \text{NE}$ | $: \iff$ | $s = \emptyset;$ |
| $M, s \vDash \neg \varphi$ | : ⇔ | $M, s = \varphi;$ |
| $M, s \dashv \neg \varphi$ | : ⇔ | $M, s \vDash \varphi;$ |
| $M, s \vDash \varphi \land \psi$ | : ⇔ | $M, s \vDash \varphi$ and $M, s \vDash \psi$; |
| $M, s \dashv \varphi \land \psi$ | : ⇔ | there exist t, u s.t. $s = t \cup u$ and $M, t = \varphi$ and $M, u = \psi$; |
| $M, s \vDash \varphi \lor \psi$ | : ⇔ | there exist t, u s.t. $s = t \cup u$ and $M, t \models \varphi$ and $M, u \models \psi$; |

 $\begin{array}{ll} M, s = \varphi \lor \psi & : \Longleftrightarrow & M, s = \varphi \text{ and } M, s = \psi; \\ M, s \models \varphi \lor \psi & : \Longleftrightarrow & M, s \models \varphi \text{ or } M, s \models \psi; \\ M, s = \varphi \lor \psi & : \Longleftrightarrow & M, s = \varphi \text{ and } M, s = \psi; \\ M, s \models \Diamond \varphi & : \Longleftrightarrow & \text{for all } w \in s \text{ there exists } t \subseteq R[w] \text{ s.t. } t \neq \emptyset \text{ and } M, t \models \varphi; \\ M, s = \Diamond \varphi & : \Longleftrightarrow & \text{for all } w \in s \text{ there exists } t \subseteq R[w] = \varphi. \end{array}$

We also refer to support by *s* as *truth in s* for convenience, although we caution that most interpretations of team logics rely on distinguishing the way in which a team satisfies a formula (often called 'support') from the way in which an element of a team satisfies one (often called 'truth'; see Fact 4.2.4). We write $\varphi \models \psi$ and say φ *entails* ψ if for all *M* and all *s* on *M*, *M*,*s* $\models \varphi$ implies *M*,*s* $\models \psi$. If both $\varphi \models \psi$ and $\psi \models \varphi$, then we write $\varphi \equiv \psi$ and say that φ and ψ are *equivalent*. If both $\varphi \equiv \psi$ and $\neg \varphi \equiv \neg \psi$, then φ and ψ are said to be *bi-equivalent* or *strongly equivalent*, written $\varphi \equiv^{\pm} \psi$.

We refer to the atom \perp as the *weak contradiction*. We also define the *strong contradiction* $\perp := \perp \land \text{NE}$, and $\top := \neg \perp$. The weak contradiction is true only in the empty team; the strong contradiction in no team; \top in all teams. Note that *ex falso* with respect to \perp holds for all formulas φ : $\perp \models \varphi$; with respect to \perp it holds only for formulas ψ with the empty team property (see below): $\perp \models \psi$ if $M, \varnothing \models \psi$ for all M. We let $\lor \varnothing := \perp$; $\lor \forall \varnothing := \bot$; and $\Box \varphi := \neg \diamondsuit \neg \varphi$. Observe that $\neg(\varphi \lor \psi) \equiv \neg \varphi \land \neg \psi \equiv \neg(\varphi \lor \psi); \neg \neg \varphi \equiv \varphi$; and $\neg \diamondsuit \varphi \equiv \Box \neg \varphi$. It follows from these equivalences that each formula is equivalent to one in negation normal form.

4.2.3. DEFINITION (Closure properties). We say that a formula φ

- is *downward closed*, provided $[M, s \vDash \varphi \text{ and } t \subseteq s] \Longrightarrow M, t \vDash \varphi;$
- is *convex*, provided $[M, s \models \varphi; M, t \models \varphi; and t \subseteq u \subseteq s] \implies M, u \models \varphi;$
- is union closed, provided [*M*,*s* ⊨ φ for all *s* ∈ *S* ≠ \emptyset] \implies *M*, \bigcup *S* ⊨ φ;
- has the *empty team property*, provided $M, \emptyset \vDash \varphi$ for all M;
- is *flat*, provided *M*,*s* ⊨ $φ \iff M$, {*w*} ⊨ φ for all *w* ∈ *s*.

It is easy to check that a formula is flat if and only if it is downwards closed and union closed, and has the empty team property. One can show by induction that, in the context of the connectives we have introduced, all \forall -free formulas (in particular, all formulas of BSML) are union closed and convex, and that all NE-free formulas are downward closed and have the empty team property (so that all formulas of ML are flat). Formulas of BSML or BSML^{\forall} clearly need not be downward closed or have the empty team property (consider $p \land NE$); formulas of BSML^{\forall} need not be union closed or convex (consider $(a \land ((p \land NE) \lor \top)) \lor (b \land ((q \land NE) \lor \top))$. Another easy induction shows: **4.2.4.** FACT (Conservativity of ML over standard semantics). For any $\alpha \in ML$:

$$M, s \models \alpha \quad \iff \quad M, \{w\} \models \alpha \text{ for all } w \in s \quad \iff \quad M, w \models \alpha \text{ for all } w \in s$$

where \models on the right is the standard single-world truth relation for ML. This implies that for $\alpha, \beta \in ML$: $\alpha \models \beta$ in the team-semantics sense iff $\alpha \models \beta$ in the usual single-world-semantics sense; we may therefore use the symbol ' \models ' to refer interchangeably to either type of entailment when discussing ML-formulas.

We write $\chi[\xi]$ to refer to a specific occurrence of the subformula ξ in χ , and $\chi[\psi/\xi]$ for the result of replacing this occurrence of ξ in χ with φ . Our Burgess theorems will build on the fact that preservation of equivalence under replacement fails in negative contexts: if $\chi[p]$ is within the scope of an odd number of \neg in χ , $\varphi \equiv \psi$ need not imply $\chi[\varphi/p] \equiv \chi[\psi/p]$. For instance, $\bot \equiv \neg \text{NE}$ but $\neg \bot \equiv \top \neq \text{NE} \equiv \neg \neg \text{NE}$, and $\neg(\varphi \lor \psi) \equiv \neg(\varphi \lor \psi)$ but $\neg \neg(\varphi \lor \psi) \equiv \varphi \lor \psi \neq \varphi \lor \psi \equiv \neg \neg(\varphi \lor \psi)$. Bi-equivalence, on the other hand, is preserved under replacement: if $\varphi \equiv^{\pm} \psi$, then $\chi[\varphi/p] \equiv^{\pm} \chi[\psi/p]$. (This implies that if we take meanings to be represented by pairs $\|\varphi\|^{\pm} = (\|\varphi\|, \|\neg \varphi\|)$)—see below for the definition of this notation—the semantics is compositional in Hodges' sense. See Section 4.1.3.) And, crucially, replacement does hold in non-negated and other positive contexts:

4.2.5. FACT (Replacement in positive contexts). If $\chi[p]$ is in the scope of an even number of \neg in χ , then $\varphi \equiv \psi$ implies $\chi[\varphi/p] \equiv \chi[\psi/p]$.

We officially reserve the term *dual negation* to refer to any negation operator that engenders failure of preservation of equivalence under replacement in negated contexts (as noted above, in BSML and BSML^{\vee} this failure occurs only in *negative* negated contexts—contexts in the scope of odd number of negations). However, for simplicity, we also occasionally use this term to refer to any negation operator with bilateral semantics (such as the negation of ML as defined above).

A *pointed model* over X is a pair (M, w) where M is a model over X and $w \in W$. We define k-bisimilarity (for $k \in \mathbb{N}$) between pointed models with respect to a set of propositional variables X in the usual way:

$$-M, w \rightleftharpoons_{0}^{\mathsf{X}} M, w : \iff \text{ for all } p \in \mathsf{X} \text{ we have } M, w \vDash p \text{ iff } M', w' \vDash p.$$

$$-M, w \rightleftharpoons_{k+1}^{\mathsf{X}} M', w' : \iff M, w \rightleftharpoons_{0} M', w' \text{ and}$$

$$- \text{ [forth] for all } v \in R[w] \text{ there is a } v' \in R'[w'] \text{ such that } M, v \rightleftharpoons_{k} M', v';$$

$$- \text{ [back] for all } v' \in R'[w'] \text{ there is a } v \in R[w] \text{ such that } M, v \rightleftharpoons_{k} M', v',$$

and we write $M, w \rightleftharpoons_k^X M', w'$ (or simply $w \rightleftharpoons^k w'$) if (M, w) and (M', w') are *k*-bisimilar with respect to X. The *modal depth md*(φ) of a formula φ is defined as usual; note that we put $md(NE) \coloneqq 0$ and $md(\varphi \lor \psi) \coloneqq max\{md(\varphi), md(\psi)\}$. We say that (M, w)and (M, w') are X,*k*-equivalent, written $M, w \equiv_k^X M', w'$ (or simply $w \equiv_k w'$), if for all $\alpha(X) \in ML$ with $md(\alpha) \leq k$: $M, w \models \alpha \iff M', w' \models \alpha$. We define the *k*-th Hintikka formula or characteristic formula $\chi_{M,w}^{X,k} \in ML$ (or simply χ_w^k) of (M, w) as follows:

$$\begin{array}{ll} \chi_{M,w}^{\mathsf{X},0} & \coloneqq & \bigwedge\{p \mid p \in \mathsf{X}, w \in V(p)\} \land \bigwedge\{\neg p \mid p \in \mathsf{X}, w \notin V(p)\};\\ \chi_{M,w}^{\mathsf{X},k+1} & \coloneqq & \chi_{w}^{k} \land \bigwedge_{v \in R[w]} \diamondsuit \chi_{v}^{k} \land \Box \bigvee_{v \in R[w]} \chi_{v}^{k}. \end{array}$$

We then have that:

4.2.6. THEOREM. (See, for instance, [24, 56].)

$$w \equiv_k w' \quad \iff \quad w \rightleftharpoons_k w' \quad \iff \quad w' \vDash \chi_w^k \quad \iff \quad \chi_w^k \equiv \chi_{w'}^k.$$

Similarly, a *pointed (team) model* over X is a pair (M, s) where M is a model over X and s is a team on M. For a given $k \in \mathbb{N}$, (M, s) and (M, s') are X, k-equivalent in a logic L, written $M, s \equiv_k^X M', s'$, if for all $\varphi(X) \in L$ with $md(\varphi) \leq k : M, s \models \varphi \iff M', s' \models \varphi$. It can be shown that $s \equiv_k^X s'$ in BSML iff $s \equiv_k^X s'$ in BSML^W, so we use $s \equiv_k^X s'$ to refer to equivalence in either of these two logics. We say that (M, s) and (M', s') are *(team)* k-bisimilar with respect to X, and write $M, s \rightleftharpoons_k^X M', s'$ (or simply $s \rightleftharpoons_k s'$) if for each $w \in s$ there is some $w' \in s'$ such that $w \rightleftharpoons_k w'$; and for each $w' \in s'$ there is some $w \in s$ such that $w \rightleftharpoons_k w'$. We define the *(weak)* k-th Hintikka formula $\chi_{M,s}^{X,s}$ (or simply χ_s^k) of (M, s) by $\chi_{M,s}^{X,k} := \bigvee_{w \in s} \chi_w^k$. We have:

4.2.7. THEOREM. [71, 7]

$$[s \equiv_k s' \iff s \rightleftharpoons_k s']$$
 and $[s' \vDash \chi_s^k \iff s' \rightleftharpoons_k t \text{ for some } t \subseteq s].$

A (world) property (over X) is a class of pointed models over X. Each formula $\alpha \in ML$ expresses a property $[\![\alpha]\!]_X$ (or simply $[\![\alpha]\!]$) over any $X \supseteq P(\alpha)$, where

$$\llbracket \alpha \rrbracket_{\mathsf{X}} := \{ (M, w) \text{ over } \mathsf{X} \mid M, w \vDash \alpha \}.$$

A (*team*) property (over X) is a class of pointed team models over X. Each formula φ expresses a property $\|\varphi\|_X$ (or simply $\|\varphi\|$) over any $X \supseteq P(\varphi)$, where

$$\|\varphi\|_{\mathsf{X}} \coloneqq \{(M,s) \text{ over } \mathsf{X} \mid M, s \vDash \varphi\}$$

The ground team of a team property \mathcal{P} , denoted $\bigcup \mathcal{P}$, is the world property

$$\bigcup \mathcal{P} \coloneqq \{(M,w) \mid \exists (M,s) \in \mathcal{P} : w \in s\},\$$

and the *ground team* of a formula φ (over X), denoted $|\varphi|_X$, is defined by $|\varphi|_X := \bigcup \|\varphi\|_X$, so that

$$|\varphi|_{\mathsf{X}} = \{(M, w) \mid \exists (M, s) \in \|\varphi\|_{\mathsf{X}} : w \in s\}.$$

(Note that $\bigcup \mathcal{P}$ is essentially a team modulo bisimulation¹¹, whence the name.)¹² Observe that for classical α , $[\alpha] = |\alpha|$.

We say that a logic *L* is *expressively complete* for a class of (team) properties \mathbb{P} , written $||L|| = \mathbb{P}$, if for each finite X, the class \mathbb{P}_X of properties over X in \mathbb{P} is precisely the class of properties over X expressible by formulas of *L*, that is, if

$$||L||_{\mathsf{X}} = \mathbb{P}_{\mathsf{X}}$$
, where $||L||_{\mathsf{X}} := \{ ||\varphi||_{\mathsf{X}} \mid \varphi \in L \}$.

The definition of closure properties (4.2.3) is extended to team properties in the obvious way; for instance, a property \mathcal{P} is *union closed* just in case $[(M,s) \in \mathcal{P} \text{ for all } s \in S \neq \emptyset]$ implies $(M, \bigcup S) \in \mathcal{P}$. We additionally say that \mathcal{P} is *invariant under* X, *k*-*bisimulation* $(k \in \mathbb{N})$ if $[(M,s) \in \mathcal{P} \text{ and } M, s \rightleftharpoons_k^X M', s']$ implies $(M',s') \in \mathcal{P}$, and that \mathcal{P} over X is *invariant under bounded bisimulation* if \mathcal{P} is invariant under X, *k*-bisimulation for some $k \in \mathbb{N}$.

4.2.8. THEOREM (Expressive completeness of BSML and BSML^{\vee}). [16, 7]

¹¹ $\cup \mathcal{P}$ is bisimilar to the pointed model $(\bigcup \{M' \mid (M', w') \in \cup \mathcal{P}\}, \{w' \mid (M', w') \in \cup \mathcal{P}\})$, where \uplus denotes disjoint union, in the sense that for each $(M, w) \in \cup \mathcal{P}$ there is a $w' \in \{w' \mid (M', w') \in \cup \mathcal{P}\}$ such that (M, w) is bisimilar to $(\bigcup \{M' \mid (M', w') \in \cup \mathcal{P}\}, w')$, and vice versa.

¹²While the other notions in Section 4.2.1 are from the literature, the term 'ground team' is introduced in this paper. The notion itself is, however, essentially identical or very similar to a number of other notions in the literature. In this endnote we list some of these notions to stave off confusion, to help the reader make connections, and also to highlight the fact that ground teams (and hence potentially also the incompatibility notions making use of ground teams—see Sections 4.2.2 and 4.3) can (at least in some settings) be given meaningful, interesting interpretations. (We will discuss some possible interpretations of ground teams and the incompatibility notions in more detail in Section 4.4.)

Hodges [78] defines (a first-order analogue) of the following notion: the *flattening operator* \downarrow is such that $\|\downarrow \varphi\| = \{(M,t) \mid \forall w \in t : \exists (M,s) \in \|\varphi\| : w \in s\}$. In other words, the flattening $\downarrow \varphi$ of φ expresses the property consisting of teams consisting of elements of $|\varphi|$, the ground team of φ . This is essentially the power set of the ground team—it is the flat team property corresponding to the world property $|\varphi|$. Hodges' flattening is similar to but distinct from Väänänen's [119] *flattening* φ^f of φ (see Sections 4.3.3 and 4.3.4)— φ^f is, roughly, φ with each nonclassical subformula ψ replaced with a classical subformula α such that $\psi \models \alpha$ and $|\psi| = |\alpha|$.

In inquisitive semantics/logic (see [41, 42, 38, 35], as well as Sections 4.3.1 and 4.4.4; we consider here only the propositional setting for simplicity), the union $\bigcup \mathcal{P}$ of a property \mathcal{P} is referred to and represents the *informative content* of the (inquisitive) proposition represented by \mathcal{P} (in the propositional setting—see Section 4.3—we define $\bigcup \mathcal{P} := \bigcup \mathcal{P}$, so $\bigcup \mathcal{P}$ is the same notion as the ground team of \mathcal{P}). Similarly, the informative content of φ is $\bigcup \|\varphi\|$. There is an operator on propositions ! such that $!\mathcal{P} = \wp(\bigcup \mathcal{P})$ and a corresponding connective ! such that $\|!\varphi\| = \wp(\bigcup \|\varphi\|)$, and hence ! φ is equivalent to the Hodges' flattening $\downarrow \varphi$ of φ . Applied to \mathcal{P}/φ , the operator/connective yields a proposition/formula that has the same informative content as \mathcal{P}/φ , but with no *inquisitive content*. The set of worlds w such that $\{w\} \models \varphi$ is referred to as the *truth-set* of φ (as in BSML, the fundamental semantic notion, defined with respect to teams, is *support*; truth is defined as support with respect to singleton teams). Due to the downward closure of inquisitive logic, the truth-set of φ is equal to its ground team/informative content. The truth-set/informative content of φ can also represent the information that φ presupposes, if φ expresses a question. Inquisitive logic also features a notion analogous to Väänänen's flattening of φ ; this is called the *classical variant* of φ .

- (i) BSML is expressively complete for the class $\{\mathcal{P} \mid \mathcal{P} \text{ is convex, union closed, and invariant under bounded bisimulation}\}$.
- (ii) BSML^{\vee} is expressively complete for the class { $\mathcal{P} \mid \mathcal{P}$ is invariant under bounded *bisimulation*}.

4.2.2 Bicompleteness and Incompatibility

In this section, we reformulate Burgess' theorem as an expressive completeness result for pairs satisfying a specific notion of incompatibility, or pair property. We then distinguish between multiple different notions of incompatibility, examine how they are related, and show that some of them are not sufficiently strong to yield Burgess theorems for BSML and BSML^V due to the failure of downward closure and the empty team property in these logics.

Burgess [29] formulates his result as follows: if two sentences φ and ψ are incompatible in that they share no models, then there is a sentence θ such that $\varphi \equiv \theta$ and $\psi \equiv \neg \theta$. Kontinen and Väänänen [94] (working in D and IF, whose formulas are downward closed and have the empty team property), make use of an equivalent notion of incompatibility, generalized to make it applicable to arbitrary formulas (that may contain free variables): the formulas φ and ψ are incompatible if, for any given model and (first-order) team on that model, if φ and ψ are both true in the team, the team must be empty.¹³ Let us say that two formulas which are incompatible in this way are \perp -*incompatible*. In our modal setting:

4.2.9. DEFINITION (\perp -incompatibility). φ_0 and φ_1 are \perp -incompatible (\perp -*I*) if [$s \models \varphi_0$ and $s \models \varphi_1$] implies $s = \emptyset$ (equivalently, if $\varphi_0, \varphi_1 \models \perp$).

Table 4.1 lists all the pair properties/incompatibility notions we consider in this paper, formulated for the simpler propositional setting (see Section 4.3). Figure 4.2 displays some of the implications between the incompatibility notions.

We noted in Section 4.1 that a sentence (of D or IF) and its dual negation share no models. The analogous result also holds for arbitrary formulas: any formula φ and its dual negation $\neg \varphi$ are \perp -incompatible: $\varphi, \neg \varphi \models \bot$. Observe that this fact immediately yields the converse of Kontinen and Väänänen's Burgess theorem: for any formulas φ and ψ , if there is a formula θ such that $\varphi \equiv \theta$ and $\psi \equiv \neg \theta$, then since θ and $\neg \theta$ are \perp -incompatible, φ and ψ must also be \perp -incompatible.

We may view a Burgess theorem, together with its converse, as a type of expressive completeness result. In our modal setting, we let $\|\varphi\|_X^{\pm,\neg}$ (or simply $\|\varphi\|_X^{\pm}$) denote the pair $(\|\varphi\|_X, \|\neg \varphi\|_X)$. A *pair property* \mathscr{P} over X is a class of pairs of team properties $(\mathcal{P}, \mathcal{Q})$, where \mathcal{P} and \mathcal{Q} are over X. We say that a logic L is *bicomplete* for a pair

¹³For sentences, Kontinen and Väänänen's notion is equivalent to Burgess'. This follows from the fact that a sentence, in first-order team semantics, is defined to be true in a model just in case it is true in all first-order teams on the model, and this is equivalent to it being true in some nonempty team on the model.

| Pair property | Definition(s) | Bicomplete logics | |
|--|---|---|--|
| \perp -incompatible (\perp -I) | • $[s \vDash \varphi_0 \text{ and } s \vDash \varphi_1] \implies s = \emptyset$ | D, IF | |
| | • $\varphi_0, \varphi_1 \vDash \bot$ | | |
| | • NE-incompatible or ⊥-incompatible | | |
| Ground-incompatible (G-I) | • $[s \vDash \varphi_0 \text{ and } t \vDash \varphi_1] \implies s \cap t = \emptyset$ | D, IF, BSML, | |
| | • $ \boldsymbol{\varphi}_0 \cap \boldsymbol{\varphi}_1 = \varnothing$ | $BSML^{\mathbb{V}}, PL(NE, \mathbb{V})$ | |
| ⊥-incompatible (⊥-I) | • φ_0 and φ_1 never jointly true | | |
| | • $\pmb{\varphi}_0, \pmb{\varphi}_1 \vDash \mathbb{I}$ | | |
| | • $\ \boldsymbol{\varphi}_0\ \cap \ \boldsymbol{\varphi}_1\ = \boldsymbol{\varnothing} = \ \boldsymbol{\bot}\ $ | | |
| NE-incompatible (NE-I) | • $[s \vDash \varphi_0 \text{ and } s \vDash \varphi_1] \iff s = \emptyset$ | D, IF | |
| | • $\varphi_0, \varphi_1 \vDash \bot$ and $\varphi_0, \varphi_1 \nvDash \bot$ | | |
| | • $\ \boldsymbol{\varphi}_0\ \cap \ \boldsymbol{\varphi}_1\ = \{\varnothing\} = \ \bot\ $ | | |
| World-incompatible (W-I) | • $\{w\} \vDash \varphi_0 \iff \{w\} \nvDash \varphi_1$ | $PL(=(\cdot)), PL$ | |
| Team-incompatible (T-I) | • $s \vDash \varphi_i \iff s \nvDash \varphi_{1-i}$ | $PL(\sim)$ (w.r.t. \sim) | |
| | • $\ \boldsymbol{\varphi}_i\ = \ \top\ \setminus \ \boldsymbol{\varphi}_{1-i}\ $ | | |
| Flat-incompatible (F-I) | • world-incompatible and φ_0, φ_1 flat | PL | |
| | $\bullet \ \boldsymbol{\varphi}_i \ = \wp(\top \smallsetminus \boldsymbol{\varphi}_{1-i})$ | | |
| | • $\ \boldsymbol{\varphi}_i\ = \{s \mid t \models \boldsymbol{\varphi}_{1-i} \implies s \cap t = \emptyset\}$ | | |
| | • $\ \boldsymbol{\varphi}_i\ = \bigcup \{ \mathcal{P} \subseteq \ T\ \mid \mathcal{P}, \ \boldsymbol{\varphi}_{1-i}\ \text{ G-I} \}$ | | |
| φ_1 down-set incompatibility | • $s \vDash \varphi_1 \iff$ | InqB (w.r.t. \neg_i), | |
| (D-I) of φ_0 | $\left[\left[t \vDash \varphi_0 \text{ and } t \subseteq s\right] \implies t = \emptyset\right]$ | PL | |
| Down-set incompatibilities | • φ_1 D-I of φ_0 or φ_0 D-I of φ_1 | HS, PL | |
| (on either side) (E-D-I) | | | |
| Ground-complementary | • $ \boldsymbol{\varphi}_i = \top \setminus \boldsymbol{\varphi}_{1-i} $ | $PL(=(\cdot)), PL$ | |
| (G-C) | | | |
| Ground-complementary | • $ \varphi_i = \top \setminus \varphi_{1-i} $ or $\varphi_0 \equiv \bot$ or $\varphi_1 \equiv \bot$ | $PL(NE), PL(=(\overline{\cdot})),$ | |
| $\underline{\text{modulo}} \perp (\text{G-C mod} \perp)$ | | PL | |
| All pairs | | $PL(NE^*, W)$ | |

Table 4.1: Pair properties/incompatibility notions, formulated for the propositional setting. Each bulleted (\bullet) item is a definition for the relevant notion; all definitions for a given notion are equivalent.

property \mathscr{P} (with respect to \neg), written $||L||^{\pm, \neg} = \mathscr{P}$ (or simply $||L||^{\pm} = \mathscr{P}$), if for all finite $X \subseteq Prop$,

$$\|L\|_{\mathsf{X}}^{\pm,\neg} = \mathscr{P}_{\mathsf{X}}, \text{ where}$$
$$\|L\|_{\mathsf{X}}^{\pm,\neg} \coloneqq \{(\|\varphi\|_{\mathsf{X}}^{\pm,\neg} \mid \varphi \in L\} \text{ and } \mathscr{P}_{\mathsf{X}} \coloneqq \{(\mathcal{P}, \mathcal{Q}) \in \mathscr{P} \mid \mathcal{P}, \mathcal{Q} \text{ over } \mathsf{X}\},\$$

and we say it is *bicomplete modulo expressive power* (or simply bicomplete, written in monospace font) for \mathscr{P} (with respect to \neg) if it is bicomplete for $\mathscr{P}_L := \mathscr{P} \cap \{(\|\varphi\|, \|\psi\|) \mid \varphi, \psi \in L\}$ (with respect to \neg). A Burgess theorem for an incompatibility of the product of the



Figure 4.2: Some implications among the incompatibility notions. Black arrows indicate implications which always hold. Dashed arrows in other colors indicate implications conditional on closure properties (e.g., DC = downward closure, ET = empty team property, etc.). Implications which trivialize the notions have been omitted; for instance, in a setting with the empty team property, no pair can be team-incompatible, so team-incompatibility with the empty team property implies all other notions. Observe also that if a pair φ_0, φ_1 is NE-incompatible, then both formulas have the empty team property. Similarly flat-incompatibility forces the flatness of both φ_0 and φ_1 , while D-I forces the downward closure and the empty team property of φ_1 . Each of team-incompatibility and $\mathbb{1}$ -incompatibility is inconsistent with both formulas having the empty team property.

ity notion (pair property) \mathscr{P} yields an inclusion (abusing our notation) $\mathscr{P}_L \subseteq ||L||^{\pm}$, and its converse yields the converse inclusion $||L||^{\pm} \subseteq \mathscr{P}_L$; together, then, they constitute a bicompleteness theorem. Say that L_1 and L_2 are *(expressively) bi-equivalent* or *strongly equivalent* (with respect to \neg_1 and \neg_2) if $||L_1||_X^{\pm, \neg_1} = ||L_2||_X^{\pm, \neg_2}$ for all finite $X \subseteq Prop.$

We extend the definition of \perp -incompatibility to pairs of properties in the obvious way: \mathcal{P} and \mathcal{Q} are \perp -*incompatible* if $(M,s) \in \mathcal{P} \cap \mathcal{Q}$ implies $s = \emptyset$. The first-order analogues of these notions (formulated for all formulas rather than only sentences) give us the following reformulation of Kontinen and Väänänen's result:

4.2.10. THEOREM (Bicompleteness of D and IF). [94] Each of the logics D and IF is bicomplete for \perp -incompatible pairs.

Let us now consider the logics BSML and $BSML^{\vee}$. While it does hold, in these logics, that a formula and its dual negation are \perp -incompatible, \perp -incompatibility does not yield a Burgess theorem or bicompleteness theorem for either of these logics. To show why, we make use of the following proposition:

4.2.11. PROPOSITION. If $s \models \varphi$ and $t \models \neg \varphi$, then $s \cap t = \emptyset$.

Proof:

By induction on the complexity of φ , which we may assume to be in negation normal form. We only show some cases; for the rest, see [12, Proposition 3.3.9].

4.2. Burgess Theorems for BSML and $BSML^{\vee}$

If $\varphi = \psi \lor \chi$, we have $s = s_1 \lor s_2$ where $s_1 \vDash \psi$ and $s_2 \vDash \chi$, and $t \dashv \psi \lor \chi$, whence $t \dashv \psi$ and $t \dashv \chi$, and therefore $t \vDash \neg \psi$ and $t \vDash \neg \chi$. By the induction hypothesis, $s_1 \cap t = s_1 \cap t = \emptyset$, so that also $s \cap t = (s_1 \cup s_2) \cap t = (s_1 \cap t) \cup (s_2 \cap t) = \emptyset$.

If $\varphi = \Diamond \psi$, assume for contradiction that $s \cap t \neq \emptyset$. Let $w \in s \cap t$. By $s \models \Diamond \varphi$, there is a nonempty $t \subseteq R[w]$ such that $t \models \psi$. By $t \models \neg \Diamond \psi$, we have $t \Rightarrow \Diamond \psi$ whence $R[w] \Rightarrow \psi$ so that also $R[w] \models \neg \psi$. But then by the induction hypothesis, $t = t \cap R[w] = \emptyset$, contradicting the fact that *t* is nonempty.

Now let $\varphi := p$ and $\psi := ((p \land NE) \lor (\neg p \land NE))$. Then $\varphi, \psi \models \bot \models \bot$, so φ and ψ are \bot -incompatible. A Burgess theorem employing \bot -incompatibility would then produce a θ such that $\varphi \equiv \theta$ and $\psi \equiv \neg \theta$. But consider the teams $\{w_p\}$ and $\{w_p, w_{\neg p}\}$ (where $w_p \models p$, etc.). We have $\{w_p\} \models \varphi$, so $\{w_p\} \models \theta$; and $\{w_p, w_{\neg p}\} \models \psi$, so $\{w_p, w_{\neg p}\} \models \neg \theta$. Therefore, by Proposition 4.2.11, $\{w_p\} \cap \{w_p, w_{\neg p}\} = \{w_p\} = \emptyset$, a contradiction.

Drawing on Proposition 4.2.11, we can define an incompatibility notion more appropriate for BSML and $BSML^{\vee}$:

4.2.12. DEFINITION (Ground-incompatibility). φ_0 and φ_1 are *ground-incompatible* (*G*-*I*) if $[s \models \varphi_0 \text{ and } t \models \varphi_1]$ implies $s \cap t = \emptyset$ (equivalently, if $|\varphi_0|_X \cap |\varphi_1|_X = \emptyset = |\bot|_X$ for any $X \supseteq P(\varphi_0) \cup P(\varphi_1)$).

We will show in Section 4.2.3 that ground-incompatibility does yield Burgess theorems for BSML and $BSML^{\vee}$. It is easy to see that ground-incompatibility is strictly stronger than \perp -incompatibility:

4.2.13. FACT. If φ and ψ are ground-incompatible, then they are \perp -incompatible, and the converse implication does not hold in general.

Proof:

Given ground-incompatible φ and ψ , if there is some *s* such that $s \models \varphi$ and $s \models \psi$, this implies $s = \emptyset$ whence $\varphi, \psi \models \bot$. Otherwise, if there is no such *s*, then $\varphi, \psi \models \bot \models \bot$. Either way, $\varphi, \psi \models \bot$.

We have already seen that the converse does not hold (in the example directly below Proposition 4.2.11). \Box

The analogous fact clearly also holds in the first-order setting. This means that the Burgess theorems for D and IF also hold with respect to ground-incompatibility. In general, if a Burgess theorem holds with respect to a given incompatibility notion—a given pair property—it also holds with respect to all stronger notions: if $\mathcal{P}_1 \subseteq \mathcal{P}_2$, then $\mathcal{P}_{2L} \subseteq ||L||^{\pm}$ implies $\mathcal{P}_{1L} \subseteq ||L||^{\pm}$. In a downward-closed setting (such as that of D and IF), ground-incompatibility is in fact equivalent to \bot -incompatibility:

4.2.14. FACT. For downward-closed φ and ψ : φ and ψ are ground-incompatible iff they are \perp -incompatible.

Proof:

Given downward-closed and \perp -incompatible φ and ψ , if $s \models \varphi$ and $t \models \psi$, then by downward closure, $s \cap t \models \varphi \land \psi$, whence $s \cap t \models \varphi$.

This, in turn, means that the converses of the Burgess theorems for D and IF also hold with respect to ground-incompatibility, and hence that D and IF are also bicomplete for ground-incompatible pairs. In general we clearly have that if the converse of a Burgess theorem holds with respect to a given notion, it also holds with respect to all weaker notions: if $\mathscr{P}_1 \subseteq \mathscr{P}_2$, then $\|L\|^{\pm} \subseteq \mathscr{P}_{1L}$ implies $\|L\|^{\pm} \subseteq \mathscr{P}_{2L}$.

So \perp -incompatibility and ground-incompatibility are equivalent in a downwardclosed setting, but come apart when downward closure fails. Are there any natural interpretations of \perp -/ground-incompatibility that would also yield some intuitive understanding of this fact? In Section 4.4, we provide rough sketches of some possible sets of interpretations. It will be helpful, for the discussion in Section 4.4, to further dissect \perp -incompatibility into two more natural notions of incompatibility; this we do now.

4.2.15. DEFINITION (*L*-incompatibility and NE-incompatibility).

- (i) φ_0 and φ_1 are $\mathbb{1}$ -*incompatible* ($\mathbb{1}$ -*I*) if there is no team (including the empty team) that makes both φ_0 and φ_1 true (equivalently, if $\varphi_0, \varphi_1 \models \mathbb{1}$; equivalently, if $\|\varphi_0\|_X \cap \|\varphi_1\|_X = \emptyset = \|\mathbb{1}\|_X$ for any $X \supseteq P(\varphi_0) \cup P(\varphi_1)$).
- (ii) Incompatible in non-empty teams or NE-incompatible (NE-I) if $[[s \models \varphi_0 \text{ and } s \models \varphi_1]$ if and only if $s = \emptyset$] (equivalently, if $\varphi_0, \varphi_1 \models \bot$ and $\varphi_0, \varphi_1 \not\models \bot$; equivalently, if $\|\varphi_0\|_X \cap \|\varphi_1\|_X = \|\bot\|_X$ for any $X \supseteq P(\varphi_0) \cup P(\varphi_1)$).

Clearly φ and ψ are \perp -incompatible if and only if they are \parallel -incompatible or they are NE-incompatible, and they are NE-incompatible if and only if they are \perp -incompatible and each of them has the empty team property. Therefore, in a setting with both downward closure and the empty team property such as that of D/IF, ground-incompatibility, \perp -incompatibility and NE-incompatibility are all equivalent (and so D and IF are also bicomplete for NE-incompatible pairs). We further have:

4.2.16. FACT. For convex φ and ψ , if φ and ψ are NE-incompatible, then they are ground-incompatible. NE-incompatibility does not imply ground-incompatibility in general.

Proof:

If φ and ψ are NE-incompatible, then each has the empty team property. The empty team property together with convexity implies downward closure, so the first claim now follows by Fact 4.2.14. For the second claim, consider the formulas $\neg p \lor ((p \land NE) \lor (\neg p \land NE))$ and p.

In particular, NE-incompatibility implies ground-incompatibility in BSML. The Burgess theorem with respect to ground-incompatibility in Section 4.2.3 will, therefore, also imply a Burgess theorem with respect to NE-incompatibility for BSML. The converse of this theorem clearly does not hold, since, for instance, for $\varphi := NE$, φ and $\neg \phi$ are not NE-incompatible.

4.2.3 The Theorems

In this section, we prove the Burgess- and bicompleteness theorems for BSML and $BSML^{\mathbb{V}}$.

Burgess' original result is essentially a corollary of Craig's interpolation for classical first-order logic. Similarly, our results follow from interpolation for classical modal logic ML (that is, interpolation for the smallest normal modal logic K—as with the other logics in this paper, we use 'ML' to refer both to the language as well as the logic consisting of all ML-formulas which are valid in all Kripke frames). This allows us to prove the following variant of interpolation for ground-incompatible formulas (c.f. the similar results for IF [75, pp. 59–60] and D [119, Theorem 6.7]; and note that restricted to ML-formulas, the following is equivalent to interpolation for ML/K).

4.2.17. THEOREM (Separation theorem). If φ and ψ are ground-incompatible, then there is a classical formula $\gamma \in ML$ such that $P(\gamma) \subseteq P(\phi) \cap P(\psi)$, $\phi \models \gamma$, and $\psi \models \neg \gamma$.

Proof:

Take $\alpha := \bigvee_{u \in \|\varphi\|_{P(\varphi)}} \chi_u^{P(\varphi), md(\varphi)}$. We show that $\varphi \models \alpha$. If there are no (M, t) such that $M, t \models \varphi$, we have $\alpha = \bot$ and $\varphi \equiv \bot \models \bot$. Otherwise let $M, t \models \varphi$, where (M, t) is over $X \supseteq P(\varphi)$. This implies $M', t \models \varphi$, where M' is the restriction of M to $P(\varphi)$, i.e., $(M', t) \in \|\varphi\|_{P(\varphi)}$. Clearly $M, t \rightleftharpoons_{P(\varphi)}^{md(\varphi)} M', t$ so $M, t \models \chi_{M', t}^{P(\varphi), md(\varphi)}$ by Theorem 4.2.7. Then by the empty team property of ML, $M, t \models \alpha$. Similarly, for $\beta := \bigvee_{u \in ||\psi||} \chi_u^{\mathsf{P}(\psi), md(\psi)}$, we have $\psi \models \beta$.

We now show that α and β are ground-incompatible, so let $M, s \models \alpha$ and $M, t \models \beta$, where M is over $X \supseteq P(\varphi) \cup P(\psi)$, and assume for contradiction that $w \in s \cap t$. This implies that there are $(M_{\varphi}, u_{\varphi}) \in \|\varphi\|_{\mathsf{P}(\varphi)}$ and $w_{\varphi} \in u_{\varphi}$ such that $w \models \chi_{w_{\varphi}}^{\mathsf{P}(\varphi), md(\varphi)}$; and $(M_{\psi}, u_{\psi}) \in \|\psi\|_{\mathsf{P}(\psi)}$ and $w_{\psi} \in u_{\psi}$ such that $w \models \chi^{\mathsf{P}(\psi), md(\psi)}_{w_{\psi}}$. By Theorem 4.2.6, $w \rightleftharpoons_{md(\phi)}^{\mathsf{P}(\phi)} w_{\phi}$ and $w \rightleftharpoons_{md(\psi)}^{\mathsf{P}(\psi)} w_{\psi}$. Let (M_{ϕ}^*, u_{ϕ}^*) be a pointed model over X defined by taking $(M_{\varphi}, u_{\varphi})$ and assigning arbitrary valuations for all $p \in X \setminus P(\varphi)$, and similarly for (M_{ψ}^*, u_{ψ}^*) . Let M' be the disjoint union $M \uplus M_{\varphi}^* \uplus M_{\psi}^*$, and let $u'_{\varphi} := (u_{\varphi}^* \setminus \{w_{\varphi}^*\}) \cup$ $\{w\} \text{ and } u'_{\psi} \coloneqq (u^*_{\psi} \setminus \{w^*_{\psi}\}) \cup \{w\}. \text{ Clearly } M', u'_{\varphi} \rightleftharpoons_{md(\varphi)}^{\mathsf{P}(\varphi)} M_{\varphi}, u_{\varphi} \text{ and } M', u'_{\psi} \rightleftharpoons_{md(\psi)}^{\mathsf{P}(\psi)} M_{\psi}, u_{\psi}, \text{ whence } u'_{\varphi} \vDash \varphi \text{ and } u'_{\psi} \vDash \psi \text{ by Theorem 4.2.7. But we have } w \in u'_{\varphi} \cap u'_{\psi},$ contradicting the assumption that φ and ψ are ground-incompatible.

We have shown $|\alpha|_X \cap |\beta|_X = \emptyset$; therefore also $[\alpha]_X \cap [\beta]_X = \emptyset$, and so we have $\alpha \models \neg \beta$. By interpolation for ML, there is an ML-formula γ such that $\alpha \models \gamma \models \neg \beta$ and

$$\mathsf{P}(\gamma) = \mathsf{P}(\alpha) \cap \mathsf{P}(\beta) = \mathsf{P}(\phi) \cap \mathsf{P}(\psi). \text{ Then also } \phi \vDash \alpha \vDash \gamma \text{ and } \psi \vDash \beta \vDash \neg \gamma. \square$$

The modal complications in the proof above might distract from the structure of the argument. In the analogous proof for the propositional setting (see Section 4.3 for details on this setting), to show $|\alpha|_X \cap |\beta|_X = \emptyset$, one simply notes that $|\varphi|_{P(\varphi)} = |\alpha|_{P(\varphi)}$ and $|\psi|_{P(\psi)} = |\beta|_{P(\psi)}$, which implies $|\varphi|_X = |\alpha|_X$ and $|\psi|_X = |\beta|_X$, whence $|\alpha|_X \cap |\beta|_X = \emptyset$. (In the modal case, we have $|\varphi|_{P(\varphi)} \subseteq |\alpha|_{P(\varphi)}$, but $|\varphi|_{P(\varphi)} \supseteq |\alpha|_{P(\varphi)}$ might only hold modulo bisimulation.)

We require one more simple lemma.

4.2.18. LEMMA. For any formula φ , there is a formula φ' such that $\varphi' \equiv \varphi$, and $\neg \varphi'$ has the empty team property.

Proof:

Define φ' by putting φ in negation normal form and then replacing each occurrence of \neg NE with \bot . An easy induction shows that φ' is as required.

We can now prove the Burgess theorems for BSML and BSML^{\vee}. We give one version of the proof which is completely analogous to Burgess' and which works for both logics, and then provide an alternative proof for BSML^{\vee}. The alternative proof does not make essential use of modal operators, and it is easy to see that its propositional analogue establishes the analogous result for the propositional fragment of BSML^{\vee}. The propositional analogue of this result does not hold for the propositional fragment of BSML; see Section 4.3.3. We show both the Burgess theorems as well as their converses.

4.2.19. THEOREM (Burgess theorems for BSML and BSML^{\vee}). In both of the logics BSML and BSML^{\vee}, the following are equivalent:

- (i) φ and ψ are ground-incompatible.
- (*ii*) There is a formula θ such that $\varphi \equiv \theta$ and $\psi \equiv \neg \theta$ (and $P(\theta) = P(\varphi) \cup P(\psi)$).

Proof:

(ii) \implies (i) by Proposition 4.2.11. (i) \implies (ii): Let $\theta_0 \coloneqq \diamondsuit (\square \lor \neg \square)$ so that $\theta_0 \equiv \diamondsuit \amalg \equiv \square$ and $\neg \theta_0 \equiv \square \neg (\square \lor \neg \blacksquare) \equiv \square (\neg \square \land \blacksquare) \equiv \square \amalg \equiv \bot$. By Lemma 4.2.18, let φ' be such that $\varphi \equiv \varphi'$ and $\neg \varphi'$ has the empty team property, and similarly for ψ' . Let $\varphi_0 \coloneqq \varphi' \lor \theta_0$ and $\psi_0 \coloneqq \psi' \lor \theta_0$ so that $\varphi_0 \equiv \varphi$ and $\neg \varphi_0 \equiv \bot$ (clearly $\neg \varphi_0 \equiv \neg \varphi' \land \neg \theta_0 \models \bot$, and by the empty team property of $\neg \varphi'$, we also have $\bot \models \neg \varphi' \land \neg \theta_0$), and similarly for ψ_0 . By Theorem 4.2.17, let $\gamma \in ML$ be such that $\varphi_0 \models \gamma$ and $\psi_0 \models \neg \gamma$. Finally, let $\theta \coloneqq (\varphi_0 \land \gamma) \lor \neg \psi_0$. Then:

Alternatively, for $\varphi, \psi \in \text{BSML}^{\vee}$, by Theorem 4.2.17 we can let γ be such that $\varphi \models \gamma$ and $\psi \models \neg \gamma$. Let $\varphi_{\perp} \coloneqq \neg \varphi \lor \perp$ and $\psi_{\perp} \coloneqq \neg \psi \lor \perp$ so that $\varphi_{\perp} \equiv \perp$ and $\neg \varphi_{\perp} \equiv \varphi$, and similarly for ψ_{\perp} . Finally, let $\theta \coloneqq \neg (\varphi_{\perp} \lor \neg (\psi_{\perp} \lor \gamma))$. Then:

Burgess' proof (as well as those of Dechesne, Kontinen and Väänänen, and Mann) likewise makes use of a formula θ_0 such that $\theta_0 \equiv \perp \equiv \neg \theta_0$; let us briefly discuss the sentence used by Burgess (originally due to Väänänen) to allow for easy comparison with our θ_0 . Formulated in IF, this sentence is $\theta_0 := \forall x (\exists y/\forall x) (x = y)$. This is equivalent to the Σ_1^1 -sentence $\exists f \forall x (x = f(x))$, and its dual negation is equivalent to the FO-sentence $\exists x \forall y (x \neq y)$. Clearly, then, θ_0 is true in a model just in case the domain of the model has less than two elements; and $\neg \theta_0$ is never true. Therefore, ignoring empty models and models of size one, we have $\theta_0 \equiv \bot \equiv \neg \theta_0$.

We extend the definition of ground-incompatibility to pairs of properties in the obvious way: \mathcal{P} and \mathcal{Q} are ground-incompatible if $\bigcup P \cap \bigcup \mathcal{Q} = \emptyset$. By Theorems 4.2.8 and 4.2.19,

4.2.20. COROLLARY (Bicompleteness of BSML and $BSML^{\vee}$).

- *(i)* BSML *is bicomplete for* {(\mathcal{P}, \mathcal{Q}) | \mathcal{P}, \mathcal{Q} *are convex, union closed, and invariant under bounded bisimulation;* \mathcal{P}, \mathcal{Q} *are ground-incompatible*}.
- (*ii*) BSML^{\vee} *is bicomplete for* {(\mathcal{P}, \mathcal{Q}) | \mathcal{P}, \mathcal{Q} *are invariant under bounded bisimulation;* \mathcal{P}, \mathcal{Q} *are ground-incompatible*}.
- (iii) Each of BSML and $BSML^{\vee}$ is bicomplete for ground-incompatible pairs.

Note that a Burgess theorem can be viewed as a kind of strengthening of the relevant Separation theorem (4.2.17): given ground-incompatible φ and ψ , there is a θ such that not only $\varphi \models \theta$ and $\psi \models \neg \theta$, but also $\theta \models \varphi$ and $\neg \theta \models \psi$.¹⁴ However, to get also these converse entailments, one must go from $P(\theta) \subseteq P(\varphi) \cap P(\psi)$ to $P(\theta) = P(\varphi) \cup P(\psi)$.

4.3 **Burgess Theorems for Propositional Team Logics**

In this section, we prove Burgess/bicompleteness theorems for Hawke and Steinert-Threlkeld's semantic expressivist logic for epistemic modals HS (Section 4.3.2); for PL(NE) and PL(NE, W), the propositional fragments of BSML and $BSML^{V}$, respectively (Section 4.3.3); as well as for propositional dependence logic $PL(=(\cdot))$ (Section 4.3.4). We also comment (Section 4.3.1) on the bicompleteness of team logics with

¹⁴I am grateful to Maria Aloni for pointing this out to me.

negations that do not exhibit failure of replacement: team-based classical propositional logic PL, propositional inquisitive logic InqB, and the extension $PL(\sim)$ of PL with the Boolean negation ~ (also known as *propositional team logic*—see endnote 5), and give an example (in Section 4.3.3) of a logic bicomplete for all pairs.

4.3.1 **Bicompleteness Without Failure of Replacement**

The notion of bicompleteness can also be applied to logics in which replacement of equivalents under negation does not fail. We briefly consider three such logics by way of illustration.

4.3.1. DEFINITION (Syntax of PL, PL(~), and InqB). The set of formulas of *classical propositional logic* PL is generated by:

$$\alpha ::= p \mid \perp \mid \neg \alpha \mid (\alpha \land \alpha) \mid (\alpha \lor \alpha)$$

where $p \in \mathsf{Prop}$, and Prop is a countable set of propositional variables, as before.

The set of formulas of *propositional logic with the Boolean negation* $PL(\sim)$ is generated by:

$$\boldsymbol{\varphi} \coloneqq p \mid \perp \mid \neg \alpha \mid (\boldsymbol{\varphi} \land \boldsymbol{\varphi}) \mid (\boldsymbol{\varphi} \lor \boldsymbol{\varphi}) \mid \sim \boldsymbol{\varphi}$$

where $p \in \text{Prop}$ and $\alpha \in \text{PL}$.

The set of formulas of *propositional inquisitive logic* InqB is generated by:

$$\boldsymbol{\varphi} \coloneqq p \mid \perp \mid (\boldsymbol{\varphi} \land \boldsymbol{\varphi}) \mid (\boldsymbol{\varphi} \to \boldsymbol{\varphi}) \mid (\boldsymbol{\varphi} \lor \boldsymbol{\varphi})$$

where $p \in \mathsf{Prop}$.

We use α, β to refer exclusively to formulas of PL as well as the \forall -free fragment of InqB (*classical formulas*). In InqB, let $\neg_i \varphi := \varphi \rightarrow \bot$ and $\forall \forall \varphi := \bot$. As before, let $\top := \neg_{\perp}$ or $\top := \neg_i \top$, and $\forall \varphi := \bot$.

A (propositional) *team* is a set of valuations $s \subseteq 2^X$, where $X \subseteq Prop$. We formulate the semantics of PL and PL(~) in terms of support/anti-support conditions as in BSML. The conditions for the atoms and connectives of PL are the obvious propositional analogues of their modal support/anti-support conditions. The support condition for the Boolean negation is given by

$$s \models \sim \varphi : \iff s \not\models \varphi.$$

The Boolean negation is not permitted to occur in the scope of the dual negation, so it does not have anti-support conditions.¹⁵. We only formulate support conditions for InqB; the support conditions for \rightarrow , the *intuitionistic implication*, are as follows:

$$s \vDash \varphi \to \psi : \iff \forall t \subseteq s : [t \vDash \varphi \implies t \vDash \psi]$$

¹⁵One could of course formulate anti-support conditions for ~ (i.e., truth/support conditions for $\neg \sim \phi$), but it is not clear what they should be. The anti-support conditions for the connectives in BSML are

The resulting support conditions for \neg_i , the *intuitionistic negation*, are:

$$s \models \neg_i \varphi \iff \forall t \subseteq s : [t \models \varphi \implies t = \emptyset].$$

We define the propositional analogues of the team semantic closure properties, team properties, expressive completeness, and the notations $[\![\alpha]\!]_X$, $\|\varphi\|_X$, $||\varphi||_X$, etc. in the obvious way. Note that now $\bigcup \mathcal{P} = \bigcup \mathcal{P}$. We let $\|\varphi\|_X^{\pm, \neg} := (\|\varphi\|_X, \|\neg \varphi\|_X)$, $\|\varphi\|_X^{\pm, \neg} := (\|\varphi\|_X, \|\neg \varphi\|_X)$, and $\|\varphi\|_X^{\pm, \neg_i} := (\|\varphi\|_X, \|\neg_i \varphi\|_X)$.

As with ML, PL-formulas are flat, and the analogue of Fact 4.2.4 holds for these formulas. PL is indeed, therefore, simply classical propositional logic with team semantics, and the negation \neg is the classical negation with respect to worlds. PL(~) additionally incorporates the Boolean negation, which is essentially the classical negation with respect to teams. InqB features a negation \neg_i defined in terms of the intuitionistic implication \rightarrow . Note that, unlike with all other negations considered in this paper, double negation elimination is not valid for \neg_i (as, for instance, $\neg_i \neg_i (p \lor q) \equiv p \lor q \not\models p \lor q)$. The \lor -free fragment of InqB is an alternative way of formulating team-based classical propositional logic; one can show that this fragment is flat and that an analogue of 4.2.4 holds. Formulas InqB are downward closed and have the empty team property.

The usual formulations of PL and PL(~) (as well as those of the logics in Sections 4.3.3 and 4.3.4) in the literature do not include the dual negation; instead, these formulations feature what we can call the *restricted classical negation* \neg_c . This negation may only precede classical formulas, and its semantics are given by:

$$s \models \neg_c \alpha : \iff \forall w \in s : \{w\} \not\models \alpha$$

One can show that for all classical α (where α may also include \neg, \neg_i, \neg_c): $\neg \alpha \equiv \neg_i \alpha \equiv \neg_c \alpha$. These formulations are therefore expressively equivalent in the usual sense (of being able to express the same properties) to the present formulations; let us express this by writing, for instance, $\|PL_{\neg}\| = \|PL_{\neg_c}\|$. Since, in PL and PL(~), the dual negation only appears in front of classical formulas, the \neg_c -formulations of these logics are also bi-equivalent to the present ones (with respect to both \neg, \neg_c as well as \sim, \sim): $\|PL_{\neg}\|^{\pm, \neg} = \|PL_{\neg_c}\|^{\pm, \neg_c}$; $\|PL(\sim)_{\neg_c}\|^{\pm, \neg} = \|PL(\sim)_{\neg_c}\|^{\pm, \neg_c}$; and $\|PL(\sim)_{\neg}\|^{\pm, \sim} = \|PL(\sim)_{\neg_c}\|^{\pm, \sim}$. This second fact does not, however, hold for the logics we introduce in Sections 4.3.3 and 4.3.4—in these logics, the dual negation may occur in front of non-classical formulas, so the dual-negation formulations allow one to capture more pairs than the \neg_c -formulations. So for instance, for the propositional fragment of BSML, which we

motivated by empirical considerations, whereas those in all other logics we consider ultimately derive from the game-theoretic semantics for IF and D. In these semantics, the dual negation corresponds to an in-game move in which the players switch their verifier/falsifier roles. On the other hand, in the extensions of IF and D with ~, ~ φ is true just in case the verifier does not have a winning strategy for φ . The Boolean negation is evaluated from outside the game, so there is no way to interpret $\neg \sim \varphi$ in these semantics. For more discussion, see [75, 119]. See also [20], where the extension of D with ~ is provided with game-theoretic semantics in an indirect way, and Section 4.3.2, featuring the logic HS, which has a bilateral negation and in which ~ φ is uniformly definable (using the HS-definition of ~ φ from Section 4.3.2, we get $s \Rightarrow ~ \varphi \iff \forall t \subseteq s : t \models \varphi$)

denote by PL(NE), while we do have $||PL(NE)_{\neg}|| = ||PL(NE)_{\neg c}||$, it is not the case that $||PL(NE)_{\neg}||^{\pm,\neg} = ||PL(NE)_{\neg c}||^{\pm,\neg c}$. We prove Burgess theorems for the dual-negation formulations; no such theorems hold for the \neg_c -formulations (we will clarify below what we mean by 'Burgess theorems' in this context).

Returning to the properties of the logics PL, PL(~) and InqB, it can be shown:

4.3.2. THEOREM (Expressive completeness of PL, InqB, and PL(~)). [41, 140]

- (*i*) PL is expressively complete for $\{\mathcal{P} \mid \mathcal{P} \text{ is flat}\}$.
- (ii) InqB is expressively complete for $\{\mathcal{P} \mid \mathcal{P} \text{ is downward closed and has the empty team property}\}$.
- (iii) PL(~) is expressively complete for the class of all team properties.

We introduce the following natural incompatibility notions corresponding to the two types of classical negation \neg and \sim :

4.3.3. DEFINITION (World-incompatibility and Team-incompatibility).

- (i) φ_0 and φ_1 are *world-incompatible* (*W-I*) if $\{w\} \models \varphi_0 \iff \{w\} \not\models \varphi_1$.
- (ii) φ_0 and φ_1 are *team-incompatible* (*T-I*) if $s \models \varphi_0 \iff s \not\models \varphi_1$ (equivalently, if $\|\varphi_0\|_{\mathsf{X}} = \|\mathsf{T}\|_{\mathsf{X}} \setminus \|\varphi_1\|_{\mathsf{X}}$ for all $\mathsf{X} \supseteq \mathsf{P}(\varphi_0) \cup \mathsf{P}(\varphi_1)$).

Observe that, e.g., p and $\neg p$ are world-incompatible but not team-incompatible (since $\emptyset \vDash p \land \neg p$), whereas p and $\sim p$ are both team-incompatible as well as world-incompatible. These notions yield bicompleteness theorems for PL and PL(\sim), respectively, but we also introduce a stronger notion for PL; this will allow us to differentiate between the pairs expressible (modulo expressive power) in PL from those expressible in propositional dependence logic (Section 4.3.4).

4.3.4. DEFINITION (Flat-incompatibility). φ_0 and φ_1 are *flat-incompatible (F-I)* if they are world-incompatible and flat; or, equivalently, if any of the following holds for all $X \supseteq P(\varphi) \cup P(\psi)$ and $i \in \{0, 1\}$:

- (i) $\|\boldsymbol{\varphi}_i\|_{\mathsf{X}} = \wp(|\mathsf{T}|_{\mathsf{X}} \setminus |\boldsymbol{\varphi}_{1-i}|_{\mathsf{X}}).$
- (ii) $\|\varphi_i\|_{\mathsf{X}} = \{s \subseteq 2^{\mathsf{X}} \mid t \models \varphi_{1-i} \implies s \cap t = \emptyset\}.$
- (iii) $\|\varphi_i\|_{\mathsf{X}} = \bigcup \{\mathcal{P} \subseteq 2^{2^{\mathsf{X}}} \mid \mathcal{P} \text{ and } \|\varphi_{1-i}\|_{\mathsf{X}} \text{ are ground-incompatible} \}.$

Observe that p and $\neg p$ are flat-incompatible, whereas p and $\sim p$ are not.

As for the intuitionistic negation \neg_i , the truth conditions yield the following incompatibility notion:

4.3.5. DEFINITION ((One-sided) Down-set Incompatibility). φ_1 is a *down-set incompatibility* (*D-I*) of φ_0 if $[s \models \varphi_1 \text{ iff [for all } t \subseteq s, t \models \varphi_0 \text{ implies } t = \emptyset]].$

Unlike the other notions we consider, this is not symmetric—note, for instance, that $\neg p \land \neg q$ (which is equivalent to $\neg_i(p \lor q)$) is a down-set incompatibility of $p \lor q$, but not vice versa: we have that $[t \subseteq \{w_{p,\neg q}, w_{\neg p,q}\}$ and $t \models \neg p \land \neg q]$ implies $t = \emptyset$, but $\{w_{p,\neg q}, w_{\neg p,q}\} \not\models p \lor q$. This reflects the fact that double negation elimination is not sound for \neg_i . It is easy to check that D-I implies world-incompatibility, and that in a flat setting (such as that of PL), flat-incompatibility, D-I, and world-incompatibility are equivalent; we will therefore have that PL is bicomplete for each of the corresponding pair properties. That D-I implies W-I tracks the fact that in InqB the intuitionistic negation \neg_i behaves classically with respect to worlds/singletons; for instance, double negation elimination holds with respect to singletons: $\{w\} \models \neg_i \neg_i \varphi \iff \{w\} \models \varphi$.

Extending these incompatibility notions to pairs of properties in the obvious way, it is easy to see that the following bicompleteness results hold:

4.3.6. COROLLARY (Bicompleteness of PL, InqB, and PL(~)).

- (i) PL is bicomplete for F-I/W-I/D-I pairs of flat properties, and it is therefore also bicomplete for F-I/W-I/D-I pairs (with respect to ¬).
- (ii) InqB is bicomplete for D-I pairs of downward closed properties with the empty team property, and it is therefore also bicomplete for D-I pairs (with respect to \neg_i).
- (iii) $PL(\sim)$ is both bicomplete and bicomplete for T-I pairs (with respect to \sim).

Inclusions such as $\{(\mathcal{P}, \mathcal{Q}) \mid (\mathcal{P}, \mathcal{Q}) \text{ world-incompatible}_{PL} \subseteq \|PL\|^{\pm, \neg}$, which form part of the bicompleteness facts above, are of the same form as Burgess theorems. However, world-incompatibility relativized to pairs of flat properties is a pair property such that the first element of the pairs $(\mathcal{P}, \mathcal{Q})$ in the property determines the second element: $\mathcal{Q} = \wp(|| \top || \setminus \bigcup \mathcal{P})$. Given the converse inclusion, then, this inclusion is trivial: if $(\mathcal{P}, \mathcal{Q}) \in \{(\mathcal{P}, \mathcal{Q}) \mid (\mathcal{P}, \mathcal{Q}) \text{ world-incompatible}\}_{PL}$, then $\mathcal{Q} = \wp(||\top|| \setminus \bigcup \mathcal{P})$ and $\mathcal{P}, \mathcal{Q} \in \mathcal{Q}$ $\|PL\|$, whence $\mathcal{P} = \|\alpha\|$ and $\mathcal{Q} = \|\beta\|$ where $\alpha, \beta \in PL$. By the converse inclusion, $\|\alpha\|^{\pm} \in \mathbb{R}$ $\{(\mathcal{P}, \mathcal{Q}) \mid (\mathcal{P}, \mathcal{Q}) \text{ world-incompatible}_{PL}, \text{ whence } \|\neg \alpha\| = \wp(\|\top\| \setminus \bigcup \|\alpha\|) = \wp(\|\neg\| \cap \bigcap \|\alpha\|) = \wp(\| \cap \bigcap \|\alpha\|) = \wp(\|\neg\| \cap \bigcap \|\beta\|) = \wp(\|\neg\| \cap \bigcap \|\beta\|$ $(\mathcal{P}) = \mathcal{Q} = \|\beta\|, \text{ so } (\mathcal{P}, \mathcal{Q}) = (\|\alpha\|, \|\beta\|) = (\|\alpha\|, \|\neg\alpha\|) \in \|\mathrm{PL}\|^{\pm, \neg}.$ The same holds for all incompatibility notions considered in this section. Symmetric reasoning shows that the Burgess inclusion is similarly trivial if the second element always determines the first. In order to keep the term 'Burgess theorem' meaningful, let us reserve it for a theorem which shows an inclusion of this sort for a property of pairs containing at least one pair whose first element does not determine the second, and at least one pair whose second element does not determine the first. Note that this allows for a Burgess theorem with respect to a property of pairs such that in each pair, either the first element determines the second or the second the first (as long as the first does not always determine the second, etc.). We will see such a theorem in Section 4.3.2.

4.3.2 Hawke and Steinert-Threlkeld's Logic

Hawke and Steinert-Threlkeld's Logic HS [67] is essentially PL together with a operator \diamond intended to represent epistemic modals such as the 'might' in the sentence 'It might be raining' (cf. the operator \bullet in Section 4.4.2). The anti-support clauses for the conjunction and disjunction are different from those defined in Section 4.3.1; we will therefore use distinct symbols for the conjunction and disjunction of HS to avoid confusion.

4.3.7. DEFINITION (Syntax of HS). The set of formulas of HS is generated by:

$$\varphi ::= p \mid \bot \mid \neg \varphi \mid (\varphi \land \varphi) \mid (\varphi \lor \varphi) \mid \otimes \varphi$$

where $p \in \mathsf{Prop}$.

As before, let $\top := \neg \bot$. The support clauses for p, \bot , and $\neg \varphi$ are as before; the support clauses for \land and \lor are the same as those for \land and \lor , respectively. For \diamondsuit , we have:

$$s \vDash \Diamond \varphi : \iff s \neq \varphi.$$

The anti-support clause for $\neg \phi$ is as before. For all formulas ϕ whose main connective is not \neg :

$$s \dashv \varphi : \iff \forall t \subseteq s : [t \models \varphi \implies t = \emptyset].$$

Observe that this anti-support clause is equivalent to our usual one if φ is p or \bot .

We have made two alterations to the system as presented in [67]: first, we have formulated the semantics in terms of support and anti-support conditions (Hawke and Steinert-Therlkeld's semantics are essentially bilateral, but not formulated in terms of support/anti-support conditions). Second, we have added the constant \bot . This addition does not substantially change the logic— \bot is strongly uniformly definable in the \bot free fragment using any given $p \in Prop$ in that $\bot \equiv^{\pm} p \land \neg p$ —but it allows us to secure a simple expressive completeness theorem for the logic (by giving us the means to express all team properties over the empty set of propositional variables).

Many of the propositional analogues of the results in Section 4.2.1 hold for this logic; we note in particular that the natural analogue of Fact 4.2.4 holds for the \diamond -free fragment, meaning that this fragment is expressively equivalent (in the usual positive sense) to PL. The expressive completeness theorem mentioned above follows from this together with the fact that we can uniformly define ~ in HS:

4.3.8. THEOREM (Expressive completeness of HS). HS *is expressively complete for the class of all team properties.*

Proof:

Observe that $s \models \otimes \neg \varphi \iff s \neq \neg \varphi \iff s \neq \varphi \iff s \models \neg \varphi$. The result now follows from Theorem 4.3.2 (iii) together with the fact that the \otimes -free fragment of HS is expressively equivalent to PL.

Replacement of equivalents fails in negated contexts; note, for instance, that $\neg p \equiv \neg \otimes p$, but $\neg \neg p \equiv p \neq \otimes p \equiv \neg \neg \otimes p$. The incompatibility notion appropriate for HS is a kind of weaker two-sided version of down-set incompatibility:

4.3.9. DEFINITION ((Either-sided) Down-set Incompatibility). φ_0 and φ_1 are *down-set incompatible (on either side) (E-D-I)* if φ_0 is a down-set incompatibility of φ_1 or φ_1 is a down-set incompatibility of φ_0 .

4.3.10. PROPOSITION. For any $\varphi \in HS$, φ and $\neg \varphi$ are down-set incompatible (on either side).

Proof:

Case 1: φ is of the form $\neg^n \psi$, where \neg^n denotes a string of symbols \neg of length *n*, and *n* is even (possibly 0); and the main connective of ψ is not \neg . Then $\varphi \equiv \psi$ and $\neg \varphi \equiv \neg \psi$. We have $s \models \neg \psi$ iff $\forall t \subseteq s : [t \models \psi \implies t = \emptyset]$, and so $\neg \psi$ is a down-set incompatibility of ψ ; similarly $\neg \varphi$ is a down-set incompatibility of φ .

Case 2: φ is of the form $\neg^n \psi$, where \neg^n and ψ are as above, but *n* is odd. Then $\varphi \equiv \neg \psi$ and $\neg \varphi \equiv \psi$. Similarly to the above, we now have that φ is a down-set incompatibility of $\neg \varphi$.

Note that while the first element of a D-I pair determines the second, it is not the case that the first element of an E-D-I pair always determines the second (recall our example from above: $\neg p \equiv \neg \otimes p$, but $\neg \neg p \equiv p \neq \otimes p \equiv \neg \neg \otimes p$, so an E-D-I pair with first element $\|\neg p\|$ does not determine the second element) or that the second element always determines the first. It is, however, the case that for any given E-D-I pair, either the first element determines the second, or vice versa.

Clearly D-I implies E-D-I, and it is also easy to see that, like D-I, E-D-I implies W-I. For other implications, see Figure 4.2. Note that in a flat context, E-D-I is equivalent to F-I as well as D-I, so that PL is also bicomplete for E-D-I pairs.

We now show Burgess and bicompleteness theorems for HS using E-D-I. This time our proof will not be analogous to Burgess'; rather, the result follows easily from the basic properties of the negation.

4.3.11. THEOREM (Burgess theorem for HS). In HS, the following are equivalent:

(i) φ and ψ are down-set incompatible (on either side).

(*ii*) There is a formula θ such that $\varphi \equiv \theta$ and $\psi \equiv \neg \theta$ (and $P(\theta) \subseteq P(\varphi) \cup P(\psi)$).

Proof:

(ii) \implies (i) by Proposition 4.3.10. (i) \implies (ii): If ψ is D-I of φ , let $\theta := \varphi \land \top$. Then $\theta \equiv \varphi$ and

 $s \vDash \neg \theta \iff \forall t \subseteq s : [t \vDash \varphi \land \top \implies t = \emptyset] \iff \forall t \subseteq s : [t \vDash \varphi \implies t = \emptyset],$

so that $\neg \theta \equiv \psi$. Else if φ is D-I of ψ , let $\theta \coloneqq \neg(\psi \land \top)$. Similarly to the previous case, $\theta \equiv \varphi$ and $\neg \theta \equiv \psi$.

4.3.12. COROLLARY (Bicompleteness of HS). HS is both bicomplete and bicomplete for E-D-I pairs.

Note that by Corollaries 4.3.6 (iii) and 4.3.12, we have $||PL(\sim)|| = ||HS||$, whereas $||PL(\sim)||^{\pm,\sim} \neq ||HS||^{\pm,\sim}$.

Interestingly, Hawke and Steinert-Threlkeld also consider a variant of HS which uses \land and \lor instead of \land and \lor , but they dismiss it since the pairs it produces are not \bot -incompatible (among other reasons): "the altered system egregiously allows for the assertibility of contradictions" [67]. (HS allows for a formula and its negation to be simultaneously true in the empty team, but in the variant this also holds for nonempty teams. An example from [67]: $\neg(\otimes p \land \otimes \neg p) \land (\otimes p \land \otimes \neg p)$, which is true in any team *s* such that $s \nvDash p$ and $s \nvDash \neg p$.) The bicompleteness of this variant is an open problem.

4.3.3 The Propositional Fragments of BSML and BSML^{\vee}

Let PL(NE) (the propositional fragment of BSML) and PL(NE, W) (the propositional fragment of $BSML^{\mathbb{W}}$) be the extension of PL with NE, and with NE and W, respectively. In this section, we show that while PL(NE, W) is, like BSML and $BSML^{\mathbb{W}}$, bicomplete for ground-incompatible pairs, PL(NE) is bicomplete for pairs conforming to a stronger notion of incompatibility. We also note that a variant of the logic PL(NE, W) is both bicomplete and bicomplete for all pairs.

Define, as before, $\pm = \pm \land NE$ and $\bigvee \varnothing := \pm$. The support and anti-support conditions are again the obvious propositional analogues of the modal ones. The propositional analogues of all results in Section 4.2.1 hold; we note in particular that each formula is equivalent to one in negation normal form, that replacement holds in positive contexts, and that we have the following propositional analogue of Theorem 4.2.8:

4.3.13. THEOREM (Expressive completeness of PL(NE) and PL(NE, W)). [16, 140]

- (i) PL(NE) is expressively complete for $\{\mathcal{P} \mid \mathcal{P} \text{ is convex and union closed}\}$.
- (ii) PL(NE, W) is expressively complete for the class of all team properties.

As already noted in Section 4.2.3, the BSML^{\vee} Burgess theorem makes no essential use of the modal operators, and so the propositional analogues of Theorem 4.2.17 and Theorem 4.2.19 (using the second proof given for this theorem) give us:

4.3.14. THEOREM (Burgess theorem for PL(NE, W)). In PL(NE, W), the following are equivalent:

- (i) φ and ψ are ground-incompatible.
- (*ii*) There is a formula θ such that $\varphi \equiv \theta$ and $\psi \equiv \neg \theta$ (and $P(\theta) = P(\varphi) \cup P(\psi)$).

4.3.15. COROLLARY (Bicompleteness of PL(NE, W)). PL(NE, W) is both bicomplete and bicomplete for ground-incompatible pairs.

However, the propositional analogue of Theorem 4.2.19 does not hold for PL(NE). To show why, we recursively define the *flattening*¹⁶ $\varphi^f \in PL$ of $\varphi \in PL(NE)$ as follows: $p^f := p; \perp^f := \perp; NE^f := \top; (\neg \varphi)^f = \neg \varphi^f; (\varphi \land \psi)^f := \varphi^f \land \psi^f; \text{ and } (\varphi \lor \psi)^f := \varphi^f \lor \psi^f.$ Note that since φ^f is classical, we have $|\varphi^f| = [\![\varphi^f]\!]$. We require the following lemma for the sequel:

4.3.16. LEMMA. Let φ' be the negation normal form of $\varphi \in PL(NE)$. Then $\varphi^f \equiv \varphi'^f$.

Proof:

By induction on the complexity of φ . The cases for the atoms, \wedge , and \vee are all immediate. The cases for negated atoms are likewise immediate.

If $\varphi = \neg \neg \psi$, then $\varphi' = \psi'$. By the induction hypothesis, $\psi^f \equiv \psi'^f$. We then have $\varphi^f = \neg \neg \psi^f \equiv \psi'^f \equiv \psi'^f = \varphi'^f$.

If $\varphi = \neg(\psi \land \chi)$, then $\varphi' = (\neg \psi)' \lor (\neg \chi)'$. By the induction hypothesis, $(\neg \psi)^f \equiv (\neg \psi)'^f$ and $(\neg \chi)^f \equiv (\neg \chi)'^f$. We then have

$$\varphi^{f} = \neg(\psi^{f} \land \chi^{f}) \equiv \neg \psi^{f} \lor \neg \chi^{f} = (\neg \psi)^{f} \lor (\neg \chi)^{f} \equiv (\neg \psi)^{\prime f} \lor (\neg \chi)^{\prime f} = \varphi^{\prime f}.$$

The case for $\varphi = \neg(\psi \lor \chi)$ is similar to that for $\varphi = \neg(\psi \land \chi)$.

Using which we can show:

4.3.17. LEMMA. For all $\varphi \in PL(NE)$ and all $X \supseteq P(\varphi)$, either $|\varphi|_X = |\varphi^f|_X$, or $\varphi \equiv \bot$.

Proof:

By induction on the complexity of φ , which we may assume to be in negation normal form by Lemma 4.3.16. The base cases—those for atoms and negated atoms—are obvious; note in particular that $|NE| = |T| = |NE^f|$ since for any $w, w \in \{w\} \models NE$, and $|\neg NE| = |\bot| = |\neg T| = |(\neg NE)^f|$.

Consider $\varphi = \psi \land \chi$. If either $\psi \equiv \bot$ or $\chi \equiv \bot$, then $\psi \land \chi \equiv \bot$. Otherwise we have both $|\psi| = |\psi^f|$ and $|\chi| = |\chi^f|$. If there is no *s* such that $s \models \psi \land \chi$, then $\psi \land \chi \equiv \bot$, and we are done; we may therefore assume there is *s* with $s \models \psi \land \chi$. We now show $|\psi \land \chi| =$ $|\psi| \cap |\chi|$. $|\psi \land \chi| \subseteq |\psi| \cap |\chi|$ is immediate; for the converse inclusion, let $w \in |\psi| \cap |\chi|$. Then $w \in t \models \psi$ and $w \in u \models \chi$. Let *s* be such that $s \models \psi \land \chi$. By union closure, $s \cup t \models \psi$ and $s \cup u \models \chi$. Since $s \subseteq s \cup \{w\} \subseteq s \cup t$ and $s \subseteq s \cup \{w\} \subseteq s \cup u$, by convexity we have $s \cup \{w\} \models \psi$ and $s \cup \{w\} \models \chi$, so $w \in |\psi \land \chi|$. We then have

$$|\psi \wedge \chi| = |\psi| \cap |\chi| = |\psi^f| \cap |\chi^f| = \llbracket \psi^f \rrbracket \cap \llbracket \chi^f \rrbracket = \llbracket \psi^f \wedge \chi^f \rrbracket = \llbracket (\psi \wedge \chi)^f \rrbracket = |(\psi \wedge \chi)^f|.$$

Now consider $\varphi = \psi \lor \chi$. If either $\psi \equiv \bot$ or $\chi \equiv \bot$, then $\psi \lor \chi \equiv \bot$. Otherwise we have both $|\psi| = |\psi^f|$ and $|\chi| = |\chi^f|$. We show $|\psi \lor \chi| = |\psi| \cup |\chi|$. For $|\psi \lor \chi| \subseteq |\psi| \cup |\chi|$, if $w \in |\psi \lor \chi|$, then $w \in s \models \psi \lor \chi$. We have that $s = s_1 \cup s_2$ where $s_1 \models \psi$ and $s_2 \models \chi$, so $w \in s_1$

¹⁶The name 'flattening' is taken from Väänänen, who defines an analogous notion for D in [119, p. 42]. See also endnote 10.

or $w \in s_2$. Either way, $w \in |\psi| \cup |\chi|$. To show the converse inclusion, let $w \in |\psi| \cup |\chi|$. Then $w \in s$ where $s \models \psi$ or $s \models \chi$; assume without loss of generality that $s \models \psi$. Since $\chi \not\equiv \bot$, there is *t* such that $t \models \chi$. Then $s \cup t \models \psi \lor \chi$, so $s \in |\psi \lor \chi|$. We then have:

$$|\psi \lor \chi| = |\psi| \sqcup |\chi| = |\psi^f| \sqcup |\chi^f| = \llbracket \psi^f \rrbracket \sqcup \llbracket \chi^f \rrbracket = \llbracket \psi^f \lor \chi^f \rrbracket = \llbracket (\psi \lor \chi)^f \rrbracket = |(\psi \lor \chi)^f|.$$

Note that the BSML-analogue of Lemma 4.3.17 fails, since, for instance, extending the definition of flattening in the obvious way, we would have $|\Box \bot \bot| = |\bot|$ and $\Box \bot \bot \equiv \bot$, but $|(\Box \bot)^f| = |\Box (\bot \land \top)| = |\Box \bot | \neq |\bot|$.

Now assume for contradiction that the propositional analogue of Theorem 4.2.19 holds for PL(NE), i.e., that for any ground-incompatible φ and ψ , there is a θ such that $\varphi \equiv \theta$ and $\psi \equiv \neg \theta$. Let $\varphi \coloneqq p$ and $\psi \coloneqq \neg p \land q$. Clearly these are ground-incompatible, so there is a θ such that $\varphi \equiv \theta$ and $\psi \equiv \neg \theta$. By Lemma 4.3.17,

$$|\varphi| = |\theta| = |\theta^{f}| = [\![\theta^{f}]\!] = [\![\top]\!] \setminus [\![\neg \theta^{f}]\!] = |\top| \setminus |\neg \theta^{f}| = |\top| \setminus |(\neg \theta)^{f}| = |\top| \setminus |\neg \theta| = |\top| \setminus |\psi|.$$

But this leads to contradiction, since, for instance, $w_{\neg p,\neg q} \notin |\varphi|$ and $w_{\neg p,\neg q} \notin |\psi|$.

PL(NE)-formulas and their negations conform, instead, to the following incompatibility notion.

4.3.18. DEFINITION (Ground-complementariness (modulo **L**)).

- (i) φ_0 and φ_1 are ground-complementary if $|\varphi_i|_X = |\top|_X \setminus |\varphi_{1-i}|_X$ for all $i \in \{0, 1\}$ and $X \supseteq P(\varphi) \cup P(\psi)$.
- (ii) φ_0 and φ_1 are ground-complementary modulo \perp if

(a) they are ground-complementary, (b) $\varphi_0 \equiv \bot$, or (c) $\varphi_1 \equiv \bot$.

Similarly, let \mathcal{P} and \mathcal{Q} be ground-complementary (modulo \bot) if $\bigcup \mathcal{P} = |\top| \setminus \bigcup \mathcal{Q}$ (or $\mathcal{P} = \emptyset$ or $\mathcal{Q} = \emptyset$). We have:

4.3.19. PROPOSITION. For any $\varphi \in PL(NE)$, φ and $\neg \varphi$ are ground-complementary modulo \bot .

Proof:

If $\varphi \equiv \bot$ or $\neg \varphi \equiv \bot$, we are done. Otherwise by Lemma 4.3.17,

$$|\varphi| = |\varphi^f| = \llbracket \varphi^f \rrbracket = \llbracket \top \rrbracket \smallsetminus \llbracket \neg \varphi^f \rrbracket = |\mathsf{T}| \smallsetminus |\neg \varphi^f| = |\mathsf{T}| \smallsetminus |(\neg \varphi)^f| = |\mathsf{T}| \smallsetminus |\neg \varphi|. \qquad \Box$$

Note that we have the following connection between ground-complementariness and world-incompatibility:

4.3.20. FACT. For downward-closed φ , ψ : φ and ψ are world-incompatible if and only if they are ground-complementary. Neither implication holds for all formulas.

Proof:

Let φ and ψ be world-incompatible, and let $w \in |\varphi|$. Assume for contradiction that $w \in |\psi|$. We then have that $w \in s \models \varphi$ and $w \in t \models \psi$, whence by downward closure, $\{w\} \models \varphi$ and $\{w\} \models \psi$, contradicting world-incompatibility. So $|\varphi| \subseteq |T| \setminus |\psi|$. Now let $w \in |T| \setminus |\psi|$. By world-incompatibility, either $\{w\} \models \varphi$ or $\{w\} \models \psi$. The latter clearly implies $w \in |\psi|$, so we must have $\{w\} \models \varphi$. Therefore, $|T| \setminus |\psi| \subseteq |\varphi|$ and so $|\varphi| = |T| \setminus |\psi|$.

Conversely, let φ and ψ be ground-complementary. Let $\{w\} \models \varphi$ and assume for contradiction that $\{w\} \models \psi$. Then $w \in |\varphi| \cap |\psi|$, contradicting ground-complementariness. So $\{w\} \models \varphi \implies \{w\} \not\models \psi$. Now let $\{w\} \not\models \psi$. Assume for contradiction that $w \in |\psi|$. Then $w \in t \models \psi$, so by downward closure, $\{w\} \models \psi$, contradicting our assumption. So $w \notin |\psi|$, whence $w \in |\varphi|$ by ground-complementariness.

To see why world-incompatibility does not imply ground-complementariness in general, consider, for instance, the pair \top and $(p \land NE) \lor (\neg p \land NE)$, or the pair q and $\neg q \lor ((p \land NE) \lor (\neg p \land NE))$.

To see why ground-complementariness does not imply world-incompatibility in general, consider, for instance, the pair \perp and $(p \land NE) \lor (\neg p \land NE)$, or the pair $q \land ((p \land NE) \lor (\neg p \land NE))$ and $\neg q \land ((p \land NE) \lor (\neg p \land NE))$.

Furthermore, for formulas with the empty team property, G-C modulo \perp and G-C are clearly equivalent. Therefore, given the fact above, G-C modulo \perp , G-C, and W-I are all equivalent for downward-closed properties with the empty team property. In a flat context, each of these is also equivalent to each of F-I, D-I, and E-D-I, so that PL is also bicomplete for G-C/G-C mod \perp pairs.

We now show a Burgess theorem for PL(NE) using ground-complementariness modulo \bot . As in the preceding section, our proof will not be analogous to Burgess'; this time we derive our result by making a small modification to the proof of Theorem 4.3.13 (i). This proof makes use of the propositional analogues $\chi_w^X := \bigwedge \{p \mid p \in X, w \models p\} \land \bigwedge \{\neg p \mid p \in X, w \neq p\}$ and $\chi_s^X := \bigvee_{w \in s} \chi_w^X$ of Hintikka formulas. As with Hintikka formulas, we have $w' \models \chi_w^X \iff w' \upharpoonright X = w \upharpoonright X$ (so that $w' \models \chi_w^X \iff w' = w$ if $w, w' \in 2^X$) and $s' \models \chi_s^X \iff s' \upharpoonright X = t$ for some $t \subseteq s \upharpoonright X$, where $s \upharpoonright X := \{v \upharpoonright X \mid v \in s\}$, etc. (so that $s' \models \chi_s^X \iff s' \subseteq s$ if $s, s' \subseteq 2^X$). Given a property $\mathcal{P} = \{t_1, \ldots, t_n\}$, define

$$\delta_{\mathcal{P}}^{\mathsf{X}} \coloneqq \bigwedge_{w_1 \in t_1, \dots, w_n \in t_n} \left(\left(\left(\chi_{w_1}^{\mathsf{X}} \lor \dots \lor \chi_{w_n}^{\mathsf{X}} \right) \land \operatorname{NE} \right) \lor \top \right).$$

Theorem 4.3.13 (i) essentially follows from:

4.3.21. PROPOSITION. [16] For any nonempty, convex, and union-closed property $\mathcal{P} = \{t_1, \ldots, t_n\}$ over $X \subseteq \text{Prop}$ (where X is finite), $\mathcal{P} = \|\bigvee_{t \in \mathcal{P}} \chi_t^X \wedge \delta_{\mathcal{P}}^X\|_X$.

Using which we get:

4.3.22. PROPOSITION. For any nonempty, convex, and union-closed properties $\mathcal{P} = \{t_1, \ldots, t_n\}, \mathcal{Q} = \{s_1, \ldots, s_m\}$ over $X \subseteq \text{Prop}$ (where X is finite) such that \mathcal{P}, \mathcal{Q} are ground-complementary modulo $\mathbb{L}, (\mathcal{P}, \mathcal{Q}) = \|(\bigvee_{t \in \mathcal{P}} \chi_t^X \land \delta_{\mathcal{P}}^X) \lor \neg \delta_{\mathcal{Q}}^X\|_X^{\pm}$.

Proof:

It is easy to check that $\neg \delta_Q \equiv \bot$, whence

$$\left(\bigvee_{t\in\mathcal{P}}\chi_t\wedge\delta_{\mathcal{P}}\right)\vee\neg\delta_{\mathcal{Q}}\equiv\left(\bigvee_{t\in\mathcal{P}}\chi_t\wedge\delta_{\mathcal{P}}\right)\vee\bot\equiv\bigvee_{t\in\mathcal{P}}\chi_t\wedge\delta_{\mathcal{P}},$$

and therefore, by Proposition 4.3.21, $\|\bigvee_{t\in\mathcal{P}}\chi_t \wedge \delta_{\mathcal{P}} \vee \neg \delta_{\mathcal{Q}}\| = \mathcal{P}$. It remains to show that $\|\neg((\bigvee_{t\in\mathcal{P}}\chi_t \wedge \delta_{\mathcal{P}}) \vee \neg \delta_{\mathcal{Q}})\| = \mathcal{Q}$. We have:

$$\neg ((\bigvee_{t\in\mathcal{P}}\chi_t \wedge \delta_{\mathcal{P}}) \vee \neg \delta_{\mathcal{Q}}) \equiv (\neg \bigvee_{t\in\mathcal{P}}\chi_t \vee \neg \delta_{\mathcal{P}}) \wedge \delta_{\mathcal{Q}} \equiv (\neg \bigvee_{t\in\mathcal{P}}\chi_t \vee \bot) \wedge \delta_{\mathcal{Q}} \equiv \neg \bigvee_{t\in\mathcal{P}}\chi_t \wedge \delta_{\mathcal{Q}}.$$

Observe that $\bigcup \mathcal{P} = |\bigvee_{t \in \mathcal{P}} \chi_t| = [\![\bigvee_{t \in \mathcal{P}} \chi_t]\!]$, and similarly for \mathcal{Q} . Since \mathcal{P} and \mathcal{Q} are ground-complementary modulo \mathbb{I} , and since they are nonempty, $\bigcup \mathcal{Q} = |\mathsf{T}| \setminus \bigcup \mathcal{P}$. We therefore have:

$$\left[\!\left[\neg\bigvee_{t\in\mathcal{P}}\chi_{t}\right]\!\right]=\left[\!\left[\top\right]\!\right]\smallsetminus\left[\!\left[\bigvee_{t\in\mathcal{P}}\chi_{t}\right]\!\right]=\left[\!\left[\top\right]\!\right]\smallsetminus\bigcup\mathcal{P}=\bigcup\mathcal{Q}=\left[\!\left[\bigvee_{s\in\mathcal{Q}}\chi_{s}\right]\!\right],$$

from which it follows by flatness that $\|\neg \bigvee_{t \in \mathcal{P}} \chi_t\| = \|\bigvee_{s \in \mathcal{Q}} \chi_s\|$. We therefore have

$$\neg ((\bigvee_{t\in\mathcal{P}}\chi_t\wedge\delta_{\mathcal{P}})\vee\neg\delta_{\mathcal{Q}})\equiv\bigvee_{s\in\mathcal{Q}}\chi_s\wedge\delta_{\mathcal{Q}},$$

so that by Proposition 4.3.21, $\|\neg((\bigvee_{t\in\mathcal{P}}\chi_t \land \delta_{\mathcal{P}}) \lor \neg \delta_{\mathcal{Q}})\| = \mathcal{Q}.$

The Burgess and bicompleteness theorems now follow easily:

4.3.23. THEOREM (Burgess theorem for PL(NE)). In PL(NE), the following are equivalent:

- (*i*) φ and ψ are ground-complementary modulo \bot .
- (*ii*) There is a formula θ such that $\varphi \equiv \theta$ and $\psi \equiv \neg \theta$ (and $P(\theta) = P(\varphi) \cup P(\psi)$).

Proof:

(ii) \implies (i) by Proposition 4.3.19. For (i) \implies (ii), if $\psi \equiv \bot$, let $\theta := \varphi \land \neg \bot$. Then $\theta \equiv \varphi \land (\neg \bot \lor \neg NE) \equiv \varphi \land (\top \lor \bot) \equiv \varphi$ and $\neg \theta \equiv \neg \varphi \lor \bot \equiv \bot \equiv \psi$. The case for $\varphi \equiv \bot$ is similar. If $\varphi \not\equiv \bot$ and $\psi \not\equiv \bot$, let $X := P(\varphi) \cup P(\psi)$, and let $\theta := (\bigvee_{t \in \mathcal{P}} \chi_t^X \land \delta_{\mathcal{P}}^X) \lor \neg \delta_{\mathcal{Q}}^X$. By Proposition 4.3.22, θ is as desired.

4.3.24. COROLLARY (Bicompleteness of PL(NE)). PL(NE) is bicomplete for

 $\{(\mathcal{P}, \mathcal{Q}) | \mathcal{P}, \mathcal{Q} \text{ are convex and union closed;} \\ \mathcal{P}, \mathcal{Q} \text{ are ground-complementary modulo } \mathbb{I}\},$

and hence bicomplete for pairs which are ground-complementary modulo \bot .

We conclude this section by showing that a variant of PL(NE, W) is both bicomplete and bicomplete for all pairs. Let $PL(NE^*, W)$ be PL(NE, W) with NE swapped out for an atom NE* with the following support/anti-support clauses: $s \models NE^* : \iff s \neq \emptyset$ and $s \exists NE^* : \iff s \neq \emptyset$.¹⁷ Clearly many of the properties of PL(NE, W) are preserved in this variant: we still have a negation normal form, replacement in non-negated contexts, and expressive completeness for all properties.

We define $\top := \neg \top$, $\bot ^* := \neg ((NE^* \lor \bot) \lor \top)$, and $NE^{*-} := \neg ((NE^* \land \bot ^*) \lor \bot)$. One can then check that $\bot ^* \equiv \bot$ and $NE^{*-} \equiv NE$.

4.3.25. THEOREM (Burgess theorem for $PL(NE^*, W)$). For any $\varphi, \psi \in PL(NE^*, W)$, there is a $\theta \in PL(NE^*, W)$ such that $\varphi \equiv \theta$ and $\psi \equiv \neg \theta$.

Proof:

Let φ', ψ' be the negation normal forms of φ, ψ respectively. Let φ^* be the result of replacing each occurrence of NE^{*} in φ' with NE^{*-}, and each occurrence of \neg NE^{*} with \bot , and similarly for ψ^* . We then have, as in Lemma 4.2.18, that $\varphi^* \equiv \varphi$, and that $\neg \varphi^*$ has the empty team property, and similarly for ψ^* . Now let $\varphi_T := \neg(\neg \varphi^* \lor \neg(NE^* \lor \bot))$, and similarly for ψ_T . Then:

where the final equivalence holds because $\neg \phi^*$ has the empty team property. Similarly, $\psi_{\top} \equiv \psi$ and $\neg \psi_{\top} \equiv \top$. Finally, let $\theta \coloneqq \phi_{\top} \lor (\bot^* \lor \neg \psi_{\top})$. Then:

| θ | = | $arphi_{	op}$ $arphi$ $arphi$ $arphi$ $arphi$ $arphi$ $arphi$ $arphi$ $arphi$ $arphi$ | Ξ | $oldsymbol{arphi}_{	op}$ | Ξ | φ; | |
|---------------|---|--|---|--|---|--|--|
| $\neg \theta$ | = | $\neg(\varphi_{\scriptscriptstyle \top} \! \lor \! (\! \bot^* \! \lor \! \neg \psi_{\scriptscriptstyle \top}))$ | Ξ | $\neg \varphi_{\scriptscriptstyle \top} \land \neg ({\scriptscriptstyle \bot\!$ | ≡ | ${}^{\top} \wedge (\neg {}^{\!$ | |
| | Ξ | ${\scriptscriptstyle \top} \land ({\scriptscriptstyle \top} \land \psi_{\scriptscriptstyle \top})$ | Ξ | $oldsymbol{\psi}_{	op}$ | Ξ | ψ. | |

In particular, for any φ there is a θ such that $\theta \equiv \varphi \equiv \neg \theta$.

4.3.26. COROLLARY (Bicompleteness of $PL(NE^*, W)$). $PL(NE^*, W)$ is both bicomplete and bicomplete for all pairs.

4.3.4 Propositional Dependence Logic

Propositional dependence logic $PL(=(\cdot))$ is PL extended with the n + 1-ary $(n \ge -1)$ connectives $=(p_1...,p_n,q)$, where $p_1,...,p_n,q \in Prop$. The formulas $=(p_1...,p_n,q)$ are called *dependence atoms*. In this section, we show that $PL(=(\cdot))$ is bicomplete for ground-complementary pairs (recall that, in contrast, first-order dependence logic D is bicomplete for \perp -incompatible pairs).

¹⁷NE^{*} was introduced by Tomasz Klochowicz in unpublished work.

The support/anti-support conditions for dependence atoms are as follows:

$$s \models=(p_1 \dots, p_n, q) \quad : \iff \quad \forall v, w \in s : [v \models p_i \iff w \models p_i \text{ for all } \forall 1 \le i \le n] \implies [v \models q \iff w \models q];$$
$$s \models=(p_1 \dots, p_n, q) \quad : \iff \quad s = \emptyset.$$

In other words, a dependence atom $=(p_1...,p_n,q)$ is true/supported in a team *s* if, whenever two valuations *v* and *w* in *s* agree on the truth values of all the p_i , they also agree on the truth value of *q*—in other words, the values of $p_i,...,p_n$ jointly determine the value of *q* in any valuation in the team. $=(p_1...,p_n,q)$ is anti-supported in a team just in case the team is empty.¹⁸ We call a unary dependence atom =(p) a *constancy atom*. =(p) expresses that the value of *p* is constant in all valuations in a team.

The propositional analogues of many results in Section 4.2.1 also hold in the context of $PL(=(\cdot))$; we note in particular that each formula is equivalent to one in negation normal form, and that replacement holds in positive contexts. As with first-order dependence logic D, formulas of $PL(=(\cdot))$ are downward closed and have the empty team property, which, by Fact 4.2.14 and the definition of NE-incompatibility, means that ground-incompatibility, \perp -incompatibility, and \emptyset -incompatibility are equivalent for these formulas. Formulas of $PL(=(\cdot))$ need not be union closed: consider =(p). We have the following expressive completeness theorem:

4.3.27. THEOREM (Expressive completeness of $PL(=(\cdot))$). [139] $PL(=(\cdot))$ is expressively complete for $\{\mathcal{P} \mid \mathcal{P} \text{ is downward closed and has the empty team property}\}$.

In contrast with first-order dependence logic D, a Burgess theorem that employs \perp -incompatibility does not hold for PL(=(·)). We can show this, as we did the analogous fact for PL(NE), using flattenings of formulas. Define the flattening $\varphi^f \in PL$ of $\varphi \in PL(=(\cdot))$ as in Section 4.3.3, together with = $(p_1 \dots, p_n, q)^f := \top$. As in Section 4.3.3, it can be shown:

Note also that Väänänen's clause preserves the ground-incompatibility and NE-incompatibility of φ and $\neg \varphi$, whereas the variant above does not.

¹⁸The anti-support clause for $=(p_1..., p_n, q)$ might seem puzzling at first glance. What is the reason, one might ask, for adopting the clause we use rather than, say, $s ==(p_1..., p_n, q)$ iff $s \neq =(p_1..., p_n, q)$? (Pietro Galliani, in recent unpublished work [53], considers a variant of D with a first-order version of this anti-support clause.) The clause we use has been adapted from the analogous first-order clause for D in [119]. This first-order clause was, in turn, chosen (it seems to me) for the following the reasons: it (unlike the variant clause above) preserves the equivalence between IF and D, as well as the empty team property and downward closure. Väänänen [119, p. 24] also offers the following explanation (notation amended; adapted for the propositional setting):

Why not allow $s \models (p_1 \dots, p_n, q)$ for non-empty *s*? The reason is that if we negate "for all $w, w' \in s$ such that $w \models p_1 \iff w' \models p_1, \dots, w \models p_n \iff w' \models p_n$, we have $w \models q \iff w' \models q$ ", maintaining the analogy with $[s \models p \text{ iff for all } w \in s : w \not\models p]$, we get "for all $w, w' \in s$ we have $w \models p_1 \iff w' \models p_1, \dots, w \models p_n \iff w' \models p_n$ and $w \models q \iff w' \models q$ ", which is only possible if $s = \emptyset$.
4.3.28. LEMMA. Let φ' be the negation normal form of $\varphi \in PL(=(\cdot))$. Then $\varphi^f \equiv \varphi'^f$.

And we have:

4.3.29. LEMMA. For all $\varphi \in PL(=(\cdot))$ and all $X \supseteq P(\varphi)$, $|\varphi|_X = |\varphi^f|_X$.

Proof:

By induction on the complexity of φ , which we may assume to be in negation normal form by Lemma 4.3.28. The base cases—those for atoms and negated atoms—are obvious; note that $|=(p_1...,p_n,q)| = |\top| = |=(p_1...,p_n,q)^f|$ since $w \in \{w\} \models =(p_1...,p_n,q)$ for any w, and $|\neg =(p_1...,p_n,q)| = |\bot| = |(\neg =(p_1...,p_n,q))^f|$.

Consider $\varphi = \psi \land \chi$. We show $|\psi \land \chi| = |\psi| \cap |\chi|$; the result will then follow as in the proof of Lemma 4.3.17. $|\psi \land \chi| \subseteq |\psi| \cap |\chi|$ is immediate; for the converse inclusion, let $w \in |\psi| \cap |\chi|$. Then $w \in t \models \psi$ and $w \in u \models \chi$. By downward closure, $\{w\} \models \psi$ and $\{w\} \models \chi$, so $w \in |\psi \land \chi|$.

Now consider $\varphi = \psi \lor \chi$. We show $|\psi \lor \chi| = |\psi| \cup |\chi|$; the result will then follow as in the proof of Lemma 4.3.17. $|\psi \lor \chi| \subseteq |\psi| \cup |\chi|$ follows as in the proof of Lemma 4.3.17. For the converse inclusion, let $w \in |\psi| \cup |\chi|$. Then $w \in s$ where $s \models \psi$ or $s \models \chi$; assume without loss of generality that $s \models \psi$. By the empty team property, $s = s \cup \emptyset \models \psi \lor \chi$, so $w \in |\psi \lor \chi|$.

One can then show that a Burgess theorem employing \perp -incompatibility would lead to contradiction using essentially the same argument as that used for the analogous fact in Section 4.3.3. We could instead show a Burgess theorem for PL(=(·)) using ground-complementariness modulo \perp , as we did for PL(NE), but, as can be gleaned from the lemma above, or, alternatively, from the fact that formulas of PL(=(·)) have the empty team property, 'modulo \perp ' is redundant here—we may simply use groundcomplementariness. Note that since PL(=(·)) is downward closed, by Fact 4.3.20 we could also equivalently use world-incompatibility. We have:

4.3.30. PROPOSITION. For any $\varphi \in PL(=(\cdot))$, φ and $\neg \varphi$ are ground-complementary.

Proof:

By Lemma 4.3.29; almost the same as the proof of Lemma 4.3.19.

We now show a Burgess theorem for $PL(=(\cdot))$ using ground-complementariness. As with PL(NE), we prove this by modifying the proof of the relevant expressive completeness theorem, Theorem 4.3.27. For further details on what follows, see [139].

For a finite $X \subseteq Prop$, define $\gamma_0^X := \bot$, $\gamma_1^X := \bigwedge_{p \in X} = (p)$, and for $n \ge 2$, $\gamma_n^X := \bigvee_n \gamma_1$. Then for $s \subseteq 2^X$, we have $s \models \gamma_n^X \iff |s| \le n$, where |s| is the size of s. Note that $\neg \gamma_0^X \equiv \top$ and $\neg \gamma_n^X \equiv \bot$ for $n \ge 1$.

and $\neg \gamma_n^{\mathsf{X}} \equiv \bot$ for $n \ge 1$. For a nonempty $s \subseteq 2^{\mathsf{X}}$, define $\xi_s^{\mathsf{X}} := \gamma_{|s|-1}^{\mathsf{X}} \lor \chi_{|\mathsf{T}|_{\mathsf{X}} \smallsetminus s}^{\mathsf{X}}$. It can be shown that for $t \subseteq 2^{\mathsf{X}}$, $t \models \xi_s^{\mathsf{X}} \iff s \notin t$. Note that if |s| = 1, then $\xi_s^{\mathsf{X}} \equiv \chi_{|\mathsf{T}|_{\mathsf{X}} \smallsetminus s}^{\mathsf{X}}$ so $|\xi_s^{\mathsf{X}}|_{\mathsf{X}} = |\chi_{|\mathsf{T}|_{\mathsf{X}} \smallsetminus s}^{\mathsf{X}}|_{\mathsf{X}}$, and if |s| > 2,

then $|\xi_s^X|_X = |T|_X$. Note also that if |s| = 1 (say $s = \{w\}$), then $\neg \xi_s^X \equiv T \land \neg \chi_{|T|_X \land s}^X \equiv \chi_s^X = \chi_w^X$ and if |s| > 1, then $\neg \xi_s^X \equiv \bot$.

Theorem 4.3.27 essentially follows from:

4.3.31. PROPOSITION. [139] For any downward-closed property $\mathcal{P} = \{t_1, \ldots, t_n\}$ with the empty team property over $X \subseteq \text{Prop}$ (where X is finite), $\mathcal{P} = \| \bigwedge_{s \in \|T\|_X \smallsetminus \mathcal{P}} \xi_s^X \|_X$.

For a property $\mathcal{P} = \{t_1, \dots, t_n\}$ over a finite X, define $\mathcal{P}^1 := \{s \in \mathcal{P} \mid |s| = 1\}$ and $\mathcal{P}^{>1} := \{s \in \mathcal{P} \mid |s| > 1\}$. We have the following corollary of Proposition 4.3.31:

4.3.32. PROPOSITION. For any downward-closed properties $\mathcal{P} = \{t_1, \ldots, t_n\}$ and $\mathcal{Q} = \{s_1, \ldots, s_m\}$ with the empty team property over $X \subseteq \text{Prop}$ (where X is finite) such that \mathcal{P}, \mathcal{Q} are ground-complementary,

$$(\mathcal{P},\mathcal{Q}) = \left\| \bigwedge_{s \in \|\mathbb{T}\|_{\mathsf{X}} \sim \mathcal{P}} \xi_{s}^{\mathsf{X}} \vee \neg \bigwedge_{s \in (\|\mathbb{T}\|_{\mathsf{X}} \sim \mathcal{Q})^{>1}} \xi_{s}^{\mathsf{X}} \right\|_{\mathsf{X}}^{\pm}.$$

Proof:

It is easy to check that $\neg \bigwedge_{s \in (||T|| \setminus Q)^{>1}} \xi_s \equiv \bot$, whence

$$\bigwedge_{s \in \|\mathsf{T}\| \smallsetminus \mathcal{P}} \xi_s \lor \neg \bigwedge_{s \in (\|\mathsf{T}\| \smallsetminus \mathcal{Q})^{>1}} \xi_s \equiv \bigwedge_{s \in \|\mathsf{T}\| \smallsetminus \mathcal{P}} \xi_s \lor \bot \equiv \bigwedge_{s \in \|\mathsf{T}\| \smallsetminus \mathcal{P}} \xi_s,$$

and therefore by Proposition 4.3.31, $\| \bigwedge_{s \in \| \mathbb{T} \| \setminus \mathcal{P}} \xi_s \vee \neg \bigwedge_{s \in (\| \mathbb{T} \| \setminus \mathcal{Q})^{>1}} \xi_s \| = \mathcal{P}$. It remains to show $\| \neg (\bigwedge_{s \in \| \mathbb{T} \| \setminus \mathcal{P}} \xi_s \vee \neg \bigwedge_{s \in (\| \mathbb{T} \| \setminus \mathcal{Q})^{>1}} \xi_s) \| = \mathcal{Q}$. We first prove:

For any team
$$s, \chi_s^{\mathsf{X}} \equiv \bigwedge_{w \in |\mathsf{T}|_{\mathsf{X} \times s}} \chi_{|\mathsf{T}|_{\mathsf{X} \times \{w\}}}^{\mathsf{X}}.$$
 (4.2)

Proof of (4.2):

$$\chi_{s} = \bigvee_{v \in s} \chi_{v} \equiv \bigwedge_{w \in |\mathsf{T}| \smallsetminus s} \neg \chi_{w} \equiv \bigwedge_{w \in |\mathsf{T}| \smallsetminus s} \bigvee_{w' \in |\mathsf{T}| \smallsetminus \{w\}} \chi_{w'} \equiv \bigwedge_{w \in |\mathsf{T}| \smallsetminus s} \chi_{|\mathsf{T}| \smallsetminus \{w\}}. \quad \dashv$$

We can now prove $\left\|\neg\left(\bigwedge_{s\in\|\mathbb{T}\|\sim\mathcal{P}}\xi_s\vee\neg\bigwedge_{s\in(\|\mathbb{T}\|\sim\mathcal{Q})^{>1}}\xi_s\right)\right\|=\mathcal{Q}$. We have:

$$= \bigcap_{s \in \|T\| \setminus \mathcal{P}} \xi_{s} \vee \neg \bigwedge_{s \in (\|T\| \setminus \mathcal{Q})^{>1}} \xi_{s})$$

$$= \bigcap_{s \in \|T\| \setminus \mathcal{P}} \xi_{s} \wedge \bigwedge_{s \in (\|T\| \setminus \mathcal{Q})^{>1}} \xi_{s}$$

$$= \bigcap_{s \in (\|T\| \setminus \mathcal{P})^{1}} \xi_{s} \wedge \bigwedge_{s \in (\|T\| \setminus \mathcal{P})^{>1}} \xi_{s}) \wedge \bigwedge_{s \in (\|T\| \setminus \mathcal{Q})^{>1}} \xi_{s}$$

$$= \bigcap_{s \in (\|T\| \setminus \mathcal{P})^{1}} \xi_{s} \vee \neg \bigwedge_{s \in (\|T\| \setminus \mathcal{P})^{>1}} \xi_{s}) \wedge \bigwedge_{s \in (\|T\| \setminus \mathcal{Q})^{>1}} \xi_{s}$$

$$= \left(\neg \bigwedge_{s \in (\|T\| \setminus \mathcal{P})^{1}} \xi_{s} \vee 1 \right) \wedge \bigwedge_{s \in (\|T\| \setminus \mathcal{Q})^{>1}} \xi_{s}$$

$$= \neg \bigwedge_{s \in (\|T\| \setminus \mathcal{P})^{1}} \xi_{s} \wedge \bigwedge_{s \in (\|T\| \setminus \mathcal{Q})^{>1}} \xi_{s}$$

$$= \bigvee_{s \in (\|T\| \setminus \mathcal{P})^{1}} \neg \xi_{s} \wedge \bigwedge_{s \in (\|T\| \setminus \mathcal{Q})^{>1}} \xi_{s}$$

$$= \bigvee_{s \in (\|T\| \setminus \mathcal{P})^{1}} \chi_{s} \wedge \bigwedge_{s \in (\|T\| \setminus \mathcal{Q})^{>1}} \xi_{s}$$

$$= \bigvee_{\{w\} \in (\|T\| \setminus \mathcal{P})^{1}} \chi_{w} \wedge \bigwedge_{s \in (\|T\| \setminus \mathcal{Q})^{>1}} \xi_{s}$$

$$= \bigvee_{\{w\} \in (\|T\| \setminus \mathcal{P})^{1}} \chi_{w} \wedge \bigwedge_{s \in (\|T\| \setminus \mathcal{Q})^{>1}} \xi_{s}$$

$$= \bigvee_{\{w\} \in (\|T\| \setminus \mathcal{Q})^{>1}} \chi_{w} \wedge \bigwedge_{s \in (\|T\| \setminus \mathcal{Q})^{>1}} \xi_{s}$$

$$= (\text{left-to-right by downward closure of } \mathcal{P})$$

$$= \bigwedge_{w \in \cup \mathcal{P}} \chi_{|T| \setminus \{w\}} \wedge \bigwedge_{s \in (\|T\| \setminus \mathcal{Q})^{>1}} \xi_{s}$$

$$= \bigwedge_{w \in \cup \mathcal{P}} \xi_{\{w\}} \wedge \bigwedge_{s \in (\|T\| \setminus \mathcal{Q})^{>1}} \xi_{s}$$

$$= \bigwedge_{w \in |T| \setminus \cup \mathcal{Q}} \xi_{\{w\}} \wedge \bigwedge_{s \in (\|T\| \setminus \mathcal{Q})^{>1}} \xi_{s}$$

$$= \bigwedge_{w \in |T| \setminus \cup \mathcal{Q}^{1}} \xi_{\{w\}} \wedge \bigwedge_{s \in (\|T\| \setminus \mathcal{Q})^{>1}} \xi_{s}$$

$$(\text{right-to-left by downward closure of } \mathcal{Q})$$

$$= \bigwedge_{s \in \|T\| \setminus \mathcal{Q}} \xi_{s}$$

The result now follows by Proposition 4.3.31.

And so we have our Burgess and bicompleteness theorems:

4.3.33. THEOREM (Burgess theorem for $PL(=(\cdot))$). In the logic $PL(=(\cdot))$, the following are equivalent:

- (i) φ and ψ are ground-complementary.
- (*ii*) There is a formula θ such that $\varphi \equiv \theta$ and $\psi \equiv \neg \theta$ (and $P(\theta) = P(\varphi) \cup P(\psi)$).

Proof:

(ii) \implies (i) by Proposition 4.3.30. For (i) \implies (ii), let $X := P(\varphi) \cup P(\psi)$, and let $\theta := \bigwedge_{s \in ||T||_X \searrow \mathcal{P}} \xi_s^X \lor \neg \bigwedge_{s \in (||T||_X \searrow \mathcal{Q})^{>1}} \xi_s^X$. By Proposition 4.3.32, θ is as desired. \Box

4.3.34. COROLLARY (Bicompleteness of $PL(=(\cdot))$). $PL(=(\cdot))$ is bicomplete for

 $\{(\mathcal{P}, \mathcal{Q}) | \mathcal{P}, \mathcal{Q} \text{ are downward closed and have the empty team property;}$ $\mathcal{P} \text{ and } \mathcal{Q} \text{ are } G\text{-}C/G\text{-}C \text{ mod } \mathbb{L}/W\text{-}I\},$

and hence bicomplete for pairs which are G-C/G-C mod \parallel /W-I.

Note that by Corollaries 4.3.6 (ii) and 4.3.34, we have $\|\text{InqB}\| = \|\text{PL}(=(\cdot))\|$, but $\|\text{InqB}\|^{\pm,\neg_i} \neq \|\text{PL}(=(\cdot))\|^{\pm,\neg}$.

To conclude, let us note that it is trivial to find a propositional logic which, like D in the first-order setting, is bicomplete for \perp -incompatible pairs, \emptyset -incompatible pairs, and ground-incompatible pairs: for instance, one could extend $PL(=(\cdot))$ with a constant θ_0 such that $s \models \theta_0 \iff s = \theta_0 \iff s = \emptyset$. It is not clear whether there is some interesting and non-*ad hoc* propositional logic bicomplete for these pairs.

4.4 Interpretations of the Incompatibility Notions

In this section, we provide sketches of some possible intuitive interpretations of the incompatibility notions, drawing mainly on [116, 123, 129, 130, 42, 25, 107, 38, 35, 67, 6, 45]. (I am particularly indebted to Yalcin [129, 130]; I borrow both from his analysis as well as his perspicuous explanations.) Our main focus will be on the relationship between \perp -incompatibility and ground-incompatibility (Facts 4.2.13 and 4.2.14), and the related notions of \parallel -incompatibility and NE-incompatibility (and Fact 4.2.16), although we will also comment briefly on some of the other notions. Our discussion must be brief so we ignore many potential complications¹⁹; the aim is provide some basis for further interpretations, and to give the reader some intuitive grasp on our technical notions.

4.4.1 Information States; Factual Information; Persistency

A common interpretation of team logics is one in which possible worlds/propositional valuations represent possible states of affairs, and teams represent information states. A valuation determines a collection of formulas which are true according to the valuation, and this collection stands in for the state of affairs in which precisely whatever is represented by the formulas in the collection obtains. A set of these valuations, or a propositional team (see Section 4.3), then corresponds to an information state in that if one knows that the valuation representing the actual world is one of the worlds in a team s, one knows that whatever holds in all worlds in s must be the case. For instance, if it is raining in all worlds in s, the information embodied in s includes the information that it is raining. A modal team also incorporates information about what is necessary/possible (or obligatory/permissible, etc.) given that the actual world is in the state. The empty information state is a state of absurdity—a state which establishes inconsistent information.

¹⁹Note in particular that while much of our presentation follows that of Yalcin [129, 130], we make no principled distinctions between the *compositional semantic value* of a linguistic expression and its *informational content* or its *object of assertion*; these distinctions (see, for instance, [129, 130, 131, 67]) are key for Yalcin. We use the terms 'information' and 'content' to refer, essentially, to (possibly pragmatically enriched) compositional semantic values.

Each formula determines a constraint on information states which we may think of as the information or content expressed by the formula: a state satisfies the constraint corresponding to a given formula just in case the team which represents the state supports the formula. A classical formula determines a constraint on possible worlds in a similar fashion; the constraint it determines on states is parasitic on this constraint on worlds: each world in the state must satisfy the constraint on worlds (as per Fact 4.2.4). The type of information expressed by classical propositional formulas—call this factual information-is, then, information concerning which facts (as captured by valuations) hold in a given world or according to a given information state. Similarly, classical modal formulas express information concerning which facts hold at accessible worlds (modal factual information). As we have seen, classical formulas are downward closed. Downward closure may be thought of as corresponding to the fact that as one becomes more informed (by ruling out certain possibilities), then, at least as far this type of information is concerned, one does not lose the information one has already acquired. (Downward closure and analogous notions are, accordingly, often called *persistency* [35, 123, 25, 67].) If s supports the assertion that it is raining, moving to a subteam t of s by ruling out some possible worlds does not erase this information: t also supports the assertion that it is raining.

This is the basic picture. It already allows us to indicate in broad strokes what some of our notions are expressing. ψ is a down-set incompatibility of φ just in case one whenever an information state establishes that ψ ($s \models \psi$), φ is ruled out in that there is no further information or content that could consistently establish φ (there is no $t \subseteq s$ such that $t \neq \emptyset$ and $t \models \varphi$). Trivially, φ and ψ are team-incompatible if one is established in a state iff the other is not—if the constraint on states determined by ψ is (expressible as) the classical negation (on states) of that determined by φ , and vice versa. Analogously for world-incompatibility and constraints on worlds. φ and ψ are flat-incompatible (φ and ψ are W-I, and both formulas are flat) just in case the world-constraint determined by one is the classical negation (on worlds) of the worldconstraint determined by the other (W-I); and they both express essentially factual information: the state-constraints they determine are parasitic on their world-constraints (flat).

However, for most of the other notions we consider, it is helpful, before providing interpretations of these notions, to first formulate what we take properly state-based constraints to represent—what is the nature of the *non-factual information/content* which is not expressible by classical formulas? In Section 4.2.2, we observed that \perp -incompatibility and ground-incompatibility are equivalent for downward-closed formulas, but when downward closure fails, there are some pairs which are \perp -incompatible but which fail to be ground-incompatible (Facts 4.2.13 and 4.2.14). We additionally noted that each \perp -incompatible pair is either \parallel -incompatible or NE-incompatible, and that NE-incompatibility entails ground-incompatibility in a convex setting (Fact 4.2.16). There are, accordingly, two types of pairs which are \perp -incompatible but not ground-incompatible: \parallel -I but not G-I, and NE-I but not G-I, with the latter type only possible when convexity is violated. We will set as our goal, in the remainder of this

section, to augment the basic picture with interpretations of non-classical formulas in a way that allows us to make intuitive sense of these facts. In Section 4.4.2, we focus on \parallel -I but not G-I pairs. We assume a convex setting, and examine an expressivist epistemic interpretation for formulas which fail downward closure (due essentially to Yalcin [129, 130] as well as Hawke and Steinert-Threlkeld [67]); on this interpretation, \parallel -I but not G-I pairs represent *epistemic contradictions*. We also comment briefly on what this interpretation tells us about G-C (mod \parallel) pairs. In Section 4.4.3, we again assume convexity, and consider a pragmatic interpretation (essentially due to Aloni [6]) which leads to \parallel -I but not G-I pairs being *pragmatic contradictions*. In Section 4.4.4, we build on the interpretation of Section 4.4.2 to also provide an account of NE-I but not G-I pairs. We adapt the interpretation of the inquisitive disjunction \bowtie and the notion of *inquisitive content* from inquisitive semantics [38, 35], and conclude that, assuming the context of the rest of our interpretations, NE-I but not G-I pairs are examples of a specific type of contradiction with inquisitive content.

4.4.2 Expressivist Epistemic Information; Factual Contradictions vs. Epistemic Contradictions

(We work in the convex setting of BSML in this section.) Not all types of information need be persistent. Define an *epistemic might* operator \diamond as follows: $\diamond \varphi := (\varphi \land NE) \lor \top$. Then is $\diamond \varphi$ supported by a team just in case the team contains a nonempty subteam supporting φ . This operator can be used to represent the 'might' in sentences such as 'It might not be raining': if there is a nonempty subteam *t* of *s* consisting of worlds in which it is not raining ($s \models \diamond \neg r$), then, for all that one knows given the information embodied in *s*, it might not be raining. The information that something might be the case *can* come to be invalidated upon receipt of further information: if one moves from *s* to a subteam *u* consisting only of worlds in which it is raining (that is, say, if one's information state supports the assertion that it might not be raining, and one then learns that it is raining), $\diamond \neg r$ is no longer supported. (For similar accounts in the literature, see, e.g., [129, 130, 25, 107, 67]; it seems the first account along these lines was Veltman's [123].)

Whereas the type of constraint on information states corresponding to factual information is local in the sense that it is parasitic on a constraint on worlds, the type of constraint on information states determined by $\bullet \varphi$ is global: the state as a whole must have a non-empty φ -substate. The constraint also pertains only to information states, rather than to both information states and worlds: it does not seem to make sense, for instance, to speak of a (fully determined) state of affairs in which might be raining. In this sense, the type of information expressed by $\bullet \varphi$ is non-factual: it does not serve to demarcate the collection of possible states of affairs in which some specific facts hold, and an assertion of $\bullet \varphi$ does not purport to (directly) help one's interlocutors better situate the actual world in the space of possibilities; rather, it only indicates that whatever state of information conforms to it is such that φ cannot be ruled out

(which state this is taken to apply to—often the speaker's information state—is determined pragmatically via the context). Following Yalcin's [129, 130] and Hawke and Steinert-Threlkeld's [67] expressivism about epistemic modality, let us call the type of information expressed by $\bullet \varphi$ expressivist epistemic information. Compare 'It might be raining' $\bullet r$ with 'For all I know, it is raining', which we might formalize, using an epistemic Kripke modality \diamond_a tracking what is compatible with what the utterer agent a knows in the usual way, as $\diamond_a r$. The latter (expressing what we can call factual epistemic information) does directly express a constraint on worlds: it excludes all worlds in which a's epistemic possibilities do not include r-worlds.²⁰ Indeed, the information that for all agent a knows, it is raining is persistent—it is for instance not invalidated by the information that it is not raining. One might pragmatically infer, from agent a's utterance of (something formalized as) $\bullet r$, that $\diamond_a r$, but the formula only directly serves to describe a constraint on information states. See [129, 130] for more on this distinction.

In this interpretation we have, then, factual information expressible using classical (downward closed) formulas and expressivist epistemic information expressed using \bullet (which breaks downward closure). In this context, we may think of the ground team $|\varphi|$ of φ as representing the factual information expressed by φ : the ground-team of φ is a classical proposition which is true precisely in all the possible worlds which compose the information states in which φ in assertible, and which does not directly communicate any expressivist epistemic information.²¹ 'It might be raining' expresses no factual information since $|\bullet r| = |\top|$ —knowing that it might be raining gives one no information as to whether it actually is raining. Similarly, the factual information expressed by 'it is not hailing but it might be raining' $\neg h \land \bullet r$ is simply that it is not hailing: $|\neg h \land \bullet r| = |\neg h|$.

We may, accordingly, think of ground-incompatibility as contradiction in factual

²⁰Note that epistemic puzzles such as the muddy children puzzle [17] turn on the factuality of the type of information expressed by formulas of the form $\neg \Box_a \varphi$.

²¹The ground team can be thought of as indirectly communicating expressivist epistemic information via the modal operators. For instance, consider $\Diamond(\bullet r \land \bullet h)$. The ground team $|\Diamond(\bullet r \land \bullet h)|$ excludes all worlds which do not see an information state that supports $\bullet r \land \bullet h$ —all worlds in which it is not possible (in whatever way \diamond represents) that it simultaneously might (expressivist-epistemically) be the case that r and might be the case that h. In the modal setting, the ground team therefore communicates not only factual information concerning what is the case in the actual world and factual modal information concerning what is factually the case in accessible worlds, but also factual (in the sense of defining a constraint on possible worlds) modal information concerning what might (expressivist-epistemically) be the case in modally accessible states. This is essentially because the semantics of the modalities we employ make use of state-based support and anti-support conditions, and so they are sensitive to constraints on states (such as those generated by \bullet), but the constraints the modalities themselves define are constraints on worlds. To make some intuitive sense of the formula above, assume we are defining rules for a system which deals in weather probabilities using a modality \diamond . We can then take, say, $\neg \diamond (r \land \neg c)$ to describe a rule to the effect that no information state of the system in which it is raining and there are no clouds is permissible/possible, whereas $\diamondsuit(\bullet r \land \bullet h)$ states that being in a state according to which it might rain and it might hail is permissible (and grants the system factual information concerning which kinds of information states are accessible from the actual world).

information: if φ and ψ are ground-incompatible ($|\varphi| \cap |\psi| = \emptyset$), the factual information expressed by φ rules out that expressed by ψ , and vice versa. On the other hand, it seems that a *i*-incompatibility that is not also a ground-incompatibility (which, assuming we are in a convex setting, must be a *1*-incompatibility by Fact 4.2.16) such as the pair r, $\bullet \neg r$ (we have $r, \bullet \neg r \models \bot \models \bot$ and $|r| \cap |\bullet \neg r| = |r| \neq \emptyset$) would then have to be a contradiction involving expressivist epistemic information, or an epistemic contra*diction* in the sense of [129, 130]. Indeed, $r \wedge \bullet \neg r$ corresponds to the classic epistemic contradiction 'it is raining and it might not be raining', and other typical examples similarly clearly constitute pairs which are *I*-I but not G-I.²² We then have the following intuitive explanation for Facts 4.2.13 and 4.2.14: G-I-factual contradictorinessand \perp -I—contradictoriness in some more general sense—come apart when expressive epistemic information is admitted since in addition to conflicts in purely fact-based constraints we now have also conflicts also involving expressive epistemic constraints, which include the typical examples of epistemic contradictions. In a setting with only factual, downward-closed information the latter type of conflict does not arise, so G-I and \perp -I coincide.

Given the association between the failure of downward closure and expressive epistemic information, one might be tempted to further stipulate that two formulas are in epistemic contradiction precisely when they are $\mathbb{1}$ -incompatible (setting aside formulas which are themselves equivalent to $\mathbb{1}$, the only way two formulas can be $\mathbb{1}$ -I is if one of them is not downward closed). Note, however, that $r \land \blacklozenge q, \neg r \models \mathbb{1}$, yet clearly with this pair the incompatibility arises due to r and $\neg r$ being factually contradictory. Perhaps the most we can say is that if a pair is $\mathbb{1}$ -I, the pair is contradictory and at least one of the formulas involved expresses expressivist epistemic information.

To conclude this section, let us also comment briefly on what it means for two formulas φ and ψ to be G-C ($|\varphi| = |T| \setminus |\psi|$ and $|\psi| = |T| \setminus |\varphi|$) (or G-C mod \bot) in the context of this interpretation. We can say that either one of the formulas is (if it is equivalent to \bot), by itself, a contradiction that incorporates expressive epistemic information in some way, or that the factual information expressed by the two formulas is not only contradictory, but also classically complementary: there is a classical formula α such that α expresses the same factual information as φ , and $\neg \alpha$ expresses the same factual information as ψ .

We can also say, given Lemmas 4.3.17 and 4.3.29, that, in the logics PL(NE) and PL(=(·)) (as long as φ is not equivalent to \bot) the flattening φ^f of φ can properly be regarded as the classical variant of φ (as per the name given to the analogue of flatten-

²²Compare epistemic contradictions with Moorean sentences such as 'It is raining and I do not know that it is raining', which we might formalize as $r \land \neg \Box_a r$. This sentence entails a ground-incompatibility on the pragmatic assumption that if agent *a* asserts *r*, they are also thereby asserting $\Box_a r$. It is therefore, by our lights, not a contradiction involving expressive epistemic information, but rather a kind of pragmatic factual contradiction, dovetailing Yalcin's [129, 130] distinction between epistemic contradictions and Moorean sentences. One might also arguably pragmatically infer $r \land \neg \Box_a r$ from *a*'s utterance of $r \land \blacklozenge \neg r$, so one might argue that $r \land \blacklozenge \neg r$ involves the same kind of pragmatic contradiction as $r \land \neg \Box_a r$ in addition to being an epistemic contradiction.

ing in inquisitive logic; see endnote 10) in that φ and φ^f express the same information regarding valuations/worlds. Propositions 4.3.19 and 4.3.30, then, say that in PL(NE) and PL(=(·)) the dual negation behaves classically at least to the extent that (as long as $\varphi \neq \bot$ and $\psi \neq \bot$) the classical variant of $\neg \varphi$ is simply the classical negation of the classical variant of φ , and on the level of valuations/worlds, there is no failure of determination.

4.4.3 Pragmatic Information via the Neglect-zero Tendency; Factual Contradictions vs. Pragmatic Contradictions

(We again work in BSML.) Let us consider another possible type of non-downwardclosed information. Aloni and her collaborators [6, 45] use BSML to account for *modal disjunction inferences* such as the following:

a. Sue went to the beach or to the cinema.

The speaker deems it possible that Sue went to the beach and the speaker deems it possible that Sue went to the cinema.

b. b∨c → ◊_sb ∧ ◊_sc

Aloni's proposal is that speakers, when interpreting language, create mental structures that represent reality; and that inferences like (1) are due to a tendency in human cognition to disregard empty structures in this process of interpretation (the *neglectzero tendency*). This tendency is modelled using a *pragmatic enrichment function* $[]^+: ML \rightarrow BSML$ which, given a classical formula α , recursively appends $\wedge NE$ to each subformula of α (for instance, $[b \lor c]^+ = ((b \land NE) \lor (c \land NE)) \land NE)$). One can then check that, for instance, the team $\{w_b\}$ is such that $\{w_b\} \models b \lor c$ but $\{w_b\} \notin [b \lor c]^+$ the team only supports $b \lor c$ by virtue of the fact that the empty team \emptyset supports c, whereas it does not support $[b \lor c]^+$ since pragmatic enrichment rules out empty verifying structures (*zero-models*).

The modal disjunction inference above can then be accounted for as follows. We model the speaker's information state using a pointed model (M,t) with an accessibility relation R_s corresponding to a modality \diamond_s tracking what is epistemically possible given the speaker's knowledge. R_s must be such that for all $w \in t$, R[w] = t (in Aloni's terminology, R_s is *state-based* in t)—this tracks the facts that R_s is intended to be an epistemic accessibility relation, and that t is the speaker's information state (note that if R_s is state-based in (M,t), then $M,t \models \diamond_s \varphi$ iff $M,t \models \bullet \varphi$). In such a pointed model, $[b \lor c]^+$ is assertible just in case $\diamond_s b \land \diamond_s c$ is as well—restricting entailment to pointed models with state-based accessibility relations R_s , we have $[b \lor c]^+ \models \diamond_s b \land \diamond_s c$.

Formulas which are not pragmatically enriched (these are all classical formulas downward-closed formulas with the empty team property), then, express information which does not incorporate any pragmatic component of the kind encoded by the pragmatic enrichment function, whereas pragmatically enriched formulas (these are non-downward-closed formulas without the empty team property) express information which does have this kind of component. This *pragmatic information* is global and non-factual in a similar way as the expressive epistemic information of the previous section.

A sentence such as

a. Sue went to the beach or to the cinema, but she did not go to the cinema.
b. (b∨c) ∧¬c

could, then, be classified as a *pragmatic contradiction*, or a contradiction involving pragmatic content, in that there is no contradiction on the non-pragmatic level— $(b \lor c) \land \neg c \nvDash \bot$ —but there is one when pragmatic content is taken into account: $[(b \lor c) \land \neg c]^+ \vDash \bot$ —but there is one when pragmatic content is taken into account: $[(b \lor c) \land \neg c]^+ \vDash \bot$ Many pragmatic contradictions, on this definition, (including the present example) involve pairs that are \bot -I but not G-I (and hence \bot -I, assuming convexity), and, assuming a setting with only ML-formulas and pragmatically enriched formulas, all such pairs must clearly be pragmatic contradictions. When no pragmatic enrichment is present (when all formulas are downward closed and have the empty team property), there are no pragmatic contradictions, so ground-incompatibility and \bot -incompatibility coincide (Fact 4.2.14). We can think of ground-incompatibility as corresponding to contradiction in factual information as before, and of \bot -incompatibility without ground-incompatibility as contradiction involving pragmatic information.

Note, however, that on this definition of pragmatic contradiction, there are some pragmatic contradictions in BSML involving ground-incompatible and hence factually contradictory pairs. This type of contradiction can be generated, for instance, by the type of sentence involved in *free choice inferences* (see, e.g., [128, 87]) such as the following:

(3) a. You may go to the beach or to the cinema.
 → You may go to the beach and you may go to the cinema.
 b. ◊(b∨c) → ◊b ∧ ◊c

As with modal disjunction inferences, Aloni [6] accounts for inferences such as (3) using pragmatic enrichment—the following entailment holds: $[\diamondsuit(b \lor c)]^+ \vDash \diamondsuit b \land \diamondsuit c$ (where \diamondsuit is deontic).

Now consider:

- (4) a. You may go to the beach or to the cinema, but you may not go to the cinema.
 - b. $\diamondsuit(b \lor c) \land \neg \diamondsuit c$
 - c. $[\diamondsuit(b \lor c) \land \neg \diamondsuit c]^+$

By our definition, this is a pragmatic contradiction: $\Diamond(b \lor c) \land \neg \Diamond c \nvDash \bot$ and $[\Diamond(b \lor c) \land \neg \Diamond c]^+ \vDash \bot$ however, the relevant pair in this case is a ground-incompatibility $(|[\Diamond(b \lor c)]^+| \cap |[\neg \Diamond c]^+| = \emptyset;$ this is because if $w \in s \vDash [\Diamond(b \lor c)]^+$, then there must be a $t \subseteq R[w]$ such that $t \vDash c \land NE$, but if also $w \in u \vDash [\neg \Diamond c]^+$, we must have $R[w] \vDash \neg c$, conflicting with the existence of a *t* as above), indicating, by the lights of our

interpretation, that it involves a contradiction in factual information.

To clarify, our interpretation gives rise to the following characterization of this situation: the pair in question is a contradiction in factual information (in the sense we have been considering: arising from a conflict in constraints on possible worlds), with the factual information in question engendered by pragmatic enrichment (and so it is in this sense a pragmatic contradiction). The modalities in BSML are sensitive to restrictions on states such as those generated by pragmatic enrichment, but they define constraints on worlds (see endnote 19). (4) c defines a constraint on worlds that involves restrictions (which incorporate some pragmatic content) on the kinds of states that can be deontically accessible from a world if it is to meet the constraint. The restrictions on accessible states are mutually exclusive due to the pragmatic content (we must have both $\exists t \subseteq R[w] : t \models c \land NE$; and $R[w] \models \neg c$; it seems to the interpreter that the speaker is simultaneously claiming that there are non-empty accessible cinema-states, and that there no non-empty accessible cinema-states), and so the factual constraint is impossible to meet; therefore (4) c is a factual contradiction.

In this example, we also have NEs outside the scope of the modalities, and hence (4) c also places a non-factual pragmatic constraint on information states. This results in (4) c being not only a factual contradiction (a ground-incompatibility), but also a contradiction involving pragmatic information in some way (a \bot -incompatibility; cf. the pair $r \land \bullet q, \neg r$ in Section 4.4.2). In what way? One way to conceive of NE as it figures in pragmatically enriched formulas is as an explicit marker of the pragmatic understanding that speakers usually represent themselves as not uttering contradictions (in Aloni's [6] succinct formulation, 'avoid \bot '). The interpreter applies this understanding to all structures involved in interpreting (4), and, as we have just seen, applying it in the scope of the modals leads to the conclusion that the speaker has expressed something contradictory: $\diamondsuit((b \land NE) \lor (c \land NE)) \land \neg \diamondsuit(c \land NE) \models \bot$. Applying 'avoid \bot ' outside the scope of the modals then engenders a further pragmatic violation: the speaker is supposedly presenting themselves as uttering no contradictions ($\land NE$), but they just expressed a contradiction (\bot). We may think of this further pragmatic violation as being represented by \bot .

4.4.4 Contradictions with Inquisitive Content

How about NE-I but not G-I pairs such as $\neg p \lor ((p \land NE) \lor (\neg p \land NE))$ and p? By Fact 4.2.16, such pairs require violation of convexity—in order to consider these pairs, we therefore now move to the non-convex setting of PL(NE, \lor) (we omit the modalities to simplify our discussion; we also assume that all formulas are in negation normal form, again for the sake of simplicity). We extend the picture of Section 4.4.2 by adapting the interpretation of \lor from inquisitive logic/semantics [38, 35] to provide an interpretation of NE-I but not G-I pairs.

Inquisitive semantics provides an account of the meanings of questions using teams (information states). On this account, while to understand the meaning of a declarative sentence (or *statement*) is to understand when the sentence is true—to know its truth

conditions—to understand the meaning of a question is to understand what information it would take to resolve the question—to know its *resolution conditions*. Truth conditions and resolution conditions may both be represented using support conditions: a statement is supported by s if it is true in s, while a question is supported by s if it is resolved in s (s supporting φ may be thought of as s settling or establishing φ in a general sense applicable to both statements and questions).

Each resolution to a question is taken to be expressed by a statement, so the resolution conditions of a question are of the form: one of the statements expressing a resolution to this question is true. This is modelled using the global/inquisitive disjunction: $p \vee \neg p$ corresponds to the question whether p or $\neg p$ is the case; it is resolved (supported) just in case either p is true (supported) or $\neg p$ is true. Intuitively, the characteristic which distinguishes questions from statements on this account is that whereas a statement can only be settled in one way in an information state (a statement is settled in *s* just in case it is true according to the information in *s*), a question can be settled in multiple ways, with each distinct resolution to a question providing a distinct way of settling the question (if an expression does not raise the question as to which of its resolutions is true—if it has only one resolution—it is not a question).

Formally, we recursively define the set of *resolutions* $\mathcal{R}(\varphi)$ of $\varphi \in PL(NE, W)$ in negation normal form as follows: $\mathcal{R}(\psi) \coloneqq \psi$ if ψ is an atom or a negated atom; $\mathcal{R}(\psi \land | \lor \chi) \coloneqq \{\psi' \land | \lor \chi' \in \mathcal{R}(\psi), \chi' \in \mathcal{R}(\chi)\}$; $\mathcal{R}(\psi \lor \chi) \coloneqq \mathcal{R}(\psi) \cup \mathcal{R}(\chi)$. Note that clearly each resolution is W-free (and hence union closed and convex); and that we allow for equivalent but distinct resolutions. One can show that:

4.4.1. PROPOSITION. For each $\varphi \in PL(NE, W)$ (in negation normal form), $\varphi \equiv W/\mathcal{R}(\varphi)$.

Proof:

Follows from the fact that \land and \lor distribute over \lor —see [7, 35].

We will say that φ is a *statement* if $|\mathcal{R}(\varphi)| = 1$, and that it is a *question* if $|\mathcal{R}(\varphi)| > 1$. We now have, in addition to informative content (comprising both factual information and expressive epistemic information) also *inquisitive content*: φ has inquisitive content just in case it is a question. Say that φ has *trivial inquisitive content* if it is a union-closed and convex question, and that it has *non-trivial inquisitive content* if it is a question that is either not union closed or not convex (examples of questions with trivial inquisitive content: $p \lor (p \land (q \lor \neg q)); (p \land q) \lor (p \land NE)$). (These definitions are based on the fact that union-closed and convex formulas are equivalent to ones which do not contain the question-forming connective \lor (Theorems 4.2.8 (i) and 4.3.13 (i)); we therefore take it that they do not express genuine inquisitive content. On the other hand, if we assume that \lor only contributes inquisitive content, any formula not expressible in the fragment of the language not containing \lor must express non-trivial inquisitive content.)²³

²³The notions introduced in Section 4.4.4 are adapted from notions used in inquisitive logic, but many of our definitions (including those of statement and question) depart from those most commonly

We will take it that the ground team $|\varphi|$ of φ still expresses the factual information conveyed by φ —in effect, moving from $||\varphi||$ to $|\varphi|$ now cancels both inquisitive content and expressive epistemic information. The factual information expressed by a question φ amounts to the constraint that in each world forming part of an information state in which φ is resolved, (the factual component of) at least one of the resolutions of φ is true: by Proposition 4.4.1, for each $\varphi \in PL(NE, W)$ (in negation normal form), $|\varphi| = |W\mathcal{R}(\varphi)| = \bigcup_{\varphi' \in \mathcal{R}(\varphi)} |\varphi'|$, so $w \in |\varphi|$ iff $w \in |\varphi'|$ for some $\varphi' \in \mathcal{R}(\varphi)$. This can be thought of as representing the fact that the question as to which element of $\mathcal{R}(\varphi)$ is true would seem to presuppose that at least one of the elements of $\mathcal{R}(\varphi)$ is indeed true—the factual information expressed by a question is the factual information that the question presupposes.

Now note that:

4.4.2. FACT. If $\varphi, \psi \in PL(NE, W)$ are NE-I and not G-I, then

- (i) $\phi \wedge \psi$ has trivial inquisitive content;
- (ii) for each $\varphi' \land \psi' \in \mathcal{R}(\varphi \land \psi)$, the pair φ', ψ' is either G-I or \bot -I; and
- (iii) there is at least one $\varphi' \land \psi' \in \mathcal{R}(\varphi \land \psi)$ such that φ', ψ' are G-I and not \bot -I, and at least one $\varphi' \land \psi' \in \mathcal{R}(\varphi \land \psi)$ such that φ', ψ' are \bot -I and not G-I.

Proof:

(i) First assume for contradiction that $\varphi \land \psi$ is not a question. Then $|\mathcal{R}(\varphi \land \psi)| = 1$, whence also $|\mathcal{R}(\varphi)| = |\mathcal{R}(\psi)| = 1$. By Proposition 4.4.1, $\varphi \equiv \bigvee / \mathcal{R}(\varphi)$. Since $|\mathcal{R}(\varphi)| = 1$, φ is simply equivalent to the only element of $\mathcal{R}(\varphi)$, and is therefore convex. Similarly for ψ . Then by Fact 4.2.16, φ and ψ cannot NE-I without being G-I, contradicting our assumptions. Therefore, $\varphi \land \psi$ must be a question. Since φ, ψ are NE-I we have $\varphi \land \psi \equiv \bot$, so $\varphi \land \psi$ is union-closed and convex, and hence has trivial inquisitive content.

(ii) First assume for contradiction that for some $\varphi' \land \psi' \in \mathcal{R}(\varphi \land \psi)$, φ', ψ' are not \bot -I. Then there is a non-empty team *s* such that $s \models \varphi' \land \psi'$. But then by Proposition 4.4.1, also $s \models \varphi \land \psi$, contradicting the fact that φ and ψ are NE-I. So we must have that φ', ψ' are \bot -I. Then φ', ψ' are either NE-I or \bot -I. Since $\varphi' \in \mathcal{R}(\varphi)$ and $\psi' \in \mathcal{R}(\psi)$, these formulas are convex; therefore, if they are NE-I, they must also be G-I by Fact 4.2.16.

(iii) Since φ, ψ are not \bot -I, there is some *s* such that $s \models \varphi \land \psi$. By Proposition 4.4.1, also $s \models \bigvee \mathcal{R}(\varphi \land \psi)$, so $s \models \varphi' \land \psi'$ for some $\varphi' \land \psi' \in \mathcal{R}(\varphi \land \psi)$. But then φ', ψ' are not \bot -I, and by (ii) they are G-I.

used in inquisitive logic. One reason for this is that the definitions used in inquisitive logic typically assume a downward-closed setting; another is that providing a syntactic rather than semantic criterion for questionhood and considering semantic questionhood separately via the distinction between trivial and nontrivial inquisitive content makes it easier to formulate our characterization of NE-I but not G-I pairs. It may also be surprising that we have not introduced a notion of *alternatives*. The standard treatment of alternatives assumes downward closure; it should be possible to formulate an analogue of this notion which works in our non-downward setting, but this is orthogonal to our aims here.

On the other hand, since φ, ψ are not G-I, there is some $w \in |\varphi| \cap |\psi|$, so that $w \in t \models \varphi$ and $w \in u \models \psi$ for some *t* and *u*. By Proposition 4.4.1, $t \models \bigvee \mathcal{R}(\varphi)$ and $u \models \bigvee \mathcal{R}(\psi)$, so for some $\varphi'' \in \mathcal{R}(\varphi)$ and some $\psi'' \in \mathcal{R}(\psi)$ we have $t \models \varphi''$ and $u \models \psi''$. But then we have $\varphi'' \wedge \psi'' \in \mathcal{R}(\varphi \wedge \psi)$, and $w \in |\varphi''| \cap |\psi''|$ so that φ'', ψ'' are not G-I. By (ii), φ'', ψ'' must be \bot -I.

We therefore have the following characterization of NE-I but not G-I pairs such as $p \lor \diamond q, \neg p \land \neg q$. At least one member of the pair must be a question (otherwise both are convex, contradicting Fact 4.2.16). Settling both elements of the pair simultaneously (that is, resolving the member that is a question, and either resolving or establishing the truth of the other member) in a consistent manner—in a non-empty information state—is not possible because the pair is 1-I. This is because each possible way of resolving both members simultaneously (each $\varphi' \land \psi' \in \mathcal{R}(\varphi \land \psi)$) is either a factual contradiction (G-I and not 11-I), or an epistemic contradiction (11-I and not G-I). But it is not the case that the factual information expressed by either of the members rules out the other (the pair is not G-I), or that the conflict between the pair is solely expressiveepistemic in nature (there is some $\varphi' \land \psi' \in \mathcal{R}(\varphi \land \psi)$ such that φ', ψ' are G-I and not 11-I). Rather, simultaneous settling of the pair involves resolving inquisitive content $(\phi \land \psi$ is a question), with some possible resolutions being factual contradictions and some possible resolutions being epistemic contradictions. The pair expresses inquisitive content in that we can formulate ways of simultaneously settling both members (the resolutions of $\varphi \land \psi$) and there are more than one of these ways (again, $\varphi \land \psi$ is a question by our syntactic definition), but the inquisitive content is ultimately trivial because the pair is contradictory: simultaneous settling in a consistent manner is not possible, and there is (semantically) only one way of resolving $\phi \land \psi$ —by being in the inconsistent state \emptyset .

We also present a toy natural language example for concreteness, although we concede that our formalization of this example is potentially controversial. Consider the following question-answer pair: 'Is Mary already home, or is it possible that she is still at the airport?' 'She's on the train.' We formalize these as $h \vee \bullet a$ and $\neg h \wedge \neg a$, respectively. This pair of formulas is an NE-incompatibility, and the resolutions of $(h \vee \bullet a) \wedge (\neg h \wedge \neg a)$ are $h \wedge (\neg h \wedge \neg a)$ and $\bullet a \wedge (\neg h \wedge \neg a)$. The first resolution $h \wedge (\neg h \wedge \neg a)$ involves a G-I (and not \bot -I) pair, and hence a factual contradiction: it is not possible to simultaneously resolve the question with the answer that Mary is at home, and to accommodate the information that Mary is on the train (and hence not at home or at the airport) because this would involve a contradiction in factual information. The second resolution $\bullet a \wedge (\neg h \wedge \neg a)$ involves a \bot -I and not G-I pair and hence an epistemic contradiction: it cannot be the case, according to any information state, both that Mary might be at the airport and that she is on the train. The pair is therefore contradictory (\bot -I)—one cannot consistently simultaneously resolve the question as presented and accept the statement.

4.5 Concluding Remarks

In this paper, we proved analogues of Burgess' theorem for BSML and BSML^{\vee}, for the propositional fragments of these logics, for Hawke and Steinert-Threlkeld's semantic expressivist logic for epistemic modals, as well as for propositional dependence logic with the dual negation. We saw that the notion of incompatibility employed to secure a Burgess theorem had to be adjusted according to the logic in question, formulated notions suitable for each of the logics we considered, and examined the relationships between these notions. We defined the notion of bicompleteness to succinctly describe our results: a logic is bicomplete with respect to a class of pairs of the relevant sort and conforming to a particular incompatibility notion just in case both a Burgess theorem employing that incompatibility notion as well as its converse hold for the logic. We also applied the notion of bicompleteness to logics which do not exhibit failure of determination of negated meanings by positive meanings, and gave an example of a logic which is bicomplete for all pairs.

The incompatibility notions and bicompleteness are interesting in their own right, and might warrant further study. We conclude with some final remarks on these notions, and on the interpretation and potential uses of our results.

We noted in Section 4.1 that Burgess characterized his theorem as concerning the degree of failure of contrariness/the dual negation (in H_p , and equivalently in IF and in D) to correspond to any operation on classes of models. One can think of our bicompleteness results and incompatibility notions as giving us a way of measuring this degree in different logics, with bicompleteness with respect to a weaker notion of incompatibility corresponding to a stronger degree of failure. On one end of the scale we have the classical PL, which exhibits no such failure-the class of teams $\|\varphi\|$ on which φ is true determines that $\|\neg\varphi\|$ on which $\neg\varphi$ is true since, as per flat-incompatibility, $\|\neg \phi\| = \wp(|\top| \setminus |\phi|)$ —moreover, $\|\neg \phi\|$ also determines $\|\phi\|$: $\|\varphi\| = \wp(|T| \setminus |\neg \varphi|)$. To get bicompleteness for IngB, we must go from the stronger flat-incompatibility to the weaker D-I $\|\neg \varphi\| = \{s \mid [t \subseteq s \text{ and } t \models \varphi] \implies t = \emptyset\} \longrightarrow \|\varphi\|$ still determines $\|\neg \phi\|$, but the converse is no longer true. Further along the scale, we have mild failure of determination: for HS and E-D-I, it is not the case that $\|\boldsymbol{\varphi}\|$ always determines $\|\neg \phi\|$ or vice versa, but it is always the case that one of the two determines the other. Propositional dependence logic violates determination in a different way with ground-complements $|\neg \phi| = |\top| \setminus |\phi|$. Neither $\|\phi\|$ nor $\|\neg \phi\|$ determines the other, but $\|\varphi\|$ does determine the ground team $|\neg \varphi|$ of the negation, and similarly for $\|\neg \varphi\|$ and $|\varphi|$. Even further along we have first-order dependence logic D, and we must now weaken ground-complements to \perp -incompatibility $\varphi, \neg \varphi \models \bot$. Properties and ground teams are no longer determined, but we do at least know that $\|\varphi\|$ and $\|\neg\varphi\|$ are disjoint modulo the empty team. At the very far end of the scale we have $PL(NE^*, \forall)$, bicomplete for all pairs: $\|\varphi\|$ does not constrain $\|\neg\varphi\|$ in any way.

We saw that dual negations in seemingly similar logics can conform to very different incompatibility notions. For instance, whereas BSML is bicomplete for groundincompatible pairs—pairs $(\mathcal{P}, \mathcal{Q})$ in which \mathcal{P} places only a very weak constraint on \mathcal{Q} $(\bigcup \mathcal{P} \cap \bigcup \mathcal{Q} = \emptyset)$ —simply removing the modalities gives us a logic (PL(NE)) which is bicomplete for pairs which are ground-complements modulo \bot —pairs (\mathcal{P}, \mathcal{Q}) such that, as long as $\mathcal{P} \neq \emptyset \neq \mathcal{Q}$, all the valuation-level information concerning \mathcal{P} can be recovered from \mathcal{Q} ($\bigcup \mathcal{P} = |\top| \setminus \bigcup \mathcal{Q}$). Similarly, whereas first-order dependence logic D is bicomplete for \bot -incompatible pairs, the natural propositional analogue PL(=(·)) is bicomplete for ground-complementary pairs.

On the one hand, these types of results are to be expected whenever the semantics of the negation $\neg \phi$ of ϕ are defined in the bilateral way we have been observing, with the support conditions of $\neg \phi$ not depending in a uniform manner on the positive semantic value $\|\varphi\|$ of the argument φ , but depending, rather, on the antisupport conditions specific to the main connective or atom of φ . The pair properties that a given logic conforms to are a function of each of the anti-support clauses; if new atoms or connectives are added, their anti-support clauses must be configured to guarantee or bring about whatever properties are desired (for instance, switching out NE, in PL(NE, W) with the atom NE^{*}, whose semantics clearly do not conform to any of our incompatibility notions, allowed us to violate all of these notions).²⁴ On the other hand, however, the results mentioned above demonstrate the surprising ways in which the anti-support conditions of different connectives can interact to violate aspects of determination which the conditions of any given connective do not violate on their own. If we add the BSML-modalities to PL, we have ML, which, it is easy to see, is bicomplete for flat-incompatible pairs. If, instead of the modalities, we add NE, we will thereby violate determination of $\|\neg \phi\|$, but will still have determination of $|\neg \phi|$. But if we add both the modalities and NE, we have BSML, which violates determination in a more radical way.

The standard expressive completeness theorems with respect to team-semantic closure properties (such as 4.2.8, 4.3.2, 4.3.13, and 4.3.27) play an important role in the study of team logics, allowing for concise and tractable characterizations of these logics, and providing useful resources for axiomatization completeness proofs (see, e.g., [139, 140]) and the proofs of other properties such as uniform interpolation [43]. Are there any similar technical applications for bicompleteness theorems? It seems more

²⁴The dependence of the semantics of the dual negation on the main logical symbol of the argument is obvious in the bilateral support/anti-support semantics we have been considering. It is also obvious in the team semantics for IF [78] and D [119], and Väänänen's team-based game semantics for D [119, Section 5.2]—each of these systems features two clauses/rules for each connective or atom, one applied when the symbol in question appears in a positive context, the other when it appears in a negative context. However, it should be noted that this dependence is far less clear in the original assignmentbased game semantics for IF [75] and D [119, Section 5.3]. This type of semantics has a single game rule for each connective or atom, and the behavior of the negation is brought about by the interaction of these rules. The team semantics for these logics (whether bilateral or game-theoretical), accordingly, also make it easier to see why Burgess theorems hold, and enable one to easily adjust the properties of the dual negation as described in the main text. It seems that making similar adjustments would be more challenging in an assignment-, world-, or valuation-based game semantics, and it is not clear to me whether one can formulate (reasonable) world- or valuation-based game semantics for the logics we consider in this paper.

likely to me that theorems of this type would find conceptual application—we have observed for instance, that we may think of them as providing a measure of the degree of failure of determination of negative meanings by positive ones, and that which incompatibility notions a given logic conforms to can provide insight into which types of information are clashing in a contradiction $\varphi \land \neg \varphi$ of the logic. To that end, it may prove interesting and fruitful to establish bicompleteness results for more logics in the philosophical logic and formal semantics literature featuring bilateral negations, including logics which do not employ team semantics.

Chapter 5

Axiomatizing Modal Inclusion Logic and its Variants

This chapter is based on:

Aleksi Anttila, Matilda Häggblom, and Fan Yang. "Axiomatizing modal inclusion logic and its variants". In: *Arch. Math. Logic* (2024). Forthcoming. arXiv: 2312.02285 [math.L0]. URL: https://arxiv.org/abs/2312.02285

Abstract We provide a complete axiomatization of modal inclusion logic team-based modal logic extended with inclusion atoms. We review and refine an expressive completeness and normal form theorem for the logic, define a natural deduction proof system, and use the normal form to prove completeness of the axiomatization. Complete axiomatizations are also provided for two other extensions of modal logic with the same expressive power as modal inclusion logic: one augmented with a might-operator and the other with a single-world variant of the might-operator.

5.1 Introduction

In this article, we axiomatize modal inclusion logic and two other expressively equivalent logics. Modal inclusion logic extends the usual modal logic with *inclusion atoms*—non-classical atoms the interpretation of which requires the use of *team semantics*. In team semantics—introduced by Hodges [78, 79] to provide a compositional semantics for Hintikka and Sandu's *independence-friendly logic* [77, 75], and developed further by Väänänen in his work on *dependence logic* [119]—formulas are interpreted with respect to sets of evaluation points called *teams*, as opposed to single evaluation points. In Hodges' and Väänänen's first-order setting, teams are sets of variable assignments; we will mainly work in *modal team semantics* (first considered in [120]), in which teams are sets of possible worlds in a Kripke model. The shift to teams enables one to express that certain relationships hold between the truth/assignment

values obtained by formulas/variables in the worlds/assignments in a team, rendering team-based logics generally more expressive than their singularly evaluated counterparts. The articulation of these relationships is typically accomplished by furnishing team-based logics with various *atoms of dependency* expressing different kinds of relationships. The first such atoms to be considered were Väänänen's *dependence atoms* [119, 120]. Another example of note are Grädel and Väänänen's *independence atoms* [59].

Inclusion atoms were introduced by Galliani in [51] to import the notion of inclusion dependencies from database theory (see, e.g., [30]) into team semantics. In the modal setting, an inclusion atom is a formula of the form $a \subseteq b$, where a and b are finite sequences of formulas of modal logic of the same length. The atom $a \subseteq b$ is satisfied in a team (i.e., a set of possible worlds) if any sequence of truth values assigned to the formulas in a by some world in the team is also assigned to the sequence b by some world in the team. Consider a database in which various facts about the stores in some town have been collected. Each row (world) of the database contains data about a single store, and each column (propositional symbol) corresponds to a category of collected data such as whether a store sells flowers: if a store sells flowers, then the row for the store has a 1 in the Sell_flowers column. If the team consisting of all rows corresponding to stores in some neighbourhood satisfies the atom Sell_flowers \subseteq Sell_mulch, we know that if there is a store in the neighbourhood that sells flowers, there is one that sells mulch (and that if there is one that does not sell flowers, there is one that does not sell mulch). The atom $\top \perp \subseteq$ Sell_flowers Sell_food, on the other hand, would express that there is a store that sells flowers but not food.

Our main focus in this article is on modal logic extended with inclusion atoms, or *modal inclusion logic*, which we will denote by $ML(\subseteq)$. The expressive power of $ML(\subseteq)$ has been studied in [73, 89] and its complexity in [69, 70], but the logic has so far not been axiomatized. We fill this gap in the literature by providing a sound and complete natural deduction system for the logic. Our system is an extension of the system for propositional inclusion logic introduced in [137].

In addition to $ML(\subseteq)$, we axiomatize two other logics with the same expressive power: $ML(\bigtriangledown)$ and $ML(\bigtriangledown)$, or modal logic extended with a *might-operator* \bigtriangledown (first considered in the team semantics literature in [73]) and with a *singular might-operator* \bigtriangledown that we introduce in this article, respectively. The names reflect the fact that similar operators have been used to model the meanings of epistemic possibility modalities such as the "might" in "It might be raining"—see, for instance, [129, 123].

Each of these three logics— $ML(\subseteq)$, $ML(\bigtriangledown)$, and $ML(\bigtriangledown)$ —is *closed under unions*: if a formula is true in all teams in a nonempty collection of teams, it is true in the team formed by union of the collection. The most well-known team-based logics such as dependence logic are not union closed, but union-closed logics have recently been receiving more attention in the literature (see, e.g., [6, 7, 54, 57, 69, 70, 73, 80, 136, 137]). Hella and Stumpf proved in [73] that each of $ML(\subseteq)$ and $ML(\bigtriangledown)$ is expressively complete for team properties that are closed under unions and bounded bisimulations,

5.2. Preliminaries

and which contain the empty team. We revisit this proof and also show that $ML(\nabla)$ is complete for this class of properties. These proofs of expressive completeness also yield normal forms for each of the three logics. The normal forms play a crucial role in the completeness proofs for our axiomatizations.

The structure of the article is as follows: in Section 5.2, we define the syntax and semantics of modal inclusion logic $ML(\subseteq)$ and the might-operator logics $ML(\bigtriangledown)$ and $ML(\bigtriangledown)$, and discuss some basic properties of these logics. In Section 5.3 we prove the expressive completeness results discussed above. In Section 5.4, we introduce natural deduction systems for each of the logics and show that they are complete. We conclude the article and discuss directions for possible further research in Section 5.5. In an appendix (Section 5.6) we provide a translation of $ML(\subseteq)$ into first-order inclusion logic.

This article is partly based on Häggblom's master's thesis [63], supervised by Yang and Anttila, which contains preliminary versions of some of the results.

5.2 Preliminaries

In this section, we define the syntax and semantics of modal inclusion logic $ML(\subseteq)$ and the two might-operator logics $ML(\nabla)$ and $ML(\nabla)$. We also discuss some basic properties of these logics.

Fix a (countably infinite) set Prop of propositional symbols.

5.2.1. DEFINITION. The syntax for the usual modal logic (ML) is given by the grammar:

$$\alpha \coloneqq p \mid \bot \mid \neg \alpha \mid (\alpha \lor \alpha) \mid (\alpha \land \alpha) \mid \otimes \alpha \mid \Box \alpha,$$

where $p \in \text{Prop.}$ Define $\top := \neg \bot$. We also call formulas of ML *classical formulas*. Throughout the article, we reserve the first Greek letters α and β for classical formulas.

The syntax for modal inclusion logic $(ML(\subseteq))$ is given by:

$$\varphi ::= p \mid \bot \mid (\alpha_1 \dots \alpha_n \subseteq \beta_1 \dots \beta_n) \mid \neg \alpha \mid (\varphi \lor \varphi) \mid (\varphi \land \varphi) \mid \otimes \varphi \mid \Box \varphi,$$

where $p \in \text{Prop}$, and $\alpha, \alpha_i, \beta_i$ (for all $1 \le i \le n$) range over classical formulas. We write $\text{Prop}(\varphi)$ for the set of propositional symbols appearing in φ , and $\varphi(X)$ if $\text{Prop}(\varphi) \subseteq X \subseteq \text{Prop}$.

The above syntax of $ML(\subseteq)$ deserves some comment. First, we only allow negation to occur in front of classical formulas. Next, our inclusion atoms $\alpha_1 \dots \alpha_n \subseteq \beta_1 \dots \beta_n$ with α_i and β_j being (possibly complex) classical formulas are known in the literature also as *extended inclusion atoms*, and the logic $ML(\subseteq)$ defined above is sometimes also referred to as *extended modal inclusion logic*—in contexts which use this terminology, a distinction is drawn between extended inclusion atoms and inclusion atoms *simpliciter*, the latter referring to atoms which may only contain propositional symbols as subformulas (modal inclusion logic *simpliciter*, in these contexts, is then the variant which only allows these simpler atoms). In this article we only study the extended variant. We do not allow nested inclusion atoms; for example, $p \subseteq (p \subseteq q)$ is not a formula of ML(\subseteq). We usually omit the parentheses around inclusion atoms, and stipulate higher precedence for the inclusion symbol \subseteq than the other connectives; for instance, $p \land q \subseteq r$ has the subformula $q \subseteq r$.

Modal inclusion logic is interpreted on standard Kripke models, but we use teams (sets of worlds) rather than single worlds as points of evaluation. A (*Kripke*) model M = (W, R, V) (over $X \subseteq Prop$) consists of a set W of possible worlds, a binary accessibility relation $R \subseteq W \times W$ and a valuation function $V : X \rightarrow \mathcal{P}(W)$. A team T of M is a subset $T \subseteq W$ of the set of worlds in M. The image of T (denoted R[T]) and the preimage of T (denoted $R^{-1}[T]$) are defined as

$$R[T] = \{v \in W \mid \exists w \in T : wRv\}, \text{ and}$$
$$R^{-1}[T] = \{w \in W \mid \exists v \in T : wRv\}.$$

We say that a team S of M is a successor team of T, written TRS, if

$$S \subseteq R[T]$$
 and $T \subseteq R^{-1}[S]$;

that is, every world in S is accessible from a world in T, and every world in T has an accessible world in S.

5.2.2. DEFINITION. For any Kripke model *M* over X and team *T* of *M*, the satisfaction relation $M, T \models \varphi$ (or simply $T \models \varphi$) for an ML(\subseteq)-formula $\varphi(X)$ is given inductively by the following clauses:

$$\begin{split} M,T &\models p \iff T \subseteq V(p). \\ M,T &\models \bot \iff T = \emptyset. \\ M,T &\models \alpha_1 \dots \alpha_n \subseteq \beta_1 \dots \beta_n \iff \text{for all } w \in T, \text{ there exists } v \in T \text{ such that for all } \\ &1 \leq i \leq n, M, \{w\} \models \alpha_i \text{ iff } M, \{v\} \models \beta_i. \\ M,T &\models \neg \alpha \iff M, \{w\} \notin \alpha \text{ for all } w \in T. \\ M,T &\models \varphi \lor \psi \iff M, T_1 \models \varphi \text{ and } M, T_2 \models \psi \text{ for some } T_1, T_2 \subseteq T \\ & \text{ such that } T_1 \cup T_2 = T. \\ M,T &\models \varphi \land \psi \iff M, T \models \varphi \text{ and } M, T \models \psi. \\ M,T &\models \otimes \varphi \iff M, S \models \varphi \text{ for some } S \text{ such that } TRS. \\ M,T &\models \Box \varphi \iff M, R[T] \models \varphi. \end{split}$$

We say that a set of formulas Γ *entails* φ , written $\Gamma \vDash \varphi$, if for all models M and teams T of M, if $M, T \vDash \gamma$ for all $\gamma \in \Gamma$, then $M, T \vDash \varphi$. We write simply $\varphi_1, \ldots, \varphi_n \vDash \varphi$ for $\{\varphi_1, \ldots, \varphi_n\} \vDash \varphi$ and $\vDash \varphi$ for $\emptyset \vDash \varphi$, where \emptyset is the empty set of formulas. If both $\varphi \vDash \psi$ and $\psi \vDash \varphi$, we say that φ and ψ are *equivalent*, and write $\varphi \equiv \psi$.

5.2. Preliminaries

We often write $T \vDash \varphi$ instead of $M, T \vDash \varphi$. It is easy to verify that formulas of ML(\subseteq) have the following properties:

Union closure: If $M, T_i \vDash \varphi$ for all $i \in I \neq \emptyset$, then $M, \bigcup_{i \in I} T_i \vDash \varphi$.

Empty Team Property: $M, \emptyset \vDash \varphi$ for all models *M*.

Classical formulas α (i.e., formulas of ML) are, in addition, *downward closed*, meaning that $M, T \models \alpha$ implies $M, S \models \alpha$ for all $S \subseteq T$. It is easy to show that a formula has the downward closure property, union closure property and the empty team property if and only if it has the *flatness property*:

Flatness: $M, T \models \varphi$ iff $M, \{w\} \models \varphi$ for every $w \in T$.

Classical formulas are thus flat. It is also straightforward to verify that for any classical formula α ,

$$M, \{w\} \models \alpha \iff M, w \models \alpha,$$

where \vDash on the right is the usual single-world-based satisfaction relation for ML. It follows from this that ML (with team semantics) coincides with the usual single-world based modal logic, and hence that ML(\subseteq) is *conservative* over the usual modal logic.

5.2.3. PROPOSITION. For any set $\Gamma \cup \{\alpha\}$ of ML-formulas,

$$\Gamma \vDash \alpha \iff \Gamma \vDash^c \alpha$$
,

where \models^{c} is the usual entailment relation for ML (over the single-world semantics).

Given these facts, we use the notations $M, \{w\} \models \alpha$ and $M, w \models \alpha$ interchangeably whenever α is classical, and similarly for $\Gamma \models \alpha$ and $\Gamma \models^c \alpha$.

Let us briefly comment on the truth conditions for the modalities. Recall our store database example from Section 5.1. If the database is also equipped with an accessibility relation detailing the (epistemically) possible future inventories of each store, we could use a modal statement such as $(T \perp \subseteq Sell_flowersSell_food)$ to express that there might in the future be a store that sells flowers and does not sell food. Note that if *S* is a successor team of *T*, there may, for any given world *w* in *T*, be multiple worlds accessible to *w* in *S*. Interpretations of the modalities must take this into account—for instance, in our database example, a successor team may contain multiple inconsistent records for each store.

One may also consider alternative team-semantic truth conditions for the diamond, such as the following (called the *strict semantics* for the diamond, whereas the semantics above are the *lax semantics*):

$$T \models \bigotimes_{s} \varphi \iff \exists f : T \to R[T] \text{ s.t. } \forall w \in T : wRf(w) \text{ and } f[T] \models \varphi,$$

where $f[T] = \{f(w) | w \in T\}$. With the strict semantics, $ML(\subseteq)$ is no longer union closed, and it also fails to be bisimulation-invariant for the notion of team bisimulation

established in the literature.¹ In this article we only consider the lax semantics. For more on strict and lax semantics, see [51, 137, 69, 70].² For other alternative sets of team-semantic truth conditions for the modalities, see [42, 34, 6, 7].

We next discuss *primitive inclusion atoms*—inclusion atoms of a specific restricted form which play a crucial role in our expressive completeness results. We often abbreviate a sequence $\langle \alpha_1, ..., \alpha_n \rangle$ of classical formulas as a; similarly, b is short for $\langle \beta_1, ..., \beta_n \rangle$, etc. We also often write x for a sequence $\langle x_1, ..., x_n \rangle$ in which each x_i is one of the constants \perp or \top . The *arity* of an inclusion atom $a \subseteq b$ is defined as the length |a| of the sequence a. *Primitive inclusion atoms* are inclusion atoms of the form $x \subseteq a$ —that is, they are atoms with only the constants \perp and \top on the left-hand side. We additionally call primitive inclusion atoms $\top ... \top \subseteq \alpha_1 ... \alpha_n$ with only the constant \top on the left-hand side *top inclusion atoms*. Primitive inclusion atoms $x \subseteq a$ are clearly *upward closed* (modulo the empty team property), i.e., for any nonempty teams T and S, if $T \models x \subseteq a$ and $S \supseteq T$, then $S \models x \subseteq a$.

Primitive inclusion atoms are relatively tractable, as the next proposition shows. Hereafter, for any $x \in \{\top, \bot\}$, we write α^{\top} for α , and α^{\bot} for $\neg \alpha$. For $x_1, \ldots, x_n \in \{\top, \bot\}$, we abbreviate $\alpha_1^{x_1} \land \cdots \land \alpha_n^{x_n}$ as a^{\times} .

5.2.4. PROPOSITION. For any nonempty team T,

 $T \vDash x \subseteq a$ iff there exists $v \in T$ such that $v \vDash a^{x}$.

Proof:

Suppose $T \neq \emptyset$ and $T \models x \subseteq a$. Let $w \in T$. Then there is a $v \in T$ such that $w \models x_i$ iff $v \models \alpha_i$ for all i = 1, ..., n. Let $i \in \{1, ..., n\}$. If $x_i = \top$, then $w \models x_i$, so $v \models \alpha_i$. Hence $v \models \alpha_i^{\top}$. If $x_i = \bot$, then $w \notin x_i$, so $v \notin \alpha_i$. Hence $v \models \alpha_i^{\bot}$. So for all $i = 1, ..., n, v \models \alpha_i^{x_i}$. We conclude $v \models a^{\times}$. The other direction is similar.

5.2.5. COROLLARY. $\bot \subseteq \alpha \equiv \top \subseteq \neg \alpha$.

The following example illustrates an interesting consequence of the upward closure (modulo the empty team property) of primitive inclusion atoms in our modal setting.

5.2.6. EXAMPLE. Consider the model M = (W, R, V) as illustrated in the figure below, where the relation *R* is represented by the arrows, and $V(p) = \{v'\}$. Consider the teams $T = \{u, v\}$ and $S = \{u', v'\}$. Since $v' \in S$ and $v' \models p$, by Proposition 5.2.4, $S \models \top \subseteq p$. It is also easy to verify that *TRS*, and hence $T \models \otimes (\top \subseteq p)$. Similarly $R[T] \models \top \subseteq p$, whereby

¹For the failure of union closure, consider the formula $\varphi \coloneqq \bigotimes_s (pq \subseteq rs)$ and a model with $W = \{w_1, w_2, w_3, w_4\}$, $R = \{(w_1, w_1), (w_2, w_2), (w_2, w_3), (w_4, w_4)\}$, and $V(p) = \{w_1, w_2\}$, $V(q) = \{w_3, w_4\}$, $V(r) = \{w_2\}$, $V(s) = \{w_3\}$. We have $\{w_1, w_2\} \models \varphi$ and $\{w_2, w_4\} \models \varphi$, but $\{w_1, w_2, w_4\} \not\models \varphi$. For the failure of bisimulation invariance, see Section 5.3.

²In the literature, the term "strict semantics" typically refers to the adoption of different truth conditions not only for the diamond but also for the disjunction.

5.2. Preliminaries

 $T \models \Box(\top \subseteq p)$. In fact, we have in general $\otimes (x \subseteq a) \models \Box(x \subseteq a)$ for an arbitrary primitive inclusion atom $x \subseteq a$. Moreover, observe that the subteam $\{u\}$ of *T* does not satisfy either $\otimes (\top \subseteq p)$ or $\Box(\top \subseteq p)$, illustrating the failure of downward closure in ML(\subseteq).



It was proved in [137] in the propositional context that arbitrary inclusion atoms can be defined in terms of primitive inclusion atoms, and further in terms of unary primitive inclusion atoms (inclusion atoms of the form $\top \subseteq \alpha$ or $\perp \subseteq \alpha$). This result can be extended to our current modal context by the same argument.

5.2.7. LEMMA ([137]). Let a, b be sequences of ML-formulas, and let x, y be sequences each of whose elements is either \top or \bot . Then

- (i) $\bigwedge_{x \in \{T, \bot\}^{|a|}} (\neg a^x \lor x \subseteq b) \equiv a \subseteq b.$
- (*ii*) $xy \subseteq ab \equiv x \subseteq a \land ((y \subseteq b \land a^{x}) \lor \neg a^{x}).$

Given Corollary 5.2.5, we can further conclude that an arbitrary inclusion atom can be defined in terms of unary top inclusion atoms $\top \subseteq \alpha$. Indeed, we will see in Section 5.3 that every formula of ML(\subseteq) is equivalent to one in a normal form which contains no inclusion atoms save for unary top inclusion atoms.

In addition to modal inclusion logic, we also consider two other extensions of the usual modal logic: $ML(\nabla)$, or modal logic with the *might-operator* ∇ , and $ML(\nabla)$, or modal logic with the *singular might-operator* ∇ . We define the team semantics of these operators as follows:

$$M, T \models \nabla \varphi \iff T = \emptyset$$
 or there exists a nonempty $S \subseteq T$ such that $M, S \models \varphi$.
 $M, T \models \nabla \varphi \iff T = \emptyset$ or there exists $w \in T$ such that $M, \{w\} \models \varphi$.

The operator ∇ was first considered in the team semantics literature in [73] and the other operator ∇ is introduced in the present article, but very similar operators have been employed in philosophical logic and formal semantics to model the meanings of epistemic possibility modalities such as the "might" in "It might be raining" (see, for instance [129, 123]). One can also make sense of this interpretation in the team

semantics setting. A team *T* can be conceived of as corresponding to an *information state*: to know that the actual world *v* is one of the worlds in *T* is to know that whatever holds in all worlds in *T* must be the case. If it is raining in all worlds in *T*, this information state supports the assertion that it is raining. If, on the other hand, there is some nonempty subteam *S* of *T* consisting of worlds in which it is not raining, then for all that one knows given the information embodied in *T*, it might not be raining. Both operators may be thought of as expressing this kind of epistemic modality, with some subtle differences in meaning between the two (note, for instance, that if $\nabla \varphi$ appears within the scope of another ∇ in a propositional (non-modal) formula, it can be substituted *salva veritate* with φ , whereas the analogous fact does not hold for ∇).³

The singular might $\forall \varphi$ clearly entails the (non-singular) might $\forall \varphi$; the converse direction $\forall \varphi \models \forall \varphi$ holds whenever φ is downward closed, and in particular whenever φ is classical. For classical formulas α , both $\forall \alpha$ and $\forall \alpha$ coincide with the unary top inclusion atom $\top \subseteq \alpha$.

5.2.8. FACT. For any ML-formula α , $\top \subseteq \alpha \equiv \forall \alpha \equiv \forall \alpha$.

In general, the two might-operators behave differently; in particular, there are subtle differences when the operators are iterated: $\nabla(\nabla \phi \land \nabla \psi) \equiv \nabla \phi \land \nabla \psi$, whereas $\nabla \phi \land \nabla \psi \neq \nabla (\nabla \phi \land \nabla \psi)$; $\nabla (\phi \land \nabla \psi) \equiv \nabla (\phi \land \psi)$, whereas $\nabla (\phi \land \nabla \psi) \neq \nabla (\phi \land \psi)$.

As is the case with $ML(\subseteq)$, the logics $ML(\nabla)$ and $ML(\nabla)$ are union closed and have the empty team property; we further show in Section 5.3 that the three logics are in fact expressively equivalent.

As with many other logics with team semantics, the three logics we consider in this article are not closed under uniform substitution. Recall that a *substitution* σ for a logic \mathcal{L} is a mapping from the set of formulas \mathcal{L} to itself that commutes with the connectives (and connective-like atoms such inclusion atoms) of \mathcal{L} . We say that \mathcal{L} is *closed under* substitution σ if for any set $\Gamma \cup {\varphi}$ of formulas of \mathcal{L} ,

$$\Gamma \vDash \varphi$$
 implies $\{\sigma(\gamma) \mid \gamma \in \Gamma\} \vDash \sigma(\varphi)$.

A logic \mathcal{L} is *closed under uniform substitution* if it is closed under all substitutions. Note that due to our syntactic restrictions, any map mapping a ML(\subseteq)-formula $p \subseteq q$ to a non-formula $(r \subseteq s) \subseteq q$ is not considered a valid substitution for ML(\subseteq). This does not mean that substitution into inclusion atoms is disallowed in general: a valid substitution might map $p \subseteq q$ to $(r \land s) \subseteq q$.

To see why ML(\subseteq) is not closed under uniform substitution, note that clearly $(p \lor \neg p) \land q \models (p \land q) \lor (\neg p \land q)$ holds. But when we substitute $\top \subseteq p$ for q on both sides,

³The requirement in the truth conditions that $\nabla \varphi$ and $\nabla \varphi$ also be true in the empty team is added to preserve the empty team property. One may also consider variants of these operators without this requirement. The nonempty variant of ∇ is sometimes called the *exists operator* E and is discussed in, for instance, [89, 97]. Logics with both the nonempty variant ∇_{NE} of ∇ and the disjunction \vee may be thought of as variants of logics which include \vee and the *nonemptiness atom* NE (where $T \models \text{NE}$ iff $T \neq \emptyset$) due to the equivalences $\nabla_{\text{NE}} \varphi \equiv (\varphi \land \text{NE}) \lor \top$ and $\text{NE} \equiv \nabla_{\text{NE}} \top$; for more on these logics, see [140, 7].

the entailment no longer holds: $(p \lor \neg p) \land \top \subseteq p \notin (p \land \top \subseteq p) \lor (\neg p \land \top \subseteq p)$. For a counterexample to the entailment, consider the team $R[T] = \{u', v', w'\}$ from Example 5.2.6. Clearly $R[T] \models (p \lor \neg p) \land \top \subseteq p$, but there are no subteams $T_1, T_2 \subseteq R[T]$ such that $T_1 \cup T_2 = R[T]$ with $T_1 \models p \land \top \subseteq p$ and $T_2 \models \neg p \land \top \subseteq p$. Similar counterexamples can also be found for ML(\heartsuit) and ML(\heartsuit): in the above example, instead of using the top inclusion atom $\top \subseteq p$, we can equivalently (by Fact 5.2.8) use the formula ∇p or the formula ∇p .

Nevertheless, the three logics we consider in the article are closed under *classical* substitutions, namely, substitutions σ such that $\sigma(p)$ is a classical formula for all $p \in$ Prop. We now prove this by using the same strategy as in [140], where it was proved that a number of propositional team logics (including propositional inclusion logic) are closed under classical substitutions.

5.2.9. LEMMA. Let φ be a formula in ML(\subseteq), ML(\heartsuit), or ML(\heartsuit), and σ a classical substitution for the relevant logic. For any model M = (W, R, V) over $X \supseteq \operatorname{Prop}(\sigma(\varphi))$ and team T of M,

$$M,T \vDash \sigma(\varphi) \iff M_{\sigma}, T \vDash \varphi,$$

where $M_{\sigma} = (W, R, V_{\sigma})$ is any model over $Y \supseteq \operatorname{Prop}(\varphi)$ satisfying $M_{\sigma}, w \models p$ iff $M, w \models \sigma(p)$, for all $p \in Y$ and $w \in W$.

Proof:

We prove the lemma by induction on φ . The case in which $\varphi = p$ follows by flatness of $\sigma(p)$. The other cases follow by the induction hypothesis. In particular, for $\varphi = \otimes \psi$, we have $M, T \models \otimes \sigma(\psi)$ if and only if there is a team *S* such that *TRS* and $M, S \models \sigma(\psi)$. By the induction hypothesis, this is the case if and only if there is a team *S* such that *TRS* and $M_{\sigma}, S \models \psi$, which holds if and only if $M_{\sigma}, T \models \otimes \psi$.

5.2.10. THEOREM. Let $\Gamma \cup \{\varphi\}$ be a set of formulas in ML(\subseteq), ML(∇), or ML(∇). For any classical substitution σ for the relevant logic, if $\Gamma \vDash \varphi$, then $\{\sigma(\gamma) | \gamma \in \Gamma\} \vDash \sigma(\varphi)$.

Proof:

If $\Gamma \vDash \varphi$, then for any model *M* and team *T* of *M*,

$$M, T \models \sigma(\gamma) \text{ for all } \gamma \in \Gamma \implies M_{\sigma}, T \models \gamma \text{ for all } \gamma \in \Gamma \qquad \text{(Lemma 5.2.9)}$$
$$\implies M_{\sigma}, T \models \varphi \qquad \text{(By assumption)}$$
$$\implies M, T \models \sigma(\varphi) \qquad \text{(Lemma 5.2.9)}$$

Hence $\{\sigma(\gamma)|\gamma\in\Gamma\} \models \sigma(\varphi)$.

As with standard modal logic, one can provide a first-order translation of $ML(\subseteq)$ and its variants; see [89] for a translation (via normal form) of many modal team-based logics into classical first-order logic, and see Section 5.6 for a translation of $ML(\subseteq)$ into first-order inclusion logic.

5.3 Expressive Completeness and Normal Forms

It was proved in [73] that each of $ML(\subseteq)$ and $ML(\bigtriangledown)$ is expressively complete with respect to the class \mathbb{U} of union-closed modally definable team properties with the empty team property. In this section, we review this result, and show that our new variant $ML(\bigtriangledown)$ has the same expressive power. The proofs of these results also yield a normal form for each logic. These normal forms are crucial for proving the completeness of our axiomatizations for the logics—see Section 5.4.

As in the single-world setting, in the team-based setting modal definability can be characterized by bisimulation invariance: the modally definable properties are precisely those invariant (i.e., closed) under bisimulation. We begin by recalling the standard notion of bisimulation as well as that of Hintikka formulas—characteristic formulas which serve to capture bisimilarity in the syntax of ML—and then proceed to construct analogous notions for teams.

Throughout this section, we make use of a fixed finite set $X \subseteq$ Prop of propositional symbols; we often omit mention of X in definitions and results to keep notation light. If *M* is a Kripke model (over X) and $w \in W$, we call (M, w) a *pointed model* (over X).

5.3.1. DEFINITION. For any (M, w), (M', w'), and $k \in \mathbb{N}$, (M, w) and (M', w') are X, *k*-bisimilar, written $M, w \rightleftharpoons_k^X M', w'$ (or simply $w \rightleftharpoons_k w'$), if the following recursively defined relation holds:

- (i) $M, w \rightleftharpoons_0^X M', w'$ iff $M, w \models p \iff M', w' \models p$ for all $p \in X$.
- (ii) $M, w \rightleftharpoons_{k=1}^{\mathsf{X}} M', w'$ if $M, w \rightleftharpoons_{0}^{\mathsf{X}} M', w'$ and:

(Forth condition) For every world v of M with wRv there is a world v' of M' with w'Rv' such that $M, v \rightleftharpoons_{k}^{X} M', v'$.

(Back condition) For every world v' of M' with w'Rv' there is a world v of M with wRv such that $M, v \rightleftharpoons_k^X M', v'$.

The modal depth $md(\varphi)$ of a formula φ is defined by the following clauses:

$$md(p) = md(\bot) = 0,$$

$$md(\neg \alpha) = md(\alpha),$$

$$md(\psi_1 \lor \psi_2) = md(\psi_1 \land \psi_2) = \max\{md(\psi_1), md(\psi_2)\},$$

$$md(\otimes \psi) = md(\Box \psi) = md(\psi) + 1, \text{ and}$$

$$md(\alpha_1 \dots \alpha_n \subseteq \beta_1 \dots \beta_n) = \max\{md(\alpha_1), \dots md(\alpha_n), md(\beta_1) \dots md(\beta_n)\}$$

We say that two pointed models (M, w) and (M', w') are X,*k*-equivalent, written $M, w \equiv_k^X M', w'$ (or simply $w \equiv_k w'$), if they satisfy the same ML-formulas with propositional symbols among those in X up to modal depth k, i.e., if $M, w \models \alpha$ iff $M', w' \models \alpha$ for every $\alpha(X) \in ML$ with $md(\alpha) \le k$.

5.3.2. DEFINITION. Let $k \in \mathbb{N}$ and let (M, w) be a pointed model over $Y \supseteq X$. The *k*th *Hintikka formula* $\chi_{M,w}^{X,k}$ (or simply χ_w^k) of (M, w) is defined recursively by:

$$\chi_{M,w}^{\mathsf{X},0} \coloneqq \bigwedge \{ p \mid p \in \mathsf{X} \text{ and } w \in V(p) \} \land \bigwedge \{ \neg p \mid p \in \mathsf{X} \text{ and } w \notin V(p) \};$$
$$\chi_{M,w}^{\mathsf{X},k+1} \coloneqq \chi_{M,w}^k \land \bigwedge_{v \in R[w]} \otimes \chi_{M,v}^k \land \boxtimes \bigvee_{v \in R[w]} \chi_{M,v}^k.$$

It is not hard to see that there are only finitely many non-equivalent *k*th Hintikka formulas for a given finite X. This is why we may assume that the conjunction and the disjunction in $\chi_{M,w}^{X,k+1}$ are finite and hence that $\chi_{M,w}^{X,k+1}$ is well-defined⁴.

5.3.3. THEOREM (See, e.g., [56]). For any $k \in \mathbb{N}$ and any pointed models (M, w) and (M', w'):

$$w \equiv_k w' \iff w \rightleftharpoons_k w' \iff w' \vDash \chi_w^k.$$

We now define team-based analogues to the preceding world-based notions. A *model with a team* (over X) is a pair (M,T), where M is a model (over X) and T is a team of M. Team bisimulation (introduced in [71, 89]) is a straightforward generalization of world bisimulation:

5.3.4. DEFINITION. For any (M,T) and (M',T'), and any $k \in \mathbb{N}$, (M,T) and (M',T') are (team) X,*k*-bisimilar, written $M,T \rightleftharpoons_k^X M',T'$ (or simply $T \rightleftharpoons_k T'$), if:

(Forth condition) For every $w \in T$ there exists a $w' \in T'$ such that $w \rightleftharpoons_k^X w'$.

(Back condition) For every $w' \in T'$ there exists a $w \in T$ such that $w \rightleftharpoons_k^X w'$.

Clearly, both world and team k-bisimulation are equivalence relations. Moreover, if $w \rightleftharpoons_k w'$, then $w \rightleftharpoons_n w'$ for all $n \le k$, and similarly if $T \rightleftharpoons_k T'$, then $T \rightleftharpoons_n T'$ for all $n \le k$. The following lemma lists some further simple facts about team bisimulation.

5.3.5. LEMMA ([71]). If $T \rightleftharpoons_{k+1} T'$, then

- (i) For every S such that TRS, there is a S' such that T'R'S' and $S \rightleftharpoons_k S'$.
- (*ii*) $R[T] \rightleftharpoons_k R'[T']$.
- (iii) For all $T_1, T_2 \subseteq T$ such that $T_1 \cup T_2 = T$ there are $T'_1, T'_2 \subseteq T'$ such that $T'_1 \cup T'_2 = T'$ and $T_i \rightleftharpoons_{k+1} T'_i$ for $i \in \{1, 2\}$.

⁴ To be more precise, this assumption amounts to the following: if R[w] is infinite, we choose a finite set $T \subseteq R[w]$ such that for all $v \in R[w]$, there is a $v' \in T$ with $\chi_{M,v}^{X,k} \equiv \chi_{M,v'}^{X,k}$, and define $\chi_{M,w}^{X,k+1}$ using this finite representative set T in place of R[w]. Clearly, the specific choice of this finite representative set T makes no difference for the Hintikka formula. Similar remarks apply to other finiteness assumptions made in the sequel.

We say that the models with teams (M,T) and (M',T') are X, *k*-equivalent (in a logic \mathcal{L}), written $M, T \equiv_k^X M', T'$ (or simply $T \equiv_k T'$), if for every formula $\varphi(X)$ in \mathcal{L} with $md(\varphi) \leq k, M, T \models \varphi$ iff $M', T' \models \varphi$. It will follow from results we show that $T \equiv_k T'$ in any of the logics we consider iff $T \equiv_k T'$ in either of the other logics; we therefore simply write $T \equiv_k T$ without specifying the logic. It is easy to show that *k*-bisimilarity implies *k*-equivalence for all three logics:

5.3.6. THEOREM (Bisimulation invariance). If $T \rightleftharpoons_k T'$, then $T \equiv_k T'$.

Proof:

We show by induction on the complexity of formulas φ in ML(\subseteq), ML(∇), or ML(∇) that if $T \rightleftharpoons_k T'$, then $T \equiv_k T'$ for $k = md(\varphi)$. See [73] for details. In particular, items (i), (ii) and (iii) in Lemma 5.3.5 are used in the diamond, box and disjunction cases, respectively. For $\varphi = \nabla \psi$, we have $T \models \nabla \psi$ iff $T = \emptyset$ or $\{w\} \models \psi$ for some $w \in T$. If $T = \emptyset$, then by $T \rightleftharpoons_k T'$ also $T' = \emptyset$ so that $T' \models \nabla \psi$. Otherwise by $T \rightleftharpoons_k T'$ we have $\{w\} \rightleftharpoons_k \{w'\}$ for some $w' \in T'$, and then $\{w'\} \models \psi$ by the induction hypothesis, whence $T' \models \nabla \psi$. The other direction is similar.

Let us also demonstrate why ML(\subseteq), ML(∇), and ML(∇) with the strict semantics for the diamond (see Section 5.2) are not bisimulation-invariant for the notion of team bisimulation we have adopted. Consider a model M = (W, R, V) over $\{p\}$ where W = $\{w_1, w_2, w_3\}$; $R = \{(w_1, w_1), (w_1, w_3), (w_2, w_2), (w_2, w_3)\}$; and $V(p) = \{w_3\}$. Clearly $w_1 \rightleftharpoons_k w_2$ for all $k \in \mathbb{N}$, so that also $\{w_1\} \rightleftharpoons_k \{w_1, w_2\}$ for all $k \in \mathbb{N}$. Defining $f(w_1) \coloneqq w_3$ and $f(w_2) \coloneqq w_2$, we have $f[\{w_1, w_2\}] \vDash \top \subseteq p \land \bot \subseteq p$ so that $\{w_1, w_2\} \vDash \bigotimes_s (\top \subseteq p \land \bot \subseteq p)$. Clearly $\{w_1\} \notin \bigotimes_s (\top \subseteq p \land \bot \subseteq p)$, so $\{w_1\}_s \notin_1 \{w_1, w_2\}$ where $_s \equiv_1$ is 1-equivalence defined with respect to the strict semantics. An analogous argument can be conducted in ML(∇) as well as in ML(∇).

With a suitable notion of bisimilarity at hand, we now proceed to show our expressive completeness results. We measure the expressive power of the logics in terms of the *properties* they can express. A *(team) property* (over X) is a class of models with teams (over X). For each formula $\varphi(X)$ we denote by $\|\varphi\|_X$ (or simply $\|\varphi\|$) the property (over X) expressed or defined by φ , i.e.,

$$\|\varphi\|_{\mathsf{X}} \coloneqq \{(M,T) \text{ over } \mathsf{X} \,|\, M, T \vDash \varphi\}.$$

A property \mathcal{P} is *invariant under* X,*k-bisimulation* if $(M,T) \in \mathcal{P}$ and $M,T \rightleftharpoons_k^X M',T'$ imply that $(M',T') \in \mathcal{P}$; is *closed under unions* if $(M,T_i) \in \mathcal{P}$ for all $i \in I \neq \emptyset$ implies that $(M,\bigcup_{i\in I}T_i) \in \mathcal{P}$; and has the *empty team property* if $(M,\emptyset) \in \mathcal{P}$ for all M.

We say that a logic \mathcal{L} is *expressively complete* for a class of properties \mathbb{P} , written $\mathbb{P} = \|\mathcal{L}\|$, if for each finite $X \subseteq Prop$, the class \mathbb{P}_X of properties over X in \mathbb{P} is precisely the class of properties over X definable by formulas in \mathcal{L} , i.e., if

$$\mathbb{P}_{\mathsf{X}} = \{ \| \varphi \|_{\mathsf{X}} \mid \varphi \in \mathcal{L} \text{ and } \mathsf{Prop}(\varphi) = \mathsf{X} \}.$$

Our goal is to show that each of the three logics is expressively complete for the class \mathbb{U} of properties \mathcal{P} such that \mathcal{P} is union closed; \mathcal{P} has the empty team property; and \mathcal{P} is invariant under bounded bisimulation, meaning that if \mathcal{P} is a property over X, then \mathcal{P} is invariant under X,*k*-bisimulation for some $k \in \mathbb{N}$. That is, we show:

$$\mathbb{U} = \|\mathrm{ML}(\subseteq)\| = \|\mathrm{ML}(\bigtriangledown)\| = \|\mathrm{ML}(\bigtriangledown)\|.$$

Clearly, any property $\|\varphi\|_X$ expressible in any of the three logics is union closed and has the empty team property, and by Theorem 5.3.6 it is also invariant under X,*k*bisimulation for some $k \in \mathbb{N}$ (take any $k \ge md(\varphi)$). Therefore, we have $\{\|\varphi\|_X | \varphi \in \mathcal{L} \text{ where } \operatorname{Prop}(\varphi) = X\} \subseteq \mathbb{U}_X$ for \mathcal{L} being any of the three logics. For the converse inclusion, we must show that any property $\mathcal{P} \in \mathbb{U}_X$ is definable by some formula in each of the logics. Let us focus on ML(\subseteq) first.

As a first step, we will construct characteristic formulas in $ML(\subseteq)$ for teams analogous to the Hintikka formulas (characteristic formulas for worlds) defined above. These formulas and the accompanying lemma are essentially as in [73]; we have simplified them somewhat using an idea presented in [89].

5.3.7. LEMMA. For any $k \in \mathbb{N}$ and any (M,T) over X, there are formulas $\eta_{M,T}^{X,k} \in ML$ and $\zeta_{M,T}^{X,k}, \theta_{M,T}^{X,k} \in ML(\subseteq)$ such that:

(i)
$$M', T' \models \zeta_{M,T}^{X,k}$$
 iff $T' = \emptyset$ or for all $w \in T$ there is a $w' \in T'$ such that $M, w \rightleftharpoons_k^X M', w'$.

- (ii) $M', T' \vDash \eta_{M,T}^{X,k}$ iff for all $w' \in T'$ there is a $w \in T$ such that $M, w \rightleftharpoons_k^X M', w'$.
- (iii) $M', T' \vDash \theta_{M,T}^{X,k}$ iff $M, T \rightleftharpoons_k^X M', T'$ or $T' = \emptyset$.

Proof:

(i) Let

$$\zeta_{M,T}^{\mathsf{X},k} \coloneqq \bigwedge_{w \in T} (\top \subseteq \boldsymbol{\chi}_{M,w}^{\mathsf{X},k}),$$

where we stipulate $\bigwedge \emptyset = \top$. Since there are only a finite number of non-equivalent *k*th Hintikka formulas for a given finite X, we can assume that the conjunction $\bigwedge_{w \in T} (\top \subseteq \chi_{M,w}^{X,k})$ is finite and that $\zeta_{M,T}^k$ is well-defined. If $T' \neq \emptyset$, then:

$$T' \models \bigwedge_{w \in T} (\top \subseteq \chi_w^k) \iff \forall w \in T : T' \models \top \subseteq \chi_w^k$$
$$\iff \forall w \in T \ \forall v' \in T' \ \exists w' \in T' : v' \models \top \iff w' \models \chi_w^k$$
$$\iff \forall w \in T \ \exists w' \in T' : w' \models \chi_w^k \qquad (Since \ T' \neq \emptyset)$$
$$\iff \forall w \in T \ \exists w' \in T' : w \rightleftharpoons_k w'. \qquad (Theorem \ 5.3.3)$$

(ii) Let

$$\eta_{M,T}^{\mathsf{X},k} \coloneqq \bigvee_{w \in T} \chi_{M,w}^{\mathsf{X},k}$$

where we stipulate $\bigvee \emptyset = \bot$.

$$T' \vDash \bigvee_{w \in T} \chi_w^k \iff T' = \bigcup_{w \in T} T'_w \text{ where } T'_w \vDash \chi_w^k$$
$$\iff \forall w' \in T' \exists w \in T : w' \vDash \chi_w^k \qquad \text{(By flatness)}$$
$$\iff \forall w' \in T' \exists w \in T : w \rightleftharpoons_k w'. \qquad \text{(Theorem 5.3.3)}$$

(iii) Let

$$\theta_{M,T}^{\mathsf{X},k} \coloneqq \eta_{M,T}^{\mathsf{X},k} \wedge \zeta_{M,T}^{\mathsf{X},k}$$

If $T' = \emptyset$, then $T' \models \theta_T^k$ by the empty team property. Else if $T' \neq \emptyset$, then $T' \models \eta_T^k \land \zeta_T^k$ if and only if the back and forth conditions in Definition 5.3.4 hold (by items (i) and (ii)), which is the case if and only if $T \rightleftharpoons_k T'$.

We call the characteristic formulas θ_T^k for teams obtained in Lemma 5.3.7 (iii) *Hintikka formulas for teams*. As with standard Hintikka formulas, it is easy to see that for a given finite X, there are only a finite number of non-equivalent *k*th Hintikka formulas for teams.

We are now ready to show also the inclusion $\mathbb{U}_X \subseteq \{ \|\varphi\|_X | \varphi \in ML(\subseteq) \text{ and } Prop(\varphi) = X \}.$

5.3.8. THEOREM ([73]). $\mathbb{U} = \|\mathrm{ML}(\subseteq)\|$. That is, for each finite $X \subseteq \operatorname{Prop}$, $\mathbb{U}_X = \{\mathcal{C} \text{ over } X \mid \mathcal{C} \text{ is union closed, has the empty team property, and is invariant under <math>\rightleftharpoons_k^X \text{ for some } k \in \mathbb{N}\} = \{\|\varphi\|_X \mid \varphi \in \mathrm{ML}(\subseteq) \text{ and } \operatorname{Prop}(\varphi) = X\}.$

Proof:

The direction \supseteq follows by Theorem 5.3.6 and the fact that formulas in ML(\subseteq) are union closed and have the empty team property.

For the direction \subseteq , let $C \in \mathbb{U}_X$, and let $k \in \mathbb{N}$ be such that C is invariant under \rightleftharpoons_k^X . Let

$$\varphi' \coloneqq \bigvee_{(M,T)\in\mathcal{C}} \theta_{M,T}^{\mathsf{X},k},$$

where $\theta_{M,T}^{X,k}$ is defined as in the proof of Lemma 5.3.7 (iii). Since there are only finitely many non-equivalent *k*th Hintikka formulas for teams for a given finite X, we may assume the disjunction in φ' to be finite and φ' to be well-defined. We show $C = \|\varphi'\|_X$. Note that since both C and $\|\varphi'\|_X$ have the empty team property, neither is empty.

Let $(M,T) \in \mathcal{C}$. Clearly $T \rightleftharpoons_k T$, so by Lemma 5.3.7 (iii), $T \models \theta_T^k$, whence $T \models \varphi'$.

Now let $M', T' \models \varphi'$. Then there are subteams $T'_T \subseteq T'$ such that $T' = \bigcup_{(M,T) \in \mathcal{C}} T'_T$ and $M', T'_T \models \theta^k_T$. By Lemma 5.3.7 (iii) it follows that for a given $(M,T) \in \mathcal{C}$, either $M, T \rightleftharpoons_k M', T'_T$ or $T'_T = \emptyset$. If $T'_T = \emptyset$, then by the empty team property $(M', T'_T) \in \mathcal{C}$. If

156

 $M,T \rightleftharpoons_k M',T'_T$, then since C is invariant under *k*-bisimulation, $(M',T'_T) \in C$. So for all $(M,T) \in C$ we have $(M',T'_T) \in C$; then since C is closed under unions, we conclude that $(M',T') \in C$.

It follows from the proof of Theorem 5.3.8 that each $ML(\subseteq)$ -formula is equivalent to a formula in the normal form

$$\bigvee_{(M,T)\in\mathcal{C}} \boldsymbol{\theta}_T^k = \bigvee_{(M,T)\in\mathcal{C}} (\bigvee_{w\in T} \boldsymbol{\chi}_w^k \wedge \bigwedge_{w\in T} (\top \subseteq \boldsymbol{\chi}_w^k)).$$
(NF)

Note that in this normal form, the extended inclusion atoms $\top \subseteq \chi_w^k$ play a substantial role that cannot be replicated using inclusion atoms *simpliciter*, i.e., inclusion atoms $p_1 \dots p_n \subseteq q_1 \dots q_n$ with propositional symbols only. It was observed in [137] that the variant of propositional inclusion logic with only these simpler inclusion atoms is strictly less expressive than extended propositional inclusion logic, as, e.g., in one propositional symbol p, the former contains only one (trivial) inclusion atom $p \subseteq p$ and is thus flat, while the latter can express non-flat properties using extended inclusion atoms atoms such as $\top \subseteq p$ and $\perp \subseteq p$. The analogous fact clearly also holds in the modal case.

Observe also that only unary top inclusion atoms (atoms of the form $\top \subseteq \alpha$) are required in the above normal form to express any given property in U. Given that $\top \subseteq \alpha \equiv \nabla \alpha \equiv \nabla \alpha$ (Fact 5.2.8), we can derive the expressive completeness of ML(∇) as well as that of ML(∇) as a corollary to Theorem 5.3.8.

5.3.9. THEOREM. $\mathbb{U} = \|ML(\nabla)\| = \|ML(\nabla)\|$.

And we clearly have the following normal forms for these logics:

$$\bigvee_{(M,T)\in\mathcal{C}} (\bigvee_{w\in T} \chi_w^k \wedge \bigwedge_{w\in T} \nabla \chi_w^k).$$
 (\(\normalf NF)\)

$$\bigvee_{(M,T)\in\mathcal{C}} (\bigvee_{w\in T} \chi_w^k \wedge \bigwedge_{w\in T} \nabla \chi_w^k).$$
 (\(\not\)NF)

The expressive completeness of $ML(\nabla)$ was also proved in [73]; we have added the result for $ML(\nabla)$. Let us briefly comment on how the structure of the normal form motivates the introduction of the operator ∇ . Given a team *T*, the formula $\zeta_T^k = \bigwedge_{w \in T} (\top \subseteq \chi_w^k)$ is intended to express the forth-condition of \Rightarrow_k from *T* to a team *T'* in the sense that $T' \models \zeta_T^k$ iff for all $w \in T$ there is a $w' \in T$ such that $w \Rightarrow_k w'$ (see the proof of Lemma 5.3.7 (i)). The formula $\bigwedge_{w \in T} \nabla \chi_w^k$ articulates precisely what *T'* needs to satisfy in order to fulfill this condition. Inclusion atoms are clearly stronger than is strictly necessary if we only wish to express this condition. The operator ∇ is also *prima facie* stronger than ∇ in the sense that for any $\varphi \in ML(\nabla)$, there is an $\alpha \in ML$ such that $\nabla \varphi \equiv \nabla \alpha$,⁵ whereas, for instance, there is no $\alpha \in ML$ such that $\nabla(\nabla p \wedge \nabla q) \equiv \nabla \alpha$.

⁵This follows from the fact that ML is expressively complete for the class of flat properties invariant under *k*-bisimulation for some $k \in \mathbb{N}$ (see, e.g., [134]). A team property \mathcal{P} is *flat* if $(M,T) \in \mathcal{P} \iff$ $(M,\{w\}) \in \mathcal{P}$ for all $w \in T$. Given any $\varphi \in ML(\nabla)$, $\{(M,T) \mid M, \{w\} \models \varphi$ for all $w \in T\}$ is clearly flat and invariant under *k*-bisimulation for $k = md(\varphi)$, so by the expressive completeness fact we have $\{(M,T) \mid M, \{w\} \models \varphi \text{ for all } w \in T\} = \|\alpha\|$ for some $\alpha \in ML$, and then $\nabla \varphi \equiv \nabla \alpha$.

the singular might, in a sense, cancels out non-classical content within its scope. It is interesting, therefore, that $ML(\nabla)$ has the same expressive power as the other two logics.

One important corollary to the results in this section is that the logics we consider are compact. This follows from the fact that any team property invariant under bounded bisimulation can be expressed in classical first-order logic—see [89]. (It should be noted that for many team-based logics, compactness formulated in terms of satisfiability need not coincide with compactness formulated in terms of entailment, but the logics we consider are compact in both senses; see Section 5.6 for further discussion.) It also follows readily from the expressive completeness results that each of the three logics admits uniform interpolation. It was proved in [43] that any team-based modal logic which is *local* and *forgetting* admits uniform interpolation. The logics we consider are all local, and it follows from the expressive completeness theorems that they are also forgetting; they therefore admit uniform interpolation. We refer the reader to [43] for the detailed proof and more discussion about the notion of uniform interpolation; see [137] for clarification on the role of locality.

5.4 Axiomatizations

In this section, we introduce sound and complete natural deduction systems for the logics $ML(\subseteq)$, $ML(\bigtriangledown)$ and $\mathcal{ML}(\bigtriangledown)$. We prove the completeness of the axiomatizations by means of a strategy commonly used for propositional and modal team-based logics which involves showing that each formula is provably equivalent to its normal form. This strategy is used in [137] for the propositional inclusion logic system; we also adapt many of the lemmas and other details of the proof presented in [137].

5.4.1 ML(⊆)

We start by axiomatizing our core logic, modal inclusion logic $ML(\subseteq)$. Our natural deduction system for $ML(\subseteq)$ comprises standard rules for connectives and modalities for the smallest normal modal logic \mathcal{K} , modified to account for special features of our setting such as the failure of downward closure, as well as rules governing the behaviour of inclusion atoms and their interaction with the other connectives. Note that since $ML(\subseteq)$ is not closed under uniform substitution (see Section 5.2), the system does not admit the usual uniform substitution rule.

5.4.1. DEFINITION. The natural deduction system for $ML(\subseteq)$ consists of all axioms and rules presented in Tables 5.1-5.3. We write $\Gamma \vdash_{ML(\subseteq)} \varphi$ (or simply $\Gamma \vdash \varphi$) if φ is derivable from formulas in Γ using this system. We write $\varphi \vdash \psi$ for $\{\varphi\} \vdash \psi$ and $\vdash \varphi$ for $\emptyset \vdash \varphi$, and say that φ and ψ are *provably equivalent*, denoted by $\varphi \dashv \vdash \psi$, if $\varphi \vdash \psi$ and $\psi \vdash \varphi$.



Table 5.1: Rules for classical modal logic.

Table 5.1 lists the rules which operate only on the connectives and operators of ML. The soundness of the disjunction elimination rule as well as that of the negation rules involving assumptions requires that the undischarged assumptions in the subderivations are downward closed, hence the restriction to ML-formulas.

Table 5.2 lists the propositional rules for inclusion atoms. All rules in this table are adapted from rules or results in the propositional inclusion logic system in [137], but we have considerably simplified the propositional system (see Proposition 5.4.4, in which we show that the full set of rules from [137] is derivable in our system). The rule $\subseteq_{\neg}E$ captures the fact that $x \subseteq a$ and $\neg a^x$ lead to a contradiction, since (for a nonempty team) $x \subseteq a$ implies that a^x is true somewhere in the team (see Proposition 5.2.4), while $\neg a^x$ implies that a^x is not true anywhere in the team. The rules $\subseteq Ext$ and $\subseteq Rdt$ allow us to reduce arbitrary inclusion atoms to equivalent formulas in which



Table 5.2: Rules for inclusion.



Table 5.3: Rules for modal operators and inclusion.

all non-classical subformulas are primitive inclusion atoms (see Lemma 5.2.7) and to utilize the properties of classical formulas to derive properties of inclusion atoms (see Proposition 5.4.4). The rule $\lor_{\subseteq} E$ models the upward closure (modulo the empty team property) of primitive inclusion atoms: if $T \vDash x \subseteq a \lor \psi$ and $T \notin \psi$, then $T = S \cup U$ where $S \vDash x \subseteq a$, $U \vDash \psi$ and $S \neq \emptyset$, whence also $T \vDash x \subseteq a$. Primitive inclusion atoms are not downward closed so we do not in general have the converse direction: $(\varphi \lor \psi) \land x \subseteq a \notin \emptyset$
5.4. Axiomatizations

 $(\varphi \land x \subseteq a) \lor \psi$. However, $(\varphi \lor \psi) \land x \subseteq a \models ((\varphi \lor a^x) \land x \subseteq a) \lor \psi$ does hold, as reflected in the rule \subseteq Distr.

The rules in Table 5.3 concern modal operators and inclusion atoms. The rule \bigotimes_{\subseteq} Distr allows us to distribute the diamond over the inclusion atom. The converse direction of this rule is not sound, as, e.g., $\top \subseteq \bigotimes_{p} \neq \bigotimes_{q} (\top \subseteq p)$ (consider a team with a world $w \models \bigotimes_{p}$ but without a successor team). If we ensure that there is a successor team by requiring $\bigotimes_{\varphi} \phi$ to hold for some φ , the converse is sound. The rule \subseteq_{\bigotimes} Distr generalizes this fact, and indeed we have:

$$\vdash \otimes (\top \subseteq \mathsf{a}^{\mathsf{X}}) \tag{($Mon)}$$

The rule $\bigotimes \square_{\subseteq} \text{Exc}$ allows us to derive a box formula from a top inclusion formula with a diamond formula on the right. The converse is not sound—consider a nonempty team T with $R[T] = \emptyset$. By the empty team property, $T \models \square(\top \subseteq \alpha)$, whereas $T \notin \top \subseteq \bigotimes \alpha$, since the worlds in T have no accessible worlds. We can ensure that $R[T] \neq \emptyset$ in case $T \neq \emptyset$ by requiring $\top \subseteq \bigotimes \beta$ to hold for some β —this yields the rule $\square \bigotimes_{\subseteq} \text{Exc}$. The rule $\square \lor_{\subseteq} \text{E}$ is similar to $\lor_{\subseteq} \text{E}$ in Table 5.2, but it applies to box formulas.

5.4.2. THEOREM (Soundness). *If* $\Gamma \vdash \varphi$, *then* $\Gamma \models \varphi$.

Proof:

Soundness proofs for most rules in Tables 5.1 and 5.2 can be found in [137]. The soundness of \bigotimes_{\subseteq} Distr, $\square \bigotimes_{\subseteq} Exc$ and $\bigotimes_{\square_{\subseteq}} Exc$ (Table 5.3) is easy to show using Proposition 5.2.4. We only give detailed proofs for \subseteq_{\bigotimes} Distr and $\square \lor_{\subseteq} E$ (Table 5.3).

- $(\subseteq_{\bigotimes} \text{Distr}) \text{ It suffices to show } \bigotimes \varphi \land \land_{1 \le i \le n} (x_i \subseteq \bigotimes \alpha_i) \vDash \bigotimes ((\varphi \lor \bigvee_{1 \le i \le n} \alpha_i^{x_i}) \land \land_{1 \le i \le n} (x_i \subseteq \alpha_i)). \text{ Suppose that } T \vDash \bigotimes \varphi \land \land_{i \in I} (x_i \subseteq \bigotimes \alpha_i). \text{ If } T = \emptyset, \text{ it satisfies the conclusion} \text{ by the empty team property so we may suppose } T \ne \emptyset. \text{ Then there is some } S \text{ such that } TRS \text{ and } S \vDash \varphi, \text{ and for each } 1 \le i \le n, T \vDash x_i \subseteq \bigotimes \alpha_i. \text{ If } x_i = \top, \text{ then there is some } v_i \in R[T] \text{ such that } v_i \vDash \alpha_i, \text{ i.e., } v_i \vDash \alpha_i^{x_i}. \text{ If, on the other hand,} x_i = \bot, \text{ then by Proposition 5.2.4 there is some } w_i \in T \text{ such that } w_i \vDash \neg \bigotimes \alpha_i, \text{ i.e.,} w_i \vDash [m \neg \alpha_i, we have that v_i \vDash \neg \alpha_i, \text{ i.e., } v_i \vDash \alpha_i^{x_i}. \text{ Let } S' \coloneqq S \cup \{v_i\}_{1 \le i \le n}. \text{ Clearly } S' \vDash (\varphi \lor \bigvee_{1 \le i \le n} \alpha_i^{x_i}) \land \land_{1 \le i \le n} \alpha_i^{x_i}) \land \land_{1 \le i \le n} \alpha_i^{x_i}) \land \land_{1 \le i \le n} \alpha_i^{x_i} \land \land_{1 \le i \le n} \alpha_i^{x_i}) \land \land_{1 \le i \le n} \alpha_i^{x_i})$
- $(\square \lor_{\subseteq} E)$ Suppose that $\Gamma, \top \subseteq \otimes a^{\times} \models \chi$, and $\Gamma, \square \psi \models \chi$. Let $T \models \square (x \subseteq a \lor \psi)$ and $T \models \gamma$ for all $\gamma \in \Gamma$. We show that $T \models \chi$.

We have that $R[T] \models x \subseteq a \lor \psi$, so $R[T] = T_1 \cup T_2$ where $T_1 \models x \subseteq a$ and $T_2 \models \psi$. If $T_1 = \emptyset$, then $T_2 = R[T]$, so $R[T] \models \psi$, whence $T \models \Box \psi$; therefore also $T \models \chi$. Else if $T_1 \neq \emptyset$, then by $T_1 \models x \subseteq a$ and Proposition 5.2.4 it follows that there is some $v \in T_1$ such that $v \models a^x$. By $v \in R[T]$, there is a $w \in T$ with wRv, whence $w \models \otimes a^x$. Therefore $T \models \top \subseteq \otimes a^x$, whence also $T \models \chi$.

It is easy to verify that the rules in Table 5.1 constitute a system that is equivalent to other axiomatizations of \mathcal{K} ,⁶ and so, given soundness, Proposition 5.2.3, and the fact that \mathcal{K} is complete for the class of all Kripke models, we have:

5.4.3. PROPOSITION (Classical completeness). Let $\Gamma \cup \{\alpha\}$ be a set of ML-formulas. *Then*

$$\Gamma \vDash \alpha \iff \Gamma \vdash_{\mathrm{ML}(\subseteq)} \alpha.$$

In the following proposition we derive some useful properties of inclusion atoms. These properties allow us to manipulate the atoms with ease, and we occasionally make use of them without explicit reference to this proposition. All items in the proposition were included as rules in the propositional system in [137]; the proposition shows that the simpler propositional system yielded by the propositional fragment of the rules in Tables 5.1 and 5.2 suffices.⁷

5.4.4. PROPOSITION.

- (*i*) $a \subseteq b, b \subseteq c \vdash a \subseteq c$.
- $(ii) a_0a_1a_2 \subseteq b_0b_1b_2 \vdash a_1a_0a_2 \subseteq b_1b_0b_2.$
- (*iii*) $a_0a_1 \subseteq b_0b_1 \vdash a_0a_0a_1 \subseteq b_0b_0b_1$.
- (*iv*) $ab \subseteq cd \vdash a \subseteq c$.
- (v) $a \subseteq b, \alpha(b) \vdash \alpha(a)$, where α , or $\alpha(c)$, is classical and propositional, the propositional symbols and constants occurring in $\alpha(c)$ are among those in c, and $\alpha(a)$ and $\alpha(b)$ denote the result of replacing each element of c in α with the corresponding element in a and b, respectively.

Proof:

(i) By ⊆Ext, it suffices to derive ¬a^x ∨ x ⊆ c for all x ∈ {T,⊥}^{|a|}. For a given x, we have a ⊆ b ⊢ ¬a^x ∨ x ⊆ b and b ⊆ c ⊢ ¬b^x ∨ x ⊆ c by ⊆Rdt. By ∨_⊆E it now suffices to show:

(a) $\neg a^{x} \vdash \neg a^{x} \lor x \subseteq c$.

⁶For a proof that a similar system is equivalent to a Hilbert-style system for \mathcal{K} , see [134]. Note that some of the rules in Table 5.1 such as $\otimes \vee$ Distr are derivable for the classical fragment of our axiomatization and are therefore not necessary for this equivalence.

⁷The system in [137] did not feature our $\subseteq_{\neg}E$ as a rule; it was instead shown to be derivable in that system using a rule corresponding to our Proposition 5.4.4 (v). An alternative simplified propositional system would then replace $\subseteq_{\neg}E$ with Proposition 5.4.4 (v). The rule corresponding to Proposition 5.4.4 (v) was originally introduced in the first-order system in [64].

- (b) $x \subseteq b, \neg b^x \vdash \neg a^x \lor x \subseteq c$.
- (c) $x \subseteq c \vdash \neg a^x \lor x \subseteq c$.

We have that (a) and (c) follow by $\lor I$, and (b) by $\subseteq _E$.

(ii) By \subseteq Ext, it suffices to derive $\neg a_1a_0a_2^{x_1x_0x_2} \lor x_1x_0x_2 \subseteq b_1b_0b_2$ for all $x_1x_0x_2 \in \{\top, \bot\}^{|a_1a_0a_2|}$. For a given $x_1x_0x_2$, we have $a_0a_1a_2 \subseteq b_0b_1b_2 \vdash \neg a_0a_1a_2^{x_0x_1x_2} \lor x_0x_1x_2 \subseteq b_0b_1b_2$ by \subseteq Rdt. By Proposition 5.4.3 we have that $\neg a_0a_1a_2^{x_0x_1x_2} \dashv \vdash \neg a_1a_0a_2^{x_1x_0x_2}$, and therefore also

$$\neg a_0 a_1 a_2^{x_0 x_1 x_2} \lor x_0 x_1 x_2 \subseteq b_0 b_1 b_2 \vdash \neg a_1 a_0 a_2^{x_1 x_0 x_2} \lor x_0 x_1 x_2 \subseteq b_0 b_1 b_2. \quad (*)$$

We also have $b_1b_0b_2 \subseteq b_1b_0b_2$ by \subseteq Id, and then $\neg b_1b_0b_2^{x_1x_0x_2} \lor x_1x_0x_2 \subseteq b_1b_0b_2$ by \subseteq Rdt. By Proposition 5.4.3, we have $\neg b_0b_1b_2^{x_0x_1x_2} \dashv \neg \neg b_1b_0b_2^{x_1x_0x_2}$ so that

$$\neg b_1 b_0 b_2^{x_1 x_0 x_2} \lor x_1 x_0 x_2 \subseteq b_1 b_0 b_2 \vdash \neg b_0 b_1 b_2^{x_0 x_1 x_2} \lor x_1 x_0 x_2 \subseteq b_1 b_0 b_2. \quad (**)$$

By $\lor_{\subseteq} E$ applied to (*) and (**) it now suffices to show:

- (a) $\neg a_1 a_0 a_2^{x_1 x_0 x_2} \vdash \neg a_1 a_0 a_2^{x_1 x_0 x_2} \lor x_1 x_0 x_2 \subseteq b_1 b_0 b_2.$
- $(b) \ x_0x_1x_2 \subseteq b_0b_1b_2, \neg \, b_0b_1b_2^{\,\, x_0x_1x_2} \vdash \neg \, a_1a_0a_2^{\, x_1x_0x_2} \vee \, x_1x_0x_2 \subseteq b_1b_0b_2.$
- (c) $x_1x_0x_2 \subseteq b_1b_0b_2 \vdash \neg a_1a_0a_2^{x_1x_0x_2} \lor x_1x_0x_2 \subseteq b_1b_0b_2$.

We have that (a) and (c) follow by $\lor I$, and (b) by $\subseteq _E$.

- (iii) Similar to (ii).
- (iv) By \subseteq Ext, it suffices to derive $\neg a^x \lor x \subseteq c$ for all $x \in \{\top, \bot\}^{|a|}$. For a given x, we have $\bigwedge_{y \in \{\top, \bot\}^{|b|}} (\neg ab^{xy} \lor xy \subseteq cd)$ by \subseteq Rdt. By repeatedly applying $\lor_{\subseteq} E$, it suffices to show:
 - (a) $xy \subseteq cd \vdash \neg a^x \lor x \subseteq c$ for any given $y \in \{\top, \bot\}^{|b|}$.
 - (b) $\bigwedge_{v \in \{T, +\} | b|} \neg ab^{xy} \vdash \neg a^x \lor x \subseteq c.$

For (b), by Proposition 5.4.3 we have $\vdash \bigvee_{y \in \{\top, \bot\}^{|b|}} b^y$, and by Proposition 5.4.3 and $\lor I$, $\bigwedge_{y \in \{\top, \bot\}^{|b|}} \neg ab^{xy}$, $\bigvee_{y \in \{\top, \bot\}^{|b|}} b^y \vdash \neg a^x \lor x \subseteq c$ (for a given $y, \neg ab^{xy} = \neg (a^x \land b^y) \dashv \vdash \neg a^x \lor \neg b^y$, and $\neg a^x \lor \neg b^y$, $b^y \vdash \neg a^x$).

For (a), we have $c \subseteq c$ by \subseteq Id, and then $\neg c^{x} \lor x \subseteq c$ by \subseteq Rdt. By Proposition 5.4.3, we have $\neg c^{x} \lor x \subseteq c \vdash \bigwedge_{z \in \{\top, \bot\}^{|b|}} \neg cd^{xz} \lor x \subseteq c$ (for a given z, we have $\neg c^{x} \vdash \neg c^{x} \lor \neg d^{z} \dashv \vdash \neg (c^{x} \land d^{z}) = \neg cd^{xz}$). By $\lor_{\subseteq} E$ it therefore suffices to show:

- (c) $x \subseteq c \vdash \neg a^x \lor x \subseteq c$.
- (d) $xy \subseteq cd, \bigwedge_{z \in \{\top, \bot\}^{|b|}} \neg cd^{xz} \vdash \neg a^x \lor x \subseteq c \text{ for any given } y \in \{\top, \bot\}^{|b|}.$

We have that (c) follows by $\lor I$, and (d) by $\subseteq _E$.

- (v) Let n := |a| = |b| = |c|. Define a function f : {⊥, T}ⁿ → {⊥, T} by f(y) = T : ⇔
 w ⊨ α(y) for some w (note that since the propositional symbols and constants appearing in α(c) are among those in c, the definition of f is independent of the choice of w). From a ⊆ b we have ∧_{{x∈{T,⊥}ⁿ|f(x)=⊥}}(¬a^x ∨ x ⊆ b) by ⊆Rdt. By repeatedly applying ∨_⊆E, it suffices to show:
 - (a) $\bigwedge_{\{\mathbf{x}\in\{\top,\bot\}^n\mid f(\mathbf{x})=\bot\}} \neg \mathbf{a}^{\mathbf{x}} \vdash \alpha(\mathbf{a}).$
 - (b) $\alpha(b), x \subseteq b \vdash \alpha(a)$ for any given $x \in \{\top, \bot\}^n$ with $f(x) = \bot$.

To show (a) and (b), we show (c):

(c)
$$\bigwedge_{\{\mathbf{x}\in\{\mathsf{T},\bot\}^n\mid f(\mathbf{x})=\bot\}} \neg \mathbf{c}^{\mathbf{x}} \equiv \alpha(\mathbf{c}).$$

For the left-to-right direction, let $w \models \bigwedge_{\{x \in \{\top, \bot\}^n | f(x) = \bot\}} \neg c^x$ and assume for contradiction that $w \not\models \alpha(c)$. Then for $y \in \{\bot, \top\}^n$ given by $y_i = \top \iff w \models c_i$ we have $w \not\models \alpha(y)$ so $f(y) = \bot$, whence $w \models \neg c^y$. But clearly by the definition of y we have $w \models c^y$, a contradiction.

For the right-to-left direction, let $w \models \alpha(c)$ and let $x \in \{\top, \bot\}^n$ be such that $f(x) = \bot$. Then for some x_i we must have $w \models c_i \iff x_i = \bot$ (lest we have $w \models \alpha(x)$ whence $f(x) = \top$, contradicting $f(x) = \bot$), and therefore $w \models \neg c^x$.

Given (c), (a) follows immediately by Proposition 5.4.3.

For (b), by (c) and Proposition 5.4.3, $\alpha(b)$ gives us $\neg b^x$. By $\subseteq \neg E$ applied to $\neg b^x$ and $x \subseteq b$, we have $\alpha(a)$.

We move on to the completeness proof. Our strategy involves showing that each $ML(\subseteq)$ -formula is provably equivalent to a formula in the normal form presented in Section 5.3. Once all formulas are in normal form, completeness follows from the semantic and proof-theoretic properties of formulas in this form. That is, we show:

5.4.5. LEMMA (Provable equivalence of the normal form). For any formula $\varphi(X)$ in ML(\subseteq) and $k \ge md(\varphi)$, there is a (finite, nonempty) property C (over X) such that

$$\varphi \dashv \vdash \bigvee_{(M,T) \in \mathcal{C}} \theta_{M,T}^{\mathsf{X},k}, \qquad i.e., \qquad \varphi \dashv \vdash \bigvee_{(M,T) \in \mathcal{C}} (\bigvee_{w \in T} \chi_{M,w}^{\mathsf{X},k} \land \bigwedge_{w \in T} (\top \subseteq \chi_{M,w}^{\mathsf{X},k})).$$

The proof of the above lemma is more involved. We postpone it for now, and first focus on how it allows us to prove completeness. To that end, note the following immediate consequence of classical completeness: the kth Hintikka formulas of two k-bisimilar pointed models are provably equivalent.

5.4.6. LEMMA. If $M, w \rightleftharpoons_k M', u$, then $\chi_{M,w}^k \dashv \chi_{M',u}^k$.

Proof:

By Theorem 5.3.3, $w \rightleftharpoons_k u$ implies $w \equiv_k u$ and then $\chi_w^k \dashv \vdash \chi_u^k$ by Proposition 5.4.3. \Box

Recall from footnote 4 in Section 5.3 that in defining the (k+1)th Hintikka formula $\chi_{M,w}^{X,k+1}$, we choose an arbitrary finite set *T* of representatives from the possibly infinite set R[w]. The above lemma shows that two *k*-bisimilar worlds give rise to provably equivalent Hintikka formulas, so the specific choice of the representatives also makes no difference in our proof system. Therefore, we may continue our practice of using arbitrary finite representatives in the definition of $\chi_{M,w}^{X,k+1}$ in the context of the proof system. Similar results hold for other formulas employing finite representatives, though we henceforth omit explicit discussion of these representatives. Our next aim is to prove the analogous result for Hintikka formulas for teams—that is:

5.4.7. LEMMA. If $M, T \rightleftharpoons_k M', S$, then $\theta_{M,T}^k \dashv \theta_{M',S}^k$.

In order to establish this, we first prove some results concerning primitive inclusion atoms. We also show another unrelated result concerning primitive inclusion atoms ((iii) in the lemma below) which is required in the sequel.

5.4.8. LEMMA.

- (i) $\alpha_1, \ldots, \alpha_n \vdash \forall^n \subseteq \alpha_1 \ldots \alpha_n$, where \forall^n denotes the sequence \forall, \ldots, \forall of length n.
- (*ii*) If $\alpha_i \vdash \beta_i$, then $\top^n \subseteq \alpha_1 \ldots \alpha_i \ldots \alpha_n \vdash \top^n \subseteq \alpha_1 \ldots \beta_i \ldots \alpha_n$.
- (*iii*) $\top^n \subseteq \alpha_1^{x_1} \dots \alpha_n^{x_n} \vdash x_1 \dots x_n \subseteq \alpha_1 \dots \alpha_n$.

Proof:

- (i) $\top \subseteq \top$ by \subseteq Id; using α_n and $\top \subseteq \top$ we get $\top \top \subseteq \alpha_n \top$ by \subseteq Exp; and then $\top \top \subseteq \alpha_n \top \vdash \top \subseteq \alpha_n$ by Proposition 5.4.4 (iv). We repeat the \subseteq Exp-step with $\alpha_{n-1}, \ldots, \alpha_1$.
- (ii) By Proposition 5.4.3, $\vdash \neg(\alpha_1 \land \dots \land \alpha_n) \lor (\alpha_1 \land \dots \land \alpha_n)$. From the assumption $\alpha_i \vdash \beta_i$ together with item (i) we have that $\alpha_1 \land \dots \land \alpha_n \vdash \alpha_1 \land \dots \land \beta_i \land \dots \land \alpha_n \vdash \neg^n \subseteq \alpha_1 \dots \beta_i \dots \alpha_n$. Thus $\neg(\alpha_1 \land \dots \land \alpha_n) \lor (\alpha_1 \land \dots \land \alpha_n) \vdash \neg(\alpha_1 \land \dots \land \alpha_n) \lor (\top^n \subseteq \alpha_1 \dots \beta_i \dots \alpha_n)$. By \lor_{\subseteq} E, it suffices to show:
 - (a) $\top^n \subseteq \alpha_1 \dots \alpha_n, \neg(\alpha_1 \wedge \dots \wedge \alpha_n) \vdash \top^n \subseteq \alpha_1 \dots \beta_i \dots \alpha_n$
 - (b) $\forall^n \subseteq \alpha_1 \dots \alpha_n, \forall^n \subseteq \alpha_1 \dots \beta_i \dots \alpha_n \vdash \forall^n \subseteq \alpha_1 \dots \beta_i \dots \alpha_n$

We have that (a) follows by $\subseteq E$, and (b) is immediate.

(iii) By \subseteq Id, we have $\alpha_1 \dots \alpha_n \subseteq \alpha_1 \dots \alpha_n$ and then by \subseteq Rdt, $\neg(\alpha_1 \dots \alpha_n^{x_1 \dots x_n}) \lor x_1 \dots x_n \subseteq \alpha_1 \dots \alpha_n$. By \lor_{\subseteq} E it suffices to show $\neg(\alpha_1 \dots \alpha_n^{x_1 \dots x_n}), \top^n \subseteq \alpha_1^{x_1} \dots \alpha_n^{x_n} \vdash x_1 \dots x_n \subseteq \alpha_1 \dots \alpha_n$ and $x_1 \dots x_n \subseteq \alpha_1 \dots \alpha_n \vdash x_1 \dots x_n \subseteq \alpha_1 \dots \alpha_n$. The latter is immediate; the former follows by $\subseteq \neg$ E since $\alpha_1 \dots \alpha_n^{x_1 \dots x_n} = (\alpha_1^{x_1} \dots \alpha_n^{x_n})^{\top^n}$.

Proof of Lemma 5.4.7:

Suppose that $T \rightleftharpoons_k S$. If $T = S = \emptyset$, then $\theta_T^k = \theta_S^k$. Otherwise, both T and S are nonempty. By symmetry it suffices to show $\theta_T^k \vdash \theta_S^k$. We have that $\theta_T^k \vdash \bigvee_{w \in T} \chi_w^k$ by $\land E$. Let $w \in T$. By $T \rightleftharpoons_k S$, there is some $u \in S$ such that $w \rightleftharpoons_k u$, so that by Lemma 5.4.6, $\chi_w^k \dashv \chi_u^k$. Therefore, by $\lor I$ and $\lor E$, $\bigvee_{w \in T} \chi_w^k \vdash \bigvee_{u \in S} \chi_u^k$. Now let $u \in S$. By $T \rightleftharpoons_k S$, there is some $w_u \in T$ such that $w_u \rightleftharpoons_k u$, so that by Lemma 5.4.6, $\chi_{w_u}^k \dashv \chi_u^k$. By $\land E$ and Lemma 5.4.8 (ii), $\theta_T^k \vdash \top \subseteq \chi_{w_u}^k \vdash \top \subseteq \chi_u^k$. Therefore by $\land I$, $\theta_T^k \vdash \bigwedge_{u \in S} (\top \subseteq \chi_u) = \theta_S^k$.

Next we prove some important semantic facts concerning the relationship between formulas in normal form and disjoint unions. The *disjoint union* $\biguplus_{i \in I} M_i$ of the models M_i ($i \in I$) is defined as usual, where in particular we recall that the domain of the disjoint union is defined to be $\biguplus_{i \in I} W_i = \bigcup_{i \in I} (W_i \times \{i\})$. We also extend the notion to properties: the *disjoint union* of a property $C = \{(M_i, T_i) \mid i \in I\}$ is $\oiint C = (\biguplus_{i \in I} M_i, \biguplus_{i \in I} T_i)$, where $\biguplus_{i \in I} T_i = \bigcup_{i \in I} (T_i \times \{i\})$. To simplify notation, we define the disjoint union $\oiint \emptyset$ of the empty property to be (M^*, \emptyset) for some fixed model M^* . We refer to a world (w, i) in a disjoint union as simply w, and to a team $T \times \{i\}$ as simply T. The following result is standard (see, for instance, [24]):

5.4.9. PROPOSITION. For any $i \in I$, any $w \in W_i$, and any $k \in \mathbb{N}$: $M_i, w \rightleftharpoons_k \biguplus_{i \in I} M_i, w$. *Therefore, for any* $T \subseteq W_i$: $M_i, T \rightleftharpoons_k \biguplus_{i \in I} M_i, T$.

We have:

5.4.10. LEMMA. $M, S \models \bigvee_{(M',T) \in \mathcal{C}} \theta_{M',T}^k$ iff $M, S \rightleftharpoons_k \bigcup \mathcal{C}'$ for some $\mathcal{C}' \subseteq \mathcal{C}$.

Proof:

 $\implies: \text{ If } M, S \vDash \bigvee_{(M',T) \in \mathcal{C}} \theta_{M',T}^k, \text{ then } mS = \bigcup_{(M',T) \in \mathcal{C}} S_{M',T} \text{ where } S_{M',T} \vDash \theta_{M',T}^k. \text{ By Lemma 5.3.7 (iii), either } S_{M',T} \rightleftharpoons_k T \text{ or } S_{M',T} = \emptyset. \text{ Let } \mathcal{C}' \coloneqq \{(M',T) \mid S_{M',T} \rightleftharpoons_k T\}.$ Then $S = \bigcup_{(M',T) \in \mathcal{C}} S_{M',T} = \bigcup_{(M',T) \in \mathcal{C}'} S_{M',T}. \text{ By Proposition 5.4.9, we have:}$

$$M, S = M, \bigcup_{(M',T) \in \mathcal{C}'} S_{M',T} \rightleftharpoons_k \biguplus \{ (M',T) \mid T \in \mathcal{C}' \} = \biguplus \mathcal{C}'.$$

Note that if $S_{M',T} = \emptyset$ for all $(M',T) \in C$, then $S = \emptyset$ and $\forall C' = \forall \emptyset = (M^*,\emptyset)$ by our convention.

 $\longleftrightarrow: \text{Let } \mathcal{C}' = \{ (M_i, T_i) \mid i \in I \}. \text{ If } M, S \rightleftharpoons_k \bigcup \mathcal{C}', \text{ then by Lemma 5.3.5 (iii), there} \\ \text{are subteams } S_i \subseteq S \text{ such that } \bigcup_{i \in I} S_i = S \text{ and } M, S_i \rightleftharpoons_k \bigcup_{j \in I} M_j, T_i. \text{ By Proposition 5.4.9,} \\ \bigcup_{j \in I} M_j, T_i \rightleftharpoons_k M_i, T_i \text{ so that also } M, S_i \rightleftharpoons_k M_i, T_i. \text{ By Lemma 5.3.7 (iii), } M, S_i \vDash \theta_{M_i, T_i}^k \text{ so that } M, S \vDash \bigvee_{(M_i, T_i) \in \mathcal{C}'} \theta_{M_i, T_i}^k, \text{ and by the empty team property, } M, S \vDash \bigvee_{(M', T) \in \mathcal{C}} \theta_{M', T}^k. \Box$

5.4.11. COROLLARY. $\bigvee_{(M,T)\in\mathcal{C}} \theta_{M,T}^k \models \bigvee_{(M',S)\in\mathcal{D}} \theta_{M',S}^k$ iff for all $(M'',U)\in\mathcal{C}, M'',U \rightleftharpoons_k \cup \mathcal{D}_{(M'',U)}$ for some $\mathcal{D}_{(M'',U)} \subseteq \mathcal{D}$.

Proof:

 \implies : Let $(M'', U) \in C$. By Lemma 5.3.7 (iii), $U \models \theta_U^k$. By the empty team property, $U \models \bigvee_{(M,T)\in C} \theta_T^k$, and then by assumption, $U \models \bigvee_{(M',S)\in D} \theta_S^k$. The result now follows by Lemma 5.4.10.

 $\iff: \text{Let } M''', Y \vDash \bigvee_{(M,T) \in \mathcal{C}} \theta_T^k. \text{ By Lemma 5.4.10, } M''', Y \rightleftharpoons_k \uplus \mathcal{C}' \text{ for some } \mathcal{C}' \subseteq \mathcal{C}.$ By assumption, for each $(M'',U) \in \mathcal{C}'$ there is some $\mathcal{D}_{(M'',U)} \subseteq \mathcal{D}$ such that $M'', U \rightleftharpoons_k$ $\uplus \mathcal{D}_{(M'',U)}.$ Then $M''', Y \rightleftharpoons_k \uplus \{ \uplus \mathcal{D}_{(M'',U)} \mid (M'',U) \in \mathcal{C}' \} \rightleftharpoons_k \uplus \bigcup_{(M'',U) \in \mathcal{C}'} \mathcal{D}_{(M'',U)}, \text{ so that by Lemma 5.4.10, } Y \vDash \bigvee_{(M',S) \in \mathcal{D}} \theta_S^k.$

The final lemma required for the completeness proof is a proof-theoretic analogue of (one direction of) Lemma 5.4.10.

5.4.12. LEMMA. $\theta_{\flat,\mathcal{D}}^k \vdash \bigvee_{(M,S)\in\mathcal{D}} \theta_{M,S}^k$.

Proof:

If \mathcal{D} is empty, then $\theta_{l+\mathcal{D}}^k \dashv \vdash \bot \vdash \bigvee_{(M,S) \in \mathcal{D}} \theta_S^k$. Otherwise let $\mathcal{D} = \{S_1, \ldots, S_n\}$. We have:

$$\boldsymbol{\theta}_{\boldsymbol{t}^{\mathsf{H}}\mathcal{D}}^{k} = \bigvee_{w \in \boldsymbol{t}^{\mathsf{H}}\mathcal{D}} \boldsymbol{\chi}_{w}^{k} \wedge \bigwedge_{w \in \boldsymbol{t}^{\mathsf{H}}\mathcal{D}} (\mathsf{T} \subseteq \boldsymbol{\chi}_{w}^{k}) = \bigvee_{(M,S) \in \mathcal{D}} \bigvee_{w \in S} \boldsymbol{\chi}_{w}^{k} \wedge \bigwedge_{(M,S) \in \mathcal{D}} \bigwedge_{w \in S} (\mathsf{T} \subseteq \boldsymbol{\chi}_{w}^{k}),$$

from which we derive:

$$+ \left(\bigvee_{w \in S_{1}} \chi_{w}^{k} \vee \bigvee_{2 \leq i \leq n} \bigvee_{w \in S_{i}} \chi_{w}^{k} \right) \wedge \bigwedge_{w \in S_{1}} (\top \subseteq \chi_{w}^{k}) \wedge \bigwedge_{2 \leq i \leq n} \bigwedge_{w \in S_{i}} (\top \subseteq \chi_{w}^{k})$$

$$+ \left(\left(\left(\bigvee_{w \in S_{1}} \chi_{w}^{k} \vee \bigvee_{w \in S_{1}} \chi_{w}^{k} \right) \wedge \bigwedge_{w \in S_{1}} (\top \subseteq \chi_{w}^{k}) \right) \vee \bigvee_{2 \leq i \leq n} \bigvee_{w \in S_{i}} \chi_{w}^{k} \right) \wedge \bigwedge_{2 \leq i \leq n} \bigwedge_{w \in S_{i}} (\top \subseteq \chi_{w}^{k})$$

$$+ \left(\theta_{S_{1}}^{k} \vee \bigvee_{2 \leq i \leq n} \bigvee_{w \in S_{i}} \chi_{w}^{k} \right) \wedge \bigwedge_{2 \leq i \leq n} \bigwedge_{w \in S_{i}} (\top \subseteq \chi_{w}^{k})$$

$$= \theta_{S_{1}}^{k} \vee \ldots \vee \theta_{S_{n}}^{k}.$$

5.4.13. THEOREM (Completeness). *If* $\Gamma \vDash \psi$, *then* $\Gamma \vdash \psi$.

Proof:

Suppose that $\Gamma \vDash \psi$. Since ML(\subseteq) is compact, there is a finite subset $\Gamma_0 \subseteq \Gamma$ such that $\Gamma_0 \vDash \psi$. It suffices to show that $\varphi \vdash \psi$ where $\varphi = \bigwedge_{\gamma \in \Gamma_0} \gamma$.

Let $k \ge \max\{md(\varphi), md(\psi)\}$. By Lemma 5.4.5,

$$\varphi \dashv \vdash \bigvee_{(M,T) \in \mathcal{C}} \theta_{M,T}^k \text{ and } \psi \dashv \vdash \bigvee_{(M',S) \in \mathcal{D}} \theta_{M',S}^k.$$

for some finite nonempty properties C and D. By soundness and $\varphi \models \psi$,

$$\bigvee_{(M,T)\in\mathcal{C}} \theta^k_{M,T} \vDash \bigvee_{(M',S)\in\mathcal{D}} \theta^k_{M',S}$$

Let $(M, T) \in C$. By Corollary 5.4.11, $M, T \rightleftharpoons_k \bigcup D_T$ for some $D_T \subseteq D$. By Lemma 5.4.7, Lemma 5.4.12, and $\lor I$,

$$\theta_{M,T}^k \vdash \theta_{\mathfrak{t} \mid \mathcal{D}_T}^k \vdash \bigvee_{(M',S) \in \mathcal{D}_T} \theta_{M',S}^k \vdash \bigvee_{(M',S) \in \mathcal{D}} \theta_{M',S}^k.$$

By $\forall E$ we get that $\bigvee_{(M,T)\in\mathcal{C}} \theta_{M,T}^k \vdash \bigvee_{(M',S)\in\mathcal{D}} \theta_{M',S}^k$, and conclude that $\varphi \vdash \psi$. \Box

of Lemma 5.4.5° provat

We dedicate the rest of this subsection to the proof of Lemma 5.4.5: provable equivalence of the normal form. We first prove some technical lemmas. The rules $\lor_{\subseteq} E$ and $\Box \lor_{\subseteq} E$ can be generalized as follows:

5.4.14. LEMMA. (*i*) If

- (a) $\Gamma, x_1 \subseteq a_1, \ldots, x_k \subseteq a_k \vdash \chi$ and
- (b) $\Gamma, \psi \vdash \chi$,

then Γ , $(x_1 \subseteq a_1 \land \cdots \land x_k \subseteq a_k) \lor \psi \vdash \chi$.

(ii) Let I be a finite index set and for each $i \in I$, let ι_i be a conjunction of finitely many primitive inclusion atoms. If for every nonempty $J \subseteq I$,

$$\Gamma, \bigvee_{j\in J} \varphi_j, \bigwedge_{j\in J} \iota_j \vdash \chi,$$

then Γ , $\forall_{i \in I} (\varphi_i \land \iota_i) \vdash \chi$.

(iii) If

- (a) $\Gamma, \top \subseteq \otimes a_1, \ldots, \top \subseteq \otimes a_k \vdash \chi$ and
- (b) $\Gamma, \Box \psi \vdash \chi$,

then Γ , \Box (($x_1 \subseteq a_1 \land \cdots \land x_k \subseteq a_k$) $\lor \psi$) $\vdash \chi$.

(iv) Let I be a finite index set and for each $i \in I$, let ι_i be a conjunction of finitely many primitive inclusion atoms. For $\iota_i = \bigwedge_{k \in K_i} (\mathsf{x}_k \subseteq \mathsf{a}_k)$, define $\iota_{\otimes i} = \bigwedge_{k \in K_i} (\top \subseteq \otimes \mathsf{a}_k^{\times_k})$. If for every nonempty $J \subseteq I$

$$\Gamma, \Box \bigvee_{j \in J} \varphi_j, \bigwedge_{j \in J} \iota_{\otimes j} \vdash \chi,$$

then Γ , $\Box \bigvee_{i \in I} (\varphi_i \wedge \iota_i) \vdash \chi$.

Proof:

We prove (i) and (ii); the proofs of (iii) and (iv) are similar.

- (i) We have that $(x_1 \subseteq a_1 \land \dots \land x_k \subseteq a_k) \lor \psi \vdash ((x_1 \subseteq a_1 \land \dots \land x_{k-1} \subseteq a_{k-1}) \lor \psi) \land (x_k \subseteq a_k \lor \psi)$ by $\lor E$, $\lor I$, $\land E$, and $\land I$. To show $\Gamma, (x_1 \subseteq a_1 \land \dots \land x_{k-1} \subseteq a_{k-1}) \lor \psi, x_k \subseteq a_k \lor \psi \vdash \chi$, by $\lor_{\subseteq} E$ it suffices to show:
 - (a) Γ , $(x_1 \subseteq a_1 \land \cdots \land x_{k-1} \subseteq a_{k-1}) \lor \psi, \psi \vdash \chi$ and
 - (b) Γ , $(x_1 \subseteq a_1 \land \cdots \land x_{k-1} \subseteq a_{k-1}) \lor \psi, x_k \subseteq a_k \vdash \chi$.

Showing $\Gamma, \psi \vdash \chi$ suffices for (a). As for (b), similarly to the above, we have $((x_1 \subseteq a_1 \land \dots \land x_{k-1} \subseteq a_{k-1}) \lor \psi) \land x_k \subseteq a_k \vdash ((x_1 \subseteq a_1 \land \dots \land x_{k-2} \subseteq a_{k-2}) \lor \psi) \land x_k \subseteq a_k \land (x_{k-1} \subseteq a_{k-1} \lor \psi)$. To show (b), it therefore suffices, by $\lor_{\subseteq} E$, to show:

(c) Γ , $(x_1 \subseteq a_1 \land \dots \land x_{k-2} \subseteq a_{k-2}) \lor \psi, x_k \subseteq a_k, \psi \vdash \chi$ and

(d)
$$\Gamma$$
, $(\mathsf{x}_1 \subseteq \mathsf{a}_1 \land \dots \land \mathsf{x}_{k-2} \subseteq \mathsf{a}_{k-2}) \lor \psi, \mathsf{x}_k \subseteq \mathsf{a}_k, \mathsf{x}_{k-1} \subseteq \mathsf{a}_{k-1} \vdash \chi$.

Showing $\Gamma, \psi \vdash \chi$ would again suffice for (c). Continuing in the same manner, one eventually finds that it suffices to show $\Gamma, \psi \vdash \chi$ and $\Gamma, x_1 \subseteq a_1, \ldots, x_k \subseteq a_k \vdash \chi$.

- (ii) Let $I = \{1, ..., n\}$. We have that $\bigvee_{1 \le i \le n-1} (\varphi_i \land \iota_i) \vdash (\bigvee_{1 \le i \le n-1} (\varphi_i \land \iota_i) \lor \varphi_n) \land (\bigvee_{1 \le i \le n-1} (\varphi_i \land \iota_i) \lor \iota_n)$ by $\lor E$, $\lor I$, $\land E$, and $\land I$. To show $\Gamma, \bigvee_{1 \le i \le n-1} (\varphi_i \land \iota_i) \lor \varphi_n, \bigvee_{1 \le i \le n-1} (\varphi_i \land \iota_i) \lor \iota_n \vdash \chi$, by (i) it suffices to show:
 - (a) $\Gamma, \bigvee_{1 \le i \le n-1} (\varphi_i \land \iota_i) \lor \varphi_n, \bigvee_{1 \le i \le n-1} (\varphi_i \land \iota_i) \vdash \chi$ and
 - (b) $\Gamma, \bigvee_{1 \leq i \leq n-1} (\varphi_i \wedge \iota_i) \vee \varphi_n, \iota_n \vdash \chi$.

By \lor I, showing Γ , $\lor_{1 \le i \le n-1} (\varphi_i \land \iota_i) \vdash \chi$ suffices for (a). Furthermore, by a similar manipulation as used above, (c) suffices for (a), and (d) suffices for (b).

(c) $\Gamma, \bigvee_{1 \leq i \leq n-2} (\varphi_i \wedge \iota_i) \lor \varphi_{n-1}, \bigvee_{1 \leq i \leq n-2} (\varphi_i \wedge \iota_i) \lor \iota_{n-1} \vdash \chi$.

(d)
$$\Gamma, \bigvee_{1 \leq i \leq n-2} (\varphi_i \wedge \iota_i) \lor \varphi_{n-1} \lor \varphi_n, \bigvee_{1 \leq i \leq n-2} (\varphi_i \wedge \iota_i) \lor \varphi_n \lor \iota_{n-1}, \iota_n \vdash \chi.$$

By (i), to show (c) it suffices to show (e) and (f), and to show (d) it suffices to show (g) and (h):

- (e) $\Gamma, \bigvee_{1 \le i \le n-2} (\varphi_i \land \iota_i) \lor \varphi_{n-1}, \bigvee_{1 \le i \le n-2} (\varphi_i \land \iota_i) \vdash \chi.$
- (f) $\Gamma, \bigvee_{1 \leq i \leq n-2} (\varphi_i \wedge \iota_i) \vee \varphi_{n-1}, \iota_{n-1} \vdash \chi$.
- (g) $\Gamma, \bigvee_{1 \le i \le n-2} (\varphi_i \land \iota_i) \lor \varphi_{n-1} \lor \varphi_n, \bigvee_{1 \le i \le n-2} (\varphi_i \land \iota_i) \lor \varphi_n, \iota_n \vdash \chi.$
- (h) $\Gamma, \bigvee_{1 \leq i \leq n-2} (\varphi_i \wedge \iota_i) \vee \varphi_{n-1} \vee \varphi_n, \iota_{n-1}, \iota_n \vdash \chi.$

Showing that $\Gamma, \bigvee_{1 \le i \le n-2} (\varphi_i \land \iota_i) \vdash \chi$ would suffice to show (e), and showing that $\Gamma, \bigvee_{1 \le i \le n-2} (\varphi_i \land \iota_i) \lor \varphi_n, \iota_n \vdash \chi$ would suffice to show (g). Continuing in the same manner, one eventually finds that it suffices to show $\Gamma, \bigvee_{j \in J} \varphi_j, \bigwedge_{j \in J} \iota_j \vdash \chi$ for every nonempty $J \subseteq I$.

Next, we show that the Hintikka formulas of non-bisimilar pointed models/models with teams are contradictory in our proof system.

5.4.15. LEMMA.

- (i) If $M, w \neq_k M', u$, then $\chi_{M,w}^k, \chi_{M',u}^k \vdash \bot$.
- (ii) If $M, T \neq_k M', S$, then $\theta_{M,T}^k, \theta_{M',S}^k \vdash \bot$.

Proof:

- (i) Assume for contradiction that \(\chi_w\), \(\chi_u^k\) \(\not\) \(\pm\). By Proposition 5.4.3\), there is some model *M*'' and a nonempty team *T* of *M*'' such that *M*'', *T* ⊨ \(\chi_w\) and *M*'', *T* ⊨ \(\chi_u^k\) and *M*'', *T* ⊨ \(\chi_u^k\) and *M*'', *V* ⊨ \(\chi_u^k\) for all \(v \in T\), from which it follows by Theorem 5.3.3 that *M*, w \(\Rightarrow_k M'', v \(\Rightarrow_k M'', u\), a contradiction.
- (ii) W.l.o.g., we assume that there is some $w \in T$ such that $M, w \neq_k M', u$ for all $u \in S$. By item (i), $\chi_w^k, \chi_u^k \vdash \bot$ for all $u \in S$, whence by $\lor E$, $\bigvee_{u \in S} \chi_u^k, \chi_w^k \vdash \bot$. By $\neg I$, we derive $\bigvee_{u \in S} \chi_u^k \vdash \neg \chi_w^k$. Then by $\land E$, we have $\theta_S^k \vdash \bigvee_{u \in S} \chi_u^k \vdash \neg \chi_w^k$. We also derive $\theta_T^k \vdash \top \subseteq \chi_w^k$ by $\land E$. Finally, we use $\subseteq_{\neg} E$ to derive $\top \subseteq \chi_w^k, \neg \chi_w^k \vdash \bot$. \Box

Finally, we note the following simple consequence of classical completeness:

5.4.16. LEMMA. If $M, w \models \alpha$, then $\chi_{M_w}^k \vdash \alpha$, where $k \ge md(\alpha)$.

Proof:

By Proposition 5.4.3, it suffices to show $\chi_w^k \models \alpha$. Let $T \models \chi_w^k$ so that $u \models \chi_w^k$ for all $u \in T$ by flatness. By Theorem 5.3.3, it follows that $w \equiv_k u$ whence also $u \models \alpha$ for all $u \in T$. Using flatness again, we conclude $T \models \alpha$.

We are now ready to prove the main lemma.

Proof of Lemma 5.4.5:

By induction on φ .

- If $\varphi = \alpha \in ML$ and $k \ge md(\alpha)$, letting $\mathcal{D} := \{(M, w) \mid w \models \alpha\}$, by a standard normal form result for ML (see, e.g., [56]), we have that $\alpha \equiv \bigvee_{(M,w)\in\mathcal{D}} \chi_w^k$. Then by Proposition 5.4.3, $\alpha \dashv \vdash \bigvee_{(M,w)\in\mathcal{D}} \chi_w^k$. If $\mathcal{D} = \emptyset$, we have $\alpha \dashv \vdash \bigvee_{(M,w)\in\mathcal{D}} \chi_w^k = \bot = \theta_{(M^*,\emptyset)}^k$. Otherwise, we show that $\bigvee_{(M,w)\in\mathcal{D}} \chi_w^k \dashv \vdash \bigvee_{(M,w)\in\mathcal{D}} \theta_{\{w\}}^k = \bigvee_{(M,w)\in\mathcal{D}} (\chi_w^k \land \top \subseteq \chi_w^k)$. The direction \dashv is easy. For the other direction \vdash , we derive $\chi_w^k \vdash \chi_w^k \land \top \subseteq \chi_w^k$ by Lemma 5.4.8 (i); the result then follows by $\lor E$ and $\lor I$.
- · Let $\varphi = \psi_1 \lor \psi_2$ and let $k \ge md(\varphi)$. Then $k \ge md(\psi_1), md(\psi_2)$, so by the induction hypothesis there are nonempty \mathcal{C}, \mathcal{D} such that $\psi_1 \dashv \vdash \bigvee_{(M,T)\in\mathcal{C}} \theta_T^k$ and $\psi_2 \dashv \vdash \bigvee_{(M',S)\in\mathcal{D}} \theta_S^k$. Clearly $\psi_1 \lor \psi_2 \dashv \vdash \bigvee_{(M'',U)\in\mathcal{C}\cup\mathcal{D}} \theta_U^k$.

5.4. Axiomatizations

· Let $\varphi = \psi_1 \land \psi_2$ and let $k \ge md(\varphi)$. Then $k \ge md(\psi_1), md(\psi_2)$, so by the induction hypothesis there are nonempty \mathcal{C}, \mathcal{D} such that $\psi_1 \dashv \vdash \bigvee_{(M,T)\in\mathcal{C}} \theta_T^k$ and $\psi_2 \dashv \vdash \bigvee_{(M',S)\in\mathcal{D}} \theta_S^k$. Let

$$\mathcal{Y} = \{ \biguplus \mathcal{C}' \mid \mathcal{C}' \subseteq \mathcal{C} \text{ and } \oiint \mathcal{C}' \rightleftharpoons_k \biguplus \mathcal{D}', \text{ for some } \mathcal{D}' \subseteq \mathcal{D} \},\$$

that is, \mathcal{Y} contains each disjoint union of (models with) teams in \mathcal{C} which is *k*-bisimilar to some disjoint union of teams in \mathcal{D} . By union closure and bisimulation invariance, both ψ_1 and ψ_2 will hold in each team in \mathcal{Y} , and it is also easy to see that each team in which both ψ_1 and ψ_2 hold must be *k*-bisimilar to some team in \mathcal{Y} . We show $\psi_1 \wedge \psi_2 \dashv \vdash \bigvee_{(M'',X) \in \mathcal{Y}} \theta_X^k$.

 (\vdash) By Lemma 5.4.14 (ii) it suffices to show

$$\bigvee_{(M,T)\in\mathcal{C}'} \bigvee_{w\in T} \chi_{w}^{k}, \bigwedge_{(M,T)\in\mathcal{C}'} \bigwedge_{w\in T} (\top \subseteq \chi_{w}^{k}), \bigvee_{(M',S)\in\mathcal{D}'} \bigvee_{w\in S} \chi_{w}^{k}, \bigwedge_{(M',S)\in\mathcal{D}'} \bigwedge_{w\in S} (\top \subseteq \chi_{w}^{k})$$
$$\vdash \bigvee_{(M'',X)\in\mathcal{Y}} \theta_{X}^{k},$$

for all nonempty $\mathcal{C}' \subseteq \mathcal{C}$ and all nonempty $\mathcal{D}' \subseteq \mathcal{D}$. This reduces to showing $\theta_{\bigcup \mathcal{C}'}, \theta_{\bigcup \mathcal{D}'} \vdash \bigvee_{(M'',X) \in \mathcal{Y}} \theta_X^k$ for all nonempty $\mathcal{C}' \subseteq \mathcal{C}$ and all nonempty $\mathcal{D}' \subseteq \mathcal{D}$. For a given \mathcal{C}' and \mathcal{D}' , if $\bigcup \mathcal{C}' \neq_k \bigcup \mathcal{D}'$, then $\theta_{\bigcup \mathcal{C}'}, \theta_{\bigcup \mathcal{D}'} \vdash \bot \vdash \bigvee_{(M'',X) \in \mathcal{Y}} \theta_X^k$ by Lemma 5.4.15 (ii). If $\bigcup \mathcal{C}' \rightleftharpoons_k \bigcup \mathcal{D}'$, then $\bigcup \mathcal{C}' \in \mathcal{Y}$, whence $\theta_{\bigcup \mathcal{C}'} \vdash \bigvee_{(M'',X) \in \mathcal{Y}} \theta_X^k$ by $\lor \lor$.

(¬) Let $\forall C' \in \mathcal{Y}$. By Lemma 5.4.12, we have $\theta_{\forall C'}^k \vdash \bigvee_{(M,T)\in \mathcal{C}'} \theta_T^k$, and by $\lor I$, $\bigvee_{(M,T)\in \mathcal{C}'} \theta_T^k \vdash \bigvee_{(M,T)\in \mathcal{C}} \theta_T^k \vdash \psi_1$. Similarly $\theta_{\forall C'}^k \vdash \psi_2$, so by $\lor E$, $\bigvee_{\forall C'\in \mathcal{Y}} \theta_{\forall C'}^k \vdash \psi_1 \land \psi_2$.

· Let $\varphi = a \subseteq b$, where $a = \alpha_1 \dots \alpha_n$ and $b = \beta_1 \dots \beta_n$, and let $k \ge md(\varphi)$. By \subseteq Rdt and \subseteq Ext,

$$a \subseteq b \dashv \vdash \bigwedge_{x \in \{\top, \bot\}^{|a|}} (\neg \, a^x \lor x \subseteq b).$$

Given the induction cases for ML-formulas, conjunction and disjunction, it therefore suffices to show that each primitive inclusion atom is provably equivalent to a formula in the normal form. We show that $x \subseteq b \rightarrow \bigvee_{(M,T) \in \mathcal{Y}} \theta_T^k$, where

$$\mathsf{Y} = \{ (M, T) \mid \exists w \in T \text{ such that } w \models \mathsf{b}^{\mathsf{x}} \}.$$

Note that the provable equivalence we wish to show corresponds to the semantic fact in Proposition 5.2.4.

 $(\vdash) \text{ Let } \mathcal{M} \coloneqq \{(M', w) \mid w \models \top\}. \text{ Then } \vDash \bigvee_{(M', w) \in \mathcal{M}} \chi_w^k \text{ so that } \vdash \bigvee_{(M', w) \in \mathcal{M}} \chi_w^k \text{ by Proposition 5.4.3. We have } \chi_w^k \vdash \bigvee_{(M', w) \in \mathcal{M}} (\chi_w^k \land \top \subseteq \chi_w^k) \text{ by Lemma 5.4.8 (i) and}$

 \lor I, so $\vdash \bigvee_{(M',w)\in\mathcal{M}}(\chi_w^k \land \top \subseteq \chi_w^k)$ by \lor E. To show $\bigvee_{(M',w)\in\mathcal{M}}(\chi_w^k \land \top \subseteq \chi_w^k), x \subseteq b \vdash \bigvee_{(M,T)\in\mathcal{Y}} \theta_T^k$, by Lemma 5.4.14 (ii) it suffices to show that for all nonempty teams *S*,

$$\bigvee_{w \in S} \chi_w^k, \bigwedge_{w \in S} (\top \subseteq \chi_w^k), \mathsf{x} \subseteq \mathsf{b} \vdash \bigvee_{(M,T) \in \mathcal{Y}} \theta_T^k$$

If $S \in \mathcal{Y}$, the result follows by $\lor I$. If $S \notin \mathcal{Y}$, then for any $w \in S$, $w \models \neg b^{\times}$. We have that $md(\neg b^{\times}) = md(x \subseteq b) \leq md(a \subseteq b) = md(\varphi) \leq k$, whence $\chi_{w}^{k} \vdash \neg b^{\times}$ by Lemma 5.4.16. Therefore $\bigvee_{w \in S} \chi_{w}^{k} \vdash \neg b^{\times}$ by $\lor E$. By $\subseteq \neg E$, we have $\neg b^{\times}, x \subseteq b \vdash \bigvee_{(M,T) \in \mathcal{Y}} \theta_{T}^{k}$.

(¬) Let $(M,T) \in \mathcal{Y}$. Then there is some $w \in T$ such that $w \models b^{\times}$ so that since $md(b^{\times}) = md(x \subseteq b) \leq md(a \subseteq b) = md(\varphi) \leq k$, we have $\chi_w^k \vdash b^{\times}$ by Lemma 5.4.16. We have $\theta_T^k \vdash \top \subseteq \chi_w^k$ by $\wedge E$, and $\top \subseteq \chi_w^k \vdash \top \subseteq b^{\times}$ by $\chi_w^k \vdash b^{\times}$ and Lemma 5.4.8 (ii). Then:

$$\top \subseteq \mathbf{b}^{\mathsf{x}} \vdash \mathsf{T}^{|\mathbf{b}|} \subseteq \mathbf{b}^{\mathsf{x}} \dots \mathbf{b}^{\mathsf{x}}$$
 (Prop. 5.4.4 (iii))

$$\vdash \mathsf{T}^{|\mathbf{b}|} \subseteq \beta_{1}^{x_{1}} \dots \beta_{n}^{x_{n}}$$
 (Lemma 5.4.8 (ii))

$$\vdash x_{1} \dots x_{n} \subseteq \beta_{1} \dots \beta_{n}.$$
 (Lemma 5.4.8 (iii))

where $x \subseteq b = x_1 \dots x_n \subseteq \beta_1 \dots \beta_n$. Therefore $\bigvee_{(M,T) \in \mathcal{Y}} \theta_T^k \vdash x \subseteq b$ by $\lor E$.

· Let $\varphi = \otimes \psi$ and let $k \ge md(\varphi)$. Then $n := k - 1 \ge md(\psi)$ and by the induction hypothesis there is a nonempty \mathcal{D} such that $\psi \dashv \vdash \bigvee_{(M,T)\in\mathcal{D}} \theta_T^n$. By \otimes Mon, $\otimes \psi \dashv \vdash \otimes \bigvee_{(M,T)\in\mathcal{D}} \theta_T^n$. We show that

$$\bigotimes \bigvee_{(M,T)\in\mathcal{D}} \theta^n_T \dashv \vdash \bigvee_{(M,T)\in\mathcal{D}} \bigotimes \theta^n_T \dashv \vdash \bigvee_{(M,T)\in\mathcal{D}} (\bigotimes \bigvee_{w\in T} \chi^n_w \land \bigwedge_{w\in T} (\top \subseteq \bigotimes \chi^n_w)).$$
(5.1)

The modal depth of the formula on the very right is $\leq n + 1 = k$, so the result then follows by the induction cases for ML-formulas, inclusion atoms, conjunction, and disjunction.

The first equivalence in (5.1) follows from the more general equivalence $\otimes \varphi \lor \otimes \psi \dashv \vdash \otimes (\varphi \lor \psi)$, whose direction \dashv can be derived by $\otimes \lor$ Distr, and the converse direction \vdash by applying \lor E to $\otimes \varphi \vdash \otimes (\varphi \lor \psi)$ and $\otimes \psi \vdash \otimes (\varphi \lor \psi)$ (which are given by \otimes Mon).

For the second equivalence in (5.1), by \lor I and \lor E, it suffices to derive $\otimes \theta_T^n \dashv \vdash \otimes \bigvee_{w \in T} \chi_w^n \land \bigwedge_{w \in T} (\top \subseteq \otimes \chi_w^n)$ for each $T \in \mathcal{D}$, i.e.,

$$\bigotimes \left(\bigvee_{w \in T} \chi_{w}^{n} \wedge \bigwedge_{w \in T} \left(\top \subseteq \chi_{w}^{n} \right) \right) \to \bowtie \bigotimes_{w \in T} \chi_{w}^{n} \wedge \bigwedge_{w \in T} \left(\top \subseteq \bigotimes \chi_{w}^{n} \right).$$
 (5.2)

Intuitively, if a team S satisfies the formula on the right in (5.2), this means that S has a successor team that is a subset (modulo bisimulation) of T (captured by the left conjunct), and that all elements in T can be seen from S (captured by the

5.4. Axiomatizations

right conjunct). Combining these facts, one gets that *S* has the team *T* (modulo bisimulation) as a successor team—which is what the formula $\otimes \theta_T^n$ on the left of the equivalence expresses.

Now, the direction \vdash of (5.2) can be derived by:

$$\bigotimes (\bigvee_{w \in T} \chi_w^n \land \bigwedge_{w \in T} (\top \subseteq \chi_w^n)) \vdash \bigotimes \bigvee_{w \in T} \chi_w^n \land \bigwedge_{w \in T} \bigotimes (\top \subseteq \chi_w^n)$$
 (\$Mon)

$$\vdash \bigotimes \bigvee_{w \in T} \chi_w^n \wedge \bigwedge_{w \in T} (\top \subseteq \bigotimes \chi_w^n), \qquad (\bigotimes_{\subseteq} \text{Distr})$$

while the other direction \dashv is derived by:

$$\bigotimes_{w \in T} \chi_w^n \wedge \bigwedge_{w \in T} (\top \subseteq \bigotimes \chi_w^n) \vdash \bigotimes ((\bigvee_{w \in T} \chi_w^n \lor \bigvee_{w \in T} \chi_w^n) \wedge \bigwedge_{w \in T} (\top \subseteq \chi_w^n)) \quad (\subseteq_{\bigotimes} \text{Distr})$$

$$\vdash \bigotimes (\bigvee \chi_w^n \wedge \bigwedge (\top \subseteq \chi_w^n)). \quad (\bigotimes \text{Mon})$$

$$\otimes (\bigvee_{w \in T} \chi_w^n \wedge \bigwedge_{w \in T} (\top \subseteq \chi_w^n)).$$
 (\otimes Mon)

· Let $\varphi = \Box \psi$ and let $k \ge md(\varphi)$. Then $n = k - 1 \ge md(\psi)$, so by the induction hypothesis there is a nonempty \mathcal{D} such that $\psi \dashv \vdash \bigvee_{(M,T)\in\mathcal{D}} \theta_T^n$. By \Box Mon, we have that $\Box \psi \dashv \vdash \Box \bigvee_{(M,T)\in\mathcal{D}} \theta_T^n$. We show that

$$\square \bigvee_{(M,T)\in\mathcal{D}} \theta^n_T \dashv \vdash \bigvee_{\mathcal{C}\subseteq\mathcal{D}} (\square \bigvee_{w\in t \not = \mathcal{C}} \chi^n_w \land \bigwedge_{w\in t \not = \mathcal{C}} (\top \subseteq \otimes \chi^n_w)).$$

The modal depth of the formula on the right is $\leq n + 1 = k$, so the result then follows by the induction cases for ML-formulas, inclusion atoms, conjunction, and disjunction. Intuitively, if a team *S* satisfies the formula on the left of the above equivalence, this means that R[S] is bisimilar to the (disjoint) union of some teams in \mathcal{D} , i.e., $R[S] \rightleftharpoons_n \biguplus \mathcal{C}$ for some $\mathcal{C} \subseteq \mathcal{D}$ (see Lemma 5.4.10). In other words, by the forth condition, R[S] is bisimilar to a subteam of $\biguplus \mathcal{C}$ (which is captured by the left conjunct of the disjunct corresponding to \mathcal{C} of the formula on the right of the equivalence), and, by the back condition, each world in $\biguplus \mathcal{C}$ can be seen from *S* (which is captured by the right conjunct of the relevant disjunct).

 (\vdash) By Lemma 5.4.14 (iv) it suffices to show

$$\square \bigvee_{(M,T)\in\mathcal{E}} \bigvee_{w\in T} \chi_w^n, \bigwedge_{(M,T)\in\mathcal{E}} \bigwedge_{w\in T} (\top \subseteq \otimes \chi_w^n) \vdash \bigvee_{\mathcal{C}\subseteq\mathcal{D}} (\square \bigvee_{w\in \boxdot \mathcal{C}} \chi_w^n \land \bigwedge_{w\in \biguplus \mathcal{C}} (\top \subseteq \otimes \chi_w^n)),$$

for all nonempty $\mathcal{E} \subseteq \mathcal{D}$. This reduces to showing

$$\Box \bigvee_{w \in \mathfrak{t} \in \mathcal{E}} \chi_{w}^{n}, \bigwedge_{w \in \mathfrak{t} \in \mathcal{E}} (\top \subseteq \mathfrak{d} \chi_{w}^{n}) \vdash \bigvee_{\mathcal{C} \subseteq \mathcal{D}} (\Box \bigvee_{w \in \mathfrak{t} \in \mathcal{C}} \chi_{w}^{n} \land \bigwedge_{w \in \mathfrak{t} \in \mathcal{C}} (\top \subseteq \mathfrak{d} \chi_{w}^{n})),$$

for all nonempty $\mathcal{E} \subseteq \mathcal{D}$, which is given by $\vee I$.

 (\neg) Let $\mathcal{C} \subseteq \mathcal{D}$. We have:

$$\square \bigvee_{w \in \mathfrak{t} \mid \mathcal{C}} \chi_w^n \land \bigwedge_{w \in \mathfrak{t} \mid \mathcal{C}} (\top \subseteq \bigotimes \chi_w^n)$$

The result then follows by \lor E.

5.4.2 ML(\heartsuit) and ML(\heartsuit)

By adapting relevant rules from the system for modal inclusion logic $ML(\subseteq)$, we obtain sound and complete systems for the two expressively equivalent might-operator logics $ML(\nabla)$ and $ML(\nabla)$.

5.4.17. DEFINITION. The natural deduction system for $ML(\nabla)/ML(\nabla)$ consists of the classical rules in Table 5.1 and the ∇/∇ -rules in Table 5.4.

Most rules in Table 5.4 correspond to rules or derivable results for the ML(\subseteq)system applied to inclusion atoms of the form $\top \subseteq \alpha$ (recall the equivalence $\top \subseteq \alpha \equiv \bullet \alpha$, where $\bullet \in \{\nabla, \nabla\}$). Given that there is no syntactic restriction on what may appear in the scope of \bullet (unlike with inclusion atoms of the corresponding form), these rules may now be generalized to also apply to formulas $\bullet \varphi$ where φ is non-classical; this has been done whenever the generalization in question is sound. The rules $\bullet_{\neg} E$, $\lor_{\bullet} E$, $\bullet Distr$, $\Box \otimes_{\bullet} Exc$, $\boxtimes \lor_{\bullet} E$, $\bigotimes_{\bullet} Distr$, and $\bullet_{\otimes} Distr$ correspond to the rules $\subseteq_{\neg} E$, $\lor_{\subseteq} E$, $\subseteq Distr$, $\Box \otimes_{\subseteq} E$, $\bigotimes_{\Box} E$, $\bigotimes_{\Box} Distr$, and $\subseteq_{\otimes} Distr$, respectively. The rule $\bullet I$ corresponds to Lemma 5.4.8 (i), and $\bullet Mon$ (restricted to classical formulas) corresponds to Lemma 5.4.8 (ii).

For both systems, we add a rule $\bullet \lor$ Distr asserting the distributivity of \bullet over \lor . The rules \triangledown Join and $\triangledown \land$ Simpl reflect the entailments pointed out in Section 5.2 and serve to differentiate the systems. The \triangledown -version of the rule \triangledown E is also sound, and it is derivable using $\triangledown \land$ Simpl and \triangledown Mon.

5.4.18. THEOREM (Soundness). *If* $\Gamma \vdash \varphi$ *, then* $\Gamma \models \varphi$ *.*

Proof:

Most cases are analogous to those for $ML(\subseteq)$. We only prove some of the more interesting cases. By the empty team property, it suffices to check soundness for an arbitrary nonempty team T.



Table 5.4: Rules for ML(\bullet) with $\bullet \in \{\nabla, \nabla\}$.

- (•Mon) Let $T \vDash \bullet \varphi$ and $T \vDash \gamma$ for all $\gamma \in \Gamma$, and assume that $\Gamma, \varphi \vDash \psi$, where Γ consists of ML-formulas. Then there is a nonempty (singleton) subteam $T' \subseteq T$ such that $T' \vDash \varphi$. By downward closure of the formulas in Γ , we have $T' \vDash \gamma$ for all $\gamma \in \Gamma$. It follows that $T' \vDash \psi$, and hence $T \vDash \bullet \psi$.
- (• \lor Distr) Let $T \vDash \bullet (\varphi \lor \psi)$. Then there is a nonempty (singleton) subteam $T' \subseteq T$ such that $T' \vDash \varphi \lor \psi$. Then there are $T_1, T_2 \subseteq T'$ such that $T_1 \cup T_2 = T', T_1 \vDash \varphi$ and $T_2 \vDash \psi$. W.l.o.g. suppose that T_1 is nonempty. Then $T \vDash \bullet \varphi$ so that $T \vDash \bullet \varphi \lor \bullet \psi$.

- (∇E) Let $T \models \nabla \nabla \varphi$. Then there are nonempty subteams $T_1, T_2 \subseteq T$ such that $T_1 \subseteq T_2 \subseteq T$ and $T_1 \models \varphi$. Thus $T \models \nabla \varphi$.
- (∇ Join) Let $T \models \nabla \varphi \land \nabla \psi$. Then there are nonempty subteams $T_1, T_2 \subseteq T$ such that $T_1 \models \varphi$ and $T_2 \models \psi$. Then $T' = T_1 \cup T_2$ is a nonempty subteam of T such that $T' \models \varphi \lor \psi, T' \models \nabla \varphi$ and $T' \models \nabla \psi$. Thus $T \models \nabla ((\varphi \lor \psi) \land \nabla \varphi \land \nabla \psi)$.
- $(\forall \land \text{Simpl})$ Let $T \models \forall (\forall \varphi \land \psi)$. Then there is a $w \in T$ such that $\{w\} \models \psi$ and $\{w\} \models \forall \varphi$, from which it follows that $\{w\} \models \varphi$. Hence $T \models \forall (\varphi \land \psi)$.

5.4.19. PROPOSITION.

(*i*) $\bullet \phi \lor \bullet \psi \dashv \vdash \bullet (\phi \lor \psi).$

(*ii*) $\bullet(\varphi \land \psi) \vdash \bullet \varphi \land \bullet \psi$.

Proof:

The direction \vdash in item (i) follows by $\lor E$, •Mon and $\lor I$, the other direction is by • \lor Distr. Item (ii) follows by •Mon, \land E and \land I.

We show completeness using the same strategy as used for $ML(\subseteq)$. Many of the details are analogous to parts of the $ML(\subseteq)$ -proof; we omit most of these details and focus on the steps that specifically concern the might-operators.

5.4.20. LEMMA (Provable equivalence of the normal form). For any φ in ML(•) and $k \ge md(\varphi)$, there is a (finite, nonempty) property C such that

$$\varphi \dashv \vdash \bigvee_{(M,T)\in\mathcal{C}} (\bigvee_{w\in T} \chi_w^k \wedge \bigwedge_{w\in T} \bullet \chi_w^k).$$

Proof:

By induction on φ . All inductive cases except the one for • are similar to the corresponding proofs for ML(\subseteq) (see Lemma 5.4.5). Inspecting the proof of Lemma 5.4.5, one sees that with the exception of the case for inclusion atoms, the lemma is derivable using rules which have analogues in the ML(•)-systems, together with the proof-theoretic results in Proposition 5.4.3 and Lemmas 5.4.6, 5.4.7, 5.4.8 (i), 5.4.8 (ii), 5.4.12, 5.4.14, 5.4.15, and 5.4.16. These results, in turn, are also (with the exception of Lemmas 5.4.8 (i), 5.4.8 (ii)) derivable using rules which have analogues in the ML(•)-system. Any steps making use of Lemma 5.4.8 (i) can be replaced by applications of •I; similarly with Lemma 5.4.8 (ii) and •Mon. We add the cases for •.

Let $\varphi = \bullet \psi$. We derive

•
$$\Psi \dashv \vdash \bullet \bigvee_{(M,T)\in\mathcal{C}} (\bigvee_{w\in T} \chi_{w}^{k} \land \bigwedge_{w\in T} \bullet \chi_{w}^{k})$$
 (Induction hypothesis, • Mon)
$$\dashv \vdash \bigvee_{(M,T)\in\mathcal{C}} \bullet (\bigvee_{w\in T} \chi_{w}^{k} \land \bigwedge_{w\in T} \bullet \chi_{w}^{k})$$
 (Prop. 5.4.19 (i))

We will now show:

176

(a)
$$\bigvee_{(M,T)\in\mathcal{C}} \nabla (\bigvee_{w\in T} \chi_w^k \wedge \bigwedge_{w\in T} \nabla \chi_w^k) \to \bigvee_{(M,T)\in\mathcal{C}} \bigwedge_{w\in T} \nabla \chi_w^k$$
 and

(b)
$$\bigvee_{(M,T)\in\mathcal{C}} \nabla (\bigvee_{w\in T} \chi_w^k \wedge \bigwedge_{w\in T} \nabla \chi_w^k) \to \bigvee_{(M,T)\in\mathcal{C}} \nabla \bigwedge_{w\in T} \chi_w^k$$
 and

(c) every formula in the form $\bullet \alpha$ is provably equivalent to a formula in normal form.

The result will then follow by the induction cases for conjunction and disjunction.

We first show (a). By \vee E and \vee I, it suffices to show that $\nabla(\bigvee_{w\in T} \chi_w^k \wedge \bigwedge_{w\in T} \nabla \chi_w^k) \dashv (M,T) \in \mathcal{C}$. For the direction \vdash , we derive $\nabla(\bigvee_{w\in T} \chi_w^k \wedge \bigwedge_{w\in T} \chi_w^k) \vdash \nabla \bigwedge_{w\in T} \nabla \chi_w^k \vdash \bigwedge_{w\in T} \nabla \chi_w^k \vdash \bigwedge_{w\in T} \nabla \chi_w^k$ by ∇ Mon, \wedge E, Proposition 5.4.19 (ii), and ∇ E. The direction \dashv follows by ∇ Join.

We now show (b). As above, it suffices to show that $\nabla(\bigvee_{w\in T} \chi_w^k \wedge \bigwedge_{w\in T} \nabla \chi_w^k) \dashv \nabla \bigwedge_{w\in T} \chi_w^k$ for an arbitrary $(M,T) \in C$. For the direction \vdash , we derive $\nabla(\bigvee_{w\in T} \chi_w^k \wedge \bigwedge_{w\in T} \chi_w^k) \vdash \nabla \bigwedge_{w\in T} \nabla \chi_w^k \vdash \nabla \bigwedge_{w\in T} \chi_w^k$ by ∇ Mon, $\wedge E$ and $\nabla \wedge$ Simpl. For the direction \dashv , we note that $\bigwedge_{w\in T} \chi_w^k \vdash \bigvee_{w\in T} \chi_w^k \wedge \bigwedge_{w\in T} \nabla \chi_w^k$ by $\wedge E$, $\vee I$, and ∇I . Therefore $\nabla \bigwedge_{w\in T} \chi_w^k \vdash \nabla(\bigvee_{w\in T} \chi_w^k \wedge \bigwedge_{w\in T} \nabla \chi_w^k)$ by ∇ Mon.

The proof of (c) is analogous to the proof of the primitive inclusion atom case in Lemma 5.4.5 (with an application of •Mon replacing the appeal to Lemma 5.4.8 (ii)). \Box

5.4.21. THEOREM (Completeness). If $\Gamma \vDash \psi$, then $\Gamma \vdash \psi$.

Proof:

Similar to the proof of Theorem 5.4.13. Inspecting this proof, one sees that it can be conducted using only rules which have analogues in the $ML(\bullet)$ -systems, together with the proof-theoretic results in Proposition 5.4.3 and Lemmas 5.4.6, 5.4.7, 5.4.8 (i), 5.4.8 (ii), 5.4.12 and 5.4.5. We have proved an analogue to Lemma 5.4.5 in Lemma 5.4.20, and as noted above, analogues to the other required results can be obtained in $ML(\bullet)$.

5.5 Concluding Remarks and Directions for Further Research

In this article, we have addressed a recognized gap in the literature on team-based modal logics by presenting an axiomatization for modal inclusion logic $ML(\subseteq)$. This logic, together with modal dependence logic and modal independence logic, are commonly considered to be the core team-based modal logics. While modal dependence logic has already been axiomatized in previous work [134], modal independence logic (see [90]) as well as propositional independence logic are still missing an axiomatization.

We also studied two other union-closed extensions of modal logic—the two mightoperator logics $ML(\nabla)$ and $ML(\nabla)$. The logics $ML(\subseteq)$ and $ML(\nabla)$ were shown to be expressively complete for the same class of properties and hence expressively equivalent in [73]; we reviewed and refined this result and showed that the new variant $ML(\nabla)$ with the singular might-operator ∇ is likewise expressively complete for this class of properties. We also provided axiomatizations for $ML(\nabla)$ and $ML(\nabla)$. Note that one can obtain expressive completeness results, axiomatizations, and completeness proofs for the propositional variants of the might-operator logics via a straightforward adaptation of the results in this article.

All our axiomatizations are presented in natural deduction style—as opposed to the Hilbert style commonly used for modal logics—mainly because we do not have an *implication* connective in the languages of the logics we consider. To turn our natural deduction rules into Hilbert-style axioms, one could consider extending the languages with an implication. The implication would have to preserve union closure and the empty team property, and presumably also satisfy other desiderata for an implication such as the deduction theorem. Whether such a team-based implication exists is currently unknown. To better study the proof-theoretic properties of these logics, it would also be desirable to introduce sequent calculi for them. To this end, finding more elegant versions of the proof systems presented in this article might be useful. A point of difficulty could be that the logics do not admit uniform substitution, although sequent calculi have been developed for other team-based logics—see, e.g., [50, 31, 100].

We observed in Section 5.3 that with the strict semantics for the diamond, the three logics are not invariant under the notion of team bisimulation established in the literature. This raises the question whether one can formulate a relation between models with teams—a *strict* team bisimulation—that would be strong enough to ensure invariance with respect to strict semantics but that would still respect the local character of modal logic and not imply first-order invariance.

As we now have a better understanding of the logical properties of these three union closed team-based modal logics, it is natural to ask about their possible applications in other fields. In connection with this, we note that certain closely related logics have recently found application in formal semantics. The two-sorted first-order team semantics framework in [9] employs first-order inclusion atoms together with existential quantification to represent epistemic modalities in a manner similar to how the might-operators function. In [6], the usual modal logic is extended with a *nonemptiness atom* NE satisfied by all but the empty team; this logic (which is union closed but does not have the empty team property, and is in some ways very similar to the might-operator logics—see footnote 3) is then used to account for *free choice inferences* and related natural-language phenomena.

5.6 Appendix: Translation of ML(⊆) into First-order Inclusion Logic.

In this appendix, we provide a translation of $ML(\subseteq)$ into first-order inclusion logic $(FO(\subseteq))$. Before we do so, however, let us issue a note of caution. One way to utilize the standard translation of ML into classical first-order logic is to derive the compactness of ML from that of first-order logic. The article [134], similarly, provides a translation from modal dependence logic into first-order dependence logic, and uses this translation and the fact that first-order dependence logic is compact [106] to conclude that modal dependence logic is likewise compact. The presentation in [134] is erroneous because it conflates compactness formulated in terms of satisfiability (if each finite subset of Γ is satisfiable, then Γ is satisfiable) with compactness formulated in terms of entailment (if $\Gamma \vDash \varphi$, then there is a finite $\Gamma_0 \subseteq \Gamma$ such that $\Gamma_0 \vDash \varphi$). These notions need not coincide for a logic that is not closed under classical negation, and, indeed, while first-order dependence logic is satisfiability-compact, it is not entailmentcompact. The same applies to $FO(\subseteq)$, which is shown to be satisfiability-compact in [106], but which is not entailment-compact.⁸ One ought not, therefore, use the translation in this section to argue that $ML(\subseteq)$ is entailment-compact. As noted in Section 5.3, however, we can derive both the satisfiability- and the entailment-compactness of $ML(\subseteq)$ (as well as those of its variants, and those of modal dependence logic) from the fact, proved in [89], that team properties invariant under bounded bisimulation can be expressed in classical first-order logic.

We briefly recall the syntax and semantics of $FO(\subseteq)$; for detailed discussion of the logic, see., e.g., [51]. Fix an infinite set Var of first-order variables, and a first-order vocabulary τ . The set of first-order τ -terms is defined as usual. The syntax for $FO(\subseteq)$ over τ is given by:

 $\varphi ::= t_1 = t_2 | R(t_1, \dots, t_n) | \vec{x} \subseteq \vec{y} | \neg t_1 = t_2 | \neg R(t_1, \dots, t_n) | (\varphi \lor \varphi) | (\varphi \land \varphi) | \exists x \varphi | \forall x \varphi,$

where each t_i is a τ -term, R is an n-ary relation symbol in τ , and \vec{x} and \vec{y} are two finite sequences of variables of the same length. Define, as usual, $\perp := \forall x(\neg x = x)$ and $\top := \forall x(x = x)$.

Let \mathcal{M} be a τ -model with domain W. An assignment over W with domain $V \subseteq Var$ is a function $s : V \to W$. We abbreviate $s(\vec{x}) := \langle s(x_1), \dots, s(x_n) \rangle$, where $\vec{x} = \langle x_1, \dots, x_n \rangle$. Given $a \in W$, the modified assignment s(a/x) is defined as s(a/x)(y) := a for y = x, and s(a/x)(y) := s(y) otherwise. Given $V \subseteq Var$, a (first-order) team X of \mathcal{M} with domain V is a set of assignments $s : V \to W$. Given $A \subseteq W$ and a team X of \mathcal{M} , we let $X(A/x) := \{s(a/x) \mid s \in X, a \in A\}$.

⁸To see why, consider the $FO(\subseteq)$ -sentence $\varphi_{nwf} \coloneqq \exists x \exists y (x \subseteq y \land x < y)$ where < is a binary relation symbol. Now φ_{nwf} expresses that there is an infinite descending <-chain, i.e., < is not well-founded (see, e.g., [136] for details). Let φ_{lo} be a first-order sentence expressing that < is a strict linear order with a (unique) greatest element. For each $n \in \mathbb{N}$, define the first-order sentence $\varphi_n \coloneqq \exists! z \forall x ((x = z \lor x < z) \land \exists y_1 \ldots \exists y_n (y_n < \cdots < y_1 < z))$, expressing that there is a <-chain of length n below the greatest element. Thus $\{\varphi_{lo}, \varphi_n \mid n \in \mathbb{N}\} \vDash \varphi_{nwf}$, but the entailment does not hold for any finite subset of $\{\varphi_{lo}, \varphi_n \mid n \in \mathbb{N}\}$.

The team semantics of $FO(\subseteq)$ is given by (here we follow the convention for first-order team-based logics and write the team on the right-hand side of the satisfaction symbol):

$$\mathcal{M} \vDash_X \alpha \iff \text{ for all } s \in X, \mathcal{M} \vDash_s \alpha, \text{ if } \alpha \text{ is a first-order literal;}$$
$$\mathcal{M} \vDash_X \vec{x} \subseteq \vec{y} \iff \text{ for all } s \in X \text{ there exists } s' \in X \text{ such that } s(\vec{x}) = s'(\vec{y});$$
$$\mathcal{M} \vDash_X \varphi \lor \psi \iff \text{ there exist } Y, Z \subseteq X \text{ such that } X = Y \cup Z, \mathcal{M} \vDash_Y \varphi \text{ and } \mathcal{M} \vDash_Z \psi;$$
$$\mathcal{M} \vDash_X \varphi \land \psi \iff \mathcal{M} \vDash_X \varphi \text{ and } \mathcal{M} \vDash_X \psi;$$
$$\mathcal{M} \vDash_X \exists x \varphi \iff \text{ there exists a function } F : X \to \mathcal{P}(W) \smallsetminus \{\varnothing\} \text{ such that }$$
$$\mathcal{M} \vDash_{X(F/x)} \varphi, \text{ where } X(F/x) \coloneqq_S \{s(a/x) \mid s \in X, a \in F(s)\};$$
$$\mathcal{M} \vDash_X \forall x \varphi \iff \mathcal{M} \vDash_{X(W/x)} \varphi.$$

It is easy to check that all formulas of $FO(\subseteq)$ are union closed and have the empty team property, and that formulas without inclusion atoms (formulas of FO) are additionally downward closed and flat (the first-order versions of these notions are defined analogously to the modal versions).

5.6.1. DEFINITION. For any formula φ in ML(\subseteq), its standard translation $ST_x(\varphi)$ into first-order inclusion logic $FO(\subseteq)$ (with respect to the first-order variable *x*) is defined inductively as follows:

$$ST_{x}(p) \coloneqq Px;$$

$$ST_{x}(\perp) \coloneqq \perp;$$

$$ST_{x}(\neg \alpha) \coloneqq \neg ST_{x}(\alpha);$$

$$ST_{x}(\varphi \land \psi) \coloneqq ST_{x}(\varphi) \land ST_{x}(\psi);$$

$$ST_{x}(\varphi \lor \psi) \coloneqq ST_{x}(\varphi) \lor ST_{x}(\psi);$$

$$ST_{x}(\otimes \varphi) \coloneqq \exists y(xRy \land ST_{y}(\varphi));$$

$$ST_{x}(\boxtimes \varphi) \coloneqq \exists y(\neg xRy \lor (xRy \land ST_{y}(\varphi)));$$

$$ST_{x}(\equiv \varphi) \coloneqq \forall y(\neg xRy \lor (xRy \land ST_{y}(\varphi)));$$

$$ST_{x}(a \subseteq b) \coloneqq \bigwedge_{z \in \{\top, \bot\}^{|a|}} (ST_{x}(\neg a^{z}) \lor \exists y(y \subseteq x \land ST_{y}(b^{z})).$$

For $X \subseteq Prop$, let σ_X be the first-order signature containing a binary relation symbol R_0 and a unary relation symbol P for each $p \in X$. There is a one-to-one correspondence between Kripke models over X and first-order σ_X -structures: M = (W, R, V) over X corresponds to the σ_X -structure $\mathcal{M} = (W, R_0^{\mathcal{M}}, \{P^{\mathcal{M}}\}_{p \in X})$, where $P^{\mathcal{M}} = V(p)$ is the interpretation of each unary P, and where $R_0^{\mathcal{M}} = R$.

We conclude with a lemma establishing that the translation functions as desired.

5.6.2. LEMMA. Let φ be a formula in modal inclusion logic ML(\subseteq). For any Kripke model M and team T of M, and any variable x,

$$M, T \vDash \varphi \iff \mathcal{M} \vDash_{T_x} ST_x(\varphi),$$

where $T_x = \{\{(x,w)\} | w \in T\}$ is a first-order team with domain $\{x\}$.

Proof:

The proof is by induction on φ ; we refer to [97] and add the case for a \subseteq b.

We want to show that $M, T \models a \subseteq b \iff \mathcal{M}, T_x \models \bigwedge_{z \in \{\top, \bot\}^{|a|}} (ST_x(\neg a^z) \lor \exists y(y \subseteq x \land ST_y(b^z))$. By Lemma 5.2.7 (i), we have $a \subseteq b \equiv \bigwedge_{z \in \{\top, \bot\}^{|a|}} (\neg a^z \lor z \subseteq b)$. It therefore suffices to show that for any $z \in \{\top, \bot\}^{|a|}, M, T \models z \subseteq b \iff \mathcal{M} \models_{T_x} \exists y(y \subseteq x \land ST_y(b^z))$, for then the result follows by the other induction cases. Let $z \in \{\top, \bot\}^{|a|}$.

The case in which $T = \emptyset$ is trivial. Suppose that $T \neq \emptyset$ whence also $T_x \neq \emptyset$. For the left-to-right direction, by Proposition 5.2.4 there is a $w \in T$ such that $\{w\} \models b^z$. By the induction hypothesis, $\mathcal{M} \models_{\{w\}_x} ST_x(b^z)$. It is easy to verify that this is equivalent to $\mathcal{M} \models_{T_x(\{w\}/y)} ST_y(b^z)$. Define a function $F : T_x \to \mathcal{P}(W) \setminus \{\emptyset\}$ by $F(s) = \{w\}$ for all $s \in T_x$. Clearly $\mathcal{M} \models_{T_x(F/y)} ST_y(b^z)$. Since $w \in T$, $\mathcal{M} \models_{T_x(F/y)} y \subseteq x$, whence $\mathcal{M} \models_{T_x}$ $\exists y(y \subseteq x \land ST_y(b^z))$.

For the other direction, we have that there is a function $F: T_x \to \mathcal{P}(W) \setminus \{\emptyset\}$, such that $\mathcal{M} \vDash_{T_x(F/y)} (y \subseteq x \land ST_y(b^z))$. We have that $T_x \neq \emptyset$ whence also $T_x(F/y) \neq \emptyset$ so let $s \in T_x(F/y)$. By $\mathcal{M} \vDash_{T_x(F/y)} y \subseteq x$, there is a $s' \in T_x(F/y)$ with s(y) = s'(x). We must have $s'(x) = w \in T$; then s = s''(w/y) for some $s'' \in T_x$. By downward closure, $\mathcal{M} \vDash_{\{s''(w/y)\}} ST_y(b^z)$. It follows that $\mathcal{M} \vDash_{\{w\}_y} ST_y(b^z)$. By the induction hypothesis, $\mathcal{M}, \{w\} \vDash_{\{v\}_y}$ so that by Proposition 5.2.4, $\mathcal{M}, T \vDash_{z} \subseteq b$.

Chapter 6

A Deep-Inference Sequent Calculus for a Propositional Team Logic

This chapter is based on:

Aleksi Anttila, Rosalie Iemhoff, and Fan Yang. Deep-inference Sequent Calculi for Propositional Team Logics. Manuscript. 2024

Abstract We introduce a sequent calculus for the propositional team logic with both the split disjunction and the inquisitive disjunction consisting of a G3-style system for classical propositional logic together with deep-inference rules for the inquisitive disjunction. We show that the system satisfies various desirable properties: it admits height-preserving weakening, contraction and inversion; it supports a procedure for constructing cutfree proofs and countermodels similar to that for G3cp; and cut elimination holds as a corollary of cut elimination for the G3-style subsystem together with a normal form theorem for cutfree derivations in the system.

6.1 Introduction

Logics such as *dependence logic* [119] and *inquisitive logic* [41] are typically interpreted using *team semantics* [78, 79]: formulas are interpreted with respect to sets of evaluation points (valuations/assignments/possible worlds) called *teams*, rather than single evaluation points as in the usual Tarskian semantics. Team semantics was originally introduced by Hodges [78, 79] to provide a compositional semantics for Hintikka and Sandu's *independence-friendly logic* [77, 75]; independently, Ciardelli, Groenendijk, and Roelofsen developed *inquisitive logic* [60, 41, 38, 35] which also essentially employs team semantics (see [133, 34]). The use of teams allows for simple and natural ways of formalizing notions such as question meaning (the question as to whether *p* is the case can be formalized in propositional inquisitive logic as $p \lor \neg p$,

where \forall is the *inquisitive* or *global disjunction*) and dependence ("the value of q functionally depends on the value of p" can be formalized in propositional dependence logic [139] using a *dependence atom* as =(p,q)). We will refer to logics primarily intended to be interpreted using team semantics as *team logics*.

In this paper, we focus on propositional team logics. While there are a great number of natural deduction- and Hilbert-style axiomatizations of propositional team logics in the literature [41, 112, 107, 139, 33, 140, 96, 39, 137], the development of sequent calculus systems and of proof theory in general for these logics has been slower. The sequent calculi that have been constructed have all been for variants of propositional inquisitive logic; these include multiple labelled systems [111, 31, 100, 18] as well as the multi-type display calculus in [50]. There is additionally one natural deduction system with normalization [100].¹

As observed in [50], one of the main difficulties in providing standard sequent calculi for team logics is that typically these logics are not *closed uniform substitution*. Due to this failure, axiomatizations for team logics typically feature rules that may only be applied to some subclass of formulas, and these axiomatizations do not admit the usual uniform substitution rule. Many proof-theoretic techniques depend on the universal applicability of the rules, so it is not immediately obvious how to apply these techniques to most team logics (we discuss the difficulties with cut elimination in a setting with restricted rules in some more detail in Section 6.7)—often some specialized machinery has to be introduced to handle this issue. For instance, the construction of the multi-type display calculus in [50] involves the introduction of a new language featuring two types of formulas for the team logic axiomatized, with closure under substitution holding within each of these types. It is also not even clear how a sequent should be interpreted in the setting of team semantics—there are, for instance, multiple disjunctions available to interpret the commas in the succedent Δ of a sequent $\Gamma \Rightarrow \Delta$.

In this paper, we introduce a sequent calculus with cut elimination (i.e., cut is admissible in the cutfree fragment) for the propositional team logic PL(W) (studied in, for instance, [139, 112, 33, 138]). This logic features both the *split disjunction* \vee (also known as the *tensor* or *local disjunction*)—this is the canonical disjunction employed in dependence logic and other logics in the lineage of dependence logic—and the *inquisitive disjunction* \forall already mentioned above, which is used in inquisitive logic to model the meanings of questions. The semantics for these connectives are as follows (here *t* is a propositional team—a set of valuations):

$$t \vDash \varphi \lor \psi \qquad \text{iff} \qquad \text{there are } s, u \text{ such that } t = s \cup u, s \vDash \varphi, \text{ and } u \vDash \psi$$
$$t \vDash \varphi \lor \psi \qquad \text{iff} \qquad t \vDash \varphi \text{ or } t \vDash \psi$$

That is, a split disjunction $\varphi \lor \psi$ is true in a team just in case the team can be split into two subteams, with each disjunct being true in at least one of the subteams; and an

¹These systems are all for *propositional inquisitive logic* InqB [41]. There is also an earlier labelled system [111] for an early version propositional inquisitive logic [60] interpreted on ordered pairs (essentially teams of size 2) rather than on teams in general.

6.1. Introduction

inquisitive disjunction is true in a team just in case one of the disjuncts is. The logic PL(W) is a conservative extension of classical propositional logic PL, with the split disjunction \vee extending the classical disjunction.

The calculus we construct features a standard Gentzen-style system for PL with some syntactic restrictions to the effect that certain active formulas and context sets must be classical (\forall -free). Adapting an idea from the natural deduction systems in [139, 140], this Gentzen-style system is supplemented with *deep-inference* (see, e.g., [62, 27, 28]) rules for the inquisitive disjunction w—that is, rules which allow one to introduce the inquisitive disjunction (almost) anywhere within a formula, rather than only as its main connective. The deep-inference rules allow for cutfree completeness of the system and for many standard proof-theoretic techniques to be applied despite the limited applicability of the restricted rules: essentially, cutfree proofs can be constructed by first constructing cutfree classical proofs, and then inserting inquisitive disjunctions as required; and procedures involving the commuting of sequents which depend on the universal applicability of the rules (such as cut elimination) can be conducted in such a way that they only involve commuting sequents in the classical part of the calculus, in which the rules *are* universally applicable. The cutfree fragment of the calculus has a weak subformula property; we define the relevant notion of weak subformula by generalizing the notion of *resolutions* from inquisitive logic [35].

Our aim was to develop a simple system with cut elimination that departs as little as possible from a Gentzen-style calculus. Our deep-inference approach accomplishes this in the following ways.

First, we avoid both importing the semantics into the system in the form of labels (like the labelled systems in [31, 100, 18]) and extending the syntax of the logic (like the multi-type display calculus in [50]).

Second, our system consists of only a single pair of rules for each connective: an introduction (right) rule and an elimination (left) rule (this is in contrast with the frequently very complicated natural deduction systems for team logics, including the natural deduction system for PL(\mathbb{W}) [139]).

Third, the system simply extends a well-known Gentzen-style system for PL (a variant of G3cp without the implication rules—see, e.g., [118]) with rules for the inquisitive disjunction \mathbb{V} —this means that the fact, mentioned above, that PL(\mathbb{V}) is an extension of PL with \mathbb{V} is directly reflected in a straightforward way in the calculus, and the calculus allows us to see immediately and transparently exactly what is required to be added to an axiomatization of PL to axiomatize PL(\mathbb{V}). It should be pointed out that this structure is possible due to a key design decision: instead of interpreting a sequent $\Gamma \Rightarrow \Delta$ as $\Lambda \Gamma \models \mathbb{V}\Delta$ (i.e., taking the sequent $\Gamma \Rightarrow \Delta$ to be valid just in case whenever each formula in Γ is true in a team *t*, at least one formula of Δ is true in *t*), we interpret $\Gamma \Rightarrow \Delta$ as $\Lambda \Gamma \models \mathbb{V}\Delta$ (i.e., $\Gamma \Rightarrow \Delta$ is valid just in case whenever each formula in Γ is true in a team *t*, at least one formula of Δ is true in *t*). That is, we interpret the comma in the succedent of a sequent not as the inquisitive disjunction \mathbb{V} , but, rather, as the split disjunction \vee . To reiterate, while the inquisitive disjunction \mathbb{V} has the standard disjunction semantics (with respect to

teams), it is the split disjunction \vee that extends the classical disjunction of PL; therefore, in order to employ, as we have done, a calculus for PL in the team setting without making extensive changes, the succedent comma must be interpreted as \vee . The labelled systems in [31, 100, 18], in contrast, interpret the succedent comma as W; the multi-type display calculus in [50] essentially interprets the succedent comma as \vee for one type of the system, and as W for the other.

Fourth and finally, due to the third point above, we can show many proof-theoretic results for our systems as easy extensions or corollaries of the analogous results for the classical Gentzen-style system. The following results in this paper are examples of this: the admissibility (with some restrictions) of height-preserving weakening, contraction and inversion (Section 6.5), the G3cp-style proof of cutfree completeness and countermodel construction in Section 6.6; and our cut elimination procedure (Section 6.7).

The system also has other interesting features. For instance, due to the decision to interpret the succedent comma as the split disjunction, there is a correspondence between certain structural rules of the calculus and certain *team-semantic closure properties*—semantic properties of formulas which play an important role in the study of team logics. The closure properties which have structural correspondents in our system are the *empty team property* and *union closure*. The logic PL(W) has the empty team property (meaning that the empty team satisfies all PL(W)-formulas), and its classical fragment is union closed (meaning that the truth of a classical formula in a collection of teams implies its truth in the union of the collection); PL(W) as a whole is not union closed. The empty team property corresponds to the soundness of weakening on the right, and union closure corresponds to the soundness of contraction on the right. Therefore, weakening on the right is sound for all PL(W)-formulas), whereas contraction on the right is only guaranteed to be sound (and admissible in the cutfree fragment) for classical formulas.

The paper is structured as follows. In Section 6.2.1, we define the syntax and semantics of the logic PL(W) and recall some basic facts about team semantics and this logic. In Section 6.2.2, we adapt the notion of *resolutions* from inquisitive logic to our setting—resolutions turn out to be a convenient tool for describing how our system functions. In Section 6.3, we introduce our deep-inference sequent calculus GT for PL(W). We also provide additional motivation for the deep-inference approach by discussing why a straightforward sequent calculus-translation of the natural deduction system for PL(W) would not be cutfree complete, and how the deep-inference rules provide for a natural cutfree extension of this translation. In Section 6.4, we generalize the notion of resolutions to what we call *partial resolutions*, and use partial resolutions to define a weak subformula property for the cutfree fragment of our system as well as to prove that the system is cutfree complete (that is, we provide in this section a semantic proof of cut elimination). In Section 6.5 we prove some basic properties of the system: the admissibility (with some restrictions) of height-preserving weakening, contraction, and inversion in the cutfree fragment. In Section 6.6, we make use of

6.2. Preliminaries

the (semantic) invertibility of the rules in our system to provide a procedure for constructing cutfree proofs and countermodels that is similar to the analogous procedure for G3cp (this yields a second semantic proof of cut elimination). In Section 6.7, we give a syntactic proof of cut elimination—a cut elimination procedure. This makes use of a *normal form theorem* for derivations (or what is known in the deep inferenceliterature as a *decomposition theorem*): each derivation in the cutfree fragment of GT can be transformed into a derivation in which one first applies only rules within the classical subsystem of GT, then applies only the right deep-inference rule for \forall , and finally applies only the left deep-inference rule for \forall . Using this theorem, each cut in GT can be transformed into cuts within the classical subsystem of GT, which can then be eliminated using the cut elimination procedure for this subsystem. In Section 6.8, we define a variant of GT featuring independent-context rules instead of sharedcontext rules with syntactic restrictions. Section 6.9 concludes with some discussion concerning the applicability of the deep-inference approach to other team logics.

6.2 Preliminaries

6.2.1 Syntax, Semantics, and Closure Properties

We fix a (countably infinite) set Prop of propositional variables.

6.2.1. DEFINITION (Syntax). The set of formulas α of *classical propositional logic* PL is generated by:

$$\alpha ::= p \mid \bot \mid \neg \alpha \mid \alpha \land \alpha \mid \alpha \lor \alpha$$

where $p \in \mathsf{Prop}$.

The set of formulas φ of *propositional logic with the inquisitive disjunction* PL(\mathbb{W}) is generated by:

$$\varphi ::= \alpha \mid \varphi \land \varphi \mid \varphi \lor \varphi \mid \varphi \lor \varphi$$

where $\alpha \in PL$.

We let $\top := \neg \bot$, $\bigvee \emptyset := \bot$, and $\bigwedge \emptyset := \top$. \lor is the *split disjunction* (also known as the *tensor disjunction* and the *local disjunction*) and \lor is the *inquisitive disjunction* (also known as the *global disjunction*). We write $\mathsf{P}(\varphi)$ for the set of propositional variables appearing in φ , and $\varphi(p_1, \ldots, p_n)$ if $\{p_1, \ldots, p_n\} \subseteq \mathsf{P}(\varphi)$. For a set/multiset of formulas Γ , we let $\mathsf{P}(\Gamma) := \bigcup_{\varphi \in \Gamma} \mathsf{P}(\Gamma)$. We also call formulas of PL *classical formulas*. We reserve the first lowercase Greek letters α, β for classical formulas, and the uppercase Greek letters $\Xi, \Lambda, \Theta, \Omega$ for multisets of classical formulas. The set of *subformulas* of a formula is defined in the standard way.

Let N \subseteq Prop. An *team with domain* N is a set $t \subseteq 2^{N}$ of valuations over N.

6.2.2. DEFINITION (Semantics). For any formula φ and any team *t* with domain $\supseteq P(\varphi)$, the satisfaction relation $t \models \varphi$ is defined inductively by:

- $t \models p$ iff for all $v \in t$, v(p) = 1;
- $t \models \bot$ iff $t = \emptyset$;
- $t \vDash \neg \alpha$ iff for all $v \in t$, $\{v\} \nvDash \alpha$;
- $t \models \phi \land \psi$ iff $t \models \phi$ and $t \models \psi$;
- $t \models \varphi \lor \psi$ iff there exist $r, s \subseteq t$ such that $t = r \cup s, r \models \varphi$ and $s \models \psi$;
- $t \models \phi \lor \psi$ iff $t \models \phi$ or $t \models \psi$.

For any multiset $\Gamma \cup \varphi$ of formulas, we write $t \models \Gamma$ if $t \models \psi$ for all $\psi \in \Gamma$, and we write $\Gamma \models \varphi$ if $t \models \Gamma$ implies $t \models \varphi$. We write $\varphi \models \psi$ for $\{\varphi\} \models \psi$. We write $\models \varphi$ if $t \models \varphi$ for all teams *t* with domain $\supseteq P(\varphi)$.

Let φ be a formula. We define the following standard *closure properties*:

Empty team property $\emptyset \vDash \varphi$.

Downward closure If $t \models \varphi$ and $s \subseteq t$, then $s \models \varphi$.

Union closure If $t \models \varphi$ and $s \models \varphi$, then $t \cup s \models \varphi$.

Flatness $t \models \varphi \iff$ for all $v \in t$, $\{v\} \models \varphi$.

It is easy to see that φ is flat iff it satisfies the empty team, downwards-closure, and union-closure properties.

6.2.3. PROPOSITION. All formulas of PL(W) satisfy the empty team and downwardclosure properties. All formulas of PL additionally satisfy the union-closure property and hence also the flatness property.

We also have that the team semantics of formulas of PL on singletons coincide with their standard single-valuation semantics: $\{v\} \models \alpha \iff v \models \alpha$. We therefore have that for any $\alpha \in PL$:

 $t \models \alpha \iff \{v\} \models \alpha \text{ for all } v \in t \iff v \models \alpha \text{ for all } v \in t.$

6.2.4. COROLLARY. For any set/multiset $\Lambda \cup \{\alpha\}$ of classical formula, $\Lambda \models^c \alpha \iff \Lambda \models \alpha$, where \models^c stands for the usual (single-valuation) entailment relation.

To see why formulas including \forall might fail to be union closed, consider $p \forall \neg p$. If $v_p \vDash p$ and $v_{\overline{p}} \vDash \neg p$, then $\{v_p\} \vDash p \forall \neg p$ and $\{v_{\overline{p}}\} \vDash p \forall \neg p$, but $\{v_p, v_{\overline{p}}\} \vDash p \forall \neg p$.

PL(\mathbb{W}) does not admit uniform substitution: writing $\varphi(\chi/p)$ for the result of replacing all occurrences of p in $\varphi(p)$ with χ , it is not the case that $\varphi(p) \models \psi(p)$ implies $\varphi(\chi/p) \models \psi(\chi/p)$. For instance, we have $p \lor p \models p$, but $(p \lor \neg p) \lor (p \lor \neg p) \not\models p \lor \neg p$.

6.2.2 **Resolutions and Normal Form**

In this section, we adapt the notion of *resolutions* from inquisitive logic [35], and list pertinent results. The proofs of these results are analogous to their proofs in propositional inquisitive logic; see [35] for details.

As discussed in Section 6.1, the inquisitive disjunction \mathbb{V} is used in inquisitive logic as a question-forming connective—for instance, the question as to whether or not p is the case is represented by $p \mathbb{V} \neg p$. Resolutions serve mainly a technical purpose, but it is helpful to think of the resolutions of a formula φ as representing the answers to φ (the different possible ways of truthfully resolving φ) whenever φ represents a question. If φ does not contain instances of \mathbb{V} , it does not represent a question, and its only resolution is φ itself. In Section 6.4, we generalize the notion of resolutions to *partial resolutions*, and use partial resolutions to prove a weak subformula property for and the cutfree completeness of our sequent calculus.

6.2.5. DEFINITION (Resolutions). The set $\mathcal{R}(\varphi)$ of *resolutions* of a formula φ is defined recursively as follows:

- $\mathcal{R}(p) \coloneqq \{p\};$
- $\mathcal{R}(\bot) \coloneqq \{\bot\};$
- $\mathcal{R}(\neg \alpha) \coloneqq \{\neg \beta \mid \beta \in \mathcal{R}(\alpha)\};$
- $\mathcal{R}(\varphi \land \psi) \coloneqq \{\alpha \land \beta \mid \alpha \in \mathcal{R}(\varphi) \text{ and } \beta \in \mathcal{R}(\psi)\};$
- $\mathcal{R}(\varphi \lor \psi) \coloneqq \{ \alpha \lor \beta \mid \alpha \in \mathcal{R}(\varphi) \text{ and } \beta \in \mathcal{R}(\psi) \};$
- $\mathcal{R}(\varphi \lor \psi) \coloneqq \mathcal{R}(\varphi) \cup \mathcal{R}(\psi).$

Clearly for $\varphi \in PL(W)$, $\mathcal{R}(\varphi) \subseteq PL$; and for $\alpha \in PL$, $\mathcal{R}(\alpha) = \{\alpha\}$.

6.2.6. DEFINITION (Resolutions for multisets). A *resolution function for a multiset of* formulas $\Gamma \subseteq PL(\mathbb{W})$ is a map $f : \Gamma \to PL$ such that for each $\varphi \in \Gamma$, $f(\varphi) \in \mathcal{R}(\varphi)$. The set (of multisets) $\mathcal{R}(\Gamma)$ of *resolutions* of Γ is:

 $\mathcal{R}(\Gamma) \coloneqq \{f[\Gamma] \mid f \text{ if a resolution function for } \Gamma\}.$

As above, for $\Gamma \subseteq PL(\mathbb{W})$, $\mathcal{R}(\Gamma) \subseteq PL$; and for $\Lambda \subseteq PL$, $\mathcal{R}(\Lambda) = \{\Lambda\}$.

Each formula is equivalent to the inquisitive disjunction of its resolutions (and hence to an inquisitive disjunction of classical formulas), and similarly for multisets of formulas.

6.2.7. PROPOSITION (Normal form). $[\varphi \equiv \bigvee \mathcal{R}(\varphi)]$ and $[t \models \Gamma \iff t \models \bigcup \mathcal{R}(\Gamma)]$.

Proof:

Follows from the fact that \land and \lor distribute over \lor .

The *Split property* is a generalization of the disjunction property (with respect to \forall): a classical multiset of formulas entails a \forall -disjunction just in case it entails one of the disjuncts.

6.2.8. PROPOSITION (Split property). For $\Lambda \subseteq PL$ and $\varphi, \psi \in PL(\mathbb{W})$:

$$\Lambda \vDash \phi \lor \psi \iff [\Lambda \vDash \phi \text{ or } \Lambda \vDash \psi].$$

Proof:

By the union closure of PL and the downward closure of PL(W).

Finally, a multiset of formulas Γ entails ψ just in case each resolution of Γ entails some resolution of ψ :

6.2.9. THEOREM (Resolution theorem).

$$\Gamma \vDash \psi \iff$$
 for each $\Lambda \in \mathcal{R}(\Gamma)$ there is some $\alpha \in \mathcal{R}(\psi)$ such that $\Lambda \vDash \alpha$.

Proof:

By Propositions 6.2.7 and 6.2.8.

6.3 The System GT

We introduce the sequent calculus GT for PL ('G' stands for Gentzen, 'T' for team). This is a G3-style system (structural rules are incorporated into the logical rules; see Section 6.8 for an alternative system featuring some explicit structural rules) with deep-inference style rules for the inquisitive disjunction \mathbb{V} .

Given formulas $\varphi_1, \ldots, \varphi_n$ and multisets of formulas Γ and Δ , we use the notation $\varphi_1, \ldots, \varphi_n, \Gamma$ to denote the multiset $\{\varphi_1, \ldots, \varphi_n\} \cup \Gamma$, and the notation Γ, Δ to denote the multiset $\Gamma \cup \Delta$ (in both cases, \cup denotes the multiset union operation). A *sequent* is an expression of the form $\Gamma \Rightarrow \Delta$, where Γ and Δ are finite multisets of formulas. In $\Gamma \Rightarrow \Delta$, Γ is the *antecedent* and Δ is the *succedent*. The intended interpretation of $\Gamma \Rightarrow \Delta$ is $\land \Gamma \models \lor \Delta$; we say that $\Gamma \Rightarrow \Delta$ is *valid* if $\land \Gamma \models \lor \Delta$. Let us emphasize that the commas in the succedent of a sequent $\varphi_1, \ldots, \varphi_n \Rightarrow \psi_1, \ldots, \varphi_m$ are interpreted as the split disjunction \lor (while the commas in the antecedent are interpreted as conjunction \land); the sequent $\varphi_1, \ldots, \varphi_n \Rightarrow \psi_1, \ldots, \varphi_m$ is valid just in case for any team *t*, if $t \models \varphi_1 \land \ldots \land \varphi_n$, then $t = t_1 \cup \ldots \cup t_m$, where $t_1 \models \psi_1, \ldots, t_m \models \psi_m$.

Our system makes use of a notion of formula contexts to implement deep-inference rules. A *(formula) context* $\varphi\{\cdot_1\} \dots \{\cdot_n\}$ is a formula φ with designated atoms $\cdot_1 \dots \cdot_n$ (each of which occurs only once in φ). We do not allow contexts in which any \cdot_i is in

the scope of a negation \neg . We write $\varphi\{\psi_1\} \dots \{\psi_n\}$ for the result of replacing \cdot_1, \dots, \cdot_n in $\varphi\{\cdot_1\} \dots \{\cdot_n\}$ with ψ_1, \dots, ψ_n , respectively. For instance, if $\psi\{\cdot\} = p \land (\cdot \lor r)$, then $\psi\{q \lor p\} = p \land ((q \lor p) \lor r)$.

6.3.1. DEFINITION (The sequent calculus GT).

| Axioms | | | | | |
|--|--|--|--|--|--|
| $\Gamma, p \Rightarrow p, \Delta$ At | $\Gamma, \bot \Rightarrow \Delta$ Li | | | | |
| Logical rules | | | | | |
| $\frac{\Gamma \Rightarrow \alpha, \Delta}{\Gamma, \neg \alpha \Rightarrow \Delta} L \neg$ | $\frac{\Gamma, \alpha \Rightarrow \Delta}{\Gamma \Rightarrow \neg \alpha, \Delta} R_{\neg}$ | | | | |
| $\frac{\Gamma, \varphi, \psi \Rightarrow \Delta}{\Gamma, \varphi \land \psi \Rightarrow \Delta} L \land$ | $\frac{\Gamma \Rightarrow \varphi, \Lambda \qquad \Gamma \Rightarrow \psi, \Lambda}{\Gamma \Rightarrow \varphi \land \psi, \Lambda, \Delta} \operatorname{R}_{\wedge}$ | | | | |
| $\frac{\Gamma, \varphi \Rightarrow \Lambda \qquad \Gamma, \psi \Rightarrow \Lambda}{\Gamma, \varphi \lor \psi \Rightarrow \Lambda, \Delta} L \lor$ | $\frac{\Gamma \Rightarrow \varphi, \psi, \Delta}{\Gamma \Rightarrow \varphi \lor \psi, \Delta} R \lor$ | | | | |
| $\frac{\Gamma, \chi\{\varphi_L\} \Rightarrow \Delta \qquad \Gamma, \chi\{\varphi_R\} \Rightarrow \Delta}{\Gamma, \chi\{\varphi_L \lor \varphi_R\} \Rightarrow \Delta} \sqcup \lor$ | $\frac{\Gamma \Rightarrow \chi\{\varphi_i\}, \Delta}{\Gamma \Rightarrow \chi\{\varphi_L \lor \varphi_R\}, \Delta} R \lor$ | | | | |
| Cut | | | | | |
| $\frac{\Gamma \Rightarrow \varphi, \Delta \qquad \Pi, \varphi \Rightarrow \Sigma}{\Pi, \Gamma \Rightarrow \Delta, \Sigma} \operatorname{Cut}$ | | | | | |

The sequent(s) above the line in each rule (application) are called the *premise(s)* of that rule (application); the sequent below the line is the *conclusion* of the rule. The formulas $\varphi, \psi, \alpha, \chi\{\varphi_i\}$ in the premise(s) are the *active formulas* of the rule; the newly introduced formula in the conclusion is the *principal formula* of the rule. Γ, Δ, Λ are *side formulas* or the *left/right context* ((*multi)set*) (left: Γ ; right: Δ, Λ). In L \forall and R \forall , we call the subformula(s) φ_i in the premise(s) the *active subformulas* of the rule and the subformula(s) φ_i in the conclusion the *principal subformula* of the rule; for all other rules, their active subformula(s) are the same as their active formula(s), and similarly for their principal subformula(s). The formula φ in Cut is the *cutformula*, and we say Cut is a *cut on* φ .²

The rules $L \neg$, $R \neg$, $R \land$, and $L \lor$ feature syntactic restrictions. The restrictions on the negation rules are due to the corresponding restriction in the syntax of PL(w). The

 $^{^{2}}$ We will often refer to formula occurrences simply as formulas for simplicity, and disambiguate when it aids clarity.

restrictions that the premise right context in $\mathbb{R} \wedge$ and $\mathbb{L} \vee$ be classical are required for soundness—note, for instance, that $\mathbb{L} \vee$ without the restriction would not be sound, essentially due to the failure of union closure of non-classical formulas: $p \models p \lor \neg p$ and $\neg p \models p \lor \neg p$, but $p \lor \neg p \nvDash p \lor \neg p$. The restriction to a classical (and hence unionclosed) premise right context ensures soundness. These two rules also feature implicit weakening (Δ is added to the right context) to make up for the loss in strength incurred due to the syntactic restrictions—they allow us to prove that right weakening is admissible in the system (see Lemma 6.5.1).

Relatedly, observe that with the comma on the right of a sequent corresponding to the split disjunction, we have the following correspondences between structural rules on the right and team-semantic closure properties. Right weakening $[\models \Gamma \Rightarrow \Delta, \xi$ implies $\models \Gamma \Rightarrow \Delta, \xi, \xi$] is sound for ξ if ξ has the empty team property; and right contraction $[\models \Gamma \Rightarrow \Delta, \xi, \xi$ implies $\models \Gamma \Rightarrow \Delta, \xi$] is sound for ξ if ξ is union closed. In our setting, all formulas have the empty team property and classical formulas are union closed; accordingly, right weakening is sound for all formulas, and right contraction is sound for classical formulas. To see why right contraction is not sound in general with respect to formulas which are not union closed, note that $(p \lor \neg p) \lor (p \lor \neg p) \Rightarrow$ $(p \lor \neg p), (p \lor \neg p)$ is valid whereas $(p \lor \neg p) \lor (p \lor \neg p) \Rightarrow (p \lor \neg p)$ is not.

LW and RW are our deep-inference rules. They are standard classical disjunction rules, strengthened to allow the active/principal subformula(s) to appear within any position (not in the scope of a negation) in the active/principle formula(s). The soundness of these rules follows from the fact that \land and \lor distribute over W. Example application:

$$\frac{\Gamma, p \land (q \lor r) \Rightarrow \Delta}{\Gamma, p \land ((q \lor r) \lor (s \lor (q \land \neg p)) \Rightarrow \Delta} \sqcup \forall$$

Given any set of sequent calculus rules C, we let C also denote the system consisting of the rules in C. Let G3cp (for PL) be the system $GT \setminus \{L \otimes, R \otimes\}$. This is a G3style cutfree complete system for PL—the soundness (given the standard semantics for classical propositional logic) and cutfree completeness of this system follow easily from that for the usual G3cp (see, e.g., [118]). We write C⁻ for the Cut-free fragment of C, i.e., C⁻ := C \ Cut. We write $\vdash_C \Gamma \Rightarrow \Delta$ (or simply $\vdash \Gamma \Rightarrow \Delta$ when the system is clear from the context) if there is a derivation of $\Gamma \Rightarrow \Delta$ in calculus C. If \mathcal{D} is a derivation of $\Gamma \Rightarrow \Delta$ in calculus C, we say that \mathcal{D} witnesses $\vdash_C \Gamma \Rightarrow \Delta$. We define the *height* of a derivation \mathcal{D} inductively: if \mathcal{D} consists of a single axiom, the height of \mathcal{D} is one; otherwise the height of \mathcal{D} is 1 plus the maximum of the heights of the subderivations ending in the premises of the final rule application in \mathcal{D} .

We now verify that the system is sound given the team semantics for $PL(\mathbb{V})$.

6.3.2. THEOREM (Soundness of GT). $\vdash \Gamma \Rightarrow \Delta$ implies $\land \Gamma \vDash \lor \Delta$.

Proof:

By induction on the height of derivations; we separate into different cases based on the final rule applied in the derivation.

- At: $p \models p \lor \lor \Delta$ by $p \models p$ and the empty team property.
- L1: $\perp \models \lor \Delta$ by the empty team property.
- L¬: Assume $\land \Gamma \vDash \alpha \lor \lor \Delta$. We want to show $\neg \alpha \land \land \Gamma \vDash \lor \Delta$, so let $t \vDash \neg \alpha \land \land \Gamma$. By $\land \Gamma \vDash \alpha \lor \lor \Delta$, we have $t \vDash \alpha \lor \lor \Delta$, so $t = t_{\alpha} \cup t_{\Delta}$ where $t_{\alpha} \vDash \alpha$ and $t_{\Delta} \vDash \lor \Delta$. By $t_{\alpha} \vDash \alpha$, $t \vDash \neg \alpha$, and the flatness of α and $\neg \alpha$, we have that for each $v \in t_{\alpha}$, $v \vDash \alpha \land \neg \alpha$; therefore $t_{\alpha} = \emptyset$, whence $t = t_{\Delta} \vDash \lor \Delta$.
- R¬: Assume $\wedge \Gamma \wedge \alpha \models \lor \Delta$. We want to show $\wedge \Gamma \models \neg \alpha \lor \lor \Delta$, so let $t \models \wedge \Gamma$. Let $t_{\alpha} := \{v \in t \mid v \models \alpha\}$ and $t_{\neg \alpha} := t \lor t_{\alpha}$. By the flatness of α we have $t_{\alpha} \models \alpha$, and by downward closure we have $t_{\alpha} \models \wedge \Gamma$; therefore by $\wedge \Gamma \wedge \alpha \models \lor \Delta$, we have $t_{\alpha} \models \lor \Delta$. By the flatness of $\neg \alpha$, we have $t_{\neg \alpha} \models \neg \alpha$, so that by $t = t_{\neg \alpha} \cup t_{\alpha}$, we have $t \models \neg \alpha \lor \lor \Delta$.
- LA: Follows immediately from the induction hypothesis.
- RA: Assume $\land \Gamma \vDash \varphi \lor \lor \land$ and $\land \Gamma \vDash \psi \lor \lor \land$. We want to show $\land \Gamma \vDash (\varphi \land \psi) \lor \lor \land$, so let $t \vDash \land \Gamma$. Then $t \vDash \varphi \lor \lor \land \land$ and $t \vDash \psi \lor \lor \land \land$, so $t = t_{\varphi} \cup t_{\Lambda 1}$ where $t_{\varphi} \vDash \varphi$ and $t_{\Lambda 1} \vDash \lor \land \land$, and $t = t_{\psi} \cup t_{\Lambda 2}$ where $t_{\psi} \vDash \psi$ and $t_{\Lambda 2} \vDash \lor \land \land$. Then also $t = (t_{\varphi} \cap t_{\psi}) \cup t_{\Lambda 1} \cup t_{\Lambda 2}$. By downward closure, $t_{\varphi} \cap t_{\psi} \vDash \varphi \land \psi$, and by the union closure of $\lor \land \land$, we have $t_{\Lambda 1} \cup t_{\Lambda 2} \vDash \lor \land \land$. Therefore, $t \vDash (\varphi \land \psi) \lor \lor \land$.
- Lv: Assume $\wedge \Gamma \wedge \varphi \models \vee \Lambda$ and $\wedge \Gamma \wedge \psi \models \vee \Lambda$. We want to show $\wedge \Gamma \wedge (\varphi \vee \psi) \models \vee \Lambda$, so let $t \models \wedge \Gamma \wedge (\varphi \vee \psi)$. Then $t = t_{\varphi} \cup t_{\psi}$ where $t_{\varphi} \models \varphi$ and $t_{\psi} \models \psi$, and $t \models \wedge \Gamma$. By downward closure, $t_{\varphi} \models \wedge \Gamma$ so that $t_{\varphi} \models \vee \Lambda$; similarly $t_{\psi} \models \vee \Lambda$. By the union closure of $\vee \Lambda$, we have $t \models \vee \Lambda$.
- $R\lor$: Follows immediately from the induction hypothesis.
- RW: Follows immediately from the induction hypothesis and the fact that ∧ and ∨ distribute over W.
- L \mathbb{W} : Assume $\wedge \Gamma \wedge \chi\{\varphi_L\} \models \vee \Delta$ and $\wedge \Gamma \wedge \chi\{\varphi_R\} \models \vee \Delta$. We want to show $\wedge \Gamma \wedge \chi\{\varphi_L \lor \varphi_R\} \models \vee \Delta$, so let $t \models \wedge \Gamma \wedge \chi\{\varphi_L \lor \varphi_R\}$. Since \wedge and \vee distribute over \mathbb{W} , $\chi\{\varphi_L \lor \varphi_R\} \equiv \chi\{\varphi_L\} \lor \chi\{\varphi_R\}$, so $t \models \chi\{\varphi_L\}$ or $\chi\{\varphi_R\}$. In either case, by our assumptions $t \models \vee \Delta$.
- Cut: Assume $\land \Gamma \vDash \varphi \lor \lor \Delta$ and $\land \Pi \land \varphi \vDash \lor \Sigma$. We want to show $\land \Gamma \land \land \Pi \vDash \lor \Delta \lor \lor \Sigma$, so let $t \vDash \land \Gamma \land \land \Pi$. By $\land \Gamma \vDash \varphi \lor \lor \Delta$, $t \vDash \varphi \lor \lor \Delta$, whence $t = t_{\varphi} \cup t_{\Delta}$, where $t_{\varphi} \vDash \varphi$ and $t_{\Delta} \vDash \lor \Delta$. By downward closure, $t_{\varphi} \vDash \lor \Pi$, whence by $\land \Pi \land \varphi \vDash \lor \Sigma$, $t_{\varphi} \vDash \lor \Sigma$. Therefore, $t \vDash \lor \Delta \lor \lor \Sigma$.

6.3.3. THEOREM (Cut elimination for G3cp). [118] If \mathcal{D} witnesses $\vdash_{G3cp} \Xi \Rightarrow \Lambda$, there is an effective procedure for transforming \mathcal{D} into a derivation \mathcal{D}' witnessing $\vdash_{G3cp} \Xi \Rightarrow \Lambda$.

6.3.4. COROLLARY (Cutfree classical completeness). $\land \Xi \vDash \lor \land implies \vdash_{\mathsf{G3cp}^{-}} \Xi \Rightarrow \land$ (whence also $\vdash_{\mathsf{GT}^{-}} \Xi \Rightarrow \land and \vdash_{\mathsf{GT}} \Xi \Rightarrow \land$).

Why Deep Inference?

One possible natural translation of the natural deduction rules for PL(W) [139] into sequent calculus rules would give us the system $G3cp \cup \{LW^-, RW^-, LDstr, RDstr\}$, where:

$$\frac{\Gamma, \varphi_L \Rightarrow \Delta \qquad \Gamma, \varphi_R \Rightarrow \Delta}{\Gamma, \varphi_L \lor \varphi_R \Rightarrow \Delta} \sqcup_{\mathbb{V}^-} \qquad \frac{\Gamma \Rightarrow \varphi_i, \Delta}{\Gamma \Rightarrow \varphi_L \lor \varphi_R, \Delta} \: \mathsf{R} \lor_-$$

$$\frac{\Gamma, \varphi \lor \psi_L \Rightarrow \Delta \qquad \Gamma, \varphi \lor \psi_R \Rightarrow \Delta}{\Gamma, \varphi \lor (\psi_L \lor \psi_R) \Rightarrow \Delta} \: \mathsf{LDstr} \qquad \frac{\Gamma \Rightarrow \varphi \lor (\psi_L \lor \psi_R), \Delta}{\Gamma \Rightarrow (\varphi \lor \psi_L) \lor (\varphi \lor \psi_R), \Delta} \: \mathsf{RDstr}$$

This system is not cutfree complete. For instance, the following valid sequent is not derivable:

$$(r \land x) \lor (((p \land x) \lor (q \land x)) \lor (y \land x)) \Rightarrow (x \land (r \lor (p \lor y))) \lor (x \land (r \lor (q \lor y))).$$

There is no rule of this system that any putative derivation of this sequent could have concluded with. Any putative derivation could not have concluded with either of the distributivity rules because the putative principal formula of such a rule application would not be of the right form in either case. It could not have concluded with RW^- or LV because none of the possible premises of such a rule application are valid. Finally, it could not have concluded with any other rule because the main connective of the putative principal formula of such a rule application would not be the connective of the rule.

The problem is that the distributivity rules do not reach deep enough into the active/principal formula (we conjecture that one can formulate a notion of context depth and show that a system with the distributivity rules is cutfree complete for formulas in which W only appears up to certain depth). The deep-inference rules LW and RWfix this issue and have the added benefit of allowing for a system consisting only of a single introduction (right) and single elimination (left) rule for each connective.

6.4 Partial Resolutions and Cutfree Completeness

GT⁻ has a weak subformula property. We can formulate this property in terms of what we will call *partial resolutions*—a generalization of resolutions (Section 6.2.2). Partial resolutions also yield an easy way of showing the cutfree completeness of GT.

In order to define partial resolutions, it is helpful to first define some more notation for the type of substitutions carried out via the deep inference rules. A \vee -*labelled formula (occurrence)* is a formula occurrence φ together with a function f mapping each occurrence of \vee in φ to \mathbb{N} . Examples: $p \vee_0(q \vee_1 r)$; $(p \vee_2 x) \vee_2(q \vee_1 r)$. We will frequently refer to \forall -labelled occurrences as simply formula occurrences or formulas, and switch between labelled and unlabelled occurrences when it is convenient to do so. The \forall -labelling φ_{\forall} of a formula occurrence φ is the \forall -labelled formula occurrence consisting of the formula φ and the function mapping each occurrence of \forall in φ according to its position in the sequence of such occurrences; e.g., if $\varphi = p \forall (q \forall r)$, then $\varphi_{\forall} = p \forall_0 (q \forall_1 r)$. Below, we will use \forall -labellings φ_{\forall} to define other \forall -labelled formulas. We use the notation $\psi[\chi_1 \lor_i \chi_2]^{\varphi}$ (or simply $\psi[\chi_1 \lor_i \chi_2]$ or $\psi[i]^{\varphi}$ or $\psi[i]$) to refer to a subformula occurrence $\chi_L \lor_i \chi_R$ in ψ with φ_{\forall} -label *i*. We let $\psi[i/j]^{\varphi}$ (or simply $\psi[i/j]$), where $i \in \{L, R\}$, denote the result of replacing each $\psi[\chi_L \lor_j \chi_R]^{\varphi}$ in ψ (if any such exist) with χ_i . For instance, for φ as above, let $\psi := \varphi[R/1]^{\varphi} = p \lor_0 r$. Then $\psi[0]^{\varphi} = \psi, \ \psi[L/0] = p$, and $\psi[L/1] = \psi$. We let $|\varphi|_{\forall}$ denote the number of occurrences of \forall in φ .

6.4.1. DEFINITION (Partial resolutions). The set $\mathcal{PR}_n(\varphi)$ of *partial resolutions of* φ *of degree* $n \in \{0, ..., |\varphi|_W\}$ is defined by

$$\mathcal{PR}_n(\varphi) \coloneqq \{ \varphi[i_1/j_1]^{\varphi} \dots [i_n/j_n]^{\varphi} | \{i_x\}_{1 \le x \le n} \subseteq \{L, R\}, \\ \{j_x\}_{1 \le x \le n} \subseteq \{0, \dots, |\varphi|_{\mathbb{W}} - 1\}, \\ [x \neq y \text{ implies } j_x \neq j_y] \}.$$

The set $\mathcal{PR}(\varphi)$ of *partial resolutions of* φ is defined by

$$\mathcal{PR}(\boldsymbol{\varphi}) \coloneqq \bigcup_{0 \le n \le |\boldsymbol{\varphi}|_{\mathcal{W}}} \mathcal{PR}_n(\boldsymbol{\varphi}).$$

Example: for $\varphi_{\mathbb{W}} = p \mathbb{V}_0(q \mathbb{V}_1 r)$,

$$\begin{aligned} \mathcal{PR}_{0}(\varphi) &= \{\varphi\} = \{p \otimes_{0}(q \otimes_{1}r)\}; \\ \mathcal{PR}_{1}(\varphi) &= \{\varphi[L/0], \varphi[R/0], \varphi[L/1], \varphi[R/1]\} \\ &= \{p, q \otimes_{1}r, p \otimes_{0}q, p \otimes_{0}r\}; \\ \mathcal{PR}_{2}(\varphi) &= \{\varphi[L/0][L/1], \varphi[L/0][R/1], \varphi[R/0][L/1], \varphi[R/0][R/1], \varphi[R/0][R/1]], \varphi[L/1][L/0], \varphi[R/1][R/0]\} \\ &= \{p, q, r\}. \end{aligned}$$

It is easy to see that $\mathcal{PR}_{|\varphi|_{W}}(\varphi) = \mathcal{R}(\varphi)$; we generalize this in Proposition 6.4.5 below. The guiding intuition behind partial resolutions is that if \forall represents a question-forming operator as in inquisitive logic, each formula in $\mathcal{PR}_{n}(\varphi)$ represents an attempt to resolve *n* "questions" in φ .³ The formulas in $\mathcal{R}(\varphi) = \mathcal{PR}_{|\varphi|_{W}}(\varphi)$, then, resolve all the questions in φ .

³The notion of "question" here is such that each occurrence $\forall \forall$ corresponds to one question. The formula $a \forall_0 (b \forall_1 c)$ (which could be thought to correspond roughly to "Is Mary in Amsterdam or is she in Beijing or in Copenhagen?") has two questions. Each of the partial resolutions *a* ("Mary is in Amsterdam") and $b \forall_1 c$ ("Is Mary in Beijing or in Copenhagen?") resolves question 0. The first partial resolution *a*, since it itself contains no questions, also constitutes a (full) resolution to φ .

We may now formulate the weak subformula property for GT^- . Observe that using labelled formulas we may reformulate the deep-inference rules $L \otimes$ and $R \otimes$ as follows:

$$\frac{\Gamma, \chi[L/j]^{\varphi} \Rightarrow \Delta \qquad \Gamma, \chi[R/j]^{\varphi} \Rightarrow \Delta}{\Gamma, \chi \Rightarrow \Delta} L \vee \qquad \qquad \frac{\Gamma \Rightarrow \chi[i/j]^{\varphi}, \Delta}{\Gamma \Rightarrow \chi, \Delta} R \vee$$

The following are therefore immediate from the shape of the rules:

6.4.2. LEMMA.

- (*i*) If $\Gamma \Rightarrow \chi[i/j]^{\varphi}, \Delta$, then $\Gamma \Rightarrow \chi, \Delta$.
- (*ii*) If $\Gamma, \chi[L/j]^{\varphi} \Rightarrow \Delta$ and $\Gamma, \chi[R/j]^{\varphi} \Rightarrow \Delta$, then $\Gamma, \chi \Rightarrow \Delta$.

6.4.3. PROPOSITION (Partial resolution subformula property). For any derivation \mathcal{D} witnessing $\Gamma \vdash_{\mathsf{GT}^-} \Delta$, each formula occurring in \mathcal{D} is a subformula of a partial resolution of some formula occurring in Γ, Δ .

It is also easy to see from the reformulation above—together with the fact that $\mathcal{PR}_{|\varphi|_{W}}(\varphi) = \mathcal{R}(\varphi)$ —that the cutfree completeness of GT follows essentially immediately from the resolution theorem (Theorem 6.2.9) in conjunction with cutfree classical completeness (Corollary 6.3.4). To make this more precise, we now also define partial resolutions for multisets. Given a multiset $\Gamma = \{\varphi_0, \dots, \varphi_n\}$, we assume Γ comes with some canonical ordering of its elements and write $\vec{\Gamma} = \langle \varphi_0, \dots, \varphi_n \rangle$ for the sequence corresponding to this canonical ordering. We write $\Gamma[i/j]_k^{\varphi}$ (or simply $\Gamma[i/j]_k$, or $\Gamma[i/j]_{\varphi_k}$) for the multiset consisting of the elements of the sequence $\langle \varphi_0, \dots, \varphi_k[i/j]^{\varphi}, \dots, \varphi_n \rangle$, where $\vec{\Gamma} = \langle \varphi_0, \dots, \varphi_n \rangle$. E.g., for $\varphi = p \vee_0 (q \vee_1 r)$ and $\psi = p \vee_0 r = \psi[0]^{\varphi}$ and $\vec{\Gamma} = \langle \varphi, \varphi, \psi \rangle$, we have $\Gamma[L/0]_2^{\varphi} = \{\varphi, \varphi, p\}$; $\Gamma[L/1]_1 = \{\varphi, p \vee_0 q, \psi\}$; and $\Gamma[L/1]_3 = \Gamma$. We let $|\Gamma|_{W} \coloneqq \sum_{\varphi \in \Gamma} |\varphi|_W$.

6.4.4. DEFINITION (Partial resolutions for multisets). The set (of multisets) $\mathcal{PR}_n(\Gamma)$ of *partial resolutions of* $\Gamma = \{\varphi_0, \dots, \varphi_q\}$ *of degree* $n \in \{0, \dots, |\Gamma|_W\}$ is defined by

$$\mathcal{PR}_{n}(\Gamma) \coloneqq \{ \Gamma[i_{1}/j_{1}]_{k_{1}}^{\varphi^{1}} \dots [i_{n}/j_{n}]_{k_{n}}^{\varphi^{n}} | \text{for each } x \in \{1, \dots, n\} \colon \varphi^{x} \in \Gamma, \\ \{i_{x}\}_{1 \le x \le n} \subseteq \{L, R\}, \\ \{k_{x}\}_{1 \le x \le n} \subseteq \{0, \dots, q\}, \\ \text{for each } w \in \{0, \dots, q\} \colon \\ \{j_{x} \mid \varphi^{x} = \varphi_{w}\}_{1 \le x \le n} \subseteq \{0, \dots, |\varphi_{w}|_{W} - 1\} \text{ and} \\ [[x \neq y \text{ and } \varphi^{x} = \varphi^{y} = \varphi_{w}] \text{ implies } j_{x} \neq j_{y}] \}.$$

The set (of multisets) $\mathcal{PR}(\varphi)$ of *partial resolutions of* φ is defined by

$$\mathcal{PR}(\boldsymbol{\varphi}) \coloneqq \bigcup_{0 \leq n \leq |\Gamma|_{W}} \mathcal{PR}_{n}(\Gamma).$$
Example: for $\Gamma = \{\varphi_0, \varphi_1\} = \{p \lor_0(q \lor_1 r), s \lor_0 r\}$, where we have $\vec{\Gamma} = \langle \varphi_0, \varphi_1 \rangle =$ $\langle p \otimes_0 (q \otimes_1 r), s \otimes_0 r \rangle$,

$$\begin{aligned} \mathcal{PR}_{0}(\Gamma) = \{\Gamma\} = \{\{p \otimes_{0}(q \otimes_{1}r), s \otimes_{0}r\}\}; \\ \mathcal{PR}_{1}(\Gamma) = \{\Gamma[L/0]_{0}, \Gamma[R/0]_{0}, \Gamma[L/1]_{0}, \Gamma[R/1]_{0}, \Gamma[L/0]_{1}, \Gamma[R/0]_{1}\} \\ = \{\{p, s \otimes_{0}r\}, \{q \otimes_{1}r, s \otimes_{0}r\}, \{p \otimes_{0}q, s \otimes_{0}r\}, \{p \otimes_{0}r, s \otimes_{0}r\}, \{p \otimes_{0}q \otimes_{1}r), s\}, \{p \otimes_{0}(q \otimes_{1}r), r\}\}; \\ \mathcal{PR}_{2}(\Gamma) = \{\{p, s \otimes_{0}r\}, \{p, s\}, \{p, r\}, \{q, s \otimes_{0}r\}, \{r, s \otimes_{0}r\}, \{q \otimes_{1}r, s\}, \{q \otimes_{1}r, r\}, \{p \otimes_{0}q, s\}, \{p \otimes_{0}q, r\}, \{p \otimes_{0}r, s\}, \{p \otimes_{0}r, r\}\}; \\ \mathcal{PR}_{3}(\Gamma) = \{\{p, s\}, \{p, r\}, \{q, s\}, \{q, r\}, \{r, s\}, \{r, r\}\} = \mathcal{R}(\Gamma). \end{aligned}$$

One can check that $\mathcal{PR}_n(\{\varphi\}) = \{\{\psi\} \mid \psi \in \mathcal{PR}_n(\varphi)\}$, so this notion generalizes that for formulas. It is easy to see that we have the following generalization of the fact we observed above:

6.4.5. PROPOSITION. $\mathcal{PR}_{|\Gamma|_{\mathbb{N}}}(\Gamma) = \mathcal{R}(\Gamma).$

And we have:

6.4.6. LEMMA.

- (i) For any $n \in \{0, ..., |\Delta_{W}| 1\}$, if there is a $\Delta_{n+1} \in \mathcal{PR}_{n+1}(\Delta)$ such that $\vdash_{\mathsf{GT}^{-}} \Gamma \Rightarrow$ Δ_{n+1} , then there is a $\Delta_n \in \mathcal{PR}_n(\Delta)$ such that $\vdash_{\mathsf{GT}^-} \Gamma \Rightarrow \Delta_n$.
- (ii) For any $n \in \{0, \dots, |\Delta_{W}| 1\}$, if for all $\Gamma_{n+1} \in \mathcal{PR}_{n+1}(\Delta)$ we have $\vdash_{\mathsf{GT}^{-}} \Gamma_{n+1} \Rightarrow \Delta$, then for all $\Gamma_n \in \mathcal{PR}_n(\Gamma)$ we have $\vdash_{\mathbf{GT}^-} \Gamma_n \Rightarrow \Delta$.

Proof:

Follows by the definition of partial resolutions together with Lemma 6.4.2.

6.4.7. THEOREM (Cutfree completeness). $\land \Gamma \vDash \lor \Delta$ implies $\vdash_{GT^{-}} \Gamma \Rightarrow \Delta$.

Proof:

By the resolution theorem (Theorem 6.2.9), for each $\Xi \in \mathcal{R}(\Gamma)$, there is some $\alpha \in$ $\mathcal{R}(\vee \Delta)$ such that $\Xi \models \alpha$, i.e., there is some $\Lambda \in \mathcal{R}(\Delta)$ such that $\Lambda \Xi \models \vee \Lambda$. By cutfree classical completeness (Corollary 6.3.4), also $\vdash_{\mathsf{GT}^-} \Xi \Rightarrow \Lambda$. So by Proposition 6.4.5, for each $\Xi \in \mathcal{PR}_{|\Gamma|_{W}}(\Gamma)$, there is some $\Lambda \in \mathcal{PR}_{|\Delta|_{W}}(\Delta)$ such that $\vdash_{\mathsf{GT}^{-}} \Xi \Rightarrow \Lambda$. By repeated applications of Lemma 6.4.6 (i), for each $\Xi \in \mathcal{PR}_{|\Gamma|_{W}}(\Gamma)$, $\vdash_{\mathsf{GT}^{-}} \Xi \Rightarrow \Delta$. By repeated applications of Lemma 6.4.6 (ii), $\vdash_{GT^{-}} \Gamma \Rightarrow \Delta$.

6.5 Weakening, Inversion, Contraction

We mentioned in Section 6.3 that in our setting, right weakening is sound for all formulas, whereas right contraction is only (guaranteed to be) sound for union-closed (and hence classical) formulas. In this section, we further show that (left and right) weakening is height-preserving admissible in GT^- ; that inversion for all rules except R Wis height-preserving admissible; and that left contraction is height-preserving admissible with respect to all formulas, whereas right contraction is only (guaranteed to be) admissible with respect to classical formulas.

Let $\vdash_{\mathsf{GT}^-}^n \Gamma \Rightarrow \Delta$ (or simply $\vdash^n \Gamma \Rightarrow \Delta$) denote the fact that there is a GT^- -derivation of $\Gamma \Rightarrow \Delta$ of height at most *n*.

6.5.1. LEMMA (Height-preserving weakening). Both left and right weakening are height-preserving admissible:

- Left weakening: $\vdash^n \Gamma \Rightarrow \Delta$ implies $\vdash^n \Gamma, \xi \Rightarrow \Delta$.
- *Right weakening*: $\vdash^{n} \Gamma \Rightarrow \Delta$ *implies* $\vdash^{n} \Gamma \Rightarrow \xi, \Delta$.

Proof:

By induction on *n*. For both items, if n = 1, the derivation \mathcal{D} witnessing the antecedent of the item consists of a single axiom, and then the consequent of the item follows by the same axiom. For the inductive step, we assume both items hold for *n* and prove that they hold for n + 1. Letting *R* denote the final rule applied in $\vdash^{n+1} \Gamma \Rightarrow \Delta$, if *R* is not \mathbb{R}^{\wedge} or \mathbb{L}^{\vee} , both items follow by applying the induction hypothesis to the (subderivation(s) ending in the) premise(s) of *R*, and then applying *R*. If *R* is \mathbb{R}^{\wedge} or \mathbb{L}^{\vee} , left weakening follows by the induction hypothesis and *R*, as above, and right weakening follows immediately by the implicit weakening in *R*. For instance, if *R* is \mathbb{R}^{\wedge} , it is of the form:

$$\frac{\Gamma \Rightarrow \varphi, \Lambda \qquad \Gamma \Rightarrow \psi, \Lambda}{\Gamma \Rightarrow \varphi \land \psi, \Lambda, \Delta'} \mathsf{R}$$

Left weakening: By the induction hypothesis, $\vdash^n \Gamma, \xi \Rightarrow \varphi, \Lambda$ and $\vdash^n \Gamma, \xi \Rightarrow \psi, \Lambda$, whence by \mathbb{R}_{Λ} , we have $\vdash^{n+1} \Gamma, \xi \Rightarrow \varphi \land \psi, \Lambda, \Delta'$.

Right weakening: By R \wedge applied to the premises, $\vdash^{n+1} \Gamma \Rightarrow \xi, \varphi \land \psi, \Lambda, \Delta'$. \Box

Height-preserving inversion is admissible for all rules except for $\mathbb{R} \vee$; $\mathbb{R} \vee$ instead has an inversion-like property which corresponds to the Split property (Proposition 6.2.8). The failure of invertibility of $\mathbb{R} \vee$ is as desired: the inverted rule is not sound as, for instance, $p \vee \neg p \Rightarrow p \vee \neg p$ is valid whereas $p \vee \neg p \Rightarrow p$ is not. Observe that for $\mathbb{L} \vee$ and $\mathbb{R} \vee$, the inverted rules, unlike the original rules, have no syntactic restrictions and feature no implicit weakening (whereas the inversion-like property for $\mathbb{R} \vee$ has a syntactic restriction on the left context).

Note that in the following proof we use the fact that if a formula is of the form $\varphi\{\psi_1\}\{\psi_2\}$, this indicates that neither the subformula occurrence ψ_1 replacing \cdot_1 in

 $\varphi\{\cdot_1\}\{\cdot_2\}$ nor the occurrence ψ_2 replacing \cdot_2 is a suboccurrence of the other in the formula $\varphi\{\psi_1\}\{\psi_2\}$. For instance, for $\varphi_1\{\cdot_1\}\{\cdot_2\} = p \wedge \cdot_1 \wedge \cdot_2$ we have $\varphi_1\{q\}\{r\} = p \wedge q \wedge r$, in which neither q nor r is a suboccurrence of the other. We also use the following convention: if we know that ψ_2 is a suboccurrence of ψ_1 in $\varphi\{\psi_1\}$, we write $\varphi\{\psi_1'\{\psi_2\}\}$, where $\psi_1'\{\cdot\}$ is constructed in the obvious way. For instance, for $\varphi\{\cdot\} = p \wedge \cdot$ and $\psi = q \wedge r$, we have that r is a suboccurrence of $\varphi\{\psi\} = p \wedge q \wedge r$, whence we write $\varphi\{\psi_1'\{r\}\}$, where $\psi_1' = q \wedge \cdot$.

6.5.2. LEMMA (Height-preserving inversion).

 $(L\neg) \vdash^{n} \Gamma, \neg \alpha \Rightarrow \Delta \text{ implies} \vdash^{n} \Gamma \Rightarrow \alpha, \Delta;$ $(R\neg) \vdash^{n} \Gamma \Rightarrow \neg \alpha, \Delta \text{ implies} \vdash^{n} \Gamma, \alpha \Rightarrow \Delta;$ $(L\wedge) \vdash^{n} \Gamma, \varphi \land \psi \Rightarrow \Delta \text{ implies} \vdash^{n} \Gamma, \varphi, \psi \Rightarrow \Delta;$ $(R\wedge) \vdash^{n} \Gamma \Rightarrow \varphi \land \psi, \Delta \text{ implies} \vdash^{n} \Gamma \Rightarrow \varphi, \Delta \text{ and } \vdash^{n} \Gamma \Rightarrow \psi, \Delta;$ $(L\vee) \vdash^{n} \Gamma, \varphi \lor \psi \Rightarrow \Delta \text{ implies} \vdash^{n} \Gamma, \varphi \Rightarrow \Delta \text{ and } \vdash^{n} \Gamma, \psi \Rightarrow \Delta;$ $(R\vee) \vdash^{n} \Gamma \Rightarrow \varphi \lor \psi, \Delta \text{ implies} \vdash^{n} \Gamma \Rightarrow \varphi, \psi, \Delta;$ $(L\vee) \vdash^{n} \Gamma, \chi \{\varphi_{L} \lor \varphi_{R}\} \Rightarrow \Delta \text{ implies} \vdash^{n} \Gamma, \chi \{\varphi_{L}\} \Rightarrow \Delta \text{ and } \vdash^{n} \Gamma, \chi \{\varphi_{R}\} \Rightarrow \Delta;$ $(R\vee) \vdash^{n} \Xi \Rightarrow \chi \{\varphi_{L} \lor \varphi_{R}\}, \Delta \text{ implies} \vdash^{n} \Xi \Rightarrow \chi \{\varphi_{L}\}, \Delta \text{ or } \vdash^{n} \Xi \Rightarrow \chi \{\varphi_{R}\}, \Delta.$

Proof:

Each item is proved by induction on *n*. For each item, if n = 1, the derivation \mathcal{D} witnessing the antecedent of the item consists of a single axiom. In all cases, the consequent then follows by the same axiom. We now assume that each item holds for *n* and prove that it holds for n + 1. We separate subcases for each item as follows (note that not all subcases exist for each item):

Case 1: The active formula of the inversion (e.g., the formula $\neg \alpha$ in the item $L \neg$) is not principal in the final rule *R* applied in \mathcal{D} .

Case 1.1: The active formula of the inversion is in the left/right context of each premise of R. In this case, the result follows by the induction hypothesis applied to the (subderivation(s) ending in the) premise(s) of R followed by application(s) of R. For instance, for the item $R\wedge$, if R is also $R\wedge$, it is, in this case, of the form:

$$\frac{\Gamma \Rightarrow \eta, \varphi \land \psi, \Lambda' \qquad \Gamma \Rightarrow \xi, \varphi \land \psi, \Lambda'}{\Gamma \Rightarrow \eta \land \xi, \varphi \land \psi, \Lambda', \Delta'} \mathsf{R}_{\land},$$

where $\varphi \land \psi, \Lambda' = \Lambda$ is the right context of each premise of *R*. Applying the induction hypothesis to the (subderivations ending in the) premises yields $\vdash^n \Gamma \Rightarrow \eta, \varphi, \Lambda'; \vdash^n \Gamma \Rightarrow \eta, \varphi, \Lambda'; \vdash^n \Gamma \Rightarrow \xi, \varphi, \Lambda';$ and $\vdash^n \Gamma \Rightarrow \xi, \psi, \Lambda'$. Then two applications of $\mathbb{R} \land$ yield $\vdash^{n+1} \Gamma \Rightarrow \eta \land \xi, \varphi, \Lambda', \Delta'$ and $\vdash^{n+1} \Gamma \Rightarrow \eta \land \xi, \psi, \Lambda', \Delta'$.

200 Chapter 6. A Deep-Inference Sequent Calculus for a Propositional Team Logic

Case 1.2: The active formula of the inversion is introduced by the implicit weakening in *R*. In this case, the right context of the conclusion of *R* is of the form Λ , φ_a, Δ' , where φ_a is the active formula of the inversion, and the right context of each premise is Λ . The result follows by an application of *R* to the premises with a conclusion right context of the form $\Lambda, \varphi_p, \Delta'$, where φ_p is a principal formula of the inversion. For instance, for the item \mathbb{R}^{\wedge} , if *R* is also \mathbb{R}^{\wedge} , it is, in this case, of the form:

$$\frac{\Gamma \Rightarrow \eta, \Lambda}{\Gamma \Rightarrow \eta, \Lambda} \xrightarrow{\Gamma \Rightarrow \xi, \Lambda} \mathsf{R}$$

By applying R \wedge to the premises, we get $\vdash^{n+1} \Gamma \Rightarrow \eta \land \xi, \Lambda, \varphi, \Delta'$ and $\vdash^{n+1} \Gamma \Rightarrow \eta \land \xi, \Lambda, \psi, \Delta'$.

Case 2: The active formula of the inversion is principal in *R*.

Case 2.1: The active subformula of the inversion (e.g., the formula $\neg \alpha$ in the item L \neg , and the formula $\varphi_L \otimes \varphi_R$ in L \otimes) is the active subformula of *R*. In this case, the (subderivation(s) ending in the) premise(s) of *R* already yield the desired result (if *R* is R \land or L \lor , right weakening must also be applied to the premises). For instance, for the item R \land , *R* must in this case also be R \land , and it is of the form:

$$\frac{\Gamma \Rightarrow \varphi, \Lambda \qquad \Gamma \Rightarrow \psi, \Lambda}{\Gamma \Rightarrow \varphi \land \psi, \Lambda, \Delta'} \mathsf{R} \land$$

We get $\vdash^n \Gamma \Rightarrow \varphi, \Lambda, \Delta'$ and $\vdash^n \Gamma \Rightarrow \psi, \Lambda, \Delta'$ (whence also $\vdash^{n+1} \Gamma \Rightarrow \varphi, \Lambda, \Delta'$ and $\vdash^{n+1} \Gamma \Rightarrow \psi, \Lambda, \Delta'$) by applying height-preserving right weakening (Lemma 6.5.1) to the premises.

Case 2.2: The active subformula of the inversion is not the active subformula of R. In this case, the result follows by the induction hypothesis applied to the premises of R followed by an application (or multiple applications) of R. For instance, for the item $L\lor$, R must in this case be L𝔅, and we may assume without loss of generality that it is of the form:

$$\frac{\Gamma, \varphi\{\varphi_L\} \lor \psi \Rightarrow \Delta}{\Gamma, \varphi\{\varphi_L \lor \varphi_R\} \lor \psi \Rightarrow \Delta} L \bowtie$$

Applying the induction hypothesis to the premises yields $\vdash^n \Gamma, \varphi\{\varphi_L\} \Rightarrow \Delta; \vdash^n \Gamma, \psi \Rightarrow \Delta$; and $\vdash^n \Gamma, \varphi\{\varphi_R\} \Rightarrow \Delta$. Then we already have $\vdash^{n+1} \Gamma, \psi \Rightarrow \Delta$, and an application of $L \lor$ yields $\vdash^{n+1} \Gamma, \varphi\{\varphi_L \lor \varphi_R\} \Rightarrow \Delta$.

For the items L \otimes and R \otimes , the subcases of Case 2.2 in which *R* is one of the deepinference rules follow essentially as above, but this is less easy to see due to the complicated structure of the formulas involved, and there are some minor additional complications. We now provide some further details on these subcases. In each of these subcases, the principal formula of both the inversion and *R* is $\chi{\{\varphi_L \otimes \varphi_R\}}$, and the principal subformula $\varphi_L \otimes \varphi_R$ of the inversion is not the principal subformula $\xi_L \otimes \xi_R$ of *R*. The formula $\chi{\{\varphi_L \otimes \varphi_R\}}$ is then of one of the following forms: (i) $\chi{\{\varphi_L \otimes \varphi_R\}}{\{\xi_L \otimes \xi_R\}}$; (ii) $\chi{\{(\varphi_L \otimes \varphi_R)'\{\xi_L \otimes \xi_R\}\}}$; or (iii) $\eta{\{(\xi_L \otimes \xi_R)'\{\varphi_L \otimes \varphi_R\}}$.

Case 2.2 of item LW; *R* is LW. In all of the subcases (i), (ii), (iii), the result follows essentially, as above, by applying the induction hypothesis to the premises and then

applying *R*. We provide the details for (iii). We may assume without loss of generality that $\eta\{(\xi_L \otimes \xi_R)'\{\varphi_L \otimes \varphi_R\}\} = \eta\{\xi'_L\{\varphi_L \otimes \varphi_R\} \otimes \xi_R\}$. Then *R* is of the form:

$$\frac{\Gamma, \eta\{\xi'_L\{\varphi_L \lor \varphi_R\}\} \Rightarrow \Delta}{\Gamma, \eta\{\xi'_L\{\varphi_L \lor \varphi_R\} \lor \xi_R\} \Rightarrow \Delta} L \bowtie$$

By the induction hypothesis applied to the left premise, $\vdash^n \Gamma, \eta\{\xi'_L\{\varphi_L\}\} \Rightarrow \Delta$ and $\vdash^n \Gamma, \eta\{\xi'_L\{\varphi_R\}\} \Rightarrow \Delta$, and from the right premise we have $\vdash^n \Gamma, \eta\{\xi_R\} \Rightarrow \Delta$. Then by LW, we have $\vdash^{n+1} \Gamma, \eta\{\xi'_L\{\varphi_L\} \otimes \xi_R\} \Rightarrow \Delta$ (i.e., $\vdash^{n+1} \Gamma, \chi\{\varphi_L\} \Rightarrow \Delta$) and $\vdash^{n+1} \Gamma, \eta\{\xi'_L\{\varphi_R\} \otimes \xi_R\} \Rightarrow \Delta$ (i.e., $\vdash^{n+1} \Gamma, \chi\{\varphi_R\} \Rightarrow \Delta$).

Case 2.2 of item $\mathbb{R} \otimes \mathbb{R}$ is $\mathbb{R} \otimes \mathbb{R}$. In all of the subcases (i), (ii), (iii), the result follows essentially, as above, by applying the induction hypothesis to the premises and then applying *R*. We provide the details for (iii). We may assume without loss of generality that $\eta\{(\xi_L \otimes \xi_R)'\{\varphi_L \otimes \varphi_R\}\} = \eta\{\xi'_L\{\varphi_L \otimes \varphi_R\} \otimes \xi_R\}$. There are two subcases. In the first, *R* is of the form:

$$\frac{\Xi \Rightarrow \eta\{\xi_L'\{\varphi_L \lor \varphi_R\}\}, \Delta}{\Xi \Rightarrow \eta\{\xi_L'\{\varphi_L \lor \varphi_R\} \lor \xi_R\}, \Delta} \mathsf{R} \lor$$

By the induction hypothesis applied to the premise, we have $\vdash^n \Xi \Rightarrow \eta\{\xi'_L\{\varphi_L\}\}, \Delta$ or $\vdash^n \Xi \Rightarrow \eta\{\xi'_L\{\varphi_R\}\}, \Delta$. If the former, then $\vdash^{n+1} \Xi \Rightarrow \eta\{\xi'_L\{\varphi_L\} \lor \xi_R\}, \Delta$ (i.e., $\vdash^{n+1} \Xi \Rightarrow \chi\{\varphi_L\}, \Delta$) by $\mathbb{R} \lor$; if the latter, then $\vdash^{n+1} \Xi \Rightarrow \eta\{\xi'_L\{\varphi_R\} \lor \xi_R\}, \Delta$ (i.e., $\vdash^{n+1} \Xi \Rightarrow \chi\{\varphi_R\}, \Delta$) by $\mathbb{R} \lor$.

In the second subcase, R is of the form

$$\frac{\Xi \Rightarrow \eta\{\xi_R\}, \Delta}{\Xi \Rightarrow \eta\{\xi_L'\{\varphi_L \lor \varphi_R\} \lor \xi_R\}, \Delta} \mathsf{Rw}$$

Then by $\mathbb{R} \otimes \mathbb{V}$ applied to the premise, $\vdash^{n+1} \Xi \Rightarrow \eta \{\xi'_L \{\varphi_L\} \otimes \xi_R\}, \Delta$ (i.e., $\vdash^{n+1} \Xi \Rightarrow \chi \{\varphi_L\}, \Delta$).

We have concluded the proof, but it is worth making more explicit why the induction goes through for item $\mathbb{R} \vee$ in the cases in which *R* has two premises. Note first that *R* cannot be $\mathbb{L} \vee$ since the left context of the conclusion of *R* must be classical (in order to serve as the antecedent for item $\mathbb{R} \vee$). So *R* must be $\mathbb{R} \wedge$ or $\mathbb{L} \vee$. If the principal formula $\chi \{\varphi_L \vee \varphi_R\}$ of the inversion is a side formula of *R*, it must then have been introduced via the implicit weakening in *R*, and we proceed as in Case 1.2 (instead of introducing $\chi \{\varphi_L \vee \varphi_R\}$, we introduce $\chi \{\varphi_L\}$ or $\chi \{\varphi_R\}$). If, on the other hand, $\chi \{\varphi_L \vee \varphi_R\}$ is an active formula of *R*, *R* must be $\mathbb{R} \wedge$, and we may assume without loss of generality that $\chi \{\varphi_L \vee \varphi_R\} = \chi_1 \{\varphi_L \vee \varphi_R\} \wedge \chi_2$ and that *R* is of the form:

$$\frac{\Xi \Rightarrow \chi_1\{\varphi_L \lor \varphi_R\}, \Lambda}{\Xi \Rightarrow \chi_1\{\varphi_L \lor \varphi_R\} \land \chi_2, \Lambda, \Delta'} \xrightarrow{\Xi \Rightarrow \chi_2, \Lambda} \mathsf{R} \land,$$

We proceed essentially as in Case 2.2, by first applying the induction hypothesis to the left premise and then applying R.

6.5.3. LEMMA (Height-preserving contraction). Left contraction is height-preserving admissible, and right contraction is height-preserving admissible with respect to classical formulas:

- Left contraction: $\vdash^{n} \Gamma, \xi, \xi \Rightarrow \Delta$ implies $\vdash^{n} \Gamma, \xi \Rightarrow \Delta$.
- *Right contraction*: $\vdash^{n} \Gamma \Rightarrow \alpha, \alpha, \Delta$ *implies* $\vdash^{n} \Gamma \Rightarrow \alpha, \Delta$.

Proof:

By simultaneous induction on *n*. For both items, if n = 1, the derivation \mathcal{D} witnessing the antecedent of the item consists of a single axiom, and then the consequent of the item follows by the same axiom. We now assume that both items hold for *n* and prove that they hold for n + 1. Let *R* denote the final rule applied in \mathcal{D} . We separate subcases for each item as follows (note that not all subcases exist for each item):

Case 1: The formula (occurrence) to be contracted is not the principal formula of R. In this case, the result follows by the induction hypothesis applied to the (subderivation(s) ending in the) premise(s) of R followed by an application of R. For instance, for right contraction, if R is $L \neg$, it is of the form:

$$\frac{\Gamma' \Rightarrow \beta, \alpha, \alpha, \Delta}{\Gamma', \neg \beta \Rightarrow \alpha, \alpha, \Delta} L \neg$$

By the induction hypothesis applied to the premise followed by an application of L¬, we have $\vdash^{n+1} \Gamma', \neg \beta \Rightarrow \alpha, \Delta$.

For right contraction, when *R* is $L \vee$ or $R \wedge$, Case 1 has two subcases.

Case 1.1: Both of the formula occurrences α to be contracted are in the right context of each premise of *R*. The result follows as in Case 1. For instance, if *R* is $R \wedge$, it is of the form:

$$\frac{\Gamma \Rightarrow \alpha, \alpha, \varphi, \Lambda \qquad \Gamma \Rightarrow \alpha, \alpha, \psi, \Lambda}{\Gamma \Rightarrow \alpha, \alpha, \varphi \land \psi, \Lambda, \Delta'} \mathsf{R}^{\wedge}$$

By the induction hypothesis applied to the premises followed by an application of \mathbb{R}_{\wedge} , we have $\vdash^{n+1} \Gamma \Rightarrow \alpha, \varphi \land \psi, \Lambda, \Delta'$.

Case 1.2: At least one of the formula occurrences α to be contracted is introduced by the implicit weakening in *R*. In this case, we replace the application of *R* by another application in which one fewer occurrence is introduced. For instance, if *R* is $R \wedge$, it is of one of the following forms:

$$\frac{\Gamma \Rightarrow \varphi, \Lambda \qquad \Gamma \Rightarrow \psi, \Lambda}{\Gamma \Rightarrow \alpha, \alpha, \varphi \land \psi, \Lambda, \Delta'} \mathsf{R} \land \qquad \frac{\Gamma \Rightarrow \alpha, \varphi, \Lambda \qquad \Gamma \Rightarrow \alpha, \psi, \Lambda}{\Gamma \Rightarrow \alpha, \alpha, \varphi \land \psi, \Lambda, \Delta'} \mathsf{R} \land$$

In either case, by \mathbb{R}_{\wedge} applied to the premises, $\vdash^{n+1} \Gamma \Rightarrow \alpha, \varphi \land \psi, \Lambda, \Delta'$.

Case 2: The formula occurrence to be contracted is the principal formula of R. In this case, the result follows by application(s) of the inversion lemma to the (subderivations ending in the) premise(s), followed by application(s) of the induction hypothesis (for one of the two items), followed by an application of R. We give three examples.

6.6. Countermodel Construction

For left contraction, if *R* is $L \neg$, we have that $\xi = \neg \beta$ and *R* is of the form:

$$\frac{\Gamma,\neg\beta\Rightarrow\beta,\Delta}{\Gamma,\neg\beta,\neg\beta\Rightarrow\Delta} \mathsf{L}^{-1}$$

By the L¬-case of Lemma 6.5.2 applied to the premise, $\vdash^n \Gamma \Rightarrow \beta, \beta, \Delta$. Then by the induction hypothesis (for right contraction), we have $\vdash^n \Gamma \Rightarrow \beta, \Delta$, whence by L¬, we have $\vdash^{n+1} \Gamma, \neg \beta \Rightarrow \Delta$.

For right contraction, if *R* is $R \lor$, we have that $\alpha = \beta_1 \lor \beta_2$ and *R* is of the form:

$$\frac{\Gamma \Rightarrow \beta_1, \beta_2, \beta_1 \lor \beta_2, \Delta}{\Gamma \Rightarrow \beta_1 \lor \beta_2, \beta_1 \lor \beta_2, \Delta} \mathsf{R} \lor$$

By the R \lor -case of Lemma 6.5.2 applied to the premise, $\vdash^n \Gamma \Rightarrow \beta_1, \beta_2, \beta_1, \beta_2, \Delta$. Then by the induction hypothesis, we obtain first $\vdash^n \Gamma \Rightarrow \beta_1, \beta_1, \beta_2, \Delta$ and then $\vdash^n \Gamma \Rightarrow \beta_1, \beta_2, \Delta$, whence by R \lor , we have $\vdash^{n+1} \Gamma \Rightarrow \beta_1 \lor \beta_2, \Delta$.

For left contraction, if *R* is L \mathbb{W} , we have that $\xi = \chi \{\varphi_L \otimes \varphi_R\}$ and *R* is of the form:

$$\frac{\Gamma, \chi\{\varphi_L \lor \varphi_R\}, \chi\{\varphi_L\} \Rightarrow \Delta \qquad \Gamma, \chi\{\varphi_L \lor \varphi_R\}, \chi\{\varphi_R\} \Rightarrow \Delta}{\Gamma, \chi\{\varphi_L \lor \varphi_R\}, \chi\{\varphi_L \lor \varphi_R\}, \chi\{\varphi_L \lor \varphi_R\} \Rightarrow \Delta} \sqcup \lor$$

By the L \otimes -case of Lemma 6.5.2 applied to the premises, $\vdash^n \Gamma, \chi\{\varphi_L\}, \chi\{\varphi_L\} \Rightarrow \Delta$ and $\vdash^n \Gamma \Rightarrow \chi\{\varphi_R\}, \chi\{\varphi_R\} \Rightarrow \Delta$. Then by the induction hypothesis, $\vdash^n \Gamma, \chi\{\varphi_L\} \Rightarrow \Delta$ and $\vdash^n \Gamma, \chi\{\varphi_R\} \Rightarrow \Delta$, whence by L \otimes , we have $\vdash^{n+1} \Gamma, \chi\{\varphi_L \otimes \varphi_R\} \Rightarrow \Delta$.

Note that for right contraction, if *R* is $\mathbb{R} \lor$, there is no Case 2 since the principal formula of *R* is not classical and hence cannot be the target formula occurrence of the contraction.

6.6 Countermodel Construction

Due to the (semantic) invertibility of the rules, together with the Split property (Proposition 6.2.8), there is a simple way of proving the cutfree completeness of GT that is similar to one for G3cp [118, pp. 104–105]: roughly, given a sequent S, one applies inverted versions of the rules until one arrives at *atomic sequents*—sequents consisting solely of propositional variables and \perp . Given semantic invertibility and Split, if S is valid, so are these atomic sequents, whence they are axioms; one will then have constructed a derivation of S consisting of these axioms and the non-inverted versions of the inverted rules applied in the process. Conversely, if S is invalid, some of these atomic sequents must also be invalid, and these invalid atomic sequents then enable one to construct a countermodel for S.

6.6.1. LEMMA.

- There is an effective procedure for constructing a countermodel to the conclusion of any rule apart from R∧, L∨ and Rℕ from a countermodel to any premise of that rule.

- 204 Chapter 6. A Deep-Inference Sequent Calculus for a Propositional Team Logic
 - $\mathbb{R}\wedge$: There is an effective procedure for constructing a countermodel to $\Gamma \Rightarrow \varphi \wedge \psi, \Lambda$ from a countermodel to either $\Gamma \Rightarrow \varphi, \Lambda$ or $\Gamma \Rightarrow \psi, \Lambda$.
 - L \lor : There is an effective procedure for constructing a countermodel to $\Gamma, \phi \lor \psi \Rightarrow \Lambda$ from a countermodel to either $\Gamma, \phi \Rightarrow \Lambda$ or $\Gamma, \psi \Rightarrow \Lambda$.
 - $\mathbb{R} \otimes :$ There is an effective procedure for constructing a countermodel to $\Xi \Rightarrow \chi\{\psi_L \otimes \psi_R\}$ given countermodels to both $\Xi \Rightarrow \chi\{\psi_L\}, \Delta$ and $\Xi \Rightarrow \chi\{\psi_R\}, \Delta$.

Proof:

- L¬: Let $t \models \wedge \Gamma$ and $t \not\models \alpha \lor \lor \Delta$. Let $t_{\neg \alpha} \coloneqq \{v \in t \mid v \models \neg \alpha\}$; this is our countermodel. By downward closure, $t_{\neg \alpha} \models \wedge \Gamma \land \neg \alpha$. And if $t_{\neg \alpha} \models \lor \Delta$, then $t \models \alpha \lor \lor \Delta$, a contradiction; therefore $t_{\neg \alpha} \not\models \lor \Delta$.
- $\mathbb{R} \neg$: Let $t \models \bigwedge \Gamma \land \alpha$ and $t \not\models \bigvee \Delta$. *t* is our countermodel. Clearly $t \models \bigwedge \Gamma$. Assume for contradiction that $t \models \neg \alpha \lor \bigvee \Delta$. Then $t = t_{\neg \alpha} \cup t_{\Delta}$ where $t_{\neg \alpha} \models \neg \alpha$ and $t_{\Delta} \models \bigvee \Delta$. But then by the flatness of α and $\neg \alpha$, we have that for each $v \in t_{\neg \alpha}$, $v \models \alpha \land \neg \alpha$; therefore $t_{\neg \alpha} = \emptyset$, whence $t = t_{\Delta} \models \bigvee \Delta$, a contradiction. Therefore, $t \not\models \neg \alpha \lor \lor \Delta$.
- LA: Immediate from the countermodel to the premise.
- R \land : Assume $t \models \land \Gamma$ and $t \not\models \varphi \lor \lor \land \Lambda$. *t* is our countermodel: clearly $t \not\models (\varphi \land \psi) \lor \lor \land \Lambda$. Similarly for the other premise.
- L \lor : Assume $t \models \land \land \land \varphi$ and $t \not\models \lor \land \land t$ is our countermodel: by the empty team property, $t \models \land \land \land \land (\varphi \lor \psi)$. Similarly for the other premise.
- R_{\vee} : Immediate from the countermodel to the premise.
- LW: Assume $t \models \land \Gamma \land \chi \{ \varphi_L \}$ and $t \not\models \lor \Delta$. *t* is our countermodel: clearly $t \models \land \Gamma \land \chi \{ \varphi_L \lor \varphi_R \}$. Similarly for the other premise.
- R_W: Assume $t_1 \models \land \Xi$ and $t_1 \not\models \chi\{\varphi_L\}, \Delta$ and $t_2 \models \land \Xi$ and $t_2 \not\models \chi\{\varphi_R\}, \Delta$. $t_1 \cup t_2$ is our countermodel. By the union closure of $\land \Xi, t_1 \cup t_2 \models \land \Xi$. By downward closure, $t_1 \cup t_2 \not\models \chi\{\varphi_L\}, \Delta$ and $t_1 \cup t_2 \not\models \chi\{\varphi_R\}, \Delta$ so $t_1 \cup t_2 \not\models \chi\{\varphi_L\}, \Delta$. \Box

We introduce some metalanguage notation for convenience. We use \wedge as a metalanguage 'and', and \vee as a metalanguage 'or'. We call a collection of sequents joined together by the metalanguage connectives a *Boolean combinations of sequents*. For instance, $\Gamma \Rightarrow \varphi, \Lambda \land \Gamma \Rightarrow \psi, \Lambda$ is a Boolean combination of sequents. We will use these combinations to express derivability and semantic invertibility facts in the following manner:

$$\mathsf{R} \land \ (\Gamma \Rightarrow \varphi, \Lambda \land \Gamma \Rightarrow \psi, \Lambda) = \vdash_{\mathsf{R} \land} \Gamma \Rightarrow \varphi \land \psi, \Lambda;$$

$$\mathsf{R} \lor (\Xi \Rightarrow \chi\{\varphi_L\}, \Delta \lor \Xi \Rightarrow \chi\{\varphi_R\}, \Delta) \preccurlyeq \vdash_{\mathsf{R} \lor \mathsf{V}} \Xi \Rightarrow \chi\{\varphi_L \lor \varphi_R\}, \Delta.$$

The left-to-right direction of the first line above reformulates the rule $\mathbb{R}\wedge$ (without implicit weakening): given both of the sequents $\Gamma \Rightarrow \varphi, \Lambda$ and $\Gamma \Rightarrow \psi, \Lambda$, one can derive the sequent $\Gamma \Rightarrow \varphi \land \psi, \Lambda$ using only $\mathbb{R}\wedge$. The right-to-left direction expresses the semantic invertibility of $\mathbb{R}\wedge$ (restricted to classical right contexts): if $\Gamma \Rightarrow \varphi \land \psi, \Lambda$ is valid, then both $\Gamma \Rightarrow \varphi, \Lambda$ and $\Gamma \Rightarrow \psi, \Lambda$ are valid. Similarly, the second line expresses that one can derive $\Xi \Rightarrow \chi\{\varphi_L \lor \varphi_R\}, \Delta$ via $\mathbb{R} \lor$ using either $\Xi \Rightarrow \chi\{\varphi_L\}, \Delta$ or $\Xi \Rightarrow \chi\{\varphi_R\}, \Delta$, and that if $\Xi \Rightarrow \chi\{\varphi_L \lor \varphi_R\}, \Delta$ is valid, so is either $\Xi \Rightarrow \chi\{\varphi_L\}, \Delta$ or $\Xi \Rightarrow \chi\{\varphi_R\}, \Delta$.

We say that a Boolean combination S of sequents is *valid* if, replacing each valid sequent in S with 1 and each invalid sequent with 0, the Boolean function determined by the metalanguage connectives in S outputs 1. We write $S_1 \models S_2$ to denote the fact that the validity of S_1 implies the validity of S_2 . We say that a subcombination of S is a witnessing combination for S if the validity of the subcombination entails the validity of S (e.g., S_1 is a witnessing combination for $S_1 \ \forall S_2$). We write $S_1 \vdash_C S_2$ to denote the fact that there is some witnessing combination for S_1 such that given derivation(s) in GT^- for the sequents in S_1 , one can extend these derivations using rules in C to construct derivation(s) of a witnessing combination for S_2 . We write $S_1 C_2 \dashv \vdash_{C_1} S_2$ if $S_1 \vdash_{C_1} S_2$ and $S_2 \vdash_{C_2} S_1$, and $S_1 \dashv \vdash_{C} S_2$ if $S_1 \vdash_{C_1} S_2$ and $S_2 \vDash S_1$. We write simply \vdash for \vdash_{GT^-} . We write $\vdash S$ if there are derivations for the sequents in a witnessing combination for S (from axioms) in GT⁻. We say that a derivation Dwitnesses $\vdash_{C_1} S_1 \vdash_{C_2} S_2 \vdash_{C_3} \ldots \vdash_{C_n} S_n$ if \mathcal{D} consists of derivations of the sequents in a witnessing combination for S_1 using rules in C_1 , extensions of these derivations using rules in C₂ such that the combined derivations constitute derivations of the sequents in a witnessing combination for S_2 , and so on (\mathcal{D} witnessing $S_1 \vdash_{C_2} S_2 \vdash_{C_3} \ldots \vdash_{C_n} S_n$ is defined in a similar way).

The following list of results (given, for instance, by the rules, by Lemma 6.5.2, and by soundness) summarizes the facts we make use of in our proof of the cutfree completeness/countermodel construction theorem (Theorem 6.6.2):

$$\begin{split} \mathsf{L}\neg \ \Gamma &\Rightarrow \alpha, \Delta \dashv \vdash_{\mathsf{L}\neg} \Gamma, \neg \alpha \Rightarrow \Delta; \\ \mathsf{R}\neg \ \Gamma, \alpha \Rightarrow \Delta \dashv \vdash_{\mathsf{R}\neg} \Gamma \Rightarrow \neg \alpha, \Delta; \\ \mathsf{L}\land \ \Gamma, \varphi, \psi \Rightarrow \Delta \dashv \vdash_{\mathsf{L}\land} \Gamma, \varphi \land \psi \Rightarrow \Delta; \\ \mathsf{R}\land \ (\Gamma \Rightarrow \varphi, \Lambda \land \Gamma \Rightarrow \psi, \Lambda) \dashv \vdash_{\mathsf{R}\land} \Gamma \Rightarrow \varphi \land \psi, \Lambda; \\ \mathsf{L}\lor \ (\Gamma, \varphi \Rightarrow \Lambda \land \Gamma, \psi \Rightarrow \Lambda) \dashv \vdash_{\mathsf{L}\lor} \Gamma, \varphi \lor \psi \Rightarrow \Lambda; \\ \mathsf{R}\lor \ \Gamma \Rightarrow \varphi, \psi, \Delta \dashv \vdash_{\mathsf{R}\lor} \Gamma \Rightarrow \varphi \lor \psi, \Delta; \\ \mathsf{L} \lor \ (\Gamma, \chi\{\varphi_L\} \Rightarrow \Delta \land \Gamma, \chi\{\varphi_L\} \Rightarrow \Delta) \dashv \vdash_{\mathsf{L}\lor} \Gamma, \chi\{\varphi_L \lor \varphi_R\} \Rightarrow \Delta; \\ \mathsf{R} \lor \ (\Xi \Rightarrow \chi\{\varphi_L\}, \Delta \lor \Xi \Rightarrow \chi\{\varphi_R\}, \Delta) \dashv \vdash_{\mathsf{R}\lor\lor} \Xi \Rightarrow \chi\{\varphi_L \lor \varphi_R\}, \Delta. \end{split}$$

6.6.2. THEOREM (Cutfree completeness and countermodel construction). *There is an effective procedure that, given a sequent* $\Gamma \Rightarrow \Delta$ *, yields a cutfree derivation of* $\Gamma \Rightarrow \Delta$ *if* $\Gamma \Rightarrow \Delta$ *is valid, and yields a countermodel to* $\Gamma \Rightarrow \Delta$ *if* $\Gamma \Rightarrow \Delta$ *is not valid.*

Proof:

By applying $L \otimes$ in reverse, and by the semantic invertibility of $L \otimes$, we can find $\Xi_1; \ldots; \Xi_n$ such that

$$\bigwedge_{1\leq i\leq n} (\Xi_i \Rightarrow \Delta) = \vdash_{\mathsf{LW}} \Gamma \Rightarrow \Delta.$$

By applying $\mathbb{R} \otimes \mathbb{V}$ in reverse, and by the Split property (Proposition 6.2.8), we can find $\Lambda_{11}; \ldots; \Lambda_{nm_n}$ such that

$$\bigwedge_{1\leq i\leq n}\bigvee_{1\leq j\leq m_i}(\Xi_i\Rightarrow\Lambda_{ij})=\vdash_{\mathsf{RW}}\bigwedge_{1\leq i\leq n}(\Xi_i\Rightarrow\Delta).$$

Finally, by applying rules in G3cp⁻ in reverse, and by the semantic invertibility of G3cp⁻, we can find atomic sequents $\Xi_{111} \Rightarrow \Lambda_{111}; \ldots; \Xi_{nm_nq_{nm_n}} \Rightarrow \Lambda_{nm_nq_{nm_n}}$ such that

$$\bigwedge_{1 \le i \le n} \bigvee_{1 \le j \le m_i} \bigwedge_{1 \le k \le q_{ij}} (\Xi_{ijk} \Rightarrow \Lambda_{ijk}) \dashv_{\mathsf{G3cp}^-} \bigwedge_{1 \le i \le n} \bigvee_{1 \le j \le m_i} (\Xi_i \Rightarrow \Lambda_{ij}).$$

Therefore,

$$\bigwedge_{1\leq i\leq n} \bigvee_{1\leq j\leq m_i} \bigwedge_{1\leq k\leq q_{ij}} (\Xi_{ijk} \Rightarrow \Lambda_{ijk}) = \vdash \Gamma \Rightarrow \Delta.$$

Then if $\Gamma \Rightarrow \Delta$ is valid, so is

$$\bigwedge_{1\leq i\leq n} \bigvee_{1\leq j\leq m_i} \bigwedge_{1\leq k\leq q_{ij}} (\Xi_{ijk} \Rightarrow \Lambda_{ijk}),$$

whence we must have

$$\vdash \bigwedge_{1 \leq i \leq n} \bigvee_{1 \leq j \leq m_i} \bigwedge_{1 \leq k \leq q_{ij}} (\Xi_{ijk} \Rightarrow \Lambda_{ijk})$$

by the axioms, whence $\vdash \Gamma \Rightarrow \Delta$. Conversely, if $\Gamma \Rightarrow \Delta$ is invalid, then by the soundness of GT^- , we also have that

$$\bigwedge_{1\leq i\leq n}\bigvee_{1\leq j\leq m_i}\bigwedge_{1\leq k\leq q_{ij}}(\Xi_{ijk}\Rightarrow\Lambda_{ijk})$$

is invalid. Since the sequents in this combination are atomic, it is clear how to find countermodels for the invalid sequents in this combination; Lemma 6.6.1 then yields a procedure for constructing a countermodel to $\Gamma \Rightarrow \Delta$ from these countermodels.

Let us consider an example. We write

$$\frac{\Gamma_1 \Rightarrow \Delta_1 \qquad \Gamma_2 \Rightarrow \Delta_2}{\Gamma_3 \Rightarrow \Delta_3} R$$

to denote $(\Gamma_1 \Rightarrow \Delta_1 \land \land \Gamma_2 \Rightarrow \Delta_2) = \vdash_R \Gamma_3 \Rightarrow \Delta_3$ for all rules *R* other than $\mathbb{R} \lor$ (and similarly for single-premise rules), and we write

$$\frac{\Xi \Rightarrow \chi\{\varphi_L\}, \Delta \qquad \Xi \Rightarrow \chi\{\varphi_R\}, \Delta}{\Xi \Rightarrow \chi\{\varphi_L \lor \varphi_R\}, \Delta} R \lor$$

to denote $(\Xi \Rightarrow \chi \{\varphi_L\}, \Delta \le \Xi \Rightarrow \chi \{\varphi_R\}, \Delta) = \vdash_{L \otimes V} \Xi \Rightarrow \chi \{\varphi_L \lor \varphi_R\}, \Delta$. Valid sequents are blue; invalid sequents are red; countermodels are written above the sequent arrows. The following demonstrates how the procedure yields a countermodel $\{v_p, v_{\overline{p}}\}$ to the invalid sequent $p \lor (p \lor \neg p) \Rightarrow p \lor \neg p$.

Or, in our inline notation,

$$((p \Rightarrow p \lor p, p \Rightarrow) \land ((p \Rightarrow p \land p, p) \lor (p, p \Rightarrow \land p \Rightarrow p))$$

$$\exists \vdash_{G3cp^{-}} ((p \Rightarrow p \lor p \Rightarrow \neg p) \land (p \lor \neg p \Rightarrow p \lor p \lor \neg p \Rightarrow \neg p))$$

$$\exists \vdash_{RW} (p \Rightarrow p \lor \neg p \land p \lor \neg p \Rightarrow p \lor \neg p)$$

$$\exists \vdash_{LW} p \lor (p \lor \neg p) \Rightarrow p \lor \neg p$$

6.7 Cut Elimination and Derivation Normal Form

In this section, we prove a normal form result for cutfree derivations and use this normal form result to describe how the cut elimination procedure for G3cp yields one for GT.

Let us introduce some standard terminology. The *level* of a cut (i.e., an application of Cut) is the sum of the heights of the deductions of its premises. The *rank* of a cut on φ is $|\varphi|$, the number of symbols in φ . The *cutrank* $cr(\mathcal{D})$ of a deduction \mathcal{D} is the maximum of the ranks of the cutformulas occurring in \mathcal{D} .

The standard approach to cut elimination fails due to the syntactic restrictions on the rules. Consider, for instance, the following case in a prospective standard cut elimination proof: the rightmost cut of maximal rank and maximal level among cuts of the same rank is such that neither premise is an axiom; the cutformula is not principal on the left; the final rule applied to get the left premise is L_{\vee} . In the setting of G3cp, this

cut would look as follows:

$$\frac{\begin{array}{ccc}
\mathcal{D}_{1} & \mathcal{D}_{2} \\
\underline{\Xi, \beta_{1} \Rightarrow \alpha, \Lambda} & \underline{\Xi, \beta_{2} \Rightarrow \alpha, \Lambda} \\
\underline{\Xi, \beta_{1} \lor \beta_{2} \Rightarrow \alpha, \Lambda} & \mathsf{L} \lor & \begin{array}{c}
\mathcal{D}_{3} \\
\underline{\Theta, \alpha \Rightarrow \Omega} \\
\underline{\Xi, \Theta, \beta_{1} \lor \beta_{2} \Rightarrow \Lambda, \Omega} \\
\end{array}} Cut$$

In the cut elimination procedure for G3cp, we would commute the cut upwards to obtain a derivation with the same cutrank as before ($|\alpha|$), but with either a lower maximal level among cuts of rank $|\alpha|$, or one fewer cut of maximal level of rank $|\alpha|$:

$$\frac{\begin{array}{ccc}
\mathcal{D}_{1} & \mathcal{D}_{3} & \mathcal{D}_{2} & \mathcal{D}_{3} \\
\underline{\Xi,\beta_{1} \Rightarrow \alpha,\Lambda} & \Theta,\alpha \Rightarrow \Omega \\
\underline{\Xi,\Theta,\beta_{1} \Rightarrow \Lambda,\Omega} & \mathsf{Cut} & \underline{\Xi,\beta_{2} \Rightarrow \alpha,\Lambda} & \Theta,\alpha \Rightarrow \Omega \\
\underline{\Xi,\Theta,\beta_{1} \Rightarrow \Lambda,\Omega} & \mathsf{Lv}
\end{array}} \mathsf{Cut}$$

In the general setting of GT, the cut would look as follows (in the case in which the cutformula α is not part of the right context set Δ generated by the implicit weakening):

$$\frac{\begin{array}{c} \mathcal{D}_{1} & \mathcal{D}_{2} \\ \hline \Gamma, \psi_{1} \Rightarrow \alpha, \Lambda & \Gamma, \psi_{2} \Rightarrow \alpha, \Lambda \\ \hline \hline \frac{\Gamma, \psi_{1} \lor \psi_{2} \Rightarrow \alpha, \Lambda, \Delta}{\Gamma, \Pi, \psi_{1} \lor \psi_{2} \Rightarrow \Lambda, \Delta, \Sigma} L \lor & \mathcal{D}_{3} \\ \hline \Gamma, \Pi, \psi_{1} \lor \psi_{2} \Rightarrow \Lambda, \Delta, \Sigma \end{array} Cut$$

We cannot freely commute the cut upwards because Σ might not be classical. If we tried to do so, we would end up with:

$$\frac{\begin{array}{ccc} \mathcal{D}_{1} & \mathcal{D}_{3} & \mathcal{D}_{2} & \mathcal{D}_{3} \\ \hline \Gamma, \psi_{1} \Rightarrow \alpha, \Lambda & \Pi, \alpha \Rightarrow \Sigma \\ \hline \frac{\Gamma, \Pi, \psi_{1} \Rightarrow \Lambda, \Sigma}{\Gamma, \Pi, \psi_{1} \Rightarrow \Lambda, \Sigma} \operatorname{Cut} & \frac{\Gamma, \psi_{2} \Rightarrow \alpha, \Lambda & \Pi, \alpha \Rightarrow \Sigma}{\Gamma, \Pi, \psi_{2} \Rightarrow \Lambda, \Sigma} \operatorname{Cut} \\ \hline \end{array}$$

Here the application of L \vee is not legitimate because the right context set Λ, Σ of the premises might not be classical.

We show instead that each cut can be transformed into a cut on classical sequents. This follows from a derivation normal form theorem for the cutfree system: any cutfree derivation of $\Gamma \Rightarrow \Delta$ can be transformed into a cutfree derivation in which one first derives classical sequents whose antecedents are the resolutions of Γ and whose succedents are resolutions of Δ , and then applies the deep-inference rules to derive $\Gamma \Rightarrow \Delta$. (This is type of theorem is also known as a *decomposition theorem* in the deep-inference literature [27].) Before stating the theorem, we give an example. The theorem transforms the following derivation

into the following:

$$\frac{\mathcal{D}_{1}}{x,\neg x \vee (\neg q \vee p), q \Rightarrow p} \underset{x,\neg x \vee (\neg q \vee p) \Rightarrow p, \neg q}{\frac{x,\neg x \vee (\neg q \vee p) \Rightarrow p, \neg q}{x,\neg x \vee (\neg q \vee p) \Rightarrow p \vee \neg q}} \underset{k \vee}{\mathsf{R}_{\vee}} \qquad \frac{\mathcal{D}_{2}}{\frac{x,\neg x \vee (\neg q \vee r), q \Rightarrow r}{x,\neg x \vee (\neg q \vee r), q \Rightarrow r}}{\frac{x,\neg x \vee (\neg q \vee r), q \Rightarrow r}{x,\neg x \vee (\neg q \vee r) \Rightarrow r, \neg q}} \underset{k \vee}{\mathsf{R}_{\vee}} \\ \frac{\frac{x,\neg x \vee (\neg q \vee r), q \Rightarrow r}{x,\neg x \vee (\neg q \vee r) \Rightarrow r \vee \neg q}}{\frac{x,\neg x \vee (\neg q \vee r), q \Rightarrow r \vee \neg q}{x,\neg x \vee (\neg q \vee r) \Rightarrow r \vee \neg q}} \underset{k \vee (\neg x \vee (\neg q \vee r))) \Rightarrow (p \vee r) \vee \neg q}{x \wedge (\neg x \vee (\neg q \vee r))) \Rightarrow (p \vee r) \vee \neg q}} \underset{k \vee (\neg x \vee (\neg q \vee r))) \Rightarrow (p \vee r) \vee \neg q}{\mathsf{R}_{\vee}} \\ \frac{\mathcal{D}_{1}}{x,\neg x \vee (\neg q \vee r), q \to r} \underset{k \vee (\neg q \vee r)) \Rightarrow (p \vee r) \vee \neg q}{\mathsf{R}_{\vee}} \underset{k \vee (\neg x \vee (\neg q \vee r))) \Rightarrow (p \vee r) \vee \neg q}{\mathsf{R}_{\vee}} \\ \frac{\mathcal{D}_{1}}{x,\neg x \vee (\neg q \vee r), q \to r} \underset{k \vee (\neg q \vee r)) \rightarrow (p \vee r)}{\mathsf{R}_{\vee}} \underset{k \vee (\neg q \vee r)}{\mathsf{R}_{\vee}} \\ \frac{\mathcal{D}_{1}}{x,\neg x \vee (\neg q \vee r), q \to r} \underset{k \vee (\neg q \vee r)}{\mathsf{R}_{\vee}} \underset{k \vee (\neg q \vee r)}{\mathsf{R}_{\vee}} \\ \frac{\mathcal{D}_{1}}{x,\neg x \vee (\neg q \vee r), q \to r} \underset{k \vee (\neg q \vee r)}{\mathsf{R}_{\vee}} \underset{k \vee (\neg q \vee r)}{\mathsf{R}_{\vee}}$$

where notice that $\mathcal{R}(x \land (\neg x \lor (\neg q \lor (p \lor r)))) = \{x \land (\neg x \lor (\neg q \lor p)), x \land (\neg x \lor (\neg q \lor r))\}$ and that $\mathcal{R}((p \lor r) \lor \neg q) = \{p \lor \neg q, r \lor \neg q\}.$

6.7.1. THEOREM (Derivation normal form). *There is an effective procedure (that is height-preserving) transforming any derivation witnessing* $\vdash_{\mathsf{GT}^-} \Gamma \Rightarrow \Delta$ *into a derivation witnessing*

$$\vdash_{\mathsf{G3cp}^{-}} \bigwedge_{\Xi \in \mathcal{R}(\Gamma)} (\Xi \Rightarrow f[\Xi])$$
$$\vdash_{\mathcal{R} \boxtimes} \bigwedge_{\Xi \in \mathcal{R}(\Gamma)} (\Xi \Rightarrow \Delta)$$
$$\vdash_{\mathsf{L} \boxtimes} \Gamma \Rightarrow \Delta,$$

where $f : \mathcal{R}(\Gamma) \to \mathcal{R}(\Delta)$. We say that a derivation of $\Gamma \Rightarrow \Delta$ of the form above is in normal form.

Proof:

It suffices to show that applications of $\mathbb{R} \mathbb{V}$ can be commuted below applications of rules in G3cp, and that applications $L \mathbb{V}$ can be commuted below applications of all other rules (in a way that preserves height); it is then easy to see that the rest follows by the definition of resolutions. We show this by induction on the height *n* of derivations. If n = 1, there is nothing to show. We now assume the result holds for *n* and prove that it holds for n + 1 (note that most cases do not need to make use of the induction hypothesis). We only explicitly show that applications of the \mathbb{V} -rules can be commuted below applications of $\mathbb{R} \mathbb{V}$; and that applications of $L \mathbb{V}$ can be commuted below one another. The rest of the cases are similar. In most cases below we write only the relevant rule applications, and omit the rest of the derivation.

We show that the \vee -rules can be commuted below $R \wedge$.

 $\mathbb{R} \vee$: The principal formula of $\mathbb{R} \vee$ cannot be a side formula of $\mathbb{R} \wedge$ since the former must be nonclassical and the latter classical. It is therefore an active formula of $\mathbb{R} \wedge$, and we may assume that the relevant part of \mathcal{D} is of the form:

$$\frac{\frac{\Gamma \Rightarrow \chi\{\varphi_i\}, \Lambda}{\Gamma \Rightarrow \chi\{\varphi_L \lor \varphi_R\}, \Lambda} \mathsf{R} \lor}{\Gamma \Rightarrow \chi\{\varphi_L \lor \varphi_R\}, \Lambda} \mathsf{R} \lor} \Gamma \Rightarrow \psi, \Lambda} \mathsf{R} \land$$

This is transformed into:

$$\frac{\Gamma \Rightarrow \chi\{\varphi_i\}, \Lambda \qquad \Gamma \Rightarrow \psi, \Lambda}{\Gamma \Rightarrow \chi\{\varphi_i\} \land \psi, \Lambda, \Delta} \underset{\Gamma \Rightarrow \chi\{\varphi_L \lor \varphi_R\} \land \psi, \Lambda, \Delta}{R \lor} R \lor$$

LW: The relevant part of \mathcal{D} is of the form:

$$\frac{\mathcal{D}_{1} \qquad \mathcal{D}_{2}}{\Gamma, \chi\{\varphi_{L}\} \Rightarrow \varphi, \Lambda \qquad \Gamma, \chi\{\varphi_{R}\} \Rightarrow \varphi, \Lambda} \qquad \mathcal{D}_{3} \\
\frac{\Gamma, \chi\{\varphi_{L} \lor \varphi_{R}\} \Rightarrow \varphi, \Lambda \qquad \Gamma, \chi\{\varphi_{L} \lor \varphi_{R}\} \Rightarrow \psi, \Lambda}{\Gamma, \chi\{\varphi_{L} \lor \varphi_{R}\} \Rightarrow \varphi \land \psi, \Lambda, \Delta} \qquad \mathcal{R}_{A}$$

By the induction hypothesis, we may assume that either \mathcal{D}_3 consists of a single axiom, or the final rule applied in \mathcal{D}_3 is L \otimes , the principal formula of the final rule application is $\chi\{\varphi_L \otimes \varphi_R\}$, and the principal subformula is $\varphi_L \otimes \varphi_R$. In the former case, $\Gamma, \chi\{\varphi_L\} \Rightarrow \psi, \Lambda$ and $\Gamma, \chi\{\varphi_R\} \Rightarrow \psi, \Lambda$ are also axioms, and the above is transformed into:

$$\frac{\Gamma, \chi\{\varphi_L\} \Rightarrow \varphi, \Lambda \qquad \Gamma, \chi\{\varphi_L\} \Rightarrow \psi, \Lambda}{\frac{\Gamma, \chi\{\varphi_L\} \Rightarrow \varphi \land \psi, \Lambda, \Delta}{\Gamma, \chi\{\varphi_L\} \Rightarrow \varphi \land \psi, \Lambda, \Delta}} R \land \qquad \frac{\Gamma, \chi\{\varphi_R\} \Rightarrow \varphi, \Lambda \qquad \Gamma, \chi\{\varphi_R\} \Rightarrow \psi, \Lambda}{\Gamma, \chi\{\varphi_R\} \Rightarrow \varphi \land \psi, \Lambda, \Delta} R \land$$

In the latter case, we have:

$$\frac{\begin{array}{cccc}
\mathcal{D}_{1} & \mathcal{D}_{2} & \mathcal{D}_{4} & \mathcal{D}_{5} \\
\hline \Gamma, \chi\{\varphi_{L}\} \Rightarrow \varphi, \Lambda & \Gamma, \chi\{\varphi_{R}\} \Rightarrow \varphi, \Lambda \\
\hline \frac{\Gamma, \chi\{\varphi_{L} \lor \varphi_{R}\} \Rightarrow \varphi, \Lambda & \Gamma, \chi\{\varphi_{L}\} \Rightarrow \psi, \Lambda & \Gamma, \chi\{\varphi_{R}\} \Rightarrow \psi, \Lambda \\
\hline \Gamma, \chi\{\varphi_{L} \lor \varphi_{R}\} \Rightarrow \varphi, \Lambda & \Gamma, \chi\{\varphi_{L} \lor \varphi_{R}\} \Rightarrow \psi, \Lambda & \Gamma, \chi\{\varphi_{R}\} \Rightarrow \psi, \Lambda \\
\hline \Gamma, \chi\{\varphi_{L} \lor \varphi_{R}\} \Rightarrow \varphi \land \psi, \Lambda, \Delta & R \land
\end{array}$$

This is transformed into:

We show that $L \lor can be commuted below R \lor$. The relevant part of \mathcal{D} is of the form: $\Gamma, \chi\{\varphi_L\} \Rightarrow \eta\{\psi_i\}, \Delta \qquad \Gamma, \chi\{\varphi_R\} \Rightarrow \eta\{\psi_i\}, \Delta$

This is transformed into:

$$\frac{\Gamma, \chi\{\varphi_L\} \Rightarrow \eta\{\psi_i\}, \Delta}{\Gamma, \chi\{\varphi_L\} \Rightarrow \eta\{\psi_L \otimes \psi_R\}, \Delta} R \otimes \frac{\Gamma, \chi\{\varphi_R\} \Rightarrow \eta\{\psi_i\}, \Delta}{\Gamma, \chi\{\varphi_R\} \Rightarrow \eta\{\psi_L \otimes \psi_R\}, \Delta} R \otimes \frac{\Gamma, \chi\{\varphi_R\} \Rightarrow \eta\{\psi_L \otimes \psi_R\}, \Delta}{\Gamma, \chi\{\varphi_L \otimes \varphi_R\} \Rightarrow \eta\{\psi_L \otimes \psi_R\}, \Delta}$$

We show that applications of $L \otimes$ can be commuted around one another.

Case 1: The principal formula of the first application is a side formula of the second application. We may assume that the relevant part of \mathcal{D} is of the form:

$$\frac{\Gamma, \chi\{\varphi_L\}, \eta\{\psi_L\} \Rightarrow \Delta \qquad \Gamma, \chi\{\varphi_R\}, \eta\{\psi_L\} \Rightarrow \Delta}{\Gamma, \chi\{\varphi_L \lor \varphi_R\}, \eta\{\psi_L\} \Rightarrow \Delta} \sqcup \lor \qquad \Gamma, \chi\{\varphi_L \lor \varphi_R\}, \eta\{\psi_R\} \Rightarrow \Delta} \sqcup \lor$$

This case is clearly analogous to the L \otimes -subcase of R \wedge .

Case 2: The principal formula $\chi\{\varphi_L \lor \varphi_R\}$ of the first application is an active formula $\eta\{\psi_i\}$ of the second application (we assume i = L without loss of generality). There are three subcases.

Case 2.1: $\chi\{\varphi_L \lor \varphi_R\} = \chi\{\varphi_L \lor \varphi_R\}\{\psi_L\}$. The relevant part of \mathcal{D} is of the form:

$$\frac{\Gamma, \chi\{\varphi_L\}\{\psi_L\} \Rightarrow \Delta \qquad \Gamma, \chi\{\varphi_R\}\{\psi_L\} \Rightarrow \Delta}{\frac{\Gamma, \chi\{\varphi_L \lor \varphi_R\}\{\psi_L\} \Rightarrow \Delta}{\Gamma, \chi\{\varphi_L \lor \varphi_R\}\{\psi_L\} \Rightarrow \Delta}} L \lor \qquad \Gamma, \chi\{\varphi_L \lor \varphi_R\}\{\psi_R\} \Rightarrow \Delta} L \lor$$

This case is clearly analogous to the L \otimes -subcase of R \wedge .

Case 2.2: $\chi\{\varphi_L \otimes \varphi_R\} = \chi\{(\varphi_L \otimes \varphi_R)'\{\psi_L\}\}$ (we assume without loss of generality that $\chi\{\varphi_L \otimes \varphi_R\} = \chi\{\varphi'_L\{\psi_L\} \otimes \varphi_R\}$). The relevant part of \mathcal{D} is of the form:

$$\frac{\Gamma, \chi\{\varphi'_L\{\psi_L\}\} \Rightarrow \Delta \qquad \Gamma, \chi\{\varphi_R\} \Rightarrow \Delta}{\frac{\Gamma, \chi\{\varphi'_L\{\psi_L\} \lor \varphi_R\} \Rightarrow \Delta}{\Gamma, \chi\{\varphi'_L\{\psi_L \lor \psi_R\} \lor \varphi_R\} \Rightarrow \Delta}} L \bowtie \qquad \Gamma, \chi\{\varphi'_L\{\psi_R\} \lor \varphi_R\} \Rightarrow \Delta}$$

This case is similar to the L \mathbb{W} -subcase of R \wedge .

Case 2.3: $\chi\{\varphi_L \otimes \varphi_R\} = \eta\{\psi_L\} = \eta\{\psi_L\} = \eta\{\psi_L \otimes \varphi_R\}$. The relevant part of \mathcal{D} is of the form $\Gamma, \eta \{ \psi_L' \{ \varphi_L \} \} \Rightarrow \Delta \qquad \Gamma, \eta \{ \psi_L' \{ \varphi_R \} \} \Rightarrow \Delta$

$$\frac{,\eta\{\psi_L\{\varphi_L\}\}\Rightarrow\Delta}{\Gamma,\eta\{\psi_L'\{\varphi_L\vee\varphi_R\}\}\Rightarrow\Delta} \stackrel{L}{\longrightarrow} \Gamma,\eta\{\psi_R'\}\Rightarrow\Delta}{\Gamma,\eta\{\psi_L'\{\varphi_L\vee\varphi_R\}\vee\psi_R\}\Rightarrow\Delta} L \otimes \Gamma,\eta\{\psi_R\}\Rightarrow\Delta} L \otimes \Gamma$$

This is transformed into:

$$\frac{\Gamma, \eta\{\psi_{L}'\{\varphi_{L}\}\} \Rightarrow \Delta \qquad \Gamma, \eta\{\psi_{R}\} \Rightarrow \Delta}{\Gamma, \eta\{\psi_{L}'\{\varphi_{L}\} \otimes \psi_{R}\} \Rightarrow \Delta} \ \Box \otimes \qquad \frac{\Gamma, \eta\{\psi_{L}'\{\varphi_{R}\}\} \Rightarrow \Delta \qquad \Gamma, \eta\{\psi_{R}\} \Rightarrow \Delta}{\Gamma, \eta\{\psi_{L}'\{\varphi_{R}\} \otimes \psi_{R}\} \Rightarrow \Delta} \ \Box \otimes \qquad \frac{\Gamma, \eta\{\psi_{L}'\{\varphi_{R}\} \otimes \psi_{R}\} \Rightarrow \Delta}{\Gamma, \eta\{\psi_{L}'\{\varphi_{L} \otimes \psi_{R}\} \otimes \psi_{R}\} \Rightarrow \Delta} \ \Box \otimes \qquad \frac{\Gamma, \eta\{\psi_{L}'\{\varphi_{R}\}\} \otimes \psi_{R}\} \Rightarrow \Delta}{\Gamma, \eta\{\psi_{L}'\{\varphi_{L} \otimes \psi_{R}\} \otimes \psi_{R}\} \Rightarrow \Delta} \ \Box \otimes \qquad \frac{\Gamma, \eta\{\psi_{L}'\{\varphi_{R}\}\} \otimes \psi_{R}\} \Rightarrow \Delta}{\Gamma, \eta\{\psi_{L}'\{\varphi_{L} \otimes \psi_{R}\} \otimes \psi_{R}\} \Rightarrow \Delta} \ \Box \otimes \qquad \frac{\Gamma, \eta\{\psi_{L}'\{\varphi_{R}\}\} \otimes \Delta}{\Gamma, \eta\{\psi_{L}'\{\varphi_{R}\} \otimes \psi_{R}\} \Rightarrow \Delta} \ \Box \otimes \qquad \frac{\Gamma, \eta\{\psi_{L}'\{\varphi_{R}\}\} \otimes \Delta}{\Gamma, \eta\{\psi_{L}'\{\varphi_{R}\} \otimes \psi_{R}\} \Rightarrow \Delta} \ \Box \otimes \qquad \frac{\Gamma, \eta\{\psi_{L}'\{\varphi_{R}\} \otimes \psi_{R}\} \Rightarrow \Delta}{\Gamma, \eta\{\psi_{L}'\{\varphi_{R}\} \otimes \psi_{R}\} \Rightarrow \Delta} \ \Box \otimes \qquad \frac{\Gamma, \eta\{\psi_{L}'\{\varphi_{R}\} \otimes \psi_{R}\} \Rightarrow \Delta}{\Gamma, \eta\{\psi_{L}'\{\varphi_{R}\} \otimes \psi_{R}\} \Rightarrow \Delta} \ \Box \otimes \qquad \frac{\Gamma, \eta\{\psi_{L}'\{\varphi_{R}\} \otimes \psi_{R}\} \Rightarrow \Delta}{\Gamma, \eta\{\psi_{L}'\{\varphi_{R}\} \otimes \psi_{R}\} \Rightarrow \Delta} \ \Box \otimes \qquad \frac{\Gamma, \eta\{\psi_{L}'\{\varphi_{R}\} \otimes \psi_{R}\} \Rightarrow \Delta}{\Gamma, \eta\{\psi_{L}'\{\varphi_{R}\} \otimes \psi_{R}\} \Rightarrow \Delta} \ \Box \otimes \qquad \frac{\Gamma, \eta\{\psi_{L}'\{\varphi_{R}\} \otimes \psi_{R}\} \Rightarrow \Delta}{\Gamma, \eta\{\psi_{L}'\{\varphi_{R}\} \otimes \psi_{R}\} \Rightarrow \Delta} \ \Box \otimes \qquad \frac{\Gamma, \eta\{\psi_{L}'\{\varphi_{R}\} \otimes \psi_{R}\} \Rightarrow \Delta}{\Gamma, \eta\{\psi_{L}'\{\varphi_{R}\} \otimes \psi_{R}\} \Rightarrow \Delta} \ \Box \otimes \qquad \frac{\Gamma, \eta\{\psi_{L}'\{\varphi_{R}\} \otimes \psi_{R}\} \Rightarrow \Delta}{\Gamma, \eta\{\psi_{L}'\{\varphi_{R}\} \otimes \psi_{R}\} \Rightarrow \Delta} \ \Box \otimes \qquad \frac{\Gamma, \eta\{\psi_{L}'\{\varphi_{R}\} \otimes \psi_{R}\} \Rightarrow \Delta}{\Gamma, \eta\{\psi_{L}'\{\varphi_{R}\} \otimes \psi_{R}\} \Rightarrow \Delta} \ \Box \otimes \qquad \frac{\Gamma, \eta\{\psi_{L}'\{\varphi_{R}\} \otimes \psi_{R}\} \Rightarrow \Delta}{\Gamma, \eta\{\psi_{L}'\{\varphi_{R}\} \otimes \psi_{R}\} \Rightarrow \Delta} \ \Box \otimes \qquad \frac{\Gamma, \eta\{\psi_{L}'\{\varphi_{R}\} \otimes \psi_{R}\} \Rightarrow \Delta}{\Gamma, \eta\{\psi_{L}'\{\varphi_{R}\} \otimes \psi_{R}\} \Rightarrow \Delta} \ \Box \otimes \qquad \frac{\Gamma, \eta\{\psi_{L}'\{\varphi_{R}\} \otimes \psi_{R}\} \Rightarrow \Delta}{\Gamma, \eta\{\psi_{L}'\{\varphi_{R}\} \otimes \psi_{R}\} \Rightarrow \Delta} \ \Box \otimes \qquad \frac{\Gamma, \eta\{\psi_{L}'\{\varphi_{R}\} \otimes \psi_{R}\} \Rightarrow \Delta}{\Gamma, \eta\{\psi_{L}'\{\varphi_{R}\} \otimes \psi_{R}\} \Rightarrow \Delta} \ \Box \otimes \qquad \frac{\Gamma, \eta\{\psi_{L}'\{\varphi_{R}\} \otimes \psi_{R}\} \Rightarrow \Delta}{\Gamma, \eta\{\psi_{L}'\{\varphi_{R}\} \otimes \psi_{R}\} \Rightarrow \Delta} \ \Box \otimes \qquad \frac{\Gamma, \eta\{\psi_{L}'\{\varphi_{R}\} \otimes \psi_{R}\} \Rightarrow \Delta}{\Gamma, \eta\{\psi_{L}'\{\varphi_{R}\} \otimes \psi_{R}\} \Rightarrow \Delta} \ \Box \otimes \qquad \frac{\Gamma, \eta\{\psi_{R}'\{\varphi_{R}\} \otimes \psi_{R}\} \Rightarrow \Delta}{\Gamma, \eta\{\psi_{R}'\{\varphi_{R}\} \otimes \psi_{R}\} \Rightarrow \Delta}$$

Observe that derivations in normal form have a stronger subformula property than that of cutfree derivations in general (Proposition 6.4.3):

6.7.2. PROPOSITION (Normal form derivation partial resolution subformula property). For any \mathcal{D} in normal form witnessing $\Gamma \vdash_{\mathsf{GT}^-} \Delta$, each formula occurring in \mathcal{D} is a partial resolution or a subformula of a resolution of some formula occurring in Γ, Δ .

We are now ready to prove our cut elimination theorem.

212 Chapter 6. A Deep-Inference Sequent Calculus for a Propositional Team Logic

6.7.3. THEOREM (Cut elimination). If \mathcal{D} witnesses $\vdash_{\mathsf{GT}} \Gamma \Rightarrow \Delta$, there is an effective procedure for transforming \mathcal{D} into a derivation \mathcal{D}' witnessing $\vdash_{\mathsf{GT}^-} \Gamma \Rightarrow \Delta$.

Proof:

By induction on the cutrank, with a subinduction on the level of cuts. It suffices to show that we can transform a derivation \mathcal{D} ending in a cut

$$\frac{\begin{array}{cc} \mathcal{D}_{1} & \mathcal{D}_{2} \\ \hline \Gamma \Rightarrow \varphi, \Delta & \Pi, \varphi \Rightarrow \Sigma \\ \hline \Gamma, \Pi \Rightarrow \Delta, \Sigma \end{array} \mathsf{Cut}$$

where \mathcal{D}_1 and \mathcal{D}_2 are cutfree into a cutfree derivation. By Theorem 6.7.1, we can replace $\mathcal{D}_1, \mathcal{D}_2$ with cutfree $\mathcal{D}'_1, \mathcal{D}'_2$ in normal form; i.e., \mathcal{D}'_1 witnesses

$$\vdash_{\mathsf{G3cp}^{-}} \bigwedge_{\Xi \in \mathcal{R}(\Gamma)} (\Xi \Rightarrow f(\Xi))$$
$$\vdash_{\mathcal{R} \mathbb{W}} \bigwedge_{\Xi \in \mathcal{R}(\Gamma)} (\Xi \Rightarrow \varphi, \Delta)$$
$$\vdash_{\mathsf{L} \mathbb{W}} \Gamma \Rightarrow \varphi, \Delta,$$

where $f : \mathcal{R}(\Gamma) \to \mathcal{R}(\varphi, \Delta)$, and \mathcal{D}'_2 witnesses

$$\begin{split} & \vdash_{\mathsf{G3cp}^{-}} \bigwedge_{(\Theta,\alpha) \in \mathcal{R}(\Pi,\varphi)} (\Theta, \alpha \Rightarrow g(\Theta, \alpha)) \\ & \vdash_{\mathcal{RW}} \bigwedge_{(\Theta,\alpha) \in \mathcal{R}(\Pi,\varphi)} (\Theta, \alpha \Rightarrow \Sigma) \\ & \vdash_{\mathsf{LW}} \Pi, \varphi \Rightarrow \Sigma, \end{split}$$

where $g : \mathcal{R}(\Pi, \varphi) \to \mathcal{R}(\Sigma)$, each $\Theta \in \mathcal{R}(\Pi)$, and each $\alpha \in \mathcal{R}(\varphi)$.

Write $f(\Xi) = \alpha^{\Xi}, \Lambda^{\Xi} \in \mathcal{R}(\varphi, \Delta)$. For each $(\Xi, \Theta) \in \mathcal{R}(\Gamma, \Pi)$, we can find a subderivation \mathcal{D}^{Ξ} of \mathcal{D}'_1 witnessing $\vdash_{\mathsf{G3cp}^-} \Xi \Rightarrow \Lambda^{\Xi}, \alpha^{\Xi}$ and a subderivation $\mathcal{D}^{\Xi,\Theta}$ of \mathcal{D}'_2 witnessing $\vdash_{\mathsf{G3cp}^-} \Theta, \alpha^{\Xi} \Rightarrow g(\Theta, \alpha^{\Xi})$. We can then construct the following G3cpderivation:

$$\frac{\begin{array}{ccc} \mathcal{D}^{\pm} & \mathcal{D}^{\pm,\Theta} \\ \hline \Xi \Rightarrow \Lambda^{\pm}, \alpha^{\pm} & \Theta, \alpha^{\pm} \Rightarrow g(\Theta, \alpha^{\pm}) \\ \hline \Xi, \Theta, \Rightarrow \Lambda^{\pm}, g(\Theta, \alpha^{\pm}) \end{array} \mathsf{Cut}$$

By classical cut elimination (Theorem 6.3.3), there is an effective procedure transforming this derivation into a derivation $\mathcal{D}_{\Xi,\Theta}$ witnessing $\vdash_{\mathsf{G3cp}^-} \Xi, \Theta, \Rightarrow \Lambda^{\Xi}, g(\Theta, \alpha^{\Xi})$.

Combining the derivations $\mathcal{D}_{\Xi,\Theta}$ with subderivations derived from the subderivations of \mathcal{D}'_1 and \mathcal{D}'_2 which apply $\mathbb{R} \vee$ to partial resolutions of Δ, Σ (changing the contexts as appropriate), we can construct derivations witnessing

$$\vdash_{\mathsf{G3cp}^{-}} \bigwedge_{(\Xi,\Theta)\in\mathcal{R}(\Gamma,\Pi)} (\Xi,\Theta,\Rightarrow\Lambda^{\Xi},g(\Theta,\alpha^{\Xi}))$$
$$\vdash_{\mathcal{RW}} \bigwedge_{(\Xi,\Theta)\in\mathcal{R}(\Gamma,\Pi)} (\Xi,\Theta,\Rightarrow\Delta,\Sigma)$$

Finally, combining these derivations with subderivations derived from the subderivations of \mathcal{D}'_1 and \mathcal{D}'_2 which apply $L \otimes$ to partial resolutions of Γ, Π (changing the contexts as appropriate), we can construct a derivation witnessing

$$\vdash_{\mathsf{G3cp}^{-}} \bigwedge_{(\Xi,\Theta)\in\mathcal{R}(\Gamma,\Pi)} (\Xi,\Theta,\Rightarrow\Lambda^{\Xi},g(\Theta,\alpha^{\Xi}))$$

$$\vdash_{\mathcal{R}} \bigvee \bigwedge_{(\Xi,\Theta)\in\mathcal{R}(\Gamma,\Pi)} (\Xi,\Theta,\Rightarrow\Delta,\Sigma)$$

$$\vdash_{\mathsf{L}} \bigvee \Gamma,\Pi\Rightarrow\Delta,\Sigma \qquad \Box$$

Let us consider a simple example. We wish to eliminate the following cut from a derivation \mathcal{D} , where \mathcal{D}_1 and \mathcal{D}_2 are cutfree:

$$\frac{\mathcal{D}_{1} \qquad \mathcal{D}_{2}}{a \vee (p \vee q) \Rightarrow p \vee q, a \qquad b, p \vee q \Rightarrow (b \wedge p) \vee (b \wedge q)} \operatorname{Cut}_{a \vee (p \vee q), b \Rightarrow a, (b \wedge p) \vee (b \wedge q)} \operatorname{Cut}_{a \vee (p \vee q), b \Rightarrow a, (b \wedge p) \vee (b \wedge q)}$$

The normal form theorem transforms the derivations \mathcal{D}_1 and \mathcal{D}_2 into derivations in normal form \mathcal{D}'_1 and \mathcal{D}'_2 :

$$\frac{D^{a \lor p}}{a \lor p \Rightarrow p, a} \xrightarrow{D^{a \lor q}} \frac{D^{a \lor q}}{a \lor q \Rightarrow q, a} \underset{L \lor V}{A \lor q \Rightarrow p \lor q, a} \underset{L \lor V}{R \lor q} \xrightarrow{P \lor q, a} \underset{L \lor V}{R \lor q} \xrightarrow{D^{a \lor q}} \frac{D^{a \lor q}}{a \lor q \Rightarrow q, a} \underset{L \lor V}{R \lor q} \xrightarrow{D^{a \lor q}, b} \underset{L \lor V}{B, p \Rightarrow b \land p} \underset{L \lor V}{R \lor q} \xrightarrow{D^{a \lor q}, b} \xrightarrow{D^{a \lor q}, b} \xrightarrow{D^{a \lor q}, b} \underset{L \lor V}{B, q \Rightarrow (b \land p) \lor (b \land q)} \underset{L \lor V}{R \lor q} \xrightarrow{D^{a \lor q}, b} \underset{L \lor V}{R \lor q} \xrightarrow{D^{a \lor q}, b} \xrightarrow{D^{a \lor q}, b} \underset{L \lor V}{R \lor q} \xrightarrow{D^{a \lor q}, b} \xrightarrow{D^{a \lor q}, b} \underset{L \lor V}{R \lor q} \xrightarrow{D^{a \lor q}, b} \xrightarrow{D^{a \lor q}, b} \underset{L \lor V}{R \lor q} \xrightarrow{D^{a \lor q}, b} \xrightarrow{D^{a \lor q}, b} \underset{L \lor V}{R \lor q} \xrightarrow{D^{a \lor q}, b} \xrightarrow{D^{a \lor q}, b}$$

Combining the classical subderivations of \mathcal{D}'_1 and \mathcal{D}'_2 , we have the following G3cp-derivations:

$$\frac{\mathcal{D}^{a \lor p}}{a \lor p \Rightarrow p, a} \frac{\mathcal{D}^{a \lor p, b}}{b, p \Rightarrow b \land p} \operatorname{Cut} \frac{\mathcal{D}^{a \lor q}}{a \lor q \Rightarrow q, a} \frac{\mathcal{D}^{a \lor q, b}}{b, q \Rightarrow b \land q} \operatorname{Cut}$$

By classical cut elimination, we may transform the first of these into a classical cutfree derivation $\mathcal{D}_{a \lor p,b}$ of $a \lor p, b \Rightarrow a, b \land p$, and similarly for the second. Combining these derivations with subderivations derived as appropriate from \mathcal{D}'_1 and \mathcal{D}'_2 , we have the following cutfree derivation of $a \lor (p \lor q), b \Rightarrow (b \land p) \lor (b \land q)$:

$$\frac{\mathcal{D}_{a \lor p, b}}{\frac{a \lor p, b \Rightarrow a, b \land p}{a \lor q, b \Rightarrow a, (b \land p) \lor (b \land q)}} \operatorname{Rw} \frac{\mathcal{D}_{a \lor q, b}}{a \lor q, b \Rightarrow a, b \land q} \operatorname{Rw} \frac{a \lor q, b \Rightarrow a, b \land q}{a \lor q, b \Rightarrow a, (b \land p) \lor (b \land q)} \operatorname{Rw} Lw$$

Note that we have now the following counterpart to the Resolution theorem (Theorem 6.2.9) in our system: **6.7.4.** COROLLARY (Derivability resolution theorem). There is an effective procedure transforming any derivation \mathcal{D} witnessing $\vdash_{\mathsf{GT}} \Gamma \Rightarrow \Delta$ into derivations \mathcal{D}_{Ξ} ($\Xi \in \mathcal{R}(\Gamma)$) witnessing $\vdash_{\mathsf{GT}} \bigwedge_{\Xi \in \mathcal{R}(\Gamma)} (\Xi \Rightarrow f[\Xi])$ where $f : \mathcal{R}(\Gamma) \to \mathcal{R}(\Delta)$, and vice versa.

Proof:

For the first direction, Theorem 6.7.3 yields a derivation witnessing $\vdash_{\mathsf{GT}^-} \Gamma \Rightarrow \Delta$, and the result then follows by Theorem 6.7.1. For the second direction, we can clearly construct a derivation witnessing $\vdash_{\mathsf{GT}} \Gamma \Rightarrow \Delta$ using the derivations \mathcal{D}_{Ξ} and the rules $\mathbb{R} \mathbb{V}$ and $\mathbb{L} \mathbb{V}$.

6.8 Variant System with Independent Contexts

Our system GT features shared contexts for the two-premise rules, and, as in G3cp, the structural rules of contraction are implicit, having been 'absorbed' into these shared-context rules. In this section, we briefly present a variant of GT with independent contexts for R \land and L \lor , and with explicit contraction rules. This variant features no syntactic restrictions on the rules R \land and L \lor ; the rules of the variant therefore make it explicit that the structural effect of the syntactic restrictions and the implicit weakening in R \land and L \lor is to allow for right weakening for all formulas while allowing for right contraction only for classical formulas.

6.8.1. DEFINITION (The sequent calculus GT'). The rules for GT' are as for GT except we remove the rules $R \land$ and $L \lor$, and add:



It is interesting to note that the \vee - and \wedge -rules in this variant coincide with those of the *multiplicative disjunction (par)* and *conjunction (tensor)* in linear logic, respectively. Cf. [2], in which a connection is drawn between the semantics of the split disjunction \vee and the multiplicative conjunction (recall also that \vee is often called the *tensor disjunction*). The global disjunction, on the other hand, has the rules for the *additive disjunction (plus)*, modulo deep inference.

6.8.2. THEOREM (Soundness of GT'). $\vdash \Gamma \Rightarrow \Delta$ implies $\land \Gamma \vDash \lor \Delta$.

Proof:

By induction on the depth of derivations; we look at the final rule applied in the derivation and only consider the cases Lv' and Rv'

- Lv': Assume $\wedge \Gamma_1 \wedge \varphi \models \forall \Delta_1$ and $\wedge \Gamma_2 \wedge \psi \models \forall \Delta_2$. We want to show $\wedge \Gamma_1 \wedge \Lambda \Gamma_2 \wedge (\varphi \lor \psi) \models \forall \Delta_1 \lor \forall \Delta_2$, so let $t \models \wedge \Gamma_1 \wedge \wedge \Gamma_2 \wedge (\varphi \lor \psi)$. Then $t = t_{\varphi} \cup t_{\psi}$ where $t_{\varphi} \models \varphi$ and $t_{\psi} \models \psi$, and $t \models \wedge \Gamma_1 \wedge \wedge \Gamma_2$. By downward closure, $t_{\varphi} \models \wedge \Gamma_1$ so that $t_{\varphi} \models \forall \Delta_1$; similarly $t_{\psi} \models \forall \Delta_2$, whence $t \models \forall \Delta_1 \lor \forall \Delta_2$.
- RA: Assume $\wedge \Gamma_1 \models \varphi \lor \lor \Delta_1$ and $\wedge \Gamma_2 \models \psi \lor \lor \Delta_2$. We want to show $\wedge \Gamma_1 \land \land \Gamma_2 \models (\varphi \land \psi) \lor \lor \Delta_1 \lor \lor \lor \Delta_2$, so let $t \models \land \Gamma_1 \land \land \Gamma_2$. Then $t \models \varphi \lor \lor \Delta_1$ and $t \models \psi \lor \lor \Delta_2$, so $t = t_{\varphi} \cup t_{\Delta_1}$ where $t_{\varphi} \models \varphi$ and $t_{\Delta_1} \models \lor \Delta_1$, and $t = t_{\psi} \cup t_{\Delta_2}$ where $t_{\psi} \models \psi$ and $t_{\Delta_2} \models \lor \Delta_2$. Then also $t = (t_{\varphi} \cap t_{\psi}) \cup t_{\Delta_1} \cup t_{\Delta_2}$. By downward closure, $t_{\varphi} \cap t_{\psi} \models \varphi \land \psi$, whence, $t \models (\varphi \land \psi) \lor \lor \Delta_1 \lor \lor \Delta_2$.

It is easy to see that the systems GT and GT' are equivalent:

6.8.3. PROPOSITION. $\vdash_{\mathsf{GT}^{-}} \Gamma \Rightarrow \Delta \iff \vdash_{\mathsf{GT}'^{-}} \Gamma \Rightarrow \Delta$.

Proof:

The direction \Leftarrow follows by soundness and the cutfree completeness of GT (Theorem 6.4.7). The direction \implies follows by the admissibility of R \lor and L \lor in GT^{'-}. For instance, we have:

$$\frac{\Gamma, \varphi \Rightarrow \Lambda}{\Gamma, \Gamma, \varphi \lor \psi \Rightarrow \Lambda, \Lambda} \mathsf{L} \lor'$$

Clearly $\Gamma, \Gamma, \varphi \lor \psi \Rightarrow \Lambda, \Lambda \vdash_{\mathsf{LC}} \Gamma, \varphi \lor \psi \Rightarrow \Lambda, \Lambda \vdash_{\mathsf{RC}} \Gamma, \varphi \lor \psi \Rightarrow \Lambda$. As with GT (Lemma 6.5.1), it is easy to show that weakening is admissible in GT'^- , whence $\Gamma, \varphi \lor \psi \Rightarrow \Lambda \vdash_{\mathsf{GT}'^-} \Gamma, \varphi \lor \psi \Rightarrow \Lambda, \Delta$.

6.9 Concluding Remarks

In this paper, we presented a sequent calculus GT for the team logic PL(W) consisting of a subsystem G3cp for classical propositional logic together with deep-inference rules for the inquisitive disjunction W. We generalized the notion of resolutions from inquisitive logic to define the notion of partial resolutions, and used partial resolutions to define a weak subformula property for GT^- as well as to prove the cutfree completeness of GT. We showed that GT^- admits height-preserving weakening, contraction, and inversion; with the caveats that right contraction is only admissible for classical formulas (which is as expected, given the connection we observed between right contraction and union closure), and that RW is only 'invertible' in the sense of the Split property (Proposition 6.2.8). We provided a procedure for constructing cutfree derivations and countermodels in GT that is similar to the analogous procedure for G3cp. We proved a normal form theorem for cutfree derivations, using which we provided a cut elimination procedure for GT. Finally, we defined a variant of GT with independent-context rules instead of shared-context rules with restricted contexts for the connectives \lor and \land , discussed the similarities between these variant and multiplicative linear logic, and showed that this variant is equivalent to GT.

Let us conclude by discussing some open problems, as well as the prospects of applying our deep-inference approach to provide sequent calculi for other team logics.

With regard to the first of these concerns, are there any further interesting consequences we can draw from the cutfree completeness of GT? To consider one example, can we leverage the cutfree completeness of GT to provide a constructive proof of Craig's/Lyndon's/uniform interpolation for PL(W), perhaps along the lines of [118, pp. 116–118], or [103, 85, 84]? (There is a semantic proof of uniform interpolation for PL(W) that follows essentially from results in [43].)

We can provide one limited negative result here: the procedure for proving Craig's and Lyndon's interpolation via sequent interpolants, as detailed in, for instance [118, pp. 116–118], fails for GT because the form of interpolation involved does not hold for PL(\forall). This form of interpolation states, roughly, that for each sequent $\Gamma \Rightarrow \Delta$, and each pair of partitions $\Gamma_1; \Gamma_2$ of Γ and $\Delta_1; \Delta_2$ of Δ , there is an interpolant φ in the common language of Γ_1, Δ_1 and Γ_2, Δ_2 such that $\Gamma_1 \Rightarrow \Delta_1, \varphi$ and $\Gamma_2, \varphi \Rightarrow \Delta_2$. Consider, however, the following valid sequent, where $?p := p \forall \neg p$, etc.

$$?p,?q \Rightarrow ?p \land ?q \land r,?p \land ?q \land \neg r$$

If it were the case that this form of interpolation held for PL(\forall), there would be a formula φ such that the following sequents are valid: $?p \Rightarrow ?p \land ?q \land r, \varphi$ (whence $?p \models (?p \land ?q \land r) \lor \varphi$) and $?q, \varphi \Rightarrow ?p \land ?q \land \neg r$ (whence $?q \land \varphi \models ?p \land ?q \land \neg r$). Now consider the team $t := \{v_{p,q,r}, v_{p,\overline{q},r}, v_{p,\overline{q},\overline{r}}\}$. Since $t \models ?p$, we must have $t \models (?p \land ?q \land \neg r)$ $r) \lor \varphi$, whence $t = t_1 \cup t_2$, where $t_1 \models ?p \land ?q \land r$ and $t_2 \models \varphi$. Clearly either $t_1 \subseteq \{v_{p,q,r}\}$ or $t_1 \subseteq \{v_{p,\overline{q},r}\}$, whence either $v_{p,\overline{q},r} \in t_2$ or $v_{p,q,r} \in t_2$. In the former case (the latter case is similar), by downward closure, $\{v_{p,\overline{q},r}\} \models \varphi$. Clearly also $\{v_{p,\overline{q},r}\} \models ?q$, whence $\{v_{p,\overline{q},r}\} \models ?p \land ?q \land \neg r$, but this clearly cannot be the case.

So the type of interpolation involved in the procedure in [118, pp. 116–118] fails; there might, however, be some other way of leveraging the system GT to prove interpolation results for PL(W).

We chose to employ only one set of deep-inference rules in order to keep our system as close to a Gentzen-style system as possible. However, it might be interesting to develop a system for PL(W) making more extensive use of deep-inference rules and techniques in order to make use of the powerful general results in the deep-inference literature.

With regard to extensions of our approach to other team logics, let us first note that given that resolutions are adapted from inquisitive logic, and that the analogues of the results in Section 6.2.2 hold for propositional inquisitive logic, we expect that our approach would also work for propositional inquisitive logic InqB (essentially PL(W)) without the split disjunction \lor , but with the so-called *intuitionistic implication* \rightarrow). However, given the fact that the inquisitive disjunction w does not simply distribute over all other connectives in InqB (as it does in PL(W)), which results in the resolutions in InqB having a more complicated structure than in PL(W), it is not clear whether a deep-inference calculus along these lines for InqB would be as simple and appealing as the system in this paper. As for other team logics, we suspect, given our strong focus on the properties of a single team-semantic connective-the inquisitive disjunction \vee —as well as the apparent reliance of the cutfree completeness of our system on the Resolution theorem (Theorem 6.2.9), which depends, in turn, on the downward closure of PL(W) as well as the union closure of the W-free fragment, we expect any application of this approach to logics which do not involve \vee and which do not also share these additional features to involve some further innovations and/or stronger prooftheoretic machinery. There are, however, some other interesting team logics which do incorporate w and do share these features (for instance, the extension of Team Linear *Temporal Logic* with \vee in [92]).

To summarize, our approach was tailored for PL(W), and the resulting system, with its simplicity and clarity, appears to be a good fit for this logic. Whether this approach can be generalised to other team logics remains to be explored. The intriguing links in our setting between structural properties and team-semantic closure properties do suggest that further investigation of team logics which differ in their closure properties from PL(W) may prove fruitful; it might also be interesting to change the interpretation of the comma in the antecedent of a sequent to produce correspondences between teamsemantic closure properties and structural properties on the left.

Bibliography

- Samson Abramsky, Joni Puljujärvi, and Jouko Väänänen. Team Semantics and Independence Notions in Quantum Physics. 2024. arXiv: 2107. 10817 [math.LO]. URL: https://arxiv.org/abs/2107.10817.
- [2] Samson Abramsky and Jouko Väänänen. "From IF to BI". In: Synthese 167.2 (2009), pp. 207–230. URL: https://doi.org/10.1007/ s11229-008-9415-6.
- [3] Martin Aher. "Free choice in deontic inquisitive semantics (DIS)". In: Proceedings of the 18th Amsterdam Colloquium Conference on Logic, Language and Meaning. AC'11. Amsterdam, The Netherlands: Springer-Verlag, 2011, pp. 22–31. URL: https://doi.org/10.1007/ 978-3-642-31482-7_3.
- [4] Rafael Albert and Erich Grädel. "Unifying hidden-variable problems from quantum mechanics by logics of dependence and independence". In: *Ann. Pure Appl. Logic* 173.10 (2022). Logics of Dependence and Independence, p. 103088. URL: https://doi.org/10.1016/j.apal. 2022.103088.
- [5] Maria Aloni. "Free choice, modals and imperatives". In: Natural Language Semantics 15.1 (2007), pp. 65–94. URL: https://doi.org/10. 1007/s11050-007-9010-2.
- [6] Maria Aloni. "Logic and conversation: the case of free choice". In: Semantics and Pragmatics 15.5 (2022). URL: https://doi.org/10.3765/sp.15.5.
- [7] Maria Aloni, Aleksi Anttila, and Fan Yang. "State-Based Modal Logics for Free Choice". In: *Notre Dame J. Formal Logic* (2024), pp. 1–47. URL: https://doi.org/10.1215/00294527-2024-0027.

- [8] Maria Aloni and Ivano Ciardelli. "A logical account of free choice imperatives". In: *The Dynamic, Inquisitive, and Visionary Life of φ*, ?φ, and ◊φ. Ed. by Michael Franke Maria Aloni and Floris Roelofsen. Institute for Logic, Language and Computation, 2013, pp. 1–17. URL: https://festschriften.illc.uva.nl/Festschrift-JMF/.
- [9] Maria Aloni and Marco Degano. How to be (non-)specific. Preprint. 2023. URL: https://m-degano.github.io/files/how_ nonspecific_draft.pdf.
- [10] Luis Alonso-Ovalle. "Disjunction in Alternative Semantics". PhD thesis. University of Massachusetts, Amherst, 2006. URL: https:// scholarworks.umass.edu/dissertations/AAI3242324/.
- [11] Alan Ross Anderson and Nuel D. Belnap Jr. *Entailment*. Volume I: The logic of relevance and necessity, With contributions by J. Michael Dunn and Robert K. Meyer, and further contributions by John R. Chidgey, J. Alberto Coffa, Dorothy L. Grover, Bas van Fraassen, Hugues LeBlanc, Storrs McCall, Zane Parks, Garrel Pottinger, Richard Routley, Alasdair Urquhart and Robert G. Wolf. Princeton University Press, Princeton, N.J.-London, 1975.
- [12] Aleksi Anttila. "The Logic of Free Choice. Axiomatizations of Statebased Modal Logics". MSc thesis. University of Amsterdam, 2021. URL: https://eprints.illc.uva.nl/id/eprint/1788/.
- [13] Aleksi Anttila. Further remarks on the dual negation in team logics. 2024. arXiv: 2410.07067 [math.LO]. URL: https://arxiv.org/ abs/2410.07067.
- [14] Aleksi Anttila, Matilda Häggblom, and Fan Yang. "Axiomatizing modal inclusion logic and its variants". In: Arch. Math. Logic (2024). Forthcoming. arXiv: 2312.02285 [math.L0]. URL: https://arxiv.org/ abs/2312.02285.
- [15] Aleksi Anttila, Rosalie Iemhoff, and Fan Yang. *Deep-inference Sequent Calculi for Propositional Team Logics*. Manuscript. 2024.
- [16] Aleksi Anttila and Søren Brinck Knudstorp. *Convex Propositional and Modal Team Logics*. Manuscript. 2024.
- [17] Alexandru Baltag and Bryan Renne. "Dynamic Epistemic Logic". In: The Stanford Encyclopedia of Philosophy. Winter 2016. Metaphysics Research Lab, Stanford University, 2016. URL: https:// plato.stanford.edu/archives/win2016/entries/dynamicepistemic/.
- [18] Fausto Barbero, Marianna Girlando, Fan Yang, and Valentin Müller. Labelled proof systems for inquisitive and propositional team-based logics. Manuscript. 2024.

- [19] John Barwise and Robin Cooper. "Generalized Quantifiers and Natural Language". In: *Linguistics and Philosophy* 4.2 (1981), pp. 159–219. URL: https://doi.org/10.1007/bf00350139.
- [20] Dylan Bellier, Massimo Benerecetti, Dario Della Monica, and Fabio Mogavero. "Alternating (in)dependence-friendly logic". In: Ann. Pure Appl. Logic 174.10 (2023), Paper No. 103315, 58. URL: https://doi. org/10.1016/j.apal.2023.103315.
- [21] Nuel D. Belnap Jr. "A useful four-valued logic". In: Modern uses of multiple-valued logic (Fifth Internat. Sympos., Indiana Univ., Bloomington, Ind., 1975). Vol. Vol. 2. Episteme. Reidel, Dordrecht-Boston, Mass., 1977, pp. 5–37. URL: https://doi.org/10.1007/978-94-010-1161-7_2.
- [22] Nuel D. Belnap Jr. "How a Computer Should Think". In: *Contemporary aspects of philosophy*. Ed. by Gilbert Ryle. Oriel Press, 1977.
- [23] Johan van Benthem. "Questions About Quantifiers". In: J. Symb. Log. 49.2 (1984), pp. 443–466. URL: http://www.jstor.org/stable/2274176 (visited on 10/24/2024).
- [24] Patrick Blackburn, Maarten de Rijke, and Yde Venema. *Modal logic*. Vol. 53. Cambridge Tracts in Theoretical Computer Science. Cambridge University Press, Cambridge, 2001. URL: https://doi.org/10. 1017/CB09781107050884.
- [25] Justin Bledin. "Logic informed". In: *Mind* 123.490 (2014), pp. 277–316.
 URL: https://doi.org/10.1093/mind/fzu073.
- [26] Oliver Bott, Fabian Schlotterbeck, and Udo Klein. "Empty-set effects in quantifier interpretation". In: *Journal of Semantics* 36.1 (2019), pp. 99– 163. URL: https://doi.org/10.1093/jos/ffy015.
- [27] Kai Brünnler. "Deep Inference and Symmetry in Classical Proofs". PhD thesis. Technische Universität Dresden, 2003. URL: https://people. bath.ac.uk/ag248/kai/phd.pdf.
- [28] Kai Brünnler. "Locality for Classical Logic". In: Notre Dame J. Form. Log. 47.4 (2006), pp. 557–580. URL: https://doi.org/10.1305/ ndjfl/1168352668.
- [29] John P. Burgess. "A remark on Henkin sentences and their contraries". In: Notre Dame J. Formal Logic 44.3 (2003), pp. 185–188. URL: https: //doi.org/10.1305/ndjf1/1091030856.
- [30] Marco A. Casanova, Ronald Fagin, and Christos H. Papadimitriou. "Inclusion dependencies and their interaction with functional dependencies". In: *Journal of Computer and System Sciences* 28.1 (1984), pp. 29– 59. URL: https://doi.org/10.1016/0022-0000(84)90075-8.

- [31] Jinsheng Chen and Minghui Ma. "Labelled Sequent Calculus for Inquisitive Logic". In: Proceedings of the 6th International Workshop on Logic, Rationality, and Interaction (LORI). Ed. by Jeremy Seligman Alexandru Baltag and Tomoyuki Yamada. Vol. 10455. Lecture Notes in Computer Science. Berlin: Springer, 2017, pp. 526–540. URL: https: //doi.org/10.1007/978-3-662-55665-8_36.
- [32] Ivano Ciardelli. "Inquisitive Semantics and Intermediate Logics". MSc thesis. University of Amsterdam, 2009. URL: https://eprints. illc.uva.nl/id/eprint/818/.
- [33] Ivano Ciardelli. "Dependency as Question Entailment". In: Dependence Logic: Theory and Applications. Ed. by Samson Abramsky, Juha Kontinen, Jouko Väänänen, and Heribert Vollmer. Cham: Springer International Publishing, 2016, pp. 129–181. URL: https://doi.org/10. 1007/978-3-319-31803-5_8.
- [34] Ivano Ciardelli. "Questions in Logic". PhD thesis. University of Amsterdam, 2016. URL: https://hdl.handle.net/11245/1.518411.
- [35] Ivano Ciardelli. Inquisitive logic—consequence and inference in the realm of questions. Vol. 60. Trends in Logic—Studia Logica Library. Springer, Cham, 2022. URL: https://doi.org/10.1007/978-3-031-09706-5.
- [36] Ivano Ciardelli. Inquisitive Neighborhood Logic. 2024. arXiv: 2411.04031 [math.LO]. URL: https://arxiv.org/abs/2411.04031.
- [37] Ivano Ciardelli and Fausto Barbero. "Undefinability in Inquisitive Logic with Tensor". In: Logic, Rationality, and Interaction. Ed. by Patrick Blackburn, Emiliano Lorini, and Meiyun Guo. Berlin, Heidelberg: Springer Berlin Heidelberg, 2019, pp. 29–42. URL: https://doi.org/ 10.1007/978-3-662-60292-8_3.
- [38] Ivano Ciardelli, Jeroen Groenendijk, and Floris Roelofsen. Inquisitive Semantics. Oxford University Press, 2018. URL: https://doi.org/ 10.1093/oso/9780198814788.001.0001.
- [39] Ivano Ciardelli, Rosalie Iemhoff, and Fan Yang. "Questions and dependency in intuitionistic logic". In: *Notre Dame J. Form. Log.* 61.1 (2020), pp. 75–115. URL: https://doi.org/10.1215/00294527-2019-0033.
- [40] Ivano Ciardelli and Martin Otto. "Inquisitive bisimulation". In: J. Symb. Log. 86.1 (2021), pp. 77–109. URL: https://doi.org/10.1017/jsl. 2020.77.
- [41] Ivano Ciardelli and Floris Roelofsen. "Inquisitive logic". In: J. Philos. Logic 40.1 (2011), pp. 55–94. URL: https://doi.org/10.1007/ s10992-010-9142-6.

- [42] Ivano A. Ciardelli and Floris Roelofsen. "Inquisitive dynamic epistemic logic". In: Synthese 192.6 (2015), pp. 1643–1687. URL: https://doi. org/10.1007/s11229-014-0404-7.
- [43] Giovanna D'Agostino. "Uniform interpolation for propositional and modal team logics". In: J. Logic Comput. 29.5 (2019), pp. 785–802. URL: https://doi.org/10.1093/logcom/exz006.
- [44] Francien Dechesne. "Game, set, maths : formal investigations into logic with imperfect information". PhD thesis. Tilburg University, 2005. URL: https://pure.tue.nl/ws/portalfiles/portal/2434476/ 608532584011891.pdf.
- [45] Marco Degano, Sonia Ramotowska, Paul Marty, Maria Aloni, Richard Breheny, Jacopo Romoli, and Yasutada Sudo. "The Ups and Downs of Ignorance". Manuscript. 2024. URL: https://ling.auf.net/ lingbuzz/007389.
- [46] J. Michael Dunn. "Intuitive semantics for first-degree entailments and 'coupled trees'". In: *Philos. Studies* 29.3 (1976), pp. 149–168. URL: https://doi.org/10.1007/bf00373152.
- [47] Johannes Ebbing, Lauri Hella, Arne Meier, Julian-Steffen Müller, Jonni Virtema, and Heribert Vollmer. "Extended Modal Dependence Logic EMDL". In: Logic, Language, Information, and Computation. Ed. by Leonid Libkin, Ulrich Kohlenbach, and Ruy de Queiroz. Berlin, Heidelberg: Springer Berlin Heidelberg, 2013, pp. 126–137. URL: https://doi.org/10.1007/978-3-642-39992-3_13.
- [48] Herbert B. Enderton. "Finite partially-ordered quantifiers". In: Z. Math. Logik Grundlagen Math. 16 (1970), pp. 393–397. URL: https://doi. org/10.1002/malq.19700160802.
- [49] Kit Fine. "Truthmaker Semantics". In: A Companion to the Philosophy of Language. Ed. by Bob Hale, Crispin Wright, and Alexander Miller. Wiley, 2017, pp. 556–577. URL: https://doi.org/10.1002/ 9781118972090.ch22.
- [50] Sabine Frittella, Giuseppe Greco, Alessandra Palmigiano, and Fan Yang.
 "A Multi-type Calculus for Inquisitive Logic". In: Logic, Language, Information, and Computation. Ed. by Jouko Väänänen, Åsa Hirvonen, and Ruy de Queiroz. Lecture Notes in Computer Science. Berlin, Heidelberg: Springer, 2016, pp. 215–233. URL: https://doi.org/10.1007/978-3-662-52921-8_14.
- [51] Pietro Galliani. "Inclusion and exclusion dependencies in team semantics — On some logics of imperfect information". In: Ann. Pure Appl. Logic 163.1 (2012), pp. 68–84. URL: https://doi.org/10.1016/j. apal.2011.08.005.

- [52] Pietro Galliani. "Epistemic Operators in Dependence Logic". In: *Studia Logica* 101.2 (2013), pp. 367–397. URL: https://doi.org/10.1007/s11225-013-9478-3.
- [53] Pietro Galliani. "Strongly First Order Dependencies and Dual Negation in Team Semantics". Talk presented at Dagstuhl Seminar 24111 Logics for Dependence and Independence: Expressivity and Complexity. 2024.
- [54] Pietro Galliani and Lauri Hella. "Inclusion Logic and Fixed Point Logic". In: Leibniz International Proceedings in Informatics, LIPIcs 23 (2013), pp. 281–295. URL: https://doi.org/10.4230/LIPIcs. CSL.2013.281.
- [55] Peter G\u00e4rdenfors. Conceptual Spaces: The Geometry of Thought. The MIT Press, 2000. URL: https://doi.org/10.7551/mitpress/ 2076.001.0001.
- [56] Valentin Goranko and Martin Otto. "Model theory of modal logic". In: *Handbook of modal logic*. Vol. 3. Stud. Log. Pract. Reason. Elsevier B. V., Amsterdam, 2007, pp. 249–329. URL: https://doi.org/10. 1016/S1570-2464(07)80008-5.
- [57] Erich Grädel. "Games for Inclusion Logic and Fixed-Point Logic". In: Dependence Logic: Theory and Applications. Ed. by Samson Abramsky, Juha Kontinen, Jouko Väänänen, and Heribert Vollmer. Cham: Springer International Publishing, 2016, pp. 73–98. URL: https://doi.org/ 10.1007/978-3-319-31803-5_5.
- [58] Erich Grädel and Jouko Väänänen. "Dependence, independence, and incomplete information". In: *Database theory—ICDT 2012*. ACM, New York, 2011, pp. 1–17.
- [59] Erich Grädel and Jouko Väänänen. "Dependence and Independence".
 In: Studia Logica: An International Journal for Symbolic Logic 101.2 (2013), pp. 399–410. URL: https://www.jstor.org/stable/23488329.
- [60] Jeroen Groenendijk. "Inquisitive Semantics: Two Possibilities for Disjunction". In: Logic, Language, and Computation. Ed. by Peter Bosch, David Gabelaia, and Jérôme Lang. Berlin, Heidelberg: Springer Berlin Heidelberg, 2009, pp. 80–94. URL: https://doi.org/10.1007/978-3-642-00665-4_8.
- [61] Jeroen Groenendijk and Martin Stokhof. "Dynamic predicate logic". In: Linguistics and Philosophy 14.1 (1991), pp. 39–100. URL: https:// doi.org/10.1007/BF00628304.
- [62] Alessio Guglielmi. "A system of interaction and structure". In: ACM Trans. Comput. Logic 8.1 (Jan. 2007), 1–es. URL: https://doi.org/ 10.1145/1182613.1182614.

- [63] Matilda Häggblom. "Axiomatizing Modal Inclusion Logic". MSc thesis. University of Helsinki, 2022. URL: http://hdl.handle.net/10138/ 347694.
- [64] Miika Hannula. "Axiomatizing first-order consequences in independence logic". In: Ann. Pure Appl. Logic 166.1 (2015), pp. 61–91. URL: https://doi.org/10.1016/j.apal.2014.09.002.
- [65] Miika Hannula and Juha Kontinen. "A finite axiomatization of conditional independence and inclusion dependencies". In: *Information and Computation* 249 (2016), pp. 121–137. URL: https://doi.org/10. 1016/j.ic.2016.04.001.
- [66] Miika Hannula, Juha Kontinen, and Sebastian Link. "On the finite and general implication problems of independence atoms and keys". In: *Journal of Computer and System Sciences* 82.5 (2016), pp. 856–877. URL: https://doi.org/10.1016/j.jcss.2016.02.007.
- [67] Peter Hawke and Shane Steinert-Threlkeld. "Semantic Expressivism for Epistemic Modals". In: *Linguistics and Philosophy* 44.2 (2020), pp. 475–511. URL: https://doi.org/10.1007/s10988-020-09295-7.
- [68] Irene Heim. "File Change Semantics and the Familiarity Theory of Definiteness". In: *Meaning, Use, and Interpretation of Language*. Ed. by Rainer Bäuerle, Christoph Schwarze, and Arnim von Stechow. Berlin, Boston: De Gruyter, 1983, pp. 164–189. URL: https://doi.org/10. 1515/9783110852820.164.
- [69] Lauri Hella, Antti Kuusisto, Arne Meier, and Jonni Virtema. "Model checking and validity in propositional and modal inclusion logics". In: *Journal of Logic and Computation* 29.5 (2019), pp. 605–630. URL: https://doi.org/10.1093/logcom/exz008.
- [70] Lauri Hella, Antti Kuusisto, Arne Meier, and Heribert Vollmer. "Satisfiability of Modal Inclusion Logic: Lax and Strict Semantics". In: ACM Trans. Comput. Logic 21.1 (2019). URL: https://doi.org/10.1145/ 3356043.
- [71] Lauri Hella, Kerkko Luosto, Katsuhiko Sano, and Jonni Virtema. "The expressive power of modal dependence logic". In: Advances in modal logic. Vol. 10. Coll. Publ., London, 2014, pp. 294–312. URL: http://www.aiml.net/volumes/volume10/Hella-Luosto-Sano-Virtema.pdf.
- [72] Lauri Hella, Kerkko Luosto, and Jouko Väänänen. "Dimension in team semantics". In: *Mathematical Structures in Computer Science* (2024), pp. 1–45. URL: https://doi.org/10.1017/S0960129524000021.

- [73] Lauri Hella and Johanna Stumpf. "The expressive power of modal logic with inclusion atoms". In: *Proceedings Sixth International Symposium* on Games, Automata, Logics and Formal Verification. Vol. 193. Electron. Proc. Theor. Comput. Sci. (EPTCS). EPTCS, 2015, pp. 129–143. URL: https://doi.org/10.4204/EPTCS.193.10.
- [74] Leon Henkin. "Some remarks on infinitely long formulas". In: Infinitistic Methods (Proc. Sympos. Foundations of Math., Warsaw, 1959).
 Pergamon, Oxford-London-New York-Paris, 1961, pp. 167–183. URL: https://doi.org/10.2307/2270594.
- [75] Jaakko Hintikka. The principles of mathematics revisited. With an appendix by Gabriel Sandu. Cambridge University Press, Cambridge, 1996. URL: https://doi.org/10.1017/CB09780511624919.
- [76] Jaakko Hintikka. "Negation in Logic and in Natural Language". In: Linguistics and Philosophy 25.5–6 (2002), pp. 585–600. URL: https:// doi.org/10.1023/a:1020895229197.
- [77] Jaakko Hintikka and Gabriel Sandu. "Informational independence as a semantical phenomenon". In: Logic, methodology and philosophy of science, VIII (Moscow, 1987). Vol. 126. Stud. Logic Found. Math. North-Holland, Amsterdam, 1989, pp. 571–589. URL: https://doi.org/ 10.1016/S0049-237X(08)70066-1.
- [78] Wilfrid Hodges. "Compositional semantics for a language of imperfect information". In: Log. J. IGPL 5.4 (1997), pp. 539–563. URL: https: //doi.org/10.1093/jigpal/5.4.539.
- [79] Wilfrid Hodges. "Some strange quantifiers". In: Structures in logic and computer science. Vol. 1261. Lecture Notes in Comput. Sci. Springer, Berlin, 1997, pp. 51–65. URL: https://doi.org/10.1007/3-540-63246-8_4.
- [80] Matthias Hoelzel and Richard Wilke. "On the Union Closed Fragment of Existential Second-Order Logic and Logics with Team Semantics". In: Logical Methods in Computer Science 17.3 (2021), 14:1–14:32. URL: http://doi.org/10.46298/lmcs-17(3:14)2021.
- [81] Lloyd Humberstone. "From worlds to possibilities". In: J. Philos. Logic 10.3 (1981), pp. 313–399. URL: https://doi.org/10.1007/BF00293423.
- [82] Lloyd Humberstone. "Supervenience, dependence, disjunction". In: Log. Log. Philos. 28.1 (2019), pp. 3–135. URL: https://doi.org/ 10.12775/llp.2018.007.
- [83] Tapani Hyttinen, Gianluca Paolini, and Jouko Vä"anänen. "Quantum Team Logic and Bell?s Inequalities". In: *Review of Symbolic Logic* 8.4 (2015), pp. 722–742. URL: https://10.1017/s1755020315000192.

- [84] Rosalie Iemhoff. "Uniform interpolation and sequent calculi in modal logic". In: Arch. Math. Logic 58.1 (2019), pp. 155–181. URL: https: //doi.org/10.1007/s00153-018-0629-0.
- [85] Rosalie Iemhoff. "Uniform interpolation and the existence of sequent calculi". In: Ann. Pure Appl. Logic 170.11 (2019), p. 102711. URL: https://doi.org/10.1016/j.apal.2019.05.008.
- [86] Luca Incurvati and Julian J. Schlöder. "Weak Rejection". In: Australasian Journal of Philosophy 95.4 (2017), pp. 741–760. URL: https: //doi.org/10.1080/00048402.2016.1277771.
- [87] Hans Kamp. "Free choice permission". In: Proc. Aristotelian Soc. (N.S.) 74 (1973/74), pp. 57–74. URL: https://doi.org/10.1093/ aristotelian/74.1.57.
- [88] Søren Brinck Knudstorp. *The Expressive Completeness of Bilateral State-based Modal Logic*. Manuscript. 2023.
- [89] Juha Kontinen, Julian-Steffen Müller, Henning Schnoor, and Heribert Vollmer. "A van Benthem theorem for modal team semantics". In: 24th EACSL Annual Conference on Computer Science Logic. Vol. 41. LIPIcs. Leibniz Int. Proc. Inform. Schloss Dagstuhl. Leibniz-Zent. Inform., Wadern, 2015, pp. 277–291. URL: https://doi.org/10.4230/ LIPIcs.CSL.2015.277.
- [90] Juha Kontinen, Julian-Steffen Müller, Henning Schnoor, and Heribert Vollmer. "Modal independence logic". In: J. Logic Comput. 27.5 (2017), pp. 1333–1352. URL: https://doi.org/10.1093/logcom/exw019.
- [91] Juha Kontinen and Ville Nurmi. "Team Logic and Second-Order Logic". In: Fundam. Inf. 106.2-4 (2011), pp. 259-272. URL: https://www. researchgate.net/publication/220444491_Team_Logic_and_ Second-Order_Logic.
- [92] Juha Kontinen, Max Sandström, and Jonni Virtema. Set Semantics for Asynchronous TeamLTL: Expressivity and Complexity. 2023. arXiv: 2304.10915 [cs.L0]. URL: https://arxiv.org/abs/2304.10915.
- [93] Juha Kontinen and Jouko Väänänen. "On definability in dependence logic". In: J. Log. Lang. Inf. 18.3 (2009), pp. 317–332. URL: https: //doi.org/10.1007/s10849-009-9082-0.
- [94] Juha Kontinen and Jouko Väänänen. "A remark on negation in dependence logic". In: *Notre Dame J. Form. Log.* 52.1 (2011), pp. 55–65. URL: https://doi.org/10.1215/00294527-2010-036.
- [95] Juha Kontinen and Fan Yang. "Complete Logics for Elementary Team Properties". In: J. Symb. Log. 88.2 (2023), pp. 579–619. URL: https: //doi.org/10.1017/jsl.2022.80.

- [96] Martin Lück. "Axiomatizations of team logics". In: Ann. Pure Appl. Logic 169.9 (2018), pp. 928–969. URL: https://doi.org/10.1016/ j.apal.2018.04.010.
- [97] Martin Lück. "Team Logic: Axioms, Expressiveness, Complexity". PhD thesis. Leibniz University Hannover, 2020. URL: https://doi.org/ 10.15488/9376.
- [98] Allen L. Mann, Gabriel Sandu, and Merlijn Sevenster. Independencefriendly logic. Vol. 386. London Mathematical Society Lecture Note Series. A game-theoretic approach. Cambridge University Press, Cambridge, 2011, pp. vi+208. URL: https://doi.org/10.1017/ CB09780511981418.
- [99] Julian-Steffen Müller. "Satisfiability and Model Checking in Team Based Logics". PhD thesis. Leibniz University Hannover, 2014.
- [100] Valentin Müller. "On the proof theory of inquisitive logic". MA thesis. University of Amsterdam, The Netherlands, 2022. URL: https: //eprints.illc.uva.nl/id/eprint/2278/1/MoL-2023-26. text.pdf.
- [101] Karl Nygren. "Free choice in modal inquisitive logic". In: J. Philos. Logic 52.2 (2023), pp. 347–391. URL: https://doi.org/10.1007/ s10992-022-09674-4.
- [102] Eric Pacuit and Fan Yang. "Dependence and Independence in Social Choice: Arrow's Theorem". English. In: *Dependence Logic*. Germany: Springer, June 2016, pp. 235–260. URL: https://doi.org/10.1007/ 978-3-319-31803-5_11.
- [103] Andrew M. Pitts. "On an interpretation of second order quantification in first order intuitionistic propositional logic". In: J. Symb. Log. 57.1 (1992), pp. 33–52. URL: https://doi.org/10.2307/2275175.
- [104] Huw Price. "Sense, assertion, Dummett and denial". In: *Mind* 92.366 (1983), pp. 161–173. URL: https://doi.org/10.1093/mind/XCII. 366.161.
- [105] Huw Price. "Why "not"?" In: Mind 99.394 (1990), pp. 221–238. URL: https://doi.org/10.1093/mind/XCIX.394.221.
- [106] Joni Puljujärvi and Davide Emilio Quadrellaro. "Compactness in team semantics". In: *Mathematical Logic Quarterly* (2024). URL: https:// doi.org/10.1002/malq.202200072.
- [107] Vít Punčochář. "Weak negation in inquisitive semantics". In: J. Log. Lang. Inf. 24.3 (2015), pp. 323–355. URL: https://doi.org/10. 1007/s10849-015-9219-2.

- [108] Greg Restall. "Multiple Conclusions". In: Logic, Methodology and Philosophy of Science. Ed. by Petr Hájek, Luis Valdés-Villanueva, and Dag Westerståhl. College Publications, 2005, pp. 189–205. URL: https:// consequently.org/papers/multipleconclusions.pdf.
- [109] Raine Rönnholm. Arity Fragments of Logics with Team Semantics. Acta Electronica Universitatis Tamperensis 1955. Tampere University Press, 2018. URL: https://trepo.tuni.fi/bitstream/handle/10024/ 104699/978-952-03-0912-1.pdf.
- [110] Ian Rumfitt. ""Yes" and "No"". In: *Mind* 109.436 (2000), pp. 781–823.
 URL: https://doi.org/10.1093/mind/109.436.781.
- [111] Katsuhiko Sano. "Sound and Complete Tree-Sequent Calculus for Inquisitive Logic". In: Logic, Language, Information and Computation. Ed. by Hiroakira Ono, Makoto Kanazawa, and Ruy de Queiroz. Berlin, Heidelberg: Springer Berlin Heidelberg, 2009, pp. 365–378. URL: https://doi.org/10.1007/978-3-642-02261-6_29.
- [112] Katsuhiko Sano and Jonni Virtema. "Axiomatizing Propositional Dependence Logics". In: 24th EACSL Annual Conference on Computer Science Logic (CSL 2015). Ed. by Stephan Kreutzer. Vol. 41. Leibniz International Proceedings in Informatics (LIPIcs). Dagstuhl, Germany: Schloss Dagstuhl Leibniz-Zentrum für Informatik, 2015, pp. 292–307. URL: https://drops.dagstuhl.de/entities/document/10. 4230/LIPIcs.CSL.2015.292.
- [113] Peter Schroeder-Heister. "Proof-Theoretic Semantics". In: *The Stanford Encyclopedia of Philosophy*. Ed. by Edward N. Zalta and Uri Nodelman. Fall 2023. Metaphysics Research Lab, Stanford University, 2023. URL: https://plato.stanford.edu/archives/fall2023/entries/proof-theoretic-semantics/.
- [114] Merlijn Sevenster. "Model-theoretic and computational properties of modal dependence logic". In: J. Logic Comput. 19.6 (2009), pp. 1157– 1173. URL: https://doi.org/10.1093/logcom/exn102.
- [115] Timothy Smiley. "Rejection". In: Analysis (Oxford) 56.1 (1996), pp. 1–
 9. URL: https://doi.org/10.1111/j.0003-2638.1996.00001.x.
- [116] Robert Stalnaker. "Assertion". In: *Syntax and Semantics 9: Pragmatics*. New York: New York Academic Press, 1978, pp. 315–332.
- [117] Martin Stokhof, Jeroen Groenendijk, and Frank Veltman. "Coreference and Modality". In: *The Handbook of Contemporary Semantic Theory*. Ed. by Shalom Lappin. Blackwell Reference, 1996, pp. 179–216.
- [118] A. S. Troelstra and Helmut Schwichtenberg. Basic Proof Theory. 2nd ed. New York: Cambridge University Press, 2000. URL: https://doi. org/10.1017/CB09781139168717.

- [119] Jouko Väänänen. Dependence logic. Vol. 70. London Mathematical Society Student Texts. A new approach to independence friendly logic. Cambridge University Press, Cambridge, 2007, pp. x+225. URL: https://doi.org/10.1017/CB09780511611193.
- [120] Jouko Väänänen. "Modal dependence logic". In: New perspectives on games and interaction. Vol. 4. Texts Log. Games. Amsterdam Univ. Press, Amsterdam, 2008, pp. 237–254. URL: https://hdl.handle. net/11245/1.300509.
- [121] Jouko Väänänen. "Multiverse set theory and absolutely undecidable propositions". In: *Interpreting Gödel*. Cambridge Univ. Press, Cambridge, 2014, pp. 180–208. URL: https://doi.org/10.1017/ CB09780511756306.013.
- [122] Bas C Van Fraassen. "Facts and Tautological Entailments". In: *The Journal of Philosophy* 66.15 (1969), pp. 477–487. URL: https://doi.org/10.2307/2024563.
- [123] Frank Veltman. "Defaults in Update Semantics". In: J. Philos. Logic 25.3 (1996), pp. 221–261. URL: https://doi.org/10.1007/BF00248150.
- [124] Wilbur John Walkoe Jr. "Finite partially-ordered quantification". In: J. Symbolic Logic 35 (1970), pp. 535–555. URL: https://doi.org/10.2307/2271440.
- [125] Heinrich Wansing. "Proofs, disproofs, and their duals". In: Advances in modal logic. Volume 8. Coll. Publ., London, 2010, pp. 483–505. URL: http://www.aiml.net/volumes/volume8/Wansing.pdf.
- [126] Heinrich Wansing and Sara Ayhan. "Logical multilateralism". In: J. Philos. Logic 52.6 (2023), pp. 1603–1636. URL: https://doi.org/10. 1007/s10992-023-09720-9.
- [127] Malte Willer. "Simplifying with free choice". In: *Topoi* 37.3 (2018), pp. 379–392. URL: https://doi.org/10.1007/s11245-016-9437-5.
- [128] Georg Henrik von Wright. "An essay in deontic logic and the general theory of action. With a bibliography of deontic and imperative logic". In: *Acta Philos. Fenn.* 21 (1968), p. 110.
- [129] Seth Yalcin. "Epistemic Modals". In: Mind 116.464 (2007), pp. 983– 1026. URL: https://doi.org/10.1093/mind/fzm983.
- [130] Seth Yalcin. "Nonfactualism about Epistemic Modality". In: *Epistemic Modality*. Oxford University Press, 2011, pp. 295–332. URL: https://doi.org/10.1093/acprof:oso/9780199591596.003.0011.

- [131] Seth Yalcin. "Semantics and Metasemantics in the Context of Generative Grammar". In: *Metasemantics: New Essays on the Foundations* of Meaning. Oxford University Press, 2014, pp. 17–54. URL: https: //doi.org/10.1093/acprof:oso/9780199669592.003.0002.
- [132] Fan Yang. "Expressing Second-order Sentences in Intuitionistic Dependence Logic". In: *Studia Logica* 101.2 (2013), pp. 323–342. URL: https://doi.org/10.1007/s11225-013-9476-5.
- [133] Fan Yang. "On Extensions and Variants of Dependence Logic: A study of intuitionistic connectives in the team semantics setting". PhD thesis. University of Helsinki, 2014. URL: https://helda.helsinki.fi/ items/0b455380-1a6b-47e5-99bc-910c78e06f37.
- [134] Fan Yang. "Modal dependence logics: axiomatizations and modeltheoretic properties". In: Log. J. IGPL 25.5 (2017), pp. 773–805. URL: https://doi.org/10.1093/jigpal/jzx023.
- [135] Fan Yang. "Uniform definability in propositional dependence logic". In: *Review of Symbolic Logic* 10.1 (2017), pp. 65–79. URL: https://doi. org/10.1017/s1755020316000459.
- [136] Fan Yang. "Axiomatizing first order consequences in inclusion logic". In: Mathematical Logic Quarterly 66.2 (2020), pp. 195–216. URL: https://doi.org/10.1002/malq.201900031.
- [137] Fan Yang. "Propositional union closed team logics". In: Ann. Pure Appl. Logic 173.6 (2022), Paper No. 103102, 35. URL: https://doi.org/ 10.1016/j.apal.2022.103102.
- [138] Fan Yang. "There are (other) ways to negate in propositional team semantics". In: *Exploring Negation, Modality and Proof (to appear)*. Logic in Asia: Studia Logica Library. 2025. arXiv: 2410.08413 [math.LO]. URL: https://arxiv.org/abs/2410.08413.
- [139] Fan Yang and Jouko Väänänen. "Propositional logics of dependence". In: Ann. Pure Appl. Logic 167.7 (2016), pp. 557–589. URL: https: //doi.org/10.1016/j.apal.2016.03.003.
- [140] Fan Yang and Jouko Väänänen. "Propositional team logics". In: Ann. Pure Appl. Logic 168.7 (2017), pp. 1406–1441. URL: https://doi. org/10.1016/j.apal.2017.01.007.
Samenvatting

Dit proefschrift bundelt vijf artikelen die zich voornamelijk richten op de expressieve kracht en axiomatisaties van propositionele en modale logica's met teamsemantiek. De meeste van deze artikelen richten zich op omgevingen waarin de lege-team-eigenschap niet geldt—er zijn formules die niet waar zijn in het lege team—en waarin we feiten kunnen uitdrukken en kunnen redeneren over de leegheid en niet-leegheid van teams in de objecttaal.

Het eerste artikel bestudeert de wiskundige eigenschappen van de bilaterale stategebaseerde modale logica (BSML), een modale teamlogica die is gebruikt om vrije keuzewijzen en gerelateerde linguïstische fenomenen te verklaren. Deze logica breidt de klassieke modale logica uit met een niet-leegheid-atoom NE dat waar is in een team dan en slechts dan als het team niet leeg is. In dit artikel introduceren we twee uitbreidingen van BSML, tonen we aan dat de uitbreidingen expressief compleet zijn, en ontwikkelen we axiomatisaties in natuurlijke-deductiesystemen voor de drie logica's.

In het tweede artikel bewijzen we expressieve volledigheidsresultaten voor convexe propositionele en modale teamlogica's, waarbij een logica convex is als voor elke formule geldt: als de formule waar is in twee teams, dan is de formule ook waar in alle teams tussen deze twee teams met betrekking tot de deelverzamelingrelatie. Convexiteit is een natuurlijke uitbreiding van de neerwaartse afsluiting naar een omgeving waarin de lege team-eigenschap niet geldt. We introduceren meerdere propositionele/modale logica's die expressief compleet zijn voor de klasse van alle convexe propositionele/modale teameigenschappen. We lossen ook een probleem op dat in het eerste artikel open was gebleven met betrekking tot de expressieve kracht van BSML en zijn propositionele fragment: we tonen aan dat dit propositionele fragment expressief compleet is voor de klasse van alle convexe en propositionele teameigenschappen die gesloten zijn onder vereniging, waarbij een modale analogie van dit resultaat een expressieve volledigheidsstelling voor BSML oplevert. We introduceren ook een generalisatie van uniforme definieerbaarheid en definiëren verschillende begrippen van uitbreiding door gebruik te maken van deze generalisatie om het gevoel te verduidelijken waarin onze nieuwe propositionele convexe logica's de propositionele afhankelijkheidslogica en propositionele *inquisitive logic* (propositionele logica die zich o.a. bezighoudt met de betekenis van vragen) uitbreiden.

In het derde artikel bestuderen we de eigenschappen van de negatie die in BSML wordt gebruikt, de bilaterale negatie ¬. Dit is in wezen dezelfde notie als de duale of speltheoretische negatie van onafhankelijkheidsvriendelijke logica (IF) en afhankelijkheidslogica (D). In IF en D vertoont de duale negatie een extreem hoge graad van semantische onbepaaldheid, aangezien voor elk paar zinnen φ en ψ van IF/D geldt dat: als φ en ψ onverenigbaar zijn in de zin dat ze geen modellen gemeen hebben, er een zin θ van IF/D bestaat zodat $\varphi \equiv \theta$ en $\psi \equiv \neg \theta$ (zoals oorspronkelijk aangetoond door Burgess in de equivalente context van het prenex-fragment van Henkin-kwantorlogica). We tonen aan dat door het begrip van onverenigbaarheid aan te passen (waardoor in sommige gevallen begrippen van onverenigbaarheid worden gegenereerd die beter geschikt zijn voor omgevingen waarin de lege team-eigenschap en neerwaartse afsluiting niet gelden dan de begrippen die uit neerwaarts gesloten omgevingen worden geïmporteerd), analogieën van dit resultaat voor een aantal modale en propositionele teamlogica's kunnen worden vastgesteld, waaronder BSML, de semantisch expressivistische logica van Hawke en Steinert-Threlkeld voor epistemische modaliteiten, evenals de propositionele afhankelijkheidslogica met de duale negatie. Samen met het omgekeerde van dit resultaat kan dit type resultaat worden gezien als een expressieve volledigheidsstelling met betrekking tot het relevante begrip van onverenigbaarheid; we formuleren een begrip van expressieve volledigheid voor paren van eigenschappen om dit precies te maken.

In het vierde artikel schakelen we over naar een omgeving die wel de lege teameigenschap heeft, maar waarin neerwaartse afsluiting niet geldt. We bieden een volledige axiomatisatie van de modale inclusielogica – een teamgebaseerde modale logica uitgebreid met inclusie-atomen. We herzien en verfijnen een expressieve volledigheids- en normaalvormstelling voor de logica, definiëren een natuurlijkedeductie-bewijssysteem en gebruiken de normaalvorm om de volledigheid van de axiomatisatie te bewijzen. Volledige axiomatisaties worden ook gegeven voor twee andere uitbreidingen van de modale logica met dezelfde expressieve kracht als modale inclusielogica: de een verrijkt met een 'might'-operator en de andere met een enkelewereldvariant van de 'might'-operator.

In het vijfde artikel introduceren we een sequentencalculus voor de propositionele teamlogica met zowel de gesplitste disjunctie als de *inquisitive* disjunctie, bestaande uit een G3-systeem voor de klassieke propositionele logica samen met *deep-inference* regels voor de *inquisitive* disjunctie. We tonen aan dat het systeem verschillende wenselijke eigenschappen heeft: het staat hoogte-behoudende verzwakking, contractie en inversie toe; het ondersteunt een procedure voor het construeren van snedevrije bewijzen en tegenmodellen, vergelijkbaar met die voor G3cp; en snede-eliminatie geldt als een corollarium van snede-eliminatie voor het G3-subsysteem samen met een normaalvormstelling voor snedevrije afleidingen in het systeem.

Abstract

This dissertation collects together five papers which focus primarily on the expressive power, axiomatizations, and proof theory of propositional and modal logics with team semantics. Most of these papers focus on settings in which the empty team property fails—that is, there are formulas which are not true in the empty team—and hence in which we are able to express facts and reason about the emptiness and nonemptiness of teams in the object language.

The first paper studies the mathematical properties of bilateral state-based modal logic (BSML), a modal team logic which has been used to account for free choice inferences and related linguistic phenomena. This logic extends classical modal logic with a nonemptiness atom NE which is true in a team if and only if the team is nonempty. In this paper, we introduce two extensions of BSML, show that the extensions are expressively complete, and develop natural deduction axiomatizations for the three logics.

In the second paper, we prove expressive completeness results for convex propositional and modal team logics, where a logic is convex if, for each of its formulas, if the formula is true in two teams, then it is also true in all the teams between these two teams with respect to set inclusion. Convexity is a natural generalization of the well-known property of downward closure to a setting in which the empty team property fails. We introduce multiple propositional/modal logics which are expressively complete for the class of all convex propositional/modal team properties. We also solve a problem that was left open in the first paper concerning the expressive power of BSML as well as its propositional fragment: we show that BSML is expressively complete for the class of all convex and union-closed modal team properties invariant under bounded bisimulation. We also introduce a generalization of uniform definability and define distinct notions of extension making use of this generalization in order to clarify the sense in which our novel propositional convex logics extend propositional dependence logic and propositional inquisitive logic.

In the third paper, we study the properties of the negation employed in BSML, the bilateral negation \neg . This is essentially the same notion as the dual or game-theoretical negation of independence-friendly logic (IF) and dependence logic (D). In IF and D,

the dual negation exhibits an extreme degree of semantic indeterminacy in that for any pair of sentences φ and ψ of IF/D, if φ and ψ are incompatible in the sense that they share no models, there is a sentence θ of IF/D such that $\varphi \equiv \theta$ and $\psi \equiv \neg \theta$ (as shown originally by Burgess in the equivalent context of the prenex fragment of Henkin quantifier logic). We show that by adjusting the notion of incompatibility employed (thus generating, in some cases, notions of incompatibility more suitable for settings in which the empty team property and downward closure fail than the notions imported from downward-closed settings), analogues of this result can be established for a number of modal and propositional team logics, including BSML and propositional dependence logic with the dual negation. Together with its converse, a result of this type can be seen as an expressive completeness theorem with respect to the relevant incompatibility notion; we formulate a notion of expressive completeness for pairs of properties to make this precise.

In the fourth paper, we move to a setting which does have the empty team property, but in which downward closure fails. We provide a complete axiomatization of modal inclusion logic—team-based modal logic extended with inclusion atoms. We review and refine an expressive completeness and normal form theorem for the logic, define a natural deduction proof system, and use the normal form to prove completeness of the axiomatization. Complete axiomatizations are also provided for two other extensions of modal logic with the same expressive power as modal inclusion logic: one augmented with a might-operator and the other with a single-world variant of the might-operator.

In the fifth paper, we introduce a sequent calculus for the propositional team logic with both the split disjunction and the inquisitive disjunction. The sequent calculus consists of a G3-style system for classical propositional logic together with deep-inference rules for the inquisitive disjunction. We show that the system satisfies various desirable properties: it admits height-preserving weakening, contraction and inversion; it supports a procedure for constructing cutfree proofs and countermodels similar to that for G3cp; and cut elimination holds as a corollary of cut elimination for the G3-style subsystem together with a normal form theorem for cutfree derivations in the system.

Titles in the ILLC Dissertation Series:

- ILLC DS-2020-11: Marco Del Tredici Linguistic Variation in Online Communities: A Computational Perspective
- ILLC DS-2020-12: Bastiaan van der Weij Experienced listeners: Modeling the influence of long-term musical exposure on rhythm perception
- ILLC DS-2020-13: **Thom van Gessel** *Questions in Context*
- ILLC DS-2020-14: Gianluca Grilletti Questions & Quantification: A study of first order inquisitive logic
- ILLC DS-2020-15: **Tom Schoonen** *Tales of Similarity and Imagination. A modest epistemology of possibility*
- ILLC DS-2020-16: Ilaria Canavotto Where Responsibility Takes You: Logics of Agency, Counterfactuals and Norms
- ILLC DS-2020-17: Francesca Zaffora Blando Patterns and Probabilities: A Study in Algorithmic Randomness and Computable Learning
- ILLC DS-2021-01: **Yfke Dulek** Delegated and Distributed Quantum Computation
- ILLC DS-2021-02: Elbert J. Booij The Things Before Us: On What it Is to Be an Object
- ILLC DS-2021-03: Seyyed Hadi Hashemi Modeling Users Interacting with Smart Devices
- ILLC DS-2021-04: **Sophie Arnoult** Adjunction in Hierarchical Phrase-Based Translation
- ILLC DS-2021-05: Cian Guilfoyle Chartier A Pragmatic Defense of Logical Pluralism
- ILLC DS-2021-06: **Zoi Terzopoulou** Collective Decisions with Incomplete Individual Opinions
- ILLC DS-2021-07: Anthia Solaki Logical Models for Bounded Reasoners
- ILLC DS-2021-08: Michael Sejr Schlichtkrull Incorporating Structure into Neural Models for Language Processing

ILLC DS-2021-09: **Taichi Uemura** Abstract and Concrete Type Theories

- ILLC DS-2021-10: Levin Hornischer Dynamical Systems via Domains: Toward a Unified Foundation of Symbolic and Non-symbolic Computation
- ILLC DS-2021-11: Sirin Botan Strategyproof Social Choice for Restricted Domains
- ILLC DS-2021-12: Michael Cohen Dynamic Introspection
- ILLC DS-2021-13: **Dazhu Li** Formal Threads in the Social Fabric: Studies in the Logical Dynamics of Multi-Agent Interaction
- ILLC DS-2021-14: Álvaro Piedrafita On Span Programs and Quantum Algorithms
- ILLC DS-2022-01: Anna Bellomo Sums, Numbers and Infinity: Collections in Bolzano's Mathematics and Philosophy
- ILLC DS-2022-02: Jan Czajkowski Post-Quantum Security of Hash Functions
- ILLC DS-2022-03: Sonia Ramotowska Quantifying quantifier representations: Experimental studies, computational modeling, and individual differences
- ILLC DS-2022-04: **Ruben Brokkelkamp** How Close Does It Get?: From Near-Optimal Network Algorithms to Suboptimal Equilibrium Outcomes
- ILLC DS-2022-05: Lwenn Bussière-Carae No means No! Speech Acts in Conflict

ILLC DS-2022-06: **Emma Mojet** *Observing Disciplines: Data Practices In and Between Disciplines in the 19th and Early 20th Centuries*

- ILLC DS-2022-07: Freek Gerrit Witteveen Quantum information theory and many-body physics
- ILLC DS-2023-01: Subhasree Patro Quantum Fine-Grained Complexity

- ILLC DS-2023-02: Arjan Cornelissen Quantum multivariate estimation and span program algorithms
- ILLC DS-2023-03: **Robert Paßmann** Logical Structure of Constructive Set Theories
- ILLC DS-2023-04: Samira Abnar Inductive Biases for Learning Natural Language
- ILLC DS-2023-05: **Dean McHugh** *Causation and Modality: Models and Meanings*
- ILLC DS-2023-06: **Jialiang Yan** Monotonicity in Intensional Contexts: Weakening and: Pragmatic Effects under Modals and Attitudes
- ILLC DS-2023-07: Yiyan Wang Collective Agency: From Philosophical and Logical Perspectives
- ILLC DS-2023-08: Lei Li Games, Boards and Play: A Logical Perspective
- ILLC DS-2023-09: Simon Rey Variations on Participatory Budgeting
- ILLC DS-2023-10: Mario Giulianelli Neural Models of Language Use: Studies of Language Comprehension and Production in Context
- ILLC DS-2023-11: Guillermo Menéndez Turata Cyclic Proof Systems for Modal Fixpoint Logics
- ILLC DS-2023-12: Ned J.H. Wontner Views From a Peak: Generalisations and Descriptive Set Theory
- ILLC DS-2024-01: Jan Rooduijn Fragments and Frame Classes: Towards a Uniform Proof Theory for Modal Fixed Point Logics
- ILLC DS-2024-02: **Bas Cornelissen** Measuring musics: Notes on modes, motifs, and melodies
- ILLC DS-2024-03: Nicola De Cao Entity Centric Neural Models for Natural Language Processing
- ILLC DS-2024-04: Ece Takmaz Visual and Linguistic Processes in Deep Neural Networks: A Cognitive Perspective

ILLC DS-2024-05: Fatemeh Seifan Coalgebraic fixpoint logic Expressivity and completeness result

- ILLC DS-2024-06: Jana Sotáková Isogenies and Cryptography
- ILLC DS-2024-07: Marco Degano Indefinites and their values

ILLC DS-2024-08: **Philip Verduyn Lunel** *Quantum Position Verification: Loss-tolerant Protocols and Fundamental Limits*

ILLC DS-2024-09: **Rene Allerstorfer** *Position-based Quantum Cryptography: From Theory towards Practice*

ILLC DS-2024-10: Willem Feijen Fast, Right, or Best? Algorithms for Practical Optimization Problems

ILLC DS-2024-11: Daira Pinto Prieto Combining Uncertain Evidence: Logic and Complexity

ILLC DS-2024-12: **Yanlin Chen** On Quantum Algorithms and Limitations for Convex Optimization and Lattice Problems

- ILLC DS-2024-13: Jaap Jumelet Finding Structure in Language Models
- ILLC DS-2025-01: Julian Chingoma On Proportionality in Complex Domains
- ILLC DS-2025-02: **Dmitry Grinko** *Mixed Schur-Weyl duality in quantum information*
- ILLC DS-2025-03: **Rochelle Choenni** Multilinguality and Multiculturalism: Towards more Effective and Inclusive Neural Language Models
- ILLC DS-2025-04: Aleksi Anttila Not Nothing: Nonemptiness in Team Semantics